

City of Meriden, Connecticut

Purchasing Department

Invitation to Bid

For

ARPA- Meriden Girls Inc. Renovations

Meriden, CT

B024-39

Bids Due:May 9, 2024 @ 11:00 AM

Purchasing Department 142 East Main Street, Room 210 Meriden, CT 06450 (203) 630-4115

LEGAL NOTICE

INVITATION TO BID

The City of Meriden is accepting sealed bids for:

B024-39 ARPA- Meriden Girls Inc. Renovations

The City of Meriden seeks the services of a qualified licensed and certified contractor to furnish labor and materials for interior and exterior renovations and modification of the Meriden Girls Inc in accordance with general conditions herein.

Bids shall be submitted on forms and in the manner specified. Forms and specifications may be obtained from the Purchasing Department, on the City of Meriden website (<u>www.meridenct.gov/business/bids-rfps/</u>), and on the State of Connecticut Department of Administrative Services website (<u>https://webprocure.proactiscloud.com</u>). Bids will be accepted in the Purchasing Department, 142 East Main Street-Room 210, Meriden, CT 06450 until **11:00 AM local, Eastern Standard Time on May 9, 2024**, at which time they will be publicly opened and read aloud. Any bids received after that date and time specified shall not be considered.

A NON-Mandatory Pre-Bid Meeting will be held on Wednesday, April 17, 2024 @ 10:00 AM at the Meriden Girls Inc, 130 Lincoln Street, Meriden, CT 06451.

The right is reserved to reject any or all bids, in whole or in part, to award any item, group of items, or total bid, and to waive informality or technical defects, if it is deemed to be in the best interest of the City of Meriden. No bidder may withdraw its bid within sixty (60) days of the date of the bid opening.

Each bid shall be accompanied by a Certified Check or Bid Bond in the amount of Ten (10%) percent of the amount bid.

Labor and Material Payment Bond and a Performance bond for One Hundred Percent (100%) of the contract price, with a corporate surety approved by the City of Meriden, will be required of the lowest responsible bidder.

This is an ARPA funded project and all contractors are required abide by all State and Federal Guidelines and requirement for minimum wage rates to be paid under this contract.

The City of Meriden is an Affirmative Action/Equal Opportunity Employer. Disadvantaged, minority, small, and women business enterprises are encouraged to respond.

Rawle Dummett Purchasing Officer City of Meriden, CT 06450-8022 March 28, 2024

CITY OF MERIDEN, CONNECTICUT

B024-39– ARPA- Meriden Girls Inc. Renovations

INFORMATION TO BIDDERS

1. **BIDDING PROCEDURES**

Sealed Bids shall be submitted on the forms designated by the attached proposal bid forms. Bids will be received by the City of Meriden's Purchasing Department, Room 210, City Hall, 142 East Main Street, Meriden, Connecticut, 06450-8022 until **11:00 a.m. on May 9, 2024** and thereafter immediately read in public (the "bid opening").

2. <u>BIDS</u>

Bids are to be submitted on the attached proposal forms. Please submit two copies of the proposal forms and Bidder's Qualification Statement. One shall be an original and one can be a copy. **Please submit one complete copy of your bid on a flash drive.**

BID WILL BE AUTOMATICALLY REJECTED FOR ANYONE SUBMITTING A SURETY OTHER THAN THOSE SPECIFIED.

- a. Bids must be made out and signed in the corporate, or other, name of Bidder, and must be fully and properly executed by an authorized person.
- b. The sealed envelope must denote the Bidder's name and address in the upper left hand corner and the words "BID DOCUMENT B024-39 ARPA- Meriden Girls Inc. Renovations to be opened at 11:00 a.m." in the lower left hand corner.
- c. Bids received later than the time and date specified will not be considered.
- d. Amendments to or withdrawal of bids received later than the date and time set forth in the bid opening will not be considered.
- e. All prices must be in ink or typewritten. In the event of a bidder's mathematical error in tabulating any bid prices, *the written unit prices shall govern*.

3. BIDDER QUALIFICATIONS

Bidders will be required to fill out, and include as part of its bid, any attached Bidder's Qualification Statement.

In determining the qualifications of a bidder, the City of Meriden will consider the bidder's record of performance in any prior contracts for construction work. The City of Meriden expressly reserves the right to reject a bid if the bidder's historical performance, in the sole opinion of the City of Meriden, has been unsatisfactory in any manner or if the bidder has habitually and without just cause neglected the payment of bills or has otherwise disregarded its obligations to subcontractors, suppliers, or employees.

4. EXAMINATION OF BIDDING DOCUMENTS

Bidders are to examine all documents and visit the site in order to make a thorough examination of the conditions so that the bidder may familiarize itself with all of the existing requirements, conditions, and difficulties that will affect the execution of the work in order to determine the amount of work necessary to carry out the true intent of the specifications and work shown on the drawings.

The City of Meriden and its agents do not have any responsibility for the accuracy, completeness, or sufficiency of any bid document obtained from any other source other than from the City of Meriden. Obtaining documents from any other source(s) may result in obtaining incomplete and inaccurate information. Obtaining documents from any other source may also result in failure to receive any addenda, corrections, or other revisions to the documents that may be issued.

No request shall be honored if such request is made less than seven (7) calendar days prior to the date fixed for the opening of bids. Any and all such interpretations, and any supplementary instructions, will be in the form of a written addenda to the specifications which, if issued, will be made available on the City of Meriden website (www.meridenct.gov) unless it is to change the date fixed for the opening of bids, not later than three (3) days prior to the date fixed for the opening of bids. Bidders are encouraged to check the website regularly for addenda. Failure of any bidder to receive any such addenda shall not relieve any bidder from any obligations under its bid as submitted.

Any questions about the bid document must be submitted in writing via email to <u>meridenpurchasing@meridenct.gov</u>. Any other format of question will not be answered.

5. <u>BIDS TO REMAIN OPEN</u>

No bidder may withdraw its bid within sixty (60) days of the date of the bid opening. Should there be reason why the contract cannot be awarded within the specified period, the time may be extended by mutual agreement between the City of Meriden and the successful bidder.

6. <u>AWARD OF CONTRACT</u>

The Purchasing Officer reserves the right to make an award on the bid which, by the Purchasing Officer's judgment and recommendation from the Purchasing Department following bid evaluations, best meets the specifications and is deemed to be in the best interest of the City of Meriden.

The contract will <u>not</u> be awarded to any corporation, firm, or individual which/who is in arrears to the City of Meriden by debt or contract, or who is in default as security or otherwise by any obligation to the City of Meriden.

The right is reserved to reject any or all bids, in whole or in part, to award any item, group of items, or total bid, and to waive informality or technical defects, if it is deemed to be in the best interest of the City of Meriden.

7. <u>BID PROTEST PROCEDURE</u>

In the event that any bidder wishes to protest the potential award of a bid, or any procedure of act in the advertising or soliciting of the bids, said bidder must make said protest in writing, which shall state the reason therefore and request a conference with respect thereto. Said protest must be received in the City Purchasing Office within **FIVE (5)** business days after the delivery of bid results or decisions. A conference with respect to said protest shall be scheduled by the Purchasing Officer forthwith and shall be attended by him or his designee and such other persons as the Purchasing Officer and the City Manager shall require to attend. The subject matter of said conference. Said conference shall also include a discussion of all possibilities for a resolution of dispute. The City shall make a decision in writing within three (3) business days after said conference and forward the same to the protesting bidder forthwith. In the event that any protesting bidder wishes to take legal action against the City, they must fully comply with all of these instructions to bidders.

8. <u>CITY OF MERIDEN, LOCAL PREFERENCE</u> – N/A

9. <u>EXTENSION OF AGREEMENT</u> – N/A

10. <u>TIME</u>

Inasmuch as the contract concerns a public improvement, the provisions of the contract relating to the time of performance and completion of the work are of the essence of the contract. Accordingly, the successful bidder/contractor ("Contractor") shall begin work on the day specified in paragraph 2.04 of the General Conditions and shall perform the work diligently so as to permit full use not later than the first day following the construction period established in the Contract. See paragraph 10 entitled "Liquidated Damages" of the Agreement between City of Meriden, as owner, and the Contractor.

11. SCHEDULE OF WORK

The Contractor shall schedule all work in a manner that will not disrupt City of Meriden operations. Once the work has begun, the Contractor shall work full-time until completion of the Contract.

12. <u>TAXES</u>

The City of Meriden is exempt under Connecticut General Statutes from the payment of the excise taxes imposed by the federal government and the Sales and Use Tax of the State of Connecticut; such taxes should not be included in the bid price. Upon request, exemption certificates will be furnished to the successful bidder.

13. FAIR EMPLOYMENT PRACTICES

The Contractor shall agree that neither it or its subcontractors, except in the case of a bona fide occupational qualification or need, to refuse to hire or employ or to bar or to discharge from employment any individual or to discriminate against such individual in compensation or in terms, conditions or privileges of employment because of the individual's race, color, religious creed, age, sex, gender identity or expression, marital status, national origin, ancestry, present or past history of mental disability, intellectual disability, learning disability, physical disability, including, but not limited to, blindness or status as a veteran. The aforementioned terms are obtained from Connecticut General Statutes Section 46a-60, *et seq.*, entitled "Discriminatory employment practices prohibited," as amended.

14. FORM OF AGREEMENT BETWEEN CITY OF MERIDEN AND CONTRACTOR

The Agreement for the work will be written on the Agreement between City of Meriden and Contractor, wherein the basis of payment is a stipulated sum.

15. LOCAL SUBCONTRACTORS, SUPPLIERS, etc.

Local subcontractors, material suppliers, and labor in the City of Meriden should be considered and sought out insofar as it is practical in the performance of this project.

16. CITY OF MERIDEN CODE OF ETHICS

The City of Meriden has adopted a Code of Ethics located in Chapter 21 of the Code of the City of Meriden, sections 21-1 through 21-15, inclusive, which are expressly incorporated herein by reference. The terms of the Code of Ethics shall constitute a part of any contract or agreement entered into by the City of Meriden as a result of this bid as if those terms were fully set forth in such contract or agreement.

Bidders are specifically advised that the Code of Ethics prohibits public officers and employees, as well as their immediate families and businesses, with which they are associated from participating in any transaction which is incompatible with the proper discharge of official duties or responsibilities. Bidders are also advised that the Code of Ethics contain provisions with respect to paid contractors and former employees and officials.

BIDDERS SHOULD NOTE THAT BIDS, CONTRACTS, AND AGREEMENTS ENTERED INTO OR AWARDED IN VIOLATION OF THE CODE OF ETHICS ARE VOIDABLE BY RESOLUTION OF THE CITY COUNCIL OF THE CITY OF MERIDEN.

Copies of the Code of Ethics may be obtained from the office of the City Clerk or may be found online on the City of Meriden's website.

17. NON-COLLUSION BID STATEMENT

Each bidder submitting a bid to the City of Meriden for any portion of the work contemplated by the documents on which bidding is based shall execute and attach thereto the sworn Non-Collusive Bid Statement, to the effect that the bidder has not colluded with any other person, firm, or corporation in the submission of the bid.

18. SOIL CONDITIONS

The City of Meriden does not guarantee the accuracy of any information which it may have obtained as to the kind or condition of the soil that may be encountered in the performance of the proposed work; neither does the City of Meriden represent that the plans and specifications drawn are based upon any soil data so obtained. The City of Meriden does not make any representations as to the soil data so obtained. The City of Meriden does not make any representations as to the soil conditions to be encountered or as to foundation materials.

19. AWARD IN CASE OF A TIE

In the event there are two or more responsive bidders, the decision to award will be based by the following criteria and in the following order:

- a. The incumbent will be awarded the bid over that of another bidder.
- b. In the case of a multi-item bid, if one bidder has been awarded other items from the same bid and the other bidder has not, the bidder with the multiple awards will be awarded the bid over that of another bidder.
- c. The bidder located in the State of Connecticut will be awarded the bid over that of another bidder.
- d. The winner of a coin toss will be awarded the bid over that of another bidder.

The above-referenced provisions do not apply to those situations in which more than one City-based business responsible bidder has submitted bids not more than ten (10) percent higher than the lowest bid and has agreed to accept the award of the bid at the amount of the lowest bid. Under such circumstances, the provisions of the Code of the City of Meriden, section 3-14, are controlling, as set forth under Section 8 of this 'Information to Bidders.'

20. ASSIGNMENT OF CONTRACT

No contract may be assigned without the written consent of the Purchasing Officer or designee.

21. PERMITS

The Contractor shall be responsible for obtaining any and all necessary permits required by the City of Meriden prior to the commencement of work. The Contractor may contact the City of Meriden Building Department for permit information at (203) 630-4091. For all other required permits, contact the City of Meriden Engineering Department at (203) 630-4018.

22. BID PRICE AND PAYMENT

The City of Meriden is exempt from the payment of the excise taxes imposed by the Federal government and the Sales and Use Tax of the State of Connecticut under Connecticut General Statutes; accordingly, such taxes shall not be included in the bid price.

The City of Meriden, unless stated otherwise in the bidding documents or Contract, will make payment to the Contractor not less than thirty (30) days following completion of services.

23. <u>QUALITY</u>

All materials, equipment, supplies, and services shall be subject to rigid inspection. If defective material, equipment, supplies, or services are discovered, the Contractor shall remove or make good such material, equipment, or supplies without extra compensation. It is expressly understood and agreed that any inspection by the City of Meriden will in no way lessen the responsibility of the Contractor or release Contractor from the obligation to perform and deliver to the City sound and satisfactory materials, equipment, supplies, or allow the cost to be deducted from any monies due it from the City of Meriden. All services will be performed in a workmanlike manner.

24. INSURANCE

The successful bidder shall be required to provide a Certificate of Insurance denoting general liability, automobile liability, workers compensation liability, and other coverage required by the City's Risk Manager.

25. <u>CITY HALL CLOSING</u>

If Meriden City Hall is closed due to inclement weather, or any other unforeseen event, bids will be due at the same time on the next business day that City Hall is open.

CITY OF MERIDEN, CONNECTICUT

B024-39 – ARPA- Meriden Girls Inc. Renovations

NON-COLLUSIVE BID STATEMENT/AFFIDAVIT

The undersigned bidder, having been duly sworn, does hereby depose and says:

- 1. The bid has been arrived at by the bidder independently and has been submitted without collusion and without any agreement, understanding, or planned common course of action with any other vendor of materials, supplies, equipment, or services described in the Invitation to Bid.
- 2. The contents of the bid have not been communicated by the bidder or its employees or agents to any person not an employee or agent of the bidder or its surety on any bond furnished with the bid, and will not be communicated to any such person prior to the official opening of the bid.
- 3. The undersigned bidder is duly authorized to bind the business entity identified below.

The undersigned bidder further certifies, under oath, that this statement is executed for the purposes of inducing the City of Meriden to consider the bid and make an award in accordance therewith.

Signature of Bidder

Print Legal Name of Bidder

Relationship to Business Entity Below

Business Entity Name, Address, Telephone Number, and Email Address

STATE OF CONNECTICUT

COUNTY OF

)) ss:)

Duly sworn and subscribed to before me this ____ day of _____, 2024.

Notary Public My Commission Expires: Commissioner of the Superior Court

BIDDER'S QUALIFICATION STATEMENT

This Statement of Bidder's Qualifications is to be submitted by the bidder at the time of the bid opening. All questions must be answered and the data given must be clear and comprehensive. If necessary, questions must be answered on attached sheets. The bidder may submit any additional information they desire. It is understood that when the City has executed an Agreement, to which these General Conditions are a part, it is, in part, done upon the reliance of the answers provided herein by the bidder or the agent of the bidder.

Firm Name			
Address			
Telephone		Fax	
Officers:		President Vice President Secretary Treasurer	
Bank References:			
Bond surety Compar	ny:		
If a partnership, give employee.	names of partners. If a s	ole proprietorship, give name and title	of a least one responsible
Experience: The Bic examples of similar preferences.	lder shall be qualified by projects completed within	experience to perform work of this nation the past five (5) years, with the name	ture and shall list five (5) s of responsible parties as
PROJECT	OWNER	TELEPHONE NUMBER CONTACT NAME	COST

- Minority owned business? ______ yes ______ no 1. Years organized. 2. Is your company a corporation ______ yes _____ no 3. If yes where incorporated? How many years have you been engaged in business under your present firm name? 4. 5. Former Firm Name (if any) List total number of Personnel 6. Is any principal of your firm an employee or public official of the City of Meriden, or an immediate 7. family member of an employee or public official of the City of Meriden? (Definition of immediate family includes: an individual's spouse, fiancé or fiancée; the parent, brother or sister of such individual or spouse; and the child of such individual or the spouse of such child.) _____yes _____no List Vehicles and Equipment that you will use to perform this work: (show age of vehicles and 8. equipment, sizes, capacities, etc. List the work to be performed by Subcontractors and summarize the dollar value of each subcontract. 9. 10. List the name and address of the more important contracts recently completed by you, starting the approximate gross cost for each, and the month and year completed:
- 11. General character of work performed by you_____
- 12. Have you ever failed to complete any contract awarded to you? If so, where and why?

13.	Have you ever	defaulted on a contract? If s	so where and wh	ny?
14.	Have you ever	filed bankruptcy:	Please explai	n:
15.	Will you, upon	request, furnish any inform	ation that may l	be required by the City of Meriden?
16.	The undersigne information req Bidder's Qualif	d hereby authorizes and rec uested by the City of Meric ications.	quest any person len, in verificati	a, firm or cooperation to furnish any on of the recitals comprising this Statement of
Dated	l this	day of	20	
2	day	month	,	year
				Name of Bidder
Stata	of			Title
Coun	01 tv of			
Coun				
Name	3	of	being duly	sworn deposes and says that they are
title		01	name of orga	nization
and th	hat the answers to	the forgoing question and a	all statement the	erein contained are true and correct
	Subcoribod and	sworn to before me		
this	Subscribed and	day of	20	
	day	month	year	
				Notary Public signature
My co	ommission expire	s		

Special Requirements

(A) <u>Compliance with Nondiscrimination and Affirmative Action in accordance</u> with Section 4a-60 of the Connecticut General Statutes.

(1) (a) The contractor agrees and warrants that in the performance of the contract such contractor will not discriminate or permit discrimination against any person or group of persons on the grounds of race, color, religious creed, age, marital status, nation origin, ancestry, sex, mental retardation, mental disability, or physical disability, including, but not limited to, blindness, unless it is shown by contractor that such disability prevents performance of the work involved, in any manner prohibited by the laws of the United States or of the state of Connecticut. The contractor further agrees to take affirmative action to unsure that applicants with job related qualifications are employed and that employees are treated when employed without regarded to their race, color, religious creed, age, marital status, national origin, ancestry, sex, mental retardation, mental disability, or physical disability, including, not limited to, blindness, unless it is shown by contractor that such disability prevents performance of the work involved; (b) the contractor agrees, in all solicitations or advertisements for employees placed by or on behalf of the contractor, to state that it is an "affirmative action-equal opportunity employer" in accordance with regulations adopted by the commission; (c) the contractor agrees to provide each labor union or representative of workers with which such contractor has a collective bargaining agreement or other contract or understand and each vendor with which such contractor has a contract or understand, a notice to be provided by the commission advising the labor union or workers' representative of the contractor's commitments under this section, and to post copies of the notice in conspicuous places available to employees and applicants for employment; (d) the contractor agrees to comply with each provision of this section and Sections 46a-68e and 46a-68f and with each regulation or relevant order issued by said commission pursuant to Sections 46a-56, 46a-68e and 46a-68f; (e) the contractor agrees to provide the commission on human rights and Opportunities with such information requested by the commission, and permit access to pertinent books, records, and accounts, concerning the employment practices and procedures of the contractor as relate to the provisions of this section and Section 46a-56.

(2) If the contract is a public works contract, the contractor agrees and warrants that it will make good faith efforts to employ minority business enterprises as subcontractors and suppliers of materials on such public works project.

(3) "Minority business enterprise" means any small contractor or supplier of materials fifty-one percent or more of the capital stock, if any, or assets of

which is owned by a person or persons: (a) who are active in the daily affairs of the enterprise, (b) who have the power to direct the management and policies of the enterprise and (c) who are members of a minority, as such term is defined in Section 32-9n (a) of the Connecticut General Statutes; and "good faith" means that degree of diligence which a reasonable person would exercise in the performance of legal duties and obligations. "Good faith efforts" shall include, but not be limited to, those reasonable initial efforts necessary to comply with statutory or regulatory requirements and additional or substituted efforts when it is determined that such initial efforts will not be sufficient to comply with such requirements.

(4) Determination of the contractor's good faith efforts shall include but shall not be limited to the following factors: the contractor's employment and subcontracting policies, patterns, and practices; affirmative advertising, recruitment, and training; technical assistance activities and such other reasonable activities or efforts as the commission may prescribe that are designed to ensure the participation of minority business enterprises in public works project.

(5) The contractor shall develop and maintain adequate documentation, in a manner prescribed by the commission, of its good faith efforts.

(6) The contractor shall include the provision of subsection (A)(1) and A(2) of this section in every subcontract or purchase order entered into in order to fulfill any obligation of this Agreement with the State and such provisions shall be binding on a subcontractor, vendor, or a manufacturer unless exempted by regulations or order of the commission. The contractor shall take such action with respect to any such subcontract or purchase order as the commission may direct as a mean of enforcing such provisions including sanctions or noncompliance in accordance with Section 46a-56 of the Connecticut General Statues; provided if such contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the commission, the contractor may request the State of Connecticut to enter into any such litigation or negotiation prior thereto to protect the interest of the State and the State may so enter

(B) Further Agreements re Compliance with Nondiscrimination.

(1) The contractor agrees and warrants that in the performance of the contract such contractor will not discriminate or permit discrimination against any person or group of persons on the grounds of sexual orientation, in any manner prohibited by the laws of the United States or of the State of Connecticut, and that employees are treated when employed without regard to their sexual orientation; the contract agrees to provide each labor union or representative of workers with which such contractor has a collective bargaining agreement or other contract or understanding and each vendor

with which such contractor has a contract or understanding, a notice to be provided by the Commission on Human Rights and Opportunities advising the labor union or workers' representative of the contractor's commitments under this section, and to post copies of the notice in conspicuous places available to the employees and applicants for employment; the contractor agrees to comply with each provision of this section and with each regulation or relevant order issued by said commission pursuant to Section 46a-56 of the Connecticut General Statutes; the contractor agrees to provide the Commission on Human Rights and Opportunities with such information requested by the commission and permit access to pertinent books, records and accounts, concerning the employment practices and procedures of the contractor which relate to the provisions of this section and Section 46a-56 of the Connecticut General Statutes.

(2) The contractor shall include the provision of subsection (B)(1) in every subcontract or purchase order entered into in order to fulfill any obligation of contract with the State such provisions shall be binding on a subcontractor, vendor or manufacturer unless exempted by regulations or orders of the commission. The contractor shall take such action with respect to any such subcontract or purchase order as the commission may direct as a means of enforcing such provision including sanctions for noncompliance in accordance with Section 46a-56 of the Connecticut General Statutes; provided, if such contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the commission, the contract may request the State of Connecticut to enter into any such litigation or negotiation prior thereto to protect the interest of the State and the State may so enter.

For the purposes of sections (A) and (B) of this section 2.10, "contract" (3) means the Agreement and includes any extension or modification of the contract, "contractor" includes any successors or assigns of the contractor, "marital status" means being single, married as recognized by the state of Connecticut, widowed, separated or divorce, and "mental disability" means one or more mental disorders as defined in the most recent edition of the American Psychiatric Association's "Diagnostic and Statistical Manual of Mental Disorders", or a record of or regarding a person as having one or more disorders. Further, "contract" does not include a contract where each contractor is: (i) a political subdivision of the state, including, but not limited to, a municipality; (ii) a quasi-public agency, as defined in Section 1-120 of the Connecticut General Statutes; (iii) any other state, including but not limited to any federally recognized Indian tribal governments, as defined in Section 1-267 of the Connecticut General Statutes; (iv) the federal government; (v) a foreign government; or (vi) an agency of subdivision, agency, state or government described in the immediately preceding enumerated items (i), (ii), (iii), (iv), or (v).

Executive Order No. Three. This Agreement is subject to the provisions (C) of Executive Order No. Three of Governor Thomas J. Meskill promulgated June 16, 1971 and, as such, this Agreement may be cancelled, terminated or suspended by the State Labor Commissioner for violation or of noncompliance with said Executive Order No. Three, or any State or Federal Law concerning nondiscrimination, notwithstanding that the Labor Commissioner is not a party to this Agreement. The parties to this Agreement, as part of the consideration hereof, agree that said Executive Order No. Three is incorporated herein by reference and made a part hereof. The parties agree to abide by said Executive Order and agree that the State Labor Commissioner shall have continuing jurisdiction in respect to Agreement performance in regard to nondiscrimination, until the Agreement is completed or terminated prior to completion. The Applicant agrees as part consideration hereof, that this contract is subject to the guidelines and rules issued by the State Labor Commissioner to implement policies, will file all reports as required, and will fully cooperate with the State and the State Labor Commissioner.

(D) <u>Executive Order No. Sixteen.</u> This Agreement is subject to, and Applicant hereby agrees to abide by Executive Order No. Sixteen of Governor John G. Rowland promulgated August 4, 1999, and, as such, the Agreement may be cancelled, terminated or suspended by the State for violation or noncompliance with said Executive Order No. Sixteen.

(E) <u>Executive Order No. Seventeen.</u> This Agreement is subject to the provisions of Executive Order No. Seventeen of Governor Thomas J. Meskill promulgated February 15, 1973, as, as such, this Agreement may be cancelled, terminated or suspended by the Commissioner or the State Labor Commissioner for violation of or noncompliance with said Executive Order No. Seventeen, notwithstanding that the Labor Commissioner may not be a party to this Agreement. The parties to this Agreement, as part of the consideration hereof, agree that the Executive Order No. Seventeen is incorporated herein by reference and made a part hereof. The parties agree to abide by said Executive Order and agree that the contracting agency and the State Labor Commissioner shall have joint and several continuing jurisdiction in respect to Agreement performance in regard to listing all employment opening with the Connecticut Employment Service.

(F) <u>Environmental Laws.</u> The Applicant hereby agrees to indemnify and hold harmless the State from and against any liabilities, losses, damages, costs, or expenses, including attorneys' fees, arising out of or in connection with the presence of hazardous waste on or in any of the Site or any lien or claim under Section 22a-452a of the Connecticut General Statutes, as amended, or other federal, state, or municipal statue, regulation, rule, law, or proceeding relating to environmental matters, which indemnity shall survive the termination and/or release of the Project Documents.

(G) <u>Campaign Contribution and Solicitation Prohibitions.</u> For all State contracts as defined in P.A. 07-1 having a value in a calendar year of \$50,000.00 or more or a combination or series of such agreements or contracts having a value of \$100,000.00 or more, the Applicant expressly acknowledges receipt of State Election Enforcement Commission's notice advising state contractors of state campaign contribution and solicitation prohibitions, and will cause the Applicant to inform its principals of the contents of the notice. This notice SEEC II is attached hereto as Exhibit C and made a part thereof.

6.1. <u>Insurance and Contractor Bonding.</u> All contractors engaged by the Applicants in connection with the Project shall maintain such insurance as the State deems appropriate. All such policies of insurance shall name the state as an additional insured or loss payee, as applicable. Contractors engaged by the Applicant for the Project shall also satisfy such bonding requirements as may be imposed by the Commissioner in his/her discretion.

PERFORMANCE BOND AND LABOR AND MATERIAL PAYMENT BOND

The Contractor shall, within ten (10) days from the date of the Notice of Award, furnish the City of Meriden with a <u>PERFORMANCE BOND and a LABOR AND MATERIAL PAYMENT BOND</u>, both in the amount of 100% of the amount bid, conditioned upon the performance of the Contractor on all undertaking, covenants, terms, and conditions and agreements of the contract. The bond shall be in the form of the specimen bonds annexed hereto, such bonds shall be executed by the contractor and a corporate bonding company licensed, authorized, and admitted to transact such business in the State of Connecticut and named on the current list of "Surety Companies acceptable on Federal Bonds", as published in the "Treasury Department" listed for an amount equal to the amount of the reinsurance. Written evidence of how any excess suretyship has been placed by the surety signing the bonds shall accompany the bonds. The expense of the bonds shall be borne by the Contractor. If at anytime a surety on any such bond is declared bankrupt or loses its right to do business in the State of Connecticut, or is removed from the list of Surety Companies acceptable on Federal Bonds, or for any other justifiable cause, the Contractor shall, within ten (10) days after notice from the City of Meriden to do so. substitute an acceptable bond(s) in such form and sum and signed by such other surety or sureties as may be paid by the Contractor. No payments shall be deemed due nor shall be made until the new surety or sureties have furnished an acceptable bond to the City.

If the Contractor is a partnership, the bonds shall be signed by each of the individuals who are partners; if a corporation, the bonds shall be signed in the correct corporation name by a duly authorized office, agent, or attorney-in-fact. There should be executed an appropriate number of counterparts of the bond corresponding to the number of counterparts of the contract. Each executed bond shall be accompanied by 1) appropriate acknowledgements of the respective parties; 2) appropriate duly certified copy of Power of Attorney or other certificate of authority where bond is executed by agent, officer, or other representative of Contractor or surety; 3) a duly certified extract from by-laws or resolutions or surety under which Power of Attorney or other certificates of authority of its agent, officer, or representative was issued.

The Contractor hereby agrees and understands that a Notice of Award is expressly conditional upon the receipt of these bonds and a Certificate of Insurance naming the City of Meriden (and others as appropriate) as <u>ADDITIONAL INSURED</u>. If said documents are not received by the City of Meriden within ten (10) days from the date of Notice of Award, the City of Meriden reserves the right to withdraw its conditional acceptance of the bid and cancel the Notice of Award.

PERFORMANCE BOND

KNOW ALL MEN BY THESE PRESENTS, that

(here insert full name and address or legal title of Contractor)

as Principal herinafter called contractor and

(here insert full name and address or legal title of Surety

As Surety, hereinafter called Surety, are held and firmly bound unto

(here insert full name and address or legal title of Owner)

As Obligee, hereinafter called Owner, in the amount of

for the payment whereof Contractor and Surety bind themselves, their heirs, executors, administrators, successors and assigns, jointly and severally, firmly by these presents.

WHEREAS,

Contractor has by written agreement dated (here insert full name, address and description of project)

In accordance with Drawings and Specifications prepared by

(here insert full name and address or legal title of Engineer/Architect)

Which contract is by reference made a part hereof, and is hereinafter referred to as the Contract.

\$

20 , entered into a contract with Owner for

Dollars

PERFORMANCE BOND

NOW, THEREFORE, THE CONDITION OF THIS OBLIGATION is such that, if Contractor, shall promptly and faithfully perform said Contract, then this obligation shall be null and void; otherwise it shall remain in full force and effect.

The Surety hereby waives, notice of any alteration or extension of time made by the Owner.

Whenever Contractor shall be, and declared by Owner to be in default under the Contract, the Owner having performed Owner's obligations thereunder, the surety may promptly remedy the default, or shall promptly

1) Complete the Contract in accordance with its terms and conditions, or

2) Obtain a bid or bids for completing the Contract in accordance with its terms and conditions, and upon determination by Surety of the lowest responsible bidder, or, if the Owner elects, upon determination by the Owner and the Surety jointly of the lowest responsible bidder, arrange for a contract between such bidder and Owner, and make available as Work progresses (even though there should be a default of a succession of

defaults, under the contract or contracts of completion arranged under this paragraph sufficient funds to pay the cost of completion less the balance of the contract price; but not exceeding, including other costs and damages for which the Surety may be liable hereunder, the amount set forth in the first paragraph hereof. The term "balance of the contract price," as used in this paragraph, shall mean the total amount payable by Owner to Contractor under the Contract and any amendments thereto, less the amount properly paid by Owner to Contractor.

Any suit under this bond must be instituted before the expiration of two (2) years from the date on which final payment under the Contract falls due.

No right of action shall accrue on this bond to or for the use of any person or corporation other than the Owner named herein or the heirs, executors, administrators or successors of the Owner.

Signed and sealed this	day of		20
	_	(Principal)	
(Witness)			
		(Title)	
		(Surety)	
(Witness)	-		
		(Title)	

LABOR AND MATERIAL PAYMENT BOND

KNOW ALL MEN BY THESE PRESENTS, that

(here insert full name and address or legal title of Contractor)

as Principal, herinafter called Principal, and

(here insert full name and address or legal title of Surety

As Surety, hereinafter called Surety, are held and firmly bound unto

(here insert full name and address or legal title of Owner)

As Obligee, hereinafter called Owner, for the use and benefit of claimants as hereinbelow defined, in the amount of Dollars \$_____

For the payment whereof Principal and Surety bind themselves, their heirs, executors, administrators, successors and assigns, jointly and severally, firmly by these presents.

WHEREAS,

Principal has by written agreement dated (here insert full name, address and description of project)

20 , entered into a contract with Owner for

In accordance with Drawings and Specifications prepared by

(here insert full name and address or legal title of Engineer/Architect) which contract is by reference made a part hereof, and is hereinafter referred to as the Contract.

LABOR AND MATERIAL PAYMENT BOND

NOW, THEREFORE, THE CONDITION OF THIS OBLIGATION is such that, if Principal shall promptly make payment to all claimants as hereinafter defined, for all labor and material used or reasonably required for use in the performance of the Contract, then this obligation shall be void; otherwise it shall remain in full force and effect, subject, however, to the following conditions:

1. A claimant is defined as one having a direct contract with the Principal or with a Subcontractor of the Principal for labor, material, or both, used or reasonably required for use in the performance of the Contract, labor and material being construed to include that part of water, gas, power, light, heat, oil, gasoline, telephone service or rental of equipment directly applicable to the Contract.

2. The above named Principal and Surety hereby jointly and severally agree with the Owner that every claimant as herein defined, who has not been paid in full before the expiration of a period of ninety (90) days after the date on which the last of such claimant's work or labor was done or performed or materials were furnished by such claimant, may sue on this bond for the use of such claimant, prosecute the suit to final judgment for such sum or sums as may be justly due claimant, and have execution thereon. The Owner shall not be liable for the payment of any costs or expenses of any such suit.

3. No suit or action shall be commenced hereunder by any claimant:

a) Unless claimant, other than one having a direct contact with the Principal, shall have given written notice to any two of the following: the Principal, the Owner, or the Surety above named, within ninety (90) days after such claimant did or performed the last of the work or labor, or furnished the last of the materials for which said claim is made, stating with substantial accuracy the amount claimed and the name of the party to whom the materials were furnished, or for whom the work or labor was done or performed. Such notice shall be served by mailing the same by registered mail or certified mail, postage prepaid, in an envelop addressed to the Principal Owner or Surety, at any place where an office is regularly maintained for the transaction of business, or served in any manner in which legal process may be served in the state in which the aforesaid project is located, save that such service need not be made by a public officer.

b) After the expiration of one (1) year following the date on which Principal ceased Work on said Contract, it being understood, however, that if any limitation embodied in this bond is prohibited by any law controlling the construction hereof such limitation shall be deemed to be amended so as to be equal to the minimum period of limitation permitted by such law.

c) Other than in a state court of competent jurisdiction in and for the county or other political subdivision of the state in which the Project or any part thereof, is situated, or in the United States District Court for the district in which the Project, or any part thereof is situated, and not elsewhere.

4. The amount of this bond shall be reduced by and to the extent of any payment or payments made in good faith hereunder, inclusive of the payment by Surety of mechanics' liens which may be filed of record against said improvement, whether or not claim for the amount of such lien be presented under and against this bond.

Signed and sealed this

day of

20

(Principal)

(Title)

(Surety)

(Witness)

(Witness)

(Title)

BID FORM

B024-39 ARPA- Meriden Girls Inc. Renovations

		Date of Opening: May 9, 2024 At 11 AM
To: Rawle Dummett Purchasing Officer 142 East Main Street, Roo Meriden, CT 06450	m 210	
The undersigned		, doing business in the City/Town of
, in to other Bid documents (including	he State of, l if any addendum or a	herewith, after reading thoroughly the Specifications and ddenda) submit the following proposal:
Lump sum price:		
Written Amount		Dollars & Cents
Receipt of Addenda is acknowle	edged:	
No: D	ated:	
No: D	ated:	
NAME OF BIDDER		
ADDRESS		
BY:		
Print or type nan	ne	Title
SIGNATURE		DATE
TELEPHONE	E-Mail	

PLEASE NOTE: All spaces must be filled in with figures or words or your bid may be automatically rejected.



THIS IS A PUBLIC WORKS PROJECT

Covered by the

PREVAILING WAGE LAW

CT General Statutes Section 31-53

If you have QUESTIONS regarding your wages CALL (860) 263-6790

Section 31-55 of the CT State Statutes requires every contractor or subcontractor performing work for the state to post in a prominent place the prevailing wages as determined by the Labor Commissioner.

Sec. 31-53b. Construction safety and health course. New miner training program. Proof of completion required for mechanics, laborers and workers on public works projects. Enforcement. Regulations. Exceptions. (a) Each contract for a public works project entered into on or after July 1, 2009, by the state or any of its agents, or by any political subdivision of the state or any of its agents, described in subsection (g) of section 31-53, shall contain a provision requiring that each contractor furnish proof with the weekly certified payroll form for the first week each employee begins work on such project that any person performing the work of a mechanic, laborer or worker pursuant to the classifications of labor under section 31-53 on such public works project, pursuant to such contract, has completed a course of at least ten hours in duration in construction safety and health approved by the federal Occupational Safety and Health Administration or, has completed a new miner training program approved by the Federal Mine Safety and Health Administration in accordance with 30 CFR 48 or, in the case of telecommunications employees, has completed at least ten hours of training in accordance with 29 CFR 1910.268.

(b) Any person required to complete a course or program under subsection (a) of this section who has not completed the course or program shall be subject to removal from the worksite if the person does not provide documentation of having completed such course or program by the fifteenth day after the date the person is found to be in noncompliance. The Labor Commissioner or said commissioner's designee shall enforce this section.

(c) Not later than January 1, 2009, the Labor Commissioner shall adopt regulations, in accordance with the provisions of chapter 54, to implement the provisions of subsections (a) and (b) of this section. Such regulations shall require that the ten-hour construction safety and health courses required under subsection (a) of this section be conducted in accordance with federal Occupational Safety and Health Administration Training Institute standards, or in accordance with Federal Mine Safety and Health Administration Standards or in accordance with 29 CFR 1910.268, as appropriate. The Labor Commissioner shall accept as sufficient proof of compliance with the provisions of subsection (a) or (b) of this section a student course completion card issued by the federal Occupational Safety and Health Administration Training Institute, or such other proof of compliance said commissioner deems appropriate, dated no earlier than five years before the commencement date of such public works project.

(d) This section shall not apply to employees of public service companies, as defined in section 16-1, or drivers of commercial motor vehicles driving the vehicle on the public works project and delivering or picking up cargo from public works projects provided they perform no labor relating to the project other than the loading and unloading of their cargo.

(P.A. 06-175, S. 1; P.A. 08-83, S. 1.)

History: P.A. 08-83 amended Subsec. (a) by making provisions applicable to public works project contracts entered into on or after July 1, 2009, replacing provision re total cost of work with reference to Sec. 31-53(g), requiring proof in certified payroll form that new mechanic, laborer or worker has completed a 10-hour or more construction safety course and adding provision re new miner training program, amended Subsec. (b) by substituting "person" for "employee" and adding "or program", amended Subsec. (c) by adding "or in accordance with Federal Mine

Safety and Health Administration Standards" and setting new deadline of January 1, 2009, deleted former Subsec. (d) re "public building", added new Subsec. (d) re exemptions for public service company employees and delivery drivers who perform no labor other than delivery and made conforming and technical changes, effective January 1, 2009.

- SPECIAL NOTICE -

To: All State and Political Subdivisions, Their Agents, and Contractors

Connecticut General Statute 31-55a - Annual adjustments to wage rates by contractors doing state work.

Each contractor that is awarded a contract on or after October 1, 2002, for (1) the construction of a state highway or bridge that falls under the provisions of section 31-54 of the general statutes, or (2) the construction, remodeling, refinishing, refurbishing, rehabilitation, alteration or repair of any public works project that falls under the provisions of section 31-53 of the general statutes shall contact the Labor Commissioner on or before July first of each year, for the duration of such contract, to ascertain the prevailing rate of wages on an hourly basis and the amount of payment or contributions paid or payable on behalf of each mechanic, laborer or worker employed upon the work contracted to be done, and shall make any necessary adjustments to such prevailing rate of wages and such payment or contributions paid or payable on behalf of each such employee, effective each July first.

- The prevailing wage rates applicable to any contract or subcontract awarded on or after October 1, 2002 are subject to annual adjustments each July 1st for the duration of any project which was originally advertised for bids on or after October 1, 2002.
- Each contractor affected by the above requirement shall pay the annual adjusted prevailing wage rate that is in effect each July 1st, as posted by the Department of Labor.
- It is the *contractor's* responsibility to obtain the annual adjusted prevailing wage rate increases directly from the Department of Labor's Web Site. The annual adjustments will be posted on the Department of Labor Web page: <u>www.ctdol.state.ct.us</u>. For those without internet access, please contact the division listed below.
- The Department of Labor will continue to issue the initial prevailing wage rate schedule to the Contracting Agency for the project. All subsequent annual adjustments will be posted on our Web Site for contractor access.

Any questions should be directed to the Contract Compliance Unit, Wage and Workplace Standards Division, Connecticut Department of Labor, 200 Folly Brook Blvd., Wethersfield, CT 06109 at (860)263-6790.

Minimum Rates and Classifications for Building Construction

ID#:	24-59957	Connecticut Department of Labor
		Wage and Workplace Standards

By virtue of the authority vested in the Labor Commissioner under provisions of Section 31-53 of the General Statutes of Connecticut, as amended, the following are declared to be the prevailing rates and welfare payments and will apply only where the contract is advertised for bid within 20 days of the date on which the rates are established. Any contractor or subcontractor not obligated by agreement to pay to the welfare and pension fund shall pay this amount to each employee as part of his/her hourly wages.

Project Number:	B024-39 ARPA	Project Town:	Meriden
State#:		FAP#:	
D · · · · · · ·			

Project: Girls Inc. Renovations

CLASSIFICATION	Hourly Rate	Benefits
1b) Asbestos/Toxic Waste Removal Laborers: Asbestos removal and encapsulation (except its removal from mechanical systems which are not to be scrapped), toxic waste removers, blasters.**See Laborers Group 7**		
1c) Asbestos Worker/Heat and Frost Insulator	45.56	32.65
2) Boilermaker	46.21	29.35
3a) Bricklayer, Cement Mason, Concrete Finisher (including caulking), Stone Masons	41.11	34.65 + a
3b) Tile Setter	37.1	30.52
3c) Tile and Stone Finishers	30.0	25.30
3d) Marble & Terrazzo Finishers	31.07	24.23
3e) Plasterer	42.77	29.63

4) Group 1: General laborers, carpenter tenders, concrete specialists, wrecking laborers and fire watchers.	33.5	25.59
4) Group 1a: Acetylene Burners (Hours worked with a torch)	34.5	25.59
4a) Group 2: Mortar mixers, plaster tender, power buggy operators, powdermen, fireproofer/mixer/nozzleman (Person running mixer and spraying fireproof only).	33.75	25.59
4b) Group 3: Jackhammer operators/pavement breaker, mason tender (brick), mason tender (cement/concrete), forklift operators and forklift operators (masonry).	34.0	25.59
4c) **Group 4: Pipelayers (Installation of water, storm drainage or sewage lines outside of the building line with P6, P7 license) (the pipelayer rate shall apply only to one or two employees of the total crew who primary task is to actually perform the mating of pipe sections) P6 and P7 rate is \$26.80.	34.5	25.59
4d) Group 5: Air track operator, sand blaster and hydraulic drills.	34.25	25.59
4e) Group 6: Blasters, nuclear and toxic waste removal.	36.5	25.59
4f) Group 7: Asbestos/lead removal and encapsulation (except it's removal from mechanical systems which are not to be scrapped).	36.5	25.59
4g) Group 8: Bottom men on open air caisson, cylindrical work and boring crew.	31.78	25.59
4h) Group 9: Top men on open air caisson, cylindrical work and boring crew.	31.24	25.59
4i) Group 10: Traffic Control Signalman	20.1	25.59

4j) Group 11: Toxic Waste Removers A or B With PPE	36.5	25.59
5) Carpenter, Acoustical Ceiling Installation, Soft Floor/Carpet Laying, Metal Stud Installation, Form Work and Scaffold Building, Drywall Hanging, Modular-Furniture Systems Installers, Lathers, Piledrivers, Resilient Floor Layers.	37.61	27.61
5a) Millwrights	40.56	28.87
6) Electrical Worker (including low voltage wiring) (Trade License required: E1,2 L-5,6 C-5,6 T-1,2 L-1,2 V-1,2,7,8,9)	42.6	33.21+3% of gross wage
7a) Elevator Mechanic (Trade License required: R-1,2,5,6)	64.01	39.19+a+b
LINE CONSTRUCTION		
Groundman	26.5	6.5% + 9.00
Linemen/Cable Splicer	48.19	6.5% + 22.00
8) Glazier (Trade License required: FG-1,2)	41.18	24.55 + a
9) Ironworker, Ornamental, Reinforcing, Structural, and Precast Concrete Erection	42.37	40.02 + a
OPERATORS		
Group 1: Crane Handling or Erecting Structural Steel or Stone; Hoisting Engineer (2 drums or over). (Trade License Required)	52.78	27.80 + a
Group 1a: Front End Loader (7 cubic yards or over); Work Boat 26 ft. and Over	48.37	27.80 + a

Group 2: Cranes (100 ton rate capacity and over); Bauer Drill/Caisson. (Trade License Required)	52.41	27.80 + a
Group 2a: Cranes (under 100 ton rated capacity).	51.51	27.80 + a
Group 2b: Excavator over 2 cubic yards; Pile Driver (\$3.00 premium when operator controls hammer)	48.0	27.80 + a
Group 3: Excavator; Gradall; Master Mechanic; Hoisting Engineer (all types of equipment where a drum and cable are used to hoist or drag material regardless of motive power of operation), Rubber Tire Excavator (Drott- 1085 or similar);Grader Operator; Bulldozer Finegrade. (slopes, shaping, laser or GPS, etc.). (Trade License Required)	47.1	27.80 + a
Group 4: Trenching Machines; Lighter Derrick; CMI Machine or Similar; Koehring Loader (Skooper); Goldhofer.	46.64	27.80 + a
Group 5: Specialty Railroad Equipment; Asphalt Spreader, Asphalt Reclaiming Machine; Line Grinder; Concrete Pumps; Drills with Self Contained Power Units; Boring Machine; Post Hole Digger; Auger; Pounder; Well Digger; Milling Machine (over 24 mandrel).	45.92	27.80 + a
Group 5 continued: Side Boom; Combination Hoe and Loader; Directional Driller.	45.92	27.80 + a
Group 6: Front End Loader (3 up to 7 cubic yards); Bulldozer (rough grade dozer).	45.55	27.80 + a
Group 7: Asphalt Roller; Concrete Saws and Cutters (ride on types); Vermeer Concrete Cutter; Stump Grinder; Scraper; Snooper; Skidder; Milling Machine (24" and under mandrel).	45.14	27.80 + a
Group 8: Mechanic; Grease Truck Operator; Hydroblaster; Barrier Mover; Power Stone Spreader; Welding; Work Boat Under 26 ft.; Transfer Machine; Rigger Foreman.	44.67	27.80 + a
Group 9: Front End Loader (under 3 cubic yards); Skid Steer Loader regardless of attachments; (Bobcat or Similar); Forklift, Power Chipper; Landscape Equipment (including Hydroseeder); Vacuum Excavation	44.14	27.80 + a

Truck and Hydrovac Excavation Truck (27 HG pressure or greater).

Group 10: Vibratory hammer; ice machine; diesel and air, hammer, etc.	41.69	27.80 + a
Group 11: Conveyor, earth roller, power pavement breaker (whiphammer), robot demolition equipment.	41.69	27.80 + a
Group 12: Wellpoint Operator.	41.61	27.80 + a
Group 13: Compressor Battery Operator.	40.92	27.80 + a
Group 14: Elevator Operator; Tow Motor Operator (solid tire no rough terrain).	39.54	27.80 + a
Group 15: Generator Operator; Compressor Operator; Pump Operator; Welding Machine Operator; Heater Operator.	39.06	27.80 + a
Group 16: Maintenance Engineer.	38.28	27.80 + a
Group 17: Portable Asphalt Plant Operator; Portable Crusher Plant Operator; Portable Concrete Plant Operator; Portable Grout Plant Operator; Portable Water Filtration Plant Operator.	43.46	27.80 + a
Group 18: Power Safety Boat; Vacuum Truck; Zim Mixer; Sweeper; (Minimum for any job requiring a CDL license); Rigger; Signalman.	40.54	27.80 + a
PAINTERS (Including Drywall Finishing)		
10a) Brush and Roller	37.62	24.55
10b) Taping Only/Drywall Finishing	38.37	24.55

10c) Paperhanger and Red Label	38.12	24.55
10e) Blast and Spray	40.62	24.55
11) Plumber (excluding HVAC pipe installation) (Trade License required: P-1,2,6,7,8,9 J-1,2,3,4 SP-1,2)	48.28	35.50
12) Well Digger, Pile Testing Machine	37.26	24.05 + a
13) Roofer (composition)	41.8	22.65
14) Roofer (slate & tile)	42.3	22.65
15) Sheetmetal Worker (Trade License required for HVAC and Ductwork: SM-1,SM-2,SM-3,SM-4,SM-5,SM-6)	41.89	43.22
16) Pipefitter (Including HVAC work) (Trade License required: S-1,2,3,4,5,6,7,8 B-1,2,3,4 D-1,2,3,4, G-1, G-2, G-8 & G-9)	48.28	35.50
TRUCK DRIVERS		
17a) 2 Axle, Helpers	32.16	30.51 + a
17b) 3 Axle, 2 Axle Ready Mix	32.27	30.51 + a
17c) 3 Axle Ready Mix	32.33	30.51 + a
17d) 4 Axle	32.39	30.51 + a
17e) 4 Axle Ready Mix	32.44	30.51 + a

17f) Heavy Duty Trailer (40 Tons and Over)	34.66	30.51 + a
17g) Specialized Earth Moving Equipment (Other Than Conventional Type on-the-Road Trucks and Semi-Trailers, Including Euclids)	32.44	30.51 + a
17h) Heavy Duty Trailer up to 40 tons	33.39	30.51 + a
17i) Snorkle Truck	32.54	30.51 + a
18) Sprinkler Fitter (Trade License required: F-1,2,3,4)	47.55	32.27 + a
19) Theatrical Stage Journeyman	25.76	7.34

Welders: Rate for craft to which welding is incidental.

*Note: Hazardous waste removal work receives additional \$1.25 per hour for truck drivers. **Note: Hazardous waste premium \$3.00 per hour over classified rate

Crane with 150 ft. boom (including jib) - \$1.50 extra
Crane with 200 ft. boom (including jib) - \$2.50 extra
Crane with 250 ft. boom (including jib) - \$5.00 extra
Crane with 300 ft. boom (including jib) - \$7.00 extra
Crane with 400 ft. boom (including jib) - \$10.00 extra

All classifications that indicate a percentage of the fringe benefits must be calculated at the percentage rate times the "base hourly rate".

Apprentices duly registered under the Commissioner of Labor's regulations on "Work Training Standards for Apprenticeship and Training Programs" Section 31-51-d-1 to 12, are allowed to be paid the appropriate percentage of the prevailing journeymen hourly base and the full fringe benefit rate, providing the work site ratio shall not be less than one full-time journeyperson instructing and supervising the work of each apprentice in a specific trade.

The Prevailing wage rates applicable to this project are subject to annual adjustments each July 1st for the duration of the project.

Each contractor shall pay the annual adjusted prevailing wage rate that is in effect each July 1st, as posted by the Department of Labor.

It is the contractor's responsibility to obtain the annual adjusted prevailing wage rate increases directly from the Department of Labor's website.

The annual adjustments will be posted on the Department of Labor's Web page:

www.ct.gov/dol. For those without internet access, please contact the division listed below.

The Department of Labor will continue to issue the initial prevailing wage rate schedule to the Contracting Agency for the project.

All subsequent annual adjustments will be posted on our Web Site for contractor access.

Contracting Agencies are under no obligation pursuant to State labor law to pay any increase due to the annual adjustment provision.

Effective October 1, 2005 - Public Act 05-50: any person performing the work of any mechanic, laborer, or worker shall be paid prevailing wage

All Person who perform work ON SITE must be paid prevailing wage for the appropriate mechanic, laborer, or worker classification.

All certified payrolls must list the hours worked and wages paid to All Persons who perform work ON SITE regardless of their ownership i.e.: (Owners, Corporate Officers, LLC Members, Independent Contractors, et. al)

Reporting and payment of wages is required regardless of any contractual relationship alleged to exist between the contractor and such person.

~~Unlisted classifications needed for work not included within the scope of the classifications listed may be added after award only as provided in the labor standards contract clause (29 CFR 5.5 (a) (1) (ii)).

Please direct any questions which you may have pertaining to classification of work and payment of prevailing wages to the Wage and Workplace Standards Division, telephone (860)263-6790.

As of: March 27, 2024

~NOTICE~

TO ALL CONTRACTING AGENCIES

Please be advised that Connecticut General Statutes Section 31-53, requires the contracting agency to certify to the Department of Labor, the total dollar amount of work to be done in connection with such public works project, regardless of whether such project consists of one or more contracts.

Please find the attached "Contracting Agency Certification Form" to be completed and returned to the Department of Labor, Wage and Workplace Standards Division, Public Contract Compliance Unit.

[∞] Inquiries can be directed to (860)263-6543.


CONNECTICUT DEPARTMENT OF LABOR WAGE AND WORKPLACE STANDARDS DIVISION CONTRACT COMPLIANCE UNIT

CONTRACTING AGENCY CERTIFICATION FORM

I,, acting in my offic	cial capacity as,
authorized representative	title
for, located at	,
contracting agency	address
do hereby certify that the total dollar amount of wor	rk to be done in connection with
, locate	ed at
project name and number	address
shall be <u>\$</u> , which includes all w	ork, regardless of whether such project
consists of one or more contracts.	
CONTRACTOR IN	IFORMATION
Name:	
Address:	
Authorized Representative:	
Approximate Starting Date:	_
Approximate Completion Date:	_
Signature	Date
Return To: Connecticut Department of Labor Wage & Workplace Standards Divis	ion

Contract Compliance Unit 200 Folly Brook Blvd. Wethersfield, CT 06109

Date Issued: _____

CONNECTICUT DEPARTMENT OF LABOR WAGE AND WORKPLACE STANDARDS DIVISION

CONTRACTORS WAGE CERTIFICATION FORM Construction Manager at Risk/General Contractor/Prime Contractor

I,		of
Officer, Owner, Auth	orized Rep.	Company Name
do hereby certify that the _		
		Company Name
-		Street
-		City
and all of its subcontractors	s will pay all work	kers on the
	Project Name an	nd Number
	Street and City	y
the wages as listed in the so attached hereto).	chedule of prevaili	ing rates required for such project (a copy of which is
		Signed
Subscribed and sworn to be	efore me this	day of
		Notary Public
Return to:	_	
Connecticu Wage & W 200 Folly B Wethersfiel	t Department of L orkplace Standard brook Blvd.	Labor ds Division
Rate Schedule Issued (De	ate):	

Informational Bulletin

THE 10-HOUR OSHA CONSTRUCTION SAFETY AND HEALTH COURSE

(applicable to public building contracts entered into *on or after July 1, 2007*, where the total cost of all work to be performed is at least \$100,000)

- (1) This requirement was created by Public Act No. 06-175, which is codified in Section 31-53b of the Connecticut General Statutes (pertaining to the prevailing wage statutes);
- (2) The course is required for public building construction contracts (projects funded in whole or in part by the state or any political subdivision of the state) entered into on or after July 1, 2007;
- (3) It is required of private employees (not state or municipal employees) and apprentices who perform manual labor for a general contractor or subcontractor on a public building project where the total cost of all work to be performed is at least \$100,000;
- (4) The ten-hour construction course pertains to the ten-hour Outreach Course conducted in accordance with federal OSHA Training Institute standards, and, for telecommunications workers, a ten-hour training course conducted in accordance with federal OSHA standard, 29 CFR 1910.268;
- (5) The internet website for the federal OSHA Training Institute is http://www.osha.gov/fso/ote/training/edcenters/fact_sheet.html;
- (6) The statutory language leaves it to the contractor and its employees to determine who pays for the cost of the ten-hour Outreach Course;
- (7) Within 30 days of receiving a contract award, a general contractor must furnish proof to the Labor Commissioner that all employees and apprentices performing manual labor on the project will have completed such a course;
- (8) Proof of completion may be demonstrated through either: (a) the presentation of a *bona fide* student course completion card issued by the federal OSHA Training Institute; *or* (2) the presentation of documentation provided to an employee by a trainer certified by the Institute pending the actual issuance of the completion card;
- (9) Any card with an issuance date more than 5 years prior to the commencement date of the construction project shall not constitute proof of compliance;

- (10) Each employer shall affix a copy of the construction safety course completion card to the certified payroll submitted to the contracting agency in accordance with Conn. Gen. Stat. § 31-53(f) on which such employee's name first appears;
- (11) Any employee found to be in non-compliance shall be subject to removal from the worksite if such employee does not provide satisfactory proof of course completion to the Labor Commissioner by the fifteenth day after the date the employee is determined to be in noncompliance;
- (12) Any such employee who is determined to be in noncompliance may continue to work on a public building construction project for a maximum of fourteen consecutive calendar days while bringing his or her status into compliance;
- (13) The Labor Commissioner may make complaint to the prosecuting authorities regarding any employer or agent of the employer, or officer or agent of the corporation who files a false certified payroll with respect to the status of an employee who is performing manual labor on a public building construction project;
- (14) The statute provides the minimum standards required for the completion of a safety course by manual laborers on public construction contracts; any contractor can exceed these minimum requirements; and
- (15) Regulations clarifying the statute are currently in the regulatory process, and shall be posted on the CTDOL website as soon as they are adopted in final form.
- (16) Any questions regarding this statute may be directed to the Wage and Workplace Standards Division of the Connecticut Labor Department via the internet website of http://www.ctdol.state.ct.us/wgwkstnd/wgemenu.htm; or by telephone at (860)263-6790.

THE ABOVE INFORMATION IS PROVIDED EXCLUSIVELY AS AN EDUCATIONAL RESOURCE, AND IS NOT INTENDED AS A SUBSTITUTE FOR LEGAL INTERPRETATIONS WHICH MAY ULTMATELY ARISE CONCERNIG THE CONSTRUCTION OF THE STATUTE OR THE REGULATIONS. November 29, 2006

Notice

To All Mason Contractors and Interested Parties Regarding Construction Pursuant to Section 31-53 of the Connecticut General Statutes (Prevailing Wage)

The Connecticut Labor Department Wage and Workplace Standards Division is empowered to enforce the prevailing wage rates on projects covered by the above referenced statute.

Over the past few years the Division has withheld enforcement of the rate in effect for workers who operate a forklift on a prevailing wage rate project due to a potential jurisdictional dispute.

The rate listed in the schedules and in our Occupational Bulletin (see enclosed) has been as follows:

Forklift Operator:

- Laborers (Group 4) Mason Tenders - operates forklift solely to assist a mason to a maximum height of nine feet only.

- **Power Equipment Operator (Group 9)** - operates forklift to assist any trade and to assist a mason to a height over nine feet.

The U.S. Labor Department conducted a survey of rates in Connecticut but it has not been published and the rate in effect remains as outlined in the above Occupational Bulletin.

Since this is a classification matter and not one of jurisdiction, effective January 1, 2007 the Connecticut Labor Department will enforce the rate on each schedule in accordance with our statutory authority.

Your cooperation in filing appropriate and accurate certified payrolls is appreciated.

Information Bulletin Occupational Classifications

The Connecticut Department of Labor has the responsibility to properly determine *"job classification"* on prevailing wage projects covered under C.G.S. Section 31-53(d).

Note: This information is intended to provide a sample of some occupational classifications for guidance purposes only. It is not an all-inclusive list of each occupation's duties. This list is being provided only to highlight some areas where a contractor may be unclear regarding the proper classification. If unsure, the employer should seek guidelines for CTDOL.

Below are additional clarifications of specific job duties performed for certain classifications:

<u>ASBESTOS WORKERS</u>

Applies all insulating materials, protective coverings, coatings and finishes to all types of mechanical systems.

• ASBESTOS INSULATOR

Handle, install apply, fabricate, distribute, prepare, alter, repair, dismantle, heat and frost insulation, including penetration and fire stopping work on all penetration fire stop systems.

• **BOILERMAKERS**

Erects hydro plants, incomplete vessels, steel stacks, storage tanks for water, fuel, etc. Builds incomplete boilers, repairs heat exchanges and steam generators.

• <u>BRICKLAYERS, CEMENT MASONS, CEMENT FINISHERS, MARBLE MASONS,</u> <u>PLASTERERS, STONE MASONS, PLASTERERS. STONE MASONS, TERRAZZO</u> <u>WORKERS, TILE SETTERS</u>

Lays building materials such as brick, structural tile and concrete cinder, glass, gypsum, terra cotta block. Cuts, tools and sets marble, sets stone, finishes concrete, applies decorative steel, aluminum and plastic tile, applies cements, sand, pigment and marble chips to floors, stairways, etc.

• <u>CARPENTERS, MILLWRIGHTS. PILEDRIVERMEN. LATHERS. RESILEINT FLOOR</u> <u>LAYERS, DOCK BUILDERS, DIKERS, DIVER TENDERS</u>

Constructs, erects, installs and repairs structures and fixtures of wood, plywood and wallboard. Installs, assembles, dismantles, moves industrial machinery. Drives piling into ground to provide foundations for structures such as buildings and bridges, retaining walls for earth embankments, such as cofferdams. Fastens wooden, metal or rockboard lath to walls, ceilings and partitions of buildings, acoustical tile layer, concrete form builder. Applies firestopping materials on fire resistive joint systems only. Installation of curtain/window walls only where attached to wood or metal studs. Installation of insulated material of all types whether blown, nailed or attached in other ways to walls, ceilings and floors of buildings. Assembly and installation of modular furniture/furniture systems. Free-standing furniture is not covered. This includes free standing: student chairs, study top desks, book box desks, computer furniture, dictionary stand, atlas stand, wood shelving, two-position information access station, file cabinets, storage cabinets, tables, etc.

• LABORER, CLEANING

• The clean up of any construction debris and the general (heavy/light) cleaning, including sweeping, wash down, mopping, wiping of the construction facility and its furniture, washing, polishing, and dusting.

DELIVERY PERSONNEL

• If delivery of supplies/building materials is to one common point and stockpiled there, prevailing wages <u>are not required</u>. If the delivery personnel are involved in the distribution of the material to multiple locations within the construction site then they would have to be paid prevailing wages for the type of work performed: laborer, equipment operator, electrician, ironworker, plumber, etc.

• An example of this would be where delivery of drywall is made to a building and the delivery personnel distribute the drywall from one "stockpile" location to further sub-locations on each floor. Distribution of material around a construction site is the job of a laborer or tradesman, and not a delivery personnel.

• <u>ELECTRICIANS</u>

Install, erect, maintenance, alteration or repair of any wire, cable, conduit, etc., which generates, transforms, transmits or uses electrical energy for light, heat, power or other purposes, including the Installation or maintenance of telecommunication, LAN wiring or computer equipment, and low voltage wiring. **License required per Connecticut General Statutes: E-1,2 L-5,6 C-5,6 T-1,2 L-1,2 V-1,2,7,8,9.*

• ELEVATOR CONSTRUCTORS

Install, erect, maintenance and repair of all types of elevators, escalators, dumb waiters and moving walks. *License required by Connecticut General Statutes: R-1,2,5,6.

• FORK LIFT OPERATOR

Laborers Group 4) Mason Tenders - operates forklift solely to assist a mason to a maximum height of nine (9) feet only.

Power Equipment Operator Group 9 - operates forklift to assist any trade, and to assist a mason to a height over nine (9) feet.

• <u>GLAZIERS</u>

Glazing wood and metal sash, doors, partitions, and 2 story aluminum storefronts. Installs glass windows, skylights, store fronts and display cases or surfaces such as building fronts, interior walls, ceilings and table tops and metal store fronts. Installation of aluminum window walls and curtain walls is the "joint" work of glaziers and ironworkers, which require equal composite workforce.

IRONWORKERS

Erection, installation and placement of structural steel, precast concrete, miscellaneous iron, ornamental iron, metal curtain wall, rigging and reinforcing steel. Handling, sorting, and installation of reinforcing steel (rebar). Metal bridge rail (traffic), metal bridge handrail, and decorative security fence installation. Installation of aluminum window walls and curtain walls is the "joint" work of glaziers and ironworkers which require equal composite workforce.

• INSULATOR

• Installing fire stopping systems/materials for "Penetration Firestop Systems": transit to cables, electrical conduits, insulated pipes, sprinkler pipe penetrations, ductwork behind radiation, electrical cable trays, fire rated pipe penetrations, natural polypropylene, HVAC ducts, plumbing bare metal, telephone and communication wires, and boiler room ceilings.

LABORERS

Acetylene burners, asphalt rakers, chain saw operators, concrete and power buggy operator, concrete saw operator, fence and guard rail erector (except metal bridge rail (traffic), decorative security fence (non-metal).

installation.), hand operated concrete vibrator operator, mason tenders, pipelayers (installation of storm drainage or sewage lines on the street only), pneumatic drill operator, pneumatic gas and electric drill operator, powermen and wagon drill operator, air track operator, block paver, curb setters, blasters, concrete spreaders.

• <u>PAINTERS</u>

Maintenance, preparation, cleaning, blasting (water and sand, etc.), painting or application of any protective coatings of every description on all bridges and appurtenances of highways, roadways, and railroads. Painting, decorating, hardwood finishing, paper hanging, sign writing, scenic art work and drywall hhg for any and all types of building and residential work.

• LEAD PAINT REMOVAL

- Painter's Rate
 - 1. Removal of lead paint from bridges.
 - 2. Removal of lead paint as preparation of any surface to be repainted.
 - 3. Where removal is on a Demolition project prior to reconstruction.
- Laborer's Rate
 - 1. Removal of lead paint from any surface NOT to be repainted.
 - 2. Where removal is on a *TOTAL* Demolition project only.
 - PLUMBERS AND PIPEFITTERS

Installation, repair, replacement, alteration or maintenance of all plumbing, heating, cooling and piping. **License required per Connecticut General Statutes: P-1,2,6,7,8,9 J-1,2,3,4 SP-1,2 S-1,2,3,4,5,6,7,8 B-1,2,3,4 D-1,2,3,4*.

• <u>POWER EQUIPMENT OPERATORS</u>

Operates several types of power construction equipment such as compressors, pumps, hoists, derricks, cranes, shovels, tractors, scrapers or motor graders, etc. Repairs and maintains equipment. *License required, crane operators only, per Connecticut General Statutes.

• <u>ROOFERS</u>

Covers roofs with composition shingles or sheets, wood shingles, slate or asphalt and gravel to waterproof roofs, including preparation of surface. (demolition or removal of any type of roofing and or clean-up of any and all areas where a roof is to be relaid.)

• <u>SHEETMETAL WORKERS</u>

Fabricate, assembles, installs and repairs sheetmetal products and equipment in such areas as ventilation, air-conditioning, warm air heating, restaurant equipment, architectural sheet metal work, sheetmetal roofing, and aluminum gutters. Fabrication, handling, assembling, erecting, altering, repairing, etc. of coated metal material panels and composite metal material panels when used on building exteriors and interiors as soffits, facia, louvers, partitions, canopies, cornice, column covers, awnings, beam covers, cladding, sun shades, lighting troughs, spires, ornamental roofing, metal ceilings, mansards, copings, ornamental and ventilation hoods, vertical and horizontal siding panels, trim, etc. The sheet metal classification also applies to the vast variety of coated metal material panels and composite metal material panels that have evolved over the years as an alternative to conventional ferrous and non-ferrous metals like steel, iron, tin, copper, brass, bronze, aluminum, etc. Fabrication, handling, assembling, erecting, altering, repairing, etc. of architectural metal roof, standing seam roof, composite metal roof, metal and composite bathroom/toilet partitions, aluminum gutters, metal and composite lockers and shelving, kitchen equipment, and walk-in coolers. To include testing and air –balancing ancillary to installation and construction.

• SPRINKLER FITTERS

Installation, alteration, maintenance and repair of fire protection sprinkler systems. **License required per Connecticut General Statutes: F-1,2,3,4.*

• TILE MARBLE AND TERRAZZO FINISHERS

Assists and tends the tile setter, marble mason and terrazzo worker in the performance of their duties.

• TRUCK DRIVERS

~How to pay truck drivers delivering asphalt is under <u>REVISION~</u>

Truck Drivers are requires to be paid prevailing wage for time spent "working" directly on the site. These drivers remain covered by the prevailing wage for any time spent transporting between the actual construction location and facilities (such as fabrication, plants, mobile factories, batch plant, borrow pits, job headquarters, tool yards, etc.) dedicated exclusively, or nearly so, to performance of the contract or project, which are so located in proximity to the actual construction location that it is reasonable to include them. **License required, drivers only, per Connecticut General Statutes.*

For example:

• Material men and deliverymen are not covered under prevailing wage as long as they are not directly involved in the construction process. If, they unload the material, they would then be covered by prevailing wage for the classification they are performing work in: laborer, equipment operator, etc.

• Hauling material off site is not covered provided they are not dumping it at a location outlined above.

• Driving a truck on site and moving equipment or materials on site would be considered covered work, as this is part of the construction process.

 Any questions regarding the proper classification should be directed to: Public Contract Compliance Unit Wage and Workplace Standards Division Connecticut Department of Labor 200 Folly Brook Blvd, Wethersfield, CT 06109 (860) 263-6543.

Connecticut Department of Labor Wage and Workplace Standards Division FOOTNOTES

⇒ Please Note: If the "Benefits" listed on the schedule for the following occupations includes a letter(s) (+ a or + a+b for instance), refer to the information below.

Benefits to be paid at the appropriate prevailing wage rate for the listed occupation.

If the "Benefits" section for the occupation lists only a dollar amount, disregard the information below.

Bricklayers, Cement Masons, Cement Finishers, Concrete Finishers, Stone Masons (Building Construction) and

(Residential- Hartford, Middlesex, New Haven, New London and Tolland Counties)

a. Paid Holiday: Employees shall receive 4 hours for Christmas Eve holiday provided the employee works the regularly scheduled day before and after the holiday. Employers may schedule work on Christmas Eve and employees shall receive pay for actual hours worked in addition to holiday pay.

Elevator Constructors: Mechanics

- a. Paid Holidays: New Year's Day, Memorial Day, Independence Day, Labor Day, Veterans' Day, Thanksgiving Day, Christmas Day, plus the Friday after Thanksgiving.
- b. Vacation: Employer contributes 8% of basic hourly rate for 5 years or more of service or 6% of basic hourly rate for 6 months to 5 years of service as vacation pay credit.

Glaziers

a. Paid Holidays: Labor Day and Christmas Day.

Power Equipment Operators

(Heavy and Highway Construction & Building Construction)

a. Paid Holidays: New Year's Day, Good Friday, Memorial day, Independence Day, Labor Day, Thanksgiving Day and Christmas Day, provided the employee works 3 days during the week in which the holiday falls, if scheduled, and if scheduled, the working day before and the working day after the holiday. Holidays falling on Saturday may be observed on Saturday, or if the employer so elects, on the preceding Friday.

Ironworkers

a. Paid Holiday: Labor Day provided employee has been on the payroll for the 5 consecutive work days prior to Labor Day.

Laborers (Tunnel Construction)

a. Paid Holidays: New Year's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day and Christmas Day. No employee shall be eligible for holiday pay when he fails, without cause, to work the regular work day preceding the holiday or the regular work day following the holiday.

Roofers

a. Paid Holidays: July 4th, Labor Day, and Christmas Day provided the employee is employed 15 days prior to the holiday.

Sprinkler Fitters

a. Paid Holidays: Memorial Day, July 4th, Labor Day, Thanksgiving Day and Christmas Day, provided the employee has been in the employment of a contractor 20 working days prior to any such paid holiday.

Truck Drivers

(Heavy and Highway Construction & Building Construction)

a. Paid Holidays: New Year's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Christmas day, and Good Friday, provided the employee has at least 31 calendar days of service and works the last scheduled day before and the first scheduled day after the holiday, unless excused.

CITY OF MERIDEN, CONNECTICUT

INSURANCE REQUIREMENTS

Contractor shall agree to maintain in force at all times during the contract the following minimum coverages and shall name the City of Meriden as an **Additional Insured on a primary and non-contributory basis** to all policies except Workers Compensation and Professional Liability. All policies should also include a Waiver of Subrogation. Umbrella/Excess shall state that it follows form over General Liability, Auto Liability and Workers Compensation. Insurance shall be written with Carriers approved in the State of Connecticut and with a minimum AM Best's rating of "A-"VIII. In addition, all Carriers are subject to approval by the City of Meriden.

		(Minimum Limits)
General Liability	Each Occurrence	\$1,000,000
	General Aggregate	\$2,000,000
	Products/Completed Operations Aggregate	\$2,000,000
Auto Liability	Combined Single Limit	
	Each Accident	\$1,000,000
Umbrella	Each Occurrence	\$1,000,000
(Excess Liability)	Aggregate	\$1,000,000
Workers' Compensation	and WC Statutory Limits	
Employers' Liability	EL Each Accident	\$1,000.000
1 5 5	EL Disease Each Employee	\$1,000,000
	EL Disease Policy Limit	\$1,000,000
Errors & Omissions	Each Occurrence	\$1,000,000

Original, completed Certificates of Insurance must be presented to the City of Meriden prior to contract issuance. Contractor agrees to provide replacement/renewal certificates at least 60 days prior to the expiration date of the policies.

CITY OF MERIDEN, CONNECTICUT

B024-39 – ARPA- Meriden Girls Inc. Renovations

AGREEMENT

 This AGREEMENT, made as of this ______, 2024 by and between the City of Meriden, 142

 East
 Main
 Street,
 Meriden
 CT
 06450
 (hereinafter the "City")

 and ______
 (hereinafter the "Contractor").

WHEREAS, the City has issued a certain Invitation to Bid regarding

WHEREAS, the Contractor is in the business of providing such services, is the successful bidder, and has been awarded this Agreement.

NOW THEREFORE, the parties mutually agree as follows:

- 1. <u>AGREEMENT OF THE PARTIES</u>: The City hereby contracts for and the Contractor hereby agrees to perform the work as stated in the minimum specifications as required by the City at the locations requested.
- 2. <u>SCOPE OF SERVICES</u>: As per the bid proposal documents, incorporated by reference herein.
- 3. <u>TERM</u>: This Agreement shall begin ______ and end _____, unless extended or terminated.
- 4. <u>PAYMENT</u>: The City shall pay the Contractor in accordance with the bid proposal documents, incorporated by reference herein. Work performed without authorization will not be paid for.
- 5. <u>AUDITS</u>: At any time during normal business hours, and as often as maybe deemed necessary, the Contractor shall make available for examination of all records with respect to all matters covered by this Agreement and will permit authorized City, State, and/or Federal Officials to audit, inspect, examine and make excerpts or transcripts, from such records and to make audits of all contracts, invoices, payrolls, and other data relating to all matters covered by this Agreement.
- 6. <u>INDEPENDENT</u> CONTRACTOR: It is understood that the services of the Contractor shall be as an independent contractor and not as an employee of the City, and that persons employed by said Contractor providing services under this Agreement shall be the employees of the Contractor and not of the City.
- 7. <u>INDEMNIFICATION</u>: Contractor shall indemnify and save harmless City against any and all damages to property or injuries to or death of any person or persons, including property and employees or agents of City, and shall defend, indemnify and save harmless City from

any and all claims, demands, suits, actions or proceedings of any kind or nature including workers' compensation claims, of or by anyone in any way resulting from or arising out of the operations in connection with this Agreement, including operations of subcontractors and acts or omissions of employees or agents of contractor or his or her subcontractors. Insurance coverage specified in this Agreement constitutes the minimum requirements and said requirements shall in no way lessen or limit the liability of Contractor under the terms of this Agreement. Contractor shall procure and maintain, at its own cost and expense, any additional kinds and amounts of insurance which, in its own judgment, may be necessary for his/her/its proper protection in the prosecution of the work. Contractor agrees that will well and truly save and indemnify and keep harmless the City against all liability, judgments, costs and expenses which may in any way come against owner, or which may in any way result from carelessness or neglect of contractor or its agents, employees or workmen in any respect whatever.

8. <u>INSURANCE</u>: The Contractor shall provide and maintain a Certificate of Insurance for the duration of this Agreement naming the City of Meriden as an ADDITIONAL INSURED and with the types and limits stated in the insurance requirements section. The Contractor agrees to provide Certificates of Insurance as requested by the City.

The contractor and its subcontractors shall comply with the Workers' Compensation Act of the State of Connecticut and shall provide compensation insurance to protect the Contractor, its subcontractors, and the City from and against any and all workers' compensation claims arising from performance of the Work under the Agreement. The City shall be furnished, prior to undertaking any work, copies of the certificate or certificates evidencing such insurance to be in effect.

- 9. <u>TERMINATION</u>: The City or the Contractor shall have the right, without cause, to terminate this Agreement within fifteen (15) days following the written notification to the other party to that effect by Certified Mail or personal delivery by agent, and upon the expiration of said fifteen (15) day period, this Agreement shall be null and void and all rights established herein shall end and terminate. It shall not be necessary for either party to state a reason for such termination.
- 10. <u>BID PROPOSAL DOCUMENTS</u>: The bid proposal documents are hereby incorporated by reference and made a part of this Agreement.
- 11. <u>DEFECTIVE WORK</u>: During a period of one year from and after the final acceptance of the work except as provided elsewhere in this Agreement, the Contractor shall, at its own expense, make all needed repairs or replacements due to defective work or materials which, in the judgment of the City, shall become necessary during this period. If within ten (10) days after the mailing of a written notice to the Contractor, or its agent, requesting repairs or replacement, the Contractor shall neglect to make or undertake with due diligence to make them, the City may make the repairs at the Contractor's expense; provided, however, that in the case of emergency where, in the judgment of the City, delay would cause serious loss or damage, repairs or replacement may be made without notice being sent to the Contractor, and the contractor shall pay the cost.

- 12. <u>GUARANTEE</u>: The Contractor shall further guarantee that the sidewalk repairs shall remain in good order and repair for a period of five (5) years from all causes arising from defective work and materials, and to make all repairs arising from these causes during this period without further compensation, and shall guarantee the sidewalks against defective work and materials, and shall keep them in good order and repair without further compensation for a period of two (2) years from and after completion and acceptance of them by the City. The determination of the necessity for the repair or replacement of sidewalks or any portion of them, shall rest entirely with the City, whose decision upon the matter shall be final and obligatory upon the Contractor.
- 13. <u>CLEAN UP</u>: The contractor shall remove and clean up all rubbish, debris, excess material, temporary structures, tools and equipment from streets, alleys, parkways, and adjacent property that may have been used or worked on by the Contractor in connection with the Work promptly as each section or portion is completed and ready for use, leaving the same in a neat and presentable condition. Payment of monthly or partial estimates may be withheld until this has been done to the satisfaction of the City. Final acceptance and payment for the entire project will not be made until this has been fully taken care of.

During the construction, the Contractor shall keep the site of the work and adjacent premises as free from material, debris, and rubbish as is practicable and shall remove this waste entirely and at once, if, in the opinion of the city, such material, debris, or rubbish constitutes a nuisance, a safety hazard or is objectionable in any way to the public.

Upon completion and before final acceptance of the work, the Contractor shall remove from the site of the work and adjacent premises all machinery, equipment, surplus materials, falsework, excavated and useless materials, rubbish, temporary buildings, barricades and signs, and shall restore the site to the same general conditions that existed prior to the commencement of its operations. The cost of final cleaning up will not be paid for under any specific scheduled item but shall be included in the prices bid for the various items, or included in the contract lump-sum price as the case may be.

The Contractor shall clean off all cement streaks or drippings, paint smears or drippings, rust stains, oil, grease, dirt, and any other foreign materials deposited or accumulated on any portion of its work, or existing work, due to its operations.

- 14. <u>CHARTERS, LAWS, AND ORDINANCES</u>: The Contractor shall at all times observe and comply with the provisions of the charter, ordinances and regulations of the City of Meriden and of state and federal laws, rules and regulations which in any manner limit, control, or apply to the actions or operations of the contractor, its subcontractors, or the subcontractors employees, agents or servants, engaged upon the work or affecting the materials supplied to or by them
- 15. <u>CONTRACTOR'S DUTIES</u>: The Contractor shall be responsible for the completion of the Work according to the bid proposal documents, and its responsibility shall not cease until the whole work contracted for is completed and accepted. The Contractor shall keep fully informed at all times regarding all details of the work. The Contractor shall be responsible for all delays that may result in failure to install the work in the proper manner and proper

time. It shall carefully study and compare all drawings, specifications, and other instructions and shall immediately report to the City any error or omission which the Contractor may discover, and shall subsequently proceed with the work in accordance with the instructions from the City concerning the error or omission. The Contractor shall furnish a competent and adequate staff as necessary for the proper administration, coordination and supervision of the work, organize the procurement of all the materials and equipment so that it will be available at the time they are needed for the work. The Contractor shall keep an adequate force of skilled workers on the job to complete the work in accordance with all requirements of the contract documents. Any employee who is considered incompetent or careless in his or her work shall be removed on demand of the engineer and replaced by a competent person.

- 16. <u>LIQUIDATED DAMAGES</u>. The Contractor acknowledges that time is of the essence for the Contractor to render its services and that the City will suffer financial loss if the Work is not completed within the times specified in paragraph 3 above. The Contractor acknowledges the expense and difficulties involved in proving the actual loss suffered by the City if the Work is not completed on time. Accordingly, instead of requiring any such proof, the City and Contractor agree that as liquidated damages for any delay (but not as a penalty), the Contractor shall pay the City TWO HUNDRED FIFTY AND NO/100 DOLLARS (\$250.00) for each day that expires after the time specified in Section 3 for substantial completion until the Work is substantially complete. After substantial completion, if Contractor neglects, refuses, or fails to complete the remaining Work within the time specified in Section 3, Contractor shall pay the City TWO HUNDRED FIFTY AND NO/100 DOLLARS (\$250.00) for each day that expires after the time specified in Section 3 for completion and readiness for final payment.
- 17. <u>BLUE PENCIL PROVISION</u>: In the event that any provision of this Agreement is unenforceable under applicable law, the validity or enforceability of the remaining provisions will not be affected. To the extent any provision of this Agreement is judicially determined to be unenforceable, a court of competent jurisdiction may reform any such provision to make it enforceable. The provisions of this Agreement will, where possible, be interpreted so as to sustain its legality and enforceability.

IN WITNESS WHEREOF, the parties hereto have set their hands and seals on the day and year indicated.

CONTRACTOR:

Duly Authorized
CITY OF MERIDEN:
Emily E Holland, Acting City Manager
Duly Authorized
Date:

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

Prepared by

ENGINEERS JOINT CONTRACT DOCUMENTS COMMITTEE

and

Issued and Published Jointly by









AMERICAN COUNCIL OF ENGINEERING COMPANIES

ASSOCIATED GENERAL CONTRACTORS OF AMERICA

AMERICAN SOCIETY OF CIVIL ENGINEERS

PROFESSIONAL ENGINEERS IN PRIVATE PRACTICE A Practice Division of the NATIONAL SOCIETY OF PROFESSIONAL ENGINEERS

Endorsed by



CONSTRUCTION SPECIFICATIONS INSTITUTE

Copyright © 2007 National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882 www.nspe.org

> American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474 www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723 www.asce.org

Associated General Contractors of America 2300 Wilson Boulevard, Suite 400, Arlington, VA 22201-3308 (703) 548-3118 www.agc.org

The copyright for this EJCDC document is owned jointly by the four EJCDC sponsoring organizations and held in trust for their benefit by NSPE.

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

TABLE OF CONTENTS

1.01 Defined Terms 1 1.02 Terminology 5 Article 2 – Preliminary Matters 6 2.01 Delivery of Bonds and Evidence of Insurance 6 2.02 Copies of Documents 6 2.03 Commencement of Contract Times; Notice to Proceed 6 2.04 Starting the Work 7 2.05 Before Starting Construction 7 2.06 Preconstruction Conference; Designation of Authorized Representatives 7 2.07 Initial Acceptance of Schedules 7 2.07 Initial Acceptance of Schedules 7 Article 3 – Contract Documents: Intent, Amending, Reuse 8 3.01 Intent 8 3.02 Reference Standards 8 3.03 Reporting and Resolving Discrepancies 9 3.04 Amending and Supplementing Contract Documents 9 3.05 Reuse of Documents 10 3.06 Electronic Data 10 3.07 Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental Conditions; Reference Points 11 4.01 Availability of	Article 1 – 1	Definitions and Terminology	1
1.02 Terminology 5 Article 2 – Preliminary Matters 6 2.01 Delivery of Bonds and Evidence of Insurance 6 2.02 Copies of Documents 6 2.03 Commencement of Contract Times; Notice to Proceed 6 2.04 Starting the Work 7 2.05 Before Starting Construction 7 2.06 Preconstruction Conference; Designation of Authorized Representatives 7 2.07 Initial Acceptance of Schedules 7 Article 3 – Contract Documents: Intent, Amending, Reuse 8 3.01 Intent 8 3.02 Reference Standards 8 3.03 Reporting and Resolving Discrepancies 9 3.04 Amending and Supplementing Contract Documents 9 3.05 Reuse of Documents 10 3.06 Electronic Data 10 Article 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental Conditions; Reference Points 11 4.01 Availability of Lands 11 4.02 Subsurface or Physical Conditions 12 4.04 Underground	1.01	Defined Terms	1
Article 2 - Preliminary Matters. 6 2.01 Delivery of Bonds and Evidence of Insurance. 6 2.02 Copies of Documents 6 2.03 Commencement of Contract Times; Notice to Proceed. 6 2.04 Starting the Work. 7 2.05 Before Starting Construction 7 2.06 Preconstruction Conference; Designation of Authorized Representatives. 7 2.07 Initial Acceptance of Schedules 7 Article 3 - Contract Documents: Intent, Amending, Reuse. 8 3.01 Intent 8 3.02 Reference Standards 8 3.03 Reporting and Resolving Discrepancies 9 3.04 Amending and Supplementing Contract Documents 10 3.06 Electronic Data 10 3.07 Reuse of Documents 10 3.08 Differing Subsurface and Physical Conditions; Hazardous Environmental 11 4.01 Availability of Lands; Subsurface and Physical Conditions 11 4.03 Differing Subsurface or Physical Conditions 12 4.04 Underground Facilitices 13	1.02	Terminology	5
Article 2 – Preliminary Matters.			
2.01 Delivery of Bonds and Evidence of Insurance.	Article 2 – I	Preliminary Matters	6
2.02 Copies of Documents	2.01	Delivery of Bonds and Evidence of Insurance	6
2.03 Commencement of Contract Times; Notice to Proceed	2.02	Copies of Documents	6
2.04 Starting the Work	2.03	Commencement of Contract Times; Notice to Proceed	6
2.05 Before Starting Construction .7 2.06 Preconstruction Conference; Designation of Authorized Representatives .7 2.07 Initial Acceptance of Schedules .7 Article 3 – Contract Documents: Intent, Amending, Reuse. .8 3.01 Intent .8 3.02 Reference Standards .8 3.03 Reporting and Resolving Discrepancies .9 3.04 Amending and Supplementing Contract Documents .9 3.06 Electronic Data .10 3.06 Electronic Data .10 3.06 Electronic Data .10 Article 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental Conditions: Reference Points .11 4.01 Availability of Lands .11 4.02 Subsurface and Physical Conditions .12 4.04 Underground Facilities .13 4.05 Reference Points .14 4.06 Hazardous Environmental Condition at Site .14 4.06 Hazardous Environmental Condition at Site .14 4.06 Hazardous Environmental Condition at Site .16 <td>2.04</td> <td>Starting the Work</td> <td>7</td>	2.04	Starting the Work	7
2.06 Preconstruction Conference; Designation of Authorized Representatives .7 2.07 Initial Acceptance of Schedules .7 Article 3 - Contract Documents: Intent, Amending, Reuse .8 3.01 Intent .8 3.02 Reference Standards .8 3.03 Reporting and Resolving Discrepancies .9 3.04 Amending and Supplementing Contract Documents .9 3.05 Reuse of Documents. .10 3.06 Electronic Data .10 3.07 Reference Points .11 4.01 Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental Conditions; Reference Points .11 4.01 Availability of Lands .11 4.02 Subsurface or Physical Conditions .11 4.03 Differing Subsurface or Physical Conditions .12 4.04 Underground Facilities .13 4.05 Reference Points .14 4.06 Hazardous Environmental Condition at Site .14 4.06 Hazardous Environmental Condition at Site .14 Article 5 – Bonds and Insurance .16	2.05	Before Starting Construction	7
2.07 Initial Acceptance of Schedules	2.06	Preconstruction Conference; Designation of Authorized Representatives	7
Article 3 – Contract Documents: Intent, Amending, Reuse 8 3.01 Intent 8 3.02 Reference Standards 8 3.03 Reporting and Resolving Discrepancies 9 3.04 Amending and Supplementing Contract Documents 9 3.05 Reuse of Documents. 10 3.06 Electronic Data 10 Article 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental 10 Conditions; Reference Points 11 4.01 Availability of Lands 11 4.02 Subsurface and Physical Conditions 11 4.03 Differing Subsurface or Physical Conditions 12 4.04 Underground Facilities 13 4.05 Reference Points 14 4.06 Hazardous Environmental Condition at Site 14 4.06 Hazardous Environmental Condition at Site 16 5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 </td <td>2.07</td> <td>Initial Acceptance of Schedules</td> <td>7</td>	2.07	Initial Acceptance of Schedules	7
Article 3 – Contract Documents: Intent, Amending, Reuse			
3.01 Intent	Article 3 –	Contract Documents: Intent, Amending, Reuse	8
3.02 Reference Standards	3.01	Intent	8
3.03 Reporting and Resolving Discrepancies	3.02	Reference Standards	8
3.04 Amending and Supplementing Contract Documents	3.03	Reporting and Resolving Discrepancies	9
3.05 Reuse of Documents	3.04	Amending and Supplementing Contract Documents	9
3.06 Electronic Data 10 Article 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental 11 4.01 Availability of Lands 11 4.01 Availability of Lands 11 4.02 Subsurface and Physical Conditions 11 4.03 Differing Subsurface or Physical Conditions 12 4.04 Underground Facilities 13 4.05 Reference Points 14 4.06 Hazardous Environmental Condition at Site 14 4.06 Hazardous Environmental Condition at Site 16 5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurers 16 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Deventsu 19	3.05	Reuse of Documents	10
Article 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental 11 4.01 Availability of Lands 11 4.02 Subsurface and Physical Conditions 11 4.03 Differing Subsurface or Physical Conditions 12 4.04 Underground Facilities 13 4.05 Reference Points 14 4.06 Hazardous Environmental Condition at Site 14 Article 5 – Bonds and Insurance 16 16 5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Demonstructure 19	3.06	Electronic Data	10
Article 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental 11 4.01 Availability of Lands 11 4.02 Subsurface and Physical Conditions 11 4.03 Differing Subsurface or Physical Conditions 12 4.04 Underground Facilities 13 4.05 Reference Points 14 4.06 Hazardous Environmental Condition at Site 14 4.06 Hazardous Environmental Condition at Site 16 5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurers 16 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Demonstrate 19			
Conditions; Reference Points114.01Availability of Lands114.02Subsurface and Physical Conditions114.03Differing Subsurface or Physical Conditions124.04Underground Facilities134.05Reference Points144.06Hazardous Environmental Condition at Site14Article 5 – Bonds and Insurance165.01Performance, Payment, and Other Bonds165.02Licensed Sureties and Insurers165.03Certificates of Insurance175.04Contractor's Insurance175.05Owner's Liability Insurance195.06Purporter Insurance10	Article $4 - 1$	Availability of Lands; Subsurface and Physical Conditions; Hazardous Environmental	
4.01Availability of Lands114.02Subsurface and Physical Conditions114.03Differing Subsurface or Physical Conditions124.04Underground Facilities134.05Reference Points144.06Hazardous Environmental Condition at Site14Article 5 – Bonds and Insurance165.01Performance, Payment, and Other Bonds165.02Licensed Sureties and Insurance165.03Certificates of Insurance175.04Contractor's Insurance175.05Owner's Liability Insurance195.06Demonstrate Learners10	С	onditions; Reference Points	11
4.02 Subsurface and Physical Conditions 11 4.03 Differing Subsurface or Physical Conditions 12 4.04 Underground Facilities 13 4.05 Reference Points 14 4.06 Hazardous Environmental Condition at Site 14 Article 5 – Bonds and Insurance 16 5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurers 16 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Demonstrate Learners 10	4.01	Availability of Lands	11
4.03Differing Subsurface or Physical Conditions124.04Underground Facilities134.05Reference Points144.06Hazardous Environmental Condition at Site14Article 5 – Bonds and Insurance165.01Performance, Payment, and Other Bonds165.02Licensed Sureties and Insurers165.03Certificates of Insurance175.04Contractor's Insurance175.05Owner's Liability Insurance195.06Demonstra Insurance10	4.02	Subsurface and Physical Conditions	11
4.04Underground Facilities134.05Reference Points144.06Hazardous Environmental Condition at Site14Article 5 – Bonds and Insurance165.01Performance, Payment, and Other Bonds165.02Licensed Sureties and Insurers165.03Certificates of Insurance175.04Contractor's Insurance175.05Owner's Liability Insurance19	4.03	Differing Subsurface or Physical Conditions	12
4.05Reference Points144.06Hazardous Environmental Condition at Site14Article 5 – Bonds and Insurance165.01Performance, Payment, and Other Bonds165.02Licensed Sureties and Insurers165.03Certificates of Insurance175.04Contractor's Insurance175.05Owner's Liability Insurance19	4.04	Underground Facilities	13
4.06 Hazardous Environmental Condition at Site 14 Article 5 – Bonds and Insurance 16 5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurers 16 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19	4.05	Reference Points	14
Article 5 – Bonds and Insurance165.01Performance, Payment, and Other Bonds165.02Licensed Sureties and Insurers165.03Certificates of Insurance175.04Contractor's Insurance175.05Owner's Liability Insurance195.06Demonstra Insurance10	4.06	Hazardous Environmental Condition at Site	14
5.01 Performance, Payment, and Other Bonds 16 5.02 Licensed Sureties and Insurers 16 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Property Insurance 10	Article 5	Ponds and Insurance	16
5.01 Ferrormance, Fayment, and Other Bonds 10 5.02 Licensed Sureties and Insurers 16 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Property Insurance 10	Article $J = 1$	Donds and Insurance	16
5.02 Elcensed Sureles and Insurers 10 5.03 Certificates of Insurance 17 5.04 Contractor's Insurance 17 5.05 Owner's Liability Insurance 19 5.06 Property Insurance 10	5.02	Liconsod Suration and Insuran	10
5.05 Certificates of insurance	5.02	Cartificates of Insurance	10
5.04 Contractor's Insurance	5.03	Centrates of Insurance	1/ 17
5.05 Uwner's Liability Insurance	5.04	Contractor s insurance.	1/
	5.05	Owner's Liability Insurance	19
5.06 Property insurance	5.06	Property Insurance	19
5.0/ Waiver of Rights	5.07	Waiver of Rights	20
5.08 Receipt and Application of Insurance Proceeds	5.08	Receipt and Application of Insurance Proceeds	21

Copyright © 2007 National Society of Professional Engineers for EJCDC. All rights reserved.

5.09	Acceptance of Bonds and Insurance; Option to Replace	21
5.10	Partial Utilization, Acknowledgment of Property Insurer	22
Article 6 –	Contractor's Responsibilities	
6.01	Supervision and Superintendence	
6.02	Labor; Working Hours	
6.03	Services, Materials, and Equipment	
6.04	Progress Schedule	
0.05	Substitutes and Of-Equals	
0.00	Detert Face and Develties	
6.02	Patent Fees and Royanies	
6.00	I awa and Degulations	
6.10	Laws and Regulations	
6.10	Use of Site and Other Areas	28
6.12	Record Documents	29
6.12	Safety and Protection	29
6.13	Safety Representative	30
6.15	Hazard Communication Programs	30
6.16	Emergencies	30
6.10	Shop Drawings and Samples	31
6.18	Continuing the Work	32
6.19	Contractor's General Warranty and Guarantee.	
6.20	Indemnification	
6.21	Delegation of Professional Design Services	
Article 7 –	Other Work at the Site	35
7.01	Related Work at Site	
7.02	Coordination	35
7.03	Legal Relationships	
Article 8_	Owner's Responsibilities	36
8 01	Communications to Contractor	36
8.02	Replacement of Engineer	36
8.03	Furnish Data	36
8.04	Pay When Due	36
8.05	Lands and Easements: Reports and Tests	36
8.06	Insurance	
8.07	Change Orders	
8.08	Inspections, Tests, and Approvals	
8.09	Limitations on Owner's Responsibilities	
8.10	Undisclosed Hazardous Environmental Condition	
8.11	Evidence of Financial Arrangements	
8.12	Compliance with Safety Program	
		-
Article 9 –	Engineer's Status During Construction	
9.01	Owner's Representative	

EJCDC C-700 Standard General Conditions of the Construction Contract Copyright © 2007 National Society of Professional Engineers for EJCDC. All rights reserved.

9.02	Visits to Site	37
9.03	Project Representative	
9.04	Authorized Variations in Work	
9.05	Rejecting Defective Work	
9.06	Shop Drawings, Change Orders and Payments	
9.07	Determinations for Unit Price Work	
9.08	Decisions on Requirements of Contract Documents and Acceptability of Work	
9.09	Limitations on Engineer's Authority and Responsibilities	
9.10	Compliance with Safety Program	40
Article 10 –	Changes in the Work; Claims	40
10.01	Authorized Changes in the Work	40
10.02	Unauthorized Changes in the Work	41
10.03	Execution of Change Orders	41
10.04	Notification to Surety	41
10.05	Claims	41
Article 11 –	Cost of the Work; Allowances; Unit Price Work	42
11.01	Cost of the Work	42
11.02	Allowances	45
11.03	Unit Price Work	45
Article 12 –	Change of Contract Price; Change of Contract Times	46
12.01	Change of Contract Price	46
12.02	Change of Contract Times	47
12.03	Delays	47
Article 13 –	Tests and Inspections; Correction, Removal or Acceptance of Defective Work	48
13.01	Notice of Defects	
13.02	Access to Work	48
13.03	Tests and Inspections	49
13.04	Uncovering Work	49
13.05	Owner May Stop the Work	50
13.06	Correction or Removal of Defective Work	50
13.07	Correction Period	50
13.08	Acceptance of Defective Work	51
13.09	Owner May Correct Defective Work	52
Article 14 -	Payments to Contractor and Completion	
Article 14 – 14.01	Payments to Contractor and Completion Schedule of Values	52
Article 14 – 14.01 14.02	Payments to Contractor and Completion Schedule of Values Progress Payments	
Article 14 – 14.01 14.02 14.03	Payments to Contractor and Completion Schedule of Values Progress Payments Contractor's Warranty of Title	
Article 14 – 14.01 14.02 14.03 14.04	Payments to Contractor and Completion Schedule of Values Progress Payments Contractor's Warranty of Title Substantial Completion	
Article 14 – 14.01 14.02 14.03 14.04 14.04	Payments to Contractor and Completion Schedule of Values Progress Payments Contractor's Warranty of Title Substantial Completion Partial Utilization	
Article 14 – 14.01 14.02 14.03 14.04 14.05 14.06	Payments to Contractor and Completion Schedule of Values Progress Payments Contractor's Warranty of Title Substantial Completion Partial Utilization Final Inspection	
Article 14 – 14.01 14.02 14.03 14.04 14.05 14.06 14.06	Payments to Contractor and Completion Schedule of Values Progress Payments Contractor's Warranty of Title Substantial Completion Partial Utilization Final Inspection Final Payment	

EJCDC C-700 Standard General Conditions of the Construction Contract Copyright © 2007 National Society of Professional Engineers for EJCDC. All rights reserved.

14.09	Waiver of Claims	58
Article 15 –	Suspension of Work and Termination	59
15.01	Owner May Suspend Work	59
15.02	Owner May Terminate for Cause	59
15.03	Owner May Terminate For Convenience	60
15.04	Contractor May Stop Work or Terminate	60
Article 16 –	Dispute Resolution	61
16.01	Methods and Procedures	61
Article 17 –	Miscellaneous	61
17.01	Giving Notice	61
17.02	Computation of Times	62
17.03	Cumulative Remedies	62
17.04	Survival of Obligations	62
17.05	Controlling Law.	62
17.06	Headings	62

ARTICLE 1 – DEFINITIONS AND TERMINOLOGY

1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents and printed with initial capital letters, the terms listed below will have the meanings indicated which are applicable to both the singular and plural thereof. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
 - 1. *Addenda*—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
 - 2. *Agreement*—The written instrument which is evidence of the agreement between Owner and Contractor covering the Work.
 - 3. *Application for Payment*—The form acceptable to Engineer which is to be used by Contractor during the course of the Work in requesting progress or final payments and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
 - 4. *Asbestos*—Any material that contains more than one percent asbestos and is friable or is releasing asbestos fibers into the air above current action levels established by the United States Occupational Safety and Health Administration.
 - 5. *Bid*—The offer or proposal of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
 - 6. *Bidder*—The individual or entity who submits a Bid directly to Owner.
 - 7. *Bidding Documents*—The Bidding Requirements and the proposed Contract Documents (including all Addenda).
 - 8. *Bidding Requirements*—The advertisement or invitation to bid, Instructions to Bidders, Bid security of acceptable form, if any, and the Bid Form with any supplements.
 - 9. *Change Order*—A document recommended by Engineer which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, issued on or after the Effective Date of the Agreement.
 - 10. *Claim*—A demand or assertion by Owner or Contractor seeking an adjustment of Contract Price or Contract Times, or both, or other relief with respect to the terms of the Contract. A demand for money or services by a third party is not a Claim.
 - 11. *Contract*—The entire and integrated written agreement between the Owner and Contractor concerning the Work. The Contract supersedes prior negotiations, representations, or agreements, whether written or oral.

- 12. *Contract Documents*—Those items so designated in the Agreement. Only printed or hard copies of the items listed in the Agreement are Contract Documents. Approved Shop Drawings, other Contractor submittals, and the reports and drawings of subsurface and physical conditions are not Contract Documents.
- 13. *Contract Price*—The moneys payable by Owner to Contractor for completion of the Work in accordance with the Contract Documents as stated in the Agreement (subject to the provisions of Paragraph 11.03 in the case of Unit Price Work).
- 14. *Contract Times*—The number of days or the dates stated in the Agreement to: (i) achieve Milestones, if any; (ii) achieve Substantial Completion; and (iii) complete the Work so that it is ready for final payment as evidenced by Engineer's written recommendation of final payment.
- 15. *Contractor*—The individual or entity with whom Owner has entered into the Agreement.
- 16. Cost of the Work—See Paragraph 11.01 for definition.
- 17. *Drawings*—That part of the Contract Documents prepared or approved by Engineer which graphically shows the scope, extent, and character of the Work to be performed by Contractor. Shop Drawings and other Contractor submittals are not Drawings as so defined.
- 18. *Effective Date of the Agreement*—The date indicated in the Agreement on which it becomes effective, but if no such date is indicated, it means the date on which the Agreement is signed and delivered by the last of the two parties to sign and deliver.
- 19. Engineer—The individual or entity named as such in the Agreement.
- 20. *Field Order*—A written order issued by Engineer which requires minor changes in the Work but which does not involve a change in the Contract Price or the Contract Times.
- 21. General Requirements—Sections of Division 1 of the Specifications.
- 22. *Hazardous Environmental Condition*—The presence at the Site of Asbestos, PCBs, Petroleum, Hazardous Waste, or Radioactive Material in such quantities or circumstances that may present a substantial danger to persons or property exposed thereto.
- 23. *Hazardous Waste*—The term Hazardous Waste shall have the meaning provided in Section 1004 of the Solid Waste Disposal Act (42 USC Section 6903) as amended from time to time.
- 24. *Laws and Regulations; Laws or Regulations*—Any and all applicable laws, rules, regulations, ordinances, codes, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- 25. *Liens*—Charges, security interests, or encumbrances upon Project funds, real property, or personal property.
- 26. *Milestone*—A principal event specified in the Contract Documents relating to an intermediate completion date or time prior to Substantial Completion of all the Work.

EJCDC C-700 Standard General Conditions of the Construction Contract Copyright © 2007 National Society of Professional Engineers for EJCDC. All rights reserved. Page 2 of 62

- 27. *Notice of Award*—The written notice by Owner to the Successful Bidder stating that upon timely compliance by the Successful Bidder with the conditions precedent listed therein, Owner will sign and deliver the Agreement.
- 28. *Notice to Proceed*—A written notice given by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work under the Contract Documents.
- 29. *Owner*—The individual or entity with whom Contractor has entered into the Agreement and for whom the Work is to be performed.
- 30. PCBs—Polychlorinated biphenyls.
- 31. *Petroleum*—Petroleum, including crude oil or any fraction thereof which is liquid at standard conditions of temperature and pressure (60 degrees Fahrenheit and 14.7 pounds per square inch absolute), such as oil, petroleum, fuel oil, oil sludge, oil refuse, gasoline, kerosene, and oil mixed with other non-Hazardous Waste and crude oils.
- 32. *Progress Schedule*—A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising the Contractor's plan to accomplish the Work within the Contract Times.
- 33. *Project*—The total construction of which the Work to be performed under the Contract Documents may be the whole, or a part.
- 34. *Project Manual*—The bound documentary information prepared for bidding and constructing the Work. A listing of the contents of the Project Manual, which may be bound in one or more volumes, is contained in the table(s) of contents.
- 35. *Radioactive Material*—Source, special nuclear, or byproduct material as defined by the Atomic Energy Act of 1954 (42 USC Section 2011 et seq.) as amended from time to time.
- 36. *Resident Project Representative*—The authorized representative of Engineer who may be assigned to the Site or any part thereof.
- 37. *Samples*—Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and which establish the standards by which such portion of the Work will be judged.
- 38. *Schedule of Submittals*—A schedule, prepared and maintained by Contractor, of required submittals and the time requirements to support scheduled performance of related construction activities.
- 39. *Schedule of Values*—A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.

- 40. *Shop Drawings*—All drawings, diagrams, illustrations, schedules, and other data or information which are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work.
- 41. *Site*—Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements for access thereto, and such other lands furnished by Owner which are designated for the use of Contractor.
- 42. *Specifications*—That part of the Contract Documents consisting of written requirements for materials, equipment, systems, standards and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable thereto.
- 43. *Subcontractor*—An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work at the Site.
- 44. *Substantial Completion*—The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion thereof.
- 45. Successful Bidder—The Bidder submitting a responsive Bid to whom Owner makes an award.
- 46. *Supplementary Conditions*—That part of the Contract Documents which amends or supplements these General Conditions.
- 47. *Supplier*—A manufacturer, fabricator, supplier, distributor, materialman, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or Subcontractor.
- 48. *Underground Facilities*—All underground pipelines, conduits, ducts, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or attachments, and any encasements containing such facilities, including those that convey electricity, gases, steam, liquid petroleum products, telephone or other communications, cable television, water, wastewater, storm water, other liquids or chemicals, or traffic or other control systems.
- 49. Unit Price Work—Work to be paid for on the basis of unit prices.
- 50. *Work*—The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction, and furnishing, installing, and incorporating all materials and equipment into such construction, all as required by the Contract Documents.
- 51. Work Change Directive—A written statement to Contractor issued on or after the Effective Date of the Agreement and signed by Owner and recommended by Engineer ordering an

addition, deletion, or revision in the Work, or responding to differing or unforeseen subsurface or physical conditions under which the Work is to be performed or to emergencies. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the change ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order following negotiations by the parties as to its effect, if any, on the Contract Price or Contract Times.

1.02 Terminology

- A. The words and terms discussed in Paragraph 1.02.B through F are not defined but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning.
- B. Intent of Certain Terms or Adjectives:
 - 1. The Contract Documents include the terms "as allowed," "as approved," "as ordered," "as directed" or terms of like effect or import to authorize an exercise of professional judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to describe an action or determination of Engineer as to the Work. It is intended that such exercise of professional judgment, action, or determination will be solely to evaluate, in general, the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a functioning whole as shown or indicated in the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective is not intended to and shall not be effective to assign to Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility contrary to the provisions of Paragraph 9.09 or any other provision of the Contract Documents.

C. Day:

1. The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.

D. Defective:

- 1. The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
 - a. does not conform to the Contract Documents; or
 - b. does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
 - c. has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 14.04 or 14.05).

- E. Furnish, Install, Perform, Provide:
 - 1. The word "furnish," when used in connection with services, materials, or equipment, shall mean to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
 - 2. The word "install," when used in connection with services, materials, or equipment, shall mean to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
 - 3. The words "perform" or "provide," when used in connection with services, materials, or equipment, shall mean to furnish and install said services, materials, or equipment complete and ready for intended use.
 - 4. When "furnish," "install," "perform," or "provide" is not used in connection with services, materials, or equipment in a context clearly requiring an obligation of Contractor, "provide" is implied.
- F. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

ARTICLE 2 – PRELIMINARY MATTERS

- 2.01 Delivery of Bonds and Evidence of Insurance
 - A. When Contractor delivers the executed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner such bonds as Contractor may be required to furnish.
 - B. *Evidence of Insurance:* Before any Work at the Site is started, Contractor and Owner shall each deliver to the other, with copies to each additional insured identified in the Supplementary Conditions, certificates of insurance (and other evidence of insurance which either of them or any additional insured may reasonably request) which Contractor and Owner respectively are required to purchase and maintain in accordance with Article 5.
- 2.02 Copies of Documents
 - A. Owner shall furnish to Contractor up to ten printed or hard copies of the Drawings and Project Manual. Additional copies will be furnished upon request at the cost of reproduction.
- 2.03 Commencement of Contract Times; Notice to Proceed
 - A. The Contract Times will commence to run on the thirtieth day after the Effective Date of the Agreement or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Agreement. In no event will the Contract Times commence to run later than the sixtieth day after the day of Bid opening or the thirtieth day after the Effective Date of the Agreement, whichever date is earlier.

2.04 Starting the Work

A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work shall be done at the Site prior to the date on which the Contract Times commence to run.

2.05 Before Starting Construction

- A. *Preliminary Schedules:* Within 10 days after the Effective Date of the Agreement (unless otherwise specified in the General Requirements), Contractor shall submit to Engineer for timely review:
 - 1. a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract Documents;
 - 2. a preliminary Schedule of Submittals; and
 - 3. a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work into component parts in sufficient detail to serve as the basis for progress payments during performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.

2.06 *Preconstruction Conference; Designation of Authorized Representatives*

- A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work and to discuss the schedules referred to in Paragraph 2.05.A, procedures for handling Shop Drawings and other submittals, processing Applications for Payment, and maintaining required records.
- B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit instructions, receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.

2.07 Initial Acceptance of Schedules

- A. At least 10 days before submission of the first Application for Payment a conference attended by Contractor, Engineer, and others as appropriate will be held to review for acceptability to Engineer as provided below the schedules submitted in accordance with Paragraph 2.05.A. Contractor shall have an additional 10 days to make corrections and adjustments and to complete and resubmit the schedules. No progress payment shall be made to Contractor until acceptable schedules are submitted to Engineer.
 - 1. The Progress Schedule will be acceptable to Engineer if it provides an orderly progression of the Work to completion within the Contract Times. Such acceptance will not impose on

Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or progress of the Work, nor interfere with or relieve Contractor from Contractor's full responsibility therefor.

- 2. Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
- 3. Contractor's Schedule of Values will be acceptable to Engineer as to form and substance if it provides a reasonable allocation of the Contract Price to component parts of the Work.

ARTICLE 3 – CONTRACT DOCUMENTS: INTENT, AMENDING, REUSE

- 3.01 Intent
 - A. The Contract Documents are complementary; what is required by one is as binding as if required by all.
 - B. It is the intent of the Contract Documents to describe a functionally complete project (or part thereof) to be constructed in accordance with the Contract Documents. Any labor, documentation, services, materials, or equipment that reasonably may be inferred from the Contract Documents or from prevailing custom or trade usage as being required to produce the indicated result will be provided whether or not specifically called for, at no additional cost to Owner.
 - C. Clarifications and interpretations of the Contract Documents shall be issued by Engineer as provided in Article 9.

3.02 *Reference Standards*

- A. Standards, Specifications, Codes, Laws, and Regulations
 - 1. Reference to standards, specifications, manuals, or codes of any technical society, organization, or association, or to Laws or Regulations, whether such reference be specific or by implication, shall mean the standard, specification, manual, code, or Laws or Regulations in effect at the time of opening of Bids (or on the Effective Date of the Agreement if there were no Bids), except as may be otherwise specifically stated in the Contract Documents.
 - 2. No provision of any such standard, specification, manual, or code, or any instruction of a Supplier, shall be effective to change the duties or responsibilities of Owner, Contractor, or Engineer, or any of their subcontractors, consultants, agents, or employees, from those set forth in the Contract Documents. No such provision or instruction shall be effective to assign to Owner, Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, any duty or authority to supervise or direct the performance of the Work or any duty or authority to undertake responsibility inconsistent with the provisions of the Contract Documents.

3.03 *Reporting and Resolving Discrepancies*

- A. Reporting Discrepancies:
 - 1. *Contractor's Review of Contract Documents Before Starting Work*: Before undertaking each part of the Work, Contractor shall carefully study and compare the Contract Documents and check and verify pertinent figures therein and all applicable field measurements. Contractor shall promptly report in writing to Engineer any conflict, error, ambiguity, or discrepancy which Contractor discovers, or has actual knowledge of, and shall obtain a written interpretation or clarification from Engineer before proceeding with any Work affected thereby.
 - 2. Contractor's Review of Contract Documents During Performance of Work: If, during the performance of the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the Contract Documents, or between the Contract Documents and (a) any applicable Law or Regulation, (b) any standard, specification, manual, or code, or (c) any instruction of any Supplier, then Contractor shall promptly report it to Engineer in writing. Contractor shall not proceed with the Work affected thereby (except in an emergency as required by Paragraph 6.16.A) until an amendment or supplement to the Contract Documents has been issued by one of the methods indicated in Paragraph 3.04.
 - 3. Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof.
- B. Resolving Discrepancies:
 - 1. Except as may be otherwise specifically stated in the Contract Documents, the provisions of the Contract Documents shall take precedence in resolving any conflict, error, ambiguity, or discrepancy between the provisions of the Contract Documents and:
 - a. the provisions of any standard, specification, manual, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference in the Contract Documents); or
 - b. the provisions of any Laws or Regulations applicable to the performance of the Work (unless such an interpretation of the provisions of the Contract Documents would result in violation of such Law or Regulation).
- 3.04 *Amending and Supplementing Contract Documents*
 - A. The Contract Documents may be amended to provide for additions, deletions, and revisions in the Work or to modify the terms and conditions thereof by either a Change Order or a Work Change Directive.
 - B. The requirements of the Contract Documents may be supplemented, and minor variations and deviations in the Work may be authorized, by one or more of the following ways:

- 1. A Field Order;
- 2. Engineer's approval of a Shop Drawing or Sample (subject to the provisions of Paragraph 6.17.D.3); or
- 3. Engineer's written interpretation or clarification.

3.05 *Reuse of Documents*

- A. Contractor and any Subcontractor or Supplier shall not:
 - 1. have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media editions; or
 - 2. reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project without written consent of Owner and Engineer and specific written verification or adaptation by Engineer.
- B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein shall preclude Contractor from retaining copies of the Contract Documents for record purposes.
- 3.06 *Electronic Data*
 - A. Unless otherwise stated in the Supplementary Conditions, the data furnished by Owner or Engineer to Contractor, or by Contractor to Owner or Engineer, that may be relied upon are limited to the printed copies (also known as hard copies). Files in electronic media format of text, data, graphics, or other types are furnished only for the convenience of the receiving party. Any conclusion or information obtained or derived from such electronic files will be at the user's sole risk. If there is a discrepancy between the electronic files and the hard copies, the hard copies govern.
 - B. Because data stored in electronic media format can deteriorate or be modified inadvertently or otherwise without authorization of the data's creator, the party receiving electronic files agrees that it will perform acceptance tests or procedures within 60 days, after which the receiving party shall be deemed to have accepted the data thus transferred. Any errors detected within the 60-day acceptance period will be corrected by the transferring party.
 - C. When transferring documents in electronic media format, the transferring party makes no representations as to long term compatibility, usability, or readability of documents resulting from the use of software application packages, operating systems, or computer hardware differing from those used by the data's creator.

ARTICLE 4 – AVAILABILITY OF LANDS; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS; REFERENCE POINTS

- 4.01 Availability of Lands
 - A. Owner shall furnish the Site. Owner shall notify Contractor of any encumbrances or restrictions not of general application but specifically related to use of the Site with which Contractor must comply in performing the Work. Owner will obtain in a timely manner and pay for easements for permanent structures or permanent changes in existing facilities. If Contractor and Owner are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in the Contract Price or Contract Times, or both, as a result of any delay in Owner's furnishing the Site or a part thereof, Contractor may make a Claim therefor as provided in Paragraph 10.05.
 - B. Upon reasonable written request, Owner shall furnish Contractor with a current statement of record legal title and legal description of the lands upon which the Work is to be performed and Owner's interest therein as necessary for giving notice of or filing a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
 - C. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment.
- 4.02 Subsurface and Physical Conditions
 - A. Reports and Drawings: The Supplementary Conditions identify:
 - 1. those reports known to Owner of explorations and tests of subsurface conditions at or contiguous to the Site; and
 - 2. those drawings known to Owner of physical conditions relating to existing surface or subsurface structures at the Site (except Underground Facilities).
 - B. *Limited Reliance by Contractor on Technical Data Authorized:* Contractor may rely upon the accuracy of the "technical data" contained in such reports and drawings, but such reports and drawings are not Contract Documents. Such "technical data" is identified in the Supplementary Conditions. Except for such reliance on such "technical data," Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors with respect to:
 - 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto; or
 - 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings; or
 - 3. any Contractor interpretation of or conclusion drawn from any "technical data" or any such other data, interpretations, or information.

4.03 Differing Subsurface or Physical Conditions

- A. *Notice:* If Contractor believes that any subsurface or physical condition that is uncovered or revealed either:
 - 1. is of such a nature as to establish that any "technical data" on which Contractor is entitled to rely as provided in Paragraph 4.02 is materially inaccurate; or
 - 2. is of such a nature as to require a change in the Contract Documents; or
 - 3. differs materially from that shown or indicated in the Contract Documents; or
 - 4. is of an unusual nature, and differs materially from conditions ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents;

then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 6.16.A), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such condition or perform any Work in connection therewith (except as aforesaid) until receipt of written order to do so.

- B. *Engineer's Review*: After receipt of written notice as required by Paragraph 4.03.A, Engineer will promptly review the pertinent condition, determine the necessity of Owner's obtaining additional exploration or tests with respect thereto, and advise Owner in writing (with a copy to Contractor) of Engineer's findings and conclusions.
- C. Possible Price and Times Adjustments:
 - 1. The Contract Price or the Contract Times, or both, will be equitably adjusted to the extent that the existence of such differing subsurface or physical condition causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:
 - a. such condition must meet any one or more of the categories described in Paragraph 4.03.A; and
 - b. with respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraphs 9.07 and 11.03.
 - 2. Contractor shall not be entitled to any adjustment in the Contract Price or Contract Times if:
 - a. Contractor knew of the existence of such conditions at the time Contractor made a final commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract; or
 - b. the existence of such condition could reasonably have been discovered or revealed as a result of any examination, investigation, exploration, test, or study of the Site and
contiguous areas required by the Bidding Requirements or Contract Documents to be conducted by or for Contractor prior to Contractor's making such final commitment; or

- c. Contractor failed to give the written notice as required by Paragraph 4.03.A.
- 3. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in the Contract Price or Contract Times, or both, a Claim may be made therefor as provided in Paragraph 10.05. However, neither Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors shall be liable to Contractor for any claims, costs, losses, or damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) sustained by Contractor on or in connection with any other project or anticipated project.

4.04 Underground Facilities

- A. *Shown or Indicated:* The information and data shown or indicated in the Contract Documents with respect to existing Underground Facilities at or contiguous to the Site is based on information and data furnished to Owner or Engineer by the owners of such Underground Facilities, including Owner, or by others. Unless it is otherwise expressly provided in the Supplementary Conditions:
 - 1. Owner and Engineer shall not be responsible for the accuracy or completeness of any such information or data provided by others; and
 - 2. the cost of all of the following will be included in the Contract Price, and Contractor shall have full responsibility for:
 - a. reviewing and checking all such information and data;
 - b. locating all Underground Facilities shown or indicated in the Contract Documents;
 - c. coordination of the Work with the owners of such Underground Facilities, including Owner, during construction; and
 - d. the safety and protection of all such Underground Facilities and repairing any damage thereto resulting from the Work.
- B. Not Shown or Indicated:
 - 1. If an Underground Facility is uncovered or revealed at or contiguous to the Site which was not shown or indicated, or not shown or indicated with reasonable accuracy in the Contract Documents, Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 6.16.A), identify the owner of such Underground Facility and give written notice to that owner and to Owner and Engineer. Engineer will promptly review the Underground Facility and determine the extent, if any, to which a change is required in the Contract Documents to reflect and document the

consequences of the existence or location of the Underground Facility. During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.

2. If Engineer concludes that a change in the Contract Documents is required, a Work Change Directive or a Change Order will be issued to reflect and document such consequences. An equitable adjustment shall be made in the Contract Price or Contract Times, or both, to the extent that they are attributable to the existence or location of any Underground Facility that was not shown or indicated or not shown or indicated with reasonable accuracy in the Contract Documents and that Contractor did not know of and could not reasonably have been expected to be aware of or to have anticipated. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment in Contract Price or Contract Times, Owner or Contractor may make a Claim therefor as provided in Paragraph 10.05.

4.05 *Reference Points*

A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or requires relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or property monuments by professionally qualified personnel.

4.06 Hazardous Environmental Condition at Site

- A. *Reports and Drawings:* The Supplementary Conditions identify those reports and drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at the Site.
- B. *Limited Reliance by Contractor on Technical Data Authorized:* Contractor may rely upon the accuracy of the "technical data" contained in such reports and drawings, but such reports and drawings are not Contract Documents. Such "technical data" is identified in the Supplementary Conditions. Except for such reliance on such "technical data," Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors with respect to:
 - 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures of construction to be employed by Contractor and safety precautions and programs incident thereto; or
 - 2. other data, interpretations, opinions and information contained in such reports or shown or indicated in such drawings; or
 - 3. any Contractor interpretation of or conclusion drawn from any "technical data" or any such other data, interpretations, opinions or information.

- C. Contractor shall not be responsible for any Hazardous Environmental Condition uncovered or revealed at the Site which was not shown or indicated in Drawings or Specifications or identified in the Contract Documents to be within the scope of the Work. Contractor shall be responsible for a Hazardous Environmental Condition created with any materials brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible.
- D. If Contractor encounters a Hazardous Environmental Condition or if Contractor or anyone for whom Contractor is responsible creates a Hazardous Environmental Condition, Contractor shall immediately: (i) secure or otherwise isolate such condition; (ii) stop all Work in connection with such condition and in any area affected thereby (except in an emergency as required by Paragraph 6.16.A); and (iii) notify Owner and Engineer (and promptly thereafter confirm such notice in writing). Owner shall promptly consult with Engineer concerning the necessity for Owner to retain a qualified expert to evaluate such condition or take corrective action, if any. Promptly after consulting with Engineer, Owner shall take such actions as are necessary to permit Owner to timely obtain required permits and provide Contractor the written notice required by Paragraph 4.06.E.
- E. Contractor shall not be required to resume Work in connection with such condition or in any affected area until after Owner has obtained any required permits related thereto and delivered written notice to Contractor: (i) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work; or (ii) specifying any special conditions under which such Work may be resumed safely. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, or both, as a result of such Work stoppage or such special conditions under which Work is agreed to be resumed by Contractor, either party may make a Claim therefor as provided in Paragraph 10.05.
- F. If after receipt of such written notice Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of an adjustment in Contract Price or Contract Times as a result of deleting such portion of the Work, then either party may make a Claim therefor as provided in Paragraph 10.05. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 7.
- G. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition: (i) was not shown or indicated in the Drawings or Specifications or identified in the Contract Documents to be included within the scope of the Work, and (ii) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 4.06.G shall obligate Owner to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.

- H. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 4.06.H shall obligate Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- I. The provisions of Paragraphs 4.02, 4.03, and 4.04 do not apply to a Hazardous Environmental Condition uncovered or revealed at the Site.

ARTICLE 5 – BONDS AND INSURANCE

5.01 Performance, Payment, and Other Bonds

- A. Contractor shall furnish performance and payment bonds, each in an amount at least equal to the Contract Price as security for the faithful performance and payment of all of Contractor's obligations under the Contract Documents. These bonds shall remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 13.07, whichever is later, except as provided otherwise by Laws or Regulations or by the Contract Documents. Contractor shall also furnish such other bonds as are required by the Contract Documents.
- B. All bonds shall be in the form prescribed by the Contract Documents except as provided otherwise by Laws or Regulations, and shall be executed by such sureties as are named in the list of "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Circular 570 (amended) by the Financial Management Service, Surety Bond Branch, U.S. Department of the Treasury. All bonds signed by an agent or attorney-in-fact must be accompanied by a certified copy of that individual's authority to bind the surety. The evidence of authority shall show that it is effective on the date the agent or attorney-in-fact signed each bond.
- C. If the surety on any bond furnished by Contractor is declared bankrupt or becomes insolvent or its right to do business is terminated in any state where any part of the Project is located or it ceases to meet the requirements of Paragraph 5.01.B, Contractor shall promptly notify Owner and Engineer and shall, within 20 days after the event giving rise to such notification, provide another bond and surety, both of which shall comply with the requirements of Paragraphs 5.01.B and 5.02.

5.02 *Licensed Sureties and Insurers*

A. All bonds and insurance required by the Contract Documents to be purchased and maintained by Owner or Contractor shall be obtained from surety or insurance companies that are duly licensed or authorized in the jurisdiction in which the Project is located to issue bonds or insurance policies for the limits and coverages so required. Such surety and insurance companies shall also meet such additional requirements and qualifications as may be provided in the Supplementary Conditions.

5.03 *Certificates of Insurance*

- A. Contractor shall deliver to Owner, with copies to each additional insured and loss payee identified in the Supplementary Conditions, certificates of insurance (and other evidence of insurance requested by Owner or any other additional insured) which Contractor is required to purchase and maintain.
- B. Owner shall deliver to Contractor, with copies to each additional insured and loss payee identified in the Supplementary Conditions, certificates of insurance (and other evidence of insurance requested by Contractor or any other additional insured) which Owner is required to purchase and maintain.
- C. Failure of Owner to demand such certificates or other evidence of Contractor's full compliance with these insurance requirements or failure of Owner to identify a deficiency in compliance from the evidence provided shall not be construed as a waiver of Contractor's obligation to maintain such insurance.
- D. Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor.
- E. The insurance and insurance limits required herein shall not be deemed as a limitation on Contractor's liability under the indemnities granted to Owner in the Contract Documents.

5.04 *Contractor's Insurance*

- A. Contractor shall purchase and maintain such insurance as is appropriate for the Work being performed and as will provide protection from claims set forth below which may arise out of or result from Contractor's performance of the Work and Contractor's other obligations under the Contract Documents, whether it is to be performed by Contractor, any Subcontractor or Supplier, or by anyone directly or indirectly employed by any of them to perform any of the Work, or by anyone for whose acts any of them may be liable:
 - 1. claims under workers' compensation, disability benefits, and other similar employee benefit acts;
 - 2. claims for damages because of bodily injury, occupational sickness or disease, or death of Contractor's employees;
 - 3. claims for damages because of bodily injury, sickness or disease, or death of any person other than Contractor's employees;
 - 4. claims for damages insured by reasonably available personal injury liability coverage which are sustained:

- a. by any person as a result of an offense directly or indirectly related to the employment of such person by Contractor, or
- b. by any other person for any other reason;
- 5. claims for damages, other than to the Work itself, because of injury to or destruction of tangible property wherever located, including loss of use resulting therefrom; and
- 6. claims for damages because of bodily injury or death of any person or property damage arising out of the ownership, maintenance or use of any motor vehicle.
- B. The policies of insurance required by this Paragraph 5.04 shall:
 - 1. with respect to insurance required by Paragraphs 5.04.A.3 through 5.04.A.6 inclusive, be written on an occurrence basis, include as additional insureds (subject to any customary exclusion regarding professional liability) Owner and Engineer, and any other individuals or entities identified in the Supplementary Conditions, all of whom shall be listed as additional insureds, and include coverage for the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of all such additional insureds, and the insurance afforded to these additional insureds shall provide primary coverage for all claims covered thereby;
 - 2. include at least the specific coverages and be written for not less than the limits of liability provided in the Supplementary Conditions or required by Laws or Regulations, whichever is greater;
 - 3. include contractual liability insurance covering Contractor's indemnity obligations under Paragraphs 6.11 and 6.20;
 - 4. contain a provision or endorsement that the coverage afforded will not be canceled, materially changed or renewal refused until at least 30 days prior written notice has been given to Owner and Contractor and to each other additional insured identified in the Supplementary Conditions to whom a certificate of insurance has been issued (and the certificates of insurance furnished by the Contractor pursuant to Paragraph 5.03 will so provide);
 - 5. remain in effect at least until final payment and at all times thereafter when Contractor may be correcting, removing, or replacing defective Work in accordance with Paragraph 13.07; and
 - 6. include completed operations coverage:
 - a. Such insurance shall remain in effect for two years after final payment.
 - b. Contractor shall furnish Owner and each other additional insured identified in the Supplementary Conditions, to whom a certificate of insurance has been issued, evidence satisfactory to Owner and any such additional insured of continuation of such insurance at final payment and one year thereafter.

5.05 Owner's Liability Insurance

A. In addition to the insurance required to be provided by Contractor under Paragraph 5.04, Owner, at Owner's option, may purchase and maintain at Owner's expense Owner's own liability insurance as will protect Owner against claims which may arise from operations under the Contract Documents.

5.06 *Property Insurance*

- A. Unless otherwise provided in the Supplementary Conditions, Owner shall purchase and maintain property insurance upon the Work at the Site in the amount of the full replacement cost thereof (subject to such deductible amounts as may be provided in the Supplementary Conditions or required by Laws and Regulations). This insurance shall:
 - 1. include the interests of Owner, Contractor, Subcontractors, and Engineer, and any other individuals or entities identified in the Supplementary Conditions, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, each of whom is deemed to have an insurable interest and shall be listed as a loss payee;
 - 2. be written on a Builder's Risk "all-risk" policy form that shall at least include insurance for physical loss or damage to the Work, temporary buildings, falsework, and materials and equipment in transit, and shall insure against at least the following perils or causes of loss: fire, lightning, extended coverage, theft, vandalism and malicious mischief, earthquake, collapse, debris removal, demolition occasioned by enforcement of Laws and Regulations, water damage (other than that caused by flood), and such other perils or causes of loss as may be specifically required by the Supplementary Conditions.
 - 3. include expenses incurred in the repair or replacement of any insured property (including but not limited to fees and charges of engineers and architects);
 - 4. cover materials and equipment stored at the Site or at another location that was agreed to in writing by Owner prior to being incorporated in the Work, provided that such materials and equipment have been included in an Application for Payment recommended by Engineer;
 - 5. allow for partial utilization of the Work by Owner;
 - 6. include testing and startup; and
 - 7. be maintained in effect until final payment is made unless otherwise agreed to in writing by Owner, Contractor, and Engineer with 30 days written notice to each other loss payee to whom a certificate of insurance has been issued.
- B. Owner shall purchase and maintain such equipment breakdown insurance or additional property insurance as may be required by the Supplementary Conditions or Laws and Regulations which will include the interests of Owner, Contractor, Subcontractors, and Engineer, and any other individuals or entities identified in the Supplementary Conditions, and the officers, directors,

members, partners, employees, agents, consultants and subcontractors of each and any of them, each of whom is deemed to have an insurable interest and shall be listed as a loss payee.

- C. All the policies of insurance (and the certificates or other evidence thereof) required to be purchased and maintained in accordance with this Paragraph 5.06 will contain a provision or endorsement that the coverage afforded will not be canceled or materially changed or renewal refused until at least 30 days prior written notice has been given to Owner and Contractor and to each other loss payee to whom a certificate of insurance has been issued and will contain waiver provisions in accordance with Paragraph 5.07.
- D. Owner shall not be responsible for purchasing and maintaining any property insurance specified in this Paragraph 5.06 to protect the interests of Contractor, Subcontractors, or others in the Work to the extent of any deductible amounts that are identified in the Supplementary Conditions. The risk of loss within such identified deductible amount will be borne by Contractor, Subcontractors, or others suffering any such loss, and if any of them wishes property insurance coverage within the limits of such amounts, each may purchase and maintain it at the purchaser's own expense.
- E. If Contractor requests in writing that other special insurance be included in the property insurance policies provided under this Paragraph 5.06, Owner shall, if possible, include such insurance, and the cost thereof will be charged to Contractor by appropriate Change Order. Prior to commencement of the Work at the Site, Owner shall in writing advise Contractor whether or not such other insurance has been procured by Owner.

5.07 Waiver of Rights

- A. Owner and Contractor intend that all policies purchased in accordance with Paragraph 5.06 will protect Owner, Contractor, Subcontractors, and Engineer, and all other individuals or entities identified in the Supplementary Conditions as loss payees (and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them) in such policies and will provide primary coverage for all losses and damages caused by the perils or causes of loss covered thereby. All such policies shall contain provisions to the effect that in the event of payment of any loss or damage the insurers will have no rights of recovery against any of the insureds or loss payees thereunder. Owner and Contractor waive all rights against each other and their respective officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them for all losses and damages caused by, arising out of or resulting from any of the perils or causes of loss covered by such policies and any other property insurance applicable to the Work; and, in addition, waive all such rights against Subcontractors and Engineer, and all other individuals or entities identified in the Supplementary Conditions as loss payees (and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them) under such policies for losses and damages so caused. None of the above waivers shall extend to the rights that any party making such waiver may have to the proceeds of insurance held by Owner as trustee or otherwise payable under any policy so issued.
- B. Owner waives all rights against Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them for:

- 1. loss due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other perils whether or not insured by Owner; and
- 2. loss or damage to the completed Project or part thereof caused by, arising out of, or resulting from fire or other insured peril or cause of loss covered by any property insurance maintained on the completed Project or part thereof by Owner during partial utilization pursuant to Paragraph 14.05, after Substantial Completion pursuant to Paragraph 14.04, or after final payment pursuant to Paragraph 14.07.
- C. Any insurance policy maintained by Owner covering any loss, damage or consequential loss referred to in Paragraph 5.07.B shall contain provisions to the effect that in the event of payment of any such loss, damage, or consequential loss, the insurers will have no rights of recovery against Contractor, Subcontractors, or Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them.

5.08 Receipt and Application of Insurance Proceeds

- A. Any insured loss under the policies of insurance required by Paragraph 5.06 will be adjusted with Owner and made payable to Owner as fiduciary for the loss payees, as their interests may appear, subject to the requirements of any applicable mortgage clause and of Paragraph 5.08.B. Owner shall deposit in a separate account any money so received and shall distribute it in accordance with such agreement as the parties in interest may reach. If no other special agreement is reached, the damaged Work shall be repaired or replaced, the moneys so received applied on account thereof, and the Work and the cost thereof covered by an appropriate Change Order.
- B. Owner as fiduciary shall have power to adjust and settle any loss with the insurers unless one of the parties in interest shall object in writing within 15 days after the occurrence of loss to Owner's exercise of this power. If such objection be made, Owner as fiduciary shall make settlement with the insurers in accordance with such agreement as the parties in interest may reach. If no such agreement among the parties in interest is reached, Owner as fiduciary shall adjust and settle the loss with the insurers and, if required in writing by any party in interest, Owner as fiduciary shall give bond for the proper performance of such duties.

5.09 Acceptance of Bonds and Insurance; Option to Replace

A. If either Owner or Contractor has any objection to the coverage afforded by or other provisions of the bonds or insurance required to be purchased and maintained by the other party in accordance with Article 5 on the basis of non-conformance with the Contract Documents, the objecting party shall so notify the other party in writing within 10 days after receipt of the certificates (or other evidence requested) required by Paragraph 2.01.B. Owner and Contractor shall each provide to the other such additional information in respect of insurance provided as the other may reasonably request. If either party does not purchase or maintain all of the bonds and insurance required of such party by the Contract Documents, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage. Without prejudice to any other right or remedy, the other party may elect to obtain equivalent bonds or insurance to protect such other party's

interests at the expense of the party who was required to provide such coverage, and a Change Order shall be issued to adjust the Contract Price accordingly.

5.10 Partial Utilization, Acknowledgment of Property Insurer

A. If Owner finds it necessary to occupy or use a portion or portions of the Work prior to Substantial Completion of all the Work as provided in Paragraph 14.05, no such use or occupancy shall commence before the insurers providing the property insurance pursuant to Paragraph 5.06 have acknowledged notice thereof and in writing effected any changes in coverage necessitated thereby. The insurers providing the property insurance shall consent by endorsement on the policy or policies, but the property insurance shall not be canceled or permitted to lapse on account of any such partial use or occupancy.

ARTICLE 6 – CONTRACTOR'S RESPONSIBILITIES

6.01 Supervision and Superintendence

- A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work in accordance with the Contract Documents. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction. Contractor shall not be responsible for the negligence of Owner or Engineer in the design or specification of a specific means, method, technique, sequence, or procedure of construction which is shown or indicated in and expressly required by the Contract Documents.
- B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who shall not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.

6.02 Labor; Working Hours

- A. Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall at all times maintain good discipline and order at the Site.
- B. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent thereto, and except as otherwise stated in the Contract Documents, all Work at the Site shall be performed during regular working hours. Contractor will not permit the performance of Work on a Saturday, Sunday, or any legal holiday without Owner's written consent (which will not be unreasonably withheld) given after prior written notice to Engineer.

6.03 Services, Materials, and Equipment

A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start-up, and completion of the Work.

- B. All materials and equipment incorporated into the Work shall be as specified or, if not specified, shall be of good quality and new, except as otherwise provided in the Contract Documents. All special warranties and guarantees required by the Specifications shall expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.
- C. All materials and equipment shall be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.

6.04 *Progress Schedule*

- A. Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.07 as it may be adjusted from time to time as provided below.
 - 1. Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.07) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times. Such adjustments will comply with any provisions of the General Requirements applicable thereto.
 - 2. Proposed adjustments in the Progress Schedule that will change the Contract Times shall be submitted in accordance with the requirements of Article 12. Adjustments in Contract Times may only be made by a Change Order.
- 6.05 Substitutes and "Or-Equals"
 - A. Whenever an item of material or equipment is specified or described in the Contract Documents by using the name of a proprietary item or the name of a particular Supplier, the specification or description is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or-equal" item or no substitution is permitted, other items of material or equipment or material or equipment of other Suppliers may be submitted to Engineer for review under the circumstances described below.
 - 1. "*Or-Equal*" *Items:* If in Engineer's sole discretion an item of material or equipment proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, it may be considered by Engineer as an "or-equal" item, in which case review and approval of the proposed item may, in Engineer's sole discretion, be accomplished without compliance with some or all of the requirements for approval of proposed substitute items. For the purposes of this Paragraph 6.05.A.1, a proposed item of material or equipment will be considered functionally equal to an item so named if:
 - a. in the exercise of reasonable judgment Engineer determines that:
 - 1) it is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;

- 2) it will reliably perform at least equally well the function and achieve the results imposed by the design concept of the completed Project as a functioning whole; and
- 3) it has a proven record of performance and availability of responsive service.
- b. Contractor certifies that, if approved and incorporated into the Work:
 - 1) there will be no increase in cost to the Owner or increase in Contract Times; and
 - 2) it will conform substantially to the detailed requirements of the item named in the Contract Documents.
- 2. Substitute Items:
 - a. If in Engineer's sole discretion an item of material or equipment proposed by Contractor does not qualify as an "or-equal" item under Paragraph 6.05.A.1, it will be considered a proposed substitute item.
 - b. Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is essentially equivalent to that named and an acceptable substitute therefor. Requests for review of proposed substitute items of material or equipment will not be accepted by Engineer from anyone other than Contractor.
 - c. The requirements for review by Engineer will be as set forth in Paragraph 6.05.A.2.d, as supplemented by the General Requirements, and as Engineer may decide is appropriate under the circumstances.
 - d. Contractor shall make written application to Engineer for review of a proposed substitute item of material or equipment that Contractor seeks to furnish or use. The application:
 - 1) shall certify that the proposed substitute item will:
 - a) perform adequately the functions and achieve the results called for by the general design,
 - b) be similar in substance to that specified, and
 - c) be suited to the same use as that specified;
 - 2) will state:
 - a) the extent, if any, to which the use of the proposed substitute item will prejudice Contractor's achievement of Substantial Completion on time,
 - b) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item, and

- c) whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty;
- 3) will identify:
 - a) all variations of the proposed substitute item from that specified, and
 - b) available engineering, sales, maintenance, repair, and replacement services; and
- 4) shall contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including costs of redesign and claims of other contractors affected by any resulting change.
- B. *Substitute Construction Methods or Procedures:* If a specific means, method, technique, sequence, or procedure of construction is expressly required by the Contract Documents, Contractor may furnish or utilize a substitute means, method, technique, sequence, or procedure of construction approved by Engineer. Contractor shall submit sufficient information to allow Engineer, in Engineer's sole discretion, to determine that the substitute proposed is equivalent to that expressly called for by the Contract Documents. The requirements for review by Engineer will be similar to those provided in Paragraph 6.05.A.2.
- C. *Engineer's Evaluation:* Engineer will be allowed a reasonable time within which to evaluate each proposal or submittal made pursuant to Paragraphs 6.05.A and 6.05.B. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No "or equal" or substitute will be ordered, installed or utilized until Engineer's review is complete, which will be evidenced by a Change Order in the case of a substitute and an approved Shop Drawing for an "or equal." Engineer will advise Contractor in writing of any negative determination.
- D. *Special Guarantee:* Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- E. *Engineer's Cost Reimbursement*: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor pursuant to Paragraphs 6.05.A.2 and 6.05.B. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for the reasonable charges of Engineer for waking changes in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.
- F. *Contractor's Expense*: Contractor shall provide all data in support of any proposed substitute or "or-equal" at Contractor's expense.
- 6.06 *Concerning Subcontractors, Suppliers, and Others*
 - A. Contractor shall not employ any Subcontractor, Supplier, or other individual or entity (including those acceptable to Owner as indicated in Paragraph 6.06.B), whether initially or as a replacement, against whom Owner may have reasonable objection. Contractor shall not be

required to employ any Subcontractor, Supplier, or other individual or entity to furnish or perform any of the Work against whom Contractor has reasonable objection.

- B. If the Supplementary Conditions require the identity of certain Subcontractors, Suppliers, or other individuals or entities to be submitted to Owner in advance for acceptance by Owner by a specified date prior to the Effective Date of the Agreement, and if Contractor has submitted a list thereof in accordance with the Supplementary Conditions, Owner's acceptance (either in writing or by failing to make written objection thereto by the date indicated for acceptance or objection in the Bidding Documents or the Contract Documents) of any such Subcontractor, Supplier, or other individual or entity so identified may be revoked on the basis of reasonable objection after due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor, Supplier, or other individual or entity, and the Contract Price will be adjusted by the difference in the cost occasioned by such replacement, and an appropriate Change Order will be issued. No acceptance by Owner of any such Subcontractor, Supplier, or other individual or entity, whether initially or as a replacement, shall constitute a waiver of any right of Owner or Engineer to reject defective Work.
- C. Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of the Subcontractors, Suppliers, and other individuals or entities performing or furnishing any of the Work just as Contractor is responsible for Contractor's own acts and omissions. Nothing in the Contract Documents:
 - 1. shall create for the benefit of any such Subcontractor, Supplier, or other individual or entity any contractual relationship between Owner or Engineer and any such Subcontractor, Supplier or other individual or entity; nor
 - 2. shall create any obligation on the part of Owner or Engineer to pay or to see to the payment of any moneys due any such Subcontractor, Supplier, or other individual or entity except as may otherwise be required by Laws and Regulations.
- D. Contractor shall be solely responsible for scheduling and coordinating the Work of Subcontractors, Suppliers, and other individuals or entities performing or furnishing any of the Work under a direct or indirect contract with Contractor.
- E. Contractor shall require all Subcontractors, Suppliers, and such other individuals or entities performing or furnishing any of the Work to communicate with Engineer through Contractor.
- F. The divisions and sections of the Specifications and the identifications of any Drawings shall not control Contractor in dividing the Work among Subcontractors or Suppliers or delineating the Work to be performed by any specific trade.
- G. All Work performed for Contractor by a Subcontractor or Supplier will be pursuant to an appropriate agreement between Contractor and the Subcontractor or Supplier which specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract Documents for the benefit of Owner and Engineer. Whenever any such agreement is with a Subcontractor or Supplier who is listed as a loss payee on the property insurance provided in Paragraph 5.06, the agreement between the Contractor and the Subcontractor or Supplier will contain provisions whereby the Subcontractor or Supplier waives all rights against Owner,

Contractor, Engineer, and all other individuals or entities identified in the Supplementary Conditions to be listed as insureds or loss payees (and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them) for all losses and damages caused by, arising out of, relating to, or resulting from any of the perils or causes of loss covered by such policies and any other property insurance applicable to the Work. If the insurers on any such policies require separate waiver forms to be signed by any Subcontractor or Supplier, Contractor will obtain the same.

6.07 Patent Fees and Royalties

- A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If a particular invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Owner or Engineer, its use is subject to patent rights or copyrights calling for the payment of any license fee or royalty to others, the existence of such rights shall be disclosed by Owner in the Contract Documents.
- B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights.
- C. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.

6.08 Permits

A. Unless otherwise provided in the Supplementary Conditions, Contractor shall obtain and pay for all construction permits and licenses. Owner shall assist Contractor, when necessary, in obtaining such permits and licenses. Contractor shall pay all governmental charges and inspection fees necessary for the prosecution of the Work which are applicable at the time of opening of Bids, or, if there are no Bids, on the Effective Date of the Agreement. Owner shall pay all charges of utility owners for connections for providing permanent service to the Work.

6.09 Laws and Regulations

- A. Contractor shall give all notices required by and shall comply with all Laws and Regulations applicable to the performance of the Work. Except where otherwise expressly required by applicable Laws and Regulations, neither Owner nor Engineer shall be responsible for monitoring Contractor's compliance with any Laws or Regulations.
- B. If Contractor performs any Work knowing or having reason to know that it is contrary to Laws or Regulations, Contractor shall bear all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such Work. However, it shall not be Contractor's responsibility to make certain that the Specifications and Drawings are in accordance with Laws and Regulations, but this shall not relieve Contractor of Contractor's obligations under Paragraph 3.03.
- C. Changes in Laws or Regulations not known at the time of opening of Bids (or, on the Effective Date of the Agreement if there were no Bids) having an effect on the cost or time of performance of the Work shall be the subject of an adjustment in Contract Price or Contract Times. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment, a Claim may be made therefor as provided in Paragraph 10.05.

6.10 *Taxes*

A. Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

6.11 Use of Site and Other Areas

A. Limitation on Use of Site and Other Areas:

- 1. Contractor shall confine construction equipment, the storage of materials and equipment, and the operations of workers to the Site and other areas permitted by Laws and Regulations, and shall not unreasonably encumber the Site and other areas with construction equipment or other materials or equipment. Contractor shall assume full responsibility for any damage to any such land or area, or to the owner or occupant thereof, or of any adjacent land or areas resulting from the performance of the Work.
- 2. Should any claim be made by any such owner or occupant because of the performance of the Work, Contractor shall promptly settle with such other party by negotiation or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law.
- 3. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought

by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused by or based upon Contractor's performance of the Work.

- B. *Removal of Debris During Performance of the Work:* During the progress of the Work Contractor shall keep the Site and other areas free from accumulations of waste materials, rubbish, and other debris. Removal and disposal of such waste materials, rubbish, and other debris shall conform to applicable Laws and Regulations.
- C. *Cleaning:* Prior to Substantial Completion of the Work Contractor shall clean the Site and the Work and make it ready for utilization by Owner. At the completion of the Work Contractor shall remove from the Site all tools, appliances, construction equipment and machinery, and surplus materials and shall restore to original condition all property not designated for alteration by the Contract Documents.
- D. *Loading Structures:* Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall Contractor subject any part of the Work or adjacent property to stresses or pressures that will endanger it.

6.12 Record Documents

A. Contractor shall maintain in a safe place at the Site one record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, and written interpretations and clarifications in good order and annotated to show changes made during construction. These record documents together with all approved Samples and a counterpart of all approved Shop Drawings will be available to Engineer for reference. Upon completion of the Work, these record documents, Samples, and Shop Drawings will be delivered to Engineer for Owner.

6.13 Safety and Protection

- A. Contractor shall be solely responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the Work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety Laws and Regulations. Contractor shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury or loss to:
 - 1. all persons on the Site or who may be affected by the Work;
 - 2. all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and
 - 3. other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, utilities, and Underground Facilities not designated for removal, relocation, or replacement in the course of construction.
- B. Contractor shall comply with all applicable Laws and Regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and

shall erect and maintain all necessary safeguards for such safety and protection. Contractor shall notify owners of adjacent property and of Underground Facilities and other utility owners when prosecution of the Work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property.

- C. Contractor shall comply with the applicable requirements of Owner's safety programs, if any. The Supplementary Conditions identify any Owner's safety programs that are applicable to the Work.
- D. Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.
- E. All damage, injury, or loss to any property referred to in Paragraph 6.13.A.2 or 6.13.A.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor (except damage or loss attributable to the fault of Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone employed by any of them, or anyone for whose acts any of them may be liable, and not attributable, directly or indirectly, in whole or in part, to the fault or negligence of Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly employed by any of them).
- F. Contractor's duties and responsibilities for safety and for protection of the Work shall continue until such time as all the Work is completed and Engineer has issued a notice to Owner and Contractor in accordance with Paragraph 14.07.B that the Work is acceptable (except as otherwise expressly provided in connection with Substantial Completion).

6.14 Safety Representative

A. Contractor shall designate a qualified and experienced safety representative at the Site whose duties and responsibilities shall be the prevention of accidents and the maintaining and supervising of safety precautions and programs.

6.15 Hazard Communication Programs

A. Contractor shall be responsible for coordinating any exchange of material safety data sheets or other hazard communication information required to be made available to or exchanged between or among employers at the Site in accordance with Laws or Regulations.

6.16 *Emergencies*

A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent threatened damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused thereby or are required as a result thereof. If Engineer determines that a change in the Contract Documents is

required because of the action taken by Contractor in response to such an emergency, a Work Change Directive or Change Order will be issued.

- 6.17 *Shop Drawings and Samples*
 - A. Contractor shall submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals (as required by Paragraph 2.07). Each submittal will be identified as Engineer may require.
 - 1. Shop Drawings:
 - a. Submit number of copies specified in the General Requirements.
 - b. Data shown on the Shop Drawings will be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to provide and to enable Engineer to review the information for the limited purposes required by Paragraph 6.17.D.
 - 2. Samples:
 - a. Submit number of Samples specified in the Specifications.
 - b. Clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the submittal for the limited purposes required by Paragraph 6.17.D.
 - B. Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole expense and responsibility of Contractor.
 - C. Submittal Procedures:
 - 1. Before submitting each Shop Drawing or Sample, Contractor shall have:
 - a. reviewed and coordinated each Shop Drawing or Sample with other Shop Drawings and Samples and with the requirements of the Work and the Contract Documents;
 - b. determined and verified all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect thereto;
 - c. determined and verified the suitability of all materials offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and
 - d. determined and verified all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto.

- 2. Each submittal shall bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review and approval of that submittal.
- 3. With each submittal, Contractor shall give Engineer specific written notice of any variations that the Shop Drawing or Sample may have from the requirements of the Contract Documents. This notice shall be both a written communication separate from the Shop Drawings or Sample submittal; and, in addition, by a specific notation made on each Shop Drawing or Sample submitted to Engineer for review and approval of each such variation.

D. Engineer's Review:

- 1. Engineer will provide timely review of Shop Drawings and Samples in accordance with the Schedule of Submittals acceptable to Engineer. Engineer's review and approval will be only to determine if the items covered by the submittals will, after installation or incorporation in the Work, conform to the information given in the Contract Documents and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents.
- 2. Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction (except where a particular means, method, technique, sequence, or procedure of construction is specifically and expressly called for by the Contract Documents) or to safety precautions or programs incident thereto. The review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.
- 3. Engineer's review and approval shall not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 6.17.C.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer's review and approval shall not relieve Contractor from responsibility for complying with the requirements of Paragraph 6.17.C.1.

E. Resubmittal Procedures:

1. Contractor shall make corrections required by Engineer and shall return the required number of corrected copies of Shop Drawings and submit, as required, new Samples for review and approval. Contractor shall direct specific attention in writing to revisions other than the corrections called for by Engineer on previous submittals.

6.18 *Continuing the Work*

A. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work shall be delayed or postponed pending resolution of any disputes or disagreements, except as permitted by Paragraph 15.04 or as Owner and Contractor may otherwise agree in writing.

6.19 Contractor's General Warranty and Guarantee

- A. Contractor warrants and guarantees to Owner that all Work will be in accordance with the Contract Documents and will not be defective. Engineer and its officers, directors, members, partners, employees, agents, consultants, and subcontractors shall be entitled to rely on representation of Contractor's warranty and guarantee.
- B. Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
 - 1. abuse, modification, or improper maintenance or operation by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
 - 2. normal wear and tear under normal usage.
- C. Contractor's obligation to perform and complete the Work in accordance with the Contract Documents shall be absolute. None of the following will constitute an acceptance of Work that is not in accordance with the Contract Documents or a release of Contractor's obligation to perform the Work in accordance with the Contract Documents:
 - 1. observations by Engineer;
 - 2. recommendation by Engineer or payment by Owner of any progress or final payment;
 - 3. the issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
 - 4. use or occupancy of the Work or any part thereof by Owner;
 - 5. any review and approval of a Shop Drawing or Sample submittal or the issuance of a notice of acceptability by Engineer;
 - 6. any inspection, test, or approval by others; or
 - 7. any correction of defective Work by Owner.

6.20 *Indemnification*

A. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the performance of the Work, provided that any such claim, cost, loss, or damage is attributable to bodily injury, sickness, disease, or death, or to injury to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom but only to the extent caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work or anyone for whose acts any of them may be liable .

- B. In any and all claims against Owner or Engineer or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors by any employee (or the survivor or personal representative of such employee) of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, the indemnification obligation under Paragraph 6.20.A shall not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' compensation acts, disability benefit acts, or other employee benefit acts.
- C. The indemnification obligations of Contractor under Paragraph 6.20.A shall not extend to the liability of Engineer and Engineer's officers, directors, members, partners, employees, agents, consultants and subcontractors arising out of:
 - 1. the preparation or approval of, or the failure to prepare or approve maps, Drawings, opinions, reports, surveys, Change Orders, designs, or Specifications; or
 - 2. giving directions or instructions, or failing to give them, if that is the primary cause of the injury or damage.
- 6.21 Delegation of Professional Design Services
 - A. Contractor will not be required to provide professional design services unless such services are specifically required by the Contract Documents for a portion of the Work or unless such services are required to carry out Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. Contractor shall not be required to provide professional services in violation of applicable law.
 - B. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of Contractor by the Contract Documents, Owner and Engineer will specify all performance and design criteria that such services must satisfy. Contractor shall cause such services or certifications to be provided by a properly licensed professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to Engineer.
 - C. Owner and Engineer shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications or approvals performed by such design professionals, provided Owner and Engineer have specified to Contractor all performance and design criteria that such services must satisfy.
 - D. Pursuant to this Paragraph 6.21, Engineer's review and approval of design calculations and design drawings will be only for the limited purpose of checking for conformance with performance and design criteria given and the design concept expressed in the Contract Documents. Engineer's review and approval of Shop Drawings and other submittals (except design calculations and design drawings) will be only for the purpose stated in Paragraph 6.17.D.1.

E. Contractor shall not be responsible for the adequacy of the performance or design criteria required by the Contract Documents.

ARTICLE 7 – OTHER WORK AT THE SITE

- 7.01 *Related Work at Site*
 - A. Owner may perform other work related to the Project at the Site with Owner's employees, or through other direct contracts therefor, or have other work performed by utility owners. If such other work is not noted in the Contract Documents, then:
 - 1. written notice thereof will be given to Contractor prior to starting any such other work; and
 - 2. if Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in the Contract Price or Contract Times that should be allowed as a result of such other work, a Claim may be made therefor as provided in Paragraph 10.05.
 - B. Contractor shall afford each other contractor who is a party to such a direct contract, each utility owner, and Owner, if Owner is performing other work with Owner's employees, proper and safe access to the Site, provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work, and properly coordinate the Work with theirs. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose work will be affected. The duties and responsibilities of Contractor under this Paragraph are for the benefit of such utility owners and other contractors between Owner and such utility owners and other contractors.
 - C. If the proper execution or results of any part of Contractor's Work depends upon work performed by others under this Article 7, Contractor shall inspect such other work and promptly report to Engineer in writing any delays, defects, or deficiencies in such other work that render it unavailable or unsuitable for the proper execution and results of Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent defects and deficiencies in such other work.

7.02 *Coordination*

- A. If Owner intends to contract with others for the performance of other work on the Project at the Site, the following will be set forth in Supplementary Conditions:
 - 1. the individual or entity who will have authority and responsibility for coordination of the activities among the various contractors will be identified;
 - 2. the specific matters to be covered by such authority and responsibility will be itemized; and
 - 3. the extent of such authority and responsibilities will be provided.

EJCDC C-700 Standard General Conditions of the Construction Contract
Copyright © 2007 National Society of Professional Engineers for EJCDC. All rights reserved.
Page 35 of 62

- B. Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination.
- 7.03 Legal Relationships
 - A. Paragraphs 7.01.A and 7.02 are not applicable for utilities not under the control of Owner.
 - B. Each other direct contract of Owner under Paragraph 7.01.A shall provide that the other contractor is liable to Owner and Contractor for the reasonable direct delay and disruption costs incurred by Contractor as a result of the other contractor's wrongful actions or inactions.
 - C. Contractor shall be liable to Owner and any other contractor under direct contract to Owner for the reasonable direct delay and disruption costs incurred by such other contractor as a result of Contractor's wrongful action or inactions.

ARTICLE 8 – OWNER'S RESPONSIBILITIES

- 8.01 Communications to Contractor
 - A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.
- 8.02 Replacement of Engineer
 - A. In case of termination of the employment of Engineer, Owner shall appoint an engineer to whom Contractor makes no reasonable objection, whose status under the Contract Documents shall be that of the former Engineer.
- 8.03 Furnish Data
 - A. Owner shall promptly furnish the data required of Owner under the Contract Documents.
- 8.04 Pay When Due
 - A. Owner shall make payments to Contractor when they are due as provided in Paragraphs 14.02.C and 14.07.C.
- 8.05 Lands and Easements; Reports and Tests
 - A. Owner's duties with respect to providing lands and easements and providing engineering surveys to establish reference points are set forth in Paragraphs 4.01 and 4.05. Paragraph 4.02 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of subsurface conditions and drawings of physical conditions relating to existing surface or subsurface structures at the Site.
- 8.06 *Insurance*
 - A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 5.

8.07 Change Orders

- A. Owner is obligated to execute Change Orders as indicated in Paragraph 10.03.
- 8.08 Inspections, Tests, and Approvals
 - A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 13.03.B.
- 8.09 Limitations on Owner's Responsibilities
 - A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- 8.10 Undisclosed Hazardous Environmental Condition
 - A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 4.06.
- 8.11 Evidence of Financial Arrangements
 - A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract Documents.
- 8.12 Compliance with Safety Program
 - A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed pursuant to Paragraph 6.13.D.

ARTICLE 9 – ENGINEER'S STATUS DURING CONSTRUCTION

- 9.01 Owner's Representative
 - A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract Documents.
- 9.02 Visits to Site
 - A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe as an experienced and qualified design professional the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or

continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform generally to the Contract Documents. On the basis of such visits and observations, Engineer will keep Owner informed of the progress of the Work and will endeavor to guard Owner against defective Work.

B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 9.09. Particularly, but without limitation, during or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

9.03 *Project Representative*

A. If Owner and Engineer agree, Engineer will furnish a Resident Project Representative to assist Engineer in providing more extensive observation of the Work. The authority and responsibilities of any such Resident Project Representative and assistants will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in Paragraph 9.09. If Owner designates another representative or agent to represent Owner at the Site who is not Engineer's consultant, agent or employee, the responsibilities and authority and limitations thereon of such other individual or entity will be as provided in the Supplementary Conditions.

9.04 Authorized Variations in Work

A. Engineer may authorize minor variations in the Work from the requirements of the Contract Documents which do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. These may be accomplished by a Field Order and will be binding on Owner and also on Contractor, who shall perform the Work involved promptly. If Owner or Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, or both, and the parties are unable to agree on entitlement to or on the amount or extent, if any, of any such adjustment, a Claim may be made therefor as provided in Paragraph 10.05.

9.05 *Rejecting Defective Work*

A. Engineer will have authority to reject Work which Engineer believes to be defective, or that Engineer believes will not produce a completed Project that conforms to the Contract Documents or that will prejudice the integrity of the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Engineer will also have authority to require special inspection or testing of the Work as provided in Paragraph 13.04, whether or not the Work is fabricated, installed, or completed.

- A. In connection with Engineer's authority, and limitations thereof, as to Shop Drawings and Samples, see Paragraph 6.17.
- B. In connection with Engineer's authority, and limitations thereof, as to design calculations and design drawings submitted in response to a delegation of professional design services, if any, see Paragraph 6.21.
- C. In connection with Engineer's authority as to Change Orders, see Articles 10, 11, and 12.
- D. In connection with Engineer's authority as to Applications for Payment, see Article 14.
- 9.07 Determinations for Unit Price Work
 - A. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision thereon will be final and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, subject to the provisions of Paragraph 10.05.
- 9.08 Decisions on Requirements of Contract Documents and Acceptability of Work
 - A. Engineer will be the initial interpreter of the requirements of the Contract Documents and judge of the acceptability of the Work thereunder. All matters in question and other matters between Owner and Contractor arising prior to the date final payment is due relating to the acceptability of the Work, and the interpretation of the requirements of the Contract Documents pertaining to the performance of the Work, will be referred initially to Engineer in writing within 30 days of the event giving rise to the question.
 - B. Engineer will, with reasonable promptness, render a written decision on the issue referred. If Owner or Contractor believes that any such decision entitles them to an adjustment in the Contract Price or Contract Times or both, a Claim may be made under Paragraph 10.05. The date of Engineer's decision shall be the date of the event giving rise to the issues referenced for the purposes of Paragraph 10.05.B.
 - C. Engineer's written decision on the issue referred will be final and binding on Owner and Contractor, subject to the provisions of Paragraph 10.05.
 - D. When functioning as interpreter and judge under this Paragraph 9.08, Engineer will not show partiality to Owner or Contractor and will not be liable in connection with any interpretation or decision rendered in good faith in such capacity.
- 9.09 Limitations on Engineer's Authority and Responsibilities
 - A. Neither Engineer's authority or responsibility under this Article 9 or under any other provision of the Contract Documents nor any decision made by Engineer in good faith either to exercise or not

exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer shall create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any surety for or employee or agent of any of them.

- B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
- D. Engineer's review of the final Application for Payment and accompanying documentation and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Paragraph 14.07.A will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals that the results certified indicate compliance with, the Contract Documents.
- E. The limitations upon authority and responsibility set forth in this Paragraph 9.09 shall also apply to the Resident Project Representative, if any, and assistants, if any.
- 9.10 Compliance with Safety Program
 - A. While at the Site, Engineer's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Engineer has been informed pursuant to Paragraph 6.13.D.

ARTICLE 10 - CHANGES IN THE WORK; CLAIMS

- 10.01 Authorized Changes in the Work
 - A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work by a Change Order, or a Work Change Directive. Upon receipt of any such document, Contractor shall promptly proceed with the Work involved which will be performed under the applicable conditions of the Contract Documents (except as otherwise specifically provided).
 - B. If Owner and Contractor are unable to agree on entitlement to, or on the amount or extent, if any, of an adjustment in the Contract Price or Contract Times, or both, that should be allowed as a result of a Work Change Directive, a Claim may be made therefor as provided in Paragraph 10.05.

10.02 Unauthorized Changes in the Work

A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents as amended, modified, or supplemented as provided in Paragraph 3.04, except in the case of an emergency as provided in Paragraph 6.16 or in the case of uncovering Work as provided in Paragraph 13.04.D.

10.03 Execution of Change Orders

- A. Owner and Contractor shall execute appropriate Change Orders recommended by Engineer covering:
 - 1. changes in the Work which are: (i) ordered by Owner pursuant to Paragraph 10.01.A, (ii) required because of acceptance of defective Work under Paragraph 13.08.A or Owner's correction of defective Work under Paragraph 13.09, or (iii) agreed to by the parties;
 - 2. changes in the Contract Price or Contract Times which are agreed to by the parties, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive; and
 - 3. changes in the Contract Price or Contract Times which embody the substance of any written decision rendered by Engineer pursuant to Paragraph 10.05; provided that, in lieu of executing any such Change Order, an appeal may be taken from any such decision in accordance with the provisions of the Contract Documents and applicable Laws and Regulations, but during any such appeal, Contractor shall carry on the Work and adhere to the Progress Schedule as provided in Paragraph 6.18.A.

10.04 Notification to Surety

A. If the provisions of any bond require notice to be given to a surety of any change affecting the general scope of the Work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Times), the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

10.05 Claims

- A. *Engineer's Decision Required*: All Claims, except those waived pursuant to Paragraph 14.09, shall be referred to the Engineer for decision. A decision by Engineer shall be required as a condition precedent to any exercise by Owner or Contractor of any rights or remedies either may otherwise have under the Contract Documents or by Laws and Regulations in respect of such Claims.
- B. *Notice:* Written notice stating the general nature of each Claim shall be delivered by the claimant to Engineer and the other party to the Contract promptly (but in no event later than 30 days) after the start of the event giving rise thereto. The responsibility to substantiate a Claim shall rest with the party making the Claim. Notice of the amount or extent of the Claim, with supporting data

shall be delivered to the Engineer and the other party to the Contract within 60 days after the start of such event (unless Engineer allows additional time for claimant to submit additional or more accurate data in support of such Claim). A Claim for an adjustment in Contract Price shall be prepared in accordance with the provisions of Paragraph 12.01.B. A Claim for an adjustment in Contract Times shall be prepared in accordance with the provisions of Paragraph 12.02.B. Each Claim shall be accompanied by claimant's written statement that the adjustment claimed is the entire adjustment to which the claimant believes it is entitled as a result of said event. The opposing party shall submit any response to Engineer and the claimant within 30 days after receipt of the claimant's last submittal (unless Engineer allows additional time).

- C. *Engineer's Action*: Engineer will review each Claim and, within 30 days after receipt of the last submittal of the claimant or the last submittal of the opposing party, if any, take one of the following actions in writing:
 - 1. deny the Claim in whole or in part;
 - 2. approve the Claim; or
 - 3. notify the parties that the Engineer is unable to resolve the Claim if, in the Engineer's sole discretion, it would be inappropriate for the Engineer to do so. For purposes of further resolution of the Claim, such notice shall be deemed a denial.
- D. In the event that Engineer does not take action on a Claim within said 30 days, the Claim shall be deemed denied.
- E. Engineer's written action under Paragraph 10.05.C or denial pursuant to Paragraphs 10.05.C.3 or 10.05.D will be final and binding upon Owner and Contractor, unless Owner or Contractor invoke the dispute resolution procedure set forth in Article 16 within 30 days of such action or denial.
- F. No Claim for an adjustment in Contract Price or Contract Times will be valid if not submitted in accordance with this Paragraph 10.05.

ARTICLE 11 – COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

- 11.01 Cost of the Work
 - A. *Costs Included:* The term Cost of the Work means the sum of all costs, except those excluded in Paragraph 11.01.B, necessarily incurred and paid by Contractor in the proper performance of the Work. When the value of any Work covered by a Change Order or when a Claim for an adjustment in Contract Price is determined on the basis of Cost of the Work, the costs to be reimbursed to Contractor will be only those additional or incremental costs required because of the change in the Work or because of the event giving rise to the Claim. Except as otherwise may be agreed to in writing by Owner, such costs shall be in amounts no higher than those prevailing in the locality of the Project, shall not include any of the costs itemized in Paragraph 11.01.B, and shall include only the following items:

- 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor. Such employees shall include, without limitation, superintendents, foremen, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work shall be apportioned on the basis of their time spent on the Work. Payroll costs shall include, but not be limited to, salaries and wages plus the cost of fringe benefits, which shall include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, bonuses, sick leave, vacation and holiday pay applicable thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, shall be included in the above to the extent authorized by Owner.
- 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts shall accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts shall accrue to Owner. All trade discounts, rebates and refunds and returns from sale of surplus materials and equipment shall accrue to Owner, and Contractor shall make provisions so that they may be obtained.
- 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, who will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee shall be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 11.01.
- 4. Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed for services specifically related to the Work.
- 5. Supplemental costs including the following:
 - a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
 - b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, and hand tools not owned by the workers, which are consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.
 - c. Rentals of all construction equipment and machinery, and the parts thereof whether rented from Contractor or others in accordance with rental agreements approved by Owner with the advice of Engineer, and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs shall be in accordance with the terms of

said rental agreements. The rental of any such equipment, machinery, or parts shall cease when the use thereof is no longer necessary for the Work.

- d. Sales, consumer, use, and other similar taxes related to the Work, and for which Contractor is liable, as imposed by Laws and Regulations.
- e. Deposits lost for causes other than negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
- f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of property insurance established in accordance with Paragraph 5.06.D), provided such losses and damages have resulted from causes other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable. Such losses shall include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses shall be included in the Cost of the Work for the purpose of determining Contractor's fee.
- g. The cost of utilities, fuel, and sanitary facilities at the Site.
- h. Minor expenses such as telegrams, long distance telephone calls, telephone service at the Site, express and courier services, and similar petty cash items in connection with the Work.
- i. The costs of premiums for all bonds and insurance Contractor is required by the Contract Documents to purchase and maintain.
- B. Costs Excluded: The term Cost of the Work shall not include any of the following items:
 - 1. Payroll costs and other compensation of Contractor's officers, executives, principals (of partnerships and sole proprietorships), general managers, safety managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 11.01.A.1 or specifically covered by Paragraph 11.01.A.4, all of which are to be considered administrative costs covered by the Contractor's fee.
 - 2. Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.
 - 3. Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.
 - 4. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not

limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.

- 5. Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraphs 11.01.A.
- C. *Contractor's Fee:* When all the Work is performed on the basis of cost-plus, Contractor's fee shall be determined as set forth in the Agreement. When the value of any Work covered by a Change Order or when a Claim for an adjustment in Contract Price is determined on the basis of Cost of the Work, Contractor's fee shall be determined as set forth in Paragraph 12.01.C.
- D. *Documentation:* Whenever the Cost of the Work for any purpose is to be determined pursuant to Paragraphs 11.01.A and 11.01.B, Contractor will establish and maintain records thereof in accordance with generally accepted accounting practices and submit in a form acceptable to Engineer an itemized cost breakdown together with supporting data.

11.02 Allowances

- A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.
- B. Cash Allowances:
 - 1. Contractor agrees that:
 - a. the cash allowances include the cost to Contractor (less any applicable trade discounts) of materials and equipment required by the allowances to be delivered at the Site, and all applicable taxes; and
 - b. Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment on account of any of the foregoing will be valid.
- C. Contingency Allowance:
 - 1. Contractor agrees that a contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.
- D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor on account of Work covered by allowances, and the Contract Price shall be correspondingly adjusted.

11.03 Unit Price Work

A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to

the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.

- B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Determinations of the actual quantities and classifications of Unit Price Work performed by Contractor will be made by Engineer subject to the provisions of Paragraph 9.07.
- C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.
- D. Owner or Contractor may make a Claim for an adjustment in the Contract Price in accordance with Paragraph 10.05 if:
 - 1. the quantity of any item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement; and
 - 2. there is no corresponding adjustment with respect to any other item of Work; and
 - 3. Contractor believes that Contractor is entitled to an increase in Contract Price as a result of having incurred additional expense or Owner believes that Owner is entitled to a decrease in Contract Price and the parties are unable to agree as to the amount of any such increase or decrease.

ARTICLE 12 – CHANGE OF CONTRACT PRICE; CHANGE OF CONTRACT TIMES

12.01 Change of Contract Price

- A. The Contract Price may only be changed by a Change Order. Any Claim for an adjustment in the Contract Price shall be based on written notice submitted by the party making the Claim to the Engineer and the other party to the Contract in accordance with the provisions of Paragraph 10.05.
- B. The value of any Work covered by a Change Order or of any Claim for an adjustment in the Contract Price will be determined as follows:
 - 1. where the Work involved is covered by unit prices contained in the Contract Documents, by application of such unit prices to the quantities of the items involved (subject to the provisions of Paragraph 11.03); or
 - 2. where the Work involved is not covered by unit prices contained in the Contract Documents, by a mutually agreed lump sum (which may include an allowance for overhead and profit not necessarily in accordance with Paragraph 12.01.C.2); or
 - 3. where the Work involved is not covered by unit prices contained in the Contract Documents and agreement to a lump sum is not reached under Paragraph 12.01.B.2, on the basis of the Cost of the Work (determined as provided in Paragraph 11.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 12.01.C).

- C. Contractor's Fee: The Contractor's fee for overhead and profit shall be determined as follows:
 - 1. a mutually acceptable fixed fee; or
 - 2. if a fixed fee is not agreed upon, then a fee based on the following percentages of the various portions of the Cost of the Work:
 - a. for costs incurred under Paragraphs 11.01.A.1 and 11.01.A.2, the Contractor's fee shall be 15 percent;
 - b. for costs incurred under Paragraph 11.01.A.3, the Contractor's fee shall be five percent;
 - c. where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 12.01.C.2.a and 12.01.C.2.b is that the Subcontractor who actually performs the Work, at whatever tier, will be paid a fee of 15 percent of the costs incurred by such Subcontractor under Paragraphs 11.01.A.1 and 11.01.A.2 and that any higher tier Subcontractor and Contractor will each be paid a fee of five percent of the amount paid to the next lower tier Subcontractor;
 - d. no fee shall be payable on the basis of costs itemized under Paragraphs 11.01.A.4, 11.01.A.5, and 11.01.B;
 - e. the amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in cost will be the amount of the actual net decrease in cost plus a deduction in Contractor's fee by an amount equal to five percent of such net decrease; and
 - f. when both additions and credits are involved in any one change, the adjustment in Contractor's fee shall be computed on the basis of the net change in accordance with Paragraphs 12.01.C.2.a through 12.01.C.2.e, inclusive.

12.02 Change of Contract Times

- A. The Contract Times may only be changed by a Change Order. Any Claim for an adjustment in the Contract Times shall be based on written notice submitted by the party making the Claim to the Engineer and the other party to the Contract in accordance with the provisions of Paragraph 10.05.
- B. Any adjustment of the Contract Times covered by a Change Order or any Claim for an adjustment in the Contract Times will be determined in accordance with the provisions of this Article 12.
- 12.03 Delays
 - A. Where Contractor is prevented from completing any part of the Work within the Contract Times due to delay beyond the control of Contractor, the Contract Times will be extended in an amount equal to the time lost due to such delay if a Claim is made therefor as provided in Paragraph 12.02.A. Delays beyond the control of Contractor shall include, but not be limited to, acts or

neglect by Owner, acts or neglect of utility owners or other contractors performing other work as contemplated by Article 7, fires, floods, epidemics, abnormal weather conditions, or acts of God.

- B. If Owner, Engineer, or other contractors or utility owners performing other work for Owner as contemplated by Article 7, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in the Contract Price or the Contract Times, or both. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times.
- C. If Contractor is delayed in the performance or progress of the Work by fire, flood, epidemic, abnormal weather conditions, acts of God, acts or failures to act of utility owners not under the control of Owner, or other causes not the fault of and beyond control of Owner and Contractor, then Contractor shall be entitled to an equitable adjustment in Contract Times, if such adjustment is essential to Contractor's ability to complete the Work within the Contract Times. Such an adjustment shall be Contractor's sole and exclusive remedy for the delays described in this Paragraph 12.03.C.
- D. Owner, Engineer, and their officers, directors, members, partners, employees, agents, consultants, or subcontractors shall not be liable to Contractor for any claims, costs, losses, or damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) sustained by Contractor on or in connection with any other project or anticipated project.
- E. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delays within the control of Contractor. Delays attributable to and within the control of a Subcontractor or Supplier shall be deemed to be delays within the control of Contractor.

ARTICLE 13 – TESTS AND INSPECTIONS; CORRECTION, REMOVAL OR ACCEPTANCE OF DEFECTIVE WORK

- 13.01 Notice of Defects
 - A. Prompt notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor. Defective Work may be rejected, corrected, or accepted as provided in this Article 13.
- 13.02 Access to Work
 - A. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and governmental agencies with jurisdictional interests will have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply therewith as applicable.
13.03 Tests and Inspections

- A. Contractor shall give Engineer timely notice of readiness of the Work for all required inspections, tests, or approvals and shall cooperate with inspection and testing personnel to facilitate required inspections or tests.
- B. Owner shall employ and pay for the services of an independent testing laboratory to perform all inspections, tests, or approvals required by the Contract Documents except:
 - 1. for inspections, tests, or approvals covered by Paragraphs 13.03.C and 13.03.D below;
 - 2. that costs incurred in connection with tests or inspections conducted pursuant to Paragraph 13.04.B shall be paid as provided in Paragraph 13.04.C; and
 - 3. as otherwise specifically provided in the Contract Documents.
- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.
- D. Contractor shall be responsible for arranging and obtaining and shall pay all costs in connection with any inspections, tests, or approvals required for Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work; or acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work. Such inspections, tests, or approvals shall be performed by organizations acceptable to Owner and Engineer.
- E. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation.
- F. Uncovering Work as provided in Paragraph 13.03.E shall be at Contractor's expense unless Contractor has given Engineer timely notice of Contractor's intention to cover the same and Engineer has not acted with reasonable promptness in response to such notice.

13.04 Uncovering Work

- A. If any Work is covered contrary to the written request of Engineer, it must, if requested by Engineer, be uncovered for Engineer's observation and replaced at Contractor's expense.
- B. If Engineer considers it necessary or advisable that covered Work be observed by Engineer or inspected or tested by others, Contractor, at Engineer's request, shall uncover, expose, or otherwise make available for observation, inspection, or testing as Engineer may require, that portion of the Work in question, furnishing all necessary labor, material, and equipment.

- C. If it is found that the uncovered Work is defective, Contractor shall pay all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or reconstruction (including but not limited to all costs of repair or replacement of work of others); and Owner shall be entitled to an appropriate decrease in the Contract Price. If the parties are unable to agree as to the amount thereof, Owner may make a Claim therefor as provided in Paragraph 10.05.
- D. If the uncovered Work is not found to be defective, Contractor shall be allowed an increase in the Contract Price or an extension of the Contract Times, or both, directly attributable to such uncovering, exposure, observation, inspection, testing, replacement, and reconstruction. If the parties are unable to agree as to the amount or extent thereof, Contractor may make a Claim therefor as provided in Paragraph 10.05.

13.05 Owner May Stop the Work

A. If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed Work will conform to the Contract Documents, Owner may order Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work shall not give rise to any duty on the part of Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or entity, or any surety for, or employee or agent of any of them.

13.06 Correction or Removal of Defective Work

- A. Promptly after receipt of written notice, Contractor shall correct all defective Work, whether or not fabricated, installed, or completed, or, if the Work has been rejected by Engineer, remove it from the Project and replace it with Work that is not defective. Contractor shall pay all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or removal (including but not limited to all costs of repair or replacement of work of others).
- B. When correcting defective Work under the terms of this Paragraph 13.06 or Paragraph 13.07, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.

13.07 Correction Period

A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the terms of any applicable special guarantee required by the Contract Documents) or by any specific provision of the Contract Documents, any Work is found to be defective, or if the repair of any damages to the land or areas made available for Contractor's use by Owner or permitted by Laws and Regulations as contemplated in Paragraph 6.11.A is found to be defective, Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:

- 1. repair such defective land or areas; or
- 2. correct such defective Work; or
- 3. if the defective Work has been rejected by Owner, remove it from the Project and replace it with Work that is not defective, and
- 4. satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others or other land or areas resulting therefrom.
- B. If Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. All claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others) will be paid by Contractor.
- C. In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that item may start to run from an earlier date if so provided in the Specifications.
- D. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this Paragraph 13.07, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.
- E. Contractor's obligations under this Paragraph 13.07 are in addition to any other obligation or warranty. The provisions of this Paragraph 13.07 shall not be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.

13.08 Acceptance of Defective Work

A. If, instead of requiring correction or removal and replacement of defective Work, Owner (and, prior to Engineer's recommendation of final payment, Engineer) prefers to accept it, Owner may do so. Contractor shall pay all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) attributable to Owner's evaluation of and determination to accept such defective Work (such costs to be approved by Engineer as to reasonableness) and for the diminished value of the Work to the extent not otherwise paid by Contractor pursuant to this sentence. If any such acceptance occurs prior to Engineer's recommendation of final payment, a Change Order will be issued incorporating the necessary revisions in the Contract Documents with respect to the Work, and Owner shall be entitled to an appropriate decrease in the Contract Price, reflecting the diminished value of Work so accepted. If the parties are unable to agree as to the amount thereof, Owner may make a Claim therefor as provided in Paragraph 10.05. If the acceptance occurs after such recommendation, an appropriate amount will be paid by Contractor to Owner.

13.09 Owner May Correct Defective Work

- A. If Contractor fails within a reasonable time after written notice from Engineer to correct defective Work, or to remove and replace rejected Work as required by Engineer in accordance with Paragraph 13.06.A, or if Contractor fails to perform the Work in accordance with the Contract Documents, or if Contractor fails to comply with any other provision of the Contract Documents, Owner may, after seven days written notice to Contractor, correct, or remedy any such deficiency.
- B. In exercising the rights and remedies under this Paragraph 13.09, Owner shall proceed expeditiously. In connection with such corrective or remedial action, Owner may exclude Contractor from all or part of the Site, take possession of all or part of the Work and suspend Contractor's services related thereto, take possession of Contractor's tools, appliances, construction equipment and machinery at the Site, and incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere. Contractor shall allow Owner, Owner's representatives, agents and employees, Owner's other contractors, and Engineer and Engineer's consultants access to the Site to enable Owner to exercise the rights and remedies under this Paragraph.
- C. All claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 13.09 will be charged against Contractor, and a Change Order will be issued incorporating the necessary revisions in the Contract Documents with respect to the Work; and Owner shall be entitled to an appropriate decrease in the Contract Price. If the parties are unable to agree as to the amount of the adjustment, Owner may make a Claim therefor as provided in Paragraph 10.05. Such claims, costs, losses and damages will include but not be limited to all costs of repair, or replacement of work of others destroyed or damaged by correction, removal, or replacement of Contractor's defective Work.
- D. Contractor shall not be allowed an extension of the Contract Times because of any delay in the performance of the Work attributable to the exercise by Owner of Owner's rights and remedies under this Paragraph 13.09.

ARTICLE 14 – PAYMENTS TO CONTRACTOR AND COMPLETION

- 14.01 Schedule of Values
 - A. The Schedule of Values established as provided in Paragraph 2.07.A will serve as the basis for progress payments and will be incorporated into a form of Application for Payment acceptable to Engineer. Progress payments on account of Unit Price Work will be based on the number of units completed.
- 14.02 Progress Payments
 - A. Applications for Payments:
 - 1. At least 20 days before the date established in the Agreement for each progress payment (but not more often than once a month), Contractor shall submit to Engineer for review an

Application for Payment filled out and signed by Contractor covering the Work completed as of the date of the Application and accompanied by such supporting documentation as is required by the Contract Documents. If payment is requested on the basis of materials and equipment not incorporated in the Work but delivered and suitably stored at the Site or at another location agreed to in writing, the Application for Payment shall also be accompanied by a bill of sale, invoice, or other documentation warranting that Owner has received the materials and equipment free and clear of all Liens and evidence that the materials and equipment are covered by appropriate property insurance or other arrangements to protect Owner's interest therein, all of which must be satisfactory to Owner.

- 2. Beginning with the second Application for Payment, each Application shall include an affidavit of Contractor stating that all previous progress payments received on account of the Work have been applied on account to discharge Contractor's legitimate obligations associated with prior Applications for Payment.
- 3. The amount of retainage with respect to progress payments will be as stipulated in the Agreement.
- B. *Review of Applications:*
 - 1. Engineer will, within 10 days after receipt of each Application for Payment, either indicate in writing a recommendation of payment and present the Application to Owner or return the Application to Contractor indicating in writing Engineer's reasons for refusing to recommend payment. In the latter case, Contractor may make the necessary corrections and resubmit the Application.
 - 2. Engineer's recommendation of any payment requested in an Application for Payment will constitute a representation by Engineer to Owner, based on Engineer's observations of the executed Work as an experienced and qualified design professional, and on Engineer's review of the Application for Payment and the accompanying data and schedules, that to the best of Engineer's knowledge, information and belief:
 - a. the Work has progressed to the point indicated;
 - b. the quality of the Work is generally in accordance with the Contract Documents (subject to an evaluation of the Work as a functioning whole prior to or upon Substantial Completion, the results of any subsequent tests called for in the Contract Documents, a final determination of quantities and classifications for Unit Price Work under Paragraph 9.07, and any other qualifications stated in the recommendation); and
 - c. the conditions precedent to Contractor's being entitled to such payment appear to have been fulfilled in so far as it is Engineer's responsibility to observe the Work.
 - 3. By recommending any such payment Engineer will not thereby be deemed to have represented that:
 - a. inspections made to check the quality or the quantity of the Work as it has been performed have been exhaustive, extended to every aspect of the Work in progress, or

involved detailed inspections of the Work beyond the responsibilities specifically assigned to Engineer in the Contract Documents; or

- b. there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold payment to Contractor.
- 4. Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment, will impose responsibility on Engineer:
 - a. to supervise, direct, or control the Work, or
 - b. for the means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or
 - c. for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work, or
 - d. to make any examination to ascertain how or for what purposes Contractor has used the moneys paid on account of the Contract Price, or
 - e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens.
- 5. Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated in Paragraph 14.02.B.2. Engineer may also refuse to recommend any such payment or, because of subsequently discovered evidence or the results of subsequent inspections or tests, revise or revoke any such payment recommendation previously made, to such extent as may be necessary in Engineer's opinion to protect Owner from loss because:
 - a. the Work is defective, or completed Work has been damaged, requiring correction or replacement;
 - b. the Contract Price has been reduced by Change Orders;
 - c. Owner has been required to correct defective Work or complete Work in accordance with Paragraph 13.09; or
 - d. Engineer has actual knowledge of the occurrence of any of the events enumerated in Paragraph 15.02.A.
- C. Payment Becomes Due:
 - 1. Ten days after presentation of the Application for Payment to Owner with Engineer's recommendation, the amount recommended will (subject to the provisions of Paragraph 14.02.D) become due, and when due will be paid by Owner to Contractor.

D. Reduction in Payment:

- 1. Owner may refuse to make payment of the full amount recommended by Engineer because:
 - a. claims have been made against Owner on account of Contractor's performance or furnishing of the Work;
 - b. Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and discharge of such Liens;
 - c. there are other items entitling Owner to a set-off against the amount recommended; or
 - d. Owner has actual knowledge of the occurrence of any of the events enumerated in Paragraphs 14.02.B.5.a through 14.02.B.5.c or Paragraph 15.02.A.
- 2. If Owner refuses to make payment of the full amount recommended by Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and promptly pay Contractor any amount remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, when Contractor remedies the reasons for such action.
- 3. Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld shall be treated as an amount due as determined by Paragraph 14.02.C.1 and subject to interest as provided in the Agreement.
- 14.03 Contractor's Warranty of Title
 - A. Contractor warrants and guarantees that title to all Work, materials, and equipment covered by any Application for Payment, whether incorporated in the Project or not, will pass to Owner no later than the time of payment free and clear of all Liens.
- 14.04 Substantial Completion
 - A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete (except for items specifically listed by Contractor as incomplete) and request that Engineer issue a certificate of Substantial Completion.
 - B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.
 - C. If Engineer considers the Work substantially complete, Engineer will deliver to Owner a tentative certificate of Substantial Completion which shall fix the date of Substantial Completion. There shall be attached to the certificate a tentative list of items to be completed or corrected before

final payment. Owner shall have seven days after receipt of the tentative certificate during which to make written objection to Engineer as to any provisions of the certificate or attached list. If, after considering such objections, Engineer concludes that the Work is not substantially complete, Engineer will, within 14 days after submission of the tentative certificate to Owner, notify Contractor in writing, stating the reasons therefor. If, after consideration of Owner's objections, Engineer considers the Work substantially complete, Engineer will, within said 14 days, execute and deliver to Owner and Contractor a definitive certificate of Substantial Completion (with a revised tentative list of items to be completed or corrected) reflecting such changes from the tentative certificate as Engineer believes justified after consideration of any objections from Owner.

- D. At the time of delivery of the tentative certificate of Substantial Completion, Engineer will deliver to Owner and Contractor a written recommendation as to division of responsibilities pending final payment between Owner and Contractor with respect to security, operation, safety, and protection of the Work, maintenance, heat, utilities, insurance, and warranties and guarantees. Unless Owner and Contractor agree otherwise in writing and so inform Engineer in writing prior to Engineer's issuing the definitive certificate of Substantial Completion, Engineer's aforesaid recommendation will be binding on Owner and Contractor until final payment.
- E. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the tentative list.
- 14.05 Partial Utilization
 - A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:
 - 1. Owner at any time may request Contractor in writing to permit Owner to use or occupy any such part of the Work which Owner believes to be ready for its intended use and substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 14.04.A through D for that part of the Work.
 - 2. Contractor at any time may notify Owner and Engineer in writing that Contractor considers any such part of the Work ready for its intended use and substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
 - 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of Paragraph 14.04 will apply with respect to certification of Substantial Completion of that part of the Work and the division of responsibility in respect thereof and access thereto.

- 4. No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 5.10 regarding property insurance.
- 14.06 Final Inspection
 - A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.
- 14.07 Final Payment
 - A. Application for Payment:
 - 1. After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, marked-up record documents (as provided in Paragraph 6.12), and other documents, Contractor may make application for final payment following the procedure for progress payments.
 - 2. The final Application for Payment shall be accompanied (except as previously delivered) by:
 - a. all documentation called for in the Contract Documents, including but not limited to the evidence of insurance required by Paragraph 5.04.B.6;
 - b. consent of the surety, if any, to final payment;
 - c. a list of all Claims against Owner that Contractor believes are unsettled; and
 - d. complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of or Liens filed in connection with the Work.
 - 3. In lieu of the releases or waivers of Liens specified in Paragraph 14.07.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (i) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (ii) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien.
 - B. Engineer's Review of Application and Acceptance:
 - 1. If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's review of the final Application for Payment and accompanying

documentation as required by the Contract Documents, Engineer is satisfied that the Work has been completed and Contractor's other obligations under the Contract Documents have been fulfilled, Engineer will, within ten days after receipt of the final Application for Payment, indicate in writing Engineer's recommendation of payment and present the Application for Payment to Owner for payment. At the same time Engineer will also give written notice to Owner and Contractor that the Work is acceptable subject to the provisions of Paragraph 14.09. Otherwise, Engineer will return the Application for Payment to Contractor, indicating in writing the reasons for refusing to recommend final payment, in which case Contractor shall make the necessary corrections and resubmit the Application for Payment.

C. Payment Becomes Due:

1. Thirty days after the presentation to Owner of the Application for Payment and accompanying documentation, the amount recommended by Engineer, less any sum Owner is entitled to set off against Engineer's recommendation, including but not limited to liquidated damages, will become due and will be paid by Owner to Contractor.

14.08 Final Completion Delayed

A. If, through no fault of Contractor, final completion of the Work is significantly delayed, and if Engineer so confirms, Owner shall, upon receipt of Contractor's final Application for Payment (for Work fully completed and accepted) and recommendation of Engineer, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance to be held by Owner for Work not fully completed or corrected is less than the retainage stipulated in the Agreement, and if bonds have been furnished as required in Paragraph 5.01, the written consent of the surety to the payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by Contractor to Engineer with the Application for such payment. Such payment shall be made under the terms and conditions governing final payment, except that it shall not constitute a waiver of Claims.

14.09 Waiver of Claims

- A. The making and acceptance of final payment will constitute:
 - 1. a waiver of all Claims by Owner against Contractor, except Claims arising from unsettled Liens, from defective Work appearing after final inspection pursuant to Paragraph 14.06, from failure to comply with the Contract Documents or the terms of any special guarantees specified therein, or from Contractor's continuing obligations under the Contract Documents; and
 - 2. a waiver of all Claims by Contractor against Owner other than those previously made in accordance with the requirements herein and expressly acknowledged by Owner in writing as still unsettled.

ARTICLE 15 – SUSPENSION OF WORK AND TERMINATION

15.01 Owner May Suspend Work

A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by notice in writing to Contractor and Engineer which will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be granted an adjustment in the Contract Price or an extension of the Contract Times, or both, directly attributable to any such suspension if Contractor makes a Claim therefor as provided in Paragraph 10.05.

15.02 Owner May Terminate for Cause

- A. The occurrence of any one or more of the following events will justify termination for cause:
 - 1. Contractor's persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment or failure to adhere to the Progress Schedule established under Paragraph 2.07 as adjusted from time to time pursuant to Paragraph 6.04);
 - 2. Contractor's disregard of Laws or Regulations of any public body having jurisdiction;
 - 3. Contractor's repeated disregard of the authority of Engineer; or
 - 4. Contractor's violation in any substantial way of any provisions of the Contract Documents.
- B. If one or more of the events identified in Paragraph 15.02.A occur, Owner may, after giving Contractor (and surety) seven days written notice of its intent to terminate the services of Contractor:
 - 1. exclude Contractor from the Site, and take possession of the Work and of all Contractor's tools, appliances, construction equipment, and machinery at the Site, and use the same to the full extent they could be used by Contractor (without liability to Contractor for trespass or conversion);
 - 2. incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere; and
 - 3. complete the Work as Owner may deem expedient.
- C. If Owner proceeds as provided in Paragraph 15.02.B, Contractor shall not be entitled to receive any further payment until the Work is completed. If the unpaid balance of the Contract Price exceeds all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) sustained by Owner arising out of or relating to completing the Work, such excess will be paid to Contractor. If such claims, costs, losses, and damages exceed such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when

so approved by Engineer, incorporated in a Change Order. When exercising any rights or remedies under this Paragraph, Owner shall not be required to obtain the lowest price for the Work performed.

- D. Notwithstanding Paragraphs 15.02.B and 15.02.C, Contractor's services will not be terminated if Contractor begins within seven days of receipt of notice of intent to terminate to correct its failure to perform and proceeds diligently to cure such failure within no more than 30 days of receipt of said notice.
- E. Where Contractor's services have been so terminated by Owner, the termination will not affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue. Any retention or payment of moneys due Contractor by Owner will not release Contractor from liability.
- F. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 5.01.A, the termination procedures of that bond shall supersede the provisions of Paragraphs 15.02.B and 15.02.C.
- 15.03 Owner May Terminate For Convenience
 - A. Upon seven days written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such case, Contractor shall be paid for (without duplication of any items):
 - 1. completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination, including fair and reasonable sums for overhead and profit on such Work;
 - 2. expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses;
 - 3. all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) incurred in settlement of terminated contracts with Subcontractors, Suppliers, and others; and
 - 4. reasonable expenses directly attributable to termination.
 - B. Contractor shall not be paid on account of loss of anticipated profits or revenue or other economic loss arising out of or resulting from such termination.

15.04 Contractor May Stop Work or Terminate

A. If, through no act or fault of Contractor, (i) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority, or (ii) Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (iii) Owner fails for 30 days

to pay Contractor any sum finally determined to be due, then Contractor may, upon seven days written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate the Contract and recover from Owner payment on the same terms as provided in Paragraph 15.03.

B. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 30 days to pay Contractor any sum finally determined to be due, Contractor may, seven days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The provisions of this Paragraph 15.04 are not intended to preclude Contractor from making a Claim under Paragraph 10.05 for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this Paragraph.

ARTICLE 16 – DISPUTE RESOLUTION

16.01 Methods and Procedures

- A. Either Owner or Contractor may request mediation of any Claim submitted to Engineer for a decision under Paragraph 10.05 before such decision becomes final and binding. The mediation will be governed by the Construction Industry Mediation Rules of the American Arbitration Association in effect as of the Effective Date of the Agreement. The request for mediation shall be submitted in writing to the American Arbitration Association and the other party to the Contract. Timely submission of the request shall stay the effect of Paragraph 10.05.E.
- B. Owner and Contractor shall participate in the mediation process in good faith. The process shall be concluded within 60 days of filing of the request. The date of termination of the mediation shall be determined by application of the mediation rules referenced above.
- C. If the Claim is not resolved by mediation, Engineer's action under Paragraph 10.05.C or a denial pursuant to Paragraphs 10.05.C.3 or 10.05.D shall become final and binding 30 days after termination of the mediation unless, within that time period, Owner or Contractor:
 - 1. elects in writing to invoke any dispute resolution process provided for in the Supplementary Conditions; or
 - 2. agrees with the other party to submit the Claim to another dispute resolution process; or
 - 3. gives written notice to the other party of the intent to submit the Claim to a court of competent jurisdiction.

ARTICLE 17 – MISCELLANEOUS

- 17.01 Giving Notice
 - A. Whenever any provision of the Contract Documents requires the giving of written notice, it will be deemed to have been validly given if:

- 1. delivered in person to the individual or to a member of the firm or to an officer of the corporation for whom it is intended; or
- 2. delivered at or sent by registered or certified mail, postage prepaid, to the last business address known to the giver of the notice.

17.02 Computation of Times

A. When any period of time is referred to in the Contract Documents by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.

17.03 Cumulative Remedies

A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract Documents. The provisions of this Paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.

17.04 Survival of Obligations

A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract Documents, as well as all continuing obligations indicated in the Contract Documents, will survive final payment, completion, and acceptance of the Work or termination or completion of the Contract or termination of the services of Contractor.

17.05 Controlling Law

A. This Contract is to be governed by the law of the state in which the Project is located.

17.06 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

PROJECT MANUAL

ARPA Renovations Girls Inc.

130 Lincoln Street Meriden, CT 06451



Prepared By:

Architect

Christopher Williams Architects, LLC 85 Willow Street Building 54 New Haven, CT 06511

MEP Engineer

Integrated Engineering Services LLC 33 North Plains Industrial Road Wallingford, CT 06492

HAZMAT Consultant

Enviromed 470 Murdock Avenue Meriden, CT 06450

ISSUED FOR BIDDING

March 12, 2024

TABLE OF CONTENTS

TITLE PA	AGE	
PROJEC	T MANUAL TABLE OF CONTENTS2	
LIST OF	DRAWINGS2	
DIVISION	N 01 – GENERAL REQUIREMENTS	
01 1000	SUMMARY5	
01 2200	UNIT PRICES	
01 3000	ADMINISTRATIVE REQUIRMENTS	
01 3233	PHOTOGRAPHIC DOCUMENTATION	
01 3300	SUBMITTAL PROCEDURES	
	01 3300.1 – SUBMITTAL COVER PAGE FORM1	
01 4000	QUALITY REQUIREMENTS	
01 5000	TEMPORARY FACILITIES AND CONTROLS	
01 5640	TREE PROTECTION AND PRESERVATION14	
01 6000	PRODUCT REQUIREMENTS4	
01 7000	EXECUTION REQUIREMENTS	
01 7329	CUTTING AND PATCHING	
01 7700	CLOSEOUT PROCEDURES	
DIVISION	N 02 – EXISTING CONDITIONS	
02 4119	SELECTIVE DEMOLITION	
02 8216	ASBESTOS ABATEMENT22	
02 8233	LEAD ABATEMENT4	
DIVISION	N 03 – CONCRETE	
03 3000	CAST-IN-PLACE CONCRETE	
DIVISION	N 05 – METALS	
05 5213	PIPE AND TUBE RAILINGS4	
DIVISION	1 06 – WOOD	
06 1053	MISCELLANEOUS ROUGH CARPENTRY	
06 2013	EXTERIOR FINISH CARPENTRY	
06 7300	COMPOSITE DECKING	

DIVISION 07 – THERMAL AND MOISTURE PROTECTION

07 1616	CRYSTALLINE WATERPROOFING	4
07 2500	WEATHER BARRIERS	7
07 4640	VINYL SIDING AND SOFFITS	5
07 9200	JOINT SEALANTS	7

DIVISION 08 - DOORS AND WINDOWS

08 1113	HOLLOW METAL DOOR AND FRAMES	2
08 3113	ACCESS DOORS AND FRAMES	1
08 7100	DOOR HARDWARE	4

DIVISION 09 - FINISHES

PLASTER REPAIR	12
GYPSUM BOARD	8
RESILIENT BASE AND ACCESSORIES	5
VINYL COMPOSITION FLOOR TILING	9
EXTERIOR PAINTING	9
INTERIOR PAINTING	10
	PLASTER REPAIR

DIVISION 23 – HVAC

SEE DRAWINGS

DIVISION 26 – ELECTRICAL

SEE DRAWINGS

DIVISION 31 – EARTHWORK

EROSION AND SEDIMENT CONTROL	3
EARTHWORK1	2
EXCAVATION SUPPORT AND PROTECTION	3
32 – EXTERIOR IMPROVEMENTS	
ASPHALT CONCRETE PAVING	5
CEMENT CONCRETE PAVING	4
PAVEMENT JOINT SEALANTS	4
PLANTING SOILS	8
TURF AND GRASSES	8
 	EROSION AND SEDIMENT CONTROL

END TABLE OF CONTENTS

LIST OF DRAWINGS

GENERAL

COVE	RSHEET
G001	CODE INFORMATION, GENERAL NOTES, SYMBOLS & ABBREVIATIONS
<u>CIVIL</u>	
CD101	SITE DEMOLITION PLAN
C101	SITE PLAN
C501	
C502	ENLARGED EXTERIOR RAMP PLAN & DETAILS
<u>ARCH</u>	ITECTURAL
AD100	BASEMENT DEMOLITION PLAN
AD101	FIRST FLOOR DEMOLITION PLAN
AD102	SECOND FLOOR DEMOLITION PLAN
AD103	THIRD FLOOR DEMOLITION PLAN
AD201	DEMOLITION BUILDING ELEVATIONS
A100	BASEMENT FLOOR PLAN
A101	
A102	
A103	
A120	ROOF PLAN
A201	NORTH & EAST BUILDING ELEVATIONS
A202	SOUTH & WEST BUILDING ELEVATIONS
A203	ENLARGED REAR PORCH ELEVATIONS – DEMOLITION & NEW
A204	ENLARGED PARTIAL WEST BUILDING ELEVATIONS & DETAILS
A501	DETAILS
<u>HAZAI</u>	RDOUS MATERIALS ABATEMENT
H100	BASEMENT ABATEMENT PLAN
H101	FIRST FLOOR ABATEMENT PLAN
H102	SECOND FLOOR ABATEMENT PLAN
H103	
H120	ROOF ABATEMENT PLAN
H201	BUILDING ELEVATIONS - ABATEMENT
H203	ENLARGED REAR PORCH ELEVATIONS ABATEMENT
<u>SITE L</u>	<u>ITILIITES</u>

SU.100SITE UTILITY PLAN

PLUMBING

DP.100	BASEMENT PLUMBING DEMOLITION PLAN
P.100	BASEMENT PLUMBING PLAN
P.101	FIRST FLOOR PLUMBING PLAN
P.200	
P.201	

MECHANICAL

DM.100	BASEMENT MECHANICAL DEMOLITION MECHANICAL
DM.101	FIRST FLOOR MECHANICAL DEMOLITION PLAN
DM.102	SECOND FLOOR MECHANICAL DEMOLITION PLAN
M.100	BASEMENT MECHANICAL PLAN
M.101	FIRST FLOOR MECHANICAL PLAN
M.102	SECOND FLOOR MECHANICAL PLAN
M.103	THIRD FLOOR MECHANICAL PLAN
M.200	MECHANICAL NOTES, LEGEND & DETAILS
M.201	
M.202	MECHANICAL SCHEDULES
M.203	MECHANICAL VRF SCHEDULES & DETAILS
M.204	MECHANICAL SPECIFICATIONS
ELECTRICAL	

DE.100	BASEMENT ELECTRICAL DEMOLITION PLAN
DE.101	FIRST FLOOR ELECTRICAL DEMOLITION PLAN
DE.102	2 SECOND FLOOR ELECTRICAL DEMOLITION PLAN
DE.103	THIRD FLOOR ELECTRICAL DEMOLITION PLAN
E.100	BASEMENT ELECTRICAL PLAN
E.101	FIRST FLOOR ELECTRICAL PLAN
E.102	SECOND FLOOR ELECTRICAL PLAN
E.103	
E.201	ELECTRICAL POWER RISER
E.202	ELECTRICAL SCHEDULES
E.203	ELECTRICAL NOTES, LEGEND & SCHEDULES
E.204	ELECTRICAL SPECIFICATIONS
E.205	ELECTRICAL SPECIFICATIONS CONTINUED

END OF LIST OF DRAWINGS

SECTION 01 1000 - SUMMARY

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Project information.
 - 2. Work covered by Contract Documents.
 - 1. Designated Subcontractors.
 - 2. Access to site.
 - 3. Coordination with occupants.
 - 4. Work restrictions.
 - 5. Specification and drawing conventions.
 - 6. Miscellaneous provisions.
 - B. Related Requirements:
 - 1. Section 01 5000 "Temporary Facilities and Controls" for limitations and procedures governing temporary use of Owner's facilities.
- 1.3 PROJECT INFORMATION
 - A. Project Identification: ARPA Renovations Girls Inc of Meriden
 - B. Project Location: 130 Lincoln Street Meriden, CT 06451
 - C. Owner: Girls Inc of Meriden
 - D. City of Meriden Project Manager:

Matt Peacock KBE Building Corporation 76 Batterson Park Rd, Farmington, CT 06032 (860) 284 7622 <u>mpeacock@kbebuilding.com</u>

E. Architect: Christopher Williams Architects LLC Contact: Christopher Williams 85 Willow Street New Haven, Connecticut 06511

(203) 776 0184 cwilliams@cwarchitectsllc.com

F. Architect's Consultants: The Architect has retained the following design professionals who have prepared designated portions of the Contract Documents:

MEP Integrated Engineering Services, LLC 33 North Plains Industrial Road Wallingford, CT 06492

HAZMAT EnviroMed Services INC 470 Murdock Avenue Meriden, CT 06450

G. Authorities Having Jurisdiction

City of Meriden Fire Marshal 142 East Main Street Meriden, CT 06450

City of Meriden Building Department 142 East Main Street Meriden, CT 06450

1.4 WORK COVERED BY CONTRACT DOCUMENTS

- A. Work of the Project generally consists of, but is not limited to, the following and is more specifically defined by the Contract Documents:
 - 1. Exterior concrete ramp for access to First Floor.
 - 2. Replacement of existing brick floor @ Basement with a concrete slab on grade.
 - 3. Waterproofing of Basement masonry walls
 - 4. Replacement of Third Floor Ceiling.
 - 5. Mechanical Work, including:
 - a. Installation of new underground gas service from Lincoln Street (by the utility company).
 - b. Basement boiler and water heater replacement.
 - c. Replacement of existing fan coil units at First and Second Floors.
 - d. Addition of fan coil units at select locations.
 - e. Replacement of air handling units at Gymnasium and Basement.
 - f. Installation of exhaust fans at First and Second Floor toilet rooms.
 - g. Installation of exhaust fan and ductwork at Third Floor
 - 6. Electrical Work including:
 - a. Upgrade of building electrical service
 - b. Work related to installation of mechanical equipment.
 - c. Removal and reinstallation of Third Floor heat detectors and lighting related to ceiling replacement.

- 7. Site work related to installation of exterior ramp, electrical and gas services
- 8. Hazardous Materials abatement related to architectural & MEP work
- B. Type of Contract: Project will be constructed under a single prime contract.
- C. Duration of Construction period: 180 days from issuance of building permit to substantial completion.
- 1.5 ACCESS TO SITE
 - A. General: Contractor shall have limited use of Project site for construction operations as indicated on Drawings by the Contract limits and as indicated by requirements of this Section.
 - B. Use of Site: Limit use of Project site to work in areas as the work progresses. Do not disturb portions of Project site beyond areas in which to be worked on in coordination with the Owner.
 - 1. Driveways, Walkways and Entrances: Keep driveway, parking areas, paved areas and entrances serving premises clear and available to Owner, Owner's employees, and emergency vehicles at all times. Do not use these areas for parking or storage of materials.
 - a. Schedule deliveries to minimize use of driveways and entrances by construction operations.
 - b. Schedule deliveries to minimize space and time requirements for storage of materials and equipment on-site.
 - C. Fire Alarm System: This project involves removing and reinstalling the Third Floor Heat detectors which are tied into the fire alarm system in a way that maintains fire alarm coverage at all times during occupancy. The contractor shall be responsible for maintaining a fully functional operating system, including any temporary devices and/ or wiring until the new system is inspected, tested and accepted by the Authorities with Jurisdiction.
 - D. Condition of Existing Building: Maintain portions of existing building affected by construction operations in a weathertight condition throughout construction period. Repair damage caused by construction operations.
 - a. The building must always be secure, during working and non-working hours. Maintain all affected components of the security, access control and intercom system. Provide security personnel during any shutdown times.

1.6 COORDINATION WITH OCCUPANTS

A. Full Owner Occupancy: Girls Inc will be fully occupied during entire construction period. Cooperate with Owner during construction operations to minimize conflicts and facilitate Owner usage. Perform the Work so as not to interfere with Owner's day-today operations. Maintain existing exits unless otherwise indicated.

- 1. Maintain access to existing walkways, corridors, and other adjacent occupied or used facilities. Do not close or obstruct walkways, corridors, or other occupied or used facilities without written permission from Owner and approval of authorities having jurisdiction.
- 2. Notify Owner not less than ninety-six (96) hours in advance of activities that will affect Owner's operations.

1.7 WORK RESTRICTIONS

- A. Work Restrictions, General: Comply with restrictions on construction operations.
 - 1. Comply with limitations on use of public streets and with other requirements of authorities having jurisdiction.
- B. On-Site Work Hours: Limit work in the existing building to the following, unless allowed otherwise by Owner:
 - 1. September Mid-June: Monday Friday, 9:30 am 5:00 pm
 - 2. Mid_June August: Monday Friday, 7:30 am 5:00 pm
- C. Fire Watch: At all times when the fire alarm system is disabled, or shut down for modifications, provide a fire watch in accordance with the requirements of the City of Meriden.
- D. Retain "Existing Utility Interruptions" Paragraph below for existing utilities. Coordinate with requirements for temporary utilities specified in Section 01 5000 "Temporary Facilities and Controls."
- E. Existing Utility Interruptions: Do not interrupt utilities serving facilities occupied by Owner or others unless permitted under the following conditions and then only after providing temporary utility services according to requirements indicated:
 - 1. Notify Owner not less than SEVEN (7) calendar days in advance of proposed utility interruptions.
 - 2. Obtain Owner's written permission before proceeding with utility interruptions.
- F. Noise, Vibration, and Odors: Coordinate operations that may result in high levels of noise and vibration, odors, or other disruption to Owner occupancy with Owner.
- G. Controlled Substances: Use of tobacco products and other controlled substances on Project site is not permitted.
- 1.8 Employee Identification: Will be issued by Owner. Always require personnel to use identification tags.
- 1.9 SECURITY
 - A. Any employee of a contractor that is assigned this project, shall, 1 month prior to initiating work at the work site, submit all employees and subcontractor names to the City of

Meriden, to include a copy of a driver's license or other government issued identification card, for the purpose of a background check.

- B. Following the initial submission of names and identification, the contractor shall make available all employees approved for the work site to submit their fingerprints via a digital submission for a national database search of fingerprints for any disqualifying events.
- C. Contractor agrees to have a National Sex Offender Public Registry check completed on each of its agents, employees, volunteers, and subcontractors before permitting any such individual(s) to perform any work or enter upon the premises where any work is being performed under this agreement. If any such individual(s) have committed a sex offense, then Contractor agrees to prevent and prohibit them from performing any work or entering upon the premises where any work is being performed under this agreement.
- D. Any employee or subcontractor found to have a disqualifying event on their history will not be allowed to enter or complete any work within the project.

1.10 SPECIFICATION AND DRAWING CONVENTIONS

- A. Specification Content: The Specifications use certain conventions for the style of language and the intended meaning of certain terms, words, and phrases when used in particular situations. These conventions are as follows:
 - 1. Imperative mood and streamlined language are generally used in the Specifications. The words "shall," "shall be," or "shall comply with," depending on the context, are implied where a colon (:) is used within a sentence or phrase.
 - 2. Specification requirements are to be performed by Contractor unless specifically stated otherwise.
- B. Division 01 General Requirements: Requirements of Sections in Division 01 apply to the Work of all Sections in the Specifications.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 1000

SECTION 01 2200 - UNIT PRICES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes administrative and procedural requirements for unit prices.
 - 1. A unit price is an amount stated on the Proposal Form or in the Specifications as a price per unit of measurement for materials or services that will be added to or deducted from the Contract Sum by Change Order in the event the quantities of Work required by the Contract Documents are increased or decreased. Should the number of units anticipated substantially increase by more than ten (10) percent, such unit price shall be renegotiated.
 - 2. Upon request, provide a complete breakdown of how the unit price was calculated. Unit Prices include all necessary material, delivery, equipment and manpower, overhead and profit, and applicable taxes.
 - 3. Refer to individual Specification Sections for construction activities requiring the establishment of unit prices. Methods of measurement and payment for unit prices are specified in those Sections.
- B. Related Requirements:
 - 1. Division 01 Section "Contract Modification Procedures" for procedures for submitting and handling Change Orders.
 - 2. Division 01 Section "Quality Requirements" for general testing and inspecting requirements.

1.3 DEFINITIONS

A. Unit price is an amount incorporated in the Agreement, applicable during the duration of the Work as a price per unit of measurement for materials, equipment, or services, or a portion of the Work, added to or deducted from the Contract Sum by appropriate modification, if the scope of Work or estimated quantities of Work required by the Contract Documents are increased or decreased.

1.4 PROCEDURES

A. Unit prices include all necessary material, plus cost for delivery, installation, insurance, applicable taxes, overhead, and profit.

- Β. Measurement and Payment: See individual Specification Sections for work that requires establishment of unit prices. For adjustment of the Contract Sum, track and provide supporting documentation (delivery or removal/disposal tickets) as deemed necessary to confirm Unit quantities provided or removed. Such quantities shall be tracked against any specific "Allowance" line item which may be included within the base contract value. Contractor must report status of Unit Price Allowances regularly and not exceed such allowance value without prior authorization from the Owner. Unit Price that does not have an allowance applied to it shall be treated with a CCD until all guantities have been determined. Owner reserves the right to reject Contractor's measurement of work-in-place that involves use of established unit prices and to have this work measured, at Owner's expense, by an independent surveyor acceptable to Contractor. Should it be determined that the work measured by the independent surveyor is five (5) or more percent less than the Contractor's reported measurement, Contractor shall be responsible for the cost associated with the independent survey and adjustment to the measurement.
- C. List of Unit Prices: A schedule of unit prices is included in Part 3. Specification Sections referenced in the schedule contain requirements for materials described under each unit price.

PART 2 - PRODUCTS (Not Used)

- PART 3 EXECUTION
- 3.1 SCHEDULE OF UNIT PRICES
 - A. Unit Price No. 1: Interior Painting (gypsum board and plaster partitions).
 - 1. Description: Preparation and painting of interior gypsum board partitions per Section 09 9123.
 - 2. Unit of Measurement: Square foot.

END OF SECTION 01 2200

SECTION 01 3000 - ADMINISTRATIVE REQUIREMENTS

PART 1 - GENERAL

1.1 PRECONSTRUCTION CONFERENCE

- A. The Contractor shall prepare the agenda, which shall include:
 - 1. Submittals: Those received prior to the conference and those due at the conference. Any necessary discussion covering future Submittals will be covered at this time, including the requirement for maintaining and submitting Record Drawings. At a minimum, the following Submittals will be discussed:
 - a. Certificate of Insurance.
 - b. Labor and Material Payment Bond.
 - c. Construction Schedule.
 - d. Submittal Schedule.
 - e. Schedule of Values.
 - f. Approved Applicator Certificates.
 - g. Safety Submittal.
 - h. Record Drawings.
 - i. Review of the Contractor's proposed Schedule and proposed workforce size.
 - 2. Review of final list of Subcontractors and material suppliers.
 - 3. Review of the Contract Documents to resolve errors, omissions or inconsistencies discovered.
 - 4. Review of job conditions, including:
 - a. Work by other contractors.
 - b. Work by other contractors.
 - c. Owner occupancy during construction.
 - d. Protection of existing surfaces and finishes.
 - e. Maintenance of exits.
 - f. Weather protection procedures.
 - g. Landscape protection.
 - 5. Review planned use of Owner's facilities, such as stairwells, loading docks, elevators, and driveways. Confirm Owner's requirements for scheduling such use and protecting the facilities from damage.
 - 6. Review Owner's Safety Advisory and regulatory requirements applicable to staging, work platforms, catch platforms, sidewalk bridges, safety netting, street closings, hoisting equipment, protective devices and hazardous materials.
 - 7. Review Owner's Safety Advisory and general project requirements.
 - 8. Security procedures:

- a. Confirm communications network.
- b. Personnel identification.
- c. Parking assignments.
- d. Workforce comportment.
- 9. Review temporary utility requirements:
 - a. Power.
 - b. Lighting.
 - c. Telephone.
 - d. Water.
 - e. Heat and Ventilation.
 - f. Sanitary facilities.
- 10. Review field office requirements.
- 11. Review material handling procedures such as:
 - a. Methods of transport and disposal of demolished materials.
 - b. Locations of chutes and refuse receptacles.
 - c. Delivery dates of materials.
 - d. Storage and protection of materials delivered to the site.
 - e. Storage and protection of materials delivered off site, including access for cost certification inspections.
- 12. Discuss project administration procedures.
- 13. Provisions for Owner and Design Professional's access to the work.
- 14. Scheduling of construction or operations to be done by owner with his own forces or under separate contract.
- 15. Frequency, time, and location of project meetings.

1.2 PROGRESS MEETINGS

- A. Meetings will be held as agreed to review the Work in progress. Based on the stage of the Work and matters needing resolution, the Contractor shall invite Subcontractors and Consultants of all tiers as required by the Contractor, the Design Professional and the Owner.
 - 1. Attendance shall be mandatory.
- B. The Contractor will prepare the agenda for the meeting, preside over it, record, reproduce and distribute the minutes.
- C. The Agenda shall include:
 - 1. Problems that may impede progress and procedures to maintain schedule.
 - 2. Discussion of Work planned to be done before next meeting.
 - 3. Schedule pre-installation conferences.

- 4. Discuss findings or action items identified in previous pre-installation conferences.
- 5. Review delivery schedules.
- 6. Review proposed changes.
- 7. Review Submittal status.
- 8. Review applications and payments.
- 9. Other business as required by the progress of the Project.

1.3 JOB SITE ADMINISTRATION

- A. The Design Professional may have a Designated Representative present at the site while the Work is in progress.
 - 1. The Contractor shall provide ready, easy and safe access for the Design Professional and the Owner to all parts of the Work, whenever and wherever it is in progress.
 - 2. Failure to provide access may result in rejection of that portion of the Work to which access has been denied.
 - 3. If ladders, ramps, staging, lifts, and scaffolds that are erected, maintained, or operated for the Contractor's use in the performance of the Work, or safety equipment provide by the Contractor to the Design Professional are not acceptable to the Design Professional, this will be considered as denial of access to the Work.
 - 4. The Contractor shall coordinate Work of the various sections of Specifications to assure efficient and orderly sequence of installation of construction elements, with provisions for accommodating items installed later.

PART 2 - PRODUCTS (Not used)

PART 3 - EXECUTION (Not used

END OF SECTION 01 3 000

SECTION 01 3233 - PHOTOGRAPHIC DOCUMENTATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes administrative and procedural requirements for the following:
 - 1. Preconstruction photographs.
 - 2. Periodic construction photographs.
 - 3. Final completion construction photographs.
- B. Related Requirements:
 - 1. Division 01 Section "Submittal Procedures" for submitting photographic documentation.
 - 2. Division 01 Section "Closeout Procedures" for submitting photographic documentation as project record documents at Project closeout.
 - 3. Division 02 Section "Selective Demolition" for photographic documentation before selective demolition operations commence.

1.3 INFORMATIONAL SUBMITTALS

- A. Key Plan: Submit key plan of Project site and building with notation of vantage points marked for location and direction of each photograph. Indicate elevation or story of construction. Include same information as corresponding photographic documentation.
 - 1. Digital Photographs: Submit image files within three days of taking photographs.
 - 2. Digital Camera: Minimum sensor resolution of 8 megapixels.
 - 3. Format: JPG2 with a minimum 3200 by 2400 pixels, uncompressed or layered in unaltered original files, with same aspect ratio as the sensor, uncropped, date and time stamped with key plan location number, in a folder named by date of photograph, accompanied by key plan file.
 - 4. Identification: Provide the following information with each image description in file metadata tag:
 - a. Name of Project.
 - b. Date photograph was taken.
 - c. Description of vantage point, indicating location, direction (by compass point), and elevation or story of construction.
 - d. Unique sequential identifier keyed to accompanying key plan.
- B. Construction Photographs: Submit digital images in JPG2 format of photos taken within the month with the pencil draft of invoice being submitted for review.

- C. Aerial Photographs: If requested by the Owner, submit digital images in JPG2 format as outlined for digital cameras. If it is intended to use drones, Contractor must notify the Owner prior to use.
- 1.4 USAGE RIGHTS
 - A. The photographic documentation ownership shall be the Owner. When necessary, provide or transfer copyright usage rights from photographer to Owner for unlimited reproduction of photographic documentation.
- PART 2 PRODUCTS (not used)

PART 3 - EXECUTION

- 3.1 CONSTRUCTION PHOTOGRAPHS
 - A. Photographer: Designate a competent person to take construction photographs.
 - B. General: Take photographs using the maximum range of depth of field, and that are in focus, to clearly show the Work. Photographs with blurry or out-of-focus areas will not be accepted.
 - C. Digital Images: Submit digital images exactly as originally recorded in the digital camera, without alteration, manipulation, editing, or modifications using image-editing software.
 - 1. Date and Time: Include date and time in file name for each image.
 - 2. Field Office Images: Maintain one set of images accessible in the field office at Project site, available at all times for reference. Identify images in the same manner as those submitted to the Owner.
 - D. Web Cameras: Are required for this project. When required, the system shall provide for time lapse videography with web view access at a minimum.
 - E. Preconstruction Photographs: Before commencement of demolition for renovation projects or starting new construction, take photographs of Project site and surrounding properties, including existing items to remain during construction, from different vantage points.
 - 1. Flag construction limits before taking construction photographs.
 - 2. Take not less than 20 photographs to show existing conditions adjacent to property before starting the Work.
 - 3. Take not less than 20 photographs of existing buildings either on or adjoining property to accurately record physical conditions at start of construction.
 - 4. Take additional photographs as required to record settlement or cracking of adjacent structures, pavements, and improvements.
 - 5. Contractor has the option of video-taping as opposed to digital photos.

- F. Periodic Construction Photographs: Take no less than 20 photographs daily or at significant start and finish points of construction phases with the cutoff date associated with each Application for Payment. Select vantage points to show status of construction and progress since last photographs were taken.
 - 1. Significant Critical activities include but not limited to:
 - a. Commencement of the Work
 - b. Depth and connections of all utilities of subgrade construction.
 - c. Above-grade structural framing.
 - d. Exterior building enclosure.
 - e. Interior Work, through date of Substantial Completion.
- G. Owner-Directed Construction Photographs: From time to time, Owner may instruct photographer about number and frequency of photographs and general directions on vantage points. Select actual vantage points and take photographs to show the status of construction and progress since last photographs were taken.
- H. Time-Lapse Sequence Construction Photographs: Take photographs as indicated, to show status of construction and progress since last photographs were taken.
 - 1. Frequency: Take photographs hourly or as required by industry standards, coinciding with the cutoff date associated with each Application for Payment.
- I. Vantage Points: As identified in the Key Plan and accepted by the Owner, follow select vantage points. During each of the following construction phases, Final Completion Construction Photographs: Take color photographs after date of Substantial Completion for submission as project record documents. Key plan will inform photographer of desired vantage points or as directed by the Owner.
 - 1. Do not include date stamp.
- J. Additional Photographs: Owner may request photographs in addition to periodic photographs specified.
 - 1. Three days' notice will be given, where feasible.
 - 2. In emergency situations, take additional photographs within 24 hours of request.
 - 3. Circumstances that could require additional photographs include, but are not limited to, the following:
 - a. Special events planned at Project site.
 - b. Immediate follow-up when on-site events result in construction damage or losses.
 - c. Photographs to be taken at fabrication locations away from Project site. These photographs are not subject to unit prices or unit-cost allowances.
 - d. Substantial Completion of a major phase or component of the Work.
 - e. Extra record photographs at time of final acceptance.
 - f. Owner's request for special publicity photographs.

END OF SECTION 01 3233

SECTION 01 3300 - SUBMITTAL PROCEDURES

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.

1.2 SECTION INCLUDES

- A. Work of this Section includes all labor, materials, equipment, and services necessary to complete submittal requirements as specified herein, including, but not limited to, the following:
 - 1. Construction Schedule.
 - 2. Schedule of Values.
 - 3. Submittal and Shop Drawing Schedule.
 - 4. Product data.
 - 5. Shop drawings.
 - 6. Coordination drawings.

PART 2 PRODUCTS

2.1 SHOP DRAWINGS AND SAMPLES

- A. General
 - 1. The Contractor shall be responsible for coordinating the schedule for submittal of shop drawings and samples with his progress schedule and the requirements of the Contract Schedule, and submit a coordinated schedule of submission of all shop drawings and samples to the Architect.
 - 2. Failure of the Contractor to schedule and submit shop drawings and samples in ample time for checking, correction, and rechecking will not justify any delay in the Contract Schedule. Allow ample time for items to be tested, including time for retesting if the tests or mock-ups fail.
 - 3. Samples, shop drawings, manufacturers' literature, and other required information shall be submitted in sufficient time to permit proper consideration and action on same before any materials and items are delivered on the work. Stagger submissions so that the Architect can review the documents in an orderly and timely manner. All samples of materials requiring laboratory tests shall be submitted to the laboratory for testing not less than 90 days before such materials are required to be used

in the work. All other samples, manufacturers' literature, and other sample information shall be submitted for approval not less than 30 days before such materials are required to be used in the work.

- 4. Shop drawings for each Section of the work shall be numbered consecutively, and the numbering system shall be retained throughout all revisions. Each drawing shall have a clear space for the stamps of the Contractor, Architect, and one of the Architect's consultants.
- 5. All shop drawings shall be thoroughly checked by the Contractor for compliance with the Contract Documents before submitting them to the Architect and shall bear the Contractor's stamp of approval certifying that they have been so checked. Any shop drawings submitted without this stamp of approval and certification, and shop drawings which, in the Architect's opinion, are incomplete, contain errors or have not been checked, or only checked superficially, will be returned unchecked by the Architect for re-submission by the Contractor.
- In checking shop drawings, the Contractor shall verify all dimensions and 6. field conditions and shall check and coordinate the shop drawings of any Section or trade with the requirements of all other Sections or trades whose work is related thereto, as required for proper and complete installation of the work. The Architect will review shop drawings. The Architect's acceptance of shop drawings is for design only and not method of assembly or erection. Acceptance shall in no way be construed as (1) permitting any departure whatsoever from the Contract Documents: (2) relieving the Contractor of full responsibility for any error in details, dimensions, omissions, or otherwise that may exist; (3) relieving the Contractor of full responsibility for adequate field connections, erection techniques, bracing, or deficiencies in strength; (4) relieving the Contractor of full responsibility for satisfactory performance of all work and coordination with the work of all subcontractors and other contractors; or (5) permitting departure from additional details or instructions previously furnished by the Architect. Acceptance of such drawings shall not be construed as a complete check, nor shall it relieve the Contractor from responsibility for proper fitting of the work, nor from the necessity of furnishing any work which may not be indicated on shop drawings when approved. The Contractor shall be solely responsible for any quantities which may be shown on the shop drawings.
- 7. No work shall be fabricated, manufactured, or installed from shop drawings stamped "Revise and Resubmit" or "Rejected," and such shop drawings shall be corrected and resubmitted by the Contractor until accepted by the Architect. At least one complete set of "No Exceptions Taken" and/or "Make Corrections Noted" shop drawings shall be kept at the site in the Contractor's field office for reference at all times. "Revise and Resubmit" or "Rejected" shop drawings shall not be permitted at the site.

- 8. Submittals Marked "No Exceptions Taken": Submittals which require no corrections by the Architect will be marked "No Exceptions Taken."
- 9. Submittals Marked "Make Corrections Noted": Submittals which require only a minor amount of correcting shall be marked "Make Corrections Noted." This mark shall mean that checking is complete and all corrections are obvious without ambiguity. Fabrication will be allowed on work marked "Make Corrections Noted" provided such action will expedite construction and noted corrections are adhered to. If fabrication is not made strictly in accordance with corrections noted, the item shall be rejected in the field, and the Contractor will be required to replace such work in accordance with corrected submittals.
- 10. Submittals Marked "Revise and Resubmit" or "Rejected": When submittals are contrary to contract requirements or too many corrections are required, they shall be marked "Revise and Resubmit" or "Rejected." No work shall be fabricated under this mark. The Architect shall list his reasons for rejection on the submittals or in the transmittal letter accompanying their return. The submittals must be corrected and resubmitted for approval.
- 11. All shop drawings and samples shall be identified as follows:
 - a. Date of submittal.
 - b. Title of project.
 - c. Name of Contractor and date of his approval.
 - d. Name of subcontractor or supplier and date of submittal to Contractor.
 - e. Number of submissions.
 - f. Any qualification, departure, or deviation from the requirements of the Contract.
 - g. Federal Specification or ASTM number where required.
 - h. Such additional information as may be required by the Specifications for the particular material being furnished.
- 12. If the Contractor wishes to deviate from the materials or details as shown in Specifications or Drawings, he shall submit the proposed deviation with shop drawings and/or samples stating the extent and the materials or details being replaced. The Contractor shall also submit information on the allowed credit or extra cost required for the proposed deviation, and also all information relating to the work of other Sections revised by the proposed deviation.
- 13. The Architect will review and approve shop drawings and samples for approval within 10 working days unless otherwise noted, 15 working days for Divisions 22, 23, 26, 31, 32 and 33, but only for conformance with the design concept of the work and with information contained in the Contract Documents.

- 14. Incomplete shop drawings will be returned without checking for proper submission, and this shall not be considered as cause for delay of the work or extra compensation to the Contractor.
- 15. The Contractor shall submit appropriate transmittal forms with every submittal of shop drawings, manufacturer's literature, and samples. The Contractor shall submit all required shop drawings, manufacturer's literature and samples in accordance with the procedures specified herein.
- 16. Unless otherwise specifically directed by the Architect, make all shop drawings accurately to a scale sufficiently large to show all pertinent features of the item and its method of connection to the work.
- 17. Submit newly prepared information drawn accurately to scale. Highlight, encircle, or otherwise indicate deviations from the Contract Documents. Do not reproduce Contract Documents or copy standard information as the basis of Shop Drawings. Standard information prepared without specific reference to the Project is not a Shop Drawing.
- 18. The Contractor shall submit one copy of each standard referred to in the Specifications (ASTM, Fed. Spec., etc.) with the submission of each respective shop drawing, sample, or literature.
- B. Submission of Shop Drawings
 - 1. Electronic Construction Management Software: All submittals shall be submitted electronically through an online management system such as Procore. The General Contractor shall maintain an electronic website for shop drawings, RFIs, ASIs, etc.
 - 2. Architectural Work: Submit one (1) pdf of each shop drawing to the Architect for approval through Procore or other Electronic Construction Management software. If submission is approved, the Architect will return one (1) pdf stamped "No Exceptions Taken" or "Make Corrections Noted," and the Contractor shall print the required number of copies. In the event the Architect returns one (1) pdf stamped "Revise and Resubmit" or "Rejected," the Contractor shall make indicated changes and resubmit one (1) pdf to the Architect through Procore or other Electronic Construction Management software.
 - 3. Structural Work and Mechanical Work: Submit one (1) pdf of each shop drawing to the Engineer and Architect through Procore or other Electronic Construction Management software. If submission is accepted, the Architect shall return one (1) pdf stamped "No Exceptions Taken" or "Make Corrections Noted," and the Contractor shall print the required number of copies. In the event the Architect returns one (1) pdf stamped "Revise and Resubmit" or "Rejected," the Contractor shall make indicated changes and resubmit one (1) pdf to the Engineer and Architect through Procore or other Electronic Construction Management software.
- 4. Prints: The Contractor shall provide all prints or shop drawings as reasonably required by subcontractors, material suppliers, superintendents, inspectors, and others as required for the work, or as directed by the Architect. The Contractor shall pay all costs in connection with printing and distribution of shop drawings.
- C. Submission of Manufacturer's Literature, Including Catalog, Catalog Cuts, Brochures, Charts, Test Data, and Similar Information
 - 1. Manufacturer's literature will receive consideration only when accompanied by the transmittal form properly filled out, as indicated, and listing each item of literature, as well as the Specification Section and paragraph numbers describing such materials. Any deviations from contract requirements shall be stated on the above form or attached to it.
 - 2. Architectural Work: Submit one (1) pdf copy of manufacturer's literature to the Architect for acceptance through Procore or other Electronic Construction Management software. If submission is accepted, the Architect will return one (1) copy stamped "No Exceptions Taken" or "Make Corrections Noted" through Procore or other Electronic Construction Management software. In the event the Architect returns the literature stamped "Revise and Resubmit" or "Rejected," he will return one (1) copy through Procore or other Electronic Construction Management software only. The Contractor shall resubmit one (1) copy of correct or corrected literature of all submissions stamped by the Architect "Revise and Resubmit" or "Rejected."
 - 3. Structural Work and Mechanical Work: Submit one (1) copy of manufacturer's literature to the Engineer and Architect through Procore or other Electronic Construction Management software. If submission is accepted, the Architect will return one (1) copy through Procore or other Electronic Construction Management software stamped "No Exceptions Taken" or "Make Corrections Noted." In the event the Architect stamps the literature "Revise and Resubmit" or "Rejected," he will return one (1) copy through Procore or other Electronic Construction Management software only. The Contractor shall resubmit one (1) copy of correct or corrected literature to the Engineer for all submissions stamped "Revise and Resubmit" or "Rejected" by the Engineer, with one (1) copy of correct or corrected literature with copy of the transmittal for to the Architect through Procore or other Electronic Construction Management software.
 - 4. All copies of manufacturer's literature required to be resubmitted hereunder shall be original printed material. Reproductions of printed material will not receive consideration.
- D. Submission of Samples
 - 1. All samples shall be submitted in triplicate unless otherwise indicated in the Specifications.

- 2. Samples will receive consideration only when accompanied by the transmittal form properly filled out, as indicated, and listing each sample, as well as the listing of any ASTM, Federal or other standard references specified or applicable and such additional information as may be required by the Specifications for the materials being submitted. Any deviation from the contract requirements shall be so stated on the above form or attached to it.
- 3. The Architect shall have the right to require submission of samples of any materials, whether or not specifically indicated in the various Sections of the Specifications.
- 4. Unless otherwise specified, samples of sufficient size to indicate general visual effect shall be submitted. Where samples must show a range of color, texture, finish, graining, or other similar property, the Contractor shall submit sets of pairs illustrating the full scope of the range.
- 5. One (1) sample of each submission will be returned to the Contractor. Samples stamped "Revise and Resubmit" or "Rejected" by the Architect shall be resubmitted in triplicate by the Contractor.
- 6. All samples stamped "No Exceptions Taken" or "Make Corrections Noted" shall be kept at the site in the Contractor's field office facilities for reference at all times. "Revise and Resubmit" or "Rejected" samples shall not be kept at the site.

2.2 INTEGRATED DRAWINGS

- A. The HVAC subcontractor shall prepare a Drawing or Drawings showing duct work, heating and sprinkler piping. This Drawing shall include location of grilles, registers, etc., and access doors in hung ceilings. Locations shall be fixed by elevations and dimensions from column center lines and/or walls.
- B. The HVAC subcontractor shall prepare and distribute to the Plumbing and Electrical subcontractors, the General Contractor, and to the Architect a reproducible of the above.
- C. The HVAC subcontractor shall lay out on his reproducible the reflected ceiling plan, beam soffit elevations, ceiling heights, roof openings, etc.
- D. The Plumbing subcontractor shall lay out on his reproducible the piping, valves, clean-outs, etc., indicating locations and elevations and shall indicate the necessary access doors.
- E. The Electrical subcontractor shall indicate on his reproducible the fixtures, large conduit runs, clearances, pull boxes, junction boxes, sound system speakers, etc.
- F. The General Contractor shall indicate on his reproducible any structural framing, ceiling hangers, etc.

- G. The General Contractor shall call as many meetings with the subcontractors as are necessary to resolve any conflicts that become apparent. He will call on the services of the Consultant Engineer or Architect where necessary. The General Contractor is responsible for the coordination of the Drawing or Drawings.
- H. On resolution of the conflicts, each subcontractor shall enter his own work on the HVAC subcontractor's reproducible, which shall become the master or integrated Drawings. The master reproducible shall be signed by each contributing subcontractor to indicate his acceptance of the arrangement of the work.
- I. A reproducible copy of the master integrated Drawing will be prepared by the HVAC subcontractor. The General Contractor will make distribution.
- J. Each subcontractor shall prepare his Shop Drawings in accordance with the integrated Drawings. No work will be permitted without approved Shop Drawings. It is therefore essential that this procedure be instituted as quickly as possible.

PART 3 EXECUTION

3.1 COORDINATION OF SUBMITTALS

- A. Prior to submittal for Architect's review, use all means necessary to fully coordinate all material, including the following procedures:
 - 1. Determine and verify all field dimensions and conditions, materials, catalog numbers and similar data.
 - 2. Coordinate as required with all trades and with public agencies involved.
 - 3. Secure all necessary approvals from public agencies and others and signify by stamp, or other means, that they have been secured.
 - 4. Clearly indicate all deviations from the Contract Documents.
- B. Unless otherwise specifically permitted by the Architect, make all submittals in groups containing all associated items; the Architect may reject partial submittals as not complying with the provisions of the Contract Documents.

END OF SECTION 01 3300

SUBMITTED BY: SPECIFICATION SECTION:		SUBCONTRACTOR COMPANY I	VAME
Submittal Type Resubm (Product Data/ Test Data/ Shop Drawings/ Y/N Samples/ Closeout/ Etc.)	nittal? ⊿	Submittal Description (Please be detailed)	Notes/ Comments
Contractor's Received Stamp		Consultant's Received Stamp	Architect's Received Stamp
Contractor's Review Stamp		Consultant's Review Stamp	Architect's Review Stamp

ARPA Renovations Girls Inc of Meriden SUBMITTAL COVER PAGE

The undersigned submits this package and certifies that:

1. Submittal has been reviewed and it is complete to the best of our knowledge and confirms with requirements of Contract Documents except as noted.

2. Required dimensions will be. Have been field verified and are acceptable for installation of proposed products and construction of proposed work.

3. Required quantities for products and materials covered by this submittal have been verified as correct based on current drawings, addendum and bulletins issued.

4. Fabrication processes and construction methods proposed in this submittal are acceptable for this Project and will result in a complete, functional installation.

5. Submittal has been coordinated with other submittals and work and proposed products and construction will properly interface with other construction.

Signature:

Date:

SECTION 01 4000 - QUALITY REQUIREMENTS

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.

1.2 SECTION INCLUDES

- A. This Section includes administrative and procedural requirements for quality assurance and quality control.
- 1.3 RELATED SECTIONS
 - A. Divisions 2 through 32 Sections for specific test and inspection requirements.
- 1.4 DEFINITIONS
 - A. Quality-Assurance Services: Activities, actions, and procedures performed before and during execution of the Work to guard against defects and deficiencies and ensure that proposed construction complies with requirements.
 - B. Quality-Control Services: Tests, inspections, procedures, and related actions during and after execution of the Work to evaluate that completed construction complies with requirements. Services do not include contract enforcement activities performed by Architect.
 - C. Mockups: Full-size, physical example assemblies to illustrate finishes and materials. Mockups are used to verify selections made under Sample submittals, to demonstrate aesthetic effects and, where indicated, qualities of materials and execution, and to review construction, coordination, testing, or operation; they are not Samples. Mockups establish the standard by which the Work will be judged.
 - D. Testing Agency: An entity engaged to perform specific tests, inspections, or both. Testing laboratory shall mean the same as testing agency.

1.5 DELEGATED DESIGN

- A. Performance and Design Criteria: Where professional design services or certifications by a design professional are specifically required of Contractor by the Contract Documents, provide products and systems complying with specific performance and design criteria indicated.
 - 1. If criteria indicated are not sufficient to perform services or certification required, submit a written request for additional information to Architect.

1.6 SUBMITTALS

- A. Qualification Data: For testing agencies specified in "Quality Assurance" Article to demonstrate their capabilities and experience. Include proof of qualifications in the form of a recent report on the inspection of the testing agency by a recognized authority.
- B. Delegated-Design Submittal: In addition to Shop Drawings, Product Data, and other required submittals, submit a statement, signed and sealed by the responsible design professional, for each product and system specifically assigned to Contractor to be designed or certified by a design professional, indicating that the products and systems are in compliance with performance and design criteria indicated. Include list of codes, loads, and other factors used in performing these services.
- C. Schedule of Tests and Inspections: Prepare in tabular form and include the following:
 - 1. Specification Section number and title.
 - 2. Description of test and inspection.
 - 3. Identification of applicable standards.
 - 4. Identification of test and inspection methods.
 - 5. Number of tests and inspections required.
 - 6. Time schedule or time span for tests and inspections.
 - 7. Entity responsible for performing tests and inspections.
 - 8. Requirements for obtaining samples.
 - 9. Unique characteristics of each quality-control service.
- D. Reports: Prepare and submit certified written reports that include the following:
 - 1. Date of issue.
 - 2. Project title and number.
 - 3. Name, address, and telephone number of testing agency.
 - 4. Dates and locations of samples and tests or inspections.
 - 5. Names of individuals making tests and inspections.
 - 6. Description of the Work and test and inspection method.
 - 7. Identification of product and Specification Section.
 - 8. Complete test or inspection data.
 - 9. Test and inspection results and an interpretation of test results.

- 10. Ambient conditions at time of sample taking and testing and inspecting.
- 11. Comments or professional opinion on whether tested or inspected Work complies with the Contract Document requirements.
- 12. Name and signature of laboratory inspector.
- 13. Recommendations on retesting and reinspecting.
- E. Permits, Licenses, and Certificates: For Owner's records, submit copies of permits, licenses, certifications, inspection reports, releases, jurisdictional settlements, notices, receipts for fee payments, judgments, correspondence, records, and similar documents, established for compliance with standards and regulations bearing on performance of the Work.

1.7 QUALITY ASSURANCE

- A. Fabricator Qualifications: A firm experienced in producing products similar to those indicated for this Project and with a record of successful in-service performance, as well as sufficient production capacity to produce required units.
- B. Factory-Authorized Service Representative Qualifications: An authorized representative of manufacturer who is trained and approved by manufacturer to inspect installation of manufacturer's products that are similar in material, design, and extent to those indicated for this Project.
- C. Installer Qualifications: A firm or individual experienced in installing, erecting, or assembling work similar in material, design, and extent to that indicated for this Project, whose work has resulted in construction with a record of successful inservice performance.
- D. Manufacturer Qualifications: A firm experienced in manufacturing products or systems similar to those indicated for this Project and with a record of successful in-service performance.
- E. Professional Engineer Qualifications: A professional engineer who is legally qualified to practice in the State of Connecticut and who is experienced in providing engineering services of the kind indicated. Engineering services are defined as those performed for installations of the system, assembly, or product that are similar to those indicated for this Project in material, design, and extent.
- F. Specialists: Certain sections of the Specifications require that specific construction activities shall be performed by entities who are recognized experts in those operations. Specialists shall satisfy qualification requirements indicated and shall be engaged for the activities indicated.
 - 1. Requirement for specialists shall not supersede building codes and similar regulations governing the Work, nor interfere with local trade-union jurisdictional settlements and similar conventions.

- G. Testing Agency Qualifications: An agency with the experience and capability to conduct testing and inspecting indicated, as documented by ASTM E 548, and that specializes in types of tests and inspections to be performed.
- H. Preconstruction Testing: Testing agency shall perform preconstruction testing for compliance with specified requirements for performance and test methods.
 - 1. Contractor responsibilities include the following:
 - a. Provide test specimens and assemblies representative of proposed materials and construction. Provide sizes and configurations of assemblies to adequately demonstrate capability of product to comply with performance requirements.
 - b. Submit specimens in a timely manner with sufficient time for testing and analyzing results to prevent delaying the Work.
 - c. Fabricate and install test assemblies using installers who will perform the same tasks for Project.
 - d. When testing is complete, remove assemblies; do not reuse materials on Project.
 - 2. Testing Agency Responsibilities: Submit a certified written report of each test, inspection, and similar quality-assurance service to Architect with copy to Contractor. Interpret tests and inspections and state in each report whether tested and inspected work complies with or deviates from the Contract Documents.
- I. Mockups: Before installing portions of the Work requiring mockups, build mockups for each form of construction and finish required to comply with the following requirements, using materials indicated for the completed Work:
 - 1. Build mockups in location and of size indicated or, if not indicated, as directed by Architect.
 - 2. Notify Architect seven days in advance of dates and times when mockups will be constructed.
 - 3. Demonstrate the proposed range of aesthetic effects and workmanship.
 - 4. Obtain Architect's approval of mockups before starting work, fabrication, or construction.
 - 5. Maintain mockups during construction in an undisturbed condition as a standard for judging the completed Work.
 - 6. Demolish and remove mockups, unless otherwise directed by the Architect.

PART 2 PRODUCTS (Not Applicable)

PART 3 EXECUTION

3.1 REPAIR AND PROTECTION

- A. General: On completion of testing, inspecting, sample taking, and similar services, repair damaged construction and restore substrates and finishes.
 - 1. Provide materials and comply with installation requirements specified in other Sections of these Specifications. Restore patched areas and extend restoration into adjoining areas in a manner that eliminates evidence of patching.
- B. Protect construction exposed by or for quality-control service activities.
- C. Repair and protection are Contractor's responsibility, regardless of the assignment of responsibility for quality-control services.

END OF SECTION 01 4000

SECTION 01 5000 - TEMPORARY FACILITIES AND CONTROLS

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.

1.2 SECTION INCLUDES

- A. Work of this Section includes all labor, materials, equipment, and services necessary to complete the temporary facilities and controls as shown on the drawings and specified herein, including, but not limited to, the following:
 - 1. Construction sign.
 - 2. Hoists, stairs, and ladders.
 - 3. Rodent control.
 - 4. Construction fence.
 - 5. Field office.
 - 6. Fire protection.
 - 7. Temporary utilities.
 - 8. Temporary toilets.
 - 9. Temporary site access.
 - 10. Security.
 - 11. Water and snow control.
 - 12. Environmental controls.

1.3 RELATED SECTIONS

- A. Product Requirements Section 01 6000.
- B. Execution Requirements Section 01 7300, for progress cleaning.
- PART 2 PRODUCTS
- 2.1 GENERAL
 - A. Arrange for and provide temporary facilities and controls as specified herein and as required for the proper and expeditious prosecution of the work. Pay all costs, except as otherwise specified, until final acceptance of the work unless

the Owner makes arrangements for the use of completed portions of the work after substantial completion.

- B. Make all temporary connections to utilities and services in locations acceptable to the local authorities having jurisdiction thereof; furnish all necessary labor and materials, and make all installations in a manner subject to the acceptance of such authorities; maintain such connections; remove the temporary installation and connections when no longer required; restore the services and sources of supply to proper operating condition.
- C. Unless otherwise noted, pay all costs for temporary electrical power, temporary water, and temporary heating; provide metering as necessary.
- D. A Staging Plan shall be submitted by the Contractor for approval by the Owner. The Staging Plan shall locate all temporary facilities and services, including parking for the Contractor's employees, within the limits of the staging areas, and shall allot ground space to Subcontractors for storage of materials, and the erection of sheds and tool houses. Materials and equipment can only be stored in the staging area. No parking for Contractor's or Subcontractors' employees' vehicles will be allowed in undesignated parking areas. The staging area shall be maintained in good repair, free of mud and standing water, and passable at all times. All materials stored within the project site are the responsibility of the Contractor. At the completion of the work, the staging areas shall be restored to their original condition, gravel removed, topsoil replaced and graded and reseeded.

2.2 PROJECT IDENTIFICATION

- A. No signs or advertisements will be allowed to be displayed on the premises without the approval of the Owner.
- B. One construction sign on the site shall be provided by the Contractor and shall be subject to the review of the Architect and the approval of the Owner.
- C. Erect the construction sign on the site where directed by the Architect. Provide sign approximately 4 ft. x 8 ft. in size, of 3/4 in. plywood with structural supports. Use Douglas Fir Overlaid Plywood, Grade B-B high density, exterior, good two sides, complying with PS-1. The sign shall be primed and given two coats of alkyd white paint. Lettering shall be black of a type, size, and lay-out as directed by the Owner. Sign shall contain the name of the Building, Owner, Architect, Contractor, and such other information as the Architect or Owner may require.

2.3 MATERIAL HOIST

A. Provide a material hoist as required for use by all trades. Provide all necessary guards, signals, safety devices, and so on, required for safe operations, and suitable runways from the hoists to each floor level and roof. The construction and operation of the material hoist shall comply with all applicable requirements

of ANSI A10.5, the ACG Manual of Accident Prevention in Construction and to all applicable state and municipal codes. Prohibit the use of the material hoist for transporting personnel.

2.4 RODENT CONTROL

A. Institute an effective program of rodent control for the entire site within the construction limits. Cooperate with local authorities and provide the regular services of an experienced exterminator who shall visit the site at least once a month for the entire construction period. Provide marked metal containers for all edible rubbish and enforce their use by all employees. Containers shall be emptied and the contents removed from the site as often as required to maintain an adequate rodent control program. If the program of rodent control used is not effective, take whatever steps are necessary to rid the project of rodents, and such action shall not be the basis of a claim for additional compensation or damages.

2.5 TEMPORARY CONSTRUCTION OPENINGS

A. Provide openings in slabs, walls, and partitions where required for moving in large pieces of equipment of all types. Close and/or restore all openings and finish them after the equipment is in place. Structural modification, if required, shall be subject to review by the Architect.

2.6 TEMPORARY FENCE

- A. Provide and maintain an 8 foot high temporary fence to enclose the area at the job site and to guard and close effectively the designated area. Provide gates at locations where required for access to the enclosed area. Gates shall be cross-braced, hung on heavy strap hinges, and shall have hasps and padlocks. Submit shop drawings of fence and gates for review of Architect and Owner. Paint the fence with two coats of an approved paint.
- B. Remove the fence upon completion of the work or at such time before final completion as directed by the Owner.

2.7 TEMPORARY FIELD OFFICES

A. Provide and maintain a field office with a telephone, internet access and computers at the job site with not less than 200 square feet of space. The office shall be complete with light, heat, air conditioning, toilet facilities, electric water cooler, plan racks, four-drawer metal file with lock, shelves for samples, tables, chairs, and janitor service. When it becomes possible to establish an office in the building, office accommodation of approximately the same size as those in the field offices, including the services above, shall be provided and maintained until the issuance of a certificate of substantial completion. Temporary offices shall be removed when no longer required. Provide a telephone and fax line with machine and pay all charges for installation and calls, including long distance calls.

2.8 FIRE PROTECTION

- A. Provide and maintain adequate fire protection, ready for instant use, distributed around the project.
- B. Make arrangements for periodical inspection by local fire protection authorities and insurance underwriters inspections. Cooperate with said authorities and promptly carry out their recommendations.
- C. Open fire will not be permitted within the building enclosure or on the project site.

2.9 TEMPORARY HEAT AND VENTILATION

- A. Provide temporary heat as required during construction to protect the work from freezing or frost damage, and as necessary to ensure suitable working conditions for the construction operations of all trades. In areas of the building where work is being conducted, the temperature shall be maintained as specified in the various sections of the Specifications, but not less than 45 degrees Fahrenheit. Under no circumstances shall the temperature be allowed to reach a level that will cause damage to any portion of the work which may be subject to damage by low temperatures.
- B. Until the building, or any major portion thereof, is enclosed, temporary heating shall be by smokeless portable unit heaters of type listed by Underwriter's Laboratories, Factory Mutual, and the Fire Marshall. Pay for fuel, maintenance, and attendants required in connection with the portable unit heaters. Interior or exterior surfaces damaged by the use of these space heaters shall be replaced by new materials or be refinished.
- C. The building shall be considered enclosed when it has reached the stage when exterior walls have been erected, the roof substantially completed, exterior openings closed up either by the permanently glazed windows and doors, or by adequate temporary closing, and the building is ready for interior masonry and plastering operations.
- D. After the building, or any major portion thereof, has been enclosed, the permanent heating system as specified below may be used for temporary heat.
- E. When the permanent heating system, or a suitable portion thereof, is in operating condition, the system may be used for temporary heating, provided that the Contractor assumes full responsibility for the entire heating system, and pays all costs for fuel, operation, maintenance, and restoration of the system.
- F. Provide adequate ventilation as required to keep the temperature of the building within 10 degrees Fahrenheit of the ambient outdoor temperature when such ambient temperature exceeds 70 degrees Fahrenheit, and to prevent accumulation of excess moisture or to prevent excess thermal movement in the building.

- G. When the permanent air circulation system, or a suitable portion thereof, is in operating condition, it may be used without refrigeration or chilling, provided that the Contractor assumes full responsibility for the system which he is using, and pays costs for power, operation, maintenance, and restoration of the system. Provide temporary filters to adequately filter air being distributed through the duct work to the supply outlets; disposable filters shall be placed in front of all exhaust registers to keep construction dirt out of exhaust duct work. The Contractor shall thoroughly clean the interior of the air handling units and duct work prior to acceptance of the work.
- H. Upon conclusion of the temporary heating period, remove all temporary piping, temporary heating units, or other equipment and pay all costs in connection with repairing any damage caused by the installation or removal of temporary heating equipment. Thoroughly clean and recondition those parts of permanent heating and air circulation systems used for temporary service.

2.10 TEMPORARY LIGHT AND POWER

- A. Make all arrangements with the local electric company for temporary electrical service to the construction site; provide all equipment necessary for temporary power and lighting; and pay all charges for this equipment, the installation thereof, and for current used. The electrical service shall be of 120v and 240v for single phase loads up to 30 amps for all construction tools and equipment without overloading the temporary facilities and shall be made available for power, lighting, and construction operations of all trades.
- B. In addition to the electrical service, provide power distribution as required throughout structure. The terminations of power distribution shall be at convenient locations in the building. Terminations shall be provided for each voltage supply complete with circuit breakers, disconnect switches, and other electrical devices as required to protect the power supply system.
 - 1. Provide double duplex outlets at not more than 200' o.c. both directions throughout this building.
- C. A temporary lighting system shall be furnished, installed, and maintained as required to satisfy minimum requirements of safety and security. The temporary lighting system shall afford general illumination in all building areas and shall supply not less than 150 watt lamps on 30' centers both directions of floor area for illumination in the areas of the building where work is being performed.
- D. All temporary equipment and wiring for power and lighting shall be in accordance with the applicable provisions of the governing codes. All temporary wiring shall be maintained in a safe manner and used so as not to constitute a hazard to persons or property.
- E. When the permanent electrical power and lighting systems are in operating condition, they may be used for temporary power and lighting for construction purposes, provided that the Contractor assumes full responsibility for the entire

power and lighting system, and pays costs for power, operations, maintenance, and restoration of the system.

- 2.11 TEMPORARY ACCESS TO SITE
 - A. Construct and maintain in good usable condition all required temporary access to site, and, when no longer required, remove all temporary construction and restore the site.
 - B. Where streets now in use are within or adjacent to the work, keep the passageways of such streets open to vehicular and pedestrian traffic to building fronting thereon. Maintain constant access for police, fire and ambulance service.
 - C. Mud carried off the site and into public roads shall be removed immediately by the Contractor.
 - D. Access to the site for delivery of construction material or equipment shall be made only from locations designated by the Architect.
- 2.12 TEMPORARY STAIRS, LADDERS, RAMPS, SIDEWALK BRIDGING AND RUNWAYS
 - A. Provide and maintain all equipment such as temporary stairs, ladders, ramps, runways, and chutes as required for the proper execution of the work.
 - B. All such apparatus, equipment, and construction shall meet all requirements of the Labor Law and other state or local laws applicable thereto.
 - C. As soon as permanent stairs are erected, provide temporary protective treads, handrails, and shaft protection.
 - D. Covered Walkway: Erect a structurally adequate, protective, covered walkway for passage of persons along adjacent public street. Coordinate with entrance gates, other facilities, and obstructions. Comply with regulations of authorities having jurisdiction.
 - 1. Construct covered walkways using scaffold or shoring framing.
 - 2. Provide wood plank overhead decking, protective plywood enclosure walls, handrails, barricades, warning signs, lights, safe and well drained walkways, and similar provisions for protection and safe passage.
 - 3. Extend back wall beyond the structure to complete enclosure fence.
 - 4. Paint and maintain in a manner approved by Owner and Architect.
 - 5. For safety barriers, sidewalk bridges, and similar uses, provide minimum 5/8" thick exterior plywood.

2.13 TEMPORARY TOILETS

A. Provide and maintain in a sanitary condition enclosed weathertight toilets for the use of all construction personnel at a location within the contract limits. Upon completion of the work, toilets shall be removed. Installation shall be in accordance with all applicable codes and regulations of authorities having jurisdiction. The number of toilet rooms required shall be in accordance with the ANSI Standard Safety Code for Building Construction or other local authorities.

2.14 TEMPORARY WATER SERVICE

- A. Provide at a point within 10 feet of the building (or buildings) all water necessary for construction purposes. Make all temporary connections to existing mains; provide temporary meter; and make arrangements to pay for the temporary water service including cost of installation, maintenance thereof, and water used.
- B. Furnish drinking water with suitable containers and cups for use of employees. Drinking water dispensers shall be conveniently located in the building where work is in progress.
- C. When the permanent water supply and distribution system has been installed, it may be used as a source of water for construction purposes, provided that the Contractor assumes full responsibility for the entire water distribution system, and pays costs for operation, maintenance, and restoration of the system including the cost of water used.
- D. At the completion of the construction work or at such time after the Contractor makes use of the permanent water installation, all temporary water service equipment and piping shall be removed, and all worn or damaged parts of the permanent system shall be replaced and equipment placed in first class condition equal to new.

2.15 SECURITY

- A. Provide sufficient watchman service to prevent illegal entry or damage during nights, holidays, or other periods when work is not being executed, and such other control watchmen as required during working hours.
- B. Provide all temporary enclosures required for protecting the project from the exterior, for providing passageways, for the protection of openings both exterior and interior, and any other location where temporary enclosures and protection may be required.
- C. Take adequate precautions against fire; keep flammable material at an absolute minimum; and ensure that such material is properly handled and stored. Except as otherwise provided herein, do not permit fires to be built or open salamanders to be used in any part of the work.

2.16 WATER AND SNOW CONTROL

- A. From the commencement of the construction to the completion of the work, keep all parts of the site and the project free from accumulation of water, and supply, maintain, and operate all necessary pumping and bailing equipment.
- B. Remove snow and ice as necessary for the protection and prosecution of the work, and protect the work against weather damage.
- C. The Contractor shall take over responsibility for site drainage upon entering the premises and shall maintain such drainage until completion of the work so as not to adversely affect the adjacent areas.

2.17 ENVIRONMENTAL CONTROLS

- A. The Contractor shall comply with all applicable Federal, State and local laws, regulations, ordinances, codes and standards concerning environment control. Particular attention shall be given, without limitations, to:
 - 1. Minimization of dust, containment of chemical vapors, control of engine exhaust gases, and control of smoke from temporary heaters.
 - 2. Reduction of water pollution by control of sanitary facilities, proper storage of fuels and other potential contaminants, and prevention of siltation from land erosion.
 - 3. Minimization of noise levels.
 - 4. Proper and legal disposal, off site unless otherwise provided, of waste and spoil resulting from construction activities.

PART 3 EXECUTION

3.1 REMOVAL

A. Maintain all temporary facilities and controls as long as needed for the safe and proper completion of the work. Remove all such temporary facilities and controls as rapidly as progress of the work will permit or as directed by the Architect.

END OF SECTION 01 5000

SECTION 01 5640 – TREE PROTECTION AND PRESERVATION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. This Section includes the protection and stress reduction of existing trees and vegetation that interfere with, or are affected by, execution of the Work, whether temporary or permanent. Work is to be coordinated with the contract documents which shall include a tree preservation plan authored by a certified arborist.
- B. The following specifications apply to work of the related to protection and stress reduction measures and coordination and oversight of the tree preservation Plan by the Owner. This work includes but is not limited to the following:
 - 1. Coordination of Temporary Tree and Plant Protection
 - 2. Selective tree removals for "Removal By Arborist" (RBA) (Contract Arborist) within Tree Protection Areas (TPAs)
 - 3. Root Pruning
 - 4. Temporary Site and Tree Protection Fencing and temporary sign installation referenced in Section 01-5000 Temporary Facilities and Controls
 - 5. Composted Mulching
 - 6. Liquid subsurface fertilization
 - 7. Temporary Limb Guying or Clearance Pruning for construction access
 - 8. Seasonal Supplemental Watering
 - 9. Monitoring and Treatment of Tree Health
 - 10. Supersonic Air Tool (SSAT) and Hand Excavation within the Critical Root Zones (CRZs)

1.3 DEFINITIONS

- A. Certified Arborist: Credential of an individual arborist issued and administered by the International Society of Arboriculture. This credential must be current and valid to qualify to use the copyrighted designation of "Certified Arborist". Refer to <u>www.isa-arbor.com</u> for additional information.
- B. Contract Arborist: Arboricultural firm contracted to implement the approved tree preservation plans on site. All crews' conduction arboricultural operations on site shall consist of at least one Certified Arborist who directly oversees all work by that crew. Arboricultural operations include, but are not limited to, pruning, tree protection device

installation and maintenance (fence, matting, etc.), root pruning, air tool root excavation/exploration (SSAT), soil care activities, soil testing, mulch application, tree inspections, pesticide/chemical applications and tree removal. Special qualifications submittal is required for review and approval below. Contract Arborist will be sub-contracted by the Contractor and cannot also be the individual or entity contracted as the Project Arborist for the Architect and/or Owner.

- C. Tree Protection Area (TPA): Area indicated on Drawings surrounding individual trees or groups of trees to be protected during construction.
- D. Supersonic Airtool (SSAT): Hand held tool designed to focus highly compressed air (90-125 psi) provided from a large air compressor (185-375 cfm) at speeds close to 1400 mph at the tip of the tool. Widely used by arboricultural firms and consultants for multiple purposes including but not limited to: root collar investigation, CRZ investigation, root pruning (especially large roots > 1.5" diameter or were existing underground cables or conduits are located, radial mulching and restoration of compacted soils, excavation for utilities within protected CRZs to minimize root damage from constriction.
- E. Tree Removal by Arborist: Action whereby the Contract Arborist removes trees designated for "Removal by Arborist" selected from inside the TPAs. Trees shall be taken down by hand sectionally, or directionally felled to minimize damage to adjacent tree canopies, root systems, or adjacent structures. Work shall be completed by a qualified contract arborist.
- F. Crown Pruning: Action by the Contract Arborist of pruning specific tree limbs to improve tree health, reduce hazard, and / or provide construction clearance.
- G. Supportive Cabling: Installation of supportive cabling for designated tree branches due to weak branch attachments.
- H. Root Pruning: Action indicated on Drawings to provide a more suitable cut for protected tree roots to minimize ripped or torn roots during excavations and grading with standard construction equipment. Various methods may be used.
- I. Mulching of Trees: Application of a wood mulch product to areas surrounding designated trees. Mulch increases moisture-holding capacity, helps mitigate soil compaction, and increases needed soil organic composition.
- J. Soil Amendments: Various product components applied to existing soil environment of protected trees, as indicated on Plan Notes.
- K. Tree Growth Regulator (*Paclobutrazol*): Products applied to designated trees used to regulate plant growth in such a way as to restrict canopy growth and free stored or produced energy for other uses in the tree. For highly impacted trees, more energy may be available for fibrous root growth (to combat root loss), thicker darker leaves (allowing for increased photosynthesis, and increased drought tolerance), and pest tolerance (often an issue with construction stressed trees); among other potential benefits.

- L. Limits of Disturbance (LOD) (also called Limits of Construction): Specific outer limits of all construction activities for the entire project.
- M. DBH (Diameter at Breast Height): Tree trunk diameter measured at 4.5 feet above grade.

1.4 SUBMITTALS

- A. The Contract Arborist shall provide submittals as follows:
 - 1. Product Data: For each type of product indicated
 - 2. Certification: For each phase, the Contract Arborist shall certify for each tree designated to remain has been protected during construction according to recognized standards and that trees were promptly and properly treated and repaired when damaged.
 - 3. Qualification Data: For Contract Arborist Firm Qualifications, submit firm and individual qualifications as follows:
 - a. Submit a minimum of two resumes and detailed qualifications from staff or team individuals assigned to this project as detailed under Quality Assurance below. Due to the complexity of this project, standard arboricultural experience may not qualify.
 - b. Provide references for above from a minimum of three commercial, nongovernmental or governmental projects for whom similar tree preservation programs have been successfully implemented. Include the following information:
 - 1) Project Name, size and scope
 - 2) Number and species of trees involved
 - 3) Relevant photos or aerials
 - 4) Scope of services provided
 - 5) Name and contact for project owner, designer, or contractor.
 - 4. Pedestrian / Property Protection Plan: Contract Arborist to submit a written plan describing all protective measures proposed to be used. Protection measures shall be required for all on-site tree care activities including but not limited to Supersonic Airtool excavation, root pruning, canopy pruning, etc. to minimize potential impact to pedestrians and property.
 - 5. Maintenance Prescription: Contract Arborist shall submit for care and protection of trees as a result of construction, changes in weather patterns or events, and response in health from individual trees during and after completing the Work.
 - 6. Soil Samples: Submit soil sample for analysis during site work phase of this project. Take representative soil samples from all areas of protected trees (landscape areas and street tree planting pits). Samples and procedures per local cooperative extension shall be followed. Forward reports to Engineer and Owner.

- 7. Soil Amendments: Contract Arborist shall submit specific fertilizer formulations, application rates and methods for review by Project Arborist. All fertilization and soil amendments shall be in conformance with soil test results.
- 8. Site Documentation: Submit weekly reports to the Owner containing complete documentation of all tree impacts and tree preservation activities including but not limited to: root pruning, tree protection fencing, excavation within critical root zones, tree fertilization or other treatments, etc. Documentation shall include tree numbers of trees impacted and / or treated. Complete daily photographic record is also required.
- 9. Existing Conditions: Documentation of existing trees and plantings indicated to remain, which establishes preconstruction conditions that might be misconstrued as damaged caused by construction activities.
 - a. Use sufficiently detailed photographs or videotape.
 - b. Include plans and notations to indicate specific wounds and damage conditions of each tree or other plants designated to remain.
- 10. Tree and shrub removal of additional plants not under base contract will require a "request to remove plantings" form to be submitted to the Owner for approval prior to starting the removal.

1.5 QUALITY ASSURANCE

- A. Certified Arborist (individual) Qualifications: An arborist certified by the International Society of Arboriculture (ISA) and licensed in the jurisdiction where project is located. All work performed by Contract Arborist including any oversight and documentation work, shall be performed or directly supervised by at least one on-site arborist with these minimum qualifications.
- B. Contract Arborist Firm Qualifications:
 - 1. Contract Arborist Firm shall comply with the following:
 - a. Established business with documented experience of at least five years.
 - b. Experience working on a minimum of three commercial, nongovernmental or governmental projects where similar tree preservation programs have been successfully implemented.
 - c. Properly licensed and insured to perform arboricultural work in the jurisdiction where the project is located.
 - 2. Provide names of each individual to comply with the following:
 - a. Minimum BS degrees in forestry, arboriculture, or related field and Certification in ISA.
 - b. Resumes should reflect combined 10 years full time experience on similar tree preservation projects.
 - c. Provide individual(s) names, certifications, and each anticipated role in this project. "Role(s)" shall be defined as one or more of the following:
 - 1) Project Manager
 - 2) Technical Oversight
 - 3) Field Arborist / Technician

- 3. For each staff member, list a minimum of three construction projects and a minimum three years' experience in the following technical applications:
 - a. Soil amendment prescriptions and applications
 - b. Supersonic Airtool Excavations for underground utilities exceeding 24" depth.
 - c. Root Protection Matting or similar applications
- C. Part of this work to extent referenced shall include but not be limited to the following:
 - 1. ANSI A300 Standard Practices for Trees, Shrubs, and Other Woody Plant Maintenance.
 - 2. Part 1-2001, Tree Pruning;
 - 3. Part 2-3004, Fertilization;
 - 4. Part 3-2000, Cabling, Bracing, Guying of Established Trees;
 - 5. Part 4-2002, Lightning Protection Systems;
 - 6. ANSI Z133.1 1994 and most recent updates, Tree Care Operations Safety Requirements
- D. Fertilizer and pesticide will be applied in strict accordance with the manufacturers label instructions and applicable federal, state, and local requirements. Fertilizer, soil conditioners, and pesticide applications must be approved by the owner prior to application. Safety Data Sheets (SDS) will be available for fertilizers and pesticides in the Contract Arborists' possession while on the site.
- E. Pre-Construction Meeting: Conduct meeting at the project site prior to commencement of construction related activities.
 - 1. Contract Arborist, Project Arborist, Project Design Team, Owner and Contractors shall attend.
 - 2. Review methods and procedures related to tree protection and preservation including, but not limited to, the following:
 - a. Site Logistics Plan
 - b. Construction schedule verify availability of material, personnel, and equipment needed to make progress and avoid delays.
 - c. Enforcement of requirements for tree protection areas.
 - d. Responsibilities of all parties, including coordination, access and timing requirements.
 - e. Field quality control

1.6 PROJECT CONDITIONS

- A. The following practices are prohibited within all tree protection areas except as specifically indicated herein:
 - 1. Storage or stockpiling of construction materials, chemicals, debris, or excavation materials.
 - 2. Parking vehicles, trailers or equipment.
 - 3. Foot traffic.
 - 4. Erection of sheds or structures.

- 5. Impoundment or discharge of water.
- 6. Excavation or other hand or mechanical digging unless otherwise indicated.
- 7. Attachment of signs to or wrapping materials around trees or plants unless otherwise indicated.
- B. Do not direct vehicle or equipment exhaust toward protection zones.
- C. Prohibit heat sources, flames, ignition sources, and smoking within or near protection zones and organic mulch.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. Temporary Tree Protection Fence
 - 1. Chain-Link Fence: Follow requirements on fencing outlined in 01-5000 Temporary Facilities and in 01-1000 Summary.
- B. Wood Chip Mulch
 - Double ground hardwood, aged a minimum 6 months from production, free from deleterious materials. Green chips or mulch not aged at least 6 months shall not be used. No walnut mulch shall be used. Submittal shall include original material source(s), number and type of grindings / chippings, duration of aging, timing of turning / aeration.
- C. Hardwood Destruction Borer / Beetle Control: Bifenthrin, such as Onyx or equivalent. Applied per label.
- D. Tree Growth Regulator (Paclobutrazol)
 - Paclobutrazol is a compound used to regulate plant growth in such a way as to restrict canopy growth and free stored or produced energy for other uses in the tree. For highly impacted trees, this means more energy may be made available for fibrous root growth (to combat root loss), thicker darker leaves (allowing for increased photosynthesis, and increased drought tolerance), and pest suppression (often an issue with construction stressed trees); among countless other potential benefits. Trade name Cambistat® or equal.
- E. Soil Care/Soil Amendments
 - 1. Fertilizer and soil amendment selection shall be based upon soil test results and recommendations.

PART 3 - EXECUTION

3.1 TREE REMOVAL

A. See Section on Demolition for specifics on tree shrub or hedge removal.

3.2 TREE PROTECTION AND STRESS REDUCTION MEASURES

- A. General
 - 1. Installation/implementation of the following measures shall be performed in the field by and ISA Certified Arborist as provided by the Contract Arborist
 - 2. All work, substitutions and /or modifications shall be subject to review and approval by the Owner.
 - 3. All work shall conform to applicable federal, state and local regulations and industry standards.
 - 4. The Contract Arborist shall be responsible for all items in this section.
- B. Coordination of Tree preservation plan. The work of the Contract Arborist coordination to include but not limited to the following:
 - 1. Existing underground utility marker conflicts brought to the attention of the Contractor for resolution as well uncovered underground utilities as a result of work.
 - 2. Coordinate necessary survey layout of proposed construction elements in order to provide accurate locations for tree protection measures.
 - 3. Layout location of designated tree protection based upon proposed construction and methods of construction for that area.
 - 4. Site walk with Owner and Site Superintendent to verify location of all tree protection measures prior to execution.
 - 5. Notify Site Superintendent and Owner if construction adjacent to tree protection does not appear to follow specifications or prior agreement or conflicts with tree protection seem eminent.
 - 6. Coordinate with Site Superintendent and Owner, for access of deliveries, crews, equipment, start up, and cleanup of each item of work.
 - 7. Provide "as built" of any change to location of tree protection.
 - 8. Attend progress meetings as requested.
 - 9. Provide submittals as required.

- 10. Notify Superintendent and Owner of any breach or damage to tree protection requiring attention.
- C. Pruning and Supportive Cabling
 - 1. Specific canopy pruning for tree health, risk reduction, and construction clearance per Contract documents
 - 2. Size, health, species, and impact from proposed construction will be taken into consideration in determining pruning type for each designated tree. Risk Reduction Pruning will remove dead, dying, and declining limbs 2" diameter and larger. No interior green branching including sprouts will be removed unless approved by Contract Arborist.
 - 3. Contractor, Contract Arborist, and Owner shall meet at site to determine overhead clearance conflicts between trees and construction equipment/activities to prevent breakage, impacts, or aesthetic concerns. All work shall conform to ANSI A-300 arboriculture standards. An aerial assessment shall be made for all trees climbed to report any structural weakness of concern to the Owner.
 - 4. Prior to climbing any tree a risk assessment will be performed using visual, sounding, or basic drilling as needed by the Contract Arborist. Trees deemed high risk should not be climbed; alternate methods should be used and the tree reported to the Owner immediately.
 - 5. Supportive Cabling of weak unions may be recommended by the Contract Arborist if the need is discovered during pruning operations. ANSI Standards apply. Cabling may be included only if submitted to the Engineer and approved by the Owner.
- D. Root Prune
 - 1. Purpose of the root pruning is to provide a more suitable cut so as to not rip or tear roots during excavations and grading with standard construction equipment. The exact location and depth along the LOD or edge of utility excavation will be determined during the layout by a Certified Arborist.
 - 2. Root Pruning for urban sites with specimen trees or for transplanting requires the use of SSAT excavation for hand pruning. Refer to SSAT specifications in the section
 - 3. Sufficient moisture is necessary for reducing the level of dust, increase work efficiency, and provide a hospitable environment of the tree roots and pedestrians.
 - 4. At a pre-work site inspection by the Contract Arborist more than 72 hours in advance of work start, subsurface probing to 24-36" with a tile probe or similar method will determine if sufficient soil moisture exists. If sufficient moisture is not found, immediate coordination with the site managers shall be made to irrigate the proposed work areas. Methodology may be soaker hose, sprinklers, soaker

cans with small drilled holes to release water slowly or other methods. A second follow up inspection shall be made to determine final sufficiency to begin.

- 5. All root pruning operations shall be performed by the Contract Arborist and directed in the field by and ISA Certified Arborist with documented experience in similar SSAT excavation and root pruning.
- E. Temporary Tree Protection Fence
 - 1. Type and placement of fence to be designated on the Preservation Plans and Details.
 - 2. Attach tree protection area signs at30' feet spacing, facing construction activity. For fence lower than 6' feet in height, attach owner provided flagging as directed. Consult with the Owner for sign content.
 - 3. Tree protection area signs shall be high visibility and all weather to last duration of the project / phase.
 - 4. Install tree protection after root pruning if shown, and prior to all other mobilization such as demolition, clearing and/or excavation.
 - 5. Install tree protection at 6" 12" outside (construction side) of the Root Prune line or within the Root Prune Trench.
 - 6. Silt fence will be outside (construction side) the tree protection fence, unless super silt fence is used in lieu of tree protection. Trenchless installation method shall be employed per Detail if Root Protection Matting is designated.
 - 7. Exact placement of fence will be determined in walk-through with Contractor, Project Arborist, Contract Arborist, Engineer, and Owner.
 - 8. Sequencing of the tree protection fence will be determined during the initial site walk. In any case, no construction activities shall occur in each phase or section until approved protection is installed.
- F. Hand Excavation within Tree Protection Areas
 - 1. For excavation within the critical root zone areas of trees to remain, the intent is to minimize tree and root damage from excavation activities.
 - 2. Excavation shall be performed using SSAT, hand tools (shovels, etc.), or other approved non-damaging method. Roots shall not be damaged by the excavation except for approved root pruning.
 - 3. Refer to "Supersonic Airtool Excavation" and "Construction Oversight by Arborist" specifications in this section for additional requirements.
 - 4. All work shall be directly supervised by Contract Arborist in collaboration with the Owner's trades and subcontractors.

- 5. RPM (Root Protection Matting) shall be installed along trench sides to allow for temporary soil stockpile and access.
- 6. Excavate along the edge of the proposed trench closest to the trees to be protected as shown on the plans. Roots shall be uncovered and care taken to avoid damage to roots and bark.
- 7. Contract Arborist shall prune the exposed roots. Excavation shall not extend beyond the line where roots were pruned.
- 8. Contractor may proceed with conventional excavation methods or with hand excavation methods if clearance to the tree is inadequate for equipment access.
- 9. No roots shall be cut by the contractor.
- G. Supersonic Airtool (SSAT) Excavation
 - 1. Refer to "Hand Excavation within Tree Protection Areas" specification in this section for additional requirements
 - 2. At a minimum, all SSAT work shall include the use of a barrier system such as temporary walls or tents to protect property and pedestrians from flying debris.
 - 3. Excavate along the edge of the proposed trench closest to the trees to be protected as shown on the plans. Roots shall be uncovered and care taken to avoid damage to roots and bark.
 - 4. Excavation shall proceed per the "Hand Excavation within Tree Protection Areas" specification in this section.
- H. Wood Chip Mulch
 - 1. Mulching for the duration of construction for protection and stress reduction. Mulching will increase moisture-holding capacity, minimize soil compaction, and increase needed organic composition. Mulch shall meet the specifications and shall be three (3) inches in depth.
 - 2. For individual trees designated on the TPAK within the TPS or curvilinear TPA install mulch to a radius equal to trunk diameter inches equated to mulch ring diameter in feet (24" inch trunk diameter = 24' feet diameter mulch ring). Where planting pit areas are restricted by hardscape, mulch the greatest area possible.
 - 3. For privately owned trees, any installation is contingent upon receipt of owner's permission. Owner may decline.
 - 4. For linear TPAs along LOD Install mulch strips a minimum 10' feet wide the length of critical root zones along the outside of the LOD/Root Prune line (just inside the Tree Protection Zone) for designated significant trees impacted by proposed construction.

- 5. Motorized equipment shall not enter the Tree Protection Area (TPA) unless specifically approved by the Project Arborist and specific conditions met (RPM, AlturnaMATS, etc.). Any such motorized equipment shall be operated by a certified arborist while inside the TPA.
- 6. Do not allow mulch to contact trunk / roof flare.
- 7. Mulch depth shall be 3" inches.
- I. Tree Growth Regulator (*Paclobutrazol*)
 - 1. Paclobutrazol is a compound used to regulate plant growth in such a way as to restrict canopy growth and free stored or produced energy for other uses in the tree. For highly impacted trees, this means more energy may be made available for fibrous root growth (to combat root loss), thicker darker leaves allowing for increased photosynthesis, and increased drought tolerance.
 - 2. Specific methods and dosages are contained on the label and are determined by size and species, and applied by a state licensed pesticide applicator. Designated trees are shown on the Tree Protection Action Key (TPAK).
- J. Supplemental Watering
 - 1. This action is for high impact trees of significance during seasonal drought times of project construction. Based upon the number and size of trees various strategies can be considered to maintain adequate soil moisture during these times. These strategies may include but are not limited to the following:
 - a. Fire hydrant connection battery powered timer and drip irrigation hose/tubing;
 - b. Water tank trunk and hand applied as directed;
 - c. Temporary above grade poly tank with battery-powered timers for drip or soaker hoses at each TPA.
 - d. 30-50 gallon watering cans with 6 8 drilled holes in bottom to allow slow seeping of water; spacing and rotation to reach desired gallons. Equivalent means of affectively watering trees as approved by Engineer or Project Arborist.
 - 2. Trees requiring this treatment are indicated in the TPAK. Other trees will not receive this treatment.
 - 3. Drought times shall be defined as:
 - a. Periods during the growing season of two weeks or longer, where daytime high temperatures reach 80 degrees Fahrenheit or higher and less that ³/₄" rainfall are recorded per week. Or,
 - b. Periods during the growing season designated as "abnormally dry" or "drought" of any severity, by the U.S. Drought Monitor: <u>http://droughtmonitor.unl.edu/</u> Or,
 - c. Any period of extraordinary circumstance, as determined by the project arborist or engineer

- A prescription for the number of gallons and strategy for watering designated trees will be developed. Large mature trees with impacts to root systems require as much as 100 – 250 gallons per week during 90 degree days during summer drought times.
- 5. Periodic inspections by an ISA Certified Arborist (provided by the Contract Arborist) as this time are critical. Depth of moisture in soils shall be determined by soil sample tube or other exploratory means.
- 6. Minimum watering shall be considered to be 6 applications per growing season typically July through October with the exact timing and duration to be determined by the ISA Arborist.
- K. Overhead Clearance
 - 1. Trees to remain shall be assessed prior to construction for overhead clearance for construction activities. Contract Arborist shall recommend either canopy pruning, temporary guying/tying of select limbs, or alternative construction methods.
 - 2. Pruning for clearance shall not remove branches above 12' feet or over 6" inches diameter
 - 3. All pruning proposed by the Contractor and / or Contract Arborist shall first be reviewed and approved by the Owner and Project Arborist.
 - 4. Equipment exhaust should be directed away from trees as much as possible. Stationary equipment shall not exhaust directly under or toward trees.
 - 5. Contractor shall use appropriate equipment near trees to ensure that trees are not damaged by construction. Contractor shall provide any specialized equipment needed at no additional cost to the owner.
 - 6. Any pruning shall also conform to the pruning specifications in this section.
- L. Soil Tests and Soil Care/Fertilization
 - Initial soil testing within tree protection areas is required. Conduct individual soil tests for separate tree protection areas (small adjacent areas may be tested together). Soil test shall be a representative sample from each area. Soil testing shall include a texture analysis (sand, silt, and clay percentages), soluble salts, and sodium tests.
 - Treatments to the tree protection areas for specified trees (see TPAK) shall be based on the results of the soil analysis. Fertilization should be consistent with the recommendations of the ANSI A-300 (Part 2) Tree, Shrub, and Other Woody Plant Maintenance – Standard Practices (Fertilization) 2004, except as described herein.
 - 3. Application rates shall not exceed a rate of 1 pound of actual nitrogen per 1,000 square feet annually. Fertilizer used should include humic acids, soluble seaweed extracts and soil biological inoculants (mycorrhizae, etc.).

4. Applications to confined areas (i.e. street tree planting pits) should be made by soil injection. In areas where adequate application rates cannot be achieved, injection should be made to the point of refusal.

3.3 FIELD QUALITY CONTROL AND MONITORING

- A. Tree Condition Monitoring
 - 1. An ISA Certified Arborist (provided by the Contract Arborist) shall perform monitoring twice per month year round to monitor insects, disease, soil moisture levels, weather, and health changes on all trees designated on Tree Protection Action Key.
 - 2. The monitoring will include a report that details problematic areas that have been addressed, treatments provided to reduce the problem, and anticipated treatments forecast for 30 days. This report will be forwarded to the Project Arborist, Engineer and Owner for documentation.
 - 3. Any treatments recommended by the Contract Arborist not already included in the project scope shall be noted in the reports for review by the Project Arborist, Engineer and Owner. No additional work is to be performed unless approved in writing by the Owner.
- B. Construction Oversight by Contract Arborist
 - 1. Any work within CRZs of retained trees shall be directly supervised by the Contract Arborist.
 - 2. If roots are encountered during excavation, work shall progress as directed by the Contract Arborist. Contract Arborist, in coordination with the construction and design teams, shall determine appropriate means and methods to address the roots. Options may include, but not be limited to, severing the roots, hand or SSAT excavation. Contractor shall not cut roots.
 - 3. Refer to "Hand Excavation within Tree Protection Areas" specification in the section.
 - 4. All work shall be documented thoroughly, including photo documentation. Refer to site documentation submittal requirements.

3.4 CONTRACTOR DAMAGE AND PENALTIES

- A. Remedial Measures
 - 1. Any damage caused to the trees by the work of this contract through negligence by the contractor shall be immediately remedied by the contractor. Contractor shall be responsible for any associated costs.

- 2. Remedial work may include pruning, cabling, or any other measures up to and including removal and replacement, as determined by the Project Arborist and Engineer.
- 3. Remedial work shall be performed by the Contract Arborist, as approved by the Project Arborist and Engineer.
- 4. All required remedial work shall be performed to the satisfaction of the Project Arborist and Engineer, at no additional cost to the owner.
- B. Tree Replacement
 - 1. If damage to any tree is severe, because of negligence by the contractor as determined by the Project Arborist and Engineer, it shall be replaced with a new tree of equal size caliper and species as that of the damaged tree.
 - 2. If a replacement tree of equal size and caliper is not possible as determined by the Project Arborist and Engineer, it shall be replaced on and inch by inch basis with new trees of a minimum caliper size of 2"-3".
 - 3. Replacement trees shall be supplied and installed at no additional costs to the owner, including all incidental costs including the costs of inspection of the tree at the nursery and any other incidental costs associated with tree replacement.

END OF SECTION 01 5640

SECTION 01 6000 - PRODUCT REQUIREMENTS

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.

1.2 SECTION INCLUDES

- A. Work of this Section includes all labor, materials, equipment, and services necessary to complete product requirements as specified herein, including, but not limited to, the following:
 - 1. Product delivery, storage and handling.
 - 2. Storage and protection.
 - 3. Identifying markings.
 - 4. Temporary use of equipment.
 - 5. General standards.

1.3 RELATED SECTIONS

- A. Execution Requirements Section 01 7000.
- 1.4 TRANSPORTATION AND HANDLING
 - A. Materials, products, and equipment shall be properly containerized, packaged, boxed, and protected to prevent damage during transportation and handling.
 - B. More detailed requirements for transportation and handling are specified under the technical Sections.
- 1.5 PRODUCT DELIVERY, STORAGE, AND HANDLING
 - A. Deliver, store, and handle products using means and methods that will prevent damage, deterioration, and loss, including theft and vandalism. Comply with manufacturer's written instructions.
 - B. Delivery and Handling
 - 1. Schedule delivery to minimize long-term storage at Project site and to prevent overcrowding of construction spaces.
 - 2. Coordinate delivery with installation time to ensure minimum holding time for items that are flammable, hazardous, easily damaged, or sensitive to deterioration, theft, and other losses.

- 3. Deliver products to Project site in an undamaged condition in manufacturer's original sealed container or other packaging system, complete with labels and instructions for handling, storing, unpacking, protecting, and installing.
- 4. Inspect products on delivery to determine compliance with the Contract Documents and to determine that products are undamaged and properly protected.
- C. Storage
 - 1. Store products to allow for inspection and measurement of quantity or counting of units.
 - 2. Store materials in a manner that will not endanger Project structure.
 - 3. Store products that are subject to damage by the elements, under cover in a weathertight enclosure above ground, with ventilation adequate to prevent condensation.
 - 4. Store foam plastic from exposure to sunlight, except to extent necessary for period of installation and concealment.
 - 5. Comply with product manufacturer's written instructions for temperature, humidity, ventilation, and weather-protection requirements for storage.
 - 6. Protect stored products from damage and liquids from freezing.
 - 7. Provide a secure location and enclosure at Project site for storage of materials and equipment by Owner's construction forces. Coordinate location with Owner.

1.6 IDENTIFYING MARKINGS

- A. Name plates and other identifying markings shall not be affixed on exposed surfaces of manufactured items installed in finished spaces.
- 1.7 TEMPORARY USE OF EQUIPMENT
 - A. No equipment intended for permanent installation shall be operated for temporary purposes without the written permission of the Architect.
 - B. The temporary or trial usage by the Owner of any mechanical device, machinery, apparatus, equipment or any work or materials supplied under this Contract before final completion and written acceptance by the Architect, shall not be construed as evidence of the acceptance of same by the Owner. The Owner shall have the privilege of such temporary and trial usage, for such reasonable length of time as and when the Architect shall deem to be proper for making a complete and thorough test of same and no claim for damage shall be made by the Contractor for the injury to or breaking of parts of such work which may be caused by weakness or inaccuracy of structural parts or by defective

material or workmanship. If the Contractor so elects, he may at his own expense, place a competent person or persons to make such trial usage; such trial usage shall be under the supervision of the Contractor.

1.8 GENERAL REQUIREMENTS

- A. In the event that it is necessary for the Contractor to store any materials offsite, he shall first obtain the approval of the Architect. The Contractor shall be responsible for insurance and warehousing charges of any materials stored offsite. The Contractor shall also be responsible for the cost of delivery to the job site of any materials that have been stored offsite.
- B. Materials delivered to the job site shall be carefully stored and protected from damage. Damaged material shall not be used in the work. The Contractor shall provide, where directed temporary storage facilities as may be required for the storage of all materials which might be damaged by weather.
- C. Manufactured articles, materials and equipment shall be applied, installed, connected, erected, used, cleaned and conditioned as directed by the representative manufacturers, unless otherwise specified.
- D. Equipment, plant, and appliances, such as hoists, centering, concrete lifts, construction elevators, cranes, rigging, towers, derricks, walks, ramps, chutes, scaffolding, implements, transportation, cartage and other things necessary and required for the adequate execution of the work and as required by law and applicable Union rules shall be provided and shall be maintained in good and safe mechanical working order, be responsible for their safe use, and remove them when no longer required. Applicable requirements of OSHA shall become and form a part of this document.
- E. During handling and installation of work at project site clean and protect work in progress and adjoining work on a basis of perpetual maintenance. Apply suitable protective covering on newly installed work where reasonably required to ensure freedom from damage or deterioration at time of substantial completion; otherwise, clean and perform maintenance on newly installed work as frequently as necessary through remainder of construction period. Adjust and lubricate operable components to ensure operability without damaging effects.
- F. To extent possible through reasonable control and protection methods, supervise performance of work in a manner and by means which will ensure that none of the work whether completed or in progress, will be subjected to harmful, dangerous, damaging, or otherwise deleterious exposures during construction period. Such exposures include (where applicable, but not by way of limitation) static loading, dynamic loading, internal pressures, external pressures, high or low temperatures, thermal shock, high or low humidity, air contamination or pollution, water, ice, solvents, chemicals, light, radiation, puncture, abrasion, heavy traffic, soiling, bacteria, insect infestation,

combustion, electrical current, high speed operation, improper lubrication, unusual wear, misuse, incompatible interface, destructive testing, misalignment, excessive weathering, unprotected storage, improper shipping/handling, theft and vandalism.

- G. Require installer of each major unit of work to inspect substrate to receive the work, and conditions under which the work will be performed, and to report (in writing to Contractor) unsatisfactory conditions. Do not proceed with the work until unsatisfactory conditions have been corrected in a manner acceptable to Installer.
- H. Where installations include manufactured products, comply with manufacturer's applicable instructions and recommendations for installation to whatever extent these are more explicit or more stringent than applicable requirements indicated in the Contract Documents.
- I. Inspect each item of materials or equipment immediately prior to installation and reject damaged and defective items.
- J. Provide attachment and connection devices and methods for securing work properly as it is installed; true to line and level, and within recognized industry tolerance if not otherwise indicated. Allow for expansions and building movements. Provide uniform joint widths in exposed work, organized for best possible visual effect. Refer questionable visual-effect choices to Architect for final decision.
- K. Recheck measurements and dimensions of the work as an integral step of starting each installation.
- L. Install work during conditions of temperature, humidity, exposure, forecasted weather, and status of project completion which will ensure best possible results for each unit of work in coordination with entire work. Isolate each unit of work from non-compatible work, as required to prevent deterioration.
- M. Coordinate enclosure (closing-in) of work with required inspections and tests, so as to avoid necessity of uncovering work for that purpose.
- N. Mounting Heights: Except as otherwise indicated, mount individual units of work at industry-recognized standard mounting heights, for applications indicated. In CMU walls mount units at height closest to manufacturer's recommendation so as to minimize cutting of block coursings. Refer questionable mounting height choices to Architect for final decision.

END OF SECTION 01 6000

SECTION 01 7000 - EXECUTION REQUIREMENTS

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.

1.2 SECTION INCLUDES

- A. This Section includes general procedural requirements governing execution of the Work including, but not limited to, the following:
 - 1. Construction layout.
 - 2. Field engineering and surveying.
 - 3. General installation of products.
 - 4. Progress cleaning.
 - 5. Starting and adjusting.
 - 6. Protection of installed construction.
 - 7. Correction of the Work.

1.3 RELATED SECTIONS

- A. Closeout Procedures Section 01 7700.
- 1.4 SUBMITTALS
 - A. Qualification Data: For land surveyor to demonstrate capabilities and experience. Include lists of completed projects with project names and addresses, names and addresses of architects and owners, and other information specified.
 - B. Certificates: Submit certificate signed by land surveyor certifying that location and elevation of improvements comply with requirements.
 - C. Landfill Receipts: Submit copy of receipts issued by a landfill facility, licensed to accept hazardous materials, for hazardous waste disposal.
 - D. Certified Surveys: Submit two copies signed by land surveyor.
 - E. Final Property Survey: Submit 10 copies showing the Work performed and record survey data.
1.5 QUALITY ASSURANCE

A. Land Surveyor Qualifications: A professional land surveyor who is legally qualified to practice in jurisdiction where Project is located and who is experienced in providing land-surveying services of the kind indicated.

PART 2 PRODUCTS (Not Applicable)

PART 3 EXECUTION

3.1 EXAMINATION

- A. Existing Conditions: The existence and location of site improvements, utilities, and other construction indicated as existing are not guaranteed. Before beginning work, investigate and verify the existence and location of mechanical and electrical systems and other construction affecting the Work.
 - 1. Before construction, verify the location and points of connection of utility services.
- B. Existing Utilities: Before beginning site work, investigate and verify the existence and location of underground utilities and other construction affecting the Work.
 - 1. Before construction, verify the location and invert elevation at points of connection of sanitary sewer, storm sewer, and water-service piping; and underground electrical services.
 - 2. Furnish location data for work related to Project that must be performed by public utilities serving Project site.
- C. Acceptance of Conditions: Examine substrates, areas, and conditions, with Installer or Applicator present where indicated, for compliance with requirements for installation tolerances and other conditions affecting performance. Record observations.
 - 1. Written Report: Where a written report listing conditions detrimental to performance of the Work is required by other Sections, include the following:
 - a. Description of the Work.
 - b. List of detrimental conditions, including substrates.
 - c. List of unacceptable installation tolerances.
 - d. Recommended corrections.

- 2. Verify compatibility with and suitability of substrates, including compatibility with existing finishes or primers.
- 3. Examine roughing-in for mechanical and electrical systems to verify actual locations of connections before equipment and fixture installation.
- 4. Examine walls, floors, and roofs for suitable conditions where products and systems are to be installed.
- 5. Proceed with installation only after unsatisfactory conditions have been corrected. Proceeding with the Work indicates acceptance of surfaces and conditions.

3.2 PREPARATION

- A. Existing Utility Information: Furnish information to local utility that is necessary to adjust, move, or relocate existing utility structures, utility poles, lines, services, or other utility appurtenances located in or affected by construction. Coordinate with authorities having jurisdiction.
- B. Field Measurements: Take field measurements as required to fit the Work properly. Recheck measurements before installing each product. Where portions of the Work are indicated to fit to other construction, verify dimensions of other construction by field measurements before fabrication. Coordinate fabrication schedule with construction progress to avoid delaying the Work.
- C. Space Requirements: Verify space requirements and dimensions of items shown diagrammatically on Drawings.
- D. Review of Contract Documents and Field Conditions: Immediately on discovery of the need for clarification of the Contract Documents, submit a request for information to Architect. Include a detailed description of problem encountered, together with recommendations for changing the Contract Documents.

3.3 CONSTRUCTION LAYOUT

- A. Verification: Before proceeding to lay out the Work, verify layout information shown on Drawings, in relation to the property survey and existing benchmarks. If discrepancies are discovered, notify Architect promptly.
- B. General: Engage a land surveyor to lay out the Work using accepted surveying practices.
 - 1. Establish benchmarks and control points to set lines and levels at each story of construction and elsewhere as needed to locate each element of Project.
 - 2. Establish dimensions within tolerances indicated. Do not scale Drawings to obtain required dimensions.
 - 3. Inform installers of lines and levels to which they must comply.

- 4. Check the location, level and plumb, of every major element as the Work progresses.
- 5. Notify Architect when deviations from required lines and levels exceed allowable tolerances.
- 6. Close site surveys with an error of closure equal to or less than the standard established by authorities having jurisdiction.
- C. Site Improvements: Locate and lay out site improvements, including pavements, grading, fill and topsoil placement, utility slopes, and invert elevations.
- D. Building Lines and Levels: Locate and lay out control lines and levels for structures, building foundations, column grids, and floor levels, including those required for mechanical and electrical work. Transfer survey markings and elevations for use with control lines and levels. Level foundations and piers from two or more locations.
- E. Record Log: Maintain a log of layout control work. Record deviations from required lines and levels. Include beginning and ending dates and times of surveys, weather conditions, name and duty of each survey party member, and types of instruments and tapes used. Make the log available for reference by Architect.

3.4 FIELD ENGINEERING

- A. Reference Points: Locate existing permanent benchmarks, control points, and similar reference points before beginning the Work. Preserve and protect permanent benchmarks and control points during construction operations.
 - 1. Do not change or relocate existing benchmarks or control points without prior written approval of Architect. Report lost or destroyed permanent benchmarks or control points promptly. Report the need to relocate permanent benchmarks or control points to Architect before proceeding.
 - 2. Replace lost or destroyed permanent benchmarks and control points promptly. Base replacements on the original survey control points.
- B. Benchmarks: Establish and maintain a minimum of two permanent benchmarks on Project site, referenced to data established by survey control points. Comply with authorities having jurisdiction for type and size of benchmark.
 - 1. Record benchmark locations, with horizontal and vertical data, on Project Record Documents.
 - 2. Where the actual location or elevation of layout points cannot be marked, provide temporary reference points sufficient to locate the Work.

- 3. Remove temporary reference points when no longer needed. Restore marked construction to its original condition.
- C. Certified Survey: On completion of foundation walls, major site improvements, and other work requiring field-engineering services, prepare a certified survey showing dimensions, locations, angles, and elevations of construction and site work.
- D. Final Property Survey: Prepare a final property survey showing significant features (real property) for Project. Include on the survey a certification, signed by land surveyor, that principal metes, bounds, lines, and levels of Project are accurately positioned as shown on the survey.
 - 1. Show boundary lines, monuments, streets, site improvements and utilities, existing improvements and significant vegetation, adjoining properties, acreage, grade contours, and the distance and bearing from a site corner to a legal point.
 - 2. Recording: At Substantial Completion, have the final property survey recorded by or with authorities having jurisdiction as the official "property survey."

3.5 INSTALLATION

- A. General: Locate the Work and components of the Work accurately, in correct alignment and elevation, as indicated.
 - 1. Make vertical work plumb and make horizontal work level.
 - 2. Where space is limited, install components to maximize space available for maintenance and ease of removal for replacement.
 - 3. Conceal pipes, ducts, and wiring in finished areas, unless otherwise indicated.
 - 4. Maintain minimum headroom clearance of 8 feet in spaces without a suspended ceiling.
- B. Comply with manufacturer's written instructions and recommendations for installing products in applications indicated.
- C. Install products at the time and under conditions that will ensure the best possible results. Maintain conditions required for product performance until Substantial Completion.
- D. Conduct construction operations so no part of the Work is subjected to damaging operations or loading in excess of that expected during normal conditions of occupancy.
- E. Tools and Equipment: Do not use tools or equipment that produce harmful noise levels.

- F. Anchors and Fasteners: Provide anchors and fasteners as required to anchor each component securely in place, accurately located and aligned with other portions of the Work.
 - 1. Mounting Heights: Where mounting heights are not indicated, mount components at heights directed by Architect.
 - 2. Allow for building movement, including thermal expansion and contraction.
- G. Joints: Make joints of uniform width. Where joint locations in exposed work are not indicated, arrange joints for the best visual effect. Fit exposed connections together to form hairline joints.
- H. Hazardous Materials: Use products, cleaners, and installation materials that are not considered hazardous.
- 3.6 PROGRESS CLEANING
 - A. General: Clean Project site and work areas daily, including common areas. Coordinate progress cleaning for joint-use areas where more than one installer has worked. Enforce requirements strictly. Dispose of materials lawfully.
 - 1. Comply with requirements in NFPA 241 for removal of combustible waste materials and debris.
 - 2. Do not hold materials more than 7 days during normal weather or 3 days if the temperature is expected to rise above 80 deg. F.
 - 3. Containerize hazardous and unsanitary waste materials separately from other waste. Mark containers appropriately and dispose of legally, according to regulations.
 - B. Site: Maintain Project site free of waste materials and debris.
 - C. Work Areas: Clean areas where work is in progress to the level of cleanliness necessary for proper execution of the Work.
 - 1. Remove liquid spills promptly.
 - 2. Where dust would impair proper execution of the Work, broom-clean or vacuum the entire work area, as appropriate.
 - D. Installed Work: Keep installed work clean. Clean installed surfaces according to written instructions of manufacturer or fabricator of product installed, using only cleaning materials specifically recommended. If specific cleaning materials are not recommended, use cleaning materials that are not hazardous to health or property and that will not damage exposed surfaces.
 - E. Concealed Spaces: Remove debris from concealed spaces before enclosing the space.

- F. Exposed Surfaces: Clean exposed surfaces and protect as necessary to ensure freedom from damage and deterioration at time of Substantial Completion.
- G. Cutting and Patching: Clean areas and spaces where cutting and patching are performed. Completely remove paint, mortar, oils, putty, and similar materials.
 - 1. Thoroughly clean piping, conduit, and similar features before applying paint or other finishing materials. Restore damaged pipe covering to its original condition.
- H. Waste Disposal: Burying or burning waste materials on-site will not be permitted. Washing waste materials down sewers or into waterways will not be permitted.
- I. During handling and installation, clean and protect construction in progress and adjoining materials already in place. Apply protective covering where required to ensure protection from damage or deterioration at Substantial Completion.
- J. Clean and provide maintenance on completed construction as frequently as necessary through the remainder of the construction period. Adjust and lubricate operable components to ensure operability without damaging effects.
- K. Limiting Exposures: Supervise construction operations to ensure that no part of the construction, completed or in progress, is subject to harmful, dangerous, damaging, or otherwise deleterious exposure during the construction period.
- 3.7 STARTING AND ADJUSTING
 - A. Start equipment and operating components to confirm proper operation. Remove malfunctioning units, replace with new units, and retest.
 - B. Adjust operating components for proper operation without binding. Adjust equipment for proper operation.
 - C. Test each piece of equipment to verify proper operation. Test and adjust controls and safeties. Replace damaged and malfunctioning controls and equipment.
 - D. Manufacturer's Field Service: If a factory-authorized service representative is required to inspect field-assembled components and equipment installation, comply with qualification requirements in Section 014000, "Quality Requirements."

3.8 PROTECTION OF INSTALLED CONSTRUCTION

- A. Provide final protection and maintain conditions that ensure installed Work is without damage or deterioration at time of Substantial Completion.
- B. Comply with manufacturer's written instructions for temperature and relative humidity.

C. Protect existing brick-paved sidewalk at adjoining residential tower development and those abutting Main Street and Park Avenue. Restore all areas of sidewalk damaged during construction to eliminate any evidence of damage.

3.9 CORRECTION OF THE WORK

- A. Repair or remove and replace defective construction. Restore damaged substrates and finishes. Comply with requirements in Section 017329, "Cutting and Patching."
 - 1. Repairing includes replacing defective parts, refinishing damaged surfaces, touching up with matching materials, and properly adjusting operating equipment.
- B. Restore permanent facilities used during construction to their specified condition.
- C. Remove and replace damaged surfaces that are exposed to view if surfaces cannot be repaired without visible evidence of repair.
- D. Repair components that do not operate properly. Remove and replace operating components that cannot be repaired.
- E. Remove and replace chipped, scratched, and broken glass or reflective surfaces.

END OF SECTION 01 7000

SECTION 01 7329 - CUTTING AND PATCHING

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

- A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.
- 1.2 SECTION INCLUDES
 - A. This Section includes procedural requirements for cutting and patching.

1.3 RELATED SECTIONS

- A. Refer to Divisions 3 through 32 Sections for specific requirements and limitations applicable to cutting and patching individual parts of the Work.
 - 1. Requirements in this Section apply to mechanical and electrical installations. Refer to Divisions 22, 23 and 26 Sections for other requirements and limitations applicable to cutting and patching mechanical and electrical installations.

1.4 DEFINITIONS

- A. Cutting: Removal of in-place construction necessary to permit installation or performance of other Work.
- B. Patching: Fitting and repair work required to restore surfaces to original conditions after installation of other Work.

1.5 SUBMITTALS

- A. Cutting and Patching: Submit a method describing procedures at least 10 days before the time cutting and patching will be performed, requesting approval to proceed. Include the following information:
 - 1. Extent: Describe cutting and patching, show how they will be performed, and indicate why they cannot be avoided.
 - 2. Changes to In-Place Construction: Describe anticipated results. Include changes to structural elements and operating components as well as changes in building's appearance and other significant visual elements.
 - 3. Products: List products to be used and firms or entities that will perform the Work.
 - 4. Dates: Indicate when cutting and patching will be performed.

- 5. Utilities: List utilities that cutting and patching procedures will disturb or affect. List utilities that will be relocated and those that will be temporarily out of service. Indicate how long service will be disrupted.
- 6. Structural Elements: Where cutting and patching involve adding reinforcement to structural elements, submit details and engineering calculations showing integration of reinforcement with original structure.
- 7. Architect's Approval: Obtain approval of cutting and patching before cutting and patching. Approval does not waive right to later require removal and replacement of unsatisfactory work.

1.6 QUALITY ASSURANCE

- A. Structural Elements: Do not cut and patch structural elements in a manner that could change their load-carrying capacity or load-deflection ratio.
 - 1. Provide a list of additional elements that are structural elements and that require Architect's or Construction Manager's approval of a cutting and patching proposal.
- B. Operational Elements: Do not cut and patch the following operating elements and related components in a manner that results in reducing their capacity to perform as intended or that results in increased maintenance or decreased operational life or safety.
 - 1. Primary operational systems and equipment.
 - 2. Air or smoke barriers.
 - 3. Fire-protection systems.
 - 4. Control systems.
 - 5. Communication systems.
 - 6. Conveying systems.
 - 7. Electrical wiring systems.
- C. Miscellaneous Elements: Do not cut and patch the following elements or related components in a manner that could change their load-carrying capacity, that results in reducing their capacity to perform as intended, or that results in increased maintenance or decreased operational life or safety.
 - 1. Water, moisture, or vapor barriers.
 - 2. Membranes and flashings.
 - 3. Exterior construction.
 - 4. Equipment supports.
 - 5. Piping, ductwork, vessels, and equipment.

- 6. Noise- and vibration-control elements and systems.
- D. Visual Requirements: Do not cut and patch construction in a manner that results in visual evidence of cutting and patching. Do not cut and patch construction exposed on the exterior or in occupied spaces in a manner that would, in Architect's opinion, reduce the building's aesthetic qualities. Remove and replace construction that has been cut and patched in a visually unsatisfactory manner.
- E. Cutting and Patching Conference: Before proceeding, meet at Project site with parties involved in cutting and patching, including mechanical and electrical trades. Review areas of potential interference and conflict. Coordinate procedures and resolve potential conflicts before proceeding.

1.7 WARRANTY

A. Warranties: Remove, replace, patch, and repair materials and surfaces cut or damaged during cutting and patching operations, by methods and with materials so as not to void warranties.

PART 2 PRODUCTS

- 2.1 MATERIALS
 - A. General: Comply with requirements specified in other Sections of these Specifications.
 - B. In-Place Materials: Use materials identical to existing materials. For exposed surfaces, use materials that visually match existing adjacent surfaces to the fullest extent possible.
 - 1. If identical materials are unavailable or cannot be used, use materials that, when installed, will match the visual and functional performance of in-place materials.

PART 3 EXECUTION

3.1 EXAMINATION

- A. Examine surfaces to be cut and patched and conditions under which cutting and patching are to be performed.
 - 1. Compatibility: Before patching, verify compatibility with and suitability of substrates, including compatibility with existing finishes or primers.
 - 2. Proceed with installation only after unsafe or unsatisfactory conditions have been corrected.

3.2 PERFORMANCE

- A. General: Employ skilled workers to perform cutting and patching. Proceed with cutting and patching at the earliest feasible time, and complete without delay.
 - 1. Cut in-place construction to provide for installation of other components or performance of other construction, and subsequently patch as required to restore surfaces to their original condition.
- B. Cutting: Cut in-place construction by sawing, drilling, breaking, chipping, grinding, and similar operations, including excavation, using methods least likely to damage elements retained or adjoining construction. If possible, review proposed procedures with original Installer; comply with original Installer's written recommendations.
 - 1. In general, use hand or small power tools designed for sawing and grinding, not hammering and chopping. Cut holes and slots as small as possible, neatly to size required, and with minimum disturbance of adjacent surfaces. Temporarily cover openings when not in use.
 - 2. In-Place Finished Surfaces: Cut or drill from the exposed or finished side into concealed surfaces.
 - 3. Concrete and Masonry: Cut using a cutting machine, such as an abrasive saw or a diamond-core drill.
 - 4. Excavating and Backfilling: Comply with requirements in applicable Division 31 Sections where required by cutting and patching operations.
 - 5. Proceed with patching after construction operations requiring cutting are complete.
- C. Patching: Patch construction by filling, repairing, refinishing, closing up, and similar operations following performance of other Work. Patch with durable seams that are as invisible as possible. Provide materials and comply with installation requirements specified in other Sections of these Specifications.
 - 1. Inspection: Where feasible, test and inspect patched areas after completion to demonstrate integrity of installation.
 - 2. Exposed Finishes: Restore exposed finishes of patched areas and extend finish restoration into retained adjoining construction in a manner that will eliminate evidence of patching and refinishing.
 - 3. Floors and Walls: Where walls or partitions that are removed extend one finished area into another, patch and repair floor and wall surfaces in the new space. Provide an even surface of uniform finish, color, texture, and appearance. Remove existing floor and wall coverings and replace with new materials, if necessary, to achieve uniform color and appearance.
 - a. Where patching occurs in a painted surface, apply primer and intermediate paint coats over the patch and apply final paint coat over

entire unbroken surface containing the patch. Provide additional coats until patch blends with adjacent surfaces.

- 4. Ceilings: Patch, repair, or rehang existing ceilings as necessary to provide an even-plane surface of uniform appearance.
- 5. Exterior Building Enclosure: Patch components in a manner that restores enclosure to a weathertight condition.

END OF SECTION 01 7329

SECTION 01 7700 - CLOSEOUT PROCEDURES

PART 1 GENERAL

1.1 GENERAL REQUIREMENTS

A. Work of this Section, as shown or specified, shall be in accordance with the requirements of the Contract Documents.

1.2 SECTION INCLUDES

- A. This Section includes administrative and procedural requirements for contract closeout, including, but not limited to, the following:
 - 1. Inspection procedures.
 - 2. Project Record Documents.
 - 3. Warranties.
 - 4. Instruction of Owner's personnel.
 - 5. Final cleaning.

1.3 RELATED SECTIONS

A. Execution Requirements - Section 01 7300.

1.4 SUBSTANTIAL COMPLETION

- A. Preliminary Procedures: Before requesting inspection for determining date of Substantial Completion, complete the following. List items below that are incomplete in request.
 - 1. Prepare a list of items to be completed and corrected (punch list), the value of items on the list, and reasons why the Work is not complete.
 - 2. Advise Owner of pending insurance changeover requirements.
 - 3. Submit specific warranties, workmanship bonds, maintenance service agreements, final certifications, and similar documents.
 - 4. Obtain and submit releases permitting Owner unrestricted use of the Work and access to services and utilities. Include occupancy permits, operating certificates, and similar releases.
 - 5. Prepare and submit Project Record Documents, operation and maintenance manuals, Final Completion construction photographs and photographic negatives, damage or settlement surveys, property surveys, and similar final record information.

- 6. Deliver tools, spare parts, extra materials, and similar items to location designated by Owner. Label with manufacturer's name and model number where applicable.
- 7. Make final changeover of permanent locks and deliver keys to Owner. Advise Owner's personnel of changeover in security provisions.
- 8. Complete startup testing of systems.
- 9. Submit test/adjust/balance records.
- 10. Terminate and remove temporary facilities from Project site, along with mockups, construction tools, and similar elements.
- 11. Advise Owner of changeover in heat and other utilities.
- 12. Submit changeover information related to Owner's occupancy, use, operation, and maintenance.
- 13. Complete final cleaning requirements, including touchup painting.
- 14. Touch up and otherwise repair and restore marred exposed finishes to eliminate visual defects.
- B. Inspection: Submit a written request for inspection for Substantial Completion. On receipt of request, Architect will either proceed with inspection or notify Contractor of unfulfilled requirements. Architect will prepare the Certificate of Substantial Completion after inspection or will notify Contractor of items, either on Contractor's list or additional items identified by Architect, that must be completed or corrected before certificate will be issued. A sample of the Certificate of Substantial Completion form is attached at the end of this section.
 - 1. Reinspection: Request reinspection when the Work identified in previous inspections as incomplete is completed or corrected.
 - 2. Results of completed inspection will form the basis of requirements for Final Completion.

1.5 FINAL COMPLETION

- A. Preliminary Procedures: Before requesting final inspection for determining date of Final Completion, complete the following:
 - 1. Submit certified copy of Architect's Substantial Completion inspection list of items to be completed or corrected (punch list), endorsed and dated by Architect. The certified copy of the list shall state that each item has been completed or otherwise resolved for acceptance.
 - 2. Submit evidence of final, continuing insurance coverage complying with insurance requirements.
 - 3. Submit pest-control final inspection report and warranty.

- 4. Instruct Owner's personnel in operation, adjustment, and maintenance of products, equipment, and systems. Submit demonstration and training videotapes.
- B. Inspection: Submit a written request for final inspection for acceptance. On receipt of request, Architect will either proceed with inspection or notify Contractor of unfulfilled requirements. Architect will notify Contractor of construction that must be completed or corrected before certificate will be issued.
 - 1. Reinspection: Request reinspection when the Work identified in previous inspections as incomplete is completed or corrected.
- 1.6 LIST OF INCOMPLETE ITEMS (PUNCH LIST)
 - A. Preparation: Submit digital copy of list. Include name and identification of each space and area affected by construction operations for incomplete items and items needing correction including, if necessary, areas disturbed by Contractor that are outside the limits of construction.
 - 1. Organize list of spaces in sequential order, starting with exterior areas first and proceeding from lowest floor to highest floor.
 - 2. Organize items applying to each space by major element, including categories for ceiling, individual walls, floors, equipment, and building systems.
 - 3. Include the following information at the top of each page:
 - a. Project name.
 - b. Date.
 - c. Name of Architect.
 - d. Name of Contractor.
 - e. Page number.

1.7 PROJECT RECORD DOCUMENTS

- A. General: Do not use Project Record Documents for construction purposes. Protect Project Record Documents from deterioration and loss. Provide access to Project Record Documents for Architect's reference during normal working hours.
- B. Record Drawings: Maintain and submit one set of blue- or black-line white prints of Contract Drawings and Shop Drawings.
 - 1. Mark Record Prints to show the actual installation where installation varies from that shown originally. Require individual or entity who obtained record data, whether individual or entity is Installer, subcontractor, or similar entity, to prepare the marked-up Record Prints.

- a. Give particular attention to information on concealed elements that cannot be readily identified and recorded later.
- b. Accurately record information in an understandable drawing technique.
- c. Record data as soon as possible after obtaining it. Record and check the markup before enclosing concealed installations.
- d. Mark Contract Drawings or Shop Drawings, whichever is most capable of showing actual physical conditions, completely and accurately. Where Shop Drawings are marked, show cross-reference on Contract Drawings.
- 2. Mark record sets with erasable, red-colored pencil. Use other colors to distinguish between changes for different categories of the Work at the same location.
- 3. Mark important additional information that was either shown schematically or omitted from original Drawings.
- 4. Note Construction Change Directive numbers, Change Order numbers, alternate numbers, and similar identification where applicable.
- 5. Identify and date each Record Drawing; include the designation "PROJECT RECORD DRAWING" in a prominent location. Organize into manageable sets; bind each set with durable paper cover sheets. Include identification on cover sheets.
- C. Record Specifications: Submit one copy of Project's Specifications, including addenda and contract modifications. Mark copy to indicate the actual product installation where installation varies from that indicated in Specifications, addenda, and contract modifications.
 - 1. Give particular attention to information on concealed products and installations that cannot be readily identified and recorded later.
 - 2. Mark copy with the proprietary name and model number of products, materials, and equipment furnished, including substitutions and product options selected.
 - 3. Note related Change Orders, Record Drawings and Product Data, where applicable.
- D. Record Product Data: Submit one copy of each Product Data submittal. Mark one set to indicate the actual product installation where installation varies substantially from that indicated in Product Data.
 - 1. Give particular attention to information on concealed products and installations that cannot be readily identified and recorded later.

- 2. Include significant changes in the product delivered to Project site and changes in manufacturer's written instructions for installation.
- 3. Note related Change Orders, Record Drawings and Record Specifications, where applicable.
- E. Miscellaneous Record Submittals: Assemble miscellaneous records required by other Specification Sections for miscellaneous record keeping and submittal in connection with actual performance of the Work. Bind or file miscellaneous records and identify each, ready for continued use and reference.

1.8 WARRANTIES

- A. Submittal Time: Submit written warranties on request of Architect for designated portions of the Work where commencement of warranties other than date of Substantial Completion is indicated.
- B. Partial Occupancy: Submit properly executed warranties within 15 days of completion of designated portions of the Work that are completed and occupied or used by Owner during construction period by separate agreement with Contractor.
- C. Organize warranty documents into an orderly sequence based on the table of contents of the Project Manual.
 - 1. Bind warranties and bonds in heavy-duty, 3-ring, vinyl-covered, loose-leaf binders, thickness as necessary to accommodate contents, and sized to receive 8-1/2" x 11" paper.
 - 2. Provide heavy paper dividers with plastic-covered tabs for each separate warranty. Mark tab to identify the product or installation. Provide a typed description of the product or installation, including the name of the product and the name, address, and telephone number of Installer.
 - 3. Identify each binder on the front and spine with the typed or printed title "WARRANTIES," Project name, and name of Contractor.

PART 2 PRODUCTS

2.1 MATERIALS

A. Cleaning Agents: Use cleaning materials and agents recommended by manufacturer or fabricator of the surface to be cleaned. Do not use cleaning agents that are potentially hazardous to health or property or that might damage finished surfaces.

PART 3 EXECUTION

3.1 DEMONSTRATION AND TRAINING

- A. Instruction: Instruct Owner's personnel to adjust, operate, and maintain systems, subsystems, and equipment not part of a system.
 - 1. Provide instructors experienced in operation and maintenance procedures.
 - 2. Provide instruction at mutually agreed-on times. For equipment that requires seasonal operation, provide similar instruction at the start of each season.
 - 3. Schedule training with Owner with at least seven days' advance notice.
 - 4. Coordinate instructors, including providing notification of dates, times, length of instruction, and course content.
- B. Program Structure: Develop an instruction program that includes individual training modules for each system and equipment not part of a system, as required by individual Specification Sections. For each training module, develop a learning objective and teaching outline. Include instruction for the following:
 - 1. System design and operational philosophy.
 - 2. Review of documentation.
 - 3. Operations.
 - 4. Adjustments.
 - 5. Troubleshooting.
 - 6. Maintenance.
 - 7. Repair.

3.2 FINAL CLEANING

- A. General: Provide final cleaning. Conduct cleaning and waste-removal operations to comply with local laws and ordinances and Federal and local environmental and antipollution regulations.
- B. Cleaning: Employ experienced workers or professional cleaners for final cleaning. Clean each surface or unit to condition expected in an average commercial building cleaning and maintenance program. Comply with manufacturer's written instructions.
 - 1. Complete the following cleaning operations before requesting inspection for certification of Substantial Completion for entire Project or for a portion of Project:

- a. Clean Project site, yard, and grounds, in areas disturbed by construction activities, including landscape development areas, of rubbish, waste material, litter, and other foreign substances.
- b. Sweep paved areas broom clean. Remove petrochemical spills, stains, and other foreign deposits.
- c. Rake grounds that are neither planted nor paved to a smooth, even textured surface.
- d. Remove tools, construction equipment, machinery, and surplus material from Project site.
- e. Remove snow and ice to provide safe access to building.
- f. Clean exposed exterior and interior hard-surfaced finishes to a dirtfree condition, free of stains, films, and similar foreign substances. Avoid disturbing natural weathering of exterior surfaces. Restore reflective surfaces to their original condition.
- g. Remove debris and surface dust from limited access spaces, including roofs, plenums, shafts, trenches, equipment vaults, manholes, attics, and similar spaces.
- h. Sweep concrete floors broom clean in unoccupied spaces.
- i. Vacuum carpet and similar soft surfaces, removing debris and excess nap; shampoo if visible soil or stains remain.
- j. Clean transparent materials, including mirrors and glass in doors and windows. Remove glazing compounds and other noticeable, vision-obscuring materials. Replace chipped or broken glass and other damaged transparent materials. Polish mirrors and glass, taking care not to scratch surfaces.
- k. Remove labels that are not permanent.
- I. Touch up and otherwise repair and restore marred, exposed finishes and surfaces. Replace finishes and surfaces that cannot be satisfactorily repaired or restored or that already show evidence of repair or restoration.
 - 1). Do not paint over "UL" and similar labels, including mechanical and electrical nameplates.
- m. Wipe surfaces of mechanical and electrical equipment, elevator equipment, and similar equipment. Remove excess lubrication, paint and mortar droppings, and other foreign substances.
- n. Replace parts subject to unusual operating conditions.

- o. Clean plumbing fixtures to a sanitary condition, free of stains, including stains resulting from water exposure.
- p. Replace disposable air filters and clean permanent air filters. Clean exposed surfaces of diffusers, registers, and grills.
- q. Clean ducts, blowers, and coils if units were operated without filters during construction.
- r. Clean light fixtures, lamps, globes, and reflectors to function with full efficiency. Replace burned-out bulbs, and those noticeably dimmed by hours of use, and defective and noisy starters in fluorescent and mercury vapor fixtures to comply with requirements for new fixtures.
- s. Leave Project clean and ready for occupancy.
- C. Comply with safety standards for cleaning. Do not burn waste materials. Do not bury debris or excess materials on Owner's property. Do not discharge volatile, harmful, or dangerous materials into drainage systems. Remove waste materials from Project site and dispose of lawfully.

END OF SECTION 01 7700

SECTION 02 4119 – SELECTIVE DEMOLITION

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section includes selective removal and subsequent offsite disposal of portions of existing building indicated on drawings and as required to accommodate new construction.

1.2 DEFINITIONS

- A. Remove: Detach items from existing construction and legally dispose of them off-site, unless indicated to be removed and salvaged or removed and reinstalled.
- B. Remove and Salvage: Detach items from existing construction and deliver them to Owner's designated storage area.
- C. Remove and Reinstall: Detach items from existing construction, prepare them for reuse, and reinstall them where indicated.
- D. Existing to Remain: Existing items of construction that are not to be removed and that are not otherwise indicated to be removed, removed and salvaged, or removed and reinstalled.

1.3 MATERIALS OWNERSHIP

A. Except for items or materials indicated to be reused, salvaged, reinstalled, or otherwise indicated to remain Owner's property, demolished materials shall become Contractor's property and shall be removed from Project site.

1.4 SUBMITTALS

- A. Qualification Data: For firms and persons specified in "Quality Assurance" Article to demonstrate their capabilities and experience. Include lists of completed projects with project names and addresses, names and addresses of architects and owners, and other information specified.
- B. Proposed Dust-Control and Noise-Control Measures: Submit statement or drawing that indicates the measures proposed for use, proposed locations, and proposed time frame for their operation. Identify options if proposed measures are later determined to be inadequate.
- C. Schedule of Selective Demolition Activities: Indicate the following:
 - 1. Detailed sequence of selective demolition and removal work, with starting and ending dates for each activity. Ensure Owner's on-site operations are uninterrupted.

- 2. Interruption of utility services.
- 3. Coordination for shutoff, capping, and continuation of utility services.
- 4. Use of elevator and stairs.
- 5. Locations of temporary partitions and means of egress.
- 6. Coordination of Owner's continuing occupancy of portions of existing building and of Owner's partial occupancy of completed Work.
- D. Inventory: After selective demolition is complete, submit a list of items that have been removed and salvaged.
- E. Predemolition Photographs or Videotape: Show existing conditions of adjoining construction and site improvements, including finish surfaces that might be misconstrued as damage caused by selective demolition operations. Submit before Work begins.
- F. Landfill Records: Indicate receipt and acceptance of hazardous wastes by a landfill facility licensed to accept hazardous wastes.

1.5 QUALITY ASSURANCE

- A. Demolition Firm Qualifications: An experienced firm that has specialized in demolition work similar in material and extent to that indicated for this Project.
- B. Professional Engineer Qualifications: Comply with Division 01 Section "Quality Requirements."
- C. Regulatory Requirements: Comply with governing EPA notification regulations before beginning selective demolition. Comply with hauling and disposal regulations of authorities having jurisdiction. Comply with applicable regulations, codes and ordinances.
- D. Standards: Comply with ANSI A10.6 and NFPA 241.
- E. Predemolition Conference: Conduct conference at Project site to comply with requirements in Division 01 Section "Project Management and Coordination." Review methods and procedures related to selective demolition including, but not limited to, the following:
 - 1. Inspect and discuss condition of construction to be selectively demolished.
 - 2. Review structural load limitations of existing structure.
 - 3. Review and finalize selective demolition schedule.
 - 4. Review requirements of work performed by other trades that rely on substrates exposed by selective demolition operations.

1.6 PROJECT CONDITIONS

- A. Occupied Buildings:
 - 1. Owner will occupy portions of building immediately adjacent to selective demolition area. Conduct selective demolition so Owner's operations will not be

disrupted. Provide not less than 72 hours' notice to Owner of activities that will affect Owner's operations.

- 2. Maintain access to existing walkways, corridors, and other adjacent occupied or used facilities.
- B. Owner assumes no responsibility for condition of areas to be selectively demolished. Conditions existing at time of inspection for bidding will be maintained by Owner as far as practical.
- C. Hazardous Materials: If materials suspected of containing hazardous materials are encountered, do not disturb; immediately notify Architect and Owner. Hazardous materials will be removed by Owner under a separate contract.
- D. Storage or sale of removed items or materials on-site will not be permitted.
- E. Utility Service: Maintain existing utilities indicated to remain in service and protect them against damage during selective demolition operations.
 - 1. Maintain fire-protection facilities in service during selective demolition operations.
- 1.7 WARRANTY
 - A. Existing Warranties: Remove, replace, patch, and repair materials and surfaces cut or damaged during selective demolition, by methods and with materials so as not to void existing warranties.
- PART 2 PRODUCTS
- 2.1 REPAIR MATERIALS
 - A. Use repair materials identical to existing materials.
 - 1. If identical materials are unavailable or cannot be used for exposed surfaces, use materials that visually match existing adjacent surfaces to the fullest extent possible.
 - 2. Use materials whose installed performance equals or surpasses that of existing materials.
 - B. Comply with material and installation requirements specified in individual Specification Sections.
- PART 3 EXECUTION
- 3.1 EXAMINATION
 - A. Verify that utilities have been disconnected and capped.

- B. Survey existing conditions and correlate with requirements indicated to determine extent of selective demolition required.
- C. Inventory and record the condition of items to be removed and reinstalled and items to be removed and salvaged.
- D. When unanticipated mechanical, electrical, or structural elements that conflict with intended function or design are encountered, investigate and measure the nature and extent of conflict. Promptly submit a written report to Architect.
- E. Engage a professional engineer to survey condition of building to determine whether removing any element might result in structural deficiency or unplanned collapse of any portion of structure or adjacent structures during selective demolition operations.
- F. Perform surveys as the Work progresses to detect hazards resulting from selective demolition activities.

3.2 UTILITY SERVICES

- A. Existing Utilities: Maintain services indicated to remain and protect them against damage during selective demolition operations.
- B. Occupied Buildings: Do not interrupt existing utilities serving occupied or operating facilities unless authorized in writing by Owner and authorities having jurisdiction. Provide temporary services during interruptions to existing utilities, as acceptable to Owner and to authorities having jurisdiction.
 - 1. Provide at least 72 hours' notice to Owner if shutdown of service is required during changeover.
- C. Utility Requirements: Locate, identify, disconnect, and seal or cap off indicated utilities serving areas to be selectively demolished.
 - 1. Arrange with Owner to shut off indicated utilities.
 - 2. If utility services are required to be removed, relocated, or abandoned, before proceeding with selective demolition provide temporary utilities that bypass area of selective demolition and that maintain continuity of service to other parts of building.
 - 3. Cut off pipe or conduit in walls or partitions to be removed. Cap, valve, or plug and seal remaining portion of pipe or conduit after bypassing.

3.3 PREPARATION

- A. Site Access and Temporary Controls: Conduct selective demolition and debris-removal operations to ensure minimum interference with roads, streets, walks, walkways, and other adjacent occupied and used facilities.
 - 1. Do not close or obstruct streets, walks, walkways, or other adjacent occupied or used facilities without permission from Owner and authorities having

jurisdiction. Provide alternate routes around closed or obstructed traffic ways if required by governing regulations.

- 2. Erect temporary protection, such as walks, fences, railings, canopies, and covered passageways, where required by authorities having jurisdiction.
- 3. Protect existing site improvements, appurtenances, and landscaping to remain.
- B. Temporary Facilities: Provide temporary barricades and other protection required to prevent injury to people and damage to adjacent buildings and facilities to remain.
 - 1. Provide protection to ensure safe passage of people around selective demolition area and to and from occupied portions of building.
 - 2. Provide temporary weather protection, during interval between selective demolition of existing construction on exterior surfaces and new construction, to prevent water leakage and damage to structure and interior areas.
 - 3. Protect walls, ceilings, floors, and other existing finish work that are to remain or that are exposed during selective demolition operations.
 - 4. Cover and protect furnishings, and equipment that have not been removed.
- C. Temporary Enclosures: Provide temporary enclosures for protection of existing building and construction, in progress and completed, from exposure, foul weather, other construction operations, and similar activities. Provide temporary weathertight enclosure for building exterior.
 - 1. Where heating or cooling is needed and permanent enclosure is not complete, provide insulated temporary enclosures. Coordinate enclosure with ventilating and material drying or curing requirements to avoid dangerous conditions and effects.
- D. Temporary Partitions: Erect and maintain dustproof partitions and temporary enclosures to limit dust and dirt migration and to separate areas from fumes and noise.
- E. Temporary Shoring: Provide and maintain shoring, bracing, or structural support to preserve stability and prevent movement, settlement, or collapse of construction to remain, and to prevent unexpected or uncontrolled movement or collapse of construction being demolished. Strengthen or add new supports when required during progress of selective demolition.

3.4 POLLUTION CONTROLS

- A. Dust Control: Use water mist, temporary enclosures, and other suitable methods to limit spread of dust and dirt. Comply with governing environmental-protection regulations.
 - 1. Do not use water when it may damage existing construction or create hazardous or objectionable conditions, such as ice, flooding, and pollution.
 - 2. Wet mop floors to eliminate trackable dirt and wipe down walls and doors of demolition enclosure. Vacuum carpeted areas.
- B. Disposal: Remove and transport debris in a manner that will prevent spillage on

adjacent surfaces and areas.

- 1. Remove debris from elevated portions of building by chute, hoist, or other device that will convey debris to grade level in a controlled descent.
- C. Cleaning: Clean adjacent structures and improvements of dust, dirt, and debris caused by selective demolition operations. Return adjacent areas to condition existing before selective demolition operations began.

3.5 SELECTIVE DEMOLITION

- A. General: Demolish and remove existing construction only to the extent required by new construction and as indicated. Use methods required to complete the Work within limitations of governing regulations and as follows:
 - 1. Proceed with selective demolition systematically, from higher to lower level. Complete selective demolition operations above each floor or tier before disturbing supporting members on the next lower level.
 - 2. Neatly cut openings and holes plumb, square, and true to dimensions required. Use cutting methods least likely to damage construction to remain or adjoining construction.
 - 3. Use hand tools or small power tools designed for sawing or grinding, not hammering and chopping, to minimize disturbance of adjacent surfaces. Temporarily cover openings to remain.
 - 4. Cut or drill from the exposed or finished side into concealed surfaces to avoid marring existing finished surfaces.
 - 5. Do not use cutting torches until work area is cleared of flammable materials. At concealed spaces, verify condition and contents before starting flame-cutting operations.
 - 6. Maintain portable fire-suppression devices during flame-cutting operations.
 - 7. Maintain adequate ventilation when using cutting torches.
 - 8. Remove decayed, vermin-infested, or otherwise dangerous or unsuitable materials and promptly dispose of off-site.
 - 9. Remove structural framing members and lower to ground by method suitable to avoid free fall and to prevent ground impact or dust generation.
 - 10. Locate selective demolition equipment and remove debris and materials so as not to impose excessive loads on supporting walls, floors, or framing.
 - 11. Dispose of demolished items and materials promptly.
 - 12. Return elements of construction and surfaces that are to remain to condition existing before selective demolition operations began.
- B. Existing Facilities: Comply with building manager's requirements for using and protecting elevators, stairs, walkways, loading docks, building entries, and other building facilities during selective demolition operations.
- C. Removed and Salvaged Items: Comply with the following:
 - 1. Clean salvaged items.
 - 2. Pack or crate items after cleaning. Identify contents of containers.

- 3. Store items in a secure area until delivery to Owner.
- 4. Transport items to Owner's storage area designated by Owner.
- 5. Protect items from damage during transport and storage.
- D. Removed and Reinstalled Items: Comply with the following:
 - 1. Clean and repair items to functional condition adequate for intended reuse. Paint equipment to match new equipment.
 - 2. Pack or crate items after cleaning and repairing. Identify contents of containers.
 - 3. Protect items from damage during transport and storage.
 - 4. Reinstall items in locations indicated. Comply with installation requirements for new materials and equipment. Provide connections, supports, and miscellaneous materials necessary to make item functional for use indicated.
- E. Existing Items to Remain: Protect construction indicated to remain against damage and soiling during selective demolition. When permitted by Architect, items may be removed to a suitable, protected storage location during selective demolition and reinstalled in their original locations after selective demolition operations are complete.
- 3.6 PATCHING AND REPAIRS
 - A. General: Promptly repair damage to adjacent construction caused by selective demolition operations.
 - B. Repairs: Where repairs to existing surfaces are required, patch to produce surfaces suitable for new materials.
 - 1. Completely fill holes and depressions in existing concrete or masonry that are to remain with an approved masonry patching material applied according to manufacturer's written recommendations.
- 3.7 DISPOSAL OF DEMOLISHED MATERIALS
 - A. General: Promptly dispose of demolished materials. Do not allow demolished materials to accumulate on-site.
 - B. Burning: Do not burn demolished materials.
 - C. Disposal: Transport demolished materials off Owner's property and legally dispose of
- 3.8 SELECTIVE DEMOLITION SCOPE
 - A. Refer to the drawings.

END OF SECTION 024119

SECTION 02 8216 - ASBESTOS ABATEMENT

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SCOPE

- A. The work specified herein shall be the abatement of asbestos-containing materials by persons who are knowledgeable, qualified, and trained in the removal, treatment, handling, and disposal of asbestos-containing material, and the subsequent cleaning of the affected environment. The Contractor shall have a Competent Person in control on the job site at all times during asbestos abatement work. This person must comply with applicable Federal, State and Local regulations which mandate work practices, and be capable of performing the work of this contract.
- B. The Contractor shall be licensed by the State of Connecticut in accordance with State of Connecticut Regulations, Sections 20-440-1 through 9 & 20-441. The asbestos supervisor and workers shall be licensed by the State of Connecticut in accordance with State of Connecticut Regulations, Sections 20-437 and 20-438. Should any portion of the work be subcontracted, the subcontractor must also be licensed in accordance with these regulations. The licensing requirements are available from the Environmental Health Services Division, Department of Public Health, 410 Capitol Avenue, MS#51AIR, P.O. Box 340308, Hartford, CT 06134.
- C. Girls Inc. (Owner) will retain the services of a Project Monitor for protection of its interests and those using the building. Pre-abatement, during abatement and post-abatement sampling will be conducted as deemed necessary.
- D. Deviations from this Specification require the written approval of Girls Inc.
- E. Specification Section 02 8216 was produced by EnviroMed Services, Inc., John Luby CT Licensed Asbestos Project Designer #19.
- F. The Contractor is responsible for restoring all work areas and auxiliary areas utilized during abatement to conditions equal to or better than original. Any damage caused during the performance of abatement activities shall be repaired by the Contractor (e.g., paint peeled off by barrier tape, nail holes, water damage, removal of ceiling tiles or concrete blocks, broken glass, etc.) at no additional expense to the Owner. The Contractor is responsible for protecting all objects in work areas that are permanent fixtures or too large to remove.
- G. The Contractor shall be responsible for the following general requirements:
 - 1. Obtain all approvals and permits, and submit all notifications required.

- 2. Provide, erect, and maintain all planking, bracing, shoring, barricades, and warning signs.
- 3. Unless otherwise specified, all equipment, fixtures, piping and debris resulting from demolition shall become the property of the Contractor and shall be removed from the premises.
- 4. Materials to be reused shall be removed with the utmost care to prevent damage of any kind. All material to be reused shall be stored as directed. The Contractor shall coordinate with the Owner as to the storage location.
- 5. Materials not scheduled for reuse shall be removed from the site and disposed of in accordance with all applicable Federal, State and Local requirements.
- H. It shall be the responsibility of the Contractor to protect and preserve in operating condition, all utilities traversing the building and site. Damage to any utility due to work under this Contract shall be repaired to the satisfaction of the Owner at no cost to the Owner.

1.3 DESCRIPTION OF WORK

- A. The Contractor shall supply all labor, materials, equipment, services, insurance (with specific coverage for work on asbestos), and incidentals which are necessary or required to perform the work in accordance with applicable governmental regulations and these specifications.
- B. Contractor shall remove the following asbestos-containing materials at the Girls Inc. Building as depicted on Drawing H101:

Material	<u>Quantity</u>	Location
Vinyl Floor Tile & Mastic	220 SF	Mech 115 – 1 st Floor
Flex Duct Connectors	2 EA	Mech 115 – 1 st Floor
Mudded Pipe Fitting Insulation	2 LF	Training 120 – 1 st Floor

1.4 DEFINITIONS

Adequately Wet - Sufficiently mix or penetrate with liquid to prevent the release of particulates. If visible emissions are observed coming from asbestos-containing material, then that material has not been adequately wetted. However, the absence of visible emissions is not sufficient evidence of being adequately wet.

AHERA - Asbestos Hazard Emergency Response Act - U. S. EPA regulation 40 CFR Part 763 under Section 203 of Title II of the Toxic Substances Control Act (TSCA), 15 U.S.C. 2643. This rule mandates inspections, accreditations of persons involved with

asbestos, and final air clearances following abatement in public and private schools, and public and commercial buildings.

Alternative Work Practice (AWP) - Deviation from Asbestos Standards (Sections 19a-332a-1 to 19a-332a-16 inclusive). Deviation requires a written approval letter from the State of Connecticut Department of Public Health and the Owner.

Asbestos - The term asbestos includes chrysotile, amosite, crocidolite, asbestiform tremolite, asbestos, anthophyllite asbestos, actinolite asbestos and any of these minerals that has been chemically treated and/or altered.

Asbestos Abatement - The removal, encapsulation, enclosure, renovation, repair, demolition or other disturbance of asbestos-containing materials except activities which are related to the removal or repair of asbestos cement pipe and are performed as defined in Section 25-32a of the Connecticut General Statutes.

Asbestos-Containing Material (ACM) - Any material containing greater than or equal to 1.0 percent asbestos.

Asbestos-Containing Waste Materials - Mill tailings or any waste that contains commercial asbestos and is generated by a source subject to the provisions of this subpart. This term includes filters from control devices, friable asbestos waste material, and bags or other similar packaging contaminated with commercial asbestos. As applied to demolition and renovations operations, this term also includes regulated asbestos-containing material waste and materials contaminated with asbestos including disposable equipment and clothing.

Asbestos Control Area - An area where asbestos abatement operations are performed which is isolated by physical boundaries which assist in the prevention of the uncontrolled release of asbestos dust, fibers, or debris. Two examples of an Asbestos Control Area are a "full containment" and a "glove-bag."

Asbestos Fiber - A particulate form of asbestos, tremolite, anthophyllite, actinolite, or a combination of these minerals having a length of five micrometers or longer, with a length-to-diameter ratio of at least 3 to 1.

Authorized Asbestos Disposal Facility - A location approved by the Connecticut Department of Environmental Protection for handling and disposing of asbestos waste or by an equivalent regulatory agency if the material is disposed of outside the State of Connecticut.

Category I Non-Friable Asbestos-Containing Material (ACM) -Asbestos-containing packings, gaskets, resilient floor coverings and asphalt roofing products containing greater than or equal to 1 percent asbestos as determined using the method specified in Appendix A, subpart F, 40 CFR part 763, section 1, Polarized Light Microscopy.

Category II Non-Friable ACM - Any material, excluding Category I non-friable ACM, containing greater than or equal to 1 percent asbestos as determined using the method specified in Appendix A, subpart F, 40 CFR part 763, section 1, Polarized Light Microscopy that when dry, cannot be crumbled, pulverized, or reduced to powder by hand pressure.

Competent Person - Individual capable of identifying existing asbestos, tremolite, anthophyllite, or actinolite hazards and corrective measures to eliminate them, as specified in 29 CFR 1926.32. The duties of the Competent Person include at least the following: establishing the pressure differential, ensuring its integrity, and controlling entry to and exit from the enclosure; supervising any employee exposure monitoring required by the standard; ensuring that all employees working within such an enclosure wear the appropriate personal protective equipment, are trained in the use of appropriate methods of exposure control, and use the hygiene facilities and decontamination procedures specified; and ensuring that engineering controls in use are in proper operating condition and are functioning properly.

Concealed Space - Space which is out of sight. Examples of a concealed space include area above ceilings; below floors; between double walls; furred-in areas; pipe and duct shafts; and similar spaces.

Critical Barrier - A minimum of two layers of six (6) mil polyethylene sheeting taped securely over windows, doorways, diffusers, grilles and any other openings between the Work Area and uncontaminated areas outside of the Work Area, including the outside of the building.

Decontamination Enclosure System - A series of rooms separated from the Work Area and from each other by air locks, for the decontamination of workers and equipment.

Demolition - The wrecking or taking out of any load-supporting structural member of a facility together with any related handling operations or the intentional burning of any facility.

DEEP - The Connecticut Department of Energy and Environmental Protection, 79 Elm Street, Hartford, CT 06106.

DPH - The Connecticut Department of Public Health, 410 Capitol Avenue, MS#51AIR, P.O. Box 340308, Hartford, CT 06134.

Differential Pressure - A difference in the static air pressure between the Work Area and occupied areas, and is developed by the use of HEPA filtered exhaust fans. This differential is generally in the range of 0.02 to 0.04 inches of water column.

Encapsulant - Specific materials in various forms used to chemically entrap asbestos fibers in various configurations to prevent these fibers from becoming airborne. There are four types of encapsulant as follows:

- a) Removal Encapsulant (can be used as a wetting agent).
- b) Bridging Encapsulant (used to provide a tough durable surface coating to asbestos-containing material).
- c) Penetrating Encapsulant (used to penetrate the asbestos containing material down to substrate, encapsulating all asbestos fibers).
- d) Lock-down Encapsulant (used to seal off "lock-down" minute asbestos fibers left on surfaces from which asbestos containing materials have been removed).

Encapsulation - The application of an encapsulant to asbestos-containing building materials to control the possible release of asbestos fibers into the air.

Engineering Controls - Controls to include, but not be limited to, pressure differential equipment, decontamination enclosures, critical barriers and related procedures.

Equipment Decontamination Enclosure System - The portion of a Decontamination Enclosure System designed for controlled transfer of materials and equipment into or out of the Work Area, typically consisting of a Washroom and a Holding Area.

Exposed - Open to view.

Finished Space - Space used for habitation or occupancy where rough surfaces are plastered, paneled or otherwise treated to provide a pleasing appearance.

Fixed Critical Barrier - Barrier constructed of 2" x 4" metal framing 16" O.C., with 1/2" wallboard on the occupied side and $\frac{1}{2}$ " wallboard and two layers of six (6) mil polyethylene sheeting on the Work Area side to prevent unauthorized access or air flow.

Fixed Object - A piece of equipment or furniture in the Work Area which cannot be removed from the Work Area, as determined by the Owner.

Friable Asbestos Material - Material containing greater than or equal to 1 percent asbestos as determined using the method specified in Appendix A, subpart F, 40 CFR part 763, Section 1, Polarized Light Microscopy, that when dry can be crumbled, pulverized or reduced to powder by hand pressure. If the asbestos content is less then 10 percent as determined by a method other than point counting by polarized light microscopy (PLM), verify the asbestos content by point counting using PLM.

Glove-Bag - A sealed compartment with attached inner gloves used for the handling of asbestos-containing materials. Properly installed and used glove bags provide a small Work Area enclosure typically used for small scale asbestos stripping operations. Information on glove-bag installation, equipment and supplies, and work practices is contained in the Occupational Safety and Health Administration's (OSHA's) final rule on occupational exposure to asbestos (29 CFR 1926.1101).

Glove-Bag Technique - A method with limited applications for removing small amounts of friable asbestos-containing material from HVAC ducts, short piping runs, valves, joints, elbows, and other non-planar surfaces in a non-contaminated work area. The glove-bag assembly is a manufactured or fabricated device consisting of a glove-bag (typically constructed of six (6) mil polyethylene or polyvinyl chloride plastic), two inward projecting long sleeves, an internal tool pouch, and an attached, labeled receptacle for asbestos waste. The glove-bag is constructed and installed in such a manner that it surrounds the object or material to be removed and contains all asbestos fibers released during the process.

High-efficiency particulate air (HEPA) A filter capable of trapping and retaining at least 99.97 percent of all mono-dispersed particles 0.3 microns in diameter.

Lock-down - The procedure of spraying polyethylene sheeting and building materials with an encapsulant type sealant to seal in non-visible asbestos-containing residue.

EnviroMed Services, Inc.

ISSUED FOR BIDDING March 12, 2024 Page 5 of 22 Movable Object - A piece of equipment or furniture in the Work Area which can be removed from the Work Area, as determined by the Owner.

Non-Friable Asbestos-containing Material - Material containing greater than or equal to 1 percent asbestos as determined using the method specified in Appendix A, subpart F, 40 CFR part 763, section 1, Polarized Light Microscopy, that when dry cannot be crumbled, pulverized or reduced to powder by hand pressure.

Permissible Exposure Limit (PEL) - An airborne concentration of asbestos, tremolite, anthophyllite, actinolite or a combination of these minerals of 0.1 fibers per cubic centimeter (f/cc) of air calculated as an eight (8) hour time-weighted average, as determined by Phase Contrast Microscopy.

Personal Monitoring - Air sampling within the breathing zone of an employee.

Pre-Clean - The process of cleaning an area before asbestos abatement activities begin to ensure all dust and debris in the area considered to be asbestos-containing are properly contained and disposed of. This increases the likelihood the area will pass aggressive air sampling clearance requirements after asbestos-containing materials have been removed.

Regulated Area - Area established by the employer to demarcate areas where airborne concentrations of asbestos, tremolite, anthophyllite, actinolite or a combination of these minerals exceed, or can reasonably be expected to exceed, the Permissible Exposure Limit.

Regulated Asbestos-Containing Material (RACM) - (a) Friable asbestos material, (b) Category I non-friable ACM that has become friable, (c) Category I non-friable ACM that will be or has been subjected to sanding, grinding, cutting, or abrading, or (d) Category II non-friable ACM that has a high probability of becoming or has become crumbled, pulverized, or reduced to powder by the forces expected to act on the material in the course of demolition or renovation operations.

Renovation - Altering a facility or one or more facility components in any way, including the stripping or removal of RACM from a facility component. Operations in which load-supporting members are wrecked or taken out are demolitions.

Repair - Overhauling, rebuilding, reconstructing or reconditioning of structures or substrates where asbestos, tremolite, anthophyllite or actinolite is present.

Unfinished Space - Space used for storage, utilities or work area where appearance is not a factor. Examples of an unfinished space include crawlspace; pipe tunnel and similar spaces.

Visible Emissions - Any emissions, which are visually detectable without the aid of instruments, coming from RACM or asbestos-containing waste material or from any asbestos milling, manufacturing, or fabricating operation. This does not include condensed, uncombined water vapor.

Visible Residue - Any debris or dust on surfaces in areas within the Work Area where asbestos abatement has taken place and which is visible to the unaided eye. All visible residue is assumed to contain asbestos.

Waste Generator - Any owner or operator of a source whose act or process produces asbestos-containing waste material.

Waste Shipment Record - The shipping document, required to be originated and signed by the waste generator, used to track and substantiate the disposition of asbestos-containing waste material.

Wet Cleaning - The process of eliminating asbestos contamination from building surfaces and objects by using cloths, mops, or other cleaning tools which have been dampened with water, and afterwards thoroughly decontaminated or disposed of as asbestos-contaminated waste.

Work Area - Specific area or location where the actual work is being performed or such other area of a facility which the Commissioner determines may be hazardous to public health as a result of such asbestos abatement.

Worker Decontamination Enclosure System - The portion of a Decontamination Enclosure System designed for controlled passage of workers and authorized visitors, typically consisting of a Clean Room, a Shower Room and an Equipment Room.

1.5 REFERENCES

- A. The current issue of each document shall govern. Where conflict among requirements or with these specifications exists, the more stringent requirements shall apply.
 - 1. Occupational Safety and Health Administration (OSHA)

29 CFR 1910.1001 - Asbestos, Tremolite, Anthophyllite, and Actinolite.

29 CFR 1910.134 - Respiratory Protection.

29 CFR 1926.21 - Safety Training and Education

29 CFR 1926.32 - Definitions

29 CFR 1926.51 - Sanitation

29 CFR 1910.134 - Gases, Vapors, Fumes, Dusts, and Mists

29 CFR 1926.59 - Hazard Communication.

29 CFR 1926.200 - Accident Prevention Signs and Tags.

29 CFR 1926.417 - Lockout and Tagging of Circuits.

29 CFR 1926.1101 - Asbestos

2. Environmental Protection Agency (EPA)

40 CFR 61, Subpart M - National Emission Standards for Hazardous Air Pollutants; Asbestos NESHAP Revision; Final Rule.

40 CFR 763, Subpart E - Asbestos Hazard Emergency Response Act (AHERA).

40 CFR 763, Subpart G - Worker Protection Rule.

3. State of Connecticut, Department of Public Health Regulations (DPH)

Section 19a-332a-1 through 19a-332a-16 - Standards for Asbestos Abatement.

Section 20-440-1 through 20-440-9 and 20-441 Licensure and Training.

4. American National Standards Institute (ANSI)

ANSI Z9.2 - Fundamentals Governing the Design and Operation of Local Exhaust Systems.

ANSI Z88.2 - Respiratory Protection.

5. American Society of Testing and Materials (ASTM)

ASTM E 84 - Surface Burning Characteristics of Building Materials.

ASTM E 96 - Water Vapor Transmission of Materials.

ASTM E 119 - Fire Tests of Building and Construction Materials.

ASTM E 736 - Cohesion/Adhesion of Sprayed Fire-Resistive Materials Applied to Structural Members.

ASTM E 1368 - Visual Inspection of Asbestos Abatement Projects.

ASTM E 1494 - Encapsulants for Spray- or Trowel-Applied Friable Asbestos-Containing Building Materials.

6. Underwriters Laboratories, Inc. (UL)

UL 586 - High-Efficiency, Particulate, Air Filter Units.

1.6 DOCUMENTATION

- A. Submit two copies of the following documentation to ensure compliance with the applicable regulations. An up to date copy shall be retained at the job site at all times.
- B. Manufacturer's Catalog Data:

MSDS for All Materials Delivered to the Site

C. Statements:

Connecticut Notification EPA Notification Worker Medical Certification Worker Training Certification Worker Respirator Fit Testing Worker Asbestos Licenses OSHA Laboratory Certification Landfill Approval Safety Plan Respirator Protection Plan Initial Exposure Assessment

- 1. Submit notification to the following agencies at least ten (10) working days before work commences on the project:
 - a. Department of Public Health Environmental Health Section 450 Capitol Avenue, MS#51AIR P.O. Box 340308 Hartford, CT 06134-0308
 - Asbestos Demo/Reno Notifications
 US EPA Region 1
 5 Post Office Square, Mail Code OES05-4
 Boston, MA 02109-3912
- 2. Copies of all required notifications, approvals and permits for the removal, disposal and transport asbestos-containing or contaminated materials.
- 3. Documentation from a physician certifying that all employees who may be exposed to airborne asbestos in excess of the background level have been provided with an opportunity to be medically monitored to determine whether they are physically capable of working while wearing the respirator required without suffering adverse health affects. In addition, document that personnel have received medical monitoring required in 29 CFR 1926.1101. They shall also be informed of the specific types of respirators the employee shall be required to wear and the work he/she will be required to perform as well as special work place conditions such as high temperature, high humidity and chemical contaminants which to which he/she may be exposed.
- 4. Documentation certifying that all employees have received training in the proper handling of materials that contain asbestos; understand the health implications and risks involved, including the illnesses possible from exposure to airborne asbestos fibers; understands the use and limits of respiratory equipment to be used; and understands the results of monitoring of airborne quantities of asbestos as related to health and respiratory equipment as indicated in 29 CFR 1926.1101 on an initial and annual basis.
- 5. Documentation of respiratory fit testing for all employees who must enter the Work Area. This fit testing shall be in accordance with qualitative procedures as detailed in 29 CFR 1926.1101.
- 6. Qualifications of the person proposed for air sampling to assure workers are using appropriate respiratory protection in accordance with OSHA Standard 1926.1101. Include the name and address of the testing laboratory proposed to perform air sample analysis on behalf of the Contractor, along with their NIOSH PAT Program I.D. number.
- 7. Establish and supervise in accordance with 29 CFR 1926.21, a program for the education and training of workers in the recognition, avoidance and prevention of unsafe conditions and the regulations applicable to the work environment to control or eliminate any hazards or other exposure to illness or injury. Include any site specific information to address health and safety procedures unique to this project.
- 8. Establish a written Respiratory Protection Plan in accordance with 29 CFR 1910.134. This plan shall establish procedures governing the selection and use of respirators and shall include such information as training in the proper use of respirators; medical examination of workers to determine whether or not they may be assigned an activity where respiratory protection is required; training in proper use and limitations of respirators; respirator fit testing; regular inspection and evaluation of the continued effectiveness of the program; and other elements included in the standard.
- 9. Demonstrate that employees exposure will be below the PEL's. For Class I asbestos work until the employer conducts exposure monitoring and documents that employees on that job will not be exposed in excess of the PEL's, or otherwise makes a negative exposure assessment, the employer shall presume that employees are exposed in excess of the TWA and excursion limit.
- D. Records:

Sign-in/out Logs Personal Air Sampling Results Waste Shipment Records Pressure Differential Recording Data

1.7 PERSONNEL PROTECTION

- A. Instruct workers in all aspects of personnel protection, work procedures, emergency evacuation procedures and use of equipment including procedures unique to this project.
- B. Ensure workers are fully protected with respirators and protective clothing during work in the Asbestos Control Area, where there is the possibility of disturbing asbestos-containing or asbestos-contaminated materials.
- C. Respiratory protection shall meet the requirements of OSHA as required in 29 CFR 1910.134 and 29 CFR 1926.1101. Provide appropriate respiratory protection for each

worker and ensure usage during potential asbestos exposure. As a minimum, workers shall be equipped with powered air-purifying respirators (PAPR) with HEPA filters.

- D. Select respirators from among those jointly approved as being acceptable for protection by the Mine Safety and Health Administration (MSHA) and the National Institute for Occupational Safety and Health (NIOSH) under the provisions of 30 CFR Part 11. Provide an adequate supply of filter elements for respirators in use.
- E. Minimum respiratory protection shall be as follows:

Airborne concentration of asbestos, tremolite, anthophyllite, actinolite or a combination of these minerals.	Required Respirator
Not in excess of 10 f/cc (100 x PEL)	 Any powered air purifying respirator equipped with high efficiency filters.
	2. Any supplied-air respirator operated in continuous flow mode.
Not in excess of 100 f/cc (1000 x PEL)	 Full facepiece supplied supplied air respirator operated in pressure demand mode.
Greater than 100 f/cc (>1000 x PEL) or unknown concentration	1. Full facepiece supplied air respirator operated in pressure demand mode equipped with an auxiliary positive pressure self- contained breathing apparatus.

Note: Respirators assigned for higher airborne fiber concentrations may be used at lower concentrations.

A high-efficiency filter means a filter that is at least 99.97 percent efficient against mono-dispersed particles of 0.3 micrometers in diameter or larger.

- F. Provide and require all workers to wear protective clothing in Work Areas where asbestos fiber concentrations exceed permissible limits established by OSHA. Protective clothing shall include impervious coveralls with elastic wrists and ankles, head covering, gloves and foot coverings.
- G. Provide all authorized persons entering contaminated areas with proper respirators and protective clothing.

- H. Ensure that all workers and authorized persons enter and leave the Asbestos Control Area through the Worker Decontamination Enclosure System.
- I. Ensure all contaminated protective clothing remains in the Equipment Room for reuse or disposal of as contaminated waste.
- J. Ensure workers do not eat, drink, smoke or chew gum or tobacco while in the Asbestos Control Area.

1.8 EQUIPMENT REMOVAL PROCEDURE

A. Clean surfaces of contaminated containers and equipment thoroughly by vacuuming with HEPA filtered equipment and wet wiping before moving such items into the Equipment Decontamination Enclosure System for final cleaning and removal to uncontaminated areas. Ensure that personnel do not leave the Asbestos Control Area through the Equipment Decontamination Enclosure System.

1.9 SEQUENCE OF WORK

- A. Proceed in accordance with the sequence of work as mutually agreed upon with the Construction Manager. Work shall be divided into convenient Work Areas, each of which is to be completed as a separate unit.
- B. The following sequence of work shall be used for the asbestos abatement work:
 - 1. A visual inspection of the Work Area to determine pre-existing damage to facility components.
 - 2. Release of floor area (Phase) to the Contractor.
 - 3. All temporary utilities required for the project shall be on site and operational prior to the initiation of asbestos work.
 - 4. Abatement of all asbestos-containing materials by the Contractor.
 - 5. Air sampling by the Project Monitor for reoccupancy.
 - 6. Containment tear-down and clean-up.

1.10 DELIVERY, STORAGE, AND HANDLING

A. Deliver all materials in the original packages, containers, or bundles bearing the name of the manufacturer and the brand name and product technical description. Do not use damaged or deteriorating materials. Material that becomes contaminated with asbestos shall be decontaminated or disposed of as asbestos waste.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. Fire retardant polyethylene sheet in roll size to minimize the frequency of joints, shall be delivered to job site with factory label indicating four (4) or six (6) mil.
- B. Polyethylene disposable bags shall be six (6) mil with pre-printed label. Disposable bags shall be opaque.
- C. Tape shall be capable of sealing joints in adjacent polyethylene sheets and for attachment of polyethylene sheet to finished or unfinished surfaces. Tape must be capable of adhering under both dry and wet conditions.
- D. Surfactant (wetting agent) shall consist of fifty (50) percent polyoxyethylene ether and fifty (50) percent polyoxyethylene ester, or equivalent, and shall be mixed with water to provide a concentration one (1) ounce surfactant to five (5) gallons of water or as directed by the manufacturer.
- E. Containers must be impermeable and shall be both air and watertight. Containers shall be labeled in accordance with OSHA Standard 29 CFR 1926.1101 and EPA 40 CFR Part 61.152 as appropriate.
- F. Labels and signs shall conform to OSHA Standard 29 CFR 1926.1101.
- G. Encapsulant shall be bridging or penetrating type which has been approved by the Design Consultant. Usage shall be in accordance with manufacturer's printed technical data. Encapsulant must be compatible with new materials being installed. Encapsulant shall dry clear.
- H. Glove-bag assembly shall be manufactured of six (6) mil transparent polyethylene or PVC with two (2) inward projecting long sleeve gloves, an internal pouch for tools, and an attached labeled receptacle for waste.

2.2 TOOLS AND EQUIPMENT

- A. Tools and equipment shall be suitable for asbestos removal.
- B. Protective clothing, respirators, filter cartridges, air filters and sample filter cassettes shall be provided in sufficient quantities for the project.
- C. Electrical equipment, protective devices and power cables shall conform to all applicable codes.
- D. Shower stalls and plumbing shall include sufficient hose length and drain system or an acceptable alternate. Showers shall be equipped with hot and cold or warm running water. One shower stall shall be provided for each eight workers.
- E. Exhaust air filtration units shall be equipped with HEPA filters capable of providing sufficient air exhaust to create a minimum pressure differential of 0.02 inches of water column, and to allow a sufficient flow of air through the area. An automatic warning system shall be incorporated into the equipment to indicate pressure drop or unit

failure. No air movement system or air filtering equipment shall discharge unfiltered air outside the Asbestos Control Area.

- F. Pressure differential monitoring equipment shall be provided to ensure exhaust air filtration devices provide the minimum pressure differential required between the Work Area and occupied areas of the facility.
- G. Spray equipment shall be capable of mixing wetting agent with water and capable of generating sufficient pressure and volume. Hose length shall be sufficient to reach all of the Asbestos Control Area.
- H. Vacuum units, of suitable size and capabilities for the project, shall have HEPA filters capable of trapping and retaining at least 99.97 percent of all monodispersed particles of 0.3 microns in diameter or larger.
- I. Ladders and/or scaffolds shall be of adequate length, strength and sufficient quantity to schedule.
- J. Other materials such as lumber, nails and hardware necessary to construct and dismantle the decontamination enclosures and the barriers that isolate the Work Area shall be provided as appropriate for the work.

PART 3 - EXECUTION

3.1 PREPARATION OF WORK AREA ENCLOSURE SYSTEM

- A. Prior to beginning work, the Owner, Design Consultant, and Contractor shall conduct a pre-abatement meeting, perform a visual survey of each Work Area and list all pre-existing damage to building components. The Contractor shall submit to the Owner a list which shall include all damaged areas not scheduled to be repaired under this Contract and include photographs, video tapes as applicable.
- B. Post warning signs meeting the specifications of OSHA 29 CFR 1910 and 29 CFR 1926.1101 at each Regulated Area. In addition, signs shall be posted at all approaches to Regulated Areas so that an employee may read the sign and take the necessary protective steps before entering the area. Additional signs may require posting following construction of work place enclosure barriers.
- C. Utilize engineering controls and personnel protective equipment while installing enclosures and supports when asbestos-containing materials may be disturbed.
- D. When feasible, shut down and lock out electrical power, including all receptacles and light fixtures. Protect receptacles and light fixtures remaining in the Work Area with six (6) mil polyethylene and seal with tape. Coordinate all power isolation with the Owner.
- E. Provide temporary power and lighting and ensure safe installation, including ground fault protection, of temporary power sources and equipment in compliance with applicable electrical code and OSHA requirements. The Contractor is responsible for proper connection and installation of electrical wiring.

- F. Shut down and isolate heating, cooling, and ventilating air systems to prevent contamination and fiber dispersal to other areas of the building. Seal all vents.
- G. Pre-clean movable objects within the proposed Work Areas using HEPA filtered vacuum equipment and/or wet cleaning methods as appropriate and remove such objects from Work Areas to a temporary location.
- H. Pre-clean fixed objects within the proposed Work Areas, using HEPA filtered vacuum equipment and/or wet cleaning methods as appropriate, and enclose with two layers of six (6) mil polyethylene sheeting sealed with tape.
- I. Clean the proposed Work Areas using HEPA filtered vacuum equipment and/or wet cleaning methods as appropriate. Do not use methods that raise dust, such as dry sweeping or vacuuming with equipment not equipped with HEPA filters.
- J. Seal off all windows, doorways, skylights, ducts, grilles, diffusers, and any other openings between the Work Area and the uncontaminated areas outside of the Work Area with critical barriers. Doorways and corridors which will not be used for passage during work must be sealed with fixed critical barriers.
- K. Cover floor and wall surfaces with polyethylene sheeting sealed with tape. Polyethylene shall be applied alternately to floors and walls. Cover floors first, with a layer of six (6) mil polyethylene sheeting, so that polyethylene extends at least twelve (12) inches up on walls. Cover walls with a layer of four (4) mil polyethylene sheeting to twelve (12) inches beyond the wall floor intersection, thus overlapping the floor material by a minimum of twenty-four (24) inches. Repeat the process for the second layer of polyethylene. There shall be no seams in the plastic sheet at wall-to-floor joints.
- L. Conspicuously label and maintain emergency and fire exits from the Asbestos Control Area satisfactory to fire officials.

3.2 WORKER DECONTAMINATION ENCLOSURE SYSTEM

- A. Establish contiguous to the Work Area, a Worker Decontamination Enclosure System consisting of Equipment Room, Shower Room and Clean Room in series. Access to the Work Area shall only be through this enclosure.
- B. Access between rooms in the Worker Decontamination Enclosure System shall be through double flap curtained openings (air locks). Other effective designs are permissible. The Clean Room, Shower Room and Equipment Room located within the Worker Decontamination Enclosure, shall be completely sealed ensuring sole source of air flow into the Asbestos Control Area originates from the outside uncontaminated areas.
- C. The Clean Room shall be adequately sized to accommodate workers and shall be equipped with a suitable number of hooks, lockers, shelves, etc., for workers to store personal articles and clothing. Changing areas of the Clean Room shall be suitably screened from areas occupied by the public.

- D. The Shower Room shall be of sufficient capacity to accommodate the number of workers. Supply warm water to showers. Provide one shower for each eight workers. No worker or other person shall leave an Asbestos Control Area without showering. Shower water shall be collected and filtered using best available technology and dumped down an approved drain.
- E. No personnel or equipment shall be permitted to leave the Asbestos Control Area unless just decontaminated by showering, wet cleaning or HEPA vacuuming to remove all asbestos debris. No asbestos-contaminated materials or persons shall enter the Clean Room.

3.3 EQUIPMENT DECONTAMINATION ENCLOSURE SYSTEM

A. Establish contiguous to the Work Area an Equipment Decontamination Enclosure System consisting of two (2) totally enclosed chambers divided by a double flap curtained opening. Other effective designs are permissible. This enclosure must be constructed so as to ensure that no personnel enter or exit through this unit.

3.4 SEPARATION OF WORK AREAS FROM OCCUPIED AREAS

- A. Occupied areas and/or building space not within the Asbestos Control Area shall be separated from asbestos abatement Work Areas by means of airtight barriers. Barriers at openings with dimensions exceeding two (2) feet in both directions shall be blocked with fixed critical barriers.
- B. Do not impair required building exits from any occupied building area. Where normal exits have been blocked by the asbestos work, provide temporary exit signs directing building occupants to the nearest available exit location.
- C. Visually inspect and smoke test NPE barriers to assure an effective seal. Repair defects immediately.
- D. Create a pressure differential in the range of 0.02 to 0.04 inches of water column between the Work Area and occupied areas by the use of acceptable pressure differential equipment. Provide a sufficient quantity of units to exhaust the volume of air within the Asbestos Control Area a minimum of four times per hour. Continuously monitor the pressure differential between the Work Area and occupied areas utilizing recording type equipment to ensure exhaust air filtration equipment maintains a minimum pressure differential of 0.02 inches of water column.

3.5 ASBESTOS REMOVAL – INTERIOR ABATEMENT

- A. A Competent Person shall be on the job at all times to ensure the establishment and maintenance of the NPE and proper work practices throughout the project. Before beginning work within the NPE and at the beginning of each shift, the NPE shall be inspected for breaches and smoke tested for leaks, and any leaks sealed. Results of NPE inspections shall be logged.
- B. Do not begin abatement work until authorized by the Project Monitor.

- C. Spray asbestos materials with amended water, using airless spray equipment capable of providing a "mist" application to reduce the release of fibers during the removal operation.
- D. In order to maintain indoor asbestos concentrations at a minimum, remove the wet asbestos in manageable sections. Materials shall not be allowed to dry out. Material drop shall not exceed 8 feet. For heights up to 15 feet provide inclined chutes or scaffolding to intercept drop. For heights exceeding 15 feet provide enclosed dust-proof chutes.
- E. Fill disposal containers (six (6) mil polyethylene bags or fiber drums) as removal proceeds, seal filled containers, apply caution labels and clean containers before removal to wash area. Bags shall be securely sealed to prevent accidental opening and leakage by taping in gooseneck fashion. Bags may be placed in drums for staging and transportation to the disposal site. Bags shall be decontaminated by wet cleaning and HEPA vacuuming before being placed in clean drums and sealed with locking ring tops. Vinyl asbestos floor tile removed shall be placed in polypropylene burlap bags and then double poly bagged. Small components and asbestos containing waste with sharp-edged components (e.g., nails, screws, metal lath, tin sheeting) which could tear polyethylene bags and sheeting shall be placed in polypropylene burlap bags and then double poly bagged. Wet clean each container thoroughly before moving to Holding Area. Ensure that workers do not enter from uncontaminated areas into the Washroom or the Work Area. Ensure that contaminated workers do not exit the Work Area through the Equipment Decontamination Enclosure.
- F. After completion of stripping work, all surfaces from which asbestos has been removed shall be wet brushed, using a nylon brush, wet wiped and sponged or cleaned by an equivalent method to remove all visible material (wire brushes are not permitted). During this work the surfaces being cleaned shall be kept wet.
- G. If at any time during asbestos removal, should the Project Monitor suspect contamination of areas outside the Work Area, the Contractor shall stop all abatement work and take steps to decontaminate these areas and eliminate causes of such contamination. Unprotected individuals shall be prohibited from entering contaminated areas until air sampling and visual inspections determine decontamination.
- H. Containerize asbestos-containing waste material removed daily. Do not allow ACM to remain on the floor overnight, allowing it to dry out.

3.6 ASBESTOS REMOVAL – EXTERIOR ABATEMENT

- A. A Competent Person shall be on the job at all times to ensure the establishment and maintenance of control measures and proper work practices throughout the project.
- B. Do not begin abatement work until authorized by the Project Monitor.
- C. Spray asbestos materials with amended water, using airless spray equipment capable of providing a "mist" application to reduce the release of fibers during the removal operation.

- D. Cover ground under exterior removal activity with 10' wide 6 mil polyethylene sheeting, weighted to withstand wind loading. Create a regulated area around the exterior removal area with warning tape and warning signs.
- E. Fill disposal containers (six (6) mil polyethylene bags or fiber drums) as removal proceeds, seal filled containers, apply caution labels and clean containers before removal to wash area. Bags shall be securely sealed to prevent accidental opening and leakage by taping in gooseneck fashion. Bags may be placed in drums for staging and transportation to the disposal site.
- F. If at any time during asbestos removal, should the Project Monitor suspect contamination of areas outside the Work Area, the Contractor shall stop all abatement work and take steps to decontaminate these areas and eliminate causes of such contamination. Unprotected individuals shall be prohibited from entering contaminated areas until air sampling and visual inspections determine decontamination.
- G. Containerize asbestos-containing waste material removed daily. Do not allow ACM to remain on the ground overnight.

3.7 ALTERNATIVE WORK PRACTICE (AWP) PROCEDURES

- A. The procedures described in this specification are to be utilized as the basis for bidding this project.
- B. Alternative procedures require written letters of approval from the following parties:

Department of Public Health – Asbestos Program

The Contractor may not conduct asbestos removal utilizing the Alternative Work Practice until the written Alternative Work Practice approval letter from the Department of Public Health is on the job site. Alternative Work Practice approvals shall be secured prior to implementation.

- C. Allow 21 calendar days for the processing of written requests for Alternate Work Practices by the Owner and associated review parties. Alternate Work Practices may not be utilized without Owner approval.
- D. Written requests for Alternate Work Practices must be accompanied by a written itemized credit proposal to the Owner detailing the labor and material costs that will be credited to the Contract if the Alternate Work Practice is approved. Written requests for Alternate Work Practices must be accompanied by a written assessment of the schedule impact of utilizing the proposed Alternate Work Practice.
- E. Girls Inc. reserves the right to reject any proposed Alternative Work Practice without cause.
- F. The Contractor shall be responsible for all fees associated with filing Alternative Work Practice (AWP) applications. Submission of AWP applications requires a Connecticut DPH Asbestos Project Designer license. The Contractor is responsible for retaining a licensed Asbestos Project Designer to prepare the Alternate Work Practice. The licensed Asbestos Project Designer that prepares the Alternate Work Practice may not

be an employee of the Contractor or an employee of a Subcontractor under contract with the Contractor.

G. Submit written requests for Alternate Work Practices to the Owner.

3.8 CLEAN-UP PROCEDURE

- A. Remove and containerize all visible accumulations of asbestos-containing and/or asbestos-contaminated debris which may have splattered or collected on the polyethylene wall covering.
- B. Remove contamination from the exteriors of the negative air machines, scaffolding, ladders, extension cords, hoses and other equipment inside the Work Area. Cleaning may be accomplished by brushing, HEPA vacuuming and/or wet cleaning.
- C. The Project Monitor shall conduct a thorough visual inspection utilizing a high-intensity flashlight, with the containment barriers in place, to detect visible accumulations of dust or bulk asbestos-containing materials remaining in the Work Area. Should dust, debris or residue be detected, the Contractor shall repeat the cleaning, at the Contractor's expense, until the area is in compliance. The visual inspection will detect incomplete work, damage caused by the abatement activity, and inadequate clean-up of the work site.
- D. Once the area has been recleaned, any equipment, tools or materials not required for completion of the work, shall be removed from the Work Area. Negative air filtration devices shall remain in place and operating for the remainder of the clean-up operation.
- E. Apply a lock-down encapsulant to all surfaces within the Work Area from which asbestos has been removed and the cleaned inner layer of polyethylene.
- F. Air sampling for reoccupancy clearance shall be undertaken using aggressive sampling techniques. Analysis of clearance samples shall follow State of Connecticut Regulations, Section 19a-332a-12. Areas which do not comply shall continue to be cleaned by and at the Contractors expense, until the specified Standard of Cleaning is achieved as evidenced by results of air testing. When the Work Area passes the reoccupancy clearance, controls established by this specification may be removed.
- G. Remove all remaining polyethylene, including critical barriers, and Decontamination Enclosure Systems leaving negative air filtration devices in operation. Dispose of poly sheeting as asbestos-contaminated waste. HEPA vacuum and/or wet wipe any visible residue which is uncovered during this process.

3.9 REINSTALLATION OF DISPLACED EQUIPMENT

A. After reoccupancy is granted, resecure mounted items removed during the course of the work to their former positions.

3.10 DISPOSAL OF ASBESTOS

- A. Disposal of asbestos-containing and/or asbestos contaminated material shall occur at an authorized site and must be in compliance with the requirements of, and authorized by the Office of Solid Waste Management, Department of Energy and Environmental Protection, State of Connecticut, or other designated agency having jurisdiction over solid waste disposal.
- B. Disposal approval shall be obtained prior to commencement of asbestos removal.
- C. Warning signs must be attached to vehicles used to transport asbestos-containing waste. Warning signs shall be posted during loading and unloading of disposal containers. The signs must be posted so that they are plainly visible.
- D. Waste removal dumpsters and cargo areas of transport vehicles shall be lined with a layer of six (6) mil polyethylene sheeting to prevent contamination from leaking or spilled containers. Floor sheeting shall be installed first, and shall be extended up sidewalls 12-inches. Wall sheeting shall overlap floor sheeting 24-inches and shall be taped into place. Keep dumpsters locked when not in use.
- E. Contractor is responsible for signing the asbestos waste shipment record as generator prior to each asbestos waste dumpster leaving site and giving a copy of the signed waste shipment record to the Owner. The completed waste shipment record with landfill sign-offs shall be forwarded to the Owner.

3.11 CONTRACTOR RESPONSIBILITY

A. Conduct air sampling, as necessary, to assure that workers are using appropriate respiratory protection in accordance with OSHA Standard 1926.1101. Documentation of air sampling results must be recorded at the work site within twenty-four (24) hours of receipt of results, and shall be available for review until the job is complete.

3.12 AIR SAMPLING SCHEDULE

A. At a minimum, air sampling by the Project Monitor will be conducted in accordance with the following schedule:

Abatement Activity	Pre- <u>Abatement</u>	During <u>Abatement</u>	Post- <u>Abatement</u>
Greater than 500 l.f. or 1500 s.f.	PCM	PCM	TEM
Equal to or less than 500 l.f. or 1500 s.f.	PCM	PCM	PCM

- B. Frequency and duration of the air sampling during abatement will be representative of the actual conditions during the abatement. The size of the asbestos project will be a factor in the number of samples required to monitor the abatement activities. In addition to OSHA compliance monitoring (personal sampling accomplished by the Contractor) the following minimum schedule of samples will be required:
 - 1. Background Samples:
 - a) Outside of building 2.
 - b) Adjacent Area(s) inside building 2.
 - c) Work Area 3 or if areas are separated (such as rooms) at least one (1) sample per area equalling a minimum of three (3).
 - 2. During Abatement:
 - a) Outside of building at the exhaust of air filtering device 2 per shift.
 - b) Work Area 2 per shift.
 - c) Adjacent area inside building 2 per shift.
 - d) Outside of the Equipment Decontamination Enclosure System 1 during removal of ACM waste.
 - 3. Post-Abatement:
 - a) Work Area At least five (5) per homogenous work site or one (1) per room, whichever is greater.
- C. Post-abatement clearance air monitoring requirements are as follows:
 - 1. Air sampling will not begin until at least 12 hours after wet cleaning has been completed and no visible water or condensation remain.
 - 2. Sampling equipment will be placed at random around the Work Area. If the Work Area contains the number of rooms equivalent to the number of required samples based on floor area, a sampler shall be placed in each room. When the number of rooms is greater than the number of samples a representative number of rooms will be selected.
 - 3. The representative samplers placed outside the Work Area but within the building will be located to avoid any air that might escape through the isolation barriers and will be approximately 50 feet from the entrance to the Work Area, and 25 feet from the isolation barriers.
 - 4. The following aggressive air sampling procedures will be used within the Work Area during all air clearance monitoring:
 - a) Before starting the sampling pumps, direct the exhaust from forced air equipment (such as a 1 horsepower leaf blower) against all walls,

ceilings, floors, ledges and other surfaces in the Work Area. This should take at least 5 minutes per 1000 SF of floor area.

- b) Place a 20-inch fan in the center of the room. (Use one fan per 10,000 cubic feet of room space.) Place the fan on slow speed and point it toward the ceiling.
- c) Start the sampling pumps and sample for the required time.
- d) Turn off the pump and then the fan(s) when sampling is complete.
- 5. Air volumes taken for clearance sampling shall be sufficient to accurately determine (to a 95 percent probability) fiber concentrations to 0.010 f/cc of air.
- 6. The clearance criteria for work areas cleared by PCM (Phase Contrast Microscopy) is that all 5 clearance samples must register less than or equal to 0.010 f/cc of air.
- 7. The clearance criteria for work areas cleared by TEM (Transmission Electron Microscopy) is that the average of the 5 clearance samples taken inside the work area must register less than 70 structures per square millimeter of filter area.
- 8. Each homogeneous Work Area which does not meet the clearance criteria shall be thoroughly recleaned using HEPA vacuuming and/or wet cleaning, with the negative pressure ventilation system in operation. New samples shall be collected in the Work Area as described above. The process shall be repeated until the Work Area passes the test, with the cost of repeat sampling being borne entirely by the Contractor.
- 9. For an asbestos abatement project with more than one homogeneous Work Area, the release criterion shall be applied independently to each Work Area.
- D. TEM clearance turnaround time will be 24 hours after the TEM lab receives the samples by overnight mail.

3.13 ACTION CRITERIA

A. If air samples collected outside of the Work Area during abatement activities indicate airborne fiber concentrations greater than original background levels or greater than 0.010 f/cc, as determined by Phase Contrast Microscopy, whichever is larger, an examination of the Work Area perimeter shall be conducted and the integrity of barriers shall be restored. Cleanup of surfaces outside the Work Area using HEPA vacuum equipment or wet cleaning techniques shall be done prior to resuming abatement activities.

END OF SECTION 02 8216

SECTION 02 8233 - LEAD ABATEMENT

PART 1 – GENERAL

1.1 RELATED DOCUMENTS

Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 DEFINITIONS

- A. Action Level: Employee exposure, without regard to the use of respirators, to an airborne concentration of lead of 30 micrograms per cubic meter (µg/m3) calculated as an eight hour time weighted average.
- B. Competent Person: An individual capable of identifying existing lead hazards and taking corrective measures to eliminate them. The duties of the competent person include at least the following: controlling entry to and exit from the lead abatement area, ensuring contract compliance, conducting personal air monitoring, ensuring that all employees working within the lead abatement area have lead awareness training, and ensuring that lead abatement workers use the hand washing facilities.
- C. Lead Based Paint: Paint found to contain greater than or equal to 1.0 milligrams of lead per square centimeter of paint area by XRF testing.
- D. Permissible Exposure Limit (PEL): Fifty (50) micrograms per cubic meter (µg/m3) of air averaged over an 8 hour period as determined by 29 CFR 1926.62.
- E. Personal Monitoring: Sampling of lead concentrations within the breathing zone of a worker to determine the 8-hour time weighted average concentration in accordance with 29 CFR 1926.62. Samples shall be representative of the employee's work tasks.
- F. Toxicity Characteristic Leaching Procedure (TCLP): Toxicity characteristic leaching procedure utilizing EPA Test method SW-846, Method 1311.

1.3 GENERAL

- A. Scope: Paint removal, painted component removal, paint prep, drilling & cutting through painted components and disposal in support of renovations to Girls Inc. Location of lead-based paint in the building is documented in Drawings H100, H101, H102, H103, H120, H200, & H203; and in the attached lead testing XRF data.
- B. Intent: The intent of this specification is to assist the Contractor in complying with the OSHA Construction Industry Standard for Lead (29 CFR Part 1926.62) and State DEEP regulations for waste disposal during renovations at Girls Inc.

- C. OSHA Compliance: Contractor is responsible for compliance with the OSHA Construction Industry Standard for Lead (29 CFR Part 1926.62) when conducting renovation activities at Girls Inc. Contractor is to maintain a copy of 29 CFR 1926.62 on site. OSHA regulates the disturbance of paint containing any measurable lead content.
- D. Notification: Contractor shall obtain a small quantity hazardous waste generator ID number from the State DEEP for the site.
- 1.4 LEAD PAINT IDENTIFICATION
 - A. Toxic (≥1 mg/cm2) levels of lead are identified on the drawings and in the attached lead XRF data.
- 1.5 DEBRIS DISPOSAL
 - A. The following materials shall be disposed of as hazardous lead waste with a TCLP reading >5 mg/l:
 - Paint chips
 - Painted wood
 - Painted wallboard

1.6 EMPLOYEE EXPOSURE ASSESSMENT

A. Conduct lead dust air sampling to assure that workers are protected in accordance with all applicable Federal, State, and Local regulations. Documentation of air sampling results must be recorded at the work site within five (5) working days and shall be available for review until the job is complete.

1.7 WORKER HAND WASHING FACILITY

- A. Establish a lead abatement worker hand washing facility consisting of running potable water, a wash basin, hand cleaning cream, and towels.
- B. All lead abatement workers shall wash their hands and face prior to taking breaks, going to lunch, and leaving the site at the end of the work day.

1.8 WORKER PROTECTION PROCEDURES

- A. Workers shall not eat, drink, smoke, or chew gum or tobacco while in the lead abatement area.
- 1.9 WORKER TRAINING REQUIREMENT
 - A. All Contractor employees on the job site are required to be trained regarding any hazardous materials, including lead, they may be exposed to at the work site and

health and safety hazards at the work site. Training shall meet the requirements of the Hazard Communication Standard (29 CFR 1926.59) and Safety Training and Education Standard (29 CFR 1926.21).

- All workers impacting lead-containing building materials shall receive at least 4 hours of lead awareness training meeting the requirements of 29 CFR 1926.62.
 Documentation of said training is to be kept on-site in a binder for review. The awareness training must be dated within one year of the project.
- 1.10 COMPETENT PERSON
 - A. Maintain a competent person on site for the duration of lead abatement.
- 1.11 LEAD ABATEMENT PLAN
 - A. Prior to construction, in compliance with the OSHA Lead Standard, prepare a written description of lead abatement activities planned detailing methods, equipment, engineering controls, crew size, employee job responsibilities, operating and maintenance procedures.
 - B. Submit a copy of the Lead Abatement Plan to the A/E for review. Maintain a copy on site during construction.
- PART 2 PRODUCTS NOT USED
- PART 3 EXECUTION

3.1 PAINT DISTURBANCE METHODS

- A. Equip tools used to drill, cut, or sand surfaces painted with lead-based paint with HEPA vacuum cowled dust collection.
- B. Restrict access to rooms where lead paint disturbance is taking place to persons with lead awareness training.
- 3.2 PAINT DISPOSAL
 - A. Barrel paint chips, painted wood, and painted wallboard in 55 gallon steel drums.
 - B. Label steel drums as containing hazardous lead waste.
 - C. Transport and dispose of removed paint as hazardous lead waste.
 - D. Return completed waste manifest for disposal of paint waste to Construction Manager.

INSPECTION SITE:	130 Licoln Street, Meriden, CT
INSPECTION DATE:	1/31/2024 - 2/8/2024
INSTRUMENT TYPE:	Viken Detection Pb200i XRF Lead Paint Analyzer 2575
ACTION LEVEL:	1.0 (mg/cm ²)
Job ID:	girls inc
STATEMENT:	Limited Lead Screening

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024
 Inspection Site:

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) Read # Result -->RoomChoice -->Member Substrate **Room ID** Color Condtion Structure Wall 2558 1.0 mg/cm² (CAL) 2559 1.1 mg/cm² (CAL) 2560 1.0 mg/cm² (CAL) 2561 Negative Attic Storage Room Wall Drywall Α 300 Beige Deteriorated 0.4 mg/cm^2 2562 Negative Attic Storage Room Wall Drywall С 300 Beige Intact 0.3 mg/cm² 2563 Negative Attic Storage Room Wall Drywall В 300 Beige Intact 0.4 mg/cm^2 Attic Storage Wall 0.3 mg/cm² 2564 Negative Room Drywall D 300 Beige Intact 2565 Negative Attic Storage Room Ceiling Drywall 300 Beige Intact 0.3 mg/cm^2 2566 Negative Attic Storage Room Ceiling Drywall 300 Beige Intact 0.2 mg/cm² 2567 Negative Attic Closet Room Ceiling Drywall 303 Beige Intact 0.3 mg/cm^2 2568 Negative Attic Closet Room Ceiling Drywall В 303 Beige Intact 0.4 mg/cm^2 2569 Attic Closet D 303 0.4 mg/cm^2 Negative Room Ceiling Drywall Beige Intact Attic Closet Wood 2570 Positive Room Baseboard D 303 Intact 29.3 mg/cm² Beige Baseboard Wood С 2571 Positive Attic Storage Room 300 Beige Intact 27.6 mg/cm² 2572 Positive Attic Storage Room Baseboard Wood В 300 Beige Intact 28.0 mg/cm² Wood D 2573 Positive Attic Storage Room Baseboard 300 Beige Intact 25.5 mg/cm² 2574 Positive Attic Storage Room Wall Drywall Α 304 Beige Deteriorated 3.9 mg/cm² 2575 Positive Wall Drywall Α 304 Attic Storage Room Beige Intact 5.4 mg/cm² 2576 Wall В 2.6 mg/cm² Positive Attic Storage Room Drywall 304 Beige Intact 2577 Positive Attic Storage Room Wall Drywall D 304 Beige Intact 4.6 mg/cm² 2578 Negative Attic Storage Room Ceiling Drywall 304 Beige Intact 0.2 mg/cm² 2579 Positive Attic Storage Room Baseboard Wood Α 304 Beige Intact 5.4 mg/cm² 2580 Baseboard Wood D 5.2 mg/cm² Positive Attic Storage Room 304 Beige Intact 2581 Positive Baseboard Wood В 304 Intact Attic Storage Room Beige 5.3 mg/cm² 2582 Room Ceilina Drvwall 304 0.2 ma/cm² Negative Attic Storage Beige Intact

Inspection Site:

130 Licoln Street, Meriden, CT

Inspection Date: Total Readings: Unit Started: Action Level: Unit Ended:

1/31/2024 - 2/8/2024 1.0 (mg/cm²) 481 01/31/2024 10:54:25 02/08/2024 14:40:07

Read #	Result	>RoomChoice	Structure	>Member	Substrate	Wall	Room ID	Color	Condtion	Lead (mg/cm²)
2583	Positive	Attic Storage	Room	Wall	Drywall	A	305	Beige	Deteriorated	2.1 mg/cm ²
2584	Positive	Attic Storage	Room	Wall	Drywall	U	305	Beige	Intact	3.8 mg/cm ²
2585	Positive	Attic Storage	Room	Wall	Drywall	Ξ	305	Beige	Intact	2.9 mg/cm ²
2586	Positive	Attic Storage	Room	Wall	Drywall	۵	305	Beige	Intact	3.2 mg/cm ²
2587	Negative	Attic Storage	Room	Ceiling	Drywall		305	Beige	Deteriorated	0.2 mg/cm ²
2588	Positive	Attic Storage	Room	Baseboard	Wood	U	305	Beige	Intact	5.3 mg/cm ²
2589	Positive	Attic Storage	Room	Baseboard	Wood	A	305	Beige	Intact	3.6 mg/cm ²
2590	Negative	Attic Storage	Room	Wall	Drywall	U	305	Beige	Deteriorated	0.2 mg/cm ²
2591	Negative	Attic Storage	Room	Wall	Drywall	A	305	Beige	Deteriorated	0.2 mg/cm ²
2592	Negative	Attic Storage	Room	Wall	Drywall	в	305	Beige	Deteriorated	0.8 mg/cm ²
2593	Negative	Attic Storage	Room	Wall	Drywall	۵	305	Beige	Deteriorated	0.6 mg/cm ²
2594	Negative	Attic Storage	Room	Ceiling	Drywall		305	Beige	Deteriorated	0.4 mg/cm ²
2595	Positive	Attic Storage	I-Beam		Wood		305	Blue	Intact	25.2 mg/cm ²
2596	Negative	Attic Storage	Closet	Wall	Drywall		305	Beige	Intact	0.3 mg/cm ²
2597	Positive	Attic Storage	Room	Baseboard	Wood		305	Beige	Intact	23.6 mg/cm ²
2598	Negative	Attic Hallway	Room	Ceiling	Drywall		302	Beige	Deteriorated	0.0 mg/cm ²
2599	Negative	Attic Hallway	Room	Ceiling	Wood		302	White	Deteriorated	0.1 mg/cm ²
2600	Positive	Attic Hallway	Room	Wall	Drywall	U	302	Beige	Deteriorated	2.7 mg/cm ²
2601	Positive	Attic Hallway	Room	Wall	Drywall	Ξ	302	Beige	Deteriorated	4.5 mg/cm ²
2602	Positive	Attic Hallway	Room	Wall	Drywall	۵	302	Beige	Deteriorated	3.0 mg/cm ²
2603	Positive	Attic Hallway	Room	Wall	Drywall	۷	302	Beige	Deteriorated	5.8 mg/cm ²
2604	Negative	Attic Hallway	Room	Ceiling	Drywall	A	302	Beige	Intact	0.3 mg/cm ²
2605	Positive	Attic Hallway	Room	Baseboard	Wood	Ξ	302	Beige	Intact	4.1 mg/cm ²
2606	Positive	Attic Hallway	Room	Baseboard	Wood	U	302	Beige	Intact	4.0 mg/cm ²
2607	Positive	Attic Hallway	Room	Baseboard	Wood		302	Beige	Intact	3.5 mg/cm ²
2608	Negative	Attic Stairs	Room	Ceiling	Drywall		301	White	Deteriorated	0.9 mg/cm ²

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 10.8 mg/cm² 13.7 mg/cm² 16.2 mg/cm² 11.0 mg/cm² 14.8 mg/cm² 13.8 mg/cm² 14.1 mg/cm² 18.0 mg/cm² 14.2 mg/cm² 28.0 mg/cm² 12.1 mg/cm² 13.3 mg/cm² 21.9 mg/cm² 19.9 mg/cm² 18.5 mg/cm² 12.8 mg/cm² 1.0 mg/cm² 0.5 mg/cm² 0.3 mg/cm² 0.4 mg/cm² 0.4 mg/cm² 0.0 mg/cm² 4.0 mg/cm² 3.9 mg/cm² 0.4 mg/cm² 3.4 mg/cm² Condtion Deteriorated Intact Color White Green White White White Green Beige Beige White White White White White White White Beige Beige Room ID 202 202 202 202 206 206 301 301 **301** 202 202 202 202 202 202 202 202 202 202 301 301 301 301 301 301 206 Wall ഫ ۵ < υ Δ 4 U O U ∢ ഫ ∢ ∢ ∢ ш C ∢ Substrate Drywall Drywall Drywall Drywall Drywall Drywall Drywall Drywall Drywal Nood Nood Nood Mood Mood Nood Wood Wood Wood Wood Nood Wood Wood Wood Wood Wood Wood -->Member Baseboard Baseboard Baseboard Baseboard Baseboard Baseboard Baseboard Baseboard Door Trim Door Trim Ceiling Casing Jamb Ceiling Header Wall Structure Door Trim Door Trim Room Stair Stair Door Door Door Stair Stair -->RoomChoice Attic Stairs Classroom Classroom Classroom Hallway Result Negative Negative Negative Negative Negative Positive Negative Positive Read # 2613 2617 2619 2609 2615 2618 2623 2625 2628 2633 2610 2611 2612 2614 2616 2620 2621 2622 2624 2626 2627 2629 2630 2631 2632 2634

Inspection Site:

130 Licoln Street, Meriden, CT

Inspection Date: Total Readings: Unit Started: Action Level: Unit Ended:

1/31/2024 - 2/8/2024 1.0 (mg/cm²) 481 01/31/2024 10:54:25 02/08/2024 14:40:07

:										
Kead #	Kesult	>KoomChoice	Structure	>Member	Substrate	Wall	Koom ID	Color	Condtion	Lead (mg/cm²)
2635	Positive	Classroom	Room	Wall	Drywall	В	206	Yellow	Intact	12.4 mg/cm ²
2636	Positive	Classroom	Room	Baseboard	Wood	A	206	White	Intact	15.4 mg/cm ²
2637	Positive	Classroom	Room	Baseboard	Wood	U	206	White	Intact	13.4 mg/cm ²
2638	Positive	Classroom	Room	Baseboard	Wood	в	206	White	Intact	8.7 mg/cm ²
2639	Positive	Classroom	Room	Baseboard	Wood	۵	206	White	Intact	8.2 mg/cm ²
2640	Negative	Classroom	Room	Ceiling	Drywall		206	White	Intact	0.3 mg/cm ²
2641	Negative	Classroom	Beam	Underside	Drywall		206	Beige	Intact	0.6 mg/cm ²
2642	Positive	Classroom	Window	Casing	Wood	A	206	White	Intact	18.2 mg/cm ²
2643	Negative	Classroom	Window	Sill	Wood	A	206	White	Intact	0.1 mg/cm ²
2644	Negative	Classroom	Window	Sash	Wood	A	206	White	Intact	0.2 mg/cm ²
2645	Positive	Classroom	Door	Casing	Wood	U	206	White	Intact	22.3 mg/cm ²
2646	Positive	Classroom	Door	Jamb	Wood	U	206	White	Intact	20.6 mg/cm ²
2647	Positive	Kitchen	Room	Wall	Drywall	A	208	Beige	Intact	7.9 mg/cm ²
2648	Positive	Kitchen	Room	Wall	Drywall	υ	208	Beige	Intact	5.5 mg/cm ²
2649	Positive	Kitchen	Room	Wall	Drywall	۵	208	Beige	Intact	8.7 mg/cm ²
2650	Positive	Kitchen	Room	Wall	Drywall	8	208	Beige	Intact	5.3 mg/cm ²
2651	Positive	Kitchen	Room	Wall	Porcelain	A	208	Beige	Intact	5.8 mg/cm ²
2652	Positive	Kitchen	Room	Wall	Porcelain	U	208	Beige	Intact	5.8 mg/cm ²
2653	Positive	Kitchen	Room	Wall	Porcelain	в	208	Beige	Intact	6.0 mg/cm ²
2654	Positive	Kitchen	Room	Wall	Porcelain	۵	208	Beige	Intact	6.9 mg/cm ²
2655	Positive	Kitchen	Room	Baseboard	Wood	A	208	White	Intact	11.1 mg/cm ²
2656	Positive	Kitchen	Room	Baseboard	Wood	U	208	White	Intact	13.5 mg/cm ²
2657	Positive	Kitchen	Room	Baseboard	Wood	в	208	White	Intact	13.5 mg/cm ²
2658	Positive	Kithen	Window	Casing	Wood	۵	208	White	Intact	13.1 mg/cm ²
2659	Positive	Kitchen	Door	Jamb	Wood	A	208	White	Intact	32 mg/cm ²
2660	Positive	Kitchen	Door	Buck	Wood	A	208	White	Intact	20.3 mg/cm ²

Inspection Site:

130 Licoln Street, Meriden, CT

1/31/2024 - 2/8/2024 481 Inspection Date: Total Readings: Action Level: Unit Started: Unit Ended:

01/31/2024 10:54:25 02/08/2024 14:40:07 1.0 (mg/cm²)

Lead (mg/cm²) 18.2 mg/cm² 10.5 mg/cm² 19.5 mg/cm² 15.8 mg/cm² 14.1 mg/cm² 13.4 mg/cm² 11.0 mg/cm² 19.4 mg/cm² 16.5 mg/cm² 14.6 mg/cm² 11.9 mg/cm² 14.8 mg/cm² 26.6 mg/cm² 11.8 mg/cm² 14.7 mg/cm² 9.3 mg/cm² 1.0 mg/cm² 1.2 mg/cm² 6.8 mg/cm² 0.3 mg/cm² 1.2 mg/cm² 0.4 mg/cm² 0.3 mg/cm² 0.1 mg/cm² 1.1 mg/cm² 9.0 mg/cm² Condtion Intact Color Yellow Yellow Yellow White Gray Gray Gray Gray Blue Gray Gray Gray Gray Gray Room ID 213 213 213 203 208 213 213 213 213 213 213 213 213 213 213 213 213 203 203 203 203 203 203 203 203 203 Wall ഫ ∢ U m ۵ υ Δ ۵ υ Δ υ υ U υ U U ഫ υ ∢ ∢ Substrate Drywall Drywall Drywall Drywall Drywall Drywall Drywall Drywall Drywall Mood Wood Wood Wood Mood Wood Wood Wood Wood Metal Wood Wood Wood Wood Vinyl Brick Vinyl -->Member **Crown Molding Crown Molding** Baseboard Baseboard Baseboard Baseboard Baseboard Baseboard Mantle Ceiling Casing Vertical Frame Ceiling Casing Ceiling Jamb Sill Wall Wall Wall Wall Wall Wall Wall Wall Structure Fire Place Fire Place Window Window Room Door Door Pipe -->RoomChoice Classroom Kitchen Result Negative Negative Negative Negative Positive Read # 2663 2669 2673 2675 2676 2677 2678 2679 2683 2685 2661 2664 2665 2666 2667 2668 2670 2671 2672 2674 2680 2681 2682 2684 2686 2662

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 13.4 mg/cm² 24.5 mg/cm² 11.8 mg/cm² 0.2 mg/cm² 1.2 mg/cm² 0.3 mg/cm² 0.1 mg/cm² 4.4 mg/cm² 3.0 mg/cm² 4.0 mg/cm² 4.5 mg/cm² 0.3 mg/cm² 2.0 mg/cm² 3.8 mg/cm² 0.3 mg/cm² 0.6 mg/cm² 5.9 mg/cm² 0.1 mg/cm² 1.5 mg/cm² 1.4 mg/cm² 4.4 mg/cm² 9.7 mg/cm² 3.5 mg/cm² 4.4 mg/cm² 1.4 mg/cm² 33 mg/cm² Condtion Intact Color Yellow Yellow Yellow Yellow White Gray Gray Gray Gray Gray Gray Gray Gray Room ID 204 204 204 204 **204** 204 204 207 207 207 207 207 207 207 207 209 209 209 209 209 209 211 211 211 211 211 Wall ∢ ۵ υ Δ ۵ Δ ഫ υ ഫ ∢ U ഫ ∢ υ ∢ U ш ∢ ഫ ⊲ Substrate Ceramic Ceramic Ceramic Ceramic Drywall Drywall Drywall Drywall Plaster Drywall Drywall Plaster Plaster Mood Plaster Wood Mood Wood Wood Wood Wood Wood Wood Wood Wood Vinyl -->Member Crown Molding Baseboard Baseboard Baseboard Baseboard Baseboard Baseboard Casing Casing Ceiling Ceiling Ceiling Jamb Floor Wall Sill Structure Window Window Room Door Door -->RoomChoice Hallway Hallway Hallway Hallway Hallway Room Room Room Room Room Room Toilet Toilet Toilet Room Room Toilet Result Negative Negative Negative Negative Negative Negative Negative Positive Read # 2710 2689 2693 2695 2699 2703 2705 2707 2708 2709 2711 2687 2688 2690 2691 2692 2694 2696 2697 2698 2700 2701 2702 2704 2706 2712

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 21.5 mg/cm² 14.2 mg/cm² 28.0 mg/cm² 15.3 mg/cm² 17.4 mg/cm² 28.3 mg/cm² 26.4 mg/cm² 22.6 mg/cm² 11.3 mg/cm² 22.1 mg/cm² 10.0 mg/cm² 0.2 mg/cm² 0.6 mg/cm² 0.6 mg/cm² 0.7 mg/cm² 0.6 mg/cm² 0.5 mg/cm² 1.6 mg/cm² 3.4 mg/cm² 3.0 mg/cm² 7.7 mg/cm² 32 mg/cm² 33 mg/cm² 31 mg/cm² 31 mg/cm² 32 mg/cm² Condtion Intact Color White Green Green Green Green Yellow Yellow Yellow White Beige Room ID 210 210 210 210 210 210 210 210 210 210 210 210 103 211 211 211 210 103 103 103 103 103 103 103 211 103 Wall Δ υ ш Δ Δ Δ U m O υ υ ш ۵ ഫ ∢ m Substrate Drywall Drywall Drywall Drywall Drywall Drywall Drywall Drywall Drywall Mood Mood Nood Mood Mood Wood Wood Nood Wood Nood Wood Wood Wood Wood Wood Wood Wood -->Member **Crown Molding** Baseboard **3aseboard** Baseboard Baseboard Casing Mantle Casing Ceiling Frame Casing Ceiling Frame Frame Door Jamb Door Wall Wall Wall Wall Wall Wall Wall Sill Structure Fire Place Fire Place Cabinets Window Window Column Closet Closet Closet Room Door Door Door -->RoomChoice Gathering Gathering Gathering Gathering Gathering Gathering Gtahering Gathering Gathering Hallway Hallway Hallway Hallway Office Result Negative Vegative Negative Negative Negative Negative Positive Read # 2719 2720 2713 2715 2716 2717 2728 2730 2736 2714 2718 2721 2722 2723 2724 2725 2726 2727 2729 2731 2732 2733 2734 2735 2737 2738

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 15.9 mg/cm² 24.7 mg/cm² 11.5 mg/cm² 19.6 mg/cm² 24.9 mg/cm² 26.7 mg/cm² 28.1 mg/cm² 11.5 mg/cm² 10.4 mg/cm² 1.5 mg/cm² 4.3 mg/cm² 0.2 mg/cm² 3.2 mg/cm² 5.1 mg/cm² 9.3 mg/cm² 0.5 mg/cm² 3.0 mg/cm² 1.0 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 5.4 mg/cm² 0.1 mg/cm² 1.1 mg/cm² 4.2 mg/cm² 4.7 mg/cm² Condtion Intact Color Brown Brown Brown Green Green White White White White Brown White Blue Blue Blue Blue <u></u> Lan Room ID 105 103 103 104 104 104 104 104 104 104 104 104 104 104 104 105 105 105 105 105 103 103 104 104 L04 105 Wall Δ ۵ Δ 4 U ഫ ഫ U U ∢ ∢ O ഫ ∢ C ഫ ∢ ∢ Substrate Drywall Drywall Drywall Drywall Drywall Drywall Plaster Mood Drywall Drywall Plaster Mood Mood Mood Wood Wood Wood Mood Wood Wood Wood Metal Wood Wood Wood Wood -->Member **Crown Molding** Baseboard Baseboard Baseboard Baseboard Mantle Mantle Casing Ceiling Ceiling Frame Cover Cover Jamb Buck Wall Sill Structure Fire Place Fire Place Fire Place Radiator Radiator Window Window Room Door Door -->RoomChoice Gathering Gathering Gathering Gathering Kitchen Kitchen Kitchen Kitchen Kitchen Kitchen Kitchen Office Result Negative Negative Negative Negative Negative Negative Positive Read # 2743 2747 2749 2755 2763 2739 2740 2741 2742 2744 2745 2746 2748 2750 2751 2752 2753 2754 2756 2757 2758 2759 2760 2761 2762 2764

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 30.0 mg/cm² 30.0 mg/cm² 16.9 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 1.2 mg/cm² 1.8 mg/cm² 0.4 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 9.1 mg/cm² 1.4 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.8 mg/cm² 0.1 mg/cm² 35 mg/cm² Condtion Intact Color White Green Green White White White White White White White White White Beige Beige Beige White White White Black Beige Gray Gray Gray Gray [an <u></u> Lan Room ID 105 105 102 102 105 105 105 106 106 106 106 106 106 104 104 104 L05 105 101 101 101 101 101 101 101 101 Wall ∢ Δ Δ υ <u>а</u> О ∢ ഫ U ∢ ∢ O ∢ υ ∢ Substrate Drywall Drywall Drywall Drywall Drywall Drywall Drywall Plaster Wood Mood Mood Wood Wood Wood Nood Wood Wood Wood Wood Wood Wood Wood Wood Wood Metal Brick -->Member Crown Molding Baseboard Baseboard Baseboard Baseboard Baseboard Casing Ceiling Casing Ceiling Casing Casing Apron Frame Ceiling Frame Wall Wall Wall Wall Wall Wall Wall Sill Structure Fire Place Radiator Window Window Window Window Room Door Room Room Door Door Door Room -->RoomChoice Vestibule Vestibule Vestibule Vestibule Vestibule Vestibule Vestibule Vestibule Kitchen Kitchen Kitchen Kitchen Kitchen Kitchen Kitchen Pantry Pantry Pantry Pantry Office Office Office Pantry Pantry Office Office Result Negative Vegative Positive Positive ositive Positive Positive Positive Positive Positive Read # 2773 2775 2779 2783 2785 2789 2765 2766 2767 2768 2769 2770 2771 2772 2774 2776 2777 2778 2780 2781 2782 2784 2786 2787 2788 2790

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Read #	Result	>RoomChoice	Structure	>Member	Substrate	Wall	Room ID	Color	Condtion	Lead (mg/cm²)
2791	Negative	Office	Window	Casing	Wood	A	102	White	Intact	0.0 mg/cm ²
2792	Negative	Office	Window	Sill	Wood	A	102	White	Intact	0.2 mg/cm ²
2793	Negative	Office	Door	Casing	Wood	В	102	White	Intact	0.1 mg/cm ²
2794	Negative	Office	Door	Jamb	Wood	в	102	White	Intact	0.1 mg/cm ²
2795	Negative	Office	Door	Buck	Wood	В	102	White	Intact	0.2 mg/cm ²
2796	Negative	Office	Room	Ceiling	Drywall		102	White	Intact	0.2 mg/cm ²
2797	Negative	Office	Room	Crown Molding	Wood		102	White	Intact	0.2 mg/cm ²
2798	Negative	Office	Room	Wall	Drywall	A	108	Green	Intact	0.2 mg/cm ²
2799	Negative	Office	Room	Wall	Drywall	U	108	Green	Intact	0.2 mg/cm ²
2800	Negative	Office	Room	Baseboard	Wood	۵	108	White	Intact	0.1 mg/cm ²
2801	Negative	Office	Room	Baseboard	Wood	U	108	White	Intact	0.2 mg/cm ²
2802	Negative	Office	Door	Casing	Wood	۵	108	White	Intact	0.1 mg/cm ²
2803	Negative	Office	Door	Jamb	Wood	D	108	White	Intact	0.1 mg/cm ²
2804	Negative	Office	Window	Sill	Wood	A	108	White	Intact	0.2 mg/cm ²
2805	Negative	Office	Window	Casing	Wood	A	108	White	Intact	0.2 mg/cm ²
2806	Negative	Office	Room	Ceiling	Plaster		108	White	Intact	0.2 mg/cm ²
2807	Negative	Office	Room	Crown Molding	Wood		108	White	Intact	0.0 mg/cm ²
2808	Negative	Conference	Room	Wall	Drywall	U	109	Green	Intact	0.2 mg/cm ²
2809	Positive	Conference	Room	Wall	Drywall	в	109	Green	Intact	14.8 mg/cm ²
2810	Positive	Conference	Room	Wall	Drywall	в	109	Green	Intact	13.6 mg/cm ²
2811	Positive	Conference	Room	Wall	Drywall	۵	109	Green	Intact	15.0 mg/cm ²
2812	Negative	Conference	Room	Baseboard	Wood	U	109	White	Intact	0.1 mg/cm ²
2813	Negative	Basement Mechanical	Room	Wall	Concrete	A	003	White	Deteriorated	0.2 mg/cm ²
2814	Negative	Basement Mechanical	Room	Wall	Concrete	U	003	White	Deteriorated	0.5 mg/cm ²
2815	Negative	Basement Mechanical	Room	Wall	Concrete	D	003	White	Deteriorated	0.7 mg/cm ²
2816	Negative	Basement Mechanical	Room	Wall	Brick		003	White	Deteriorated	0.6 mg/cm ²

Inspection Site:

130 Licoln Street, Meriden, CT

1/31/2024 - 2/8/2024 481 Inspection Date: Total Readings: Action Level: Unit Started: Unit Ended:

01/31/2024 10:54:25 02/08/2024 14:40:07 1.0 (mg/cm²)

Lead (mg/cm²) 11.6 mg/cm² 11.0 mg/cm² 0.3 mg/cm² 0.6 mg/cm² 0.5 mg/cm² 0.6 mg/cm² 0.5 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm^2 0.5 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.3 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.3 mg/cm² 0.5 mg/cm² 0.6 mg/cm² 0.2 mg/cm² 6.1 mg/cm² 0.5 mg/cm² 0.3 mg/cm² 0.2 mg/cm² 0.9 mg/cm² Condtion Deteriorated Color White Brown White Black Blue [an Room ID 002 002 002 004 004 004 004 003 203 001 001 001 001 002 002 002 002 002 004 004 004 004 004 201 01 204 Wall -->Member Substrate Concrete Concrete Concrete Concrete Drywall Plaster Brick Wood Metal Wood Metal Metal Metal Wood Metal Metal Wood Metal Metal Metal Brick Metal Metal Brick Brick Brick Horizontal Horizontal Horizontal Horizontal Ceiling Ceiling Vertical Railing Ceiling Ceiling Ceiling Vertical Vertical Ceiling Ceiling Ceiling Riser Wall Wall Wall Wall Wall Wall Wall Wall Wall Structure Room Pipe Pipe Pipe Pipe Pipe Pipe Stair Pipe Stair -->RoomChoice **Basement Mechanical** Basement Mechanical Basement Basement Basement Basement Basement Basement Storage Electrical Electrical Electrical Storage Electrical Electrical Electrical Electrical Electrical Electrical Electrical Storage Storage Storage Storage Storage Storage Result Negative Negative Vegative Negative Positive Positive Positive Read # 2818 2817 2819 2823 2825 2828 2829 2833 2835 2836 2837 2838 2839 2840 2841 2820 2821 2822 2824 2826 2827 2830 2831 2832 2834 2842

Inspection Site:

130 Licoln Street, Meriden, CT

1/31/2024 - 2/8/2024 1.0 (mg/cm²) 481 Inspection Date: Total Readings: Action Level: Unit Started: Unit Ended:

01/31/2024 10:54:25 02/08/2024 14:40:07

Lead (mg/cm²) 19.5 mg/cm² 0.9 mg/cm² 1.1 mg/cm² 1.2 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.3 mg/cm² 0.0 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 1.2 mg/cm² 1.1 mg/cm² 0.0 mg/cm² 1.2 mg/cm² Condtion Deteriorated Deteriorated Deteriorated Deteriorated Deteriorated Deteriorated Deteriorated Intact Room ID Color Brown Brown White White White White White White White Black Blue Blue Blue Blue Blue Blue Red Tan 307 307 307 307 307 307 307 307 307 307 307 307 307 307 307 307 307 307 Wall 4 4 4 4 4 ∢ 4 4 4 4 < ∢ ∢ ∢ 4 4 ∢ ∢ -->Member Substrate Concrete siding siding Wood Wood Nood Wood Wood Wood Wood Wood Wood Wood Wood Vinyl Vinyl Vinyl Vinyl **Exterior** Casing exterior casing Exterior Sash exterior well foundation Threshold Railing Treads Railing Casing Frame siding Risers Risers Floor Wall Jamb Sill Structure Foundation Window Window Window Window Window Porch Porch Porch Stair Door Door Door Door Stair Stair Wall Stair -->RoomChoice Building Porch Porch Result Negative Vegative Negative Positive Read # 2843 (CAL) 2844 (CAL) 2844 (CAL) 3065 (CAL) 3065 (CAL) 3066 (CAL) 3066 (CAL) 3067 (CAL) 3067 3068 3069 3070 3073 3075 3077 3081 3083 3071 3072 3074 3076 3078 3079 3080 3082 3084

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 17.1 mg/cm² 2.2 mg/cm² 0.3 mg/cm² 0.3 mg/cm² 0.0 mg/cm² 0.2 mg/cm² 0.1 mg/cm^2 0.2 mg/cm² 0.3 mg/cm² 0.0 mg/cm² 0.1 mg/cm² 0.4 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 7.2 mg/cm² 4.0 mg/cm² 0.2 mg/cm² 0.3 mg/cm² 7.2 mg/cm² 7.8 mg/cm² 0.1 mg/cm² 3.9 mg/cm² 0.5 mg/cm² 3.8 mg/cm² 9.5 mg/cm² 37 mg/cm² Condtion Deteriorated Deteriorated Deteriorated Deteriorated Deteriorated Intact Color Brown White Black Blue Tan Tan Red Red Red Room ID 308 308 308 308 308 308 308 307 307 307 307 307 307 307 307 307 307 307 307 307 307 307 308 307 808 Wall ∢ ∢ ∢ ⊲ ∢ ∢ ∢ ∢ ∢ ⊲ ∢ ∢ ∢ ∢ ∢ ∢ ∢ ∢ Substrate vinyl siding vinyl siding vinyl siding Concrete Concrete Wood Metal Wood Wood Wood Wood Vinyl Vinyl Vinyl Vinyl Vinyl Vinyl Brick Vinyl Vinyl Vinyl Vinyl Vinyl Vinyl Vinyl Vinyl -->Member exterior header **Exterior** Casing **Exterior** Casing exterior casing Exterior Sash Exterior Sash Exterior Sash Exterior Wall Vinyl Siding /inyl Siding Exterior Sill Exterior Sill Foundation foundation Column Railing Siding Casing Casing Casing Frame Frame Jamb Jamb Sill Sill Structure Foundation Foundation Window Porch Porch Wall Door Door Door Door Door Door Wall Wall Wall -->RoomChoice Building Result Negative Positive Positive ositive Positive Positive Positive Positive Positive Positive Positive Read # 3089 3093 3095 3099 3103 3105 3106 3107 3108 3109 3085 3086 3087 3088 3090 3091 3092 3094 3096 3097 3098 3100 3101 3102 3104 3110

Inspection Site:

130 Licoln Street, Meriden, CT

Inspection Date: Total Readings: Unit Started: Action Level: Unit Ended:

1/31/2024 - 2/8/2024 1.0 (mg/cm²) 481 01/31/2024 10:54:25 02/08/2024 14:40:07

						=			
Kead #	Kesult	>KoomChoice	structure	>Member	substrate W	all Koom ID	Color	Condition	Lead (mg/cm²)
3111	Negative	Building	Window	Exterior Sash	Vinyl	308	White	Intact	0.3 mg/cm ²
3112	Positive	Building	Wall	Vinyl Siding	vinyl siding	309	White	Intact	6.8 mg/cm ²
3113	Positive	Building	Wall	Vinyl Siding	vinyl siding	309	White	Intact	8.3 mg/cm ²
3114	Negative	Building	Foundation	Foundation	Concrete	309	White	Intact	0.3 mg/cm ²
3115	Positive	Building	Foundation	Foundation	Concrete	309	Red	Intact	8.9 mg/cm ²
3116	Negative	Building	Foundation	Foundation	Concrete	309	Red	Intact	0.7 mg/cm ²
3117	Negative	Building	Foundation	Foundation	Concrete	309	Red	Intact	0.9 mg/cm ²
3118	Positive	Building	Window	Exterior Sill	Vinyl	309	White	Intact	2.0 mg/cm ²
3119	Positive	Building	Window	Exterior Casing	Vinyl	309	White	Intact	22.3 mg/cm ²
3120	Negative	Building	Window	Exterior Sash	Vinyl	309	White	Intact	0.3 mg/cm ²
3121	Negative	Building	Porch	Floor	Wood	309	White	Deteriorated	0.3 mg/cm ²
3122	Positive	Building	Porch	Column	Wood	309	White	Deteriorated	36 mg/cm ²
3123	Positive	Building	Door	Frame	Wood	309	Red	Intact	19.0 mg/cm ²
3124	Positive	Building	Door	Jamb	Wood	309	White	Intact	10.2 mg/cm ²
3125	Negative	Building	Door	Threshold	Wood	309	Blue	Intact	0.3 mg/cm ²
3126	Negative	Building	Door	Frame	Metal	309	Red	Intact	0.0 mg/cm ²
3127	Negative	Building	Door	Outer Casing	Metal	309	White	Intact	0.5 mg/cm ²
3128	Negative	Building	Door	Jamb	Metal	309	White	Intact	0.5 mg/cm ²
3129	Negative	Building	Window	Sill	Metal	309	Black	Intact	0.3 mg/cm ²
3130	Negative	Building	Window	Casing	Metal	309	Black	Intact	0.3 mg/cm ²
3131	Negative	Building	Window	Jamb	Metal	309	Black	Intact	0.4 mg/cm ²
3132	Negative	Building	Door	Frame	Metal	309	Red	Intact	0.1 mg/cm ²
3133	Negative	Building	Exterior Wall		Brick	309	Red	Intact	0.3 mg/cm ²
3134	Negative	Building	Exterior Wall		Brick	310	Red	Intact	0.1 mg/cm ²
3135	Negative	Building	Window	Sill	Metal	310	Black	Intact	0.2 mg/cm ²
3136	Negative	Building	Window	Casing	Metal	310	Black	Intact	0.2 mg/cm ²

Inspection Site:

130 Licoln Street, Meriden, CT

1/31/2024 - 2/8/2024 481 Inspection Date: Total Readings: Action Level: Unit Started: Unit Ended:

01/31/2024 10:54:25 02/08/2024 14:40:07 1.0 (mg/cm²)

Lead (mg/cm²) 10.6 mg/cm² 21.9 mg/cm² 19.4 mg/cm² 18.3 mg/cm² 17.8 mg/cm² 10.6 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 9.9 mg/cm² 3.7 mg/cm² 1.1 mg/cm² 5.5 mg/cm² 9.1 mg/cm² 8.9 mg/cm² 0.3 mg/cm² 9.6 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.7 mg/cm² 9.9 mg/cm² 0.4 mg/cm² 4.1 mg/cm² 0.1 mg/cm² 32 mg/cm² 35 mg/cm² Condtion Intact Color Brown Green Green Green White White White Green White Black Gray Gray Blue Blue Tan Room ID 310 110 310 310 110 110 110 110 110 110 307 308 309 309 107 107 107 107 107 107 107 107 107 107 110 Wall Δ ш ш Δ Δ ഫ ш U U U ш ш ∢ U ۵ m Substrate Concrete Drywall Drywall Drywall Drywall Plaster Plaster Wood Wood Wood Nood Metal Wood Metal Wood Vinyl -->Member **Crown Molding** Baseboard Baseboard Baseboard Baseboard **Fhreshold** Casing Railing Casing Ceiling Ceiling Frame Casing Frame Frame Wall Jamb Jamb Jamb Sash Wall Wall Wall Wall Wall Structure Foundation Window Room Porch Room Room Room Porch Porch Porch Room Room Room Room Room Room Door -->RoomChoice Building Result Negative Negative Vegative Negative Negative Negative Vegative Negative Negative Positive Read # 3143 3145 3139 3141 3146 3147 3149 3150 3153 3154 3155 3156 3157 3158 3159 3160 3137 3138 3140 3142 3144 3148 3151 3152 3161 3162

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Read #	Result	>RoomChoice	Structure	>Member	Substrate	Wall	Room ID	Color	Condtion	Lead (mg/cm ²)
3163	Positive	Building	Fire Place	Mantle	Wood	U	110	White	Intact	12.7 mg/cm ²
3164	Positive	Building	Fire Place	Frame	Mood	υ	110	White	Intact	4.5 mg/cm ²
3165	Negative	Building	Room	Ceiling	Plaster		110	White	Intact	0.2 mg/cm ²
3166	Positive	Building	Room	Wall	Plaster	A	112	Purple	Intact	6.5 mg/cm ²
3167	Positive	Building	Room	Wall	Plaster	U	112	Purple	Intact	5.6 mg/cm ²
3168	Negative	Building	Room	Wall	Plaster	8	112	Purple	Intact	0.2 mg/cm ²
3169	Positive	Building	Room	Wall	Plaster	۵	112	Purple	Intact	6.0 mg/cm ²
3170	Positive	Building	Door	Jamb	Wood	۵	112	White	Intact	14.0 mg/cm ²
3171	Positive	Building	Door	Casing	Wood	۵	112	White	Intact	24.5 mg/cm ²
3172	Positive	Building	Door	Frame	Wood	۵	112	White	Intact	21.7 mg/cm ²
3173	Negative	Building	Door	Frame	Wood	U	112	Blue	Intact	0.1 mg/cm ²
3174	Negative	Building	Door	Jamb	Wood	U	112	Blue	Intact	0.2 mg/cm ²
3175	Negative	Building	Door	Casing	Wood	U	112	Blue	Intact	0.1 mg/cm ²
3176	Negative	Building	Room	Baseboard	Wood	U	112	White	Intact	0.0 mg/cm ²
3177	Positive	Building	Room	Wall	Plaster	A	111	Purple	Intact	3.2 mg/cm ²
3178	Positive	Building	Room	Wall	Plaster	U	111	Purple	Intact	6.0 mg/cm ²
3179	Positive	Building	Room	Wall	Plaster	Ю	111	Purple	Intact	4.4 mg/cm ²
3180	Negative	Building	Room	Wall	Plaster	۵	111	Purple	Intact	0.4 mg/cm ²
3181	Positive	Building	Room	Baseboard	Wood	A	111	White	Intact	11.6 mg/cm ²
3182	Positive	Building	Pipe	Vertical	Metal	A	111	White	Intact	7.8 mg/cm ²
3183	Positive	Building	Fire Place	Mantle	Wood	A	111	White	Intact	12.6 mg/cm ²
3184	Positive	Building	Fire Place	Frame	Wood	A	111	White	Intact	11.8 mg/cm ²
3185	Positive	Building	Door	Frame	Wood	A	111	White	Intact	11.0 mg/cm ²
3186	Positive	Building	Door	Casing	Wood	A	111	White	Intact	14.9 mg/cm ²
3187	Positive	Building	Door	Jamb	Wood	A	111	White	Intact	17.5 mg/cm ²
3188	Positive	Building	Window	Casing	Wood	υ	111	White	Intact	11.4 mg/cm ²

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Lead (mg/cm²) 0.6 mg/cm² 0.7 mg/cm² 8.0 mg/cm² 0.1 mg/cm² 0.9 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.3 mg/cm² 0.3 mg/cm² 0.1 mg/cm² 0.1 mg/cm² 0.3 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 0.1 mg/cm² 0.9 mg/cm² 0.1 mg/cm² 0.3 mg/cm² 0.1 mg/cm² 0.3 mg/cm² 0.2 mg/cm² 0.2 mg/cm² 1.9 mg/cm² 1.4 mg/cm² 31 mg/cm² 32 mg/cm² Condtion Deteriorated Intact Color White White White White White White White White White Black Black Black Black Black Black Gray Blue Blue Blue Blue Blue Blue Blue Blue Pink Pink Room ID 113 113 113 113 114 114 114 112 112 112 112 112 H 111 111 11 113 113 114 114 114 114 114 114 112 112 Wall ပ υ U υ υ Δ Δ ∢ υ ഫ ш ш ۵ ഫ ∢ Substrate Concrete Concrete Concrete Concrete Concrete Concrete Ceramic Ceramic Drywall Plaster Plaster Plaster Mood Mood Mood Mood Mood Wood Metal Metal Metal Metal Metal Wood Metal Vinyl -->Member Baseboard Ceiling Vertical Ceiling Casing Ceiling Casing Stop Frame Sill Jamb Wall Jamb Floor Floor Jamb Well Stop Wall Wall Wall Wall Wall Sill Sill Structure Window Radiator Window Room Room Room Room Room Room Room Room Pipe Room Room Room Room Door -->RoomChoice Building Building Building Buildng Building Result Vegative Negative Positive Positive Positive Positive Positive Read # 3210 3193 3195 3197 3199 3203 3205 3206 3207 3209 3211 3212 3213 3214 3189 3190 3191 3192 3194 3196 3198 3200 3201 3202 3204 3208

Inspection Site:

130 Licoln Street, Meriden, CT

 Inspection Date:
 1/31/2024 - 2/8/2024

 Action Level:
 1.0 (mg/cm²)

 Total Readings:
 481

 Unit Started:
 01/31/2024 10:54:25

 Unit Ended:
 02/08/2024 14:40:07

Read #	Result	>RoomChoice	Structure	>Member	Substrate	Wall	Room ID	Color	Condtion	Lead (mg/cm ²)
3215	Negative	Building	Door	Jamb	Wood	۵	112	White	Intact	0.1 mg/cm ²
3216	Negative	Building	Room	Wall	Concrete	A	121	White	Intact	0.3 mg/cm ²
3217	Negative	Building	Room	Wall	Concrete	D	121	White	Intact	0.3 mg/cm ²
3218	Negative	Building	Window	Sill	Metal	В	121	Black	Intact	0.1 mg/cm ²
3219	Negative	Building	Window	Jamb	Metal	В	121	Black	Intact	0.1 mg/cm ²
3220	Negative	Building	Room	Wall	Plaster	A	120	Yellow	Intact	0.3 mg/cm ²
3221	Negative	Building	Room	Wall	Plaster	U	120	Yellow	Intact	0.3 mg/cm ²
3222	Negative	Building	Radiator		Metal	A	120	Yellow	Intact	0.1 mg/cm ²
3223	Negative	Building	Room	Baseboard	Vinyl	A	120	Blue	Intact	0.3 mg/cm ²
3224	Negative	Bulding	Vent		Metal		120	Yellow	Intact	0.0 mg/cm ²
3225	Positive	Building	Window	Sill	Wood	A	120	White	Intact	6.9 mg/cm ²
3226	Negative	Building	Window	Jamb	Wood	A	120	White	Intact	0.3 mg/cm ²
3227	Negative	Building	Window	Sash	Wood	A	120	White	Intact	0.2 mg/cm ²
3228	Negative	Building	Window	Sash	Drywall	U	119	Pink	Intact	0.2 mg/cm ²
3229	Negative	Building	Window	Sash	Drywall	В	119	Pink	Intact	0.3 mg/cm ²
3230	Negative	Building	Room	Wall	Drywall	ß	119	Pink	Intact	0.2 mg/cm ²
3231	Negative	Building	Room	Wall	Drywall	U	119	Pink	Intact	0.2 mg/cm ²
3232	Negative	Building	Room	Baseboard	Wood	A	119	White	Intact	0.2 mg/cm ²
3233	Positive	Building	Room	Floor	Ceramic	A	119	Blue	Intact	1.6 mg/cm ²
3234	Positive	Building	Room	Floor	Ceramic	A	119	Blue	Intact	1.5 mg/cm ²
3235	Negative	Building	Radiator		Metal	D	119	White	Intact	0.1 mg/cm ²
3236	Negative	Building	A/C		Metal		115	Tan	Intact	0.7 mg/cm ²
3237	Negative	Building	Room	Wall	Plaster	D	115	Blue	Intact	0.8 mg/cm ²
3238	Positive	Building	Room	Wall	Plaster	U	115	Blue	Intact	1.1 mg/cm ²
3239	Negative	Building	Room	Ceiling	Wood		115	Brown	Intact	0.1 mg/cm ²
3240	Positive	Building	Window	Sill	Wood	υ	115	Blue	Intact	5.4 mg/cm ²

Inspection Site:

130 Licoln Street, Meriden, CT

1/31/2024 - 2/8/2024 1.0 (mg/cm²) 481 Inspection Date: Total Readings: Action Level: Unit Started: Unit Ended:

01/31/2024 10:54:25 02/08/2024 14:40:07

Read #	Result	>RoomChoice	Structure	>Member	Substrate	Wall	Room ID	Color	Condtion	Lead (mg/cm ²)
3241	Negative	Building	Window	Jamb	Wood	U	115	Blue	Intact	0.8 mg/cm ²
3242	Positive	Building	Door	Frame	Metal	ß	115	Brown	Intact	6.0 mg/cm ²
3243	Positive	Building	Door	Jamb	Metal	в	115	White	Intact	6.4 mg/cm ²
3244	Positive	Building	Door	Jamb	Metal	B	115	Blue	Intact	4.8 mg/cm ²
3245	Negative	Building	Room	Wall	Drywall	A	116	Gray	Intact	0.1 mg/cm ²
3246	Negative	Building	Window	Casing	Wood	A	116	White	Intact	0.1 mg/cm ²
3247	Negative	Building	Window	Jamb	Wood	A	116	White	Intact	0.2 mg/cm ²
3248	Negative	Building	Room	Baseboard	Wood	A	116	White	Intact	0.0 mg/cm ²
3249	Negative	Building	Window	Jamb	Wood	A	116	Brown	Intact	0.5 mg/cm ²
3250	Negative	Building	Window	Jamb	Wood	A	206	Brown	Intact	0.5 mg/cm ²
3251	Negative	Building	Window	Jamb	Wood	A	203	Brown	Intact	0.6 mg/cm ²
3252	Negative	Building	Window	Jamb	Wood	A	210	Brown	Intact	0.4 mg/cm ²
3253	Negative	Building	Window	Jamb	Wood	A	206	Brown	Intact	0.5 mg/cm ²
3254 (CAL)										0.9 mg/cm ²
3255 (CAL)										1.0 mg/cm ²
3256 (CAL)										0.8 mg/cm ²

----- END OF READINGS ------
SECTION 03 3000 - CAST-IN-PLACE CONCRETE

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 1 Specification Sections, apply to this Section.

1.2 SUMMARY

A. This Section includes the following

- 1. Cast-in-place concrete including formwork, reinforcement, concrete materials, mix designs, placement procedures, and finishes.
- 2. Implementation of cold weather and hot weather protection measures and all costs related thereto.
- B. Related Sections include the following:
 - 1. Division 31 Section "Earthwork"

1.3 REFERENCES

- American Association of State Highway and Transportation Officials (AASHTO): AASHTO M182, Specification for Burlap Cloth Made from Jute or Kenaf.
- B. American Concrete Institute (ACI): ACI 117, Specifications for Tolerances for Concrete Construction and Materials. ACI 301, Specification for Structural Concrete. ACI 302.1R, Guide for Concrete Floor and Slab Construction. ACI 304R, Guide for Measuring, Mixing, Transporting, and Placing Concrete. ACI 305R, Hot Weather Concreting. ACI 306.1, Specification for Cold Weather Concreting. ACI 318, Building Code Requirements for Structural Concrete. ACI 347R, Guide to Formwork for Concrete.
- C. American Society for Testing and Materials (ASTM): ASTM A82, Specification for Steel Wire, Plain, for Concrete Reinforcement. ASTM A185, Specification for Steel Welded Wire Fabric, Plain, for Concrete Reinforcement. ASTM A497, Specification for Steel Welded Wire Fabric, Deformed, for
 - ASTM A497, Specification for Steel Welded Wire Fabric, Deformed, for Concrete Reinforcement.

- ASTM A615, Specification for Deformed and Plain Billet-Steel Bars for Concrete Reinforcement.
- ASTM A706, Specification for Low-Alloy Steel Deformed and Plain Bars for Concrete Reinforcement.
- ASTM A775, Specification for Epoxy-Coated Steel Reinforcing Bars
- ASTMA884, Specification for Epoxy-Coated Steel Wire and Welded Wire Reinforcement
- ASTM C33, Specification for Concrete Aggregates.
- ASTM C39, Test Method for Compressive Strength of Cylindrical Concrete Specimens.
- ASTM C42, Test Method for Obtaining and Testing Drilled Cores and Sawed Beams of Concrete.
- ASTM C94, Specification for Ready-Mixed Concrete.
- ASTM C109, Test Method for Compressive Strength of Hydraulic Cement Mortars (Using 2-in. or Cube Specimens).
- ASTM C150, Specification for Portland Cement.
- ASTM C171, Specification for Sheet Materials for Curing Concrete.
- ASTM C219, Terminology Relating to Hydraulic Cement.
- ASTM C260, Specification for Air-Entraining Admixtures for Concrete.
- ASTM C309, Specification for Liquid Membrane-Forming Compounds for Curing Concrete.
- ASTM C330, Specification for Lightweight Aggregates for Structural Concrete.
- ASTM C494, Specification for Chemical Admixtures for Concrete.
- ASTM C567, Test Method for Density of Structural Lightweight Concrete.
- ASTM C618, Specification for Coal Fly Ash and Raw or Calcined Natural Pozzolan for Use as a Mineral Admixture in Concrete.
- ASTM C881, Specification for Epoxy-Resin-Base Bonding Systems for Concrete.
- ASTM C979, Specification for Pigments for Integrally Colored Concrete.
- ASTM C989, Specification for Ground Granulated Blast-Furnace Slag for Use in Concrete and Mortars.
- ASTM C1059, Specification for Latex Agents for Bonding Fresh to Hardened Concrete.
- ASTM C1077, Practice for Laboratories Testing Concrete and Concrete Aggregates for Use in Construction and Criteria for Laboratory Evaluation.
- ASTM C1116, Specification for Fiber-Reinforced Concrete and Shotcrete.
- ASTM C1315, Specification for Liquid Membrane-Forming Compounds Having Special Properties for Curing and Sealing Concrete.
- ASTM D1751, Specification for Preformed Expansion Joint Filler for Concrete Paving and Structural Construction (Nonextruding and Resilient Bituminous Type).
- ASTM D1752, Specification for Preformed Sponge Rubber and Cork Expansion Joint Fillers for Concrete Paving and Structural Construction.
- ASTM D2240, Test Method for Rubber Property Durometer Hardness.
- ASTM E96, Test Methods for Water Vapor Transmission of Materials.
- ASTM E154, Test Methods for Water Vapor Retarders used in Contact with Earth under Concrete Slabs, on Walls, or as Ground Cover.
- ASTM E329, Specification for Agencies Engaged in the Testing and/or Inspection of Materials Used in Construction.
- ASTM E548, Guide for General Criteria Used for Evaluating Laboratory Competence.
- ASTM E1643, Practice for Installation of Water Vapor Retarders Used in Contact with Earth and Granular Fill under Concrete Slabs.

ASTM E1745, Specification for Plastic Water Vapor Retarders Used in Contact with Soil or Granular Fill under Concrete Slabs.

- D. Concrete Reinforcing Steel Institute (CRSI): Manual of Standard Practice, current edition.
- E. Corps of Engineers (CE): CE CRD-C 513, Corps of Engineers Specifications for Rubber Waterstops. CE CRD-C 572, Corps of Engineers Specifications for Polyvinylchloride Waterstops.
- 1.4 DEFINITIONS
 - A. Cementitious Materials: Portland cement alone or in combination with one or more of the following: blended hydraulic cement, fly ash and other pozzolans, ground granular blast-furnace slag, and silica fume; subject to compliance with requirements.
- 1.5 SUBMITTALS
 - A. Product Data: For each manufactured material and product indicated.
 - B. Design Mixes: Submit separate concrete mix designs for each strength class of normal weight and lightweight concrete and for pumped concrete.
 - C. Reinforcing Steel Shop Drawings: Reinforcement shop drawings showing fabrication, bending, and placement of steel reinforcing bars. Indicate reference to elevations, column and wall grid lines, and other pertinent details of the contract documents. Wall reinforcement shall be shown on wall elevations drawn to scale. Fabrication shall not commence prior to receipt of engineer's comments on reviewed drawings.
 - D. Formwork Shop Drawings: Prepared by or under supervision of a qualified professional engineer, licensed in the State of Connecticut, detailing fabrication, assembly and support for formwork. Shoring and Reshoring: Indicate proposed schedule and sequence of stripping formwork, shoring removal, and installing and removing reshoring.
 - E. Samples: Submit for waterstops, vapor retarder, vapor barrier.
 - F. Certificates: Manufacturer's certification that the materials governed by this Section including cement, grout, admixtures, reinforcing steel and accessories shall meet the requirements of this specification.
 - G. Floor surface flatness and levelness measurements to determine compliance with specified tolerances.
 - H. Submit cold weather protection plan no less than fourteen (14) days prior to commencement of concrete work.
 - I. Submit hot weather protection plan no less than fourteen (14) days prior to commencement of concrete work.
 - J. Submit minimum 1/8" scale dimensioned plans showing proposed contraction joint

locations for concrete slabs-on-grade per Section 03 3000, 3.7, D.

- 1.6 QUALITY ASSURANCE
 - A. The work of this Section shall conform to the 2018 CT State Building Code, which shall include the 2015 International Building Code and 2018 Connecticut Amendment.
 - B. The standards of the American Society for Testing and Material and the American Concrete Institute, referred to in these specifications by their serial designation and declared to be a part of these specifications, the same as if fully set forth herein.
 - C. Manufacturer Qualifications: A firm experienced in manufacturing ready-mixed concrete products complying with ASTM C94 requirements for production facilities and equipment.
 - D. Source Limitations: Obtain each type or class of cementitious material of the same brand from same manufacturer's plant, obtain aggregate from one (1) source, and obtain admixtures through one (1) source from a single manufacturer.
 - E. The work of this Section shall conform to all requirements of ACI 301, "Specifications for Structural Concrete for Buildings". Although this Section states in condensed form, the pertinent provisions of that document, it shall be understood that it has been adopted in its entirety and may be referred to throughout the project for provisions not restated below.
 - F. Concrete floor and slab construction shall conform to the recommendations of ACI 302.IR, "Guide for Concrete Floor and Slab Construction".
 - G. Concrete Testing Service: Engage a qualified independent testing agency to perform material evaluation tests and to design concrete mixtures.
 - H. Pre-installation Conference: Conduct conference at Project site.
- 1.7 DELIVERY, STORAGE AND HANDLING
 - A. Steel Reinforcement: Deliver, store, and handle steel reinforcement to prevent bending and damage.

PART 2 - PRODUCTS

2.1 FORM-FACING MATERIALS

- A. Smooth-Formed Finished Concrete: Form-facing panels that will provide continuous, true, and smooth concrete surfaces. Furnish in largest practicable sizes to minimize number of joints. Plywood, metal, or other approved panel materials.
- B. Rough-Formed Finished Concrete: Plywood, lumber, metal, or another approved material. Provide lumber dressed on at least two (2) edges and one (1) side for tight fit.

- C. Forms for Cylindrical Columns, Pedestals, and Supports: Metal, glass-fiber-reinforced plastic, paper, or fiber tubes that will produce surfaces with gradual or abrupt irregularities not exceeding specified formwork surface class. Provide units with sufficient wall thickness to resist plastic concrete loads without detrimental deformation.
- D. Void Forms: Biodegradable paper surface, treated for moisture resistance, structurally sufficient to support weight of plastic concrete and other superimposed loads.
- E. Chamfer Strips: Wood, metal, PVC, or rubber strips, 3/4 by 3/4 inch, minimum.
- F. Rustication Strips: Wood, metal, PVC, or rubber strips, kerfed for ease of form removal.
- G. Form-Release Agent: Commercially formulated form-release agent that will not bond with, stain, or adversely affect concrete surfaces and will not impair subsequent treatments of concrete surfaces. Formulate form-release agent with rust inhibitor for steel form-facing materials.

H. Form Ties: Factory-fabricated, removable or snap-off metal or glass-fiber-reinforced plastic form ties designed to resist lateral pressure of fresh concrete on forms and to prevent spalling of concrete on removal.
Furnish units that will leave no corrodible metal closer than one (1) inch to the plane of exposed concrete surface.
Furnish ties that, when removed, will leave holes no larger than one (1) inch in diameter in concrete surface.
Furnish ties with integral water-barrier plates to walls indicated to receive dampproofing or waterproofing.

- 2.2 STEEL REINFORCEMENT
 - A. Reinforcing Bars: ASTM A615, Grade 60, deformed.
 - B. Plain-Steel Wire: ASTM A82, as drawn.
 - C. Plain-Steel Welded Wire Fabric: ASTM A185, fabricated from as-drawn steel wire into flat sheets.
- 2.3 REINFORCEMENT ACCESSORIES
 - A. Joint Dowel Bars: ASTM A615, Grade 60, plain-steel bars, cut true to length with ends square and free of burrs.
 - B. Bar Supports: Bolsters, chairs, spaces, and other devices for spacing, supporting, and fastening reinforcing bars and welded wire reinforcement in place. Manufacture bar supports from steel wire, plastic, or precast concrete in accordance with CRSI's "Manual of Standard Practice," of greater compressive strength than concrete and as follows:

For concrete surfaces exposed to view where legs of wire bar supports contact forms,

use CRSI Class 1 plastic-protected steel wire or CRSI Class 2 stainless steel bar supports.

- 2.4 CONCRETE MATERIALS
 - A. Portland Cement: ASTM C150, Type I or Type II with the following supplementary materials.
 - B. Ground Granulated Blast-Furnace Slag: ASTM C989, Grade 100 or 120.
 - C. Normal-Weight Aggregate: ASTM C33, uniformly graded from a single source, not exceeding 1-1/2-inch nominal size. Refer to ACI Standard for other limitations on aggregate size.
 - D. Water: Complying with ASTM C94.
- 2.5 ADMIXTURES
 - A. General: The maximum water-soluble chloride ion concentrations in hardened concrete contributed from all sources including water, aggregates, cementitious materials, and admixtures shall not exceed the following limits:

0.15 percent (0.0015) for reinforced concrete <u>exposed</u> to moisture in service. 1.0 percent (0.01) for reinforced concrete <u>not exposed</u> to moisture in service.

- B. Air-Entraining Admixture: ASTM C260.
- C. Water-Reducing Admixture: ASTM C494, Type A.
- D. High-Range, Water-Reducing Admixture: ASTM C494, Type F.
- E. Water-Reducing and Accelerating Admixture: ASTM C494, Type E.
- F. Water-Reducing and Retarding Admixture: ASTM C494, Type D.
- G. Concrete moisture vapor reduction admixture for <u>all</u> interior suspended slab and slab on grade construction shall be Barrier One, Inc., 522 S. Hunt Club Blvd., #303, Apopka Florida 32703. Contact Mmanufacturer's representative: 877.224.5850, Fax: 866.594.3490. Concrete mix design shall contain 14 oz. of Barrier-1 admixture per 100 pounds of cementitious material or as otherwise recommended by manufacturer.

2.6 WATERSTOPS

A. Flexible PVC Waterstops: CE CRD-C 572, for embedding in concrete to prevent passage of fluids through joints. Factory fabricate corners, intersections, and direction changes.

Manufacturers:

- a. Greenstreak.
- b. W. R. Meadows, Inc.

Profile and Dimensions: As indicated on drawings.

2.7 VAPOR BARRIER

A. Plastic Vapor Barrier: A flexible, preformed sheet membrane having a water-vapor permeance rate no greater than 0.012 perms when tested in accordance with ASTM E154, Section 7 and ASTM E96, and otherwise conforming to ASTM E1745, Class B or higher. Vapor barrier shall be no less than 10 mils thick in accordance with ACI 302.1R.

Manufacturers:

- a. Stego Industries LLC; Stego Wrap, 15-mils.
- b. W.R. Meadows, Inc.; Premoulded membrane with PLASMATIC CORE.
- c. Alumiseal; Zero-Perm.

2.8 LIQUID FLOOR TREATMENTS

A. Penetrating Liquid Floor Treatment: Clear, chemically reactive, waterborne solution of inorganic silicate or siliconate materials and proprietary components; odorless; colorless; that penetrates, hardens, and densifies concrete surfaces.

Products:

- a. Dayton Superior Corporation; Day-Chem Sure Hard.
- b. L&M Construction Chemicals, Inc.; Seal Hard.
- c. Nox-Crete Products Group, Kinsman Corporation; Duranox.

2.9 CURING MATERIALS

A. Evaporation Retarder: Waterborne, monomolecular film forming, manufactured for application to fresh concrete.

Products:

- a. Euclid Chemical Company (The); Eucobar.
- b. Lambert Corporation; Lambco Skin.
- c. W.R. Meadows, Inc.; Sealtight Evapre.
- B. Absorptive Cover: AASHTO M182, Class 2, burlap cloth made from jute or kenaf.
- C. Moisture-Retaining Cover: ASTM C171, polyethylene film or white burlap-polyethylene sheet.
- D. Water: Potable.
- E. Clear, Waterborne, Membrane-Forming Curing Compound: ASTM C309, Type 1, Class B, non-dissipating.

Products:

- a. Anti-Hydro International, Inc.; AH Clear Cure WB.
- b. Euclid Chemical Company (The); Aqua Cure VOX.

- c. W.R. Meadows, Inc.; Vocomp-20.
- F. Clear, Solvent-Borne, Membrane-Forming Curing and Sealing Compound: ASTM C1315, Type 1, Class A.

Products:

- a. Euclid Chemical Company (The); Super Diamond Clear.
- b. W.R. Meadows, Inc.; CS-309/30.
- c. Sonneborn, Div. of ChemRex; Kure-N-Seal 5.

2.10 RELATED MATERIALS

- A. Expansion and Isolation Joint-Filler Strips: ASTM D1751, Nonextruding and resilient asphalt-saturated cellulosic fiber.
- B. Bonding Agent: ASTM C1059, Type II, non-redispersible, acrylic emulsion or styrene butadiene.
- C. Epoxy Bonding Agent: ASTM C881, two-component epoxy resin, capable of humid curing and bonding to damp surfaces, of class suitable for application temperature and of grade to suit requirements, and as follows:

Types I and II, non-load bearing and Types IV and V, load bearing, for bonding hardened or freshly mixed concrete to hardened concrete.

D. Dovetail Anchor Slots: Hot-dip galvanized steel sheet, not less than 0.0336-inch thick, with bent tab anchors. Temporarily fill or cover face opening of slots to prevent intrusion of concrete or debris.

2.11 CONCRETE MIXES

A. Comply with ACI 301 requirements for concrete mixtures.

B. Slump Adjustment

Adding water to concrete mix in transit or at the construction site will not be permitted without prior approval of the Engineer. Increased slump for pumped concrete shall be made with admixtures (not water) in conformance with approved design mix.

- C. Special Requirements for Concrete Durability Concrete that will be subject to the following exposures in service shall conform to the corresponding water-cementitious materials ratio and minimum specified concrete compressive strength at 28 days.
 - a. Concrete exposed to freezing and thawing in a moist condition or to deicing chemicals maximum water-cementitious materials ratio = 0.50 and minimum compressive strength = 4,000 psi at 28 days.
 - b. Concrete intended to have low permeability when exposed to water maximum water-cementitious materials ratio = 0.50 and minimum compressive strength = 4,000 psi at 28 days.

D. Prepare design mixes, proportioned according to ACI 301, for concrete determined by either laboratory trial mix or field test data bases, as follows:
 Compressive Strength (28 Days): 4,000 psi normal weight for all slabs on grade.
 Compressive Strength (28 Days): 3,000 psi normal weight for all footings and walls.

Slump:

- a. Four (4) inches for slabs.
- b. Five (5) inches for all other concrete.
- c. Slump Limit for Concrete Containing High-Range Water-Reducing Admixture: Not more than eight (8) inches after adding admixture to plantor site-verified, 2 to 3 inch slump.
- E. Add air-entraining admixture at manufacturer's prescribed rate to result in concrete at point of placement having an air content of 6.0 percent within a tolerance of plus 1.0 or minus 1.5 percent. Air content of trowel-finished interior concrete floors shall not exceed 3.0 percent. All other interior and exterior concrete shall have the aforementioned air content percent.
- 2.12 FABRICATING REINFORCEMENT
 - A. Fabricate steel reinforcement in accordance with CRSI's "Manual of Standard Practice."
- 2.13 CONCRETE MIXING
 - A. Ready-Mixed Concrete: Comply with ASTM C94. When air temperature is between 85 and 90 degrees F, reduce mixing and delivery time from 1-1/2 hours to 75 minutes. When air temperature is above 90 degrees F, reduce mixing and delivery time to 60 minutes.
 - B. Provide batch ticket for each batch discharged and used in the Work, indicating project identification name and number, date, mix type, mix time, quantity, and amount of water added. Record approximate location of final deposit in structure.

PART 3 - EXECUTION

3.1 FORMWORK

- A. Design, construct, erect, shore, brace, and maintain formwork according to ACI 301, to support vertical, lateral, static, and dynamic loads, and construction loads that might be applied, until structure can support such loads.
- B. Construct formwork so concrete members and structures are of size, shape, alignment, elevation, and position indicated, within tolerance limits of ACI 117.
- C. Limit concrete surface irregularities, designated by ACI 347R as abrupt or gradual, as follows:

Class A, 1/8-inch for smooth-formed finished surfaces. Class B, 1/4-inch for rough-formed finished surfaces. Class C, 1/2-inch for concealed concrete.

- D. Construct forms tight enough to prevent loss of concrete mortar.
- E. Fabricate forms for easy removal without hammering or prying against concrete surfaces. Provide crush or wrecking plates where stripping may damage cast concrete surfaces. Provide top forms for inclined surfaces steeper than 1.5 horizontal to 1 vertical.
- F. Set edge forms, bulkheads, and intermediate screed strips for slabs to achieve required elevations and slopes in finished concrete surfaces. Provide and secure units to support screed strips; use strike-off templates or compacting-type screeds.
- G. Provide temporary openings for cleanouts and inspection ports where interior area or formwork is inaccessible. Close openings with panels tightly fitted to forms and securely braced to prevent loss of concrete mortar. Locate temporary openings in forms at inconspicuous locations.
- H. Chamfer exterior corners and edges of permanently exposed concrete.
- I. Form openings, chases, offsets, sinkages, keyways, blocking, screeds, and bulkheads required in the Work. Determine sizes and locations from trades providing such items.
- J. Clean forms and adjacent surfaces to receive concrete. Remove chips, wood, sawdust, dirt, and other debris just before placing concrete.
- K. Retighten forms and bracing before placing concrete, as required, to prevent mortar leaks and maintain proper alignment.
- L. Coat contact surfaces of forms with form-release agent, according to manufacturer's written instructions, before placing reinforcement.
- M. All forms used in exposed finish shall be new.

3.2 EMBEDDED ITEMS

A. Place and secure anchorage devices and other embedded items required for adjoining work that is attached to or supported by cast-in-place concrete. Use setting drawings, templates, diagrams, instructions, and directions furnished with items to be embedded.

3.3 REMOVING AND REUSING FORMS

A. General: Formwork for sides of beams, walls, columns, and similar parts of the Work that does not support weight of concrete may be removed after cumulatively curing at not less than 50 degrees F for 24 hours after placing concrete, if concrete is hard enough to not be damaged by form-removal operations and curing and protection operations are maintained.

- B. Clean and repair surfaces of forms to be reused in the Work. Split, frayed, delaminated, or otherwise damaged form-facing material will not be acceptable for exposed surfaces. Apply new form-release agent.
- C. When forms are reused, clean surfaces, remove fins and laitance, and tighten to close joints. Align and secure joints to avoid offsets. Do not use patched forms for exposed concrete surfaces unless approved by Architect.
- 3.4 SHORES AND RESHORES
 - A. Comply with ACI 318 and ACI 301 for design, installation, and removal of shoring and reshoring.
 - B. Plan sequence of removal of shores and reshore to avoid damage to concrete.

3.5 VAPOR BARRIERS

- A. Vapor Barrier: Install, protect, and repair vapor-barrier sheets according to ASTM E1643; place sheets in position with longest dimension parallel with direction of pour.
 - 1. Lap joints six (6) inches and seal with manufacturer's recommended tape.

3.6 STEEL REINFORCEMENT

- A. Comply with CRSI's "Manual of Standard Practice" for fabricating, placing, and supporting reinforcement.
 - 1. Do not cut or puncture vapor barrier. Repair damage and reseal vapor barrier before placing concrete.
- B. Clean reinforcement of loose rust and mill scale, earth, ice, and other foreign materials that would reduce bond to concrete.
- C. Accurately position, support, and secure reinforcement against displacement. Locate and support reinforcement with bar supports to maintain minimum concrete cover. Do not tack weld crossing reinforcing bars.
- D. Set wire ties with ends directed into concrete, not toward exposed concrete surfaces.
- E. Install welded wire reinforcement in longest practicable lengths on bar supports spaced to minimize sagging. Lap edges and ends of adjoining sheets at least one mesh spacing plus two (2) inches. Offset laps of adjoining sheet widths to prevent continuous laps in either direction. Lace overlaps with wire.
- 3.7 JOINTS
 - A. Construct joints true to line with faces perpendicular to surface plane of concrete.
 - B. Construction Joints: Locate and install so as not to impair strength or appearance of

concrete, at locations indicated or as approved by Architect.

- C. Isolation Joints in Slabs-on-Grade: After removing formwork, install joint-filler strips at slab junctions with vertical surfaces, such as column pedestals, foundation walls, grade beams, and other locations, as indicated.
 - 1. Extend joint fillers full width and depth of joint, terminating flush with finished concrete surface, unless otherwise indicated.
- D. Contraction Joints in Slabs-on-Grade: Form weakened-plane contraction joints along column centerlines and at a maximum spacing of 20'-0" on center in both directions. Construct contraction joints for a depth equal to at least one-fourth of the concrete thickness, as follows:
 - 1. Sawed Joints: Form contraction joints with power saws equipped with shatterproof abrasive or diamond-rimmed blades. Cut 1/8-inch wide joints into concrete when cutting action will not tear, abrade, or otherwise damage surface and before concrete develops random contraction cracks.
- E. Doweled Joints: Install dowel bars and support assemblies at joints where indicated. Lubricate or asphalt coat one-half of dowel length to prevent concrete bonding to one (1) side of joint.

3.8 WATERSTOPS

- A. Flexible Waterstops: Install in construction joints and at other joints indicated to form a continuous diaphragm. Install in longest lengths practicable. Support and protect exposed waterstops during progress of the Work. Field fabricate joints in waterstops according to manufacturer's written instructions.
- B. Self-Expanding Strip Waterstops: If permitted, install in construction joints and at other joints indicated, according to manufacturer's written instructions, adhesive bonding, mechanically fastening, and firmly pressing into place. Install in longest lengths practicable.

3.9 CONCRETE PLACEMENT

- A. Before placing concrete, verify that installation of formwork, reinforcement, and embedded items is complete and that required inspections have been performed.
- B. Comply with recommendations in ACI 304R for measuring, mixing, transporting, and placing concrete.
- C. Do not add water to concrete during delivery, at project site, or during placement.
- D. Consolidate concrete with mechanical vibrating equipment.
- 3.10 FINISHING FORMED SURFACES
 - A. Rough-Formed Finish: As-cast concrete texture imparted by form-facing material with

tie holes and defective areas repaired and patched, and fins and other projections exceeding 1/4-inch in height rubbed down or chipped off.

- 1. Apply to concrete surfaces not exposed to public view.
- B. Smooth-Formed Finish: As-cast concrete texture imparted by form-facing material, arranged in an orderly and symmetrical manner with a minimum of seams. Repair and patch tie holes and defective areas. Completely remove fins and other projections.
 - 1. Apply to concrete surfaces exposed to public view or to be covered with a coating or covering material applied directly to concrete, such as waterproofing, dampproofing, veneer plaster, or painting.
 - 2. Apply smooth-rubbed, grout-cleaned, or cork-floated finish, defined in ACI 301, to smooth-formed finished concrete as indicated or directed.
- C. Related Unformed Surfaces: At tops of walls, horizontal offsets, and similar unformed surfaces adjacent to formed surfaces, strike off smooth and finish with a texture matching adjacent formed surfaces. Continue final surface treatment of formed surfaces uniformly across adjacent unformed surfaces, unless otherwise indicated.
- D. Permanently Exposed Formed Surfaces: Remove fins and other projections exceeding 1/8" in thickness, repair and patch form tie holes and other defective areas. Apply mixture of tinted Portland cement and fine aggregate per 3.16B, and as directed by Architect.
- 3.11 FINISHING FLOORS AND SLABS
 - A. General: Comply with ACI 302.1R for screeding, re-straightening, and finishing operations for concrete surfaces. Do not wet concrete surfaces.
 - B. Screed surfaces with a straightedge and strike off. Begin initial floating using bull floats or darbies to form a uniform and open-textured surface plane before excess moisture or bleedwater appears on the surface.
 - 1. Do not further disturb surfaces before starting finishing operations.
 - C. Scratch Finish: Apply scratch finish to surfaces to receive concrete floor topping or mortar setting beds for ceramic or quarry tile, Portland cement terrazzo, and other bonded cementitious floor finish, unless otherwise indicated.
 - D. Float Finish: Apply float finish to surfaces indicated, to surfaces to receive trowel finish, and to floor and slab surfaces to be covered with fluid-applied or sheet waterproofing, built-up or membrane roofing, or sand-bed terrazzo.
 - E. Trowel Finish: After applying float finish, apply first trowel finish and consolidate concrete by hand or power driven trowel. Continue troweling passes and re-straighten until surface is free of trowel marks and uniform in texture and appearance. Grind smooth any surface defects that would telegraph through applied coatings or floor coverings.

- 1. Apply a trowel finish to surfaces indicated and to floor and slab surfaces exposed to view or to be covered with resilient flooring, carpet, ceramic or quarry tile set over a cleavage membrane, paint, or another thin film-finish coating system.
- 2. Finish surfaces to the following tolerances, measured within 24 hours according To ASTM E 1155/E 1155M for a randomly trafficked floor surface:
 - a. Specified overall values of flatness, F(F) 35; and levelness, F(L) 25; with minimum local values of flatness, F(F) 24; and levelness, F(L) 17; for slabs on grade.
 - b. Specified overall values of flatness, F(F) 30; and levelness, F(L) 20; with minimum local values of flatness, F(F) 24; and levelness, F(L) 15; for suspended slabs.
- F. Trowel and Fine-Broom Finish: Apply a partial trowel finish, stopping after second troweling, to surfaces indicated and to surfaces where ceramic or quarry tile is to be installed by either thickset or thin-set methods. Immediately after second troweling, and when concrete is still plastic, slightly scarify surface with a fine broom.
- G. Non-slip Broom Finish: Apply a non-slip broom finish to surfaces indicated and to exterior concrete platforms, steps, and ramps. Immediately after float finishing, slightly roughen trafficked surface by brooming with fiber-bristle broom perpendicular to main traffic route.

3.12 MISCELLANEOUS CONCRETE ITEMS

- A. Fill in holes and openings left in concrete structures, unless otherwise indicated, after work of other trades is in place. Mix, place, and cure concrete, as specified, to blend with in-place construction. Provide other miscellaneous concrete filling indicated or required to complete the Work.
- B. Curbs: Provide monolithic finish to interior curbs by stripping forms while concrete is still green and by steel-troweling surfaces to a hard, dense finish with corners, intersections, and terminations slightly rounded.
- C. Equipment Bases and Foundations: Provide machine and equipment bases and foundations as indicated on Drawings. Set anchor bolts for machines and equipment at correct elevations, complying with diagrams or templates from manufacturer furnishing machines and equipment.

3.13 CONCRETE PROTECTION AND CURING

- A. General: Protect freshly placed concrete from premature drying and excessive cold or hot temperatures. Comply with ACI 306.1 for cold-weather protection, and follow recommendations in ACI 305R for hot-weather protection during curing.
- B. Evaporation Retarder: Apply evaporation retarder to concrete surfaces if hot, dry, or windy conditions occur before and during finishing operations. Apply according to manufacturer's written instructions after placing, screeding, and bull floating or darbying concrete, but before float finishing.

- C. Begin curing after finishing concrete, but not before free water has disappeared from concrete surface.
- D. Cure formed and unformed concrete for at least seven (7) days as follows:
 - 1. Moisture Curing: Keep surfaces continuously moist with continuous water-fog spray or absorptive cover, water saturated and kept continuously wet.
 - 2. Moisture-Retaining-Cover Curing: Cover concrete surfaces with moistureretaining cover for curing concrete, placed in widest practicable width, with sides and ends lapped at least 12 inches, and sealed by waterproof tape or adhesive. Immediately repair any holes or tears during curing period using cover material and waterproof tape.
 - 3. Curing Compound: Apply uniformly in continuous operation by power spray or roller according to manufacturer's written instructions. Recoat areas subjected to heavy rainfall within three hours after initial application. Maintain continuity of coating and repair damage during curing period. <u>Apply at permanently exposed concrete floors only.</u>

3.14 LIQUID FLOOR TREATMENTS

- A. Penetrating Liquid Floor Treatment: Prepare, apply, and finish penetrating liquid floor treatment according to manufacturer's written instructions.
- B. Sealing Coat: Uniformly apply a continuous sealing coat of curing and sealing compound to hardened concrete by power spray or roller in accordance with manufacturer's written instructions.
- 3.15 JOINT FILLING
 - A. Prepare, clean, and install joint filler in accordance with manufacturer's written instructions.
 - B. Remove dirt, debris, saw cuttings, curing compounds, and sealers from joints; leave contact surfaces of joint clean and dry.
 - C. Where indicated, install semirigid joint filler full depth in saw-cut joints and at least two (2) inches deep in formed joints. Overfill joint and trim joint filler flush with top of joint after hardening.
 - D. Install semirigid joint filler in all saw cut joints. Overfill joint and trim joint filler flush with top of joint after hardening. Semirigid joint filler shall be Eucolastic I One Part Urethane Sealant as manufactured by Euclid Chemical Co., or approved equal.

3.16 CONCRETE SURFACE REPAIRS

- A. Defective Concrete: Repair and patch defective areas when approved by Architect. Remove and replace concrete that cannot be repaired and patched to Architect's approval.
- B. Patching Mortar: Mix dry-pack patching mortar, consisting of one (1) part Portland

cement to 2-1/2 parts fine aggregate passing a No. 16 sieve, using only enough water for handling and placing.

- C. Repairing Formed Surfaces: Surface defects include color and texture irregularities, cracks, spalls, air bubbles, honeycombs, rock pockets, fins and other projections on the surface, and stains and other discolorations that cannot be removed by cleaning.
- D. Repairing Unformed Surfaces: Test unformed surfaces, such as floors and slabs, for finish and verify surface tolerances specified for each surface. Correct low and high areas. Test surfaces sloped to drain for trueness of slope and smoothness; use a sloped template.
- E. Perform structural repairs of concrete, subject to Architect's approval, using epoxy adhesive and patching mortar.
- F. Repair materials and installation not specified in this Article are subject to Architect's approval.
- 3.17 FIELD QUALITY CONTROL
 - A. Testing Agency: Owner will engage a qualified independent testing and inspecting agency to sample materials, perform tests, and submit test reports during concrete placement.

END OF SECTION 03 3000

SECTION 05 5213 - PIPE AND TUBE RAILINGS

PART 1 - GENERAL

1.1 SUMMARY

- A. Section Includes:
 - 1. Galvanized Steel pipe and tube railings.
- B. Related Sections:
 - 1. Section 03 3000 "Cast-In-Place Concrete"

1.2 ACTION SUBMITTALS

- A. Product Data: For the following:
 - 1. Grout, anchoring cement, and paint products.
- B. Shop Drawings: Include plans, elevations, sections, details, and attachments to other work.
- C. Samples: For each type of exposed finish required.
- D. Delegated-Design Submittal: For railings, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.

1.3 INFORMATIONAL SUBMITTALS

A. Product Test Reports: For pipe and tube railings, for tests performed by a qualified testing agency, according to ASTM E 894 and ASTM E 935.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Galvanized Steel Pipe and Tube Railings:
 - 1. Provide custom fabricated railings in configuration indicated on contract drawings.

2.2 PERFORMANCE REQUIREMENTS

A. Delegated Design: Engage a qualified professional engineer, as defined in Section 01 4000 "Quality Requirements," to design railings, including attachment to building construction.

Christopher Williams Architects, LLC CWA Project No.:2313

- B. Structural Performance: Railings, including attachment to building construction, shall withstand the effects of gravity loads and the following loads and stresses within limits and under conditions indicated:
 - 1. Handrails and Top Rails of Guards:
 - a. Uniform load of 50 lbf/ ft. in any direction.
 - b. Concentrated load of 200 lbf applied in any direction.
 - c. Uniform and concentrated loads need not be assumed to act concurrently.

2.3 STEEL AND IRON

- A. Recycled Content of Steel Products: Postconsumer recycled content plus one-half of preconsumer recycled content not less than 25 percent.
- B. Tubing: ASTM A 500 (cold formed) or ASTM A 513.
- C. Pipe: ASTM A 53/A 53M, Type F or Type S, Grade A, Standard Weight (Schedule 40), unless another grade and weight are required by structural loads.
 - 1. Provide galvanized finish for exterior installations and where indicated.

2.4 MISCELLANEOUS MATERIALS

- A. Welding Rods and Bare Electrodes: Select according to AWS specifications for metal alloy welded.
- B. Low-Emitting Materials: Paints and coatings shall comply with the testing and product requirements of the California Department of Public Health's (formerly, the California Department of Health Services') "Standard Method for the Testing and Evaluation of Volatile Organic Chemical Emissions from Indoor Sources Using Environmental Chambers."
- C. Etching Cleaner for Galvanized Metal: Complying with MPI#25.
- D. Galvanizing Repair Paint: High-zinc-dust-content paint complying with SSPC-Paint 20 and compatible with paints specified to be used over it.
- E. Shop Primer for Galvanized Steel: Primer formulated for exterior use over zinc-coated metal and compatible with finish paint systems indicated.
- F. Nonshrink, Nonmetallic Grout: Factory-packaged, nonstaining, noncorrosive, nongaseous grout complying with ASTM C 1107/C 1107M. Provide grout specifically recommended by manufacturer for interior and exterior applications.

2.5 FABRICATION

- A. Cut, drill, and punch metals cleanly and accurately. Remove burrs and ease edges to a radius of approximately 1/32 inch unless otherwise indicated. Remove sharp or rough areas on exposed surfaces.
- B. Form work true to line and level with accurate angles and surfaces.
- C. Welded Connections: Cope components at connections to provide close fit, or use fittings designed for this purpose. Weld all around at connections, including at fittings.
 - 1. Use materials and methods that minimize distortion and develop strength and corrosion resistance of base metals.
 - 2. Obtain fusion without undercut or overlap.
 - 3. Remove flux immediately.
 - 4. At exposed connections, finish exposed surfaces smooth and blended so no roughness shows after finishing and welded surface matches contours of adjoining surfaces.
- D. Form changes in direction by bending.
- E. For changes in direction made by bending, use jigs to produce uniform curvature for each repetitive configuration required. Maintain cross section of member throughout entire bend without buckling, twisting, cracking, or otherwise deforming exposed surfaces of components.
- 2.6 STEEL AND IRON FINISHES
 - A. Galvanized Railings:
 - 1. Hot-dip galvanize exterior steel railings, after fabrication.
 - 2. Comply with ASTM A 123/A 123M for hot-dip galvanized railings.
 - B. Preparing Galvanized Railings for Shop Priming: After galvanizing, thoroughly clean railings of grease, dirt, oil, flux, and other foreign matter, and treat with etching cleaner.
 - C. Preparation for Shop Priming: Prepare uncoated ferrous-metal surfaces to comply with SSPC-SP 6/NACE No. 3, "Commercial Blast Cleaning."
 - D. Primer Application: Apply shop primer to prepared surfaces of railings unless otherwise indicated. Comply with requirements in SSPC-PA 1, "Shop, Field, and Maintenance Painting of Steel," for shop painting. Primer need not be applied to surfaces to be embedded in concrete or masonry.
 - 1. Do not apply primer to galvanized surfaces.
 - E. High-Performance Coating: Apply epoxy intermediate and polyurethane topcoats to prime-coated surfaces. Comply with coating manufacturer's written instructions and with requirements in SSPC-PA 1, "Shop, Field, and Maintenance Painting of Steel," for shop painting. Apply at spreading rates recommended by coating manufacturer.

1. Color: As selected by Architect from manufacturer's full range.

PART 3 - EXECUTION

3.1 INSTALLATION, GENERAL

- A. Set railings accurately in location, alignment, and elevation; measured from established lines and levels and free of rack.
 - 1. Do not weld, cut, or abrade surfaces of railing components that are coated or finished after fabrication and that are intended for field connection by mechanical or other means without further cutting or fitting.
 - 2. Set posts plumb within a tolerance of 1/16 inch in 3 feet.
 - 3. Align rails so variations from level for horizontal members and variations from parallel with rake of steps and ramps for sloping members do not exceed 1/4 inch in 12 feet.

3.2 ANCHORING POSTS

A. Form or core-drill holes not less than 5 inches deep and 3/4 inch larger than OD of post for installing posts in concrete. Clean holes of loose material, insert posts, and fill annular space between post and concrete with nonshrink, nonmetallic grout or anchoring cement, mixed and placed to comply with anchoring material manufacturer's written instructions.

3.3 ADJUSTING AND CLEANING

- A. Touchup Painting: Immediately after erection, clean field welds, bolted connections, and abraded areas of shop paint, and paint exposed areas with the same material as used for shop painting to comply with SSPC-PA 1 requirements for touching up shop-painted surfaces.
- B. Galvanized Surfaces: Clean field welds, bolted connections, and abraded areas, and repair galvanizing to comply with ASTM A 780/A 780M.

END OF SECTION 05 5213

SECTION 06 1053 – MISCELLANEOUS ROUGH CARPENTRY

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Dimensional lumber.
 - 2. Wood Blocking.
 - 3. Structural Panels.

1.3 DEFINITIONS

- A. Dimension Lumber: Lumber of 2 inches nominal or greater but less than 5 inches nominal in least dimension.
- B. Lumber grading agencies, and the abbreviations used to reference them, include the following:
 - 1. NeLMA: Northeastern Lumber Manufacturers' Association.
 - 2. NHLA: National Hardwood Lumber Association.
 - 3. NLGA: National Lumber Grades Authority.
 - 4. SPIB: The Southern Pine Inspection Bureau.
 - 5. WCLIB: West Coast Lumber Inspection Bureau.
 - 6. WWPA: Western Wood Products Association.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of process and factory-fabricated product. Indicate component materials and dimensions and include construction and application details.
 - 1. Include data for fire-retardant treatment from chemical treatment manufacturer and certification by treating plant that treated materials comply with requirements. Include physical properties of treated materials based on testing by a qualified independent testing agency.
 - 2. For fire-retardant treatments, include physical properties of treated lumber both before and after exposure to elevated temperatures, based on testing by a qualified independent testing agency according to ASTM D 5664.
 - 3. For products receiving a waterborne treatment, include statement that moisture content of treated materials was reduced to levels specified before shipment to Project site.
 - 4. Include copies of warranties from chemical treatment manufacturers for each type of treatment.

- B. Submittals:
 - 1. Chain-of-custody certificates indicating that products specified to be made from certified wood comply with forest certification requirements. Include documentation that manufacturer is certified for chain of custody by an FSC-accredited certification body. Include statement indicating cost for each certified wood product.
 - 2. For adhesives, documentation including printed statement of VOC content.
 - 3. For adhesives and plywood, documentation indicating that products comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."

1.5 INFORMATIONAL SUBMITTALS

- A. Evaluation Reports: For the following, from ICC-ES:
 - 1. Fire-retardant-treated wood.
 - 2. Power-driven fasteners.
 - 3. Powder-actuated fasteners.
 - 4. Expansion anchors.
 - 5. Metal framing anchors.
- 1.6 QUALITY ASSURANCE
- A. Testing Agency Qualifications: For testing agency providing classification marking for fire-retardant treated material, an inspection agency acceptable to authorities having jurisdiction that periodically performs inspections to verify that the material bearing the classification marking is representative of the material tested.
- 1.7 DELIVERY, STORAGE, AND HANDLING
 - A. Stack lumber flat with spacers beneath and between each bundle to provide air circulation. Protect lumber from weather by covering with waterproof sheeting, securely anchored. Provide for air circulation around stacks and under coverings.

PART 2 - PRODUCTS

2.1 WOOD PRODUCTS, GENERAL

- A. Certified Wood: Lumber and plywood shall be produced from wood obtained from forests certified by an FSC-accredited certification body to comply with FSC STD-01-001, "FSC Principles and Criteria for Forest Stewardship."
- B. Lumber: DOC PS 20 and applicable rules of grading agencies indicated. If no grading agency is indicated, provide lumber that complies with the applicable rules of any rules-writing agency certified by the ALSC Board of Review. Provide lumber graded by an agency certified by the ALSC Board of Review to inspect and grade lumber under the rules indicated.

- 1. Factory mark each piece of lumber with grade stamp of grading agency.
- 2. For exposed lumber indicated to receive a stained or natural finish, mark grade stamp on end or back of each piece or omit grade stamp and provide certificates of grade compliance issued by grading agency.
- 3. Where nominal sizes are indicated, provide actual sizes required by DOC PS 20 for moisture content specified. Where actual sizes are indicated, they are minimum dressed sizes for dry lumber.
- 4. Provide dressed lumber, S4S, unless otherwise indicated.
- C. Maximum Moisture Content of Lumber: 19 percent unless otherwise indicated.
- D. Wood Structural Panels:
 - 1. Plywood: Either DOC PS 1 or DOC PS 2, unless otherwise indicated.
 - 2. Thickness: As needed to comply with requirements specified but not less than thickness indicated.
 - 3. Comply with "Code Plus" provisions in APA Form No. E30K, "APA Design/Construction Guide: Residential & Commercial."
 - 4. Factory mark panels according to indicated standard.

2.3 FIRE-RETARDANT-TREATED MATERIALS

- A. General: All lumber and plywood shall be fire retardant treated. Use materials complying with requirements in this article, that are acceptable to authorities having jurisdiction, and with fire-test response characteristics specified as determined by testing identical products per test method indicated by a qualified testing agency.
- B. Fire-Retardant-Treated Lumber and Plywood by Pressure Process: Products with a flame spread index of 25 or less when tested according to ASTM E 84, and with no evidence of significant progressive combustion when the test is extended an additional 20 minutes, and with the flame front not extending more than 10.5 feet (3.2 m) beyond the centerline of the burners at any time during the test.
 - 1. Use treatment that does not promote corrosion of metal fasteners.
 - 2. Exterior Type: Treated materials shall comply with requirements specified above for fire- retardant-treated lumber and plywood by pressure process after being subjected to accelerated weathering according to ASTM D 2898. Use for exterior locations and where indicated.
 - 3. Interior Type A: Treated materials shall have a moisture content of 28 percent or less when tested according to ASTM D 3201 at 92 percent relative humidity. Use where exterior type is not indicated.
 - 4. Design Value Adjustment Factors: Treated lumber shall be tested according ASTM D 5664, and design value adjustment factors shall be calculated according to ASTM D
- C. Kiln-dry lumber after treatment to a maximum moisture content of 19 percent. Kiln-dry plywood after treatment to a maximum moisture content of 15 percent.

- D. Identify fire-retardant-treated wood with appropriate classification marking of testing and inspecting agency acceptable to authorities having jurisdiction.
 - 1. For exposed lumber indicated to receive a stained or natural finish, mark end or back of each piece or omit marking and provide certificates of treatment compliance issued by inspection agency.
- E. For exposed items indicated to receive a stained or natural finish, use chemical formulations that do not bleed through, contain colorants, or otherwise adversely affect finishes.
- A. Application: Treat all miscellaneous carpentry and rough carpentry unless otherwise indicated.
 - 1. Dimension lumber wood blocking, equipment support, and similar members.
 - 2. Concealed blocking.
 - 3. Plywood backing panels.

2.4 MISCELLANEOUS LUMBER

- A. General: Provide miscellaneous lumber indicated and lumber for support or attachment of other construction, including the following:
 - 1. Blocking.
 - 2. Nailers.
- B. For items of dimension lumber size, provide Construction or No. 2 grade lumber and any of the following species:
 - 1. Hem-fir (north); NLGA.
 - 2. Mixed southern pine; SPIB.
 - 3. Spruce-pine-fir; NLGA.
 - 4. Hem-fir; WCLIB or WWPA.
 - 5. Spruce-pine-fir (south); NeLMA, WCLIB, or WWPA.
- C. For concealed boards, provide lumber with 15 percent maximum moisture content and any of the following species and grades:
 - 1. Mixed southern pine, No. 2 grade; SPIB.
 - 2. Hem-fir or hem-fir (north), Construction or No. 2 Common grade; NLGA, WCLIB, or WWPA.
 - 3. Spruce-pine-fir (south) or spruce-pine-fir, Construction or No. 2 Common grade; NeLMA, NLGA, WCLIB, or WWPA.
- D. For blocking not used for attachment of other construction, Utility, Stud, or No. 3 grade lumber of any species may be used provided that it is cut and selected to eliminate defects that will interfere with its attachment and purpose.
- E. For blocking and nailers used for attachment of other construction, select and cut lumber to eliminate knots and other defects that will interfere with attachment of other work.

2.5 PLYWOOD BACKING PANELS

- A. Equipment Backing Panels: DOC PS 1, Exposure 1, C-D Plugged, fire-retardant treated, in thickness indicated or, if not indicated, not less than 3/4-inch (19-mm) nominal thickness.
 - 1. Plywood shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."

2.6 FASTENERS

- A. General: Provide fasteners of size and type indicated that comply with requirements specified in this article for material and manufacture.
- B. Nails, Brads, and Staples: ASTM F 1667.
- C. Power-Driven Fasteners: NES NER-272.
- D. Wood Screws: ASME B18.6.1.
- E. Screws for Fastening to Metal Framing: ASTM C 954, length as recommended by screw manufacturer for material being fastened.
- F. Lag Bolts: ASME B18.2.1 (ASME B18.2.3.8M).
- G. Bolts: Steel bolts complying with ASTM A 307, Grade A (ASTM F 568M, Property Class 4.6); with ASTM A 563 (ASTM A 563M) hex nuts and, where indicated, flat washers.
- H. Expansion Anchors: Anchor bolt and sleeve assembly of material indicated below with capability to sustain, without failure, a load equal to 6 times the load imposed when installed in unit masonry assemblies and equal to 4 times the load imposed when installed in concrete as determined by testing per ASTM E 488 conducted by a qualified independent testing and inspecting agency.
 - 1. Material: Carbon-steel components, zinc plated to comply with ASTM B 633, Class Fe/Zn 5.
 - Material: Stainless steel with bolts and nuts complying with ASTM F 593 and ASTM F 594, Alloy Group 1 or 2 (ASTM F 738M and ASTM F 836M, Grade A1 or A4).
- I. Metal Framing Anchors:
 - 1. Manufacturers: One of the following:
 - a. Simpson Strong Tie Co., Inc
 - b. USP Stuctural Connectors
 - c. Cleveland Steel Specialty Co.

2. Use anchors made from hot-dip galvanized steel complying with ASTM A 653/A 653M, G60 (Z180) coating designation.

PART 3 - EXECUTION

3.1 INSTALLATION, GENERAL

- A. Set carpentry to required levels and lines, with members plumb, true to line, cut, and fitted. Fit carpentry to other construction; scribe and cope as needed for accurate fit. Locate furring, nailers, blocking, grounds, and similar supports to comply with requirements for attaching other construction.
- B. Where wood-preservative-treated lumber is installed adjacent to metal decking, install continuous flexible flashing separator between wood and metal decking.
- C. Install plywood backing panels by fastening to studs; coordinate locations with utilities requiring backing panels. Install fire-retardant treated plywood backing panels with classification marking of testing agency exposed to view.
- D. Metal Framing Anchors: Install metal framing anchors to comply with manufacturer's written instructions. Install fasteners through each fastener hole.
- E. Do not splice structural members between supports unless otherwise indicated.
- F. Provide blocking and framing as indicated and as required to support facing materials, fixtures, specialty items, and trim. Drawings do not show every piece of blocking.
 - 1. Provide metal clips for fastening gypsum board or lath at corners and intersections where framing or blocking does not provide a surface for fastening edges of panels. Space clips not more than 16 inches (406 mm) o.c.
 - 2. Provide continuous, wall-to-wall plywood blocking at all bathroom walls requiring blocking for such items as grab-bars and other wall-mounted elements. GC to coordinate all blocking so that finished installations shall provide code-required loading and pull-out strength.
 - 3. Coordinate finished dimensions and framing so that blocking is incorporated behind the finished wall and any required clearances including handicapped clearances. Provide a field layout for review with architect in the field prior to closing walls requiring blocking.
- G. Sort and select lumber so that natural characteristics will not interfere with installation or with fastening other materials to lumber. Do not use materials with defects that interfere with function of member or pieces that are too small to use with minimum number of joints or optimum joint arrangement.
- H. Comply with AWPA M4 for applying field treatment to cut surfaces of preservative-treated lumber.
 - 1. Use inorganic boron for items that are continuously protected from liquid water.
 - 2. Use copper naphthenate for items not continuously protected from liquid water.

- I. Securely attach carpentry work to substrate by anchoring and fastening as indicated, complying with the following:
 - 1. State Building Code.
 - 2. NES NER-272 for power-driven fasteners.
 - 3. Table 2304.9.1, "Fastening Schedule," in ICC's International Building Code.
- J. Use steel common nails unless otherwise indicated. Select fasteners of size that will not fully penetrate members where opposite side will be exposed to view or will receive finish materials. Make tight connections between members. Install fasteners without splitting wood. Drive nails snug but do not countersink nail heads unless otherwise indicated.

3.2 WOOD BLOCKING AND NAILER INSTALLATION

- A. Install where indicated and where required for attaching other work. Form to shapes indicated and cut as required for true line and level of attached work. Coordinate locations with other work involved.
- B. Attach items to substrates to support applied loading. Recess bolts and nuts flush with surfaces unless otherwise indicated.
- C. Provide permanent grounds of dressed, pressure-preservative-treated, key-beveled lumber not less than 1-1/2 inches wide and of thickness required to bring face of ground to exact thickness of finish material. Remove temporary grounds when no longer required.

3.3 PLYWOOD PANEL INSTALLATION

- A. General: Comply with applicable recommendations contained in APA Form No. E30K, "APA Design/Construction Guide: Residential & Commercial," for types of structuraluse panels and applications indicated.
 - 1. Comply with "Code Plus" provisions in above-referenced guide.
- B. Fastening Methods: Fasten panels as indicated below:
 - 1. Sheathing:
 - a. Screw to cold-formed metal framing.
 - b. Space panels 1/8 inch apart at edges and ends.
 - 2. Plywood Backing Panels: Nail or screw to supports.

3.4 PROTECTION

A. Protect wood that has been treated with inorganic boron (SBX) from weather. If, despite protection, inorganic boron-treated wood becomes wet, apply EPA-registered borate treatment. Apply borate solution by spraying to comply with EPA-registered label.

B. Protect miscellaneous rough carpentry from weather. If, despite protection, miscellaneous rough carpentry becomes wet, apply EPA-registered borate treatment. Apply borate solution by spraying to comply with EPA-registered label.

END OF SECTION 06 1053

SECTION 06 2013 - EXTERIOR FINISH CARPENTRY

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Exterior primed hardboard and cellular PVC trim.
- B. Related Requirements:
 - 1. Section 06 1053 "Miscellaneous Rough Carpentry" for furring, blocking, and other carpentry work not exposed to view and for framing exposed to view.
 - 2. Section 07 2500 "Weather Barriers"
 - 3. Section 07 4640 "Vinyl Siding and Soffits"
 - 4. Section 09 2900 "Gypsum Board"

1.3 DEFINITIONS

- A. MDO: Plywood with a medium-density overlay on the face.
- B. PVC: Polyvinyl chloride.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of process and factory-fabricated product. Indicate component materials, dimensions, profiles, textures, and colors and include construction and application details.
 - 1. Include data for wood-preservative treatment from chemical-treatment manufacturer and certification by treating plant that treated materials comply with requirements. Indicate type of preservative used and net amount of preservative retained. Include chemical-treatment manufacturer's written instructions for finishing treated material.
 - 2. For products receiving a waterborne treatment, include statement that moisture content of treated materials was reduced before shipment to Project site to levels specified.
- B. Samples: For each exposed product and for each color and texture specified.
- C. Samples for Initial Selection: For each type of product involving selection of colors, profiles, or textures.

- D. Samples for Verification:
 - 1. For cellular PVC trim, with half of exposed surface finished; 50 sq. in.
- 1.5 INFORMATIONAL SUBMITTALS
 - A. Evaluation Reports: For the following, from ICC-ES:
 - 1. Cellular PVC trim.
 - B. Sample Warranties: For manufacturer's warranties.
- 1.6 DELIVERY, STORAGE, AND HANDLING
 - A. Stack lumber, plywood, and other panels flat with spacers between each bundle to provide air circulation.
 - 1. Protect materials from weather by covering with waterproof sheeting, securely anchored.
 - 2. Provide for air circulation around stacks and under coverings.

1.7 FIELD CONDITIONS

- A. Weather Limitations: Proceed with installation only when existing and forecast weather conditions permit work to be performed and at least one coat of specified finish can be applied without exposure to rain, snow, or dampness.
- B. Do not install finish carpentry materials that are wet, moisture damaged, or mold damaged.
 - 1. Indications that materials are wet or moisture damaged include, but are not limited to, discoloration, sagging, or irregular shape.
 - 2. Indications that materials are mold damaged include, but are not limited to, fuzzy or splotchy surface contamination and discoloration.

PART 2 - PRODUCTS

2.1 EXTERIOR TRIM

- A. Cellular PVC Trim: Extruded, expanded PVC with a small-cell microstructure, recommended by manufacturer for exterior use, made from UV- and heat-stabilized rigid material.
 - 1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - a. CertainTeed Corporation.
 - b. CPG Building Products LLC.
 - c. Gossen Corporation.

- d. Kleer Lumber, LLC.
- e. Versatex Trimboard; a Wolfpac Technologies, Inc. company.
- f. Azek Building Products
- 2. Density: Not less than 31 lb/cu. ft.
- 3. Heat Deflection Temperature: Not less than 130 deg F, according to ASTM D 648.
- 4. Coefficient of Thermal Expansion: Not more than 4.5 x 10(-5) inches/inch x deg F
- 5. Water Absorption: Not more than 1 percent, according to ASTM D 570.
- 6. Flame-Spread Index: 75 or less, according to ASTM E 84.

2.2 MISCELLANEOUS MATERIALS

- A. Fasteners for Exterior Finish Carpentry: Provide nails or screws, in sufficient length to penetrate not less than 1-1/2 inches into wood substrate.
 - 1. For face-fastening siding, provide ringed-shank siding nails or hot-dip galvanizedsteel siding nails unless otherwise indicated.
 - 2. For prefinished items, provide matching prefinished aluminum fasteners where face fastening is required.
 - 3. For pressure-preservative-treated wood, provide stainless steel fasteners.
 - 4. For applications not otherwise indicated, provide stainless steel fasteners.
- B. Wood Glue: Waterproof resorcinol glue recommended by manufacturer for exterior carpentry use.
- C. Adhesive for Cellular PVC Trim: Product recommended by trim manufacturer.
- D. Flashing: Comply with requirements in Section 07620 "Sheet Metal Flashing and Trim" for flashing materials installed in exterior finish carpentry.
 - 1. Horizontal Joint Flashing for Panel Siding: Preformed, prefinished-aluminum, Z-shaped flashing.
- E. Round Soffit Vents: Stamped aluminum louvered vents, 2-1/2 inches in diameter, made to be inserted in round holes cut in soffit.
 - 1. Finish: As Selected by the Architect
- F. Sealants: Latex, complying with ASTM C 834 Type OP, Grade NF and applicable requirements in Section 07920 "Joint Sealants," and recommended by sealant and substrate manufacturers for intended application.
 - 1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - a. Bostik, Inc.
 - b. Franklin International.
 - c. May National Associates, Inc.; a subsidiary of Sika Corporation.
 - d. Pecora Corporation.

2.3 FABRICATION

- A. Back out or kerf backs of standing and running trim wider than 5 inches, except members with ends exposed in finished work.
- B. Ease edges of lumber less than 1 inch in nominal thickness to 1/16-inch radius and edges of lumber 1 inch or more in nominal thickness to 1/8-inch radius.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, with Installer present, for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. Examine finish carpentry materials before installation. Reject materials that are wet, moisture damaged, and mold damaged.
- C. Proceed with installation only after unsatisfactory conditions have been corrected.
- D. For decking and porch flooring installations, ensure surfaces are suitable for installation of decking and that adequate structural support has been provided.
 - 1. Standard Installation: Confirm that joists are spaced at 16 inches on center maximum, and are sloped at a minimum of 1/4 inch per foot away from the building.
 - 2. Forty-five Degree Angle Installation: Confirm that joists are spaced at 12 inches on center maximum, and are sloped at a minimum of 1/4 inch per foot away from the building.

3.2 PREPARATION

- A. Clean substrates of projections and substances detrimental to application.
- B. Prime lumber and moldings to be painted, including both faces and edges, unless factory primed.
 - 1. Cut to required lengths and prime ends.
 - 2. Comply with requirements in Section 09911 "Exterior Painting."

3.3 INSTALLATION, GENERAL

- A. Do not use materials that are unsound, warped, improperly treated or finished, inadequately seasoned, or too small to fabricate with proper jointing arrangements.
 - 1. Do not use manufactured units with defective surfaces, sizes, or patterns.
- B. Install exterior finish carpentry level, plumb, true, and aligned with adjacent materials.

- 1. Use concealed shims where necessary for alignment.
- 2. Scribe and cut exterior finish carpentry to fit adjoining work.
- 3. Refinish and seal cuts as recommended by manufacturer.
- 4. Install to tolerance of 1/8 inch in 96 inches for level and plumb. Install adjoining exterior finish carpentry with 1/32-inch maximum offset for flush installation and 1/16-inch maximum offset for reveal installation.
- 5. Coordinate exterior finish carpentry with materials and systems in or adjacent to it.
- 6. Provide cutouts for mechanical and electrical items that penetrate exterior finish carpentry.
- 3.4 STANDING AND RUNNING TRIM INSTALLATION
 - A. Install cellular PVC trim to comply with manufacturer's written instructions.
 - B. Install trim with minimum number of joints as is practical, using full-length pieces from maximum lengths of lumber available. Do not use pieces less than 24 inches long, except where necessary.
 - 1. Use scarf joints for end-to-end joints.
 - 2. Stagger end joints in adjacent and related members.
 - C. Fit exterior joints to exclude water.
 - 1. Cope at returns and miter at corners to produce tight-fitting joints, with full-surface contact throughout length of joint.
 - 2. Plane backs of casings to provide uniform thickness across joints, where necessary for alignment.
 - D. Where face fastening is unavoidable, countersink fasteners, fill surface flush, and sand unless otherwise indicated.
- 3.5 DECKING INSTALLATION
 - A. Decking Installation.
 - 1. Install with grain side up for the walking surface.
 - 2. Fasten tight to joists. Provide shims if there are variations in framing.
 - 3. Countersink fasteners slightly to provide necessary clearance when installing the next board.
 - 4. Cut final boards as required for proper appearance.
 - 5. Fasten with stainless steel screws, two per deck board into each joist
- 3.6 ADJUSTING
 - A. Replace exterior finish carpentry that is damaged or does not comply with requirements.
 - 1. Exterior finish carpentry may be repaired or refinished if work complies with requirements and shows no evidence of repair or refinishing.
 - B. Adjust joinery for uniform appearance.

3.7 CLEANING

- A. Clean exterior finish carpentry on exposed and semiexposed surfaces.
- B. Touch up factory-applied finishes to restore damaged or soiled areas.

3.8 PROTECTION

- A. Protect installed products from damage from weather and other causes during construction.
- B. Remove and replace finish carpentry materials that are wet, moisture damaged, and mold damaged.
 - 1. Indications that materials are wet or moisture damaged include, but are not limited to, discoloration, sagging, or irregular shape.
 - 2. Indications that materials are mold damaged include, but are not limited to, fuzzy or splotchy surface contamination and discoloration.

END OF SECTION 06201

SECTION 06 7300 – COMPOSITE DECKING

PART 1 GENERAL

1.1 SECTION INCLUDES

A. Composite Decking (Trex Select)

1.2 RELATED SECTIONS

A. Section 06-1100 – Wood Framing

1.3 REFERENCES

- A. ASTM D-7032-04: Standard Specification for Establishing Performance Ratings for Wood-Plastic Composite Deck Boards and Guardrail Systems (Guards or Handrails), ASTM International.
- B. ASTM D-7031-04: Standard Guide for Evaluating Mechanical and Physical Properties of Wood-Plastic Composite Products, ASTM International
- C. ASTM E-84-01: Test Method for Surface Burning Characteristics of Building Materials, ASTM International.
- D. ASTM D 570: Water Absorption of Plastics
- E. ASTM D 1761: Mechanical Fasteners in Wood
- F. ASTM D -1413-99: Test method for Wood Preservatives by Laboratory Soil-block Cultures
- G. ASTM C177: Standard Test Method for Steady-State Heat Flux Measurements and Thermal Transmission Properties by Means of the Guarded-Hot-Plate Apparatus

1.4 DESIGN/PERFORMANCE REQUIREMENTS

- A. Structural Performance:
 - a. Deck: Uniform Load 100lbf/sq.ft.
 - b. Tread of Stairs: Concentrated Load: 750 lbf/sq.ft., and 1/8" max. deflection with a concentrated load of 300 lbf on area of 4 sq. in.
- B. Fire-Surface Burning Characteristics per ASTM E-84.

1.5 SUBMITTALS

- A. Product Data Indicate sizes, profiles, surface style, and performance characteristics
- B. Samples: For each product specified, one sample representing actual product color, size, and finish.
- 1.6 Delivery, Storage, and Handling
 - A. Store Trex products on a flat and level surface. Adjust support blocks accordingly

Christopher Williams Architects, LLC CWA Project No.:2313

ISSUED FOR BIDDING March 12, 2024 Page 1 of 3

- B. Support Trex bundles on supplied dunnage
- C. When stacking Trex bundles, supports should start approximately 8" from each end and be spaced approximately 2ft on center. Supports should line up vertically/perpendicular to the decking product.
- D. Do not stack Trex Select decking more than 14 bundles.
- E. Keep material covered using the provided bundle cover until time of installation.
- F. See www. Trex.com for detailed storage recommendations;
 a. http://s7d4.scene7.com/is/content/Trex/Installation%20Guide%202013pdf

1.7 Warranty

A. Provide manufactures warranty against rot, decay, splitting, checking, splintering, fungal damage, and termite damage for a period of 25 years for a residential installation and 10 years for a commercial installation. In addition, provide the Trex Fade and Stain Warranty against food staining and fading beyond 5 Delta E (CIE units) for a period of 25 years for a residential installation and 10 years for a commercial installation and 10 years for a residential installation and 10 years for a warranty against food staining and fading beyond 5 Delta E (CIE units) for a period of 25 years for a residential installation and 10 years for a commercial installation. Specific terms for warranties can be found at; www.Trex.com.

PART 2 PRODUCTS

2.1 Manufacturers

- A. Contract Documents are based on products supplied by; Trex Company, Inc., 160 Exeter Dr., Winchester, VA 22603.
- B. Substitutions: Not permitted under Division 01
- 2.2 Applications/Scope
 - A. Wood-Plastic Composite Lumber;
 - a. Material Description: Composite Decking consisting of recycled Linear Low-Density Polyethylene (LLDPE) and recycled wood. The product is extruded into shapes and sizes as follows:
 - i. Trex Select Decking Boards; 0.875" x 5.5".
 - ii. Lengths 12, 16, and 20 feet
 - iii. Color To be specified by owner from Trex' standard list of colors.
 - b. Physical and Mechanical Properties as follows:

Test	Test Method	Value	
Flame spread	ASTM E 84	85	
	ASTM D	1.9 x 10-5	
Thermal Expansion	1037	inch/inch/degreeF	
	ASTM D		
Moisture Absorption	1037	< 1.2%	
	ASTM		
Screw Withdrawal	D1761	388 lbs/in	
	ASTM		
Fungus Resistance	D1413	Rating - no decay	

Christopher Williams Architects, LLC CWA Project No.:2313
Termite Resistance	AWPAE1-72	Rating = 9.7	
		Ultimate (Typical)Values	<u>Design</u>
		*	Values
Compression Parallel	ASTM D198	1588 psi	540 psi
Compression			
Perpendicular	ASTM D143	1437 psi	540 psi
Bending Strength	ASTM D198	3280 psi	500 psi
Shear Strength	ASTM D143	1761 psi	360 psi
	ASTM		
Modulus of Elasticity	D4761	400,000psi	200,000 psi
	ASTM		
Modulus of Rupture	D4761	3750 psi	500 psi

* Ultimate strength values are not meant for design analysis. Design values are for temperatures up to 130F (54C)

2.2 Accessories

- A. Fasteners:
 - a. Trex Universal Hideaway Hidden Fasteners
 - b. Screws; See -<u>http://s7d4.scene7.com/is/content/Trex/Installation%20Guide%202013pd</u> f for the updated recommendations on fasteners.

PART 3 EXECUTION

- 3.1 Installation
 - A. Install according to Trex installation guidelines. <u>http://s7d4.scene7.com/is/content/Trex/Installation%20Guide%202013pdf</u>
 - B. Cut, drill, and rout using carbide tipped blades
 - C. Do not use composite wood material for structural applications

3.2 Cleaning

A. Following cleaning recommendations as found in Trex installation guide at; http://s7d4.scene7.com/is/content/Trex/Installation%20Guide%202013pdf

END OF SECTION 06150

SECTION 07 1616 – CRYSTALLINE WATERPROOFING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes crystalline waterproofing. On exposed faces of existing masonry walls (stone and brick).
- B. System description: crystalline type that is a blend of Portland cement, fine treated silica sand and active proprietary chemicals. When mixed with water and applied as a cementitious coating, the active chemicals diffuse into the concrete and cause a catalytic reaction which generates a non-soluble crystalline structure within the pores and capillary tracts of concrete. This crystalline system causes the concrete to become sealed against the penetration of liquids from any direction, and protects the concrete from deterioration due to harsh environmental conditions.

1.3 ACTION SUBMITTALS

A. Product Data: For each type of product.

1.4 INFORMATIONAL SUBMITTALS

- A. Qualification Data: For Applicator.
- B. Manufacturer's Certification: Provide document signed by manufacturer or manufacturer's representative certifying that the materials to be installed comply with the requirements of this specification
- C. Manufacturer's Field Report: Provide copy of report from manufacturer's representative confirming that the surfaces to which waterproofing material is to be applied are in a condition suitable to receive same.

1.5 QUALITY ASSURANCE

A. Manufacturer Qualifications: Manufacturer shall be ISO 9001 registered, and shall have no less than 10 years' experience in manufacturing the cementitious crystalline waterproofing materials for the required work. Manufacturer must be capable of providing field service representation during construction phase. Manufacturers who cannot provide ongoing field support or the performance test data specified herein will not be considered for the project.

- B. Applicator: Waterproofing applicator shall be experienced in the installation of cementitious crystalline waterproofing materials as demonstrated by previous successful installations, and shall be approved by the manufacturer in writing.
- C. Technical Consultation: The waterproofing manufacturer's representative shall provide technical consultation on waterproofing application and provide on-site support as needed

1.6 FIELD CONDITIONS

A. Comply with manufacturer's product data regarding condition of substrate to receive waterproofing, weather conditions before and during installation, and protection of the installed waterproofing system.

1.7 WARRANTY

- A. Manufacturer's Warranty: Manufacturer shall provide standard product warranty executed by authorized company official.
- B. Applicator's Warranty: Applicator shall warrant the waterproofing installation against defects caused by faulty workmanship or materials for a period of typically (specify term) years from Date of Substantial Completion. The warranty will cover the surfaces treated and will bind the applicator to repair, at his expense, any and all leaks through the treated surfaces which are not due to structural weaknesses or other causes beyond applicator's control such as fire, earthquake, tornado and hurricane. The warranty shall read as follows:

Warranty: The applicator warrants that, upon completion of the work, surfaces treated with cementitious crystalline waterproofing will be and will remain free from water leakage resulting from defective workmanship or materials for a period of (specify term) years from Date of Substantial Completion. In the event that water leakage occurs within the warranty period from such causes, the applicator shall, at his sole expense, repair, replace or otherwise correct such defective workmanship or materials. Applicator shall not be liable for consequential damages and applicator's liability shall be limited to repair, replacement or correcting of defective workmanship or materials. Applicator shall have no responsibility with respect to water leakage or other defects caused by structural failure or movement of the structure, or any other causes beyond Applicator's control

PART 2 - PRODUCTS

2.1 WATERPROOFING MATERIALS

A. Basis of design: The following products manufactured by Xypex Chemical Corporation

Christopher Williams Architects, LLC CWA Project No.:2313 ISSUED FOR BIDDING March 12, 2024 Page 2 of 4

- 1. Xypex Concentrate
- 2. Xypex Megamix I
- 3. Xypex Xycrylic Admix
- B. Other acceptable manufacturers:
 - 1. American Permaquick Inc
 - 2. Anti-Hydro International Inc
 - 3. Koster American Corporation
- C. Obtain all proprietary crystalline waterproofing products from a single manufacturer:

2.2 ACCESSORY MATERIALS

A. Water: Potable.

2.3 MIXES

A. Crystalline Waterproofing: Add prepackaged dry ingredients to water according to manufacturer's written instructions. Mix together with mechanical mixer or by hand to required consistency.

PART 3 - EXECUTION

3.1 EXAMINATION

A. Prior to waterproofing installation, arrange visit to project site with waterproofing manufacturer's representative. Representative shall inspect and certify that concrete surfaces are in acceptable condition to receive waterproofing treatment.

3.2 PREPARATION

- A. Comply with manufacturer's written instructions.
- B. Protect other work from damage caused by cleaning, preparation, and application of waterproofing. Provide temporary enclosure to confine spraying operation and to ensure adequate ambient temperatures and ventilation conditions for application.
- C. Thoroughly clean and profile all surfaces. The surface must also have an open capillary system to provide "tooth and suction" for the Xypex treatment. A minimum of CSP-3 per the International Concrete Repair Institute Concrete Surface Profile Chips or other equivalent standard is required
- D. Wet surface of masonry until a saturated surface dry (SSD) condition is obtained. Dry substrates will not absorb any additional water but has no glistening water on its surface. Maintain the masonry in an SSD condition until waterproofing is applied.

- E. Apply one coat of Xypex Concentrate per manufacture's specification
- F. Allow to set and harden for between 12 hours and 24 hours. During this period moist cure per manufacturer's product data sheet.
- G. Mix Xypex Megamix I and Xycrylic Admix per Manufacturer's instructions.
- H. Apply one coat of mix over waterproofing at a thickness of 1/8-inch.
- I. Manufacturer's Field Service: Engage a factory-authorized service representative to test and inspect completed application of waterproofing.
- J. Prepare test and inspection reports.

END OF SECTION 07 1616

SECTION 07 2500 - WEATHER BARRIERS

PART 1 - GENERAL

1.1 SECTION INCLUDES

- A. A complete weather barrier assembly for vertical building envelope protection that will maintain air/moisture resistance while maintaining moisture-vapor permeability. The assembly consists of the following components.
 - 1. Weather barrier membrane
 - 2. Seam Tape
 - 3. Flashing
 - 4. Fasteners

1.2 REFERENCES

- A. ASTM International
 - 1. ASTM C 920; Standard Specification for Elastomeric Joint Sealants
 - 2. ASTM C 1193; Standard Guide for Use of Joint Sealants
 - 3. ASTM D 882; Test Method for Tensile Properties of Thin Plastic Sheeting
 - 4. ASTM D 1117; Standard Guide for Evaluating Non-woven Fabrics
 - 5. ASTM E 84; Test Method for Surface Burning Characteristics of Building Materials
 - 6. ASTM E 96; Test Method for Water Vapor Transmission of Materials
 - 7. ASTM E 1677; Specification for Air Retarder Material or System for Framed Building Walls
 - 8. ASTM E2178; Test Method for Air Permeance of Building Materials
 - 9. ASTM E2357; Standard Test Method for Determining Air Leakage of Air Barrier Assemblies
- B. AATCC American Association of Textile Chemists & Colorists
 - 1. Test Method 127 Water Resistance: Hydrostatic Pressure Test
- C. TAPPI
 - 1. Test Method T-410; Grams of Paper and Paperboard (Weight per Unit Area)
 - 2. Test Method T-460; Air Resistance of Paper (Gurley Hill Method)
- 1.3 SUBMITTALS
 - A. Refer to Section 01 3300 Submittal Procedures.
 - B. Product Data: Submit manufacturer current technical literature for each component.
 - C. Samples: Weather Barrier Membrane, minimum 8-1/2 inches by 11 inches.
 - D. Quality Assurance Submittals

- 1. Design Data, Test Reports: Provide manufacturer test reports indicating product compliance with indicated requirements.
- 2. Manufacturer Instructions: Provide manufacturer's written installation instructions.
- 3. Manufacturer's Field Service Reports: Provide site reports from authorized field service representative, indicating observation of weather barrier assembly installation.
- E. Closeout Submittals
 - 1. Weather Barrier Warranty: Manufacturer's executed warranty form with authorized signatures and endorsements indicating date of Substantial Completion.
- 1.4 QUALITY ASSURANCE
 - A. Qualifications
 - 1. Installer shall have experience with installation of weather barrier assemblies under similar conditions.
 - 2. Installation shall be in accordance with weather barrier manufacturer's installation guidelines and recommendations.
 - 3. Source Limitations: Provide weather barrier and accessory materials produced by single manufacturer.
 - B. Mock-up
 - 1. Install mock-up using approved weather barrier assembly including fasteners, flashing, tape and related accessories per manufacturer's current printed instructions and recommendations. Mock shall include portion of wall without penetrations and typical door and window openings.
 - a. Mock-up size: 8 feet by 8 feet.
 - b. Mock-up Substrate: Match wall assembly construction, including window opening.
 - c. Mock-up may not remain as part of the work.
 - 2. Contact manufacturer's designated representative prior to weather barrier assembly installation, to perform required mock-up visual inspection and analysis as required for warranty.
 - C. Pre-installation Meeting
 - 1. Hold a pre-installation conference, two weeks prior to start of weather barrier installation. Attendees shall include Contractor, Architect, installer, Owner's Representative, and weather barrier manufacturer's designated representative.
 - 2. Review all related project requirements and submittals, status of substrate work and preparation, areas of potential conflict and interface, availability of weather barrier assembly materials and components, installer's training requirements, equipment, facilities and scaffolding, and

coordinate methods, procedures and sequencing requirements for full and proper installation, integration and protection.

- 1.5 DELIVERY, STORAGE AND HANDLING
 - A. Refer to Section 01 6000 Product Requirements.
 - B. Deliver weather barrier materials and components in manufacturer's original, unopened, undamaged containers with identification labels intact.
 - C. Store weather barrier materials as recommended by weather barrier manufacturer.
- 1.6 SCHEDULING
 - A. Review requirements for sequencing of installation of weather barrier assembly with installation of windows, doors, louvers and flashings to provide a weather-tight barrier assembly.
 - B. Schedule installation of weather barrier materials and exterior cladding within nine months of weather barrier assembly installation.
- 1.7 WARRANTY
 - A. Special Warranty
 - 1. Weather barrier manufacturer's warranty for weather barrier for a period of ten (10) years from date of purchase.
 - 2. Pre-installation meetings and jobsite observations by weather barrier manufacturer for warranty is required prior to assembly installation.

PART 2 - PRODUCTS

- 2.1 MANUFACTURER
 - A. DuPont; 4417 Lancaster Pike, Chestnut Run Plaza 728, Wilmington, DE 19805; 1-800-44-TYVEK (8-9835); http://www.construction.tyvek.com
- 2.2 MATERIALS
 - A. Basis of Design: spunbonded polyolefin, non-woven, non-perforated, weather barrier is based upon DuPont[™] Tyvek® CommercialWrap® D and related assembly components.
 - B. Performance Characteristics:
 - 1. Air Penetration: 0.001 cfm/ft² at 75 Pa when tested in accordance with ASTME2178. Type 1 when tested in accordance with ASTM E 1677. \leq 0.04 cfm/ft @ 75 Pa when tested in accordance with ASTM E2357.
 - 2. Water Vapor Transmission: 30 perms, when tested in accordance with ASTM E 96, Method B.
 - 3. Water Penetration Resistance: 235 cm when tested in accordance with AATCC Test Method 127.

- 4. Basis Weight: 2.4 oz/yd², when tested in accordance with TAPPI Test Method T-410.
- 5. Air Infiltration Resistance: Air infiltration at >750 seconds, when tested in accordance with TAPPI Test Method T-460.
- 6. Tensile Strength: 33/41 lbs/in., when tested in accordance with ASTM D 822, Method A.
- 7. Surface Burning Characteristics: Class A, when tested in accordance with ASTM E 84. Flame Spread: 15, Smoke Developed: 25.

2.3 ACCESSORIES

- A. Seam Tape: 3" DuPont[™] Tyvek® Tape as distributed by DuPont.
- B. Fasteners: Provide fasteners to suit substrate and project conditions.
 - 1. Wrap Cap Screws: 1-5/8-inch rust resistant screw with 2-inch diameter plastic cap fasteners.
 - 2. Wrap Caps: #4 nails with large 1-inch plastic cap fasteners or 1-inch minimum plastic cap staple with leg length sufficient to achieve a minimum penetration of 5/8-inch into the wood stud.
 - 3. Masonry tap-con fasteners with Wrap Caps: 2-inch diameter plastic cap fasteners.
- C. Sealants: Provide sealants compatible with weather barrier assembly.
 - 1. Refer to Section 07 9200 Joint Sealants.
 - 2. Provide sealants that comply with ASTM C 920, elastomeric polymer sealant to maintain watertight conditions.
 - 3. Products:
 - a. DuPont[™] Commercial Sealant.
 - b. Sealants recommended by the weather barrier manufacturer.
- D. Adhesives:
 - 1. Provide adhesives recommended by weather barrier manufacturer.
 - 2. Products:
 - a. Liquid Nails® LN-109
 - b. Denso Butyl Liquid
 - c. 3M High Strength 90
 - d. SIA 655
 - e. Adhesives recommend by the weather barrier manufacturer.
- E. Primers:
 - 1. Provide flashing manufacturer recommended primer to assist in adhesion between substrate and flashing.
 - 2. Products:
 - a. 3M High Strength 90
 - b. Denso Butyl Spray
 - c. SIA 655
 - d. Permagrip 105
 - e. Primers recommended by the flashing manufacturer

- F. Flashing
 - 1. DuPont[™] FlexWrap[™] NF: Flexible membrane flashing materials for window openings and penetrations.
 - 2. DuPont[™] StraightFlash[™] VF: Dual-sided flashing membrane materials for brick mold and non-flanged windows and doors.

PART 3 - EXECUTION

3.1 EXAMINATION

A. Verify substrate and surface conditions are in accordance with weather barrier manufacturer recommended tolerances prior to installation of weather barrier and accessories.

3.2 INSTALLATION - WEATHER BARRIER

- A. Install weather barrier over exterior face of exterior wall substrate in accordance with manufacturer recommendations
- B. Install weather barrier prior to installation of windows and doors.
- C. Start weather barrier installation at a building corner, leaving 6-12 inches of weather barrier extended beyond corner to overlap.
- D. Install weather barrier in a horizontal manner starting at the lower portion of the wall surface with subsequent layers installed in a shingling manner to overlap lower layers. Maintain weather barrier plumb and level
- E. Sill Plate Interface: Extend lower edge of weather barrier over sill plate interface 3-6 inches. Secure to foundation with elastomeric sealant as recommended by weather barrier manufacturer.
- F. Window and Door Openings: Extend weather barrier completely over openings.
- G. Overlap weather barrier
 - 1. Exterior corners: minimum 12 inches.
 - 2. Seams: minimum 6 inches.
- H. Weather Barrier Attachment:
 - 1. Attach weather barrier to studs through exterior sheathing. Secure using weather barrier manufacturer recommend fasteners, space 6 -18 inches vertically on center along stud line, and 24 inches on center, maximum horizontally.
- I. Apply 4 inch by 7-inch piece of flashing or weather barrier manufacturer approved alternate to weather barrier membrane prior to the installation cladding anchors.

3.3 SEAMING

- A. Seal seams of weather barrier with seam tape at all vertical and horizontal overlapping seams.
- B. Seal any tears or cuts as recommended by weather barrier manufacturer.

3.4 OPENING PREPARATION

- A. Flush cut weather barrier at edge of sheathing around full perimeter of opening.
- B. Cut a head flap at 45-degree angle in the weather barrier at window head to expose 8 inches of sheathing. Temporarily secure weather barrier flap away from sheathing with tape.

3.5 FLASHING

- A. Cut 9-inch-wide flexible membrane flashing a minimum of 12 inches longer than width of sill rough opening.
- B. Cover horizontal sill by aligning flexible membrane flashing with inside edge of sill. Adhere to rough opening across sill and up jambs a minimum of 6 inches. Secure flashing tightly into corners by working in along the sill before adhering up the jambs.
- C. Fan flexible membrane flashing at bottom corners onto face of wall. Firmly press in place.
- D. Apply 9-inch-wide strips of dual-sided flashing membrane at jambs. Align flashing with interior edge of jamb framing. Start flashing at head of opening and lap sill flashing down to the sill.
- E. Spray-apply primer to top 6 inches of jambs and exposed sheathing.
- F. Install flashing membrane flashing at opening head using same installation procedures used at sill. Overlap jamb flashing a minimum of 2 inches.
- G. Coordinate flashing with window installation.
- H. On exterior, install backer-rod in joint between window frame and flashed rough framing. Apply sealant at jambs and head, leaving sill unsealed. Apply sealants in accordance with sealant manufacturer's instructions and ASTM C 1193.
- I. Position weather barrier head flap across head flashing. Adhere using 4-inch wide dual-sided flashing membrane over the 45-degree seams.
- J. Tape top of window in accordance with manufacturer recommendations.
- K. On interior, install backer rod in joint between frame of window and flashed rough framing. Apply sealant around entire window to create air seal. Apply sealant in accordance with sealant manufacturer's instructions and ASTM C 1193.
- 3.6 FIELD QUALITY CONTROL
 - A. Notify manufacturer's designated representative to obtain periodic observations of weather barrier assembly installation.

ARPA Renovations Girls Inc of Meriden

3.7 PROTECTION

A. Protect installed weather barrier from damage.

END OF SECTION 07 2500

SECTION 07 4640 - VINYL SIDING AND SOFFITS

PART 1 GENERAL

1.1 SECTION INCLUDES

- A. Vinyl siding.
- B. Vinyl soffits.
- C. Vinyl trim and accessories.

1.2 RELATED SECTIONS

- A. Section 06 1000 Rough Carpentry: Framing and sheathing.
- B. Section 06 1600 Sheathing
- C. Section 07 2500 Weather Barriers

1.3 REFERENCES

- A. ASTM D 256 Test Method for Determining the Pendulum Impact Resistance of Notched Specimens of Plastics.
- B. ASTM D 635 Test Method for Rate of Burning and/or Extent and Time of Burning of Self-Supported Plastics in a Horizontal Position.
- C. ASTM D 638 Test Method for Tensile Properties of Plastics.
- D. ASTM D 648 Test Method for Deflection Temperature of Plastics Under Flexural Load.
- E. ASTM D 696 Test Method for Coefficient of Linear Expansion of Plastics.
- F. ASTM D 1929 Test Method for Ignition Properties of Plastics.
- G. ASTM D 2843 Test Method for Density of Smoke from the Burning or Decomposition of Plastics.
- H. ASTM D 3679 Specification for Rigid Poly Vinyl Chloride (PVC) Siding.
- I. ASTM D 4226 Test Methods for Impact Resistance of Rigid Poly Vinyl Chloride (PVC) Building Products.
- J. ASTM E 84 Test Method for Surface Burning Characteristics of Building Materials.

1.4 SUBMITTALS

- A. Submit under provisions of Section 01300.
- B. Samples: Siding/soffit design, size, and color for approval. Submit two samples, full size x 2 foot long of each siding, soffit, and accessory type in selected colors.

- C. Manufacturer's installation instructions.
- D. Certificate: Manufacturer's certification that siding/soffit as supplied meets or exceeds the conditions specified herein.

1.5 QUALITY ASSURANCE

- A. Manufacturer: Maintain rigorous production quality control standards to ensure that vinyl siding and soffit will perform as expected for its intended use.
- B. Regulatory Requirements:
 - 1. International Building Code 2012 with CT Supplements.
 - 2. CHFA Construction Guidelines 2018.
 - 3. HUD-FHA Minimum Property Standards.
 - 4. Underwriters Laboratories Listing R14214.
 - 5. ICC ESR-1258, ESR 1133.
- 1.6 DELIVERY, STORAGE, AND HANDLING
 - A. Store products in manufacturer's unopened packaging until ready for installation.
 - B. Pack siding and soffits two squares per carton and clearly mark each carton with manufacturer's name, siding style, color, identifying lot number, and VSI Certification Stamp.
 - C. Store vinyl siding, soffits, and accessories in clean, dry area, out of direct sunlight.
 - D. Handle material to prevent damage. Do not allow cartons to crease.

1.7 WARRANTY

A. Upon completion, provide a written transferable, lifetime limited warranty.

PART 2 PRODUCTS

2.1 MANUFACTURERS

- A. Basis of Design: Alside / Siding, which is located at: P. O. Box 2010; Akron, OH 44309; Toll Free Tel: 800-922-6009; Tel: 330-922-5350; Fax: 330-922-5387; Web:<u>www.alside.com</u>
- B. Requests for substitutions will be considered in accordance with provisions of Section 01 6000.

2.2 MATERIALS

- A. Vinyl Siding and Soffits General Requirements: Produced from polyvinyl chloride (PVC) compounds meeting ASTM D 3679 requirements for compound class 2.
 - 1. Vinyl Siding Institute Certified.
 - 2. Average Impact Strength: 3.86 ft. lbs./in. of notch at 73.4 degrees F, per

Christopher Williams Architects, LLC CWA Project No.:2313 ISSUED FOR BIDDING March 12, 2024 Page 2 of 5 ASTM D 256.

- 3. Average Impact Strength: 2.4 ft. lbs./in. of notch at 32 degrees F, per ASTM D 256.
- 4. Tensile Strength: 6,700 psi (46,195 kPa), per ASTM D 638.
- 5. Modulus of Elasticity: 410,000 psi, per ASTM D 638.
- 6. Deflection Temperature: 170 degrees F, per ASTM D 648.
- 7. Fire Properties:
 - a. Average Time of Burning: Less than 5 seconds, when tested in accordance with ASTM D 635.
 - b. Average Extent of Burning: Less than 5 mm, when tested in accordance with ASTM D 635.
 - c. Flame Spread Index: 20 (Class A), when tested in accordance with ASTM E 84.
 - d. Smoke Developed Index: Less than 450, when tested in accordance with ASTM E 84.
 - e. Ignition Temperature: When tested in accordance with ASTM D 1929, no self ignition and no flaming; no smoldering at less than 680 degrees F.
- 8. Typical Vinyl Siding Properties:
 - a. Camber: Alside sidings and soffits meet all requirements for camber per ASTM D 3679.
 - b. Heat Shrinkage: Alside sidings and soffits meet all requirements for heat shrinkage per ASTM D 3679.
 - c. Impact Resistance: 60 in-lbf, per ASTM D 4226, Procedure A, H.25.
 - d. Weatherability: No surface or structural defects such as peeling, cracking, or chipping when tested per ASTM D 3679.
 - e. Coefficient of Linear Expansion: 3.05 by 10-5 in/in F, per ASTM D 696.
 - f. Gloss: Alside sidings and soffits meet all requirements.
 - g. Surface Distortion: No distortion at 120 degrees F, per ASTM D 3679.
- 9. Interlock: Post-form style lock with positive interlock; both ends of panels factory cut and notched for overlap.
- 10. Nail Slots: Elongated 1-inch slots spaced approximately 1/4 inch apart in nailing hem to allow for expansion and contraction.
- 11. Weep Holes: Small holes under the bottom butt of siding panels to prevent vapor build-up and allow accumulated moisture to escape.

2.3 SIDING

- A. Horizontal Vinyl Siding: Charter Oak triple 4 1/2-inch exposure solid soffit/vertical.
 - 1. 4 1/2-inch exposure dutch lap profile.
 - 2. Each 9-inch-wide horizontal siding panel nominally configured as two 4 1/2inch panels in dutch lap style with 3/4-inch butt height.
 - 3. Tri Beam panel reinforcement system with double thick rolled over nail hem.
 - 4. Length: 12 feet 0 inches
 - 5. Field Butt Height: 3/4 inch.
 - 6. Thickness: 0.042 inch.
 - 7. Lo-gloss finish.
 - 8. Color: As selected by Architect from manufacturer's standards.
- 2.4 SOFFIT

Christopher Williams Architects, LLC CWA Project No.:2313

- A. Vinyl Soffit: Charter Oak T3.33 inch invisibly vented, reinforced soffit.
 - 1. Each 10-inch-wide panel nominally configured as three 3.33-inch panels in both aerated and solid panels with 3/4-inch (19 mm) butt height.
 - 2. Aerated panels invisibly perforated with 3.2 sq. inch net free area per l/ft.
 - 3. TriBeam panel reinforcement system.
 - 4. Length: 12 feet 0 inches.
 - 5. Width: 10 inches.
 - 6. Thickness: 0.042 inch.
 - 7. Color: As selected by Architect from manufacturer's standards.

2.5 ACCESSORIES

- A. Wide Pocket Accessories: Refer to drawings & details for locations and applications. Accessories shall include all available manufacturer's products for the specified siding types, and include but not be limited to the following:
 - 1. 4-inch traditional outside corner post with foam insert
 - 2. 3-inch beaded corner starter
 - 3. 3.5, 5, and 7 inch lineal
 - 4. 1-1/2-inch j-channel
 - 5. Two-piece j-channel
 - 6. Wide pocket J-channel
 - 7. Prodigy starter

PART 3 EXECUTION

3.1 EXAMINATION

- A. Confirm that all critical dimensions are as specified on the drawings.
- B. Beginning installation indicates Installer's acceptance of substrate as suitable to accept siding and soffits.
- 3.2 PREPARATION
 - A. Repair substrate flaws or defects before applying siding or soffits.
 - B. Where necessary, fur surfaces to an even plane and free from obstructions before application.

3.3 INSTALLATION

- A. Install siding and soffits in accordance with the latest edition of "Vinyl Siding Installation Manual," published by the Vinyl Siding Institute (VSI) and special details from the drawings.
- B. Basic Installation Guidelines are as follows: The following rules, which are stated in the Vinyl Institute Guidelines are critical for proper vinyl siding installation.
 - 1. Installed panels and accessories must move freely from side to side.
 - 2. When installing a siding panel, push up from the bottom until the lock is fully engaged with the piece below it. Without stretching the panel, reach up and

fasten it into place

- 3. Fasten nails or other fasteners in the center of the nailing slot and make sure the fastener penetrates at least 1 ¼" through wood sheathing into framing.
- 4. Do not force the panels up or down when fastening in position.
- 5. Do not drive the head of the fastener tightly against the siding nail hem. Allow approximately 1/32" clearance between the fastener head and the siding panel. Make sure the panel can move freely back and forth. Drive fasteners straight and level to prevent distortion and buckling of the panel.
- 6. Leave a minimum of ¼" clearance at all openings and stops to allow for normal expansion and contraction. When installing in temperatures for 40 degrees, increased minimum clearance to 3/8".
- 7. Do not caulk the panels when they meet the receiver of inside corners, outside corners, or J-trim. Do not caulk the overlap joints.
- 8. Do not face-nail or staple through siding. Vinyl siding expands and contracts with outside temperature changes. Face-nailing can result in ripples in the siding.
- 9. Removal of existing siding is necessary take appropriate actions to ensure a smooth and continuous surface.
- 10. Inspect substrate and ensure sheathing is even, smooth & uninterrupted. Repair as necessary per Section 06160 – Sheathing
- 11. Installation of the specific products may differ in details from these instructions. Follow the manufacturer's instructions, using parts specified by the manufacturer, to ensure proper installation.
- C. Install vinyl siding, soffits, and accessories in accordance with best practice, with all joint members plumb and true.
- D. Fasteners shall be in accordance with the Fastener Schedule for siding attachment on the drawings.

3.4 FIELD QUALITY CONTROL

- A. After installation of siding and soffits, check entire surface for obvious flaws or defects.
- B. Replace and repair any problem areas, paying close attention to the substrate for causes of the problem.
- 3.5 CLEANING
 - A. After application of siding and soffits, clean as necessary to remove all fingerprints and soiled areas.
 - B. Upon completion of siding application, clean entire area, removing all scrap, packaging, and unused materials related to this work.

END OF SECTION 07 4640

SECTION 07 9200 - JOINT SEALANTS

PART 1 - GENERAL

1.1 CONDITIONS AND REQUIREMENTS

A. The General Conditions, Supplementary Conditions, and Division 1 – General Requirements apply.

1.2 SECTION INCLUDES

- A. Joint sealants for the following applications, including those specified by reference to this Section:
 - 1. Silicone Joint Sealants.
 - 2. Urethane Joint Sealants.
 - 3. Latex Joint Sealants.
 - 4. Butyl Sealants.
 - 5. Joint Sealant Backing.
 - 6. All other accessories as required for a complete installation.

1.3 RELATED SECTIONS

- A. Section 08 1113 Hollow Metal Doors and Frames
- B. Section 09 2900 Gypsum Board

1.4 REFERENCES

- A. American Society for Testing and Materials (ASTM):
 - 1. ASTM C834, Specification for Latex Sealing Compounds.
 - 2. ASTM C919, Practice for Use of Sealants in Acoustical Applications.
 - 3. ASTM C920, Specification for Elastomeric Joint Sealants.
 - 4. ASTM C1193, Guide for Use of Joint Sealants.
 - 5. ASTM C1248, Test Method for Staining of Porous Substrate by Joint Sealants.
 - 6. ASTM C1311, Specification for Solvent Release Sealants.
 - 7. ASTM C1330, Specification for Cylindrical Sealant Backing for Use with Cold Liquid Applied Sealants.
 - 8. ASTM E90, Method for Laboratory Measurement of Airborne Sound Transmission Loss of Building Partitions and Elements.
- B. Federal Specifications (FS):
 - 1. FS TT-S-00230C(2), Sealing Compound; Elastomeric Type, Single Component (for Calking, Sealing, and Glazing in Buildings or Other Structures).

1.5 PERFORMANCE REQUIREMENTS

- A. Provide elastomeric joint sealants that establish and maintain watertight and airtight continuous joint seals without staining or deteriorating joint substrates.
- B. Provide joint sealants for interior applications that establish and maintain airtight and water-resistant continuous joint seals without staining or deteriorating joint substrates.

1.6 SUBMITTALS

- A. Submit under provisions of Section 01 3300.
- B. Product Data: For each joint-sealant product indicated.
 - 1. Include printed statement of VOC content and material safety data sheets.
- C. Samples: For each type and color of joint sealant required, provide samples with joint sealants in 1/2-inch-wide joints formed between two 6-inch-long strips of material matching the appearance of exposed surfaces adjacent to joint sealants.
- D. Preconstruction field test reports.

1.7 QUALITY ASSURANCE

- A. Manufacturer Qualifications: Company experienced in manufacturing products similar to those specified for this project and with a record of successful in-service performance, as well as sufficient production capacity to produce required products.
- B. Preconstruction Field-Adhesion Testing: Before installing elastomeric sealants, field test their adhesion to Project joint substrates according to the method in ASTM C1193 that is appropriate for the types of Project joints. Follow manufacturer's recommendations for preparation of substrates and installation of primers.
- C. Mockups: Build mockups incorporating sealant joints, as follows, to verify selections made under sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution:
 - 1. Joints in mockups of assemblies specified in other Sections that are indicated to receive elastomeric joint sealants, which are specified by reference to this Section.

1.8 WARRANTY

- A. Special Installer's Warranty: Installer's standard form in which Installer agrees to repair or replace elastomeric joint sealants that do not comply with performance and other requirements specified in this Section within specified warranty period.
 - 1. Warranty Period: Two (2) years from date of Substantial Completion.

PART 2 - PRODUCTS

2.1 JOINT SEALANTS

- A. Low-Emitting Materials: Sealants shall comply with the following limits for VOC content:
 - 1. Architectural Sealants: 250 g/L.
 - 2. Nonmembrane Roof Sealants: 300 g/L.
 - 3. Single-Ply Roof Membrane Sealants: 450 g/L.
 - 4. Other Sealants: 420 g/L.
 - 5. Sealant Primers for Nonporous Substrates: 250 g/L.
 - 6. Sealant Primers for Porous Substrates: 775 g/L.
 - 7. Modified Bituminous Sealant Primers: 500 g/L.
 - 8. Other Sealant Primers: 750 g/L.
- B. Low-Emitting Materials:
 - 1. Exterior reactive sealants shall have a VOC content of not more than 50 g/L or 4 percent by weight, whichever is greater.
 - 2. Other exterior caulks and sealants shall have a VOC content of not more than 30 g/L or 2 percent by weight, whichever is greater.
 - 3. Interior sealants shall comply with the testing and product requirements of the California Department of Health Services' "Standard Practice for the Testing of Volatile Organic Emissions from Various Sources Using Small-Scale Environmental Chambers."
- C. Compatibility: Provide joint sealants, joint fillers, and other related materials that are compatible with one another and with joint substrates under service and application conditions.

2.2 EXTERIOR JOINT SEALANTS

- A. Exterior Movement Joints in Brick or Stone Masonry:
 - 1. Single-component, neutral-curing, non-staining silicone sealant, ASTM C920, Type S; Grade NS; Class 50; for Use NT.
 - a. Products: One of the following:
 - 1) Tremco Inc; Spectrem 2 or Spectrem 3.
 - 2) Pecora Corporation; 864NST or 895NST.
 - 3) Dow Corning Corporation; DOWSIL 795.
- B. Exterior Joints Between Dissimilar Materials and All Other Non-Traffic Joints Where Another Type Is Not Specified:
 - 1. Single-Component neutral-curing, non-staining silicone sealant. ASTM C920, Type S, Grade NS, Class 50, foe Use NT.
 - a. Products: One of the foillowing:

- 1) Tremco Inc; Spectrem 1 or Spectrem 2.
- 2) Pecora Corporation; 890NST.
- 3) Dow Corning Corporation; DOWSIL 791.
- C. Sealant for Exterior Traffic-Bearing Joints, Where Slope Precludes Use of Pourable Sealant:
 - 1. Single-component, nonsag urethane sealant, ASTM C920, Type S; Grade NS; Class 25; for Use T.
 - a. Products: One of the following:
 - 1) Tremco Inc; Vulkem 116.
 - 2) BASF Building Systems; Sonolastic NP1.
 - 3) Sika Corporation, Construction Products Division; Sikaflex 1a.
- D. Sealant for Exterior Traffic-Bearing Joints, Where Slope Allows Use of Pourable Sealant:
 - 1. Single-component, pourable urethane sealant, ASTM C920, Type S; Grade P; Class 25; for Use T.
 - a. Products: One of the following:
 - 1) Tremco Inc; Vulkem 45.
 - 2) Pecora Corporation; Urexpan NR-201.
 - 3) BASF Building Systems; Sonolastic SL 1.

2.3 INTERIOR JOINT SEALANTS

- A. Sealant for Use in Interior Joints in Ceramic Tile and Other Hard Non-Porous Surfaces in Kitchens and Toilet Rooms and Around Plumbing Fixtures:
 - 1. Single-component, mildew-resistant silicone sealant, ASTM C920, Type S; Grade NS; Class 25; for Use NT; formulated with fungicide.
 - a. Products: One of the following:
 - 1) Tremco Incorporated; Tremsil 200 Sanitary.
 - 2) Pecora Corporation; 898.
 - 3) Dow Corning Corporation; DOWSIL 786.
 - 4) GE; Sanitary SCS1700.
- B. Sealant for Interior Use Where Another Type is not Specified:
 - 1. Acrylic latex or siliconized acrylic latex, ASTM C834, Type OP, Grade NF.
 - a. Products: One of the following:

- 1) Tremco Inc; Tremflex 834.
- 2) Pecora Corporation; AC-20+.
- 3) BASF Building Systems; Sonolac.

2.4 SEALANT FOR BEDDING DOOR THRESHOLDS

- A. Butyl-Rubber Based Sealant for Bedding Door Thresholds:
 - 1. ASTM 1311 cne-part butyl rubber sealant ASTM C1311
 - a. Products: One of the following;
 - 1) Tremco, Inc; Tremco Butyl Sealant.
 - 2) Pecora Corp; BC-158
 - 3) Dow Chemical Co; DOWSIL 335
- B. Sealant Colors: As selected by Architects from manufacturer's full range of colors.

2.5 JOINT-SEALANT BACKING

- A. General: Provide sealant backings of material and type that are nonstaining; are compatible with joint substrates, sealants, primers, and other joint fillers; and are approved for applications indicated by sealant manufacturer based on field experience and laboratory testing.
- B. Cylindrical Sealant Backings: ASTM C1330, Type C (closed-cell material with a surface skin), O (open-cell material), B (bicellular material with a surface skin), or any of the preceding types, as approved in writing by joint-sealant manufacturer for joint application indicated, and of size and density to control sealant depth and otherwise contribute to producing optimum sealant performance:
- C. Bond-Breaker Tape: Polyethylene tape or other plastic tape recommended by sealant manufacturer for preventing sealant from adhering to rigid, inflexible joint-filler materials or joint surfaces at back of joint where such adhesion would result in sealant failure. Provide self-adhesive tape where applicable.

2.6 MISCELLANEOUS MATERIALS

- A. Primer: Material recommended by joint-sealant manufacturer where required for adhesion of sealant to joint substrates indicated, as determined from preconstruction joint-sealant-substrate tests and field tests.
- B. Cleaners for Nonporous Surfaces: Chemical cleaners acceptable to manufacturers of sealants and sealant backing materials, free of oily residues or other substances capable of staining or harming joint substrates and adjacent nonporous surfaces in any way, and formulated to promote optimum adhesion of sealants to joint substrates.
- C. Masking Tape: Nonstaining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.

PART 3 - EXECUTION

3.1 PREPARATION

- A. Surface Cleaning of Joints: Clean out joints immediately before installing joint sealants.
 - 1. Remove all foreign material from joint substrates that could interfere with adhesion of joint sealant.
 - a. Clean porous joint substrate surfaces by brushing, grinding, blast cleaning, mechanical abrading, or a combination of these methods to produce a clean, sound substrate capable of developing optimum bond with joint sealants. Remove loose particles remaining after cleaning operations above by vacuuming or blowing out joints with oil-free compressed air.
 - 2. Remove laitance and form-release agents from concrete.
 - a. Clean nonporous surfaces with chemical cleaners or other means that do not stain, harm substrates, or leave residues capable of interfering with adhesion of joint sealants.
- B. Joint Priming: Prime joint substrates, where recommended in writing by joint-sealant manufacturer and based on preconstruction joint-sealant-substrate tests or prior experience. Apply primer to comply with joint-sealant manufacturer's written instructions. Confine primers to areas of joint-sealant bond; do not allow spillage or migration onto adjoining surfaces.
- C. Masking Tape: Use masking tape where required to prevent contact of sealant with adjoining surfaces that otherwise would be permanently stained or damaged by such contact or by cleaning methods required to remove sealant smears. Remove tape immediately after tooling without disturbing joint seal.

3.2 INSTALLATION

- A. Sealant Installation Standard: Comply with recommendations in ASTM C1193 for use of joint sealants as applicable to materials, applications, and conditions indicated.
- B. Acoustical Sealant Application Standard: Comply with recommendations in ASTM C919 for use of joint sealants in acoustical applications as applicable to materials, applications, and conditions indicated.
- C. Install sealant backings of type indicated to support sealants during application and at position required to produce cross-sectional shapes and depths of installed sealants relative to joint widths that allow optimum sealant movement capability.
 - 1. Do not leave gaps between ends of sealant backings.
 - 2. Do not stretch, twist, puncture, or tear sealant backings.

- 3. Remove absorbent sealant backings that have become wet before sealant application and replace them with dry materials.
- D. Install bond-breaker tape behind sealants where sealant backings are not used between sealants and backs of joints.
- E. Install sealants using proven techniques that comply with the following and at the same time backings are installed:
 - 1. Place sealants so they directly contact and fully wet joint substrates.
 - 2. Completely fill recesses in each joint configuration.
 - 3. Produce uniform, cross-sectional shapes and depths relative to joint widths that allow optimum sealant movement capability.
- F. Tooling of Nonsag Sealants: Immediately after sealant application and before skinning or curing begins, tool sealants according to requirements specified below to form smooth, uniform beads of configuration indicated; to eliminate air pockets; and to ensure contact and adhesion of sealant with sides of joint.
 - 1. Remove excess sealant from surfaces adjacent to joints.
 - 2. Use tooling agents that are approved in writing by sealant manufacturer and that do not discolor sealants or adjacent surfaces.
 - 3. Provide concave joint configuration per Figure 5A in ASTM C1193, unless otherwise indicated.
- G. Installation of Preformed Silicone-Sealant System: Comply with manufacturer's written instructions.
- H. Installation of Preformed Foam Sealants: Install each length of sealant immediately after removing protective wrapping, taking care not to pull or stretch material, producing seal continuity at ends, turns, and intersections of joints. For applications at low ambient temperatures where expansion of sealant requires acceleration to produce seal, apply heat to sealant in compliance with sealant manufacturer's written instructions.
- I. Clean off excess sealant or sealant smears adjacent to joints as the Work progresses by methods and with cleaning materials approved in writing by manufacturers of joint sealants and of products in which joints occur.

END OF SECTION 07 9200

SECTION 08 1113 - HOLLOW METAL DOORS AND FRAMES

PART 1 - GENERAL

1.1 SECTION REQUIREMENTS

A. Submittals: Product Data and Shop Drawings indicating sizes, profiles and details for doors, frames and anchors.

PART 2 - PRODUCTS

2.1 HOLLOW METAL DOORS AND FRAMES

- A. Manufacturers: Provide Doors and Frames by one of the following:
 - 1. Ceco Door Products; an Assa Abloy Group company.
 - 2. Curries Company; an Assa Abloy Group company.
 - 3. Pioneer Industries, Inc.
 - 4. Steelcraft; an Ingersoll-Rand company.
- B. Doors: Complying with SDI A250.8 for level and model and SDI A250.4 for physicalendurance level indicated, 1-3/4 inches (44 mm) thick unless otherwise indicated.
 - 1. Exterior Doors: Performance Level A (Extra Heavy Duty), minimum 16-gauge, Model 2 Seamless. Metallic coating, designation A60.
 - 2. Design: Flush Panel.
 - 3. Core Construction: Foamed in place polyurethane and steel stiffened laminated core with no stiffener face welds, in compliance with HMMA 867 "Laminated Core".
 - Provide 22-gauge steel stiffeners at 6 inches on-center internally welded at 5" on- center to integral core assembly. No stiffener face welding is permitted.
 - b. Thermal properties to rate at a fully operable minimum U-Factor 0.37 and R-Value 2.7, including insulated door, thermal-break frame and threshold.
 - 4. Hardware Reinforcement: Fabricate according to SDI A250.6 with reinforcement plates from same material as door face sheets.
- C. Frames: ANSI A250.8; conceal fastenings unless otherwise indicated.
 - 1. Steel Sheet for Exterior Frames: 14-gauge minimum thickness, ASTM A653, metallic coating designation A60.
 - 2. Exterior Frame Construction: Full profile welded.
 - 3. Hardware Reinforcement: Fabricate according to ANSI/SDI A250.6 with reinforcement plates from same material as frames.

- 4. Frame Anchors: Not less than 0.042 inch thick.
- D. Door Silencers: Three on strike jambs of single-door frames and two on heads of double-door frames.
- E. Grout Guards: Provide where mortar might obstruct hardware operation.
- F. Prepare doors and frames to receive mortised and concealed hardware according to SDI A250.6 and BHMA A156.115.
- G. Reinforce doors and frames to receive surface-applied hardware.
- H. Prime Finish: Manufacturer's standard, factory-applied coat of lead- and chromate-free primer complying with SDI A250.10 acceptance criteria.

2.2 MATERIALS

- A. Metallic-Coated Steel Sheet: ASTM A 653/A 653M, A60.
- B. Frame Anchors: ASTM A 879/A 879M, 4Z coating designation; mill phosphatized.
- C. Inserts, Bolts, and Fasteners: Hot-dip galvanized according to ASTM A 153/A 153M.

PART 3 - EXECUTION

3.1 INSTALLATION

- A. Install hollow metal frames to comply with SDI A250.11.
- B. Install doors to provide clearances between doors and frames as indicated in SDI A250.11.
- C. Prime-Coat Touchup: Immediately after erection, sand smooth rusted or damaged areas of prime coat and apply touchup of compatible air-drying rust-inhibitive primer. Use galvanizing repair paint for metallic coated surfaces.

END OF SECTION 08 1113

SECTION 08 3113 - ACCESS DOORS AND FRAMES

PART 1 - GENERAL

1.1 SECTION REQUIREMENTS

A. Submittals: Product Data.

PART 2 - PRODUCTS

2.1 PERFORMANCE REQUIREMENTS

- A. Fire-Rated Access Doors and Frames: Labeled by a testing and inspecting agency acceptable to authorities having jurisdiction based on testing per the following:
 - 1. Horizontal Access Doors and Frames: NFPA 288.

2.2 ACCESS DOORS AND FRAMES FOR WALLS AND CEILINGS

- A. Manufacturers: One of the following:
 - 1. Acudor Products, Inc.
 - 2. Babcock-Davis.
 - 3. Karp Associates, Inc.
 - 4. Milcor Inc.
- B. Flush Access Doors with Exposed Flanges: Prime-painted steel units.
- C. Flush Access Doors with Concealed Flanges: Prime-painted steel units with drywall bead flange.
- D. Locks: Flush to finished surface, key operated.

2.3 MATERIALS

A. Steel Sheets: ASTM A 1008/A 1008M or ASTM A 591/A 591M.

PART 3 - EXECUTION

3.1 INSTALLATION

A. Install access doors and panels accurately in position. Adjust hardware and door and panels for proper operation.

END OF SECTION 08 3113

Christopher Williams Architects, LLC CWA Project No.:2313

SECTION 08 7100 - DOOR HARDWARE

- PART 1 GENERAL
- 1.1 SUMMARY
 - A. Provide the following:
 - 1. Replacement lockset for existing exterior entry door at Rear Porch, including field verification, preparation and modification of existing door and frame to receive new door hardware.
 - 2. Hardware for hollow metal door and frame at Basement.
- 1.2 RELATED SECTIONS
 - A. Section 08 1113 Hollow Metal Doors and Frames
- 1.3 SECTION REQUIREMENTS
 - A. Submittals:
 - 1. Product data for all specified hardware components.
 - 2. Hardware schedule.
- 1.4 QUALITY ASSURANCE
 - A. Single Source Responsibility: Obtain each type of door hardware from single manufacturer.
 - B. Accessibility Requirements: Door hardware on doors in an accessible route, comply with governing accessibility regulations.

PART 2 - PRODUCTS

2.1 HARDWARE

- A. Hinges:
 - 1. Manufacturers: One of the following:
 - a. IVES: 5BB Series
 - b. McKinney Products Company; an ASSA ABLOY Group company.
 - 2. Stainless-steel hinges with stainless-steel pins for exterior doors.
 - 3. Ball-bearing hinges for doors with closers and entry doors.
 - 4. Three hinges for 1-3/4-inch-thick doors 90 inches or less in height; four hinges for doors more than 90 inches in height.

- B. Mortise Locksets:
 - 1. Manufacturers: One of the following:
 - a. Sargent 8200 Series
 - b. Schlage L9000 Series
 - 2. BHMA A156.13, Series 1000, Grade 1.
 - 3. Lever handles.
- C. Cylindrical Locks
 - 1. Manufacturers: One of the following:
 - a. Sargent 10 Line
 - b. Schlage ND Series
 - 2. BHMA A156.2, Series 4000, Grade 1.
 - 3. Lever Handles.
- D. Key locks to Owner's existing master-key system.
- E. Closers
 - 1. Manufacturers: One of the following:
 - a. SARGENT 351 Series
 - b. LCN 4040 Series
 - 2. Mount closers on interior side (room side) of door opening. Provide regular-arm, parallel-arm, or top-jamb-mounted closers as necessary.
 - 3. Adjustable delayed opening (accessible to people with disabilities) feature on closers.
- F. Provide wall stops or floor stops for doors without closers.
- G. Thresholds, seals doors sweeps and gasketing.
 - 1. Manufacturers: One of the following:
 - a. Pemko
 - b. Zero
 - 2. Thermal Break
- H. Silencers
 - 1. Manufacturers:
 - a. lves
 - b. Burns, Rockwood
 - 2. Provide "push-in" type silencers for hollow metal frames.

- 3. Provide one silencer per 30 inches of height on each single frame, and two for each pair frame.
- 4. Omit where gasketing is specified.
- I. Hardware Finishes:
 - 1. Hinges at Exterior Doors: BHMA 630 (US32D)
 - 2. Continuous Hinges: BHMA 630 (US32D)
 - 3. Protection Plates: BHMA 630 (US32D)
 - 4. Door Closers: Powder Coat to Match
 - 5. Wall Stops: BHMA 630 (US32D)
 - 6. Weatherstripping: Clear Anodized Aluminum
 - 7. Thresholds: Mill Finish Aluminum

PART 3 - EXECUTION

- A. EXAMINATION
 - 1. Field verify existing door and frame receiving new hardware. Verify that new hardware is compatible with existing door and frame preparation and existing conditions.
 - 2. Field verify size and conditions of existing masonry opening receiving new door and frame.

B. PREPARATION

- 1. Where on-site modification of doors and frames is required:
 - a. Field modify and prepare existing door and frame for new hardware being installed.
 - b. When modifications are exposed to view, use concealed fasteners, when possible.
 - c. Prepare hardware locations and reinstall in accordance with installation requirements for new door hardware.

3.2 INSTALLATION

- A. Mount hardware in locations required to comply with governing regulations and according to SDI A250.8 and DHI WDHS.3.
- B. Install each hardware item in compliance with manufacturer's instructions and recommendations, using only fasteners provided by manufacturer.
- C. Do not install surface mounted items until finishes have been completed on substrate. Protect all installed hardware during painting.
- D. Set units level, plumb and true to line and location. Adjust and reinforce attachment substrate as necessary for proper installation and operation.

- E. Drill and countersink units that are not factory prepared for anchorage fasteners. Space fasteners and anchors according to industry standards.
- F. Install operating parts so they move freely and smoothly without binding, sticking, or excessive clearance.
- G. Hinges: Install types and in quantities indicated in door hardware schedule but not fewer than quantity recommended by manufacturer for application indicated or one hinge for every 30 inches of door height, whichever is more stringent, unless other equivalent means of support for door, such as spring hinges or pivots, are provided.
- H. Lock Cylinders: Install construction cores to secure building and areas during construction period.
- I. Door Closers: Mount closers on room side of corridor doors, inside of exterior doors, and stair side of stairway doors from corridors. Mount closers so they are not visible in corridors, lobbies and other public spaces unless approved by Architect.
- J. Closer/Holders: Mount closer/holders on room side of corridor doors, inside of exterior doors, and stair side of stairway doors.
- K. Thresholds: Set thresholds in full bed of butyl sealant.
- L. Stops: Provide floor stops for doors unless wall or other type stops are indicated in door hardware schedule. Do not mount floor stops where they may impede traffic or present tripping hazard.
- M. Perimeter Gasketing: Apply to head and jamb, forming seal between door and frame.
- N. Deliver keys to Owner.

3.3 HARDWARE SCHEDULE

A. Hardware Set No. 1 (Basement door to bulkhead):

3	Hinges	T4A3386	McKinney
1	Storeroom Lock	72 8204 LNP	Sargent
1	Surface Closer	351 PD10	Sargent
1	Threshold:	2010APK	Pemko
1	Head & Jamb Gasketing	2891 APK	Pemko
1	Sill Sweep	315	Pemko

- B. Hardware Set No. 2:
 - 1 Classroom Security Lock with Lever Handles (Bored or mortise type to suit existing door cutout field verify)

END OF SECTION 08 7100

SECTION 09 0320 – PLASTER REPAIR

PART 1 - GENERAL

1.1 SUMMARY

- A. Section Includes:
 - 1. Repair and replacement of interior plaster.
 - 2. Three elements of Repair scope are part of the Base Bid:
 - a. Restoring the wall surface integrity at holes or penetrations made in the interior face of exterior walls where finishes are identified to remain.
 - b. Joining new replacement finishes on the exterior walls to adjacent existing finishes.
 - c. Restoring the wall surface integrity at holes made for the convenience of other trades to access work concealed by finishes.
- B. Predominant non-wood ceiling finishes are believed to be Gypsum Wall Board. Ceiling repair techniques for plaster are described in the event that extant horizontal plaster is uncovered. Where replacement finishes are required, gypsum plaster is specified.

1.2 PREINSTALLATION MEETINGS

- A. Preinstallation Conference at Project site.
 - 1. Review locations, methods, and procedures.
 - a. Differentiate between "old "plaster and surfaces with features or treatments that qualify as having historic interest.
 - b. Verify personnel, equipment, and facilities needed to make progress and avoid delays.
 - c. Confirm materials, application techniques, colors, patterns, and sequencing.
 - d. Fire-protection plan.
 - e. Fugitive dust control
 - f. Plasterwork historic treatment program where historic plaster is affected.
 - g. Coordination with other trades and building occupants.

1.3 SEQUENCING AND SCHEDULING

- A. Perform plaster repair in the following sequence, which includes work specified in this and other Sections:
 - 1. Remove existing surface-mounted objects and hardware that overlie plaster surfaces except items indicated to remain in place. Tag items with location identification and protect.
 - 2. Verify that temporary protection has been installed.
 - 3. Examine condition of substrate, available grounds, and mating plaster surfaces.
 - 4. Clean plaster surface and remove paint and other finishes to the extent required.

- 5. Repair and replace existing plaster and supports to the degree required for a uniform, tightly adhered surface on which to apply other finishes.
- 6. Cure repaired surfaces and allow them to dry for proper finishing.
- 7. Paint and apply other finishes.
- 8. Reinstall removed surface-mounted objects and hardware unless otherwise indicated.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product.
 - 1. Include recommendations for product application and use.
- B. Shop Drawings:
 - 1. Flat-work repairs: use photographs and interior elevation drawings to document, delineate, and quantify repairs. Include a list or schedule to correlate with Unit Prices.

1.5 INFORMATIONAL SUBMITTALS

- A. Qualification Data:
 - 1. For Contractor, list comparable recent projects
 - 2. For plaster repair mechanics, identify comparable work performed.
 - 3. Plasterwork Historic Treatment Program: Indicate whether Contractor or labor force has ever participated in work controlled by a PHTP

1.6 QUALITY ASSURANCE

- A. Mockups: Prepare mockups in place for each type of plaster repair and reconstruction work to demonstrate aesthetic effects and to set quality standards for materials and execution and for fabrication and installation.
 - 1. Locate mockups on existing surfaces where directed by Architect.
 - 2. Simulate finished lighting conditions for review of mockups.
 - 3. Approval of mockups does not constitute approval of deviations from the Contract Documents contained in mockups unless Architect specifically approves such deviations in writing.
 - 4. Subject to compliance with requirements, approved mockups may become part of the completed Work if undisturbed at time of Substantial Completion.

1.7 DELIVERY, STORAGE, AND HANDLING

- A. Deliver packaged materials to Project site in manufacturer's original and unopened containers, labeled with manufacturer's name and type of products.
- B. Store materials on elevated platforms, under cover, and in a dry location with ambient temperatures continuously maintained at not less than 45 deg F.

- C. Store hydrated lime and factory-prepared lime putty in manufacturer's original and unopened containers. Discard lime if containers have been damaged or have been opened for more than two days.
- D. Store materials not in use in tightly covered containers.
- E. Store lime putty covered with water in sealed containers.
- F. Store sand where grading and other required characteristics can be maintained and contamination avoided.
- 1.8 FIELD CONDITIONS
 - A. Comply with plaster-material manufacturers' written instructions. For gypsum plaster, also comply with ASTM C 842 requirements.
 - B. Temperatures: Maintain temperatures in work areas at not less than 55 deg F or greater than 80 deg F for at least seven days before application of plaster, continuously during application, and for seven days after plaster has set or until plaster has dried.
 - C. Avoid conditions that result in plaster drying out too quickly.
 - 1. Distribute heat evenly; prevent concentrated or uneven heat on plaster.
 - 2. Maintain relative humidity levels for prevailing ambient temperature that produce normal drying conditions.
 - 3. Ventilate work areas in a manner that prevents drafts of air from contacting surfaces during plaster application and until plaster is dry.

PART 2 - PRODUCTS

2.1 LIME-PLASTER MATERIALS

- A. Hydrated Lime: ASTM C 206, Type S or Type N.
- B. Lime Putty: Slaked hydrated lime or factory-prepared lime putty according to ASTM C 1489.
- C. Sand Aggregates: ASTM C 897.
 - 1. Finish-Coat Sand: Match size, texture, and gradation of existing sand as closely as possible. Blend several sands if necessary to achieve suitable match.
- D. Pigments for Colored Plaster: ASTM C 979/C 979M and having a record of satisfactory performance in lime plaster.
- E. Fiber: 1/2 to 1 inch in length; composed of alkali-resistant glass or polypropylene fiber; free of grease, waxes, and oils; and beaten well to separate fibers before blending into unfibered plaster material.

- 1. Proportion of Fiber to Unfibered Plaster Material: 3.5 oz./cu. ft.of unfibered plaster material, adjusted as required to produce a well-dispersed, cohesive, spreadable, stiff mix with fibers uniformly distributed.
- F. Fabric Reinforcing: Coarse, open-weave, alkali-resistant fiberglass or polypropylene fabric; free of grease, waxes, and oils.

2.2 GYPSUM PLASTER MATERIALS

- A. Gypsum Materials:
 - 1. Lightweight Gypsum Ready-Mixed Plaster: ASTM C 28/C 28M, with millmixed perlite aggregate.
 - 2. Gypsum Neat Plaster: ASTM C 28/C 28M for use with job-mixed aggregates.
 - 3. Gypsum Wood-Fibered Plaster: ASTM C 28/C 28M.
 - 4. High-Strength Gypsum Neat Plaster: ASTM C 28/C 28M; with a minimum, average, dry compressive strength of 2800 psi per ASTM C 472 for a mix of 100 lb of plaster and 2 cu. ft. of sand.
 - 5. Gypsum Gaging Plaster. ASTM C 28/C 28M.
 - 6. High-Strength Gypsum Gaging Plaster: ASTM C 28/C 28M; with a minimum, average, dry compressive strength of 5000 psi per ASTM C 472 for a neat mix.
 - 7. Gypsum Ready-Mixed Finish Plaster: ASTM C 28/C 28M; manufacturer's standard, mill-mixed, gaged, interior finish.
 - 8. Gypsum Keene's Cement: ASTM C 61/C 61M.
- B. Hydrated Lime: ASTM C 206, Type S or Type N.
- C. Aggregates:
 - 1. Aggregate for Base-Coat Plasters: ASTM C 35, perlite.
- D. Fiber: 1/2 to 1 inch in length; composed of glass or polypropylene fiber; free of grease, waxes, and oils; and beaten well to separate fibers before blending into unfibered plaster material.
 - 1. Proportion of Fiber to Unfibered Plaster Material: 3.5 oz./cu. ft.of unfibered plaster material, adjusted as required to produce a well-fibered, cohesive, spreadable, stiff mix with fibers uniformly distributed.
- E. Fabric Reinforcing: Coarse, open-weave, alkali-resistant fiberglass or polypropylene fabric; free of grease, waxes, and oils.
- F. Bonding Compound: ASTM C 631.

2.3 LATH

- A. Wood Lath: 1/4 inch by 1-1/4 inch sound, straight-grained, wood strips.
- B. Metal Lath:
 - 1. Expanded-Metal Lath: ASTM C 847, cold-rolled carbon-steel sheet, ASTM A 653/A 653M, G60, hot-dip galvanized zinc coated.
 - a. Paper Backing: Kraft paper factory bonded to back of lath.

b. Diamond-Mesh Lath: Flat, 2.5 lb/sq. yd.

2.4 TRIM ACCESSORIES

- A. General: According to ASTM C 841 for gypsum plaster; coordinate depth of trim and accessories with thicknesses and number of plaster coats required.
- B. Metal Accessories:
 - 1. Cornerite: Fabricated from expanded-metal lath with ASTM A 653/A 653M, G60, hot-dip galvanized zinc coating.
 - 2. Striplath: Fabricated from expanded-metal lath with ASTM A 653/A 653M, G60, hot-dip galvanized zinc coating.
 - 3. Cornerbeads: Fabricated from zinc-coated (galvanized) steel.
 - a. Small nose cornerbead with expanded flanges; use unless otherwise indicated.
 - b. Small nose cornerbead with perforated flanges; use on curved corners.
 - c. Small nose cornerbead with expanded flanges reinforced by perforated stiffening rib; use on columns and for finishing unit masonry corners.
 - d. Large nose corner bead with perforated flanges use on wall/window jamb transitions.
 - 4. Casing Beads: Fabricated from zinc; square-edged style; with expanded flanges.
 - Control Joints: Fabricated from zinc or zinc-coated (galvanized) steel; onepiece-type, folded pair of unperforated screeds in M-shaped configuration; with perforated flanges and removable protective tape on plaster face of control joint.
 - 6. Expansion Joints: Fabricated from zinc or zinc-coated (galvanized) steel; folded pair of unperforated screeds in M-shaped configuration; with expanded flanges.

2.5 MISCELLANEOUS MATERIALS

- A. Water for Mixing and Finishing Plaster: Potable and free of substances capable of affecting plaster set or of damaging plaster, lath, or accessories.
- B. Fasteners for Attaching Lath to Substrates:
 - 1. For Lime Plaster: ASTM C 1063.
 - 2. For Gypsum Plaster: ASTM C 841.
 - 3. For Wood Lath: ASTM C 841 requirements for wood-floor-runner or woodfurring fasteners unless otherwise indicated on Drawings.
- C. Wire Ties: ASTM A 641/A 641M, Class 1 zinc coating, soft temper, not less than 0.0475-inch diameter, unless otherwise indicated.
- D. Plaster-Stabilization Materials: Acrylic emulsion(s) and related installation products shall have proven effectiveness in reattaching delaminated plaster and shall have been used previously by installing Contractor with successful results.
- 1. Acrylic Emulsion(s), General: Aqueous emulsion(s) of acrylic polymer, adhesive to plaster and plaster substrates, nontoxic, and non-reemulsifiable after curing.
- 2. Prewet Solution: Low-viscosity acrylic emulsion.
- 3. Adhesive: Thickened acrylic emulsion; thickener as recommended in writing by resin manufacturer.
- E. Other Products: Select materials and methods of use based on the following, subject to approval of a mockup:
 - 1. Previous effectiveness in performing the work involved.
 - 2. Little possibility of damaging exposed surfaces.
 - 3. Consistency of each application.
 - 4. Uniformity of the resulting overall appearance.
 - 5. Do not use products or tools that could do the following:
 - a. Remove, alter, or in any way harm the present condition or future preservation of existing surfaces, including surrounding surfaces not in contract.
 - b. Leave an unintended residue on surfaces.

PART 3 - EXECUTION

3.1 HISTORIC TREATMENT OF PLASTER, GENERAL

- A. Appearance Standard: Completed work is to have a uniform appearance as viewed by Architect from building interior at 5 feet away from surface.
- B. Wall surfaces are anticipated to be lime plaster and will be repaired in-kind. Replication of ornamental work, if any, will be gypsum plaster.
- C. General: In treating plaster, disturb it as minimally as possible and as follows unless otherwise indicated:
 - 1. Cut loose, damaged, or deteriorated plaster, lath, and support systems that cannot be repaired.
 - 2. Verify extent of plaster deterioration or removals against that indicated on Drawings. Consult Architect on types and extent of required work.
 - 3. Verify that substrate surface conditions are suitable for repairs.
 - 4. Provide lath, furring, and support systems for plaster included in the work of this Section.
 - 5. Replace lost details in new, wet-applied and cast plaster that replicate existing or indicated plaster configurations.
 - 6. Leave repaired plasterwork in proper condition for painting or applying other finishes as indicated.
 - 7. Install temporary protective measures to protect historic surfaces that shall be treated later.
- D. Illumination: Perform plastering work with adequate, uniform illumination that does not distort the flatness or curvature of surfaces.

3.2 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, for compliance with requirements for substrate and environmental conditions, installation tolerances, and other conditions affecting performance of the Work.
 - 1. If existing substrates cannot be prepared to an acceptable condition for plastering work, notify Architect in writing.
 - 2. Notify Architect of undocumented detrimental conditions including cracks, bulges, loose backup, rotted wood, rusted metal, and other deteriorated items.
- B. Masonry Substrates: Verify that mortar joints are struck flush. Notify Architect of undocumented masonry substrate without flush joints. Proceed with plastering as directed by Architect.
- C. Begin plastering work only after unsatisfactory conditions have been corrected.
- 3.3 PREPARATION FOR PLASTERING
 - A. Substrates: Prepare according to plaster manufacturer's written instructions and as follows:
 - 1. Clean surfaces to remove dust, loose particles, grease, oil, incompatible curing compounds, form-release agents, and other foreign matter and deposits that could impair bond with plaster.
 - 2. Remove ridges and protrusions greater than 1/8 inch and fill depressions greater than 1/4 inch with patching material. Allow to set and dry.
- 3.4 PLASTER REMOVAL AND REPLACEMENT, GENERAL
 - A. Cut plaster that is damaged or deteriorated to the limits indicated. Carefully cut areas along straight edges that lie over supports, without damaging surrounding plasterwork.
 - B. Maintain lath and supporting members in an undamaged condition so far as practicable. Cut damaged lath and supports that cannot be repaired or resecured and replace with new work of same type.
 - C. Notify Architect of undocumented detrimental conditions including cracks, bulges, loose backup, rotted wood, rusted metal, and other deteriorated items.
 - D. Do not deviate more than plus or minus 1/8 inch in 10 feet from a true plane in finished plaster surfaces, as measured by a 10-footstraightedge placed on surface.
 - E. Clean substrate surfaces to remove grease, waxes, oils, waterborne staining, debris, and other foreign matter and deposits that could impair bond with repair material.
 - F. Wet substrate before plaster application with latex acrylic bonding agent. Keep substrate damp to the touch but without visible droplets.

- G. Dampen the face of plaster abutting the replacement plaster before installing new plasterwork.
- H. Finish plaster flush with metal frames and other built-in metal items or accessories that act as a plaster ground unless otherwise indicated. Where casing bead does not terminate plaster at metal frame, cut base coat free from metal frame before plaster sets and groove finish coat at junctures with metal.
- I. Provide plaster surfaces that are ready to receive field-applied finishes indicated.

3.5 FLAT LIME-PLASTER REMOVAL AND REPLACEMENT

- A. General: Cut deteriorated plaster to existing sound plaster at locations indicated on Drawings.
 - 1. Inspect for lath deterioration. If any, replace lath.
 - 2. Sand bonding surfaces of repair area, and clean the surface with a nonmetallic bristle brush.
 - 3. Wet substrate to damp condition with latex acrylic bonding agent, but without visible droplets, then install new plaster to original profiles.
- B. Lime-Plaster Base Coats:
 - 1. Scratch Coat: 1 part lime putty, 2-1/2 parts base-coat sand, and fiber. Add fiber to mix and evenly distribute it without clumps just before spreading.
 - 2. Brown Coat: 1 part lime putty, 3 parts base-coat sand.
- C. Lime-Plaster Finish Coats:
 - 1. Finish-Coat Mix for Smooth-Troweled Finish: As required to match finish of design reference sample for estimating purposes, assume 3 parts lime putty, 1 part finish-coat sand.
 - 2. Finish-Coat Mix for Smooth-Float Finish: 1 part lime putty, 1 part finish-coat sand.
- D. Lime-Plaster Finishes: Match finish(es) of design reference sample(s).
 - 1. Provide smooth-troweled finish where indicated. Apply in three layers to match adjacent existing plaster thickness.
 - 2. Provide smooth-float finish where indicated. Apply in three layers to match adjacent surfaces.
- E. Hairline cracking within the plaster or plaster separation at edge of a replacement is unacceptable. Completely remove such work and reinstall or repair as a crack repair.

3.6 FLAT GYPSUM-PLASTER REMOVAL AND REPLACEMENT

- A. General: Cut deteriorated plaster to existing sound plaster. Use replacement plaster mixes of gypsum, lime, and aggregate; and application according to ASTM C 842 unless otherwise indicated.
 - 1. Inspect for lath deterioration. If any, replace lath.

- 2. Sand or scratch-brush bonding surfaces of repair area, and clean the surface with a nonmetallic bristle brush.
- 3. Saturate substrate with acrylic latex bonding agent without visible water droplets, then install new plaster to original profiles. Refresh bonding agent with each coat.
- B. Bonding Compound: Apply on brick and plaster bases.
- C. Gypsum-Plaster Base Coats:
 - 1. Base Coats over Wood Lath: Gypsum lightweight ready-mixed plaster with fiber.
 - 2. Base Coats over Expanded-Metal Lath: High-strength gypsum plaster with job-mixed sand for scratch and brown coats. Add fiber to scratch coat.
 - 3. Base Coats over Unit Masonry: lightweight ready-mixed plaster.
- D. Gypsum-Plaster Finish Coats:
 - 1. Finish-Coat Mix for Smooth-Troweled Finishes: High-strength gypsum gaging plaster.
 - 2. Finish-Coat Mix for Float Finishes: Gypsum gaging plaster.
- E. Gypsum-Plaster Finishes: Match finish(es) of design reference sample(s).
 - 1. Provide troweled finish unless otherwise indicated.
 - 2. Provide float finish where indicated.
- 3.7 REMOVING AND INSTALLING LATH AND ACCESSORIES
 - A. General: Cut existing plaster as to expose deteriorated or rusted lath, wire ties, and support system, back to firm substrates and supports. Repair with new materials, well secured to existing lath in good condition and to building structure.
 - 1. Cutting: Cut lath so it can be taken out completely from one support to the next. Cut to avoid cracking surrounding plaster.
 - 2. Cut out existing base-coat plaster beyond the edges of the new lath to permit new plaster to extend onto the old lath. Then step subsequent plaster coats to permit new plaster to extend over the old material.
 - 3. Fasten new lath to support system and to good existing lath. Wire tie at least every 6 inches.
 - 4. Install new lath according to ASTM C 1063 for lime plaster and ASTM C 841 for gypsum plaster.
 - B. Notify Architect of undocumented detrimental conditions including cracks, bulges, loose backup, rotted wood, rusted metal, and other deteriorated items.
 - C. Wood Lath: Install wood lath in same orientation and spacing as remaining wood lath and with lath ends supported by furring or framing. Stagger ends of adjacent laths over different supports, not aligned, and secure with fasteners at each end and spaced a maximum of 24 inches o.c. into supports.
 - D. Metal Lath: Install according to ASTM C 1063 for lime plaster and ASTM C 841 for gypsum plaster.

- 1. Partition Framing and Vertical Furring: Install flat diamond-mesh lath.
- 2. Flat-Ceiling and Horizontal Framing: Install flat diamond-mesh lath.
- 3. Curved-Ceiling Framing: Install flat diamond-mesh lath.
- 4. On Solid Surfaces, Not Otherwise Furred: Install self-furring, diamond-mesh lath.

3.8 PATCH REPAIR

- A. General: Patch voids, fractured surfaces, and crushed areas in otherwise sound plaster that are larger than crackS.
 - 1. Notify Architect of undocumented detrimental conditions including cracks, bulges, loose backup, rotted wood, rusted metal, and other deteriorated items.
 - 2. Inspect for deterioration of supporting plaster and lath, and repair or replace deteriorated material as required for a sound substrate.
 - 3. Rake perimeter of hole to sound plaster, and slightly undercut existing plaster to enable replacement plaster to tuck behind existing plaster.
 - 4. Replace missing lath in kind. Bridge gaps in wood lath with expanded-metal lath, overlapping wood by 6 inches and fastening them together.
 - 5. Clean hole to remove loose materials and other foreign matter and deposits that could impair bond with repair material. Where grease, waxes, oils, waterborne staining, or other foreign matter and deposits that could impair bond with repair material have penetrated into the plaster, enlarge the hole to remove these deposits.
 - 6. Wet substrate to damp condition, but without visible water droplets, then install patch material to original profiles.
 - 7. Maintain adjacent plasterwork in an undamaged condition so far as practicable.
- B. Lime-Plaster Mix: 3 parts lime putty, 1 part gypsum neat plaster or gypsum gaging plaster for repairs under one foot in any dimension. For repairs greater than 12 inches in any dimension, 1 part lime putty, 2-1/2 parts sand, applied in two coats with fiber in first coat.
- C. Gypsum-Plaster Mix: Gypsum gaging plaster or Repair mix demonstrated in mockup.
- D. Finishing: Finish flat surfaces flush and with same texture as adjacent existing plaster. For molded plaster shapes, tool surface to restore the sharp edges and the shape of the molded shape to original contours.
- E. Hairline cracking within the plaster or plaster separation at edge of a patch is unacceptable. Completely remove such work and reinstall or repair.

3.9 CRACK REPAIR

A. General: Repair discernible cracks in otherwise sound plaster, compensated on a Unit Price Basis.

- 1. Notify Architect of undocumented detrimental conditions including cracks, bulges, loose backup, rotted wood, rusted metal, and other deteriorated items.
- 2. Maintain adjacent plasterwork in an undamaged condition so far as practicable.
- B. Existing Topcoat: Open crack in existing topcoat to at least 1/8 inchin width and check for broken fiber reinforcement in base coats.
- C. Existing Base Coats: Do not open crack wider in existing base coats unless inspection or other indication shows that the fiber reinforcement has broken. Where inspections indicate failure of fiber reinforcement, Open crack to at least 1/8 inch in width and full depth with V-groove tool, and check for bond separation or lath deterioration.
- D. Clean out crack to remove loose materials and other foreign matter and deposits that could impair bond with repair material. Where grease, waxes, oils, waterborne staining, or other foreign matter and deposits that could impair bond with repair material have penetrated into the topcoat plaster, widen the crack and sand surface of the exposed basecoat to remove these deposits.
- E. Wet substrate with latex acrylic bonding agent to damp condition, but without visible water droplets.
- F. Install finish-coat plaster or repair material demonstrated in mockup to fill crack to original plaster profile.
- G. Finishing: Finish flat surfaces flush and with same texture as adjacent existing plaster. For molded plaster shapes, tool surface to restore the sharp edges and the shape of the molded shape to original contours.
- H. Offset Cracks: If the crack is offset in surface plane by more than 1/8 inch, cut the plaster on each side of the crack, a minimum width of 6 inchesand down to the lath or other substrate. Then, repair as specified for flat-plaster removal and replacement.

3.10 REATTACHMENT OF DELAMINATED PLASTER (RDP)

- A. General: Reattach plaster that has detached from its wooden lath.
 - 1. Notify Architect of undocumented detrimental conditions including cracks, bulges, loose backup, rotted wood, rusted metal, and other deteriorated items.
 - 2. Maintain adjacent plasterwork in an undamaged condition so far as practicable.
- B. Reattachment of Delaminated Plaster in excess of the indicated scope will be compensated under the Unit Price schedule.
- C. Verify extent of detachment of plaster that has not yet fallen by tapping on plaster surface and evaluating the hollow or solid resonance.

- D. Protect floors from spillage and debris in the vicinity of work. Use materials resistant to the passage of fluids used in work.
- E. Drill 1/4-inch injection ports (holes) through the plaster spaced 3 to 6 inches apart over surface of detached plaster. Dislodge loose plaster particles, and vacuum debris from holes.
- F. Prewet injection ports, gaps at edges of lost plaster, back of plaster, and wooden lath with prewet solution.
- G. Inject adhesive into ports, enough to fill gaps between detached plaster and lath, and inject into gaps at edges of lost plaster.
- H. Clean off excess and smeared adhesive while wet.
- I. Apply temporary battens over surface of treated plaster to prevent further separation during repair work. Secure battens in place against plaster with screws through the battens and plaster and into the wood lath.
- J. Maintain temporary battens in place for a week or more, allowing adhesive to coalesce and dry.
- K. Remove battens, patch holes and missing plaster, and repair cracks.
- 3.11 INSTALLATION TOLERANCES
 - A. Completed plaster installation shall not deviate from a true plane by more than 1/16 inch as measured by a 5-foot straightedge placed at any location on a surface, except where existing plaster is retained as a substrate for new plasterwork.
- 3.12 CLEANING AND PROTECTION
 - A. Protect work of other trades against damage. Promptly remove plaster from surfaces not indicated to be repaired or plastered. Do not scratch or damage finished surfaces.
 - B. Repair floors, walls, and other surfaces stained, marred, or otherwise damaged during plastering.
 - C. Correct damage to other historic surfaces and to new work of other trades by cleaning, repairing, replacing, and refinishing, as approved by Architect, and leave in an undamaged condition.
 - D. Remove temporary protection and enclosure of other work.

END OF SECTION 09 0320

SECTION 09 2900 – GYPSUM BOARD

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Gypsum board for interior ceilings.
 - 2. Joint treatment materials.
 - 3. Trim accessories.
 - 4. All other accessories for a complete installation work of specified under this Section.
- B. Related Sections:
 - 1. Section 06 1000 Rough Carpentry
 - 2. Section 07 9200 Joint Sealants
 - 3. Section 09 9123 Interior Painting

1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Samples: For the following products:
 - 1. Trim Accessories: Full-size Sample in 12-inch long length for each trim accessory indicated.
- C. Samples for Initial Selection: For each type of trim accessory indicated.
- D. Samples for Verification: For the following products:
 - 1. Trim Accessories: Full-size Sample in 12-inch long length for each trim accessory indicated.

1.4 QUALITY ASSURANCE

- A. The following standards, as well as other standards which may be referred to in this Section, shall apply to the work of this Section:
 - 1. The Gypsum Construction Handbook, latest edition, USG.
 - 2. Construction Guide, latest edition, National Gypsum.
- B. Installer: Firm with not less than 5 years of successful experience in the installation of specified materials.
- C. Mockups: Build mockups of at least 100 sq. ft. in surface area to demonstrate aesthetic effects and to set quality standards for materials and execution.
 - 1. Build mockups for the following:
 - a. Each level of gypsum board finish indicated for use in exposed locations.
 - b. Each texture finish indicated.
 - 2. Apply or install final decoration indicated, including painting and wallcoverings, on exposed surfaces for review of mockups.
 - 3. Simulate finished lighting conditions for review of mockups.
 - 4. Subject to compliance with requirements, approved mockups may become part of the completed Work if undisturbed at time of Substantial Completion.

1.5 DELIVERY, STORAGE AND HANDLING

A. Store materials inside under cover and keep them dry and protected against weather, condensation, direct sunlight, construction traffic, and other potential causes of damage. Stack panels flat and supported on risers on a flat platform to prevent sagging.

1.6 FIELD CONDITIONS

- A. Environmental Limitations: Comply with ASTM C 840 requirements or gypsum board manufacturer's written instructions, whichever are more stringent.
- B. Do not install paper-faced gypsum panels until installation areas are enclosed and conditioned.
- C. Do not install panels that are wet, moisture damaged, and mold damaged.
 - 1. Indications that panels are wet or moisture damaged include, but are not limited to, discoloration, sagging, or irregular shape.
 - 2. Indications that panels are mold damaged include, but are not limited to, fuzzy or splotchy surface contamination and discoloration.

PART 2 - PRODUCTS

2.1 PERFORMANCE REQUIREMENTS

- A. Fire-Resistance-Rated Assemblies: For fire-resistance-rated assemblies, provide materials and construction identical to those tested in assembly indicated according to ASTM E 119 by an independent testing agency.
- B. STC-Rated Assemblies: For STC-rated assemblies, provide materials and construction identical to those tested in assembly indicated according to ASTM E 90 and classified according to ASTM E 413 by an independent testing agency.

2.2 GYPSUM BOARD, GENERAL

A. Size: Provide maximum lengths and widths available that will minimize joints in each area and that correspond with support system indicated.

2.3 INTERIOR GYPSUM BOARD

- A. Acceptable Manufacturers
 - 1. United States Gypsum Corporation
 - 2. Lafarge North America, Inc.
 - 3. National Gypsum Company
- B. Gypsum Board. Moisture/Mold-Resistant: ASTM C 1396/C 1396M.
 - 1. Core: 5/8 inch
 - 2. Long Edges: Tapered.
 - 3. Mold Resistance: ASTM D3273, score of 10 as rated according to ASTM D3274.
- C. Type X Gypsum Board. Moisture/Mold-Resistant: ASTM C 1396/C 1396M.
 - 1. Core: 5/8 inch
 - 2. Long Edges: Tapered.
 - 3. Mold Resistance: ASTM D3273, score of 10 as rated according to ASTM D3274.
- D. Gypsum Ceiling Board. Moisture/Mold-Resistant: ASTM C 1396/C 1396M.
 - 1. Core: 5/8 inch.
 - 2. Long Edges: Tapered.
 - 3. Mold Resistance: ASTM D3273, score of 10 as rated according to ASTM D3274.

2.4 TRIM ACCESSORIES

A. Interior Trim: ASTM C 1047.

- 1. Material: Galvanized or aluminum-coated steel sheet, rolled zinc, plastic, or paper-faced galvanized-steel sheet.
- 2. Shapes:
 - a. Cornerbead.
 - b. Bullnose bead.
 - c. LC-Bead: J-shaped; exposed long flange receives joint compound.
 - d. L-Bead: L-shaped; exposed long flange receives joint compound.
 - e. U-Bead: J-shaped; exposed short flange does not receive joint compound.
 - f. Expansion (control) joint.
 - g. Curved-Edge Cornerbead: With notched or flexible flanges.
- B. Mud-in End Caps.
 - 1. Acceptable Manufacturer: Trim-Tex Inc
 - 2. Performance Requirements:
 - a. Self-Extinguishing: Shall not continue to support combustion once flame source is removed.
 - b. Meet or exceed following ASTM Standards:
 - 1) ASTM E84-10 Achieve Class A rating for Smoke and Flame Spread.
 - 2) ASTM C 1047 Standard Specification for Accessories for Gypsum Wallboard.
 - 3) GA-216-10 Gypsum Association.
 - 4) Impervious to rust, galvanic corrosion, electrolysis and resistant to most chemicals.
 - c. Accessories:
 - 1) Trim-Tex 847 Drywall Spray Adhesive
 - 2) Trim-Tex Powdered Mud-Max adhesion additive
 - 3) 1/2" Divergent Staples

2.5 JOINT TREATMENT MATERIALS

- A. General: Comply with ASTM C 475/C 475M.
- B. Joint Tape:
 - 1. Interior Gypsum Board: Paper.
 - 2. Tile Backing Panels: As recommended by panel manufacturer.
- C. Joint Compound for Interior Gypsum Board: For each coat, use formulation that is compatible with other compounds applied on previous or for successive coats.
 - 1. Prefilling: At open joints, rounded or beveled panel edges, and damaged surface areas, use setting-type taping compound.
 - 2. Embedding and First Coat: For embedding tape and first coat on joints, fasteners, and trim flanges, use drying-type, all-purpose compound.

- 3. Fill Coat: For second coat, use drying-type, all-purpose compound.
- 4. Finish Coat: For third coat, use drying-type, all-purpose compound.
- 5. Skim Coat: For final coat of Level 5 finish, use drying-type, all-purpose compound.
- D. Joint Compound for Tile Backing Panels:
 - 1. Cementitious Backer Units: As recommended by backer unit manufacturer.

2.6 AUXILIARY MATERIALS

- A. General: Provide auxiliary materials that comply with referenced installation standards and manufacturer's written instructions.
- B. Steel Drill Screws: ASTM C 1002 unless otherwise indicated.
 - 1. Use screws complying with ASTM C 954 for fastening panels to steel members from 0.033 to 0.112 inch thick.
 - 2. For fastening cementitious backer units, use screws of type and size recommended by panel manufacturer.
- C. Acoustical Sealant: Paintable, nonstaining latex sealant complying with ASTM C 834. Product effectively reduces airborne sound transmission through perimeter joints and openings in building construction as demonstrated by testing representative assemblies according to ASTM E 90.
 - 1. Products: One of the following:
 - a. Tremco Inc; Hilti, Inc; Acoustical/Curtainwall Sealant
 - b. USG Corporation; SHEETROCK Acoustical Sealant.
 - c. Pecora Corporation; AC-20 FTR.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine areas and substrates including welded hollow-metal frames and support framing, with Installer present, for compliance with requirements and other conditions affecting performance of the Work.
- B. Examine panels before installation. Reject panels that are wet, moisture damaged, and mold damaged.
- C. Proceed with installation only after unsatisfactory conditions have been corrected.

3.2 INSTALLATION AND FINISHING PANELS, GENERAL

- A. Comply with ASTM C 840.
- B. Install ceiling panels across framing to minimize the number of abutting end joints and to avoid abutting end joints in central area of each ceiling. Stagger abutting end joints of adjacent panels not less than one framing member.
- C. Install panels with face side out. Butt panels together for a light contact at edges and ends with not more than 1/16 inch of open space between panels. Do not force into place.
- D. Locate edge and end joints over supports, except in ceiling applications where intermediate supports or gypsum board back-blocking is provided behind end joints. Do not place tapered edges against cut edges or ends. Stagger vertical joints on opposite sides of partitions. Do not make joints other than control joints at corners of framed openings.
- E. Form control and expansion joints with space between edges of adjoining gypsum panels.
- F. Cover both faces of support framing with gypsum panels in concealed spaces (above ceilings, etc.), except in chases braced internally.
 - 1. Unless concealed application is indicated or required for sound, fire, air, or smoke ratings, coverage may be accomplished with scraps of not less than 8 sq. ft. in area.
 - 2. Fit gypsum panels around ducts, pipes, and conduits.
 - 3. Where partitions intersect structural members projecting below underside of floor/roof slabs and decks, cut gypsum panels to fit profile formed by structural members; allow 1/4-inch to 3/8-inch wide joints to install sealant.
- G. Isolate perimeter of gypsum board applied to non-load-bearing partitions at structural abutments. Provide 1/4- to 1/2-inch wide spaces at these locations and trim edges with edge trim where edges of panels are exposed. Seal joints between edges and abutting structural surfaces with acoustical sealant.
- H. Attachment to Steel Framing: Attach panels so leading edge or end of each panel is attached to open (unsupported) edges of stud flanges first.
- I. Wood Framing: Install gypsum panels over wood framing, with floating internal corner construction. Do not attach gypsum panels across the flat grain of wide-dimension lumber, including floor joists and headers. Float gypsum panels over these members or provide control joints to counteract wood shrinkage.

3.3 INSTALLATION OF INTERIOR GYPSUM BOARD

- A. Install interior gypsum board in the following locations:
 - 1. Ceiling Type: Ceiling surfaces.

- B. Single-Layer Application:
 - 1. On ceilings, apply gypsum panels before wall/partition board application to greatest extent possible and at right angles to framing unless otherwise indicated.
 - 2. On partitions/walls, apply gypsum panels vertically (parallel to framing) unless otherwise indicated or required by fire-resistance-rated assembly, and minimize end joints.
 - a. Stagger abutting end joints not less than one framing member in alternate courses of panels.
 - b. At stairwells and other high walls, install panels horizontally unless otherwise indicated or required by fire-resistance-rated assembly.

3.4 INSTALLING TRIM ACCESSORIES

- A. General: For trim with back flanges intended for fasteners, attach to framing with same fasteners used for panels. Otherwise, attach trim according to manufacturer's written instructions.
- B. Control Joints: Install control joints according to ASTM C 840 and in specific locations approved by Architect for visual effect.
- C. Interior Trim: Install in the following locations:
 - 1. Cornerbead: Use at outside corners.
 - 2. Bullnose Bead: Use Where indicated.
 - 3. LC-Bead: Use At exposed panel edges.
 - 4. L-Bead: Use Where indicated.
 - 5. U-Bead: Use Where indicated.
 - 6. Curved-Edge Cornerbead: Use at curved openings.

3.5 FINISHING GYPSUM BOARD

- A. General: Treat gypsum board joints, interior angles, edge trim, control joints, penetrations, fastener heads, surface defects, and elsewhere as required to prepare gypsum board surfaces for decoration. Promptly remove residual joint compound from adjacent surfaces.
- B. Prefill open joints rounded or beveled edges, and damaged surface areas.
- C. Apply joint tape over gypsum board joints, except for trim products specifically indicated as not intended to receive tape.
- D. Gypsum Board Finish Levels: Finish panels to levels indicated below and according to ASTM C 840:
 - 1. Level 1: Ceiling plenum areas and concealed areas.

- 2. Level 5: At exposed areas.
 - a. Primer and its application to surfaces are specified in Section 09 9123 Interior Painting.

3.6 PROTECTION

- A. Protect adjacent surfaces from drywall compound and promptly remove from floors and other non-drywall surfaces. Repair surfaces stained, marred, or otherwise damaged during drywall application.
- B. Protect installed products from damage from weather, condensation, direct sunlight, construction, and other causes during remainder of the construction period.
- C. Remove and replace panels that are wet, moisture damaged, and mold damaged.
 - 1. Indications that panels are wet or moisture damaged include, but are not limited to, discoloration, sagging, or irregular shape.
 - 2. Indications that panels are mold damaged include, but are not limited to, fuzzy or splotchy surface contamination and discoloration.

END OF SECTION 09 2900

SECTION 09 6513 – RESILIENT BASE AND ACCESSORIES

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Resilient base.
- 1.3 ACTION SUBMITTALS
 - A. Product Data: For each type of product.
 - B. Samples: For each exposed product and for each color and texture specified, not less than 12 inches long.
 - C. Samples for Initial Selection: For each type of product indicated.
 - D. Samples for Verification: For each type of product indicated and for each color, texture, and pattern required in manufacturer's standard-size Samples, but not less than 12 inches long.
 - E. Product Schedule: For resilient base and accessory products. Use same designations indicated on Drawings.

1.4 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Furnish not less than 10 linear feet for every 500 linear feet or fraction thereof, of each type, color, pattern, and size of resilient product installed.

1.5 QUALITY ASSURANCE

- A. Mockups: Build mockups to verify selections made under Sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution.
 - 1. Coordinate mockups in this Section with mockups specified in other Sections.

1.6 DELIVERY, STORAGE, AND HANDLING

A. Store resilient products and installation materials in dry spaces protected from the weather, with ambient temperatures maintained within range recommended by manufacturer, but not less than 50 deg F or more than 90 deg F.

1.7 FIELD CONDITIONS

- A. Maintain ambient temperatures within range recommended by manufacturer, but not less than 70 deg or more than 95 deg F, in spaces to receive resilient products during the following time periods:
 - 1. 48 hours before installation.
 - 2. During installation.
 - 3. 48 hours after installation.
- B. After installation and until Substantial Completion, maintain ambient temperatures within range recommended by manufacturer, but not less than 55 deg F or more than 95 deg F,
- C. Install resilient products after other finishing operations, including painting, have been completed.

PART 2 - PRODUCTS

- 2.1 THERMOPLASTIC-RUBBER BASE:
 - A. Manufacturers: Subject to compliance with specified requirements, furnish products by one of the following:
 - 1. Armstrong World Industries, Inc.
 - 2. Johnsonite; A Tarkett Company.
 - 3. Nora Systems, Inc.
 - 4. Roppe Corporation, USA.
 - B. Product Standard: ASTM F 1861, Type TP (rubber, thermoplastic).
 - 1. Group: I solid, homogeneous.
 - 2. Style and Location:
 - a. Style B, Cove: Provide in areas with resilient flooring and as scheduled.
 - C. Thickness: 0.125 inch.
 - D. 4 inches.
 - E. Lengths: Coils in manufacturer's standard length.
 - F. Outside Corners: Preformed

- G. Inside Corners: Preformed
- H. Colors: As selected by Architect from full range of industry colors.
- 2.2 RUBBER MOLDING ACCESSORIES
 - A. Manufacturers: Subject to compliance with requirements, furnish products by one of the following:
 - 1. Johnsonite; A Tarkett Company.
 - 2. Roppe Corporation
 - 3. Nora Systems, Inc.
 - B. Basis of Design:
 - 1. Rubber transition strip (RTS) for carpet to resilient flooring: Johnsonite Model CTA-XX-A.
 - C. Locations: As scheduled
 - D. Colors and Patterns: As selected by Architect from full range of industry colors.
- 2.3 INSTALLATION MATERIALS
 - A. Trowelable Leveling and Patching Compounds: Latex-modified, portland cement based or blended hydraulic-cement-based formulation provided or approved by resilient-product manufacturer for applications indicated.
 - B. Adhesives: Water-resistant type recommended by resilient-product manufacturer for resilient products and substrate conditions indicated.
 - 1. Adhesives shall have a VOC content of 50 g/L or less except that adhesive for rubber stair treads shall have a VOC content of 60 g/L or less.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, with Installer present, for compliance with requirements for maximum moisture content and other conditions affecting performance of the Work.
 - 1. Verify that finishes of substrates comply with tolerances and other requirements specified in other Sections and that substrates are free of cracks, ridges, depressions, scale, and foreign deposits that might interfere with adhesion of resilient products.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
 - 1. Installation of resilient products indicates acceptance of surfaces and conditions.

3.2 PREPARATION

- A. Prepare substrates according to manufacturer's written instructions to ensure adhesion of resilient products.
- B. Concrete Substrates for Resilient: Prepare horizontal surfaces according to ASTM F 710.
 - 1. Verify that substrates are dry and free of curing compounds, sealers, and hardeners.
 - 2. Remove substrate coatings and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by manufacturer. Do not use solvents.
 - 3. Alkalinity and Adhesion Testing: Perform tests recommended by manufacturer. Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by manufacturer in writing, but not less than 5 or more than 10 pH.
 - 4. Moisture Testing: Proceed with installation only after substrates pass testing according to manufacturer's written recommendations, but not less stringent than the following:
 - a. Perform anhydrous calcium chloride test according to ASTM F 1869. Proceed with installation only after substrates have maximum moisturevapor-emission rate of 3 lb of water/1000 sq. ft in 24 hours.
 - b. Perform relative humidity test using in situ probes according to ASTM F 2170. Proceed with installation only after substrates have maximum 75 percent relative humidity level.
- C. Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps and ridges to produce a uniform and smooth substrate.
- D. Do not install resilient products until they are the same temperature as the space where they are to be installed.
 - 1. At least 48 hours in advance of installation, move resilient products and installation materials into spaces where they will be installed.
- E. Immediately before installation, sweep and vacuum clean substrates to be covered by resilient products.

3.3 RESILIENT BASE INSTALLATION

- A. Comply with manufacturer's written instructions for installing resilient base.
- B. Apply resilient base to walls, columns, pilasters, casework and cabinets in toe spaces, and other permanent fixtures in rooms and areas where base is required.
- C. Install resilient base in lengths as long as practical without gaps at seams and with tops of adjacent pieces aligned.

- D. Tightly adhere resilient base to substrate throughout length of each piece, with base in continuous contact with horizontal and vertical substrates.
- E. Do not stretch resilient base during installation.
- F. On masonry surfaces or other similar irregular substrates, fill voids along top edge of resilient base with manufacturer's recommended adhesive filler material.
- G. Preformed Corners: Install preformed corners before installing straight pieces.
- 3.4 RESILIENT ACCESSORY INSTALLATION
 - A. Comply with manufacturer's written instructions for installing resilient accessories.
 - B. Resilient Molding Accessories: Butt to adjacent materials and tightly adhere to substrates throughout length of each piece. Install reducer strips at edges of floor covering that would otherwise be exposed.
- 3.5 CLEANING AND PROTECTION
 - A. Comply with manufacturer's written instructions for cleaning and protecting resilient products.
 - B. Perform the following operations immediately after completing resilient-product installation:
 - 1. Remove adhesive and other blemishes from exposed surfaces.
 - 2. Sweep and vacuum horizontal surfaces thoroughly.
 - 3. Damp-mop horizontal surfaces to remove marks and soil.
 - C. Protect resilient products from mars, marks, indentations, and other damage from construction operations and placement of equipment and fixtures during remainder of construction period.
 - D. Cover resilient products subject to wear and foot traffic until Substantial Completion.

END OF SECTION 09 6513

SECTION 09 6519 – VINYL COMPOSITION FLOOR TILING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Vinyl composition floor tile.
 - 2. All adhesives and other accessories as required for a complete installation.
- B. RELATED SECTIONS
 - 1. Section 03 3000 Cast-In –Place Concrete
- C. WORK INSTALLED UNDER THIS SECTION BUT SPECIFIED ELSEWHERE
 - 1. Section 09 6513 Resilient Base and Accessories

1.3 REFERENCES

- A. ASTM International:
 - 1. ASTM E 648 Standard Test Method for Critical Radiant Flux of Floor-Covering Systems Using a Radiant Heat Energy Source
 - 2. ASTM E 1745 Standard Specification for Plastic Water Vapor Retarders Used in Contact with Soil or Granular Fill under Concrete Slabs
 - 3. ASTM F 710 Standard Practice for Preparing Concrete Floors to Receive Resilient Flooring
 - 4. ASTM F 1066 Standard Specification for Vinyl Composition Floor Tile
 - 5. ASTM F 1869 Standard Test Method for Measuring Vapor Emission Rate of Concrete Subfloor Using Anhydrous Calcium Chloride
 - 6. ASTM F 2170 Standard Test Method for Determining Relative Humidity in Concrete Floor Slabs Using in situ Probes
- B. National Fire Protection Association (NFPA):
 - 1. NFPA 253 Standard Method of Test for Critical Radiant Flux of Floor Covering Systems Using a Radiant Heat Energy Source.
- 1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Submittals:
 - 1. Product Data: For adhesives, sealants and chemical-bonding compounds, documentation including printed statement of VOC content.
 - 2. Product Data: For adhesives and chemical-bonding compounds, documentation including printed statement of VOC content.
 - 3. Product Data For resilient tile flooring, documentation from an independent testing agency indicating compliance with the FloorScore standard.
- C. Shop Drawings: For each type of floor tile. Include floor tile layouts, edges, columns, doorways, enclosing partitions, built-in furniture, cabinets, and cutouts.
 - 1. Show details of special patterns.
- D. Samples for Verification: Full-size units of each color and pattern of floor tile required.
- E. Product Schedule: For floor tile use same designations indicated on Drawings.
- F. Product Data: Submit manufacturer's product data sheets, installation guide, and maintenance guide for each material.
- G. MSDS (Material Safety Data Sheets): for all adhesives used: membrane, primer, patch, leveler, heat weld cord, cold weld, liquid wax and cleaning agents.
- 1.5 INFORMATIONAL SUBMITTALS
 - A. Qualification Data: For Installer.
- 1.6 CLOSEOUT SUBMITTALS
 - A. Maintenance Data: For each type of floor tile to include in maintenance manuals.
- 1.7 MAINTENANCE MATERIAL SUBMITTALS
 - A. Furnish extra materials that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Floor Tile: Furnish one box for every 50 boxes or fraction thereof, of each type, color, and pattern of floor tile installed.
- 1.8 QUALITY ASSURANCE
 - A. Installer Qualifications: A qualified installer who employs workers for this Project who are competent in techniques required by manufacturer for floor tile installation and seaming method indicated.
 - 1. Engage an installer who employs workers for this Project who are trained or certified by floor tile manufacturer for installation techniques required.

- B. Mockups: Build mockups to verify selections made under Sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution.
 - 1. Build mockups for floor tile including resilient base and accessories.
 - a. Size: Minimum 100 sq. ft. (9.3 sq. m) for each type, color, and pattern in locations directed by Architect.
 - 2. Approval of mockups does not constitute approval of deviations from the Contract Documents contained in mockups unless Architect specifically approves such deviations in writing.
 - 3. Subject to compliance with requirements, approved mockups may become part of the completed Work if undisturbed at time of Substantial Completion.
- C. Manufacturer's Qualifications: Provide resilient flooring manufactured by a firm with a minimum of 10 years' experience with resilient flooring of types equivalent to those specified. Manufacturers proposed for use, which are not named in this section, shall submit evidence of ability to meet performance requirements specified not less than 10 days prior to bid date.
 - 1. Color Matching: Provide resilient flooring products, including wall base and accessories, from one manufacturer to ensure color matching.
 - 2. Manufacturer capable of providing technical training and field service representation.
- D. Installer Qualifications: Installer shall be manufacturer approved for the requirements of the project or INSTALL (International Standards & Training Alliance) resilient certified for the requirements of the project.
- E. Adherence to the Quality Management System approved by Lloyd's Register Quality Assurance to the Quality Management System Standard ISO 9001:2000.
- F. Provide adhesives that are in compliance (0 grams/liter) with SCAQMD Rule 1168, < 60 grams/liter Standard for Rubber Floor Adhesives.
- 1.9 DELIVERY, STORAGE, AND HANDLING
 - A. Store floor tile and installation materials in dry spaces protected from the weather, with ambient temperatures maintained within range recommended by manufacturer, but not less than 50 deg F or more than 90 deg F. Store floor tiles on flat surfaces.
 - B. Deliver materials in labeled packages. Store and handle in strict compliance with manufacturer's recommendations. Protect from damage due to weather, excessive temperatures, and construction operations.
 - C. Deliver materials sufficiently in advance of installation to condition materials to the required temperature prior to installation
- 1.10 FIELD CONDITIONS
 - A. Maintain ambient temperatures within range recommended by manufacturer, but not

less than 70 deg F or more than 95 deg F, in spaces to receive floor tile during the following time periods:

- 1. 48 hours before installation.
- 2. During installation.
- 3. 72 hours after installation.
- B. Maintain humidity, within temperature range, at 50% RH +/- 10% in areas to receive resilient flooring.
- C. After installation and until Substantial Completion, maintain ambient temperatures within range recommended by manufacturer, but not less than 70 deg F (21 deg C) or more than 95 deg F.
- D. After installation and until Substantial Completion, maintain ambient temperatures within range recommended by manufacturer, but not less than 55 deg F or more than 95 deg F.
- E. Close spaces to traffic during floor tile installation.
- F. Close spaces to traffic for 48 hours after floor tile installation.
- G. Install floor tile after other finishing operations, including painting, have been completed.
- 1.11 WARRANTY
 - A. Provide manufacturer's standard one-year warranty against defects in manufacturing and workmanship of all flooring products. Provide manufacturer's warranty as specified under each product as applicable, including limited wear, defect and conductivity
- 1.12 EXTRA MATERIALS
 - A. Furnish full size units equal to 2 percent of quantity of resilient flooring installed as extra materials. Properly label and package extra materials. Deliver to owners designated storage area

PART 2 - PRODUCTS

2.1 BASIS-OF-DESIGN PRODUCTS

- A. The design for each resilient tile flooring type is based on the product named, Specification and Drawings to establish the quality level and design requirements for the project. Subject to compliance with requirements, provide either the named product or a comparable product by one of the other manufacturers specified.
- B. Substitution of comparable products and materials specified shall be in accordance to the 'Proposed Substitution' procedures.

2.2 PERFORMANCE REQUIREMENTS

- A. Fire-Test-Response Characteristics: For resilient tile flooring, as determined by testing identical products according to ASTM E 648 or NFPA 253 by a qualified testing agency.
 - 1. Critical Radiant Flux Classification: Class I, not less than 0.45 W/sq. cm.
- B. FloorScore Compliance: Resilient tile flooring shall comply with requirements of FloorScore certification.
- C. Low-Emitting Materials: Flooring system shall comply with the testing and product requirements of the California Department of Public Health's "Standard Method for the Testing and Evaluation of Volatile Organic Chemical Emissions from Indoor Sources Using Environmental Chambers."
- 2.3 VINYL COMPOSITION FLOOR TILE
 - A. Products: Subject to compliance with requirements, provide products by one of the following:
 - 1. Armstrong World Industries, Inc.
 - 2. Congoleum Corporation.
 - 3. Mannington Mills, Inc.
 - B. Tile Standard: ASTM F 1066, Class 2, through-pattern tile.
 - C. Wearing Surface: Smooth.
 - D. Thickness: 0.125 inch.
 - E. Size: 12 by 12 inches.
 - F. Colors and Patterns: As selected by Architect from manufacturer's full range.
- 2.4 RESILIENT TILE FLOORING SCHEDULE
 - A. Resilient Tile Flooring Schedule: Selection of resilient tile flooring products, colors, surface textures, patterns, and other appearance characteristics, are indicated in the following schedules.
 - 1. Products: Resilient tile flooring products are identified in the 'Finish Schedule'.
 - 2. Locations: Resilient tile flooring location are located in the 'Finish Schedule'.

2.5 INSTALLATION MATERIALS

- A. Use only manufacturer's approved system products as specified in their installation guidelines.
- B. Trowelable Leveling and Patching Compounds: Latex-modified, portland cement

based or blended hydraulic-cement-based formulation provided or approved by floor tile manufacturer for applications indicated.

- C. Adhesives: Water-resistant type recommended by floor tile and adhesive manufacturers to suit floor tile and substrate conditions indicated.
 - 1. Adhesives shall comply with the following limits for VOC content:
 - a. Vinyl Composition Tile Adhesives: 50 g/L or less.
- D. Floor Polish: Provide protective, liquid floor-polish products recommended by floor tile manufacturer. All floors are intended to be no wax or polish floors. Review the specific

Manufacturer's maintenance requirements with the Owner prior to any finish being applied to flooring products.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates, with Installer present, for compliance with requirements for maximum moisture content and other conditions affecting performance of the Work.
 - 1. Verify that finishes of substrates comply with tolerances and other requirements specified in other Sections and that substrates are free of cracks, ridges, depressions, scale, and foreign deposits that might interfere with adhesion of floor tile.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

3.2 PREPARATION

- A. Prepare substrates according to floor tile manufacturer's written instructions to ensure adhesion of resilient products.
- B. A building or flooring area that is water tight and fully enclosed from the elements, including roof, windows or facades and doors, that is ready for the flooring installation is required.
- C. A concrete substrate that shall be structurally sound, that is finished shrinking, cracking, curling or moving in any way is required
- D. For all concrete substrates on or below grade, provide a permanent effective vapor retarder with a low permeance (less than 0.1) having a minimum thickness of 10 mils, or meets the requirements of the latest edition of ASTM E1745 (Standard Specification for Water Vapor Retarders Used in Contact with Soil or Granular Fill under Concrete Slabs) shall be placed directly underneath the concrete above the granular fill, and shall be installed as per the manufacturer's written instructions. A letter shall be provided to the end user confirming the correct products have been used and that it is fully warranted.

- E. Provide a secure storage area that is maintained permanently or temporarily at ambient service temperature and humidity (except walk in freezers or similar), or 68°F ± 5° F and 50% ± 10% relative humidity, for at least 48 hours prior to and during the application of the flooring, so the flooring contractor can acclimate the flooring materials is required.
- F. Areas of the flooring that are subject to direct sunlight through doors or windows shall have them covered using blinds, curtains, cardboard or similar for the time of the installation and 72 hours after the installation to allow the adhesive to cure. Note these areas shall be installed using wet adhesives only.
- G. Concrete Substrates: Prepare according to ASTM F 710.
 - 1. Verify that substrates are dry and free of curing compounds, sealers, and hardeners.
 - 2. Remove substrate coatings and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by floor tile manufacturer. Do not use solvents.
 - 3. Alkalinity and Adhesion Testing: Perform tests recommended by floor tile manufacturer. Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by manufacturer in writing, but not less than 5 or more than 9 pH.
 - 4. Moisture Testing: Proceed with installation only after substrates pass testing according to floor tile manufacturer's written recommendations, but not less stringent than the following:
 - a. Perform anhydrous calcium chloride test according to ASTM F 1869. Proceed with installation only after substrates have maximum moisturevapor-emission rate of 3 lb of water/1000 sq. ft. (1.36 kg of water/92.9 sq. m) in 24 hours.
 - b. Perform relative humidity test using in situ probes according to ASTM F 2170. Proceed with installation only after substrates have a maximum 75 percent relative humidity level.
- H. Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps and ridges to produce a uniform and smooth substrate.
- I. Do not install floor tiles until they are the same temperature as the space where they are to be installed.
 - 1. At least 48 hours in advance of installation, move resilient floor tile and installation materials into spaces where they will be installed.
- J. Immediately before installation, sweep and vacuum clean substrates to be covered by resilient floor tile.
- 3.3 FLOOR TILE INSTALLATION
 - A. Comply with manufacturer's written instructions for installing floor tile.

- 1. Prevent all traffic for a minimum of 12 hours and heavy rolling loads for 72 hours to allow the adhesive to set up. If required, after 12 hours protect the flooring from damage during construction operations using Masonite, plywood or a similar product, ensuring first that the flooring surface is free of all debris. Lay panels so that the edges form a butt joint and tape the joint to prevent both movement and debris entrapment underneath them. Inspect immediately before covering and after removal for final acceptance.
- 2. Clean the flooring within 72 hours after the installation. Clean the flooring only using the method detailed in the manufacturer's Maintenance Guide.
- B. Lay out floor tiles from center marks established with principal walls, discounting minor offsets, so tiles at opposite edges of room are of equal width. Adjust as necessary to avoid using cut widths that equal less than one-half tile at perimeter.
 - 1. Lay tiles square with room axis in pattern indicated.
- C. Match floor tiles for color and pattern by selecting tiles from cartons in the same sequence as manufactured and packaged, if so numbered. Discard broken, cracked, chipped, or deformed tiles.
 - 1. Lay tiles with grain running in one direction and in pattern of colors and sizes indicated.
 - 2. Basket-weave pattern is not acceptable.
- D. Scribe, cut, and fit floor tiles to butt neatly and tightly to vertical surfaces and permanent fixtures including built-in furniture, cabinets, pipes, outlets, and door frames.
- E. Extend floor tiles into toe spaces, door reveals, closets, and similar openings. Extend floor tiles to center of door openings.
- F. Maintain reference markers, holes, and openings that are in place or marked for future cutting by repeating on floor tiles as marked on substrates. Use chalk or other nonpermanent marking device.
- G. Install floor tiles on covers for telephone and electrical ducts, building expansion-joint covers, and similar items in finished floor areas. Maintain overall continuity of color and pattern between pieces of tile installed on covers and adjoining tiles. Tightly adhere tile edges to substrates that abut covers and to cover perimeters.
- H. Adhere floor tiles to flooring substrates using a full spread of adhesive applied to substrate to produce a completed installation without open cracks, voids, raising and puckering at joints, telegraphing of adhesive spreader marks, and other surface imperfections.
- 3.4 CLEANING AND PROTECTION
 - A. Comply with manufacturer's written instructions for cleaning and protecting floor tile.
 - B. Perform the following operations immediately after completing floor tile installation:

- 1. Remove adhesive and other blemishes from exposed surfaces.
- 2. Sweep and vacuum surfaces thoroughly.
- 3. Damp-mop surfaces to remove marks and soil.
- C. Protect floor tile from mars, marks, indentations, and other damage from construction operations and placement of equipment and fixtures during remainder of construction period.
- D. Floor Finish: Remove soil, adhesive, and blemishes from floor tile surfaces before applying floor finish.
 - 1. Apply products as recommended by the flooring manufacturer after discussing all maintenance options with the Owner. All floors are specified to be no wax/ no polish and therefore specific requirements for topical treatments vary.
- E. Joint Sealant: Apply sealant to resilient terrazzo floor tile perimeter and around columns, at door frames, and at other joints and penetrations.
- F. Sealers and Finish Coats: Remove soil, visible adhesive, and surface blemishes from resilient terrazzo floor tile surfaces before applying liquid cleaners, sealers, and finish products.
 - 1. Sealer: Apply two base coats of liquid sealer.
 - 2. Finish: Apply three coats of liquid floor finish.
- G. Cover floor tile until Substantial Completion.

END OF SECTION 09 6519

SECTION 09 9113 - EXTERIOR PAINTING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes surface preparation and the application of paint systems on the following exterior substrates:
 - 1. Steel and iron.
 - 2. Galvanized metal.
 - 3. Wood.

B. Related Requirements:

1. Section 05 5213 – Pipe and Tube Railing

1.3 DEFINITIONS

- A. MPI Gloss Level 1 (Flat): Not more than five units at 60 degrees and 10 units at 85 degrees, according to ASTM D 523.
- B. MPI Gloss Level 3 (Eggshell): 10 to 25 units at 60 degrees and 10 to 35 units at 85 degrees, according to ASTM D 523.
- C. MPI Gloss Level 4 (Satin): 20 to 35 units at 60 degrees and not less than 35 units at 85 degrees, according to ASTM D 523.
- D. MPI Gloss Level 5 (Semi-gloss): 35 to 70 units at 60 degrees, according to ASTM D 523.
- E. MPI Gloss Level 6 (Gloss): 70 to 85 units at 60 degrees, according to ASTM D 523.
- F. MPI Gloss Level 7 (High Gloss): More than 85 units at 60 degrees, according to ASTM D 523.

1.4 ACTION SUBMITTALS

A. Product Data: For each type of product. Include preparation requirements and application instructions.

- 1. Include printout of current "MPI Approved Products List" for each product category specified, with the proposed product highlighted.
- 2. Indicate VOC content.
- B. Samples for Initial Selection: For each type of topcoat product.
- C. Samples for Verification: For each type of paint system and each color and gloss of topcoat.
 - 1. Submit Samples on rigid backing, 8 inches square.
 - 2. Apply coats on Samples in steps to show each coat required for system.
 - 3. Label each coat of each Sample.
 - 4. Label each Sample for location and application area.
- D. Product List: Cross-reference to paint system and locations of application areas. Use same designations indicated on Drawings and in schedules. Include color designations.

1.5 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials, from the same product run, that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Paint: 5 percent, but not less than 1 gal. of each material and color applied.

1.6 QUALITY ASSURANCE

- A. Mockups: Apply mockups of each paint system indicated and each color and finish selected to verify preliminary selections made under Sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution.
 - 1. Architect will select one surface to represent surfaces and conditions for application of each paint system.
 - a. Vertical and Horizontal Surfaces: Provide samples of at least 100 sq. ft.
 - b. Other Items: Architect will designate items or areas required.
 - 2. Final approval of color selections will be based on mockups.
 - a. If preliminary color selections are not approved, apply additional mockups of additional colors selected by Architect at no added cost to Owner.
 - 3. Approval of mockups does not constitute approval of deviations from the Contract Documents contained in mockups unless Architect specifically approves such deviations in writing.
 - 4. Subject to compliance with requirements, approved mockups may become part of the completed Work if undisturbed at time of Substantial Completion.

1.7 DELIVERY, STORAGE, AND HANDLING

- A. Store materials not in use in tightly covered containers in well-ventilated areas with ambient temperatures continuously maintained at not less than 40 deg F.
 - 1. Maintain containers in clean condition, free of foreign materials and residue.
 - 2. Remove rags and waste from storage areas daily.

1.8 FIELD CONDITIONS

- A. Apply paints only when temperature of surfaces to be painted and ambient air temperatures are between 40 and 90 deg F.
- B. Do not apply paints in snow, rain, fog, or mist; when relative humidity exceeds 85 percent; at temperatures less than 5 deg F above the dew point; or to damp or wet surfaces.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Basis-of-Design Product: Subject to compliance with requirements, provide Benjamin Moore & Co.; Products as designated in "Exterior Painting Schedule" or comparable products by one of the following:
 - 1. Behr Process Corporation.
 - 2. PPG Architectural Coatings.
 - 3. Sherwin-Williams Company (The).

2.2 PAINT, GENERAL

- A. MPI Standards: Products shall comply with MPI standards indicated and shall be listed in its "MPI Approved Products Lists."
- B. Material Compatibility:
 - 1. Materials for use within each paint system shall be compatible with one another and substrates indicated, under conditions of service and application as demonstrated by manufacturer, based on testing and field experience.
 - 2. For each coat in a paint system, products shall be recommended in writing by topcoat manufacturers for use in paint system and on substrate indicated.
- C. VOC Content: For field applications, paints and coatings shall comply with VOC content limits of authorities having jurisdiction and the following VOC content limits:
 - 1. Flat Paints and Coatings: 50 g/L.
 - 2. Nonflat Paints and Coatings: 50 g/L.

- 3. Dry-Fog Coatings: 150 g/L.
- 4. Primers, Sealers, and Undercoaters: 100 g/L.
- 5. Rust-Preventive Coatings: 150 g/L.
- 6. Zinc-Rich Industrial Maintenance Primers: 500 g/L.
- 7. Pretreatment Wash Primers: 420 g/L.
- 8. Shellacs, Clear: 730 g/L.
- 9. Shellacs, Pigmented: 550 g/L.
- D. Colors: As selected by Architect from manufacturer's full range or as indicated in a color schedule.
 - 1. Ten percent of surface area will be painted with deep tones.

2.3 SOURCE QUALITY CONTROL

- A. Testing of Paint Materials: Owner reserves the right to invoke the following procedure:
 - 1. Owner will engage the services of a qualified testing agency to sample paint materials. Contractor will be notified in advance and may be present when samples are taken. If paint materials have already been delivered to Project site, samples may be taken at Project site. Samples will be identified, sealed, and certified by testing agency.
 - 2. Testing agency will perform tests for compliance with product requirements.
 - 3. Owner may direct Contractor to stop applying paints if test results show materials being used do not comply with product requirements. Contractor shall remove noncomplying paint materials from Project site, pay for testing, and repaint surfaces painted with rejected materials. Contractor will be required to remove rejected materials from previously painted surfaces if, on repainting with complying materials, the two paints are incompatible.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates and conditions, with Applicator present, for compliance with requirements for maximum moisture content and other conditions affecting performance of the Work.
- B. Maximum Moisture Content of Substrates: When measured with an electronic moisture meter as follows:
 - 1. Fiber-Cement Board: 12 percent.
 - 2. Wood: 15 percent.
- C. Verify suitability of substrates, including surface conditions and compatibility, with existing finishes and primers.

- D. Proceed with coating application only after unsatisfactory conditions have been corrected.
 - 1. Application of coating indicates acceptance of surfaces and conditions.

3.2 PREPARATION

- A. Comply with manufacturer's written instructions and recommendations in "MPI Architectural Painting Specification Manual" applicable to substrates and paint systems indicated.
- B. Remove hardware, covers, plates, and similar items already in place that are removable and are not to be painted. If removal is impractical or impossible because of size or weight of item, provide surface-applied protection before surface preparation and painting.
 - 1. After completing painting operations, use workers skilled in the trades involved to reinstall items that were removed. Remove surface-applied protection.
- C. Clean substrates of substances that could impair bond of paints, including dust, dirt, oil, grease, and incompatible paints and encapsulants.
 - 1. Remove incompatible primers and reprime substrate with compatible primers or apply tie coat as required to produce paint systems indicated.
- D. Steel Substrates: Remove rust, loose mill scale, and shop primer if any. Clean using methods recommended in writing by paint manufacturer but not less than the following:
 - 1. SSPC-SP 2.
 - 2. SSPC-SP 3.
 - 3. SSPC-SP 7/NACE No. 4.
 - 4. SSPC-SP 11.
- E. Galvanized-Metal Substrates: Remove grease and oil residue from galvanized sheet metal by mechanical methods to produce clean, lightly etched surfaces that promote adhesion of subsequently applied paints.
- F. Wood Substrates:
 - 1. Scrape and clean knots. Before applying primer, apply coat of knot sealer recommended in writing by topcoat manufacturer for exterior use in paint system indicated.
 - 2. Sand surfaces that will be exposed to view, and dust off.
 - 3. Prime edges, ends, faces, undersides, and backsides of wood.
 - 4. After priming, fill holes and imperfections in the finish surfaces with putty or plastic wood filler. Sand smooth when dried.

3.3 APPLICATION

- A. Apply paints according to manufacturer's written instructions and recommendations in "MPI Architectural Painting Specification Manual."
 - 1. Use applicators and techniques suited for paint and substrate indicated.
 - 2. Paint surfaces behind movable items same as similar exposed surfaces. Before final installation, paint surfaces behind permanently fixed items with prime coat only.
 - 3. Paint both sides and edges of exterior doors and entire exposed surface of exterior door frames.
 - 4. Paint entire exposed surface of window frames and sashes.
 - 5. Do not paint over labels of independent testing agencies or equipment name, identification, performance rating, or nomenclature plates.
 - 6. Primers specified in painting schedules may be omitted on items that are factory primed or factory finished if acceptable to topcoat manufacturers.
- B. If undercoats or other conditions show through topcoat, apply additional coats until cured film has a uniform paint finish, color, and appearance.
- C. Apply paints to produce surface films without cloudiness, spotting, holidays, laps, brush marks, roller tracking, runs, sags, ropiness, or other surface imperfections. Cut in sharp lines and color breaks.
- D. Painting Fire Suppression, Plumbing, HVAC, Electrical, Communication, and Electronic Safety and Security Work:
 - 1. Paint the following work where exposed to view:
 - a. Uninsulated metal piping.
 - b. Uninsulated plastic piping.
 - c. Pipe hangers and supports.
 - d. Metal conduit.

3.4 FIELD QUALITY CONTROL

- A. Dry Film Thickness Testing: Owner may engage the services of a qualified testing and inspecting agency to inspect and test paint for dry film thickness.
 - 1. Contractor shall touch up and restore painted surfaces damaged by testing.
 - 2. If test results show that dry film thickness of applied paint does not comply with paint manufacturer's written recommendations, Contractor shall pay for testing and apply additional coats as needed to provide dry film thickness that complies with paint manufacturer's written recommendations.

3.5 CLEANING AND PROTECTION

A. At end of each workday, remove rubbish, empty cans, rags, and other discarded materials from Project site.

Christopher Williams Architects, LLC CWA Project No.:2313

- B. After completing paint application, clean spattered surfaces. Remove spattered paints by washing, scraping, or other methods. Do not scratch or damage adjacent finished surfaces.
- C. Protect work of other trades against damage from paint application. Correct damage to work of other trades by cleaning, repairing, replacing, and refinishing, as approved by Architect, and leave in an undamaged condition.
- D. At completion of construction activities of other trades, touch up and restore damaged or defaced painted surfaces.

3.6 EXTERIOR PAINTING SCHEDULE

- A. Steel and Iron Substrates Ferrous Metal:
 - 1. Water-Based Light Industrial Coating System [MPI EXT 5.1C] [MPI EXT 5.1M]:
 - a. Prime Coat: Primer, alkyd, anti-corrosive for metal, MPI #79.
 - 1) Benjamin Moore[®] Super Spec HP[®] Alkyd Metal Primer P06 (326 g/L), MPI #79.
 - 2) Shop primer in Section where substrate is specified.
 - b. Intermediate Coat: Light industrial coating, exterior, water based, matching topcoat.
 - c. Topcoat: Light industrial coating, exterior, water based, gloss (MPI Gloss Level 6), MPI #164.
 - 1) Benjamin Moore[®] Ultra Spec[®] HP D.T.M. (Direct-To-Metal) Acrylic Gloss Enamel HP28 (142 g/L), MPI #114, MPI #154, MPI #164.
- B. Galvanized-Metal Substrates:
 - 1. Water-Based Light Industrial Coating System MPI EXT 5.3J:
 - a. Prime Coat: Primer, galvanized, water based, MPI #134.
 - Benjamin Moore[®] Ultra Spec[®] HP Acrylic Metal Primer HP04 (48 g/L).
 - b. Intermediate Coat: Light industrial coating, exterior, water based, matching topcoat.
 - c. Topcoat: Light industrial coating, exterior, water based, gloss (MPI Gloss Level 6), MPI #164.
 - 1) Benjamin Moore[®] Ultra Spec[®] HP D.T.M. (Direct-To-Metal) Acrylic Gloss Enamel HP28 (142 g/L), MPI #114, MPI #154, MPI #164.
- C. Wood Substrates: Wood trim, Doors and Windows.
- 1. Latex over Latex Primer System [MPI EXT 6.3L]:
 - a. Prime Coat: Primer, latex for exterior wood, MPI #6.
 - 1) Benjamin Moore[®] Ultra Spec[®] EXT Primer N558 (45 g/L), MPI #6.
 - b. Intermediate Coat: Latex, exterior, matching topcoat.
 - c. Topcoat: Latex, exterior, semi-gloss (MPI Gloss Level 5), MPI #11.
 - 1) Benjamin Moore[®] Ultra Spec[®] EXT Gloss Finish N449 (46 g/L), MPI #11.
- 2. Latex System [MPI EXT 6.3A]:
 - a. Prime Coat: Primer, alkyd for exterior wood, MPI #5.
 - 1) Benjamin Moore[®] Super Spec[®] Exterior Alkyd Primer 176 (345 g/L), MPI #5.
 - b. Intermediate Coat: Latex, exterior, matching topcoat.
 - c. Topcoat: Latex, exterior, semi-gloss (MPI Gloss Level 5), MPI #11.
 - 1) Benjamin Moore[®] Ultra Spec[®] EXT Gloss Finish N449 (46 g/L), MPI #11.
- D. Wood Substrates: Wood Board Siding.
 - 1. Latex System [MPI EXT 6.3A]:
 - a. Prime Coat: Primer, alkyd for exterior wood, MPI #5.
 - Benjamin Moore[®] Super Spec[®] Exterior Alkyd Primer 176 (345 g/L), MPI #5.
 - b. Intermediate Coat: Latex, exterior, matching topcoat.
 - c. Topcoat: Latex, exterior, low sheen (MPI Gloss Level 3-4), MPI #15.
 - Benjamin Moore[®] Ultra Spec[®] EXT Satin Finish N448 (46 g/L), MPI #15.
- E. Wood Substrates: Wood-based panel products.
 - 1. Latex over Alkyd Primer System [MPI EXT 6.4G]:
 - a. Prime Coat: Primer, alkyd for exterior wood, MPI #5.
 - 1) Benjamin Moore[®] Super Spec[®] Exterior Alkyd Primer 176 (345 g/L), MPI #5.
 - b. Intermediate Coat: Latex, exterior, matching topcoat.
 - c. Topcoat: Latex, exterior, semi-gloss (MPI Gloss Level 5), MPI #11.

1) Benjamin Moore[®] Ultra Spec[®] EXT Gloss Finish N449 (46 g/L), MPI #11.

END OF SECTION 09 9113

SECTION 09 9123 - INTERIOR PAINTING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section includes surface preparation and the application of paint systems on interior the following interior substrates:
 - 1. Concrete masonry units (CMUs).
 - 2. Steel and iron.
 - 3. Wood.
 - 4. Gypsum board.
- B. Related Requirements:
 - 1. Section 09 2900 Gypsum Board

1.3 DEFINITIONS

- A. MPI Gloss Level 1 (Matte): Not more than five units at 60 degrees and 10 units at 85 degrees, in accordance with ASTM D523.
- B. MPI Gloss Level 2 (Flat): Not more than 10 units at 60 degrees and 10 to 35 units at 85 degrees, in accordance with ASTM D523.
- C. MPI Gloss Level 3 (Eggshell): 10 to 25 units at 60 degrees and 10 to 35 units at 85 degrees, in accordance with ASTM D523.
- D. MPI Gloss Level 4 (Satin): 20 to 35 units at 60 degrees and not less than 35 units at 85 degrees, in accordance with ASTM D523.
- E. MPI Gloss Level 5 (Semi-gloss): 35 to 70 units at 60 degrees, in accordance with ASTM D523.
- F. MPI Gloss Level 6 (Gloss): 70 to 85 units at 60 degrees, in accordance with ASTM D523.
- G. MPI Gloss Level 7 (High Gloss): More than 85 units at 60 degrees, in accordance with ASTM D523.

1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product. Include preparation requirements and application instructions.
 - 1. Include Printout of current "MPI Approved Products List" for each product category specified, with the proposed product highlighted.
 - 2. Indicate VOC content.
- B. Sustainable Design Submittals:
 - 1. Product Data: For paints and coatings, indicating VOC content.
 - 2. Laboratory Test Reports: For paints and coatings, indicating compliance with requirements for low-emitting materials.
- C. Samples for Initial Selection: For each type of topcoat product.
- D. Samples for Verification: For each type of paint system and in each color and gloss of topcoat.
 - 1. Submit Samples on rigid backing, 8 inches square.
 - 2. Apply coats on Samples in steps to show each coat required for system.
 - 3. Label each coat of each Sample.
 - 4. Label each Sample for location and application area.
- E. Product List: Cross-reference to paint system and locations of application areas. Use same designations indicated on Drawings and in schedules. Include color designations.

1.5 MAINTENANCE MATERIAL SUBMITTALS

- A. Furnish extra materials, from the same product run, that match products installed and that are packaged with protective covering for storage and identified with labels describing contents.
 - 1. Paint: 5 percent, but not less than 1 gal. of each material and color applied.

1.6 QUALITY ASSURANCE

- A. Mockups: Apply mockups of each paint system indicated and each color and finish selected to verify preliminary selections made under Sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution.
 - 1. Architect will select one surface to represent surfaces and conditions for application of each paint system.
 - a. Vertical and Horizontal Surfaces: Provide samples of at least 100 sq. ft.
 - b. Other Items: Architect will designate items or areas required.

- 2. Final approval of color selections will be based on mockups.
 - a. If preliminary color selections are not approved, apply additional mockups of additional colors selected by Architect at no added cost to Owner.
- 3. Approval of mockups does not constitute approval of deviations from the Contract Documents contained in mockups unless Architect specifically approves such deviations in writing.
- 4. Subject to compliance with requirements, approved mockups may become part of the completed Work if undisturbed at time of Substantial Completion.

1.7 DELIVERY, STORAGE, AND HANDLING

- A. Store materials not in use in tightly covered containers in well-ventilated areas with ambient temperatures continuously maintained at not less than 45 deg F.
 - 1. Maintain containers in clean condition, free of foreign materials and residue.
 - 2. Remove rags and waste from storage areas daily.

1.8 FIELD CONDITIONS

- A. Apply paints only when temperature of surfaces to be painted and ambient air temperatures are between 50 and 95 deg F.
- B. Do not apply paints when relative humidity exceeds 85 percent; at temperatures less than 5 deg F above the dew point; or to damp or wet surfaces.

PART 2 - PRODUCTS

2.1 MANUFACTURERS

- A. Basis-of-Design Product: Subject to compliance with requirements, provide Benjamin Moore & Co.; Products as designated in Interior Painting Schedule or comparable products by one of the following:
 - 1. Behr Process Corporation.
 - 2. Glidden Professional.
 - 3. PPG Architectural Coatings.
 - 4. Sherwin-Williams Company (The).

2.2 PAINT, GENERAL

- A. MPI Standards: Products shall comply with MPI standards indicated and shall be listed in its "MPI Approved Products Lists."
- B. Material Compatibility:

Christopher Williams Architects, LLC CWA Project No.:2313

- 1. Materials for use within each paint system shall be compatible with one another and substrates indicated, under conditions of service and application as demonstrated by manufacturer, based on testing and field experience.
- 2. For each coat in a paint system, products shall be recommended in writing by topcoat manufacturers for use in paint system and on substrate indicated.
- C. VOC Content: For field applications that are inside the weatherproofing system, paints and coatings shall comply with VOC content limits of authorities having jurisdiction and the following VOC content limits:
 - 1. Flat Paints and Coatings: 50 g/L.
 - 2. Nonflat Paints and Coatings: 50 g/L.
 - 3. Dry-Fog Coatings: 150 g/L.
 - 4. Primers, Sealers, and Undercoaters: 100 g/L.
 - 5. Rust-Preventive Coatings: 100 g/L.
 - 6. Zinc-Rich Industrial Maintenance Primers: 100 g/L.
 - 7. Pretreatment Wash Primers: 420 g/L.
 - 8. Shellacs, Clear: 730 g/L.
 - 9. Shellacs, Pigmented: 550 g/L.
- D. Low-Emitting Materials: For field applications that are inside the weatherproofing system, 90 percent of paints and coatings shall comply with the requirements of the California Department of Public Health's "Standard Method for the Testing and Evaluation of Volatile Organic Chemical Emissions from Indoor Sources Using Environmental Chambers."
- E. Colors: [As selected by Architect from manufacturer's full range or as scheduled on the Contract Drawings.
 - 1. Ten percent of surface area will be painted with deep tones.

2.3 SOURCE QUALITY CONTROL

- A. Testing of Paint Materials: Owner reserves the right to invoke the following procedure:
 - 1. Owner will engage the services of a qualified testing agency to sample paint materials. Contractor will be notified in advance and may be present when samples are taken. If paint materials have already been delivered to Project site, samples may be taken at Project site. Samples will be identified, sealed, and certified by testing agency.
 - 2. Testing agency will perform tests for compliance with product requirements.
 - 3. Owner may direct Contractor to stop applying paints if test results show materials being used do not comply with product requirements. Contractor shall remove noncomplying paint materials from Project site, pay for testing, and repaint surfaces painted with rejected materials. Contractor will be required to remove rejected materials from previously painted surfaces if, on repainting with complying materials, the two paints are incompatible.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine substrates and conditions, with Applicator present, for compliance with requirements for maximum moisture content and other conditions affecting performance of the Work.
- B. Maximum Moisture Content of Substrates: When measured with an electronic moisture meter as follows:
 - 1. Masonry (Clay and CMUs): 12 percent.
 - 2. Wood: 15 percent.
 - 3. Gypsum Board: 12 percent.
- C. Gypsum Board Substrates: Verify that finishing compound is sanded smooth.
- D. Verify suitability of substrates, including surface conditions and compatibility, with existing finishes and primers.
- E. Proceed with coating application only after unsatisfactory conditions have been corrected.
 - 1. Application of coating indicates acceptance of surfaces and conditions.

3.2 PREPARATION

- A. Comply with manufacturer's written instructions and recommendations in "MPI Architectural Painting Specification Manual" applicable to substrates and paint systems indicated.
- B. Remove hardware, covers, plates, and similar items already in place that are removable and are not to be painted. If removal is impractical or impossible because of size or weight of item, provide surface-applied protection before surface preparation and painting.
 - 1. After completing painting operations, use workers skilled in the trades involved to reinstall items that were removed. Remove surface-applied protection if any.
- C. Clean substrates of substances that could impair bond of paints, including dust, dirt, oil, grease, and incompatible paints and encapsulants.
 - 1. Remove incompatible primers and reprime substrate with compatible primers or apply tie coat as required to produce paint systems indicated.
- D. Masonry Substrates: Remove efflorescence and chalk. Do not paint surfaces if moisture content or alkalinity of surfaces or mortar joints exceeds that permitted in manufacturer's written instructions.

- E. Steel Substrates: Remove rust, loose mill scale, and shop primer, if any. Clean using methods recommended in writing by paint manufacturer but not less than the following:
 - 1. SSPC-SP 2.
 - 2. SSPC-SP 3.
 - 3. SSPC-SP 7/NACE No. 4.
 - 4. SSPC-SP 11.
- F. Shop-Primed Steel Substrates: Clean field welds, bolted connections, and areas where shop paint is abraded. Paint exposed areas with the same material as used for shop priming to comply with SSPC-PA 1 for touching up shop-primed surfaces.
- G. Galvanized-Metal Substrates: Remove grease and oil residue from galvanized sheet metal by mechanical methods to produce clean, lightly etched surfaces that promote adhesion of subsequently applied paints.
- H. Wood Substrates:
 - 1. Scrape and clean knots, and apply coat of knot sealer before applying primer.
 - 2. Sand surfaces that will be exposed to view, and dust off.
 - 3. Prime edges, ends, faces, undersides, and backsides of wood.
 - 4. After priming, fill holes and imperfections in the finish surfaces with putty or plastic wood filler. Sand smooth when dried.

3.3 APPLICATION

- A. Apply paints in accordance with manufacturer's written instructions and to recommendations in "MPI Manual."
 - 1. Use applicators and techniques suited for paint and substrate indicated.
 - 2. Paint surfaces behind movable equipment and furniture same as similar exposed surfaces. Before final installation, paint surfaces behind permanently fixed equipment or furniture with prime coat only.
 - 3. Paint front and backsides of access panels, removable or hinged covers, and similar hinged items to match exposed surfaces.
 - 4. Do not paint over labels of independent testing agencies or equipment name, identification, performance rating, or nomenclature plates.
 - 5. Primers specified in painting schedule may be omitted on items that are factory primed or factory finished if acceptable to topcoat manufacturers.
- B. Tint each undercoat a lighter shade to facilitate identification of each coat if multiple coats of same material are to be applied. Tint undercoats to match color of topcoat, but provide sufficient difference in shade of undercoats to distinguish each separate coat.
- C. If undercoats or other conditions show through topcoat, apply additional coats until cured film has a uniform paint finish, color, and appearance.
- D. Apply paints to produce surface films without cloudiness, spotting, holidays, laps, brush marks, roller tracking, runs, sags, ropiness, or other surface imperfections. Cut in sharp lines and color breaks.

- E. Painting Fire Suppression, Plumbing, HVAC, Electrical, Communication, and Electronic Safety and Security Work:
 - 1. Paint the following work where exposed in equipment rooms:
 - a. Uninsulated metal piping.
 - b. Pipe hangers and supports.
 - c. Metal conduit.
 - 2. Paint the following work where exposed in occupied spaces:
 - a. Uninsulated metal piping.
 - b. Uninsulated plastic piping.
 - c. Pipe hangers and supports.
 - d. Metal conduit.
 - 3. Paint portions of internal surfaces of metal ducts, without liner, behind air inlets and outlets that are visible from occupied spaces.

3.4 FIELD QUALITY CONTROL

- A. Dry Film Thickness Testing: Owner may engage the services of a qualified testing and inspecting agency to inspect and test paint for dry film thickness.
 - 1. Contractor shall touch up and restore painted surfaces damaged by testing.
 - 2. If test results show that dry film thickness of applied paint does not comply with paint manufacturer's written instructions, Contractor shall pay for testing and apply additional coats as needed to provide dry film thickness that complies with paint manufacturer's written instructions.

3.5 CLEANING AND PROTECTION

- A. At end of each workday, remove rubbish, empty cans, rags, and other discarded materials from Project site.
- B. After completing paint application, clean spattered surfaces. Remove spattered paints by washing, scraping, or other methods. Do not scratch or damage adjacent finished surfaces.
- C. Protect work of other trades against damage from paint application. Correct damage to work of other trades by cleaning, repairing, replacing, and refinishing, as approved by Architect, and leave in an undamaged condition.
- D. At completion of construction activities of other trades, touch up and restore damaged or defaced painted surfaces.

3.6 INTERIOR PAINTING SCHEDULE

- A. CMU Substrates:
 - 1. Latex System [MPI INT 4.2A]:
 - a. Block Filler: Block filler, latex, interior/exterior, MPI #4.
 - 1) Benjamin Moore[®] Ultra Spec[®] Hi-Build Masonry Block Filler 571 (45 g/L), MPI #4; LEED v4; qualifies for CHPS.
 - b. Intermediate Coat: Latex, interior, matching topcoat.
 - c. Topcoat: Latex, interior (MPI Gloss Level 3), MPI #52 and MPI #52 X-Green[™].
 - 1) Benjamin Moore[®] Super Hide[®] Zero-VOC Interior Eggshell 357 (0 g/L), MPI #52, MPI #52 X-Green[™]; LEED v4; qualifies for CHPS.

B. Steel Substrates:

- 1. Institutional Low-Odor/VOC Latex System [MPI INT 5.1S]:
 - a. Prime Coat: Primer, rust inhibitive, water based, MPI #107 and MPI #107 X-Green[™]].
 - Benjamin Moore[®] Ultra Spec[®] HP Acrylic Metal Primer HP04 (48 g/L), MPI #107, MPI #107 X-Green[™], MPI #134; LEED v4; qualifies for CHPS.
 - b. Intermediate Coat: Latex, interior, institutional low odor/VOC, matching topcoat.
 - c. Topcoat: Latex, interior, institutional low odor/VOC (MPI Gloss Level 4), MPI #146 and MPI #146 X-Green[™].
 - Benjamin Moore[®] Ultra Spec[®] 500 Interior Semi-Gloss N539 (0 g/L), MPI #43, MPI #43 X-Green[™], MPI #140, MPI #140 X-Green[™], MPI #146, MPI #146 X-Green[™]; LEED v4; qualifies for CHPS.
- C. Galvanized-Metal Substrates:
 - 1. Latex System [MPI INT 5.3J]:
 - a. Prime Coat: Primer, galvanized, water based, MPI #134 and MPI #107 X-Green[™].
 - 1) Benjamin Moore[®] Ultra Spec[®] HP Acrylic Metal Primer HP04-78 (48 g/L), MPI #134; LEED v4; qualifies for CHPS.
 - b. Intermediate Coat: Latex, interior, matching topcoat.

- c. Topcoat: Latex, interior (MPI Gloss Level 4), MPI #43 and MPI #43 X-Green[™].
 - 1) Benjamin Moore[®] Aura[®] Waterborne Interior Satin Finish 526 (0 g/L), MPI #43, MPI #43 X-Green[™], MPI #140, MPI #140 X-Green[™]; LEED v4; qualifies for CHPS.
 - Benjamin Moore[®] Regal[®] Select Waterborne Interior Pearl Finish 550 (0 g/L), MPI #43, MPI #43 X-Green[™], MPI #140, MPI #140 X-Green[™]; LEED v4; qualifies for CHPS.
 - 3) Benjamin Moore[®] Warerborne Satin Impervo[®] 100% Acrylic Low Lustre Enamel N314 (137 g/L), MPI #43.
 - Benjamin Moore[®] Ultra Spec[®] 500 Interior Semi-Gloss N539 (0 g/L), MPI #43, MPI #43 X-Green[™], MPI #140, MPI #140 X-Green[™], MPI #146, MPI #146 X-Green[™]; LEED v4; qualifies for CHPS.
 - 5) Benjamin Moore[®] Super Hide[®] Zero VOC Interior Semi-Gloss 358 (0 g/L), MPI #43, MPI #43 X-Green[™]; LEED v4.
- D. Wood Substrates: Wood trim, architectural woodwork, doors and windows.
 - 1. Latex over Latex Primer System [MPI INT 6.3T]:
 - a. Prime Coat: Primer, latex, for interior wood, MPI #39.
 - 1) Benjamin Moore[®] Sure Seal[™] Latex Primer Sealer 027 (48 g/L), MPI #6, MPI #39, MPI #137; LEED v4; qualifies for CHPS.
 - b. Intermediate Coat: Latex, interior, matching topcoat.
 - c. Topcoat: Latex, interior, semi-gloss (MPI Gloss Level 5), MPI #54 and MPI #54 X-Green[™].
 - Benjamin Moore[®] Ultra Spec[®] 500 Interior Gloss Finish N540 (0 g/L), MPI #54, MPI #54 X-Green[™], MPI #141, MPI #141 X-Green[™], MPI #147, MPI #147 X-Green[™]; LEED v4; qualifies for CHPS.
- E. Gypsum Board Substrates: Walls:
 - 1. Latex over Latex Sealer System [MPI INT 9.2A]:
 - a. Prime Coat: Primer sealer, latex, interior, MPI #50 and MPI #50 X-Green[™].
 - 1) Benjamin Moore[®] Fresh Start[®] Natura[®] Zero-VOC Primer 511 (0 g/L), MPI #50, MPI #50 X-Green[™], MPI #149, MPI #149 X-Green[™]; qualifies for CHPS.
 - b. Prime Coat: Latex, interior, matching topcoat.
 - c. Intermediate Coat: Latex, interior, matching topcoat.
 - d. Topcoat: Latex, interior, flat (MPI Gloss Level 1), MPI #53 and MPI #53 X-Green[™].
 - 1) Benjamin Moore[®] Super Hide[®] Zero-VOC Interior Flat 355 (0 g/L), MPI #53, MPI #53 X-Green[™]; LEED v4; qualifies for CHPS.

- e. Topcoat: Latex, interior (MPI Gloss Level 3), MPI #52 and MPI #52 X-Green[™].
 - 1) Benjamin Moore[®] Super Hide[®] Zero-VOC Interior Eggshell 357 (0 g/L), MPI #52, MPI #52 X-Green[™]; LEED v4; qualifies for CHPS.
- F. Gypsum Board Substrates: Ceilings and Soffits:
 - 1. Latex over Latex Sealer System [MPI INT 9.2A]:
 - a. Prime Coat: Primer sealer, latex, interior, MPI #50 and MPI #50 X-Green[™].
 - 1) Benjamin Moore[®] Fresh Start[®] Natura[®] Zero-VOC Primer 511 (0 g/L), MPI #50, MPI #50 X-Green[™], MPI #149, MPI #149 X-Green[™]; qualifies for CHPS.
 - b. Prime Coat: Latex, interior, matching topcoat.
 - c. Intermediate Coat: Latex, interior, matching topcoat.
 - d. Topcoat: Latex, interior, flat (MPI Gloss Level 1), MPI #53 and MPI #53 X-Green[™].
 - 1) Benjamin Moore[®] Super Hide[®] Zero-VOC Interior Flat 355 (0 g/L), MPI #53, MPI #53 X-Green[™]; LEED v4; qualifies for CHPS.

END OF SECTION 099123

SECTION 31 1750 – EROSION AND SEDIMENT CONTROL

PART 1 – GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including Rockefeller Archive Center General Conditions and any Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.
- 1.2 DESCRIPTION OF WORK:
 - A. Provide all means necessary to install, inspect and maintain, and remove temporary erosion and sediment control measures as required to minimize the erosion and unspecified transport of soil and sediment from the site. THE CONTRACTOR SHALL MINIMIZE SITE DISTURBANCE.
- 1.3 QUALITY ASSURANCE:
 - A. General
 - 1. Install and maintain all soil erosion and sediment control measures in compliance the Stormwater Permit conditions.
 - 2. Contractor shall implement all measures required prior to the start of any other activities on site.
 - 3. Grade and maintain site at all times such that all storm water runoff from disturbed areas is diverted to soil erosion and sedimentation control facilities.
 - 4. No soil, not protected by erosion and sedimentation control measures, can be disturbed at any time.
 - 5. The Contractor shall comply with applicable Federal, State, and local regulations relating to the prevention and abatement of pollution.
 - 6. The Contractor shall be responsible for maintaining all erosion and sediment control devices and shall be required to provide measures to immediately correct problems encountered in the field whether or not the measures are shown on the plan. Measures that include the installation of erosion control blankets or reducing the amount of exposed soil may be necessary.
 - B. Product Data: Submit manufacturer's catalogue cuts, specifications and installation instructions for silt fences, filter fabrics and erosion control blankets.
 - C. Product Stockpiling: Stockpiles of stabilization measures such as haybales, silt fence, 1½-inch gravel for check dams, filter fabric, and mulch shall be maintained at the site for use in stabilizing disturbed areas in advance of severe weather conditions.

PART 2 - PRODUCTS

- 2.1 INLET PROTECTION: Filtrexx inlet protection or approved equal.
- 2.2 DEWATERING PITs: Number and location to be determined by contractor.
- 2.3 SILT FENCE: Silt fence fabric shall be Mirafi 100X or equal. Wood posts shall be of sound quality hardwood, a minimum 36-inches long and 2-inches square. Metal posts shall be standard T and U section weighing not less than one pound per linear foot. Wire fence backing shall be a minimum 14½ gage with a maximum 6-inch mesh opening and securely attached to fence posts. Posts shall extend a minimum of 16-inches into the ground.
- 2.4 HAY BALE BARRIERS: Wood posts shall be of sound quality hardwood, a minimum 36-inches long. Metal posts shall be standard T and U section weighing not less than one pound per linear foot.
- 2.5 FILTER FABRIC: Filter fabric shall be Mirafi 600x.
- 2.6 TEMPORARY STABILIZATION
 - A. Establishment of Temporary Grass Cover: Prepare seed bed, scarify if compacted, remove debris and obstacles such as rocks and stumps, and seed within 24 hours. Amend soil, lime soil to pH of 6.0 and fertilize at a rate of 14 pounds per 1,000 square feet with a 5-10-10 or equivalent fertilizer. Work amendments a minimum of 4-inches into soil. If seeding in October/November, seed shall be Certified Aroostook winter rye at 100 pounds per acre, otherwise seed shall be ryegrass (annual or perennial) at 30 pounds per acre.
 - B. Mulch/RECP: Small grain straw mulch or Type 1 erosion control blankets as specified on drawing SP-5.1. Straw mulch shall be applied at a rate of two tons (100 to 120 bales) per acre. Erosion control blankets shall be BonTerra S2 installed as recommended by manufacturer.
 - C. Install Temporary Stabilization within 24-hours after the end of construction activities in an area unless there is snow cover or construction activities will resume within seven days.
 - D. Tackifier: When covering between October and April, cover exposed soils with hydroseed and tackifier with the following application rates:

1.	Slopes less than 3 Vert.:12 Horiz.	75 lbs/ac
2.	Slopes between 3 Vert.:1 Horiz. and 2 Vert.:1 Horiz.	100 lbs/ac
3.	Slopes greater than 2 Vert.:1 Horiz.	150 lbs/ac
-		

Acceptable product: Conwed Fibers Con-Tack AT Tackifier as manufactured by Profile Products, LLC (800) 366-1180

PART 3 - EXECUTION

3.1 GENERAL: Install and remove measures as required. The measures shall be maintained until permanent protection of the contributing watershed is approved by the Owner's Representative. All storm drainage outlets that have been silted due to the work will be cleaned, as required.

- 3.2 INSPECTIONS: Inspect and report measures daily and within 24-hours of the end of a rainfall event. All inspections are to be made by a NYS trained and certified contractor. Stabilized areas will be inspected monthly until the entire site is stabilized.
- 3.3 MAINTENANCE: Maintenance and deficiencies shall be completed within two calendar days of determining its need.
- 3.4 PAVEMENT: Provide temporary pavement when adjacent to traffic lanes and when directed by the Engineer.

END OF SECTION 31 1750

SECTION 31 2000 – EARTHWORK

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including Rockefeller Archive Center General Conditions and any Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.
- B. Related sections include but are not limited to:
 - 1. Section 31 1750 Erosion and Sediment Control
- C. Any conflict in requirements between the Drawings, Specifications, Geotechnical Report, or other design documents should be interpreted in favor of the most restrictive requirement unless otherwise directed by the Engineer.

1.2 SUMMARY

- A. Section Includes:
 - 1. Definition of excavation, fill and backfill materials.
 - 2. The preparation and dewatering requirements for open excavations and/or structures.
 - 3. General excavation requirements.
 - 4. Excavation requirements for buildings and structures.
 - 5. Excavation requirements for preparing subgrades for walks and pavements.
 - 6. Excavation requirements for utility trenches.
 - 7. Excavation of landscaped areas.
 - 8. Subgrade inspection requirements.
 - 9. General backfill requirements.
 - 10. Backfill requirements for utility trenches.
 - 11. Fill material requirements.
 - 12. Soil moisture control requirements.
 - 13. Compaction requirements of backfills and fill material.
 - 14. Bedding course placement under slabs-on-grade, walks and other structures.
 - 15. Subbase course placement under asphaltic concrete pavements.
 - 16. Drainage course for porous pavement and underdrain systems.
 - 17. Subsurface drainage.
 - 18. Field quality control of subgrade preparation, material backfill and compaction testing.
 - 19. Protection of excavated and graded areas.
 - 20. Storage of soil materials.
 - 21. Unauthorized excavation.
 - 22. Removal of excess and unsuitable material from the site.

1.3 DEFINITIONS

A. Backfill: Soil material or controlled low-strength material used to fill an excavation.

- 1. Initial Backfill: Backfill placed beneath, beside and over pipe in a trench, including haunches to support sides of pipe.
- 2. Final Backfill: Backfill placed over initial backfill to fill a trench to the top of the proposed subgrade as shown and detailed on the plans.
- B. Structural Fill: Fill material placed under structures after removal of unsuitable bearing materials. Material shall be 3/4-inch, clean aggregate or other material as approved by the project's geotechnical engineer.
- C. Subbase Course: Aggregate layer placed between the existing subgrade and hot-mix asphalt paving.
- D. Bedding Course: Aggregate layer placed over the excavated subgrade in a trench before laying pipe and in other open excavated areas to support new curbing, walks, concrete stairs, slabs-on-grade, manholes or other structures. The bedding material shall also be used to backfill trenches to the depths and/or limits detailed on the plans. Sand bedding material shall be used in place of the aggregate material specified where required by either the local utility, Owner's representative or Project Engineer.
- E. Borrow Soil: Satisfactory soil imported from off-site for use as fill or backfill, or soil from on-site sources meeting the specifications for fill material and determined suitable for on-site use.
- F. Drainage Course: Aggregate layer supporting the collection and transporting of water.
- G. Earth Excavation: Defined to include removal, and if required proper disposal off-site, of the following:
 - 1. Soil and all other materials encountered of any name and nature that are not classified as rock excavation or unauthorized excavation.
 - 2. Hardpan, loose or decomposed bedrock or other such material that may require intermittent drilling and wedging to increase production or facilitate handling of the material with equipment normally used in the particular excavation operation.
- H. Authorized Additional Excavation: Excavation below subgrade elevations or beyond indicated lines and dimensions as shown on the contract drawings. Authorized additional excavation and replacement material will be paid for according to Contract provisions.
- I. Bulk Excavation: Excavation more than 10 feet in width and more than 30 feet in length.
- J. Unauthorized Excavation: Excavation below subgrade elevations or beyond indicated lines and dimensions without direction by Owner, Construction Manager, or Engineer. Unauthorized excavation, as well as remedial work directed by Engineer, shall be without additional compensation. Remedial work may include, but not be limited to, replacement of the unauthorized excavation by backfilling and compacting as specified for authorized excavations of the same classification, unless otherwise directed by the Owner's Representatives.
- K. Fill: Soil materials used to raise existing grades.
- L. Rock: Rock material in beds, ledges, unstratified masses, conglomerate deposits, and boulders of rock material that exceed 1 cubic yard for bulk excavation or 3/4 cubic yard for footing, trench, and pit excavation that cannot be removed by rock excavating equipment equivalent to the following in size and performance ratings, without

systematic drilling, ram hammering, ripping, or blasting, when permitted.

- 1) Excavation of Footings, Trenches, and Pits: Late-model, track-mounted hydraulic excavator; equipped with a 42-inch wide, maximum, short-tip-radius rock bucket; rated at not less than 138-hp flywheel power with bucket-curling force of not less than 28,700 lbf and stick-crowd force of not less than 18,400 lbf with extra-long reach boom; measured according to SAE J-1179.
- 2) Bulk Excavation: Late-model, track-mounted loader; rated at not less than 230hp flywheel power and developing a minimum of 47,992-lbf breakout force with a general-purpose bare bucket; measured according to SAE J-732.
- M. Rock: Rock material in beds, ledges, unstratified masses, conglomerate deposits, and boulders of rock material 3/4 cu. yd. or more in volume that exceed a standard penetration resistance of 100 blows/2 inches when tested by a geotechnical testing agency, according to ASTM D 1586
- N. Structures: Buildings, footings, foundations, retaining walls, slabs, tanks, inlets, catch basins, manholes, curbs, mechanical and electrical appurtenances, or other man-made stationary features constructed above or below the ground surface.
- O. Subgrade: Uppermost surface of an excavation or the top surface of a fill or backfill immediately below the subbase or bedding course, drainage or reservoir course, or topsoil materials.
- P. Recycled Material: The use of recycled material from any off-site source will not be permitted.
- Q. Utilities: On-site underground pipes, conduits, ducts, tunnels, and cables, as well as underground services within buildings.

1.4 SUBMITTALS

- A. Product Data: For each type of the following manufactured products required:
 - 1. Geotextiles.
 - 2. Controlled Low Strength Material (CLSM).
 - 3. Warning tapes.
 - 4. Imported borrow fill and aggregate
- B. Samples for Verification: For the following products, in sizes indicated below:
 - 1. Geotextile: 12-inches by 12-inches.
 - 2. Warning Tape: 12-inches long; of each color.
 - 3. Earthwork, borrow fill, aggregate materials; 5-gallon pail containing a representative sample plus a sealed quart plastic bag containing a representative moisture sample of material passing the No. 4 sieve.
- C. Qualification Data: For qualified testing agency.
- D. Material Test Reports: For each on-site and borrow soil material proposed for fill and backfill as follows:
 - 1. Classification according to ASTM D 2487.

- 2. Laboratory compaction curve according to ASTM D 1557.
- E. Pre-Excavation Photographs or Videotape: Show existing conditions of adjoining construction and site improvements, including finish surfaces that might be misconstrued as damage caused by earth moving operations. Submit before earth moving begins.
- 1.5 QUALITY ASSURANCE
 - A. Geotechnical Testing Agency Qualifications: Qualified according to ASTM E 329 and ASTM D 3740 for testing indicated.

1.6 PROJECT CONDITIONS

- A. Traffic: Minimize interference with adjoining roads, streets, walks, and other adjacent occupied or used facilities during earth moving operations.
 - 1. Do not close or obstruct streets, walks, or other adjacent occupied or used facilities without permission from Owner and authorities having jurisdiction.
 - 2. Provide alternate routes around closed or obstructed traffic ways if required by Owner or authorities having jurisdiction.
- B. Utility Locator Service: Notify "Call Before You Dig" for area where Project is located before beginning earth moving operations.
- C. Do not commence earth moving operations until temporary erosion- and sedimentation-control measures are in place.
- D. Do not commence earth moving operations until plant-protection measures are in place.
- E. Do not commence earth moving operations until all subsurface utilities have been located and marked in the field.
- F. The following practices are prohibited within areas not identified as work zones shown on the approved Construction Plan:
 - 1. Storage of construction materials, debris, or excavated material.
 - 2. Parking vehicles or equipment.
 - 3. Erection of sheds or structures.
 - 4. Impoundment of water.
 - 5. Excavation or other digging unless otherwise indicated.
 - 6. Attachment of signs to or wrapping materials around trees or plants unless otherwise indicated.
- G. Subsurface Information:
 - 1. The Owner makes no predictions or representations regarding the character or extent of soil, rock or other subsurface condition to be encountered during the work. The Contractor shall make his own deductions on the subsurface conditions which may affect the methods or cost of construction of the work hereunder, and he agrees that he will make no claims for damages or compensations, except as are provided under the agreement, should he find conditions during the progress of the work different from those as calculated and/or anticipated by him. Borings and other exploratory operations may be

performed by the Contractor, at the Contractor's option and following the Owner's approval. No change in the Contract Sum will be authorized for such additional exploration undertaken by the Contractor.

- 2. The Contractor shall make his own deductions of the subsurface conditions which may affect the methods or cost of construction of the work hereunder, and he agrees that he will make no claims for damages or compensations resulting from the subsurface conditions.
- 3. The Contractor, by careful examination, shall inform himself as to the nature and location of the work; the confirmation of the ground, the nature of the subsurface conditions; the locations of the groundwater table; the character, quality and quantity of the materials to be encountered; the character of the equipment and facilities needed preliminary to and during the execution of the work; the general and local conditions, water levels and all other matters which can in any way affect the work.
- 4. The Contractor shall be held to have visited the site and to have familiarized himself with the existing conditions of adjoining properties and the proposed sequence of construction.
- 5. The Contractor shall investigate the conditions of public thoroughfares and roads as to availability, clearances, loads, limits, restrictions and other limitations affecting transportation to, ingress and egress of the site of the work. The Contractor shall conform to all New York State and Federal regulations in regard to the transportation of materials to and from and at the job site and shall secure in advance such permits as may be required.

PART 2 - PRODUCTS

2.1 SOIL MATERIALS

- A. General: Provide borrow soil materials when sufficient satisfactory soil materials are not available from excavations.
- B. Satisfactory Soils: Soil Classification Groups GW, GP, GM, SW, and SP according to ASTM D 2487, or a combination of these groups; free of rock or gravel larger than 3-inches in any dimension, debris, waste, frozen materials, vegetation, and other deleterious matter.
- C. Unsatisfactory Soils: Soil Classification Groups SM, GC, SC, CL, ML, OL, CH, MH, OH, and PT according to ASTM D 2487, or a combination of these groups.
 - 1. Unsatisfactory soils also include satisfactory soils not maintained within 2 percent of optimum moisture content at time of compaction.
 - 2. Unsatisfactory materials also include materials below structures and/or foundations determined by the Owner's Representatives to be unsatisfactory bearing materials.
- D. Subbase Course: Naturally or artificially graded mixture of natural or crushed gravel, crushed stone, and natural or crushed sand; NYSDOT Type 1. The use of recycled material from any off-site source will not be permitted and recycled material from onsite demolition may not be used without written authorization from the owner.

- E. Structural Fill: Naturally or artificially, well graded mixture of natural or crushed gravel, crushed stone, and natural or crushed sand; ASTM D 2940; with at least 90 percent passing a 1¹/₂ -inch sieve and not more than 12 percent passing a No. 200 sieve. This material must be approved by the Owner's Representatives.
- F. Bedding Course: Naturally or artificially graded mixture of natural or crushed gravel or crushed stone; ASTM D 2940; except with 100 percent passing a 1-inch sieve and 0 to 5 percent passing a No. 8 sieve.
- G. Trap Rock: Narrowly graded mixture of washed crushed stone ASTM D 448; coarseaggregate grading Size 1; with 100 percent passing a 4-inch sieve and 0 to 15 percent passing a 1¹/₂-inch sieve.
- H. Sand: ASTM C 33; fine aggregate.
- I. Topsoil and other Planting Media: See Division 32 Section 329000 Planting Media Preparation and Placement.
- J. CLSM high slump mixture of Portland cement, fly ash and fine aggregate formulated, licensed and marked K-Krete or equal.
 - 1. Provide mixture with a 28 day compressive strength of 200 psi with no measureable shrinkage or surface settlement.

2.2 GEOTEXTILES

- A. Subsurface Drainage Geotextile: Filter fabric, manufactured for subsurface drainage applications, should meet the following minimum requirements as described in the "Subsurface Investigation and Analysis Report":
 - 1. Minimum Permittivity (ASTM D4991) = 0.2 sec
 - 2. Maximum AOS (ASTM D4751) = 0.25 mm
- B. Acceptable Products:
 - 1. Mirafi 140N by Mirafi, Inc. or approved equal.

2.3 ACCESSORIES

- A. Detectable Warning Tape: Acid- and alkali-resistant, polyethylene film warning tape manufactured for marking and identifying underground utilities, a minimum of 6-inches wide and 4 mils thick, continuously inscribed with a description of the utility, with metallic core encased in a protective jacket for corrosion protection, detectable by metal detector when tape is buried up to 30-inches deep; colored as follows:
 - 1. Red: Electric.
 - 2. Yellow: Gas, oil, steam, and dangerous materials.
 - 3. Orange: Telephone and other communications.
 - 4. Blue: Water systems.
 - 5. Green: Sewer systems.

PART 3 - EXECUTION

3.1 PREPARATION

- A. Protect structures, utilities, sidewalks, pavements, and other facilities from damage caused by settlement, lateral movement, undermining, washout, and other hazards created by earth moving operations.
- B. Protect and maintain erosion and sedimentation controls during earth moving operations.
- C. Protect subgrades and foundation soils from freezing temperatures and frost. Remove temporary protection before placing subsequent materials.

3.2 DEWATERING

- A. Prevent surface water and ground water from entering excavations, from ponding on prepared subgrades, and from flooding Project site and surrounding area.
- B. Protect subgrades from softening, undermining, washout, and damage by rain or water accumulation.
 - 1. Reroute surface water runoff away from excavated areas. Do not allow water to accumulate in excavations. Do not use excavated trenches as temporary drainage ditches.

3.3 EXCAVATION, GENERAL

- A. Classified Excavation: Excavate to subgrade elevations. Material to be excavated will be classified as earth and rock. Do not excavate rock until it has been classified and cross sectioned by the Owner's Representatives. The Contract Sum will be adjusted for rock excavation according to unit prices included in the Contract Documents. Changes in the Contract Time may be authorized for rock excavation.
 - 1. Earth excavation includes excavating pavements and obstructions visible on surface; underground structures, utilities, and other items indicated to be removed; together with soil, boulders, and other materials not classified as rock or unauthorized excavation.
 - a. Intermittent drilling; if permitted; ram hammering; or ripping of material not classified as rock excavation is earth excavation.
 - 2. Rock excavation includes removal and disposal of rock if material cannot be reused on-site or found to be surplus excavated material. Remove rock to lines and subgrade elevations indicated to permitinstallation of permanent construction without exceeding the following dimensions:
 - a. 12-inches outside of minimum required dimensions of concrete cast against grade.
 - b. 12-inches beneath bottom of concrete slabs-on-grade.
 - c. 12-inches beneath pipe in trenches, and the greater of 24-inches wider than pipe or 30-inches wide.

3.4 EXCAVATION FOR STRUCTURES

- A. Excavate to indicated elevations and dimensions within a tolerance of plus or minus 1-inch. If applicable, extend excavations a sufficient distance from structures for placing and removing concrete formwork, for installing services and other construction, and for inspections.
 - 1. Excavation for Mechanical or Electrical Utility Structures: Excavate to elevations and dimensions indicated within a tolerance of plus or minus 1-inch. Do not disturb bottom of excavations intended as bearing surfaces.
- B. Excavations at Edges of Tree- and Plant-Protection Zones:
 - 1. Excavate by hand to indicated lines, cross sections, elevations, and subgrades. Use narrow-tine spading forks to comb soil and expose roots. Do not break, tear, or chop exposed roots. Do not use mechanical equipment that rips, tears, or pulls roots.

3.5 EXCAVATION FOR WALKS AND PAVEMENTS

A. Excavate surfaces under walks and pavements to indicated lines, cross sections, elevations, and subgrades.

3.6 EXCAVATION FOR UTILITY TRENCHES

- A. Excavate trenches to indicated gradients, lines, depths, and elevations.
- B. Excavate trenches to uniform widths to provide the following clearance on each side of pipe or conduit. Excavate trench walls vertically from trench bottom to 12-inches higher than top of pipe or conduit unless otherwise indicated.
 - 1. Clearance: 12-inches each side of pipe or conduit.
- C. Trench Bottoms: Excavate trenches 6-inches deeper than bottom of pipe and conduit elevations to allow for bedding course. Hand-excavate deeper for bells of pipe.

3.7 EXCAVATION OF AREAS TO BE LANDSCAPED

- A. Excavate existing soil to the depths required to prepare the subgrade to receive topsoil, amended soil, drainage course material or other material as detailed on the plans. Loosen subgrade using rototillers to a minimum depth of 6-inches. Remove all stones larger than 1-inch in any dimension and all sticks, roots, rubbish, and other extraneous matter within the planted areas and legally dispose of them off the Owner's property.
- B. If underground utilities, rock or groundwater conditions are encountered at an elevation at or above the elevation of the required subgrade the Contractor shall notify the owner's representative immediately.

3.8 BACKFILL

- A. Place and compact backfill in excavations promptly, but not before completing the following:
 - 1. Construction below finish grade including, where applicable, sub-drainage.

- 2. Surveying locations of underground utilities for Record Documents.
- 3. Testing and inspecting underground utilities.
- 4. Removing concrete formwork.
- 5. Removing trash and debris.
- 6. Removing temporary shoring and bracing, and sheeting.
- 7. Installing permanent or temporary horizontal bracing on horizontally supported walls.
- B. Place backfill on subgrades free of mud, frost, snow, or ice.

3.9 UTILITY TRENCH BACKFILL

- A. Place backfill on subgrades free of mud, frost, snow, or ice.
- B. Place and compact bedding course on trench bottoms and where indicated. Shape bedding course to provide continuous support for bells, joints, and barrels of pipes and for joints, fittings, and bodies of conduits.
- C. Backfill voids with satisfactory soil while removing shoring and bracing.
- D. Place and compact initial backfill of satisfactory soil, free of particles larger than 1-inch in any dimension, to a height of 12-inches over the pipe or conduit.
 - 1. Carefully compact initial backfill under pipe haunches and compact evenly up on both sides and along the full length of piping or conduit to avoid damage or displacement of piping or conduit. Coordinate backfilling with utilities testing.
- E. Place and compact final backfill of satisfactory soil or Control Density Backfill Material to final subgrade elevation.
- F. Install warning tape directly above utilities, 12-inches below finished grade, except 6-inches below subgrade under pavements and slabs.
- 3.10 FILL MATERIAL
 - A. Plow, scarify, bench, or break up sloped surfaces steeper than 1 vertical to 4 horizontal so fill material will bond with existing material.
 - B. Place and compact fill material in layers to required elevations as follows:
 - 1. Under grass and planted areas, use satisfactory soil material.
 - 2. Under walks and pavements, gravel material.
 - 3. Under steps and ramps, use structural fill.
 - 4. Under building slabs, use structural fill.
 - C. Place fill on subgrades free of mud, frost, snow, or ice.

3.11 SOIL MOISTURE CONTROL

- A. Uniformly moisten or aerate subgrade and each subsequent fill or backfill soil layer before compaction to within 2 percent of optimum moisture content.
 - 1. Do not place backfill or fill soil material on surfaces that are muddy, frozen, or contain frost or ice.
 - 2. Remove and replace, or scarify and air dry, otherwise satisfactory soil material

that exceeds optimum moisture content by 2 percent and is too wet to compact to specified dry unit weight.

3.12 COMPACTION OF BACKFILLS AND FILL MATERIAL

- A. Place backfill and fill materials in layers not more than 12-inches in loose depth for material compacted by heavy compaction equipment, and not more than 4-inches in loose depth for material compacted by hand-operated tampers.
- B. Place backfill and fill materials evenly on all sides of structures to required elevations, and uniformly along the full length of each structure.
- C. Compact materials to not less than the following percentages of maximum dry unit weight according to ASTM D 1557:
 - 1. Under structures, steps, and pavements, scarify and re-compact top 12-inches of existing subgrade and each layer of backfill or fill material at 95 percent.
 - 2. Under walkways, scarify and re-compact top 6-inches below subgrade and compact each layer of backfill or fill material at 95 percent.
 - 3. Under turf or unpaved areas, scarify and re-compact top 6-inches below subgrade and compact each layer of backfill or fill material at 90 percent.
 - 4. For utility trenches, compact each layer of initial and final backfill soil material.

3.13 SUBBASE COURSE UNDER ASPHALTIC CONCRETE PAVEMENTS

- A. Place subbase course on existing and/or compacted subgrades free of mud, frost, snow, or ice.
- B. On prepared subgrade, place subbase course under pavements as follows:
 - 1. Place subbase course material over subgrade and under hot-mix asphalt pavement.
 - 2. Shape subbase course to required crown elevations and cross-slope grades.
 - 3. Place subbase course 12-inches or less in compacted thickness in a single layer.
 - 4. Place subbase course that exceeds 12-inches in compacted thickness in layers of equal thickness, with no compacted layer more than 12-inches thick or less than 3-inches thick.
 - 5. Compact subbase course at optimum moisture content to required grades, lines, cross sections, and thickness to not less than 95 percent of maximum dry unit weight according to ASTM D 1557.
- 3.14 BEDDING COURSE UNDER SLABS-ON-GRADE, WALKS AND MISC. STRUCTURES
 - A. Place bedding course on subgrades free of mud, frost, snow, or ice.
 - B. On prepared subgrade, place and compact bedding course under slabs-on-grade, walks and other structures as follows:
 - 1. Place bedding course 12-inches or less in compacted thickness in a single layer.
 - 2. Place bedding course that exceeds 12-inches in compacted thickness in layers of equal thickness, with no compacted layer more than 12-inches thick or less than 3-inches thick.

3. Compact each layer of bedding course to required cross sections and thicknesses to not less than 95 percent of maximum dry unit weight according to ASTM D 698.

3.15 SUBSURFACE DRAINAGE

- A. Subsurface Drain: Place subsurface drainage geotextile, drainage course filter material and sub-drainage pipe in accordance with the details shown on the plans.
 - 1. Surround drain pipe with 6-inches minimum drainage course material (as specified herein) or as detailed on plans.

3.16 FIELD QUALITY CONTROL

- A. Soil Material Testing: A representative sample as required by the owner's representatives from a truck load of imported material shall be tested for conformance to the specifications.
- B. Imported Material that does not meet the specified gradation shall be removed from the site and replaced with conforming material.

3.17 GRADING

- A. General: Uniformly grade areas to a smooth surface, free of irregular surface changes. Comply with compaction requirements and grade to cross sections, lines, and elevations indicated.
 - 1. Provide a smooth transition between adjacent existing grades and new grades.
 - 2. Cut out soft spots, fill low spots, and trim high spots to comply with required surface tolerances.
- B. Site Rough Grading: Slope grades to direct water to prevent ponding. Finish subgrades to required elevations within the following tolerances:
 - 1. Turf Areas, Planting Areas or other Unpaved Areas: Plus or minus 1-inch of the proposed top of subgrade elevation to receive topsoil, planting and/or amended soil mix, drainage course material or other material to the depths detailed on the plans.
 - 2. Concrete Pavements, Walks and other structures: Less than 1/2 inch of the proposed bottom of the bedding course as detailed on the plans.
 - 3. Asphalt Pavements: Less than 1/2 inch of the proposed bottom of the subbase course as detailed on the plans.

3.18 PROTECTION OF THE WORK AREAS

- A. Protecting Graded Areas: Protect newly graded areas from traffic, freezing, and erosion. Keep free of trash and debris.
- B. Repair and reestablish grades to specified tolerances where completed or partially completed surfaces become eroded, rutted, settled, or where they lose compaction due to subsequent construction operations or weather conditions.
 - 1. Scarify or remove and replace soil material to depth as directed by Owners Representative; reshape and re-compact.

- C. Where settling occurs before Project correction period elapses, remove finished surfacing, backfill with additional soil material, compact, and reconstruct surfacing.
 - 1. Restore appearance, quality, and condition of finished surfacing to match adjacent work, and eliminate evidence of restoration to greatest extent possible.

3.19 STORAGE OF SOIL MATERIALS

- A. Stockpile borrows soil materials and excavated satisfactory soil materials without intermixing. Place, grade, and shape stockpiles to drain surface water. Cover to prevent windblown dust.
 - 1. Stockpile soil materials away from edge of excavations. Do not store within drip line of remaining trees.
- 3.20 UNAUTHORIZED EXCAVATION
 - A. Fill unauthorized excavation under foundations or wall footings by extending bottom elevation of concrete foundation or footing to excavation bottom, without altering top elevation. Lean concrete fill, with 28-day compressive strength of 2,500 psi, may be used when approved by the Owner's Representatives.
 - 1. Fill unauthorized excavations under other construction, pipe, or conduit as directed by the Owner's Representative.
- 3.21 DISPOSAL OF SURPLUS AND WASTE MATERIALS
 - A. Unless directed otherwise, transport surplus satisfactory and unsatisfactory soil off Owner's property.
 - 1. Remove waste materials, trash, and debris, and legally dispose of them off Owner's property.

END OF SECTION 31 2000

SECTION 31 5000 – EXCAVATION SUPPORT AND PROTECTION

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including Rockefeller Archive Center General Conditions and any Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.
- B. Related Sections:
 - 1. Division 01 Section Temporary Facilities and Controls for temporary utilities and support facilities.
 - 2. Division 31 Section Dewatering for dewatering system for excavations.

1.2 SUMMARY

A. Section includes temporary excavation support and protection systems.

1.3 PERFORMANCE REQUIREMENTS

- A. Furnish, install, monitor, and maintain excavation support and protection system capable of supporting excavation sidewalls and of resisting soil and hydrostatic pressure and superimposed and construction loads.
 - 1. Delegated Design: Design excavation support and protection system, including comprehensive engineering analysis by a qualified professional engineer, using performance requirements and design criteria indicated.
 - 2. Prevent surface water from entering excavations by grading, dikes, or other means.
 - 3. Install excavation support and protection systems without damaging existing buildings, structures, and site improvements adjacent to excavation.
 - 4. Monitor vibrations, settlements, and movements.

1.4 SUBMITTALS

- A. Shop Drawings: For excavation support and protection system.
- B. Delegated-Design Submittal: For excavation support and protection system indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.

1.5 QUALITY ASSURANCE

- A. Pre-Installation Conference: Conduct conference at Project site.
 - 1. Review methods and procedures related to excavation support and protection system including, but not limited to, the following:

- a. Existing utilities and subsurface conditions.
- b. Proposed excavations.
- c. Proposed equipment.
- d. Monitoring of excavation support and protection system.
- e. Working area location and stability.
- f. Abandonment or removal of excavation support and protection system.

1.6 PROJECT CONDITIONS

- A. Interruption of Existing Utilities: Do not interrupt any utility serving facilities occupied by Owner or others unless permitted under the following conditions and then only after arranging to provide temporary utility according to requirements indicated:
 - 1. Notify Construction Manager and Owner no fewer than three (3) days in advance of proposed interruption of utility.

PART 2 - PRODUCTS

2.1 MATERIALS

- A. General: Provide materials that are either new or in serviceable condition.
- B. Structural Steel: ASTM A 36/A 36M, ASTM A 690/A 690M, or ASTM A 992/A 992M.
- C. Steel Sheet Piling: ASTM A 328/A 328M, ASTM A 572/A 572M, or ASTM A 690/A 690M; with continuous interlocks.
- D. Wood Lagging: Lumber, mixed hardwood, nominal rough thickness of 3 inches.

PART 3 - EXECUTION

3.1 PREPARATION

- A. Protect structures, utilities, sidewalks, pavements, and other facilities from damage caused by settlement, lateral movement, undermining, washout, and other hazards that could develop during excavation support and protection system operations.
 - 1. Shore, support, and protect utilities encountered.
- B. Install excavation support and protection systems to ensure minimum interference with roads, streets, walks, and other adjacent occupied and used facilities.
 - 1. Do not close or obstruct streets, walks, or other adjacent occupied or used facilities without permission from Owner and authorities having jurisdiction. Provide alternate routes around closed or obstructed traffic ways if required by authorities having jurisdiction.

Locate excavation support and protection systems clear of permanent construction so that forming and finishing of concrete surfaces are not impeded. Monitor excavation support and protection systems daily during excavation progress and for as long as excavation remains open. Promptly correct bulges, breakage, or other evidence of movement to ensure that excavation support and protection systems remain stable.

C. Promptly repair damages to adjacent facilities caused by installing excavation support and protection systems.

3.2 SOLDIER PILES AND LAGGING

- A. Install steel soldier piles before starting excavation. Extend soldier piles below excavation grade level to depths adequate to prevent lateral movement. Space soldier piles at regular intervals not to exceed allowable flexural strength of wood lagging. Accurately align exposed faces of flanges to vary not more than 2 inches.
- B. Install wood lagging within flanges of soldier piles as excavation proceeds. Trim excavation as required to install lagging. Fill voids behind lagging with soil, and compact.

3.3 REMOVAL AND REPAIRS

- A. Remove excavation support and protection systems when construction has progressed sufficiently to support excavation and bear soil and hydrostatic pressures. Remove in stages to avoid disturbing underlying soils or damaging structures, pavements, facilities, and utilities.
 - 1. Remove excavation support and protection systems to a minimum depth of 48 inches below overlaying construction and abandon remainder.
 - 2. Fill voids immediately with approved backfill compacted to density specified in Section 31 2000 Earthwork.
 - 3. Repair or replace, as approved by Architect, adjacent work damaged or displaced by removing excavation support and protection systems.
- B. Leave excavation support and protection systems permanently in place.

END OF SECTION 31 5000

SECTION 32 1216 – ASPHALT CONCRETE PAVING

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including Rockefeller Archive Center General Conditions and any Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 STREETS AND SIDEWALKS:

- A. General Requirements
 - 1. The Contractor shall furnish, place, construct and incorporate into the work, parking and driveway pavements where indicated on the plans in accordance with the Town of Mount Pleasant requirements.
 - 2. The Contractor is further directed to the Construction Details shown on the contract drawings.

1.3 DESCRIPTION OF WORK:

- A. General
 - 1. Extent of asphalt concrete paving work is shown on the drawings including parking areas, driveways and temporary and permanent pavement replacement in areas where existing street pavements have been removed due to trenching operations.
 - 2. Prepared subbase is specified in Section 31 2000 Earthwork.
- B. Aggregate Subbase
 - 1. Where applicable, is specified in Section 31 2000 Earthwork.
- C. Material Certificates
 - 1. Provide copies of material certificates signed by material producer and Contractor, certifying that each material item complies with or exceeds specified requirements.
- D. Codes and Standards
 - 1. Comply with New York State Department of Transportation standard specifications, latest edition and with the Westchester County Departments of Public Works and Transportation governing regulations.
- E. Weather Limitations
 - 1. Apply prime and tack coats when ambient temperature is above 50°F and when temperature has not been below 35°F for 12 hours immediately prior to application. Do not apply when base is wet or contains an excess of moisture.
 - 2. Construct asphalt concrete surface course when atmospheric temperature is above 40°F and only after base course and binder course have been exposed to one

winter season. Base course may be placed when air temperature is above 30°F and rising.

- F. Grade Control
 - 1. Establish and maintain required lines and elevations.

PART 2 - PRODUCTS

- 2.1 MATERIALS:
 - A. General
 - 1. Use locally available materials and gradations as required by the Town of Mount Pleasant Department of Public Works which exhibit a satisfactory record of previous installations.
 - B. Base Course Aggregate
 - 1. Sound, angular, granular stone as specified on the plans.
 - C. Asphalt Concrete
 - 1. As specified on the plans.
 - D. Tack Coat
 - 1. Emulsified asphalt; AASHTO M 140 (ASTM D 997) or M 208 (D 2397); SS-1h or CSS-1h, diluted with one part water to one-part emulsified asphalt.

PART 3 - EXECUTION

3.1 SURFACE PREPARATION:

- A. Remove loose material from compacted subbase surface.
 - 1. Proof roll prepared subbase with a ten-ton roller to check for unstable areas and areas requiring additional compaction.
 - 2. Notify Owner's Representative of unsatisfactory conditions. Do not begin paving work until deficient subbase areas have been corrected and are ready to receive paving.
- B. Tack Coat
 - 1. Apply to contact surfaces of previously constructed asphalt concrete binder course and surfaces abutting or projecting into existing asphalt concrete pavement. Distribute at rate of 0.10 gallons per square yard of surface.
 - 2. Allow to dry until at proper condition to receive paving.
 - 3. Exercise care in applying bituminous materials to avoid smearing of adjoining concrete surfaces. Remove and clean damaged surfaces.

3.2 PLACING PAVEMENT MIX:

- A. General
 - 1. Place asphalt concrete mix on prepared surface, spread and strike-off. Spread mixture at minimum temperature of 225°F. Place inaccessible and small areas by hand. Place each course to required grade, cross-section and compacted thickness.
- B. Paver Placing

Place in strips not less than 10 feet wide, unless otherwise acceptable to Owner's Representative. After first strip has been placed and rolled, place succeeding strips and extend rolling to overlap previous strips. Complete base course and binder course for a section before placing surface course. Base course and binder course shall be exposed one winter season before applying surface course.

C. Joints

1. Make joints between old and new pavements, or between successive days' work, to ensure continuous bond between adjoining work. Construct joints to have same texture, density and smoothness as other sections of asphalt concrete course. Clean contact surfaces and apply tack coat.

3.3 ROLLING:

- A. General
 - 1. Begin rolling with a ten-ton roller when mixture will bear roller weight without excessive displacement.
 - 2. Compact mixture with hot hand tampers or vibrating place compactors in areas inaccessible to rollers.
- B. Breakdown Rolling
 - 1. Accomplish breakdown or initial rolling immediately following rolling of joints and outside edge. Check surface after breakdown rolling and repair displaced areas by loosening and filling, if required, with hot material.
- C. Second Rolling
 - 1. Follow breakdown rolling as soon as possible, while mixture is hot. Continue second rolling until mixture has been thoroughly compacted.
- D. Finish Rolling
 - 1. Perform finish rolling while mixture is still warm enough for removal of roller marks. Continue rolling until roller marks are eliminated and course has attained maximum density.
- E. Patching
 - 1. Remove and replace paving areas mixed with foreign materials and defective areas. Cut-out such areas and fill with fresh, hot asphalt concrete. Compact by rolling to maximum surface density and smoothness.

- F. Protection
 - 1. After final rolling, do not permit vehicular traffic on pavement until mixture has cooled enough not to become marked. Erect barricades to protect paving from traffic.
- 3.4 FIELD QUALITY CONTROL:
 - A. General
 - 1. Test in-place asphalt concrete courses for compliance with requirements for thickness and surface smoothness. Repair or remove and replace unacceptable paving as directed by the Owner's Representative.
 - B. Thickness
 - 1. In-place compacted nominal pavement thickness will not be acceptable if exceeding following allowable variation from required thickness:

Base Course:	1/4"
Binder Course:	1/4"
Surface Course:	1/4"

- 2. The sum total thickness of all the courses shall not vary from the total nominal thickness indicated on the plans by more than 1/2 inch.
- C. Surface Smoothness
 - Test unfinished surface of each asphalt concrete course for smoothness, using 10 foot straightedge applied parallel with and at right angles to centerline of paved area. Surfaces will not be acceptable if exceeding the following tolerances for smoothness:

Base Course Surface:	1/4"
Wearing Course Surface:	1/8"

- D. Crowned Surfaces
 - 1. Test with crowned template centered and at right angle to crown. Maximum allowable variance from template, 1/4 inch.
 - 2. Variations exceeding above tolerances shall be satisfactorily corrected at no additional cost to the Owner.
 - 3. Check surface areas at intervals as directed by the Owner's Representative.

3.5 TEMPORARY PAVEMENT:

- A. Where trench excavations have been performed within an existing street, furnish and lay a temporary pavement on an approved subgrade to the lines and grades specified herein. Cut the existing pavement with suitable tools as specified in Division 31 Section 312000 Earthwork. The construction of the temporary pavement shall consist of 2-inch compacted thickness of hot-mix asphaltic concrete unless otherwise specified by the Owner's Representative.
- B. Before the material is spread, paint all curb edgings, surfaces of manholes and other

structures which will come into contact with the new pavement with a bituminous emulsion or priming material acceptable to the Owner's Representative. Take care to prevent staining, smearing or defacing the exposed faces of the curbs and other structures during the spreading and rolling of the material.

- C. After spreading, roll the material by means of a well-balanced roller weighing not less than 15 tons. In all places not accessible to the roller, compact the material thoroughly by tampers weighing not less than 25 pounds and having a bearing area not exceeding 50 square inches.
- D. Maintain temporary pavement until such time that the final settlement of the trench shall have taken place in the opinion of the Owner's Representative. Correct any settlement taking place by furnishing, spreading and rolling additional material over that previously laid.
- 3.6 PAVEMENT REPLACEMENT:
 - A. In areas where temporary pavement has been placed, remove said temporary pavement to the subgrade line as specified. Bring the subgrade to the proper elevation and compact it.
 - B. Where specified by the Owner's Representative, excavate a shelf to provide a bearing area on all sides for the new pavement. The width of said shelf and pavement thickness shall be as shown on the plans or as directed by the Owner's Representative.

END OF SECTION 32 1216

SECTION 32 1313 – CEMENT CONCRETE PAVING

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including Rockefeller Archive Center General Conditions and any Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.
- 1.2 DESCRIPTION OF WORK:
 - A. General
 - 1. Extent of cement concrete paving is shown on drawings for all walks.
 - 2. Prepared subbase is specified in Division 31 Section 31 2000 Earthwork.

1.3 QUALITY ASSURANCE:

- A. Codes and Standards
 - 1. Comply with local governing regulations if more stringent than herein specified.
- B. Submittals
 - 1. Furnish samples, manufacturer's product data, test reports and materials' certifications as required in referenced sections for concrete and joint fillers and sealers.

1.4 JOB CONDITIONS:

A. No concrete shall be placed in the work when the temperature is below 32°F or when the temperature is likely, in the opinion of the Owner's Representative, to drop to 32°F within the next 24 hours, except with the prior approval of the Owner's Representative and under special protective measures approved by the Owner's Representative.

PART 2 - PRODUCTS

2.1 MATERIALS:

- A. Forms
 - 1. Steel, wood or other suitable material of size and strength to resist movement during concrete placement and to retain horizontal and vertical alignment until removal. Use straight forms free of distortion and defects.
 - 2. Use flexible spring steel forms or laminated boards to form radius bends as required.
 - 3. Coat forms with a non-staining form release agent that will not discolor or deface surface of concrete.
- B. Welded Wire Mesh
 - 1. Welded plain cold-drawn steel wire fabric, ASTM A 185.
- C. Reinforcing Bars
 - 1. Deformed steel bars, ASTM A 615, Grade 40.
- D. Concrete Materials

1. Comply with requirements applicable for concrete materials, admixtures, bonding materials, curing materials and others as required.

- E. Expansion Joint Materials
 - 1. Comply with requirements for preformed expansion joint fillers and sealers.

PART 3 - EXECUTION

3.1 CONCRETE MIX, DESIGN AND TESTING:

- A. General
 - 1. Comply with requirements applicable for concrete mix design, sampling and testing, and quality control, and as herein specified.
 - 2. Design mix to normal-weight concrete consisting of Portland cement, aggregate, water- reducing or high-range water-reducing admixture (super-plasticizer), air-entraining admixture and water to produce the following properties:
 - a. Compressive Strength: 4,000 psi, minimum at 28 days, unless otherwise indicated.
 - b. Slump Range: 4 inches for concrete containing HPWR admixture (superplasticizer); 4 inches for other concrete.
 - c. Air Content: 4.5 to 7.5 percent.

3.2 SURFACE PREPARATION:

- A. General
 - 1. Remove loose material from compacted subbase surface immediately before placing concrete.
 - 2. Proof-roll prepared subbase surface to check for unstable areas and need for additional compaction. Do not begin paving work until such conditions have been corrected and are ready to receive paving.

3.3 FORM CONSTRUCTION:

- A. General
 - 1. Set forms to required grades and lines, rigidly braced and secured. Install sufficient quantity of forms to allow continuous progress of work and so that forms can remain in place at least 24 hours after concrete placement.
 - 2. Check completed formwork for grade and alignment to following tolerances:

- a. Top of forms not more than 1/8 inch in 10 feet.
- b. Vertical face on longitudinal axis, not more than 1/4 inch in 10 feet.
- 3. Clean forms after each use, and coat with form release agent as often as required to ensure separation from concrete without damage.

3.4 REINFORCEMENT:

- A. General
 - 1. Locate, place and support reinforcement as shown on details, unless otherwise indicated.

3.5 CONCRETE PLACEMENT:

- A. General
 - 1. Comply with requirements for mixing and placing concrete and as herein specified.
 - 2. Do not place concrete until subbase and forms have been checked for line and grade. Moisten subbase if required to provide a uniform dampened condition at time concrete is placed. Do not place concrete around manholes or other structures until they are at required finish elevation and alignment.
 - 3. Place concrete using methods which prevent segregation of mix. Consolidate concrete along face of forms and adjacent to transverse joints with internal vibrator.
 - 4. Use bonding agent at locations where fresh concrete is placed against hardened or partially hardened concrete surfaces.
 - 5. Deposit and spread concrete in a continuous operation between transverse joints, as far as possible. If interrupted for more than 1/2 hour, place a construction joint.

3.6 JOINTS:

- A. General
 - 1. Construct expansion, weakened-plane (contraction) and construction joints true-toline with face perpendicular to surface of concrete. Construct transverse joints at right angles to the centerline, unless otherwise indicated.
 - 2. When joining existing structures, place transverse joints to align with previously placed joints, unless otherwise indicated.
- B. Construction Joints
 - 1. Place construction joints at end of placements and at locations where placement operations are stopped for a period of more than 1/2 hour, except where such placements terminate at expansion joints.
- C. Expansion Joints
 - 1. Provide premolded joint filler for expansion joints abutting concrete curbs, catch basins, manholes, inlets, structures, walks and other fixed objects, unless otherwise indicated.
 - 2. Locate expansion joints at 20 feet o.c. maximum, unless otherwise indicated.

- 3. Extend joint fillers full-width and depth of joint, recessed 1/2 inch below finished concrete surface.
- 4. Furnish joint fillers in one-place lengths for full width being place.
- 5. Protect top edge of joint filler during concrete placement with a metal cap or other temporary material. Remove protection after concrete has been placed on both sides of joint.
- D. Fillers and Sealants
 - 1. All joints shall receive joint sealants. Comply with requirements for preparation of joints, materials, installation and performance.
- 3.7 CONCRETE FINISHING:
 - A. General
 - 1. Protect and cure finished concrete paving as specified on the plans and details. Use membrane-forming curing and sealing compound or approved moist-curing methods.
 - B. Anti-Spalling Treatment
 - 1. A second coat of curing and sealing compound may be used or an anti-spalling compound applied over concrete cured by continuous moist curing methods. Apply compounds to concrete surfaces no sooner than 28 days after placement, to clean, dry concrete free of oil, dirt and other foreign material. Apply curing and sealing compound at a maximum coverage rate of 300 square feet per gallon. Apply anti-spalling compound in two sprayed applications. First application at rate of 40 square yards per gallon; second application, 60 square yards per gallon. Allow complete drying between applications.

3.8 REPAIRS AND PROTECTIONS:

- A. General
 - 1. Repair or replace broken or defective concrete, as directed by Owner's Representative.
 - 2. Drill test cores where directed by Owner's Representative when necessary to determine magnitude of cracks or defective areas. Fill drilled core holes in satisfactory pavement areas with Portland cement concrete bonded to pavement with epoxy adhesive.
 - 3. Protect concrete from damage until acceptance of work. Exclude traffic from pavement for at least 14 days after placement. When construction traffic is permitted, maintain pavement as clean as possible by removing surface stains and spillage of materials as they occur.
 - 4. Sweep concrete pavement and wash free of stains, discolorations, dirt and foreign material just prior to final inspection.

END OF SECTION 32 1313

SECTION 33 1373 – PAVEMENT JOINT SEALANTS

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including Rockefeller Archive Center General Conditions and any Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Cold-applied joint sealants.
 - 2. Hot-applied joint sealants.

1.3 SUBMITTALS

- A. Product Data: For each joint-sealant product indicated.
- B. Samples for Verification: For each kind and color of joint sealant required, provide Samples with joint sealants in 1/2-inch-wide joints formed between two 6-inch-long strips of material matching the appearance of exposed surfaces adjacent to joint sealants.
- C. Pavement-Joint-Sealant Schedule: Include the following information:
 - 1. Joint-sealant application, joint location, and designation.
 - 2. Joint-sealant manufacturer and product name.
 - 3. Joint-sealant formulation.
 - 4. Joint-sealant color.
- D. Product Certificates: For each type of joint sealant and accessory, from manufacturer.

1.4 QUALITY ASSURANCE

- A. Source Limitations: Obtain each type of joint sealant from single source from single manufacturer.
- 1.5 PROJECT CONDITIONS
 - A. Do not proceed with installation of joint sealants under the following conditions:
 - 1. When ambient and substrate temperature conditions are outside limits permitted by joint-sealant manufacturer or are below 40°F.
 - 2. When joint substrates are wet.
 - 3. Where joint widths are less than those allowed by joint-sealant manufacturer for applications indicated.

4. Where contaminants capable of interfering with adhesion have not yet been removed from joint substrates

PART 2 - PRODUCTS

2.1 MATERIALS

A. Compatibility: Provide joint sealants, backing materials, and other related materials that are compatible with one another and with joint substrates under conditions of service and application, as demonstrated by joint-sealant manufacturer based on testing and field experience.

2.2 COLD-APPLIED JOINT SEALANTS

- A. Single-Component, Nonsag, Silicone Joint Sealant for Concrete: ASTM D 5893, Type NS.
- B. Single-Component, Self-Leveling, Silicone Joint Sealant for Concrete: ASTM D 5893, Type SL.
- C. Multicomponent, Pourable, Traffic-Grade, Urethane Joint Sealant for Concrete: ASTM C 920, Type M, Grade P, Class 25, for Use T.

2.3 PRIMERS

A. Primers: Product recommended by joint-sealant manufacturer where required for adhesion of sealant to joint substrates indicated, as determined from preconstruction joint-sealant- substrate tests and field tests.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine joints indicated to receive joint sealants, with Installer present, for compliance with requirements for joint configuration, installation tolerances, and other conditions affecting joint-sealant performance. All expansion and isolation joints including work adjacent to structures and other materials shall receive a joint sealant.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

3.2 PREPARATION

A. Surface Cleaning of Joints: Clean out joints immediately before installing joint sealants to comply with joint-sealant manufacturer's written instructions.

B. Joint Priming: Prime joint substrates where indicated or where recommended in writing by joint-sealant manufacturer, based on preconstruction joint-sealant substrate tests or prior experience. Apply primer to comply with joint-sealant manufacturer's written instructions. Confine primers to areas of joint-sealant bond; do not allow spillage or migration onto adjoining surfaces.

3.3 INSTALLATION OF JOINT SEALANTS

- A. General: Comply with joint-sealant manufacturer's written installation instructions for products and applications indicated unless more stringent requirements apply.
- B. Joint-Sealant Installation Standard: Comply with recommendations in ASTM C 1193 for use of joint sealants as applicable to materials, applications, and conditions indicated.
- C. Install joint sealants using proven techniques that comply with the following and at the same time backings are installed:
 - 1. Place joint sealants so they directly contact and fully wet joint substrates.
 - 2. Completely fill recesses in each joint configuration.
 - 3. Produce uniform, cross-sectional shapes and depths relative to joint widths that allow optimum sealant movement capability.
- D. Tooling of Nonsag Joint Sealants: Immediately after joint-sealant application and before skinning or curing begins, tool sealants according to the following requirements to form smooth, uniform beads of configuration indicated; to eliminate air pockets; and to ensure contact and adhesion of sealant with sides of joint:
 - 1. Remove excess joint sealant from surfaces adjacent to joints.
 - 2. Use tooling agents that are approved in writing by joint-sealant manufacturer and that do not discolor sealants or adjacent surfaces.
- E. Provide joint configuration to comply with joint-sealant manufacturer's written instructions unless otherwise indicated.

3.4 CLEANING

A. Clean off excess joint sealant or sealant smears adjacent to joints as the Work progresses, by methods and with cleaning materials approved in writing by manufacturers of joint sealants and of products in which joints occur.

3.5 PROTECTION

A. Protect joint sealants, during and after curing period, from contact with contaminating substances and from damage resulting from construction operations or other causes so sealants are without deterioration or damage at time of Substantial Completion. If, despite such protection, damage or deterioration occurs, cut out and remove damaged or deteriorated joint sealants immediately and replace with joint sealant so installations in repaired areas are indistinguishable from the original work.

3.6 PAVEMENT-JOINT-SEALANT SCHEDULE

- A. Joint-Sealant Application: Joints within cement concrete pavement.
 - 1. Expansion and isolation joints in cast-in-place concrete pavement.
 - 2. Contraction joints in cast-in-place concrete slabs.
 - 3. Other joints as indicated.
- B. Joint-Sealant Application: Joints between cement concrete and asphalt pavement.
 - 1. Joint Location: Joints between concrete and asphalt pavement.
 - 2. Joints between concrete curbs and asphalt pavement.
 - 3. Other joints as indicated.

END OF SECTION 32 1373

SECTION 32 9100 - PLANTING SOILS

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 1 Specification Sections, apply to this section.
 - B. Section 31 2300 Earthwork
 - C. Section 32 9000 Planting
 - D. Section 32 9200 Turf and Grasses

1.2 RELATED SECTIONS

- A. Testing off-site borrow soil, existing topsoil and amendment materials for approved use in planting soil mixes. Verification testing of on-site sub-soils.
- B. Furnishing material from approved off-site source(s) as a base component for planting soil mixes and furnishing other soil amendment materials.
- C. Amending, preparing, and mixing planting soils for plant areas.
- D. Placing, spreading, and fine grading pre-mixed planting soil material of the type(s) indicated for plant areas.
- E. Protecting all plant mix installations with snow fencing, filter fabric, or other approved means, over the surface area plant bed installations, until substantial completion.
- F. Protection of finished paving, light poles utility or other finished work by means of wooden protection boards, or other approved means, over the area of construction concurrent with any and all construction operations.

1.3 SUBMITTALS

- A. Refer to and comply with General Requirements, for procedures and additional submittal criteria.
- B. Product Data: Submit technical descriptive data for each manufactured or packaged product of this Section. Include manufacturer's product testing and analysis and installation instructions for manufactured or processed items and materials.
- C. Locations: Submit locations of material sources. Submit location of mixing sites.
- D. Certificates:
 - 1. Submit certified analysis for each soil treatment, amendment, and fertilizer material specified and as used. Include guaranteed analysis and weight for packaged materials.
- E. Test Reports: Submit written reports of each sample tested. Each report shall include the following as a minimum and such other information required specific to material tested:

- 1. Date issued.
- 2. Project Title and names of Contractor and supplier.
- 3. Testing laboratory name, address and telephone number, and name(s), as applicable, of each field and laboratory inspector.
- 4. Date, place, and time of sampling or test, with record of temperature and weather conditions.
- 5. Location of material source.
- 6. Type of test.
- 7. Results of tests including identification of deviations from acceptable ranges. Identify any toxic substance(s) harmful to plant growth or life.
- F. Statement(s) of Qualifications: Submit within 45 days of notice to proceed to confirm qualifications as specified in Article 1.04, herein.
- G. Schedule and Protection Plan: Submit a detailed plan for scheduling and sequencing of all contract work and for protection of soil mixes and other completed work including coordination with contractors requiring access through the site. Indicate with schedules and plans the utilization of soil mix and subsoil protection measures (filter fabric and snow fencing) over the surface area of plant bed installations, until substantial completion. Indicate with schedules and plans the utilization of soil of finished work protection measures (wooden protection boards or other approved methods) over the work area of construction operations concurrent with all construction operations until substantial completion.

1.4 QUALITY ASSURANCE

- A. Qualifications:
 - 1. Installation and maintenance foreman on the job shall be competent Englishspeaking supervisor(s), experienced in landscape installation and maintenance. Perform work with personnel totally familiar with planting soil preparation and planting installations under the supervision of a foreman experienced with landscape work.
 - 2. Agricultural Chemist: Experienced person or persons employed by public or private soils testing laboratory, qualified and capable of performing tests, making soil recommendations, and issuing reports as specified. Testing Laboratory and Agricultural Chemist shall be as approved by the Owner's Identified Agent.
- B. References:
 - 1. Association of Official Agricultural Chemists.
 - 2. American Society for Testing and Materials (ASTM) using test criteria as specified or required by other references.
- C. Pre-installation Conferences: Person(s) responsible for soil preparation and mixes of this Section shall attend Pre-installation Conference(s) to coordinate with work of other sections.
- D. Inspections and Testing

- 1. Soil, compost, and other material testing required in this Section or additionally required by Owner's Representative shall be furnished and paid for by Contractor.
- 2. Owner's Identified Agent reserves the right to take and analyze at any time such additional samples of materials as deemed necessary for verification of conformance to specification requirements. Contractor shall furnish samples for this purpose upon request and shall perform testing as requested.
- 3. If at any time, test results indicate that any soil delivered to the site does not comply with these specifications, the contractor shall immediately remove the non-compliant soils from site at the contractor's expense.

1.5 REGULATORY REQUIREMENTS

- A. Comply with all rules, regulations, laws and ordinances of local, state, and federal authorities having jurisdiction. Provide labor, materials, equipment and services necessary to make work comply with such requirements without additional cost to Owner.
- B. Procure and pay for permits and licenses required for work of this section. Comply with the requirement of any Remedial Action Work Plan that may apply to the project.
- 1.6 PROJECT/SITE CONDITIONS
 - A. Acquaintance with Existing Site Conditions
 - 1. Through study of all Contract Documents and by careful examination of the site, become informed as to the nature and location of the Work, the nature of surface and subsurface soil conditions, the character, quality and quantity of the materials to be encountered, the character of equipment and facilities needed preliminary to and during the prosecution of the Work, the general and local conditions, and all other matters which can in any way affect the Work.
 - 2. Investigate the conditions of public thoroughfares and roads as to availability, clearances, loads, limits, restrictions, and other limitations affecting transportation to, ingress and egress of this work site. Conform to all governmental regulations in regard to the transportation of materials to, from, and at the job site, and secure in advance such permits as may be necessary.
 - 3. Should the Contractor, in the course of Work, find any discrepancies between Contract Drawings and physical conditions or any omissions or errors in Drawings, or in layout as furnished by the Owner, it will be Contractor's duty to inform the Owner's Identified Agent immediately in writing for clarification. Work done after such discovery, unless authorized by the Owner's Identified Agent, shall be done at the Contractor's risk.
 - 4. Contractor shall be familiar with any Remedial Action Work Plan for the project and be experienced in working on sites with historic fill.
 - B. Environmental Requirements:
 - 1. Perform both off-site mixing and on-site soil work only during suitable weather conditions. Do not disc, rototill, or work soil when frozen, excessively wet, or in otherwise unsatisfactory condition.

- 2. Soil mixes shall not be handled, hauled, or placed during rain or wet weather or when wet near or above field capacity.
- C. Sequencing and Scheduling: Adjust, relate together, and otherwise coordinate work of this Section with work of Project and all other Sections of Project Specifications.

1.7 PRODUCT DELIVERY, STORAGE AND HANDLING

- A. Packaged Materials: Deliver packaged materials to the location where soils are to be mixed, in unopened bags or containers, each bearing the name, guarantee, and trademark of the producer, material composition, manufacturers' certified analysis, and the weight of the materials Retain packages for the Construction Manager or Owner's Representative.
- B. Soil or amendment materials stored on site temporarily in stockpiles prior to placement shall be protected from intrusion of contaminants and erosion. All temporary storage means and methods shall be approved by Owner's Representative.
- C. After mixing, soil materials shall be covered with a tarpaulin until time of actual use.

PART 2 - PRODUCTS

2.1 PLANT MIX MATERIALS

- A. All soil material to be imported to the site shall be tested every 500 CY to ensure soils meet applicable remediation standards for clean soils. Proof of compliance shall be provided by contractor prior to delivery of soils to site.
- B. General:
 - 1. All plant mix materials shall fulfill the requirements for new plant mixes as specified.
 - 2. Samples of blended plant mixes shall be submitted by the Contractor for testing and analysis to the approved testing laboratory. Include verification testing of onsite sub-soils. Comply with specific material requirements specified.
 - a. As necessary to conform to the requirements of these specifications, make any and all soil mix amendments, retest and resubmit reports indicating results until approved.
 - 3. Owner's Representative may request additional testing by Contractor for confirmation of mix quality at any time until completion.
- C. Base Component Material
 - 1. Base Component Material shall be a mix of sandy loam and sand. Base Component Materials shall not be site salvaged unless approved by Owner's Identified Agent and CT DEEP.
 - 2. Before base component material is used for mixing with amendments, handle and pile Base Component Material in the following manner:
 - a. Homogenize to make a uniform mix, free of subsoil lenses and other irregularities.
 - b. Aerate the base material to make a friable planting medium.

- c. Separate out and remove all clay lumps, stones, stocks, roots, and other debris.
- D. Organic Matter
 - 1. Leaf Mold: Shredded leaf litter, composted for a minimum of one year (12 months) and confirming to the following characteristics:
 - a. The leaf mold must be free of debris such as plastic fragments, glass, and metal fragments.
 - b. The leaf mold must be free of stones larger than 1/2", large branches, and large roots.
 - c. Woodchips over 1" in length or diameter should be removed by screening.

2.2 SOIL AMENDMENT MATERIAL

- A. Ground Limestone: Ground Limestone as a soil amendment material will only be used pending results of analysis.
 - 1. Provide a Ground Agricultural Limestone with a minimum of 88% of calcium and magnesium carbonates.
 - 2. Ground Limestone material shall have a total 100% passing the 10 mesh sieve, minimum of 90% passing the 20 mesh sieve, and a minimum of 60% passing the 100 mesh sieve.

2.3 HERBICIDES

- A. Herbicides: May be required for use if there is seed germination after sub-grade placement and prior to planting mix installation or after subsequent plant mix installation. Under no circumstances are materials to be applied without specific instruction from the Owner's Identified Agent.
 - 1. Herbicides shall be approved before use for type and rate of application by the Owner's Identified Agent and by local and state agencies with jurisdiction.
 - 2. Post emergent herbicide shall be Roundup, as manufactured by Monsanto Agricultural Products Company, C3NJ, St. Louis, MO 63166, or an approved equal.

2.4 PLANTING SOIL MIXES

- A. Adequate quantities of mixed planting soil materials shall be provided, to supplement topsoil stockpiled from site, to attain, after compaction and natural settlement, all design finish grades. Verify quantities for placement as specified in Planting and Turf and Grasses specifications and to suit conditions.
- B. Uniformly mix ingredients as specified (Base Component Material, leaf mold, and other ingredients deemed to be necessary as a result of testing) by wind rowing/tilling on an approved hard surface area. Organic matter shall be maintained moist, not wet, during mixing.
- C. Testing of Plant Mixes:

- 1. Perform initial soil mix tests to confirm compliance with specifications. In conjunction with the requirements of the specifications, these test results, when approved, will establish the standard to which all other test results must conform.
- 2. Follow-up Testing: Have one (1) composite sample tested prior to delivery and upon arrival to the site from each 500 c.y. of material or as required by Owner's Representative intended for use in each type of lawn and plant mix to include the following:
 - a. Soil Texture / Physical Analysis: Use the hydrometer method and classify the soil.
 - b. Nutrient Analysis:
 - 1) Have nutrient levels (nitrate, nitrogen, phosphate, potassium, magnesium, calcium, ammonium, iron, and manganese) tested, and request testing laboratory recommendations for additional fertilizer requirements at both lawn and all plant areas if nutrient levels are below average.
 - 2) Nutrient deficiencies in soils of plant areas shall be corrected at time of installation.
 - 3) Nutrient deficiencies in soils of lawn areas shall be corrected both at time of lawn installations and during maintenance period as specified.
 - c. Test organic matter, pH, soluble salts, and percolation.
- D. Soil Mix Types: Provide the following planting soil mix at the locations indicated. Percentages of components, unless otherwise noted, will be established upon completion of test results of the various mixes. The controlling factor will be the percent (%) organic matter as specified for each mix. Note that percent (%) by volume of components will be, in large part, determined by the leaf mold. Specifically, the bulk density reading of the leaf mold will directly impact the organic matter readings which have been specified for each mix.
 - 1. Topsoil / Horticultural Soil Mix shall be used for planting areas on natural subsoils.
 - a. Test results must be submitted for per-cent (%) retained as well as for percent (%) passing for all sieve sizes. Failure to include any of the aforementioned criteria will be cause for rejection of the test report.
 - b. Soil Texture/ Physical Analysis
 - 1) Sand 40-65%
 - 2) Silt 25-60%
 - 3) Clay 5-20%
 - 4) Gravel content should not exceed 3% and soil should be free of glass, plastic and other foreign materials
 - c. Chemical Analysis:
 - 1) Organic matter: 4.5 to 7.5%
 - 2) Soil reaction (pH): 6.0 to 7.0
 - 3) Soluble salt content (conductivity): The soluble salts measurement

(Electric Conductivity) should not exceed 0.5 mmhos/cm measured as a 1:5 dilute.

d. Heavy metal content not to exceed the following amounts:

<u>Element</u>	Acetate Extract	HCL Extract
Iron	0.5 ppm	3.1 ppm
Manganese	0.5 ppm	15.4 ppm
Molybdenum	0.4 ppm	0.8 ppm
Zinc	0.2 ppm	4.4 ppm
Aluminum	0.2 ppm	1.2 ppm
Boron	1.1 ppm	1.7 ppm
Copper	None	0.01 ppm
Lead	01 ppm	0.4 ppm
Selenium	None	0.4 ppm
Mercury	None	None
Chromium	None	None
Cadmium	None	0.02 ppm
Nickel	None	0.04 ppm
Cobalt	None	0.05 ppm

None = none detected = below detection limits of 0.01 ppm.

- e. Percolation: 60% passing in 2 minutes, 40% retained. Soil should be saturated prior to conducting test.
- f. Other Amendments as required by test results and as approved.
- E. Stockpiling and Plant Mixes
 - General: Stockpiling and Plant Mixes utilized for planting soil on-site, off-site and at source should be restricted to no more than the needs of what can be used in a 24-hr. period. Stockpiles should be no more than 6 feet in height to prevent anaerobic conditions within the pile(s). Stockpiles shall be sheltered from weather to prevent excessive water absorption and blowing by winds as approved by Owner's Representative.

2.5 VERIFICATIONS

- A. Prior to construction and soil placement operations at planting areas, ascertain the location of all electric cables, conduits, underdrainage systems and utility lines.
- B. Take proper precautions so as not to disturb or damage sub-surface elements. Contractor failing to take these precautions shall be responsible for making requisite repairs to damaged utilities at Contractor's own expense.
- C. Verify that required underground utilities are available, in proper location, and ready for use. Coordinate with other trades.

D. Verify that all work requiring access through or adjacent to areas where plant mixes are to be placed has been completed and no further access will be required. In the event that access will be required, this must be coordinated with the Owner's Representative.

2.6 PREPARATION OF SUBGRADE

- A. Prior to depositing and spreading clean fill, sand and/or plant mix soils, the Contractor shall furnish and install grade stakes on a 10 foot grid in open areas and sufficiently spaced in other areas to insure correct line and grade of subgrade and finished grade.
 - 1. Verify as constructed or existing subgrade elevations and do whatever additional grading is necessary to bring the subgrade to a true, smooth, slope parallel to the finish grade at all areas to receive planting soil for lawns.
 - 2. Clean up subgrade and dispose of all debris and garbage prior to inspection.
 - 3. Scarify compacted subgrades to permit drainage through soil layers.
- B. Remove all vegetation on subgrade or wherever additional soil is to be added to meet grade.
- C. Any soils polluted by gasoline, oil, plaster, construction debris, unacceptable soils, or other substances which would render subgrade unsuitable for a proper lawn or plant growth, shall be removed from the premises whether or not such pollution occurs or exists prior to or during the Contract period. In the event that such material is placed, this material shall be removed and replaced with approved material. All remedial operations associated with soil mixes and controlled fill shall be reviewed and approved by the Owner's Representative responsible for remediation, when applicable.

2.7 PLACING PLANTING SOIL

- A. Remove all large clods, lumps, brush, roots, stumps, litter, and other foreign material and stones one-half inch (1/2") in diameter or larger. Dispose of removed material legally off-site.
- B. Do not place a muddy or wet soil mix.
- C. Place and spread planting soil mix of the type specified over approved subgrade to a depth sufficiently greater than the depth required for planting areas so that after natural settlement, misting and/or light rolling, as previously approved by Owner's Identified Agent, the completed work will conform to the lines, grades, and elevations shown or otherwise indicated.
- D. Topsoil/ Planting Soil Mix shall be placed at the following minimum depths, or as otherwise directed by the plans in all areas over natural subsoils.
 - 1. Lawn Areas 6" minimum
- E. Grading Tolerances: Lawn and Planting areas shall be fine graded within ±1/10 (0.10) feet of grades indicated on drawings. Maintain all "flat" areas and slopes to allow free flow of surface drainage without ponding.

END OF SECTION 32 9100

SECTION 32 9200 – TURF AND GRASSES

PART 1 - GENERAL

1.1 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 1 Specification Sections, apply to this section.
- B. Section 31 2000 Earthwork
- C. Section 32 9100 Planting Soils
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Sodding.

1.3 DEFINITIONS

- A. Duff Layer: The surface layer of native topsoil that is composed of mostly decayed leaves, twigs, and detritus.
- B. Finish Grade: Elevation of finished surface of planting soil.
- C. Manufactured Topsoil: Soil produced off-site by homogeneously blending mineral soils or sand with stabilized organic soil amendments to produce topsoil or planting soil.
- D. Pesticide: A substance or mixture intended for preventing, destroying, repelling, or mitigating a pest. This includes insecticides, miticides, herbicides, fungicides, rodenticides, and molluscicides. It also includes substances or mixtures intended for use as a plant regulator, defoliant, or desiccant.
- E. Pests: Living organisms that occur where they are not desired or that cause damage to plants, animals, or people. These include insects, mites, grubs, mollusks (snails and slugs), rodents (gophers, moles, and mice), unwanted plants (weeds), fungi, bacteria, and viruses.
- F. Subgrade: Surface or elevation of subsoil remaining after excavation is complete, or top surface of a fill or backfill before planting soil is placed.
- G. Subsoil: All soil beneath the topsoil layer of the soil profile, and typified by the lack of organic matter and soil organisms.
- H. Surface Soil: Soil that is present at the top layer of the existing soil profile at the Project site. In undisturbed areas, the surface soil is typically topsoil, but in disturbed areas such as urban environments, the surface soil can be subsoil.

1.4 ACTION SUBMITTALS

A. Product Data: For each type of product indicated.

1. Pesticides and Herbicides: Include product label and manufacturer's application instructions specific to this Project.

1.5 SUBMITTALS

- A. Certification of Lawn
 - 1. Certification of each seed mixture for turfgrass sod. Include identification of source and name and telephone number of supplier.
- B. Qualification Data: For qualified landscape Installer.
- C. Product Certificates: For soil amendments and fertilizers, from manufacturer.
- D. Material Test Reports: For standardized ASTM D 5268 topsoil existing native surface topsoil existing in-place surface soil and imported or manufactured topsoil.
- E. Maintenance Instruction Manual: Contractor to submit a manual that includes recommended procedures and schedules to be established by Owner for maintenance of turf during a calendar year. Submit before expiration of required initial maintenance periods.

1.6 QUALITY ASSURANCE

- A. Installer Qualifications: A qualified landscape Installer whose work has resulted in successful turf establishment.
 - 1. Professional Membership: Installer shall be a member in good standing of either the Professional Landcare Network or the American Nursery and Landscape Association.
 - 2. Experience: Five years' experience in turf installation in addition to requirements in Division 01 Section "Quality Requirements."
 - 3. Installer's Field Supervision: Require Installer to maintain an experienced full-time supervisor on Project site when work is in progress.
 - 4. Maintenance Proximity: Not more than two hours' normal travel time from Installer's place of business to Project site.
 - 5. Pesticide Applicator: State licensed, commercial.

1.7 DELIVERY, STORAGE, AND HANDLING

- A. Seed and Other Packaged Materials: Deliver packaged materials in original, unopened containers showing weight, certified analysis, name and address of manufacturer, and indication of conformance with state and federal laws, as applicable.
- B. Sod: Harvest, deliver, store, and handle sod according to requirements in "Specifications for Turfgrass Sod Materials" and "Specifications for Turfgrass Sod Transplanting and Installation" in TPI's "Guideline Specifications to Turfgrass Sodding." Deliver sod in time for planting within 24 hours of harvesting. Protect sod from breakage and drying.
- C. Bulk Materials:

- 1. Do not dump or store bulk materials near structures, utilities, walkways and pavements, or on existing turf areas or plants.
- 2. Provide erosion-control measures to prevent erosion or displacement of bulk materials, discharge of soil-bearing water runoff, and airborne dust reaching adjacent properties, water conveyance systems, or walkways.
- 3. Accompany each delivery of bulk fertilizers, lime, and soil amendments with appropriate certificates.

1.8 PROJECT CONDITIONS

- A. Planting Restrictions: Plant during one of the following periods. Coordinate planting periods with initial maintenance periods to provide required maintenance from date of Substantial Completion
- B. Sod Installation Season
 - 1. Spring: April 15 to July 15
 - 2. Fall: August 15 to November 1

Weather Limitations: Proceed with planting only when existing and forecasted weather conditions permit planting to be performed when beneficial and optimum results may be obtained. Apply products during favorable weather conditions according to manufacturer's written instructions.

1.9 MAINTENANCE SERVICE

- A. Initial Turf Maintenance Service: Provide full maintenance by skilled employees of landscape Installer. Maintain as required in Part 3. Begin maintenance immediately after each area is planted and continue until acceptable turf is established but for not less than the following periods:
 - 1. Seeded Turf: 90 days from date of Substantial Completion
 - a) When initial maintenance period has not elapsed before end of planting season, or if turf is not fully established, continue maintenance during next planting season.
 - 2. Sodded Turf: 60 days from date of Substantial Completion.

PART 2 - PRODUCTS

2.1 TURFGRASS SOD

- A. Turfgrass Sod: Certified complying with "Specifications for Turfgrass Sod Materials" in TPI's "Guideline Specifications to Turfgrass Sodding." Furnish viable sod of uniform density, color, and texture, strongly rooted, and capable of vigorous growth and development when planted.
- B. Turfgrass Species: Closely match seed species specified in plans

2.2 FERTILIZERS

- A. Bonemeal: Commercial, raw or steamed, finely ground; a minimum of 1 percent nitrogen and 10 percent phosphoric acid.
- B. Superphosphate: Commercial, phosphate mixture, soluble; a minimum of 20 percent available phosphoric acid.
- C. Commercial Fertilizer: Commercial-grade complete fertilizer of neutral character, consisting of fast- and slow-release nitrogen, 50 percent derived from natural organic sources of urea formaldehyde, phosphorous, and potassium in the following composition:
 - 1. Composition: 1 lb./1000 sq. ft. of actual nitrogen, 4 percent phosphorous, and 2 percent potassium, by weight.
 - 2. Composition: Nitrogen, phosphorous, and potassium in amounts recommended in soil reports from a qualified soil-testing laboratory.
- D. Slow-Release Fertilizer: Granular or pelleted fertilizer consisting of 50 percent waterinsoluble nitrogen, phosphorus, and potassium in the following composition:
 - 1. Composition: 20 percent nitrogen, 10 percent phosphorous, and 10 percent potassium, by weight.
 - 2. Composition: Nitrogen, phosphorous, and potassium in amounts recommended in soil reports from a qualified soil-testing laboratory.

2.3 MULCHES

- A. Straw Mulch: Provide air-dry, clean, mildew- and seed-free, salt hay or threshed straw of wheat, rye, oats, or barley.
- B. Fiber Mulch: Biodegradable, dyed-wood, cellulose-fiber mulch; nontoxic and free of plant-growth or germination inhibitors; with a maximum moisture content of 15 percent and a pH range of 4.5 to 6.5.

2.4 PESTICIDES

- A. General: Pesticide, registered and approved by EPA, acceptable to authorities having jurisdiction, and of type recommended by manufacturer for each specific problem and as required for Project conditions and application. Do not use restricted pesticides unless authorized in writing by authorities having jurisdiction.
- B. Pre-Emergent Herbicide (Selective and Non-Selective): Effective for controlling the germination or growth of weeds within planted areas at the soil level directly below the mulch layer.
- C. Post-Emergent Herbicide (Selective and Non-Selective): Effective for controlling weed growth that has already germinated.

2.5 EROSION-CONTROL MATERIALS

A. Erosion-Control Blankets: Biodegradable wood excelsior, straw, or coconut-fiber mat enclosed in a photodegradable plastic mesh. Include manufacturer's recommended steel wire staples, 6 inches long.

PART 3 - EXECUTION

3.1 EXAMINATION

- A. Examine areas to be planted for compliance with requirements and other conditions affecting performance.
 - 1. Verify that no foreign or deleterious material or liquid such as paint, paint washout, concrete slurry, concrete layers or chunks, cement, plaster, oils, gasoline, diesel fuel, paint thinner, turpentine, tar, roofing compound, or acid has been deposited in soil within a planting area.
 - 2. Do not mix or place soils and soil amendments in frozen, wet, or muddy conditions.
 - 3. Suspend soil spreading, grading, and tilling operations during periods of excessive soil moisture until the moisture content reaches acceptable levels to attain the required results.
 - 4. Uniformly moisten excessively dry soil that is not workable and which is too dusty.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.
- C. If contamination by foreign or deleterious material or liquid is present in soil within a planting area, remove the soil and contamination as directed by Architect and replace with new planting soil.

3.2 PREPARATION

- A. Protect structures, utilities, sidewalks, pavements, and other facilities, trees, shrubs, and plantings from damage caused by planting operations.
 - 1. Protect adjacent and adjoining areas from hydroseeding and hydromulching overspray.
 - 2. Protect grade stakes set by others until directed to remove them.
- B. Install erosion-control measures to prevent erosion or displacement of soils and discharge of soil-bearing water runoff or airborne dust to adjacent properties and walkways.

3.3 TURF AREA PREPARATION

- A. Limit turf subgrade preparation to areas to be planted.
- B. Hand-dig or air spade in all areas under the drip line of existing trees.
- C. Newly Graded Subgrades: Loosen subgrade to a minimum depth of 8 inches. Remove stones larger than 1 inch in any dimension and sticks, roots, rubbish, and other extraneous matter and legally dispose of them off Owner's property.
 - 1. Spread topsoil, apply soil amendments and fertilizer on surface, and thoroughly blend planting soil.
 - a) Delay mixing fertilizer with planting soil if planting will not proceed within a few days.
 - b) Mix lime with dry soil before mixing fertilizer.

- D. Unchanged Subgrades: If turf is to be planted in areas unaltered or undisturbed by excavating, grading, or surface-soil stripping operations, prepare surface soil as follows:
 - 1. Remove existing grass, vegetation, and turf. Do not mix into surface soil.
 - 2. Loosen surface soil to a depth of at least 8 inches. Apply soil amendments and fertilizers according to planting soil mix proportions and mix thoroughly into top 8 inches. Till soil to a homogeneous mixture of fine texture.
 - 3. Remove stones larger than 1 inch in any dimension and sticks, roots, trash, and other extraneous matter.
 - 4. Legally dispose of waste material, including grass, vegetation, and turf, off Owner's property.
- E. Finish Grading: Grade planting areas to a smooth, uniform surface plane with loose, uniformly fine texture. Grade to within plus or minus ½ inch of finish elevation. Roll and rake, remove ridges, and fill depressions to meet finish grades. Limit finish grading to areas that can be planted in the immediate future.
- F. Moisten prepared area before planting if soil is dry. Water thoroughly and allow surface to dry before planting. Do not create muddy soil.
- G. Before planting, obtain Architect's acceptance of finish grading; restore planting areas if eroded or otherwise disturbed after finish grading.

3.4 SODDING

- A. Lay sod within 24 hours of harvesting. Do not lay sod if dormant or if ground is frozen or muddy.
- B. Lay sod to form a solid mass with tightly fitted joints. Butt ends and sides of sod; do not stretch or overlap. Stagger sod strips or pads to offset joints in adjacent courses. Avoid damage to subgrade or sod during installation. Tamp and roll lightly to ensure contact with subgrade, eliminate air pockets, and form a smooth surface. Work sifted soil or fine sand into minor cracks between pieces of sod; remove excess to avoid smothering sod and adjacent grass.
 - 1. Lay sod across angle of slopes exceeding 1:3.
 - 2. Anchor sod on slopes exceeding 1:6 with wood pegs spaced as recommended by sod manufacturer but not less than 2 anchors per sod strip to prevent slippage.
- C. Saturate sod with fine water spray within two hours of planting. During first week after planting, water daily or more frequently as necessary to maintain moist soil to a minimum depth of 1-1/2 inches below sod.

3.5 TURF MAINTENANCE

A. Maintain and establish turf by watering, fertilizing, weeding, mowing, trimming, replanting, and performing other operations as required to establish healthy, viable turf. Roll, regrade, and replant bare or eroded areas and remulch to produce a uniformly smooth turf. Provide materials and installation the same as those used in the original installation.

- 1. Fill in as necessary soil subsidence that may occur because of settling or other processes. Replace materials and turf damaged or lost in areas of subsidence.
- 2. In areas where mulch has been disturbed by wind or maintenance operations, add new mulch and anchor as required to prevent displacement.
- 3. Apply treatments as required to keep turf and soil free of pests and pathogens or disease. Use integrated pest management practices whenever possible to minimize the use of pesticides and reduce hazards.
- B. Watering: Install and maintain temporary piping, hoses, and turf-watering equipment to convey water from sources and to keep turf uniformly moist to a depth of 4 inches.
 - 1. Schedule watering to prevent wilting, puddling, erosion, and displacement of seed or mulch. Lay out temporary watering system to avoid walking over muddy or newly planted areas.
 - 2. Water turf with fine spray at a minimum rate of 1 inch per week unless rainfall precipitation is adequate.
- C. Mow areas designated as lawn on the plan as soon as top growth is tall enough to cut. Repeat mowing to maintain specified height without cutting more than 1/3 of grass height. Remove no more than 1/3 of grass-leaf growth in initial or subsequent mowing. Do not delay mowing until grass blades bend over and become matted. Do not mow when grass is wet. Retain applicable subparagraphs below for mowing height. For seed mixtures, base selection on predominant species to be established.
- D. Areas hatched as 'Fescue Mix' on plan shall not be mown more that twice a year, Do not mowing shorter than 4".
- E. Turf Post fertilization: Apply fertilizer after initial mowing and when grass is dry.
 - 1. Use fertilizer that will provide actual nitrogen of at least 1 lb./1000 sq. ft. to turf area.

3.6 SATISFACTORY TURF

- A. Turf installations shall meet the following criteria as determined by Architect:
 - 1. Satisfactory Seeded Turf: At end of maintenance period, a healthy, uniform, close stand of grass has been established, free of weeds and surface irregularities, with coverage exceeding 90 percent over any 10 sq. ft. and bare spots not exceeding 5 by 5 inches.
 - 2. Satisfactory Sodded Turf: At end of maintenance period, a healthy, well-rooted, even-colored, viable turf has been established, free of weeds, open joints, bare areas, and surface irregularities.
- B. Use specified materials to reestablish turf that does not comply with requirements and continue maintenance until turf is satisfactory.

3.7 PESTICIDE APPLICATION

A. Apply pesticides and other chemical products and biological control agents in accordance with requirements of authorities having jurisdiction and manufacturer's written recommendations. Coordinate applications with Owner's operations and others in proximity to the Work. Notify Owner before each application is performed.

- B. Post-Emergent Herbicides (Selective and Non-Selective): Apply only as necessary to treat already germinated weeds and in accordance with manufacturer's written recommendations.
- 3.8 CLEANUP AND PROTECTION
 - A. Promptly remove soil and debris created by turf work from paved areas. Clean wheels of vehicles before leaving site to avoid tracking soil onto roads, walks, or other paved areas.
 - B. Erect temporary fencing or barricades and warning signs as required to protect newly planted areas from traffic. Maintain fencing and barricades throughout initial maintenance period and remove after plantings are established.
 - C. Remove nondegradable erosion-control measures after grass establishment period.

END OF SECTION 32 9200

DRAWINGS and PROJECT DETAILS

ARPA Renovations

GIRLS INC. of Meriden

130 Lincoln St, Meriden, CT 06451 сwa project NO: 2313

ISSUED FOR BIDDING MARCH 12, 2024

ARCHITECT



CHRISTOPHER WILLIAMS ARCHITECTS85 Willow StreetNew Haven, CT 06511203 776 0184cwarchitectsllc.com

MEP ENGINEER



33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370 iesllc.biz

ENVIRONMENTAL CONSULTANT



LIST OF DRAWINGS

		124
		5-20
		7
		03
		0 0
NO.	SHEET NAME	B
GENERAL		
G001	CODE INFORMATION, GENERAL NOTES, SYMBOLS & ABBREVIATIONS	•
CIVIL		
CD101	SITE DEMOLITION PLAN	•
C101	SITE PLAN	•
C501	SITE DETAILS	•
C502	ENLARGED EXTERIOR CONCRETE RAMP PLAN & DETAILS	•
ARCHITECTUR		
AD100	BASEMENT DEMOLITION PLAN	•
AD101	FIRST FLOOR DEMOLITION PLAN	•
AD102	SECOND FLOOR DEMOLITION PLAN	•
AD103	THIRD FLOOR DEMOLITION PLAN	•
AD201	BUILDING ELEVATIONS - DEMOLITION	•
A100		•
A101		•
A102		•
A103		•
A113	THIRD FLOOR REFLECTED CEILING PLANS - DEMOLITION & NEW	•
A120		•
A201		•
A202	SUUTH & WEST BUILDING ELEVATIONS	•
A203	ENLARGED REAR PORCH ELEVATIONS	•
A204	PARTIAL WEST BUILDING ELEVATIONS & DETAILS	•
A501		•
HAZARDOUS M		
H100		•
H101		•
H102	SEOND FLOOR ABATEMENT PLAN	•
H103		•
H120		•
H201		•
H202	ENLARGED REAR PORCH ELEVATIONS - ABATEMENT	•
DP.100	BASEMENT PLUMBING DEMOLITION PLAN	•
P.100		•
P.101	FIRST FLOOR PLUMBING PLAN	•
P.200	PLUMBING DETAILS, LEGENDS, NOTES & SCHEDULES	•
P.201	PLUMBING SPECIFICATIONS	•
SITE UTILITIES		
SU.100	SITE UTILITY PLAN	•
MECHANICAL		
		•
DM.101	FIRST FLOOR MECHANICAL DEMOLITION PLAN	•
DIVI.102	DECUND FLOUR MECHANICAL DEMOLTIUN PLAN	•
IVI. 100		•
IVI. 101		•
IVI. 102		•
M.103		•
M.200	MECHANICAL NOTES, LEGEND & DETAILS	•
M.201		•
M.202	MECHANICAL VDE DIDING & WIDING DIAGDAMS	•
M.203	MECHANICAL VRF PIPING & WIRING DIAGRAMIS	•
		•
DE.100		•
DE 102		•
DE 102		•
E 100		-
E 101		-
E.101		•
E.102		•
E 201		-
		-
E 202		-
E 204		-
F 205	ELECTRICAL SPECIFICATIONS CONTINUED	-
L.200		-

LOCATION MAP



Autodesk Docs://2313 Girls Inc/2313 Girls Inc - CWA - R24.rvt 3/12/2024 4:52:44 PM

		ABBRE	VIATIO	NS	GRAP
	AC ACT A.F.F. AHU ANOD	AIR CONDITIONING ACOUSTICAL TILE ABOVE FINISH FLOOR AIR HANDLING UNIT ANODIZED	MACH MAINT MAS MAT MAX	MACHINE MAINTENANCE MASONRY MATERIAL MAXIMUM	
	ALT ALUM ARCH ATTEN AWP	ALTERNATE ALUMINUM ARCHITECT ATTENUATION ACOUSTICAL WALL PANEL	MB MDF MECH MEZZ MFR	MARKER BOARD MEDIUM DENSITY FIBERBOARD MECHANICAL MEZZANINE MANUFACTURER	
D	BD BC BIT BLDG BLK	BOARD BRICK COURSE BITUMINOUS BUILDING BI OCK	MIN MISC MTD MTL N	MINIMUM MISCELLANEOUS MOUNTED METAL NORTH	1 1/4"
	BLKG BM B.O. BRG BTM	BLOCKING BEAM BOTTOM OF BEARING BOTTOM	No. NOM NPS NIC, N.I.C. NRC	NUMBER NOMINAL NOMINAL PIPE SIZE NOT IN CONTRACT NOISE REDUCTION COEFFICEINT	
	BUR CAB CAP CB CEM	BUILT-UP ROOFING CABINET CAPACITY CHALKBOARD CUBIC FEET PER MINUTE	NTS, N.T.S OC, O.C. OD, O.D. OPNG OPP	NOT TO SCALE ON CENTER OUTSIDE DIAMETER OPENING OPENSITE	
	CFMF CJ CLG CLOS COL	COLD FORMED METAL FRAMING CONTROL JOINT CEILING CLOSET COLUMN	ORD OH OZ PERIM	OVERFLOW ROOF DRAIN OVERHEAD OUNCE PERIMETER	C4 DE
	CONC CONT CORR CONSTR CPT	CONCRETE CONTINUOUS CORRIDOR CONSTRUCTION CARPET TILE	PERP PL PLAS PLAM PLUMB	PERPENDICULAR PLATE PLASTIC PLASTIC LAMINATE PLUMBING	
	CRSE CT CTR CUH	COURSE CERAMIC TILE CENTER CABINET UNIT HEATER	PLYWD PNL PR PREFAB PSF PT	PLYWOOD PANEL PAIR PREFABRICATED PER SQAURE FOOT PER SQAURE FOOT	SIM
	D DBLE DEG DEMO DF DH	DEEP DOUBLE DEGREE DEMOLITION DRINKING FOUNTAIN DOUBLE HUNG	PTD PTD PVC QT QTY	PRESSURE TREATED, PAINT PAINTED, PAPER TOWEL DISPENSER POLYVINYL CHLORIDE QUARRY TILE QUANTITY	AXXX Detail Numb
С	DIA DIM DIST DN DWG DR DS	DIAMETER DIMENSION DISTANCE DOWN DRAWING DOOR	R RAD RCP RD REF PEINE	RADIUS, RISER RADIUS REFLECTED CEILING PLAN ROOF DRAIN REFRIGERATOR REINEORCE	Drawing Nu
	E EA EF	EAST EACH EXHAUST FAN	REQ'D REV RM RTU RWL	REQUIRED REVISION, REVISED ROOM ROOFTOP UNIT RAINWATER LEADER	1 A101 1 Detail Numb
	EL ELEC ELEV EPDM	EXTERIOR INSOLATION FINISH SYSTEM ELEVATION ELECTRIC, ELECTRICAL ELEVATOR ETHYLENE PROPYLENE DIENE TERPOLYMER	S SD SCHED SECT SF SGB	SOUTH SOAP DISPENSER SCHEDULE SECTION SQUARE FOOT or FEET SWING-UP GRAB BAR	
	ETR EQ EQUIP EX EXIST'G EXP	EXISTING TO REMAIN EQUAL EQUIPMENT EXISTING EXISTING EXPANSION	SGFT SH SHT SIM SJ SND	STRUCTURAL GLAZED FACING TILE SMOKE HATCH SHEET SIMILAR SEISMIC JOINT SANITARY NAPKIN DISPENSER	F1.0
	EXT FAC FB FD FDN	EXTERIOR FACTORY FIRE BLANKET FLOOR DRAIN FOUNDATION	SNR SPEC SPRKLR SQ SS STC	SANITARY NAPKIN RECEPTACLE SPECIFICATION SPRINKLER SQUARE STAINLESS STEEL SOUND TRANSMISSION CLASS	FL
	FE F.F. FIN FIXT FLR FLR'G FP	FIRE EXTINGUISHER FIISH FLOOR FINISH, FINISHED FIXTURE FLOORING FLOORING FIRE PROTECTION	STD STL STR STRUCT SUSP SYS	STANDARD STEEL STAIR STRUCTURAL SUSPENDED or SUSPENSION SYSTEM	
в	FRGP FRP FRT FT FTG FURN	FIBER REINFORCED GYPSUM PANEL FIBERGLASS REINFORCED PANEL FIRE RETARDANT TREATED FEET, FOOT FOOTING FURNITURE	T TB T'HOLD T&G T.O. TPD	TREAD TACKBOARD THRESHOLD TONGUE & GROOVE TOP OF TOILET PAPER DISPENSER	
	GA GALV GB GWB GYP	GAUGE GALVANIZED GRAB BAR, GYPSUM BOARD GYPSUM WALLBOARD GYPSUM	TRANS TYP UH UL UON	TRANSITION TYPICAL UNIT HEATER UNDERWRITERS' LABORATORIES INC UNLESS OTHERWISE NOTED	
	H HC H'CAP HB HORIZ HM	HIGH HANDICAPPED HANDICAPPED HOSE BIBB HORIZONTAL HOLLOW METAL	UV VCT VERT VIF, V.I.F. VS	UNIT VENTILATOR VENT VINYL COMPOSITION TILE VERTICAL VERIFY IN FIELD VENT STACK	
	HP HR HT	HIGH POINT HOUR HEIGHT INSIDE DIAMETER	VTR W W/ WB	VENT THROUGH ROOF WEST or WIDE WITH WHITE BOARD	
	IN INCL INFO INT INSUL	INCH, INCHES INCLUDE, INCLUDING INFORMATION INTERIOR INSULATION	WC WD WP WWF	WATER CLOSET WOOD WATERPRROOF, WATERPROOFING WELDED WIRE FABRIC ZINC COATED COPPER	
	JT LAB LAM LAV	JOINT LABORATORY LAMINATE, LAMINATED LAVATORY	ZTC	ZINC TIN COPPER	
A	LB LF LCC LP LT LTG	POUND LINEAR FOOT LEAD COATED COPPER LOW POINT LIGHT LIGHTING			



GENERAL NOTES

- 1. THE FOLLOWING NOTES APPLY TO ALL WORK AND DRAWINGS IN PROJECT SCOPE.
- 2. EXERCISE CARE NOT TO DAMAGE EXISTING BUILDING ELEMENTS AND FINISHES TO REMAIN DURING CONSTRUCTION. IN THE EVENT THAT DAMAGE DOES OCCUR PATCH/REPAIR OR REPLACE DAMAGED ELEMENTS OR FINISHES IN KIND AND/OR TO MATCH EXISTING.

2

- 3. ALL PIPING PENETRATIONS THROUGH EXISTING MASONRY WALLS SHALL BE DRILLED AND SLEEVED. PENETRATIONS THOUGH INTERIOR WALLS SHALL BE FIRE SAFED WITH DESIGNATED UL SYSTEMS. PENETRATIONS THROUGH EXTERIOR WALLS SHALL BE SEALED WEATHERTIGHT.
- 4. PROVIDE TEMPORARY BARRICADES AND DUST ABATEMENT AS REQUIRED BY OWNER.
- 5. THE PREMISIS SHALL BE KEPT FREE FROM ACCUMULATION OF TRASH AND CONSTRUCTION DEBRIS AT ALL TIMES. DISPOSE OF EXCESS TRASH AND DEBRIS OFF-SITE IN A SAFE, ACCEPTABLE MANNER.
- 6. LOCATE ALL EXISTING, EASEMENTS, UTILITIES AND BUILDING SERVICES, EXPOSED AND CONCEALED, PRIOR TO START OF WORK.
- 7. FIELD VERIFY EXISTING CONDITIONS AND DIMENSIONS PRIOR TO COMMENCEMENT OF WORK. REPORT ANY DISCREPANCIES TO ARCHITECT PRIOR TO PROCEEDING.
- 8. DIMENSIONS INDICATED ON DRAWINGS ARE TO BE USED FOR CONSTRUCTION. DO NOT SCALE DRAWINGS.
- 9. PARTITION DIMENSIONS SHOWN ON FLOOR PLANS ARE TO FINISH FACE OF WALL.
- 10. FIELD VERIFY ALL (±) AND (V.I.F) DIMENSIONS PRIOR TO START OF WORK. REPORT ANY DISCREPANCIES THAT WILL AFFECT NEW WORK TO ARCHITECT PRIOR TO PROCEEDING.
- 11. DOOR FRAMES AT CORNERS OF ROOMS ADJACENT TO INTERSECTING WALLS SHALL BE INSTALLED WITH THE DOOR OPENING 4" FROM INSIDE CORNER OF WALLS UNLESS OTHERWISE INDICATED. MEASUREMENT SHALL BE FROM FINISH FACE OF WALL.
- 12. IF INFORMATION ON DRAWINGS IS UNCLEAR OR ADDITIONAL DIMENSIONS ARE REQUIRED NOTIFY ARCHITECT PRIOR TO PROCEEDING WITH WORK.
- 13. THE TERM "TYPICAL" OR "TYP" INDICATED ON THE CONTRACT DOCUMENTS SHALL MEAN THAT THE CONDITION IS THE SAME OR SIMILAR THROUGHOUT UNLESS OTHERWISE INDICATED.
- 14. ALL EXISTING FINISHES AND BUILDING ELEMENTS AFFECTED BY DEMOLITION AND NEW WORK SHALL BE PATCHED TO MATCH EXISTING, WHETHER OR NOT SPECIFICALLY CALLED FOR.
- 15. ALL WORK SHALL BE PERFORMED IN A UNIFORM, ACCURATE MANNER AS REQUIRED TO PROVIDE NEAT, STRAIGHT LINES FREE FROM DEFECTS, OVERLAPS AND IMPERFECTIONS. WORK SHALL BE INSTALLED LEVEL, PLUMB OR FLUSH WITH ADJOINING MATERIALS (WHERE APPLICABLE). WORK OF EACH TRADE SHALL MEET OR EXCEED NATIONALLY RECOGNIZED AND/OR PUBLISHED STANDARDS. WHERE THE CONTRACT DOCUMENTS ARE MORE STRINGENT THAN THE AFOREMENETIONED STANDARDS WORK SHALL BE PERFORMED IN ACCORDANCE WITH SUCH.
- 16. COORDINATE WITH AUTHORITIES HAVING JURISDICTION AND PROVIDE ALL REQUIRED TEMPORARY FIRE AND LIFE SAFETY PROVISIONS DURING CONSTRUCTION.
- 17. PROVIDE PORTABLE FIRE EXTINGUISHERS DURING CONSTRUCTION AS REQUIRED BY CODE AND AUTHORITY HAVING JURISDICTION. THESE NOTES APPLY TO ALL CONTRACT DRAWINGS.

GENERAL DEMOLITION NOTES

- 1. THE FOLLOWING NOTES APPLY TO ALL WORK AND DRAWINGS IN PROJECT SCOPE.
- 2. THE DEMOLITION DRAWINGS IN THIS SET ARE TO BE USED AS A GUIDE AND ARE NOT INTENDED TO BE ALL INCLUSIVE OF REQUIRED WORK. ALL WORK SHOWN IN BROKEN LINE SHALL BE REMOVED UNLESS OTHERWISE NOTED. INSPECT WORK AREA TO VERIFY FULL EXTENT OF REQUIRED DEMOLITION PRIOR TO START OF WORK. NOTIFY ARCHITECT OF ANY DISCREPANCIES OR QUESTIONS PRIOR TO PROCEEDING.
- ALL WORK TO BE DEMOLISHED SHALL BE REMOVED IN ITS ENTIRETY, INCLUDING ALL RELATED FASTENERS AND COMPONENTS, AND AS REQUIRED TO PERFORM/INSTALL NEW WORK SHOWN ON THE CONTRACT DOCUMENTS.
- 4. NOTIFY BUILDING OWNER OF ALL MATERIALS OR EQUIPMENT, INTACT OR FUNCTIONAL AFTER REMOVAL, PRIOR TO DISPOSAL OR REMOVAL FROM SITE.
- 5. BRACE, SUPPORT AND/OR TEMPORARILY SHORE EXISTING WORK, PARTICULARLY LOAD BEARING COMPONENTS, PRIOR TO AND DURING DEMOLITION AND INSTALLATION OF NEW WORK.
- 6. PROVIDE TEMPORARY PROTECTION AS REQUIRED AND EXERCISE CARE NOT TO DAMAGE EXISTING BUILDING ELEMENTS AND FINISHES TO REMAIN. IN THE EVENT THAT THIS UNAVOIDABLE ALL EXISTING WORK THAT IS DAMAGED SHALL BE PATCHED/REPAIRED TO MATCH EXISTING IN MATERIAL, FINISH AND COLOR. WORK THAT CANNOT BE PROPERLY PATCHED/REPAIRED SHALL BE REPLACED IN KIND AT THE DISCRETION OF THE ARCHITECT.
- 7. LOCATE ALL EXISTING UTILITIES, BUILDING SERVICES AND EASEMENTS WITHIN THE WORK AREA PRIOR TO START OF DEMOLITION WORK. EXERCISE CARE NOT TO DAMAGE EXISTING WORK TO REMAIN. IN THE EVENT THAT THIS UNAVOIDABLE ALL EXISTING WORK THAT IS DAMAGED SHALL BE PATCHED/REPAIRED TO MATCH EXISTING. WORK THAT CANNOT BE PROPERLY PATCHED/REPAIRED SHALL BE REPLACED IN KIND AT THE DISCRETION OF THE ARCHITECT.
- 8. PLUMBING, FIRE PROTECTION, ELECTRICAL AND HVAC SYSTEMS NOT SUBJECT TO DEMOLITION SHALL REMAIN FUNCTIONAL AND BE PROTECTED AS REQUIRED OR SHALL BE RE-ROUTED. WHERE SHUT DOWN IS UNAVOIDABLE NOTIFY BUILDING OWNER AT LEAST 72 HOURS IN ADVANCE OF ANY NECESSARY INTERRUPTION TO SERVICE PRIOR TO START OF WORK.
- 9. AT EXISTING WALLS TO BE REMOVED EXISTING ELECTRICAL RECEPTACLES AND RELATED WIRING SHALL BE DISCONNECTED PRIOR TO REMOVAL.
- 10. ALL EXPOSED NON-FUNCTIONAL PIPING, WIRING, CONDUIT, BACK BOXES, DEVICES, FIXTURES, ETC ABANDONED AS A RESULT OF THIS PROJECT SHALL BE REMOVED, WHETHER OR NOT SPECIFICALLY CALLED FOR.
- 11. ALL EXISTING UTILITIES TO BE REMOVED THAT ARE STUBBED UP FROM THE FLOOR SHALL BE ABANDONED AND CAPPED BELOW FLOOR LEVEL.
- 12. SAW CUT EXISTING BITUMINOUS PAVEMENT AND CONCRETE FLOOR SLABS TO BE REMOVED IN CLEAN, STRAIGHT LINES.
- 13. WHERE DEMOLITION WORK CAUSES UNEVENNESS OR VOIDS IN THE EXISTING CONCRETE FLOOR SLAB PATCH AND LEVEL EXISTING FLOOR SURFACE TO RECEIVE NEW FINISHES.

PROFESSIONAL SEAL **CODE INFORMATION** 1. PROJECT SCOPE: IEBC Level 2 alterations to the existing building including: • Replacment of existing HVAC equipment, including boiler, water heater, AHUs, condensing units and fan coil units. installation of new exhaust system for existing toilet rooms and Third Floor of building. Installation of a new gas service from Lincoln Street Electrical system upgrade. • Replacemnt of the Basement brick floor with a concret slab. • Waterproofing of existing Basement masonry foundation walls. • Installation of a concrete walk, ramp and stairs at the West side of the building. • Replacment of Third Floor ceiling to facilitate installation of exhaust sytem. OWNER • HAZMAT abatement at araeas affected by work of this project. 2. APPLICABLE CODES: 2022 CONNECTICUT STATE BUILDING CODE which includes: Adopted Model Codes: 2021 International Building Code 2017 ICC/ANSI A117.1 Accessible and Usable Buildings and Facilities 2021 International Existing Building Code 2021 International Plumbing Code 2021 International Mechanical Code 2021 International Energy Conservation Code 2020 National Electrical Code (NFPA 70) 130 Lincoln Street Meriden, CT 06451 Connecticut Documents: 2022 Connecticut Amendments to the Adopted Model Codes Listed 203 235 7146 www.girlsincmeriden.org 2022 CONNECTICUT STATE FIRE SAFETY CODE which includes: ARCHITECT Adopted Model Code: 2021 International Fire Code Connecticut Documents: 2022 Connecticut Amendments to the 2021 International Fire Code 2018 CONNECTICUT STATE FIRE PREVENTION CODE which includes: Adopted Model Code: NFPA 1 - Fire Code, 2022 Edition 3. USE AND OCCUPANCY B: Business CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 No change of use as part of this Project 203 776 0184 www.cwarchitectsllc.com 4. BUILDING HEIGHT AND AREA Height: Not applicable. No Change in height as part of this project. MEP ENGINEER Area: Not applicable. No change in area as part of this project. 5. CONSTRUCTION CLASSIFICATION _ INNOVATIVE ENGINEERING SERVICES, LLC -IES-Existing Construction Type: IIB, Non-combustible, unprotected AN INTEGRATED ENGINEERING + DESIGN FIRM -6. FIRE RESISTANCE RATING OF ELEMENTS (See Code Plans) 9. INTERIOR FINISHES 33 North Plains Industrial Rd Wallingford, CT 06492 Not Appolicable: No chages to existing finishes as part of this project. 203 467 4370 8. FIRE PROTECTION SYSTEMS iesllc.biz Automatic Sprinkler System: No • Notification: Audible/visible alarms. No chage as part of this project • Detection: Heat detectors & manual pull stations. No change as part of this project. · Portable Fire Extinguishers: Provide at least one portable fire extinguisher in accordance with ENVIRONMENTAL CONSULTANT Section 906 of the International Fire Code at each of the following locations in the the Project Area[.] Rooms being used for storage during construction Rooms or areas where combustible and flammable liquids are being used. • Portable fire extinguisher locations and types shall be submitted to the Authority Having Jurisdiction (AHJ) and the Architect for review and approval prior to installation. EnviroMed 9. OCCUPANT LOAD & EGRESS Cleaner environment, Safer workplace Not applicable. No change as part of this project DESIGNED BY: JOHN LUBY 10. ACCESSIBILITY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986 Designated accessible building 11. PLUMBING FIXTURE COUNT Not Applicable. No chages to plumbing fixtures as part of this project. 12. MODIFICATIONS Existing: No known existing modifications New: None **ISSUED FOR BIDDING** MARCH 12, 2024 MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE CODE INFORMATION, GENERAL NOTES, SYMBOLS & ABBREVIATIONS G001 3/12/2024 2:37:26 PM Autodesk Docs://2313 Girls Inc/2313 Girls Inc - CWA - R24.rvt



SITE²WORK GENERAL

- 1. THE FOLLOWING NOTES APPLY TO ALL W
- 2. VERIFY THE ACTUAL LOCATION OF ALL UTI CONTACT "CALL BEFORE YOU DIG", 1-800-9 OF ALL EXISTING UTILITIES PRIOR TO ANY
- ALL ADJACENT PROPERTY OWNERS AFFEC MUST BE NOTIFIED A MINIMUM OF 48 HOUF CONSTRUCTION.
- 4. CONTACT THE MERIDEN DEPARTMENT OF AT (203) 630-4018 A MINIMUM OF 24 HOURS FAILURE TO DO SO MAY RESULT IN SHUT D
- 5. PROTECT THE CITY'S INFRASTRUCTURE FOR REPAIR OF ANY DAMAGE SHALL BE THE RE
- OBTAIN APPROVAL FROM THE CITY TREE EXCAVATION WITHIN THE DRIP LINE OF TR CONTACT THE CITY PARKS DEPARTMENT NUMBER (203) 630-4259.
- NO TREES SHALL BE REMOVED WITHIN TH THE PERMISSION OF THE CITY TREE WARE WITH THE CITY PARKS DEPARTMENT – 460 (203) 630-4259. 21 DAYS ADVANCE NOTICE
- TRAFFIC CONTROL, WHEN REQUIRED, SHA POLICE OFFICER(S). CONTACT POLICE SCH
- RESTORE PAVEMENT MARKINGS DISTURB UPON COMPLETION OF THE WORK IN MAT DIVISION. TEMPORARY TAPE, IF USED, MU PERMANENT MARKINGS ARE PLACED.
- 10. ALL TEMPORARY TRAFFIC CONTROL DEVIC STANDARDS OF THE MANUAL ON UNIFORM (M.U.T.C.D.) AND THE CITY OF MERIDEN.
- 11. ALL WORK ON CITY PROPERTY SHALL CON SPECIFICATIONS AND DETAILS. CONNDOT SHALL BE USED IF CITY DETAILS AND SPEN INCLUDES, BUT IS NOT LIMITED TO; TRENC REPAIR; CURB REPLACEMENTS; AND CITY
- 12. EXISTING CURBING, SIDEWALKS, PAVEMEN SHALL BE REPAIRED OR REPLACED, IN-KIN THEY BE DAMAGED DURING CONSTRUCTION
- 13. THE CONTRACTOR IS RESPONSIBLE FOR I PERMANENT PAVEMENT REPAIR. FAILURE OF RIGHT-OF-WAY LICENSE FEE.
- 14. PAVEMENT REPAIR IS CONSIDERED TEMPO INSTALLED PER CITY OF MERIDNEN ENGIN WINTER IS REQUIRED BEFORE THE PERMA
- 15. THE FINAL AREA OF ASPHALT REPAIR AND COORDINATED AND APPROVED BY DPW.
- 16. PRIOR TO PLACEMENT OF CONCRETE, THI ENGINEERING CONSTRUCTION SURVEY TO 630-4018 FOR FORM INSPECTION.
- 17. PROVIDE EROSION AND SEDIMENT CONTE SHALL COMPLY WITH THE LATEST SEDIME PUBLISHED BY THE CONNECTICUT DEPAR ENVIRONMENTAL PROTECTION

NOTES	GENÉRAL NOTES	PROFESSIONAL SEAL
INCELUIORK AND DRAWINGS IN PROJECT SCOPE.	1. The intent of the Contract Documents is to include all items	
TILITIES PRIOR TO START OF WORK.	necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as	
Y EXCAVATION. ECTED BY WORK OF THIS CONTRACT JRS PRIOR TO THE START OF	binding as if required by all. Where a conflict within the Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum.	
F PUBLIC WORKS ENGINEERING DIVISION S PRIOR TO START OF CONSTRUCTION.	Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade.	
FOR THE DURATION OF THE PROJECT.	REFERENCE KEY NOTES and SHEET KEY NOTES are intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of	
E WARDEN PRIOR TO PERFORMING ANY REES WITHIN THE CITY'S RIGHT OF WAY. F – 460 LIBERTY STREET, TELEPHONE	required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that Work can be logically inferred by other parts of the Contract Documents	
HE CITY RIGHT OF WAY (ROW) WITHOUT RDEN. TREE APPLICATIONS MUST BE FILED 30 LIBERTY STREET, TELEPHONE NUMBER	 Do not scale drawings, except for estimation and approximation purposes. 	9115
E IS REQUIRED.	 This is an existing building; all conditions and critical dimensions must be field verified. 	
CHEDULING AT 203-630-6305.		130 Lincoln Street Meriden CT 06451
TERIALS SUITABLE TO DPW – TRAFFIC UST BE IN GOOD CONDITION UNTIL THE	D3D.06 REMOVE CONDENSING UNIT & CONCRETE PAD	203 235 7146 www.girlsincmeriden.org
VICES MUST CONFORM TO THE RM TRAFFIC CONTROL DEVICES	E1D.01 REMOVE EXISTING POST MOUNTED MAILBOX. SLAVAGE FOR RELOCATION/REINSTALLATION	
ONFORM TO THE CITY STANDARD T FORM 816 SPECIFICATIONS AND DETAILS ECIFICATIONS ARE NOT AVAILABLE. THIS ICH BACKFILL MATERIALS; PAVEMENT	G1D.01 REMOVE BITUMINOUS WALK & CONCRETE STEP UP TO PORCH G1D.02 SAWCUT IN CLEAN, STRAIGHT LINES & REMOVE PORTION OF BITUMUOUS PAVEMENT & CURBING AS	
Y SIDEWALKS. ENT AND RELATED INFRASTRUCTURE IND, AT NO COST TO THE CITY SHOULD	G1D.03 SAWCUT IN CLEAN, STRAIGHT LINES & REMOVE PORTION OF BITUMUOUS PAVEMENT & CURBING (WHERE OCCURING) AS REQUIRED TO INCTACL AND	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven. CT 06511
MAINTAINING TRENCHES PENDING E TO DO SO MAY RESULT IN FORFEITURE	G1D.04 SAWCUT IN CLEAN, STRAIGHT LINES & REMOVE	203 776 0184 www.cwarchitectsllc.com
PORARY UNTIL A PERMANENT PATCH IS	(WHERE OCCURING) AS REQUIRED TO INSTALL NEW UNDERGROUND GAS SERVICE SHOWN ON SITE &	MEP ENGINEER
MANENT PATCH CAN BE INSTALLED.	G1D.05 REMOVE RED MAPLE TREE & ADJACENT GROUND COVER	INNOVATIVE ENGINEERING SERVICES, LLC
HE CONTRACTOR MUST CONTACT THE	G1D.06 SAW CUT TO NEAREST JOINT & REMOVE PORTION OF CONCRETE WALK AS REQUIRED TO INSTALL UNDERGROUND GAS SERVICE SHOWN ON SITE &	AN INTEGRATED ENGINEERING + DESIGN FIRM
TECHNICIAN AT (203) 537-3202 OR (203)	G1D.08 SAWCUT & REMOVE PORTION OF CURBING AS REQUIRED TO INSTALL UNDERGROUND GAS SERVICE SHOWN ON SITE & SITE UTILITY PLANS	33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370
ENT AND EROSION CONTROL MANUAL RTMENT OF ENERGY AND	G3.05 EXISTING PVC CLEANOUT. FIELD VERIFY ROUTING OF UNDERGROUND PIPING PRIOR TO START OF EXCAVATION FOR CONC RAMP	iesllc.biz
		ENVIRONMENTAL CONSULTANT
		EnvíroMed
		Cleaner environment. Safer workplaces.
		CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
		ISSUED FOR BIDDING
		MARCH 12, 2024
		MARK DATE DESCRIPTION
		PROJECT NO: CWA PROJECT NO.: 2313
		DRAWN BY: CHK'D BY:
		SITE DEMOLITION PLAN
	1	CD101

Autodesk Docs://2313 Girls Inc/2313 Girls Inc - CWA - R24.rvt

3/12/2024 2:37:27 PM



SITE²WORK GENERAL

- 1. THE FOLLOWING NOTES APPLY TO ALL W
- 2. VERIFY THE ACTUAL LOCATION OF ALL UTI CONTACT "CALL BEFORE YOU DIG", 1-800-9 OF ALL EXISTING UTILITIES PRIOR TO ANY
- 3. ALL ADJACENT PROPERTY OWNERS AFFEC MUST BE NOTIFIED A MINIMUM OF 48 HOUR CONSTRUCTION.
- 4. CONTACT THE MERIDEN DEPARTMENT OF AT (203) 630-4018 A MINIMUM OF 24 HOURS FAILURE TO DO SO MAY RESULT IN SHUT D
- 5. PROTECT THE CITY'S INFRASTRUCTURE F REPAIR OF ANY DAMAGE SHALL BE THE RI
- OBTAIN APPROVAL FROM THE CITY TREE V EXCAVATION WITHIN THE DRIP LINE OF TR CONTACT THE CITY PARKS DEPARTMENT -NUMBER (203) 630-4259.
- NO TREES SHALL BE REMOVED WITHIN TH THE PERMISSION OF THE CITY TREE WARE WITH THE CITY PARKS DEPARTMENT – 460 (203) 630-4259. 21 DAYS ADVANCE NOTICE
- TRAFFIC CONTROL, WHEN REQUIRED, SHA POLICE OFFICER(S). CONTACT POLICE SCI
- RESTORE PAVEMENT MARKINGS DISTURB UPON COMPLETION OF THE WORK IN MATI DIVISION. TEMPORARY TAPE, IF USED, MUS PERMANENT MARKINGS ARE PLACED.
- 10. ALL TEMPORARY TRAFFIC CONTROL DEVIC STANDARDS OF THE MANUAL ON UNIFORM (M.U.T.C.D.) AND THE CITY OF MERIDEN.
- 11. ALL WORK ON CITY PROPERTY SHALL CON SPECIFICATIONS AND DETAILS. CONNDOT SHALL BE USED IF CITY DETAILS AND SPEC INCLUDES, BUT IS NOT LIMITED TO; TRENC REPAIR; CURB REPLACEMENTS; AND CITY
- 12. EXISTING CURBING, SIDEWALKS, PAVEMEN SHALL BE REPAIRED OR REPLACED, IN-KIN THEY BE DAMAGED DURING CONSTRUCTION
- 13. THE CONTRACTOR IS RESPONSIBLE FOR I PERMANENT PAVEMENT REPAIR. FAILURE OF RIGHT-OF-WAY LICENSE FEE.
- 14. PAVEMENT REPAIR IS CONSIDERED TEMPO INSTALLED PER CITY OF MERIDNEN ENGIN WINTER IS REQUIRED BEFORE THE PERMA
- 15. THE FINAL AREA OF ASPHALT REPAIR AND COORDINATED AND APPROVED BY DPW.
- 16. PRIOR TO PLACEMENT OF CONCRETE, THE ENGINEERING CONSTRUCTION SURVEY TE 630-4018 FOR FORM INSPECTION.
- 17. PROVIDE EROSION AND SEDIMENT CONTE SHALL COMPLY WITH THE LATEST SEDIME PUBLISHED BY THE CONNECTICUT DEPAR ENVIRONMENTAL PROTECTION

NOTES	GENERAL NOTES	PROFESSIONAL SEAL
VORK AND DRAWINGS IN PROJECT SCOPE. TILITIES PRIOR TO START OF WORK. 9-922-4455, FOR LOCATION AND MARKING Y EXCAVATION. ECTED BY WORK OF THIS CONTRACT JRS PRIOR TO THE START OF	1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in	
RS PRIOR TO START OF CONSTRUCTION. DOWN OF CONSTRUCTION ACTIVITIES. FOR THE DURATION OF THE PROJECT.	 establishing the extent of Work to be performed by any trade. 2. REFERENCE KEY NOTES and SHEET KEY NOTES are intended to indicate and clarify the extent and requirements of the Work they do not indicate avery location or accurrence of 	
RESPONSIBILITY OF THE CONTRACTOR. WARDEN PRIOR TO PERFORMING ANY REES WITHIN THE CITY'S RIGHT OF WAY.	required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that Work can be logically inferred by other parts of the Contract	OWNER
THE CITY RIGHT OF WAY (ROW) WITHOUT RDEN. TREE APPLICATIONS MUST BE FILED 60 LIBERTY STREET, TELEPHONE NUMBER E IS REQUIRED.	 Documents 3. Do not scale drawings, except for estimation and approximation purposes. 4. This is an existing building; all conditions and critical dimensions must be field verified. 	
HALL BE BY UNIFORMED CITY OF MERIDEN CHEDULING AT 203-630-6305.	KEYNOTES	
BED BY CONSTRUCTION IMMEDIATELY TERIALS SUITABLE TO DPW – TRAFFIC UST BE IN GOOD CONDITION UNTIL THE	D3.01 PAD MOUNTED CONDENSING UNIT. SEE MECHANICAL DWGS D3.02 PAD MOUNTED AIR HANDLING UNIT. SEE	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
/ICES MUST CONFORM TO THE RM TRAFFIC CONTROL DEVICES	MECHANICAL DWGS G1.01 SHADED AREA: PATCH BITUMINOUS PAVEMENT FLUSH WITH EXISTING BITUMINOUS & EDGE OF CONC WALK. PATCH BITUMINOUS CURBING FLUSH	ARCHITECT
ONFORM TO THE CITY STANDARD T FORM 816 SPECIFICATIONS AND DETAILS ECIFICATIONS ARE NOT AVAILABLE. THIS ICH BACKFILL MATERIALS; PAVEMENT Y SIDEWALKS.	W/ ADJACENT CURBING G1.02 SHADED AREA: PATCH BITUMINOUS PAVEMENT FLUSH WITH EXISTING PAVEMENT. COORDINATE WITH UNDERGROUND WORK SHOWN ON SITE UTILITIES PLAN. INSTALL PAVEMENT @ LINCOLN STREET IN ACCORDANCE WITH CITY OF MERIDEN ENGINEERING DEPARTMENT STANDARDS	CW
ENT AND RELATED INFRASTRUCTURE IND, AT NO COST TO THE CITY SHOULD FION.	G1.05 CONCRETE RAMPS/WALK & STAIRS. SEE ENLARGED PLAN & DETAILS ON DWG C502 G1.06 REPLACE PORTION OF CONCRETE WALK REMOVED TO INSTALL GAS SERVICE IN ACCORDANCE WITH	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
MAINTAINING TRENCHES PENDING E TO DO SO MAY RESULT IN FORFEITURE	G1.07 REPLACE PORTION OF CONCRETE WALK REMOVED	MEP ENGINEER
PORARY UNTIL A PERMANENT PATCH IS INEERING STANDARD DETAILS. ONE JANENT PATCH CAN BE INSTALLED.	G1.08 REPLACE CURBING REMOVED FOR INSTALLATION	INNOVATIVE ENGINEERING SERVICES, LLC
ID/OR REPLACEMENT MUST BE HE CONTRACTOR MUST CONTACT THE	G1.09 EXISTING PVC CLEANOUT. EXERCISE CARE NOT TO DISTURB/DAMAGE DURING EXCAVATION EOD/INSTALLATION OF CONCRETE RAMP	AN INTEGRATED ENGINEERING + DESIGN FIRM
TECHNICIAN AT (203) 537-3202 OR (203)	G1.10 REGRADE & RESEED LAWN AREAS AFFECTED BY INSTALLATION OF CONCRETE RAMPS, STAIRS & LANDINGS	33 North Plains Industrial Rd Wallingford, CT 06492
IENT AND EROSION CONTROL MANUAL RTMENT OF ENERGY AND	G1.11 PROVIDE 8" OF CRUSHED STONE & FILTER FABRIC AT AREA BORDERED BY CONCRETE RAMP, STAIR & LANDINGS	iesllc.biz
	G1.12 EXTERIOR COMPOSITE FORCH DECK ON TREATED WD FRAMING. SEE DETAILS G1.13 6" DIA SCHEDULE 80 PIPE BOLLARD G2.01 UNDERGROUND GAS SERVICE FROM STREET BY UTILITY COMPANY. SEE SITE UTILITY PLAN. ROUTE LATERAL TO AVOID EXISTING TREES & PLANTINGS.	ENVIRONMENTAL CONSULTANT
	FINAL ROUTING TO BE FIELD VERIFIED PRIOR TO START OF EXCAVATION. EXCAVATE, BACKFILL & RESEED PER DETAILS & SPECIFICATIONS G2.02 GAS METER BY UTILITY COMPANY. SEE SITE UTILITY PLAN	EnvíroMed
	G3.01 UNDERGROUND ELECTRICAL SERVICE LATERAL FROM EXISTING POLE AT STREET. SEE SITE UTILITY PLAN. ROUTE LATERAL TO AVOID EXISTING TREES & PLANTINGS. FINAL ROUTING TO BE FIELD VERIFIED PRIOR TO START OF EXCAVATION. EXCAVATE BACKFILL & RESEED PER DETAILS & SPECIFICATIONS	DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	G3.02PAD MOUNTED ELECTRICAL TRANSFORMER. SEE SITE UTILITY PLANG3.03LIGHT POLE (TYP OF 4). SEE ELECTRICAL DWGS. FINAL LOCATIONS TO BE DETERMINED BY ARCHITECT IN FIELD	
	G3.04 EXISTING POLE. SEE SITE UTILITY PLAN. FIELD VERIFY EXACT LOCATION	ISSUED FOR BIDDING
		MARCH 12, 2024
		MARK DATE DESCRIPTION
		PROJECT NO: CWA PROJECT NO.: 2313
		DRAWN BY: CHK'D BY: COPYRIGHT
		SHEET TITLE
		SITE PLAN
	1	C101

3/12/2024 2:51:18 PM



1	PROFESSIONAL SEAL
TILL IMPORTED OR AMENDED EXIST'G PLANTING SOIL. ROT-TILL INTO SUBGRADE FOR A DEPTH OF 12"	
	OWNER
	UITS
	Ínc.®
FROM SITE CAN BE REUSED UPON APPROVAL BY ARCHITECT. ACED PLANTING SOILS. RAISE GRADES TO ALLOW FOR	130 Lincoln Street Meriden, CT 06451 203 235 7146 www.girlsincmeriden.org
S AFFECTED BY WORK OF THE CONTRACT SHALL BE ND RAKED OF ALL DEBRIS. NO STONES. WOOD CHIPS OR WILL BE PERMITTED	ARCHITECT
TO MATCH EXISTING LAWN AREAS. IULCHED WITH SALTB HYA OR UNROTTED SMALL GRAIN IN/AC OR 90 LBS/1,000 SF.	
AS FOLLOWS: 31 FOBER 31 UNTIL AN ACCEPTABLE STAND OF COVER, AS DETEMINED BY	
L DETAILS	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
0' 6" 1' 2' 3'	203 776 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
GRADE. GRASS OR BIT CONC PAVEMENT. SEE SITE PLAN & DTLS	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370 iesllc.biz
GRANULAR BACKFILL	ENVIRONMENTAL CONSULTANT
WARNING TAPE (YELLOW) <u>NOTES</u> : 1. TRENCHES OVER 5 FT DEEP IN STABLE SOIL SHALL BE	
 REINFORCED BY OSHA METHODS. 2. BOTTOMS OF TRENCHES SHALL BE COMPACTED PRIOR TO PLACEMENT OF SAND BEDDING. 3. IF SOIL BELOW TRENCHES IS FOUND TO BE UNSUITABLE, 	EnvíroMed
 GAS LINE. PROVIDE #2 TRACER WIRE CONTRACTOR SHALL PERFORM EXCAVATION & PROVIDE SAND BEDDING & BACKFILL. 	Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY
5. ENCASE PRIMARY ELECTRICAL CONDUIT CROSSING PAVED VEHICULAR PATHS W/ MINIMUM 2" CONCRETE - ENTIRE LENGTH OF PAVEMENT 6. CAS COMPANY SHALL PROVIDE CAS PIPING	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
Image: State of the state	
	ISSUED FOR BIDDING
<u>0' 6" 1' 2' 3'</u>	MARCH 12, 2024
SURFACE	
BINDER CLASS 1 & MATCH PROFILE OF EXIST'G ADJACENT CURBIING	
SEE BIT ())))))))))))))))))	
	MARK DATE DESCRIPTION
ED PROCESSED	PROJECT NO: CWA PROJECT NO.: 2313
E DASE ED SUBGRADE <u>NOTES</u> :	DRAWN BY: CHK'D BY:
INDICATED ON 1. INSTALL CURBING WHERE INDICATED ON SITE PLAN. 2. INSTALL TACK COAT ON ALL BUTT EDGES OF EXIST'G T. SPECFICATIONS CURBING.	SHEET TITLE
NFORM TO CT TION 3.06, LATEST	SITE DETAILS
<u>CURB SECTION</u>	
<u>0' 6" 1' 2' 3'</u>	C501
1 Autodesk Docs://2313 Girls Inc/2313 Gi	rls Inc - CWA - R24.rvt 3/12/2024 2:37:30 PM





0' 4' 8' 16'

GENERAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the	
Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade. 2. REFERENCE KEY NOTES and SHEET KEY NOTES are	
intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that Work can be logically inferred by other parts of the Contract	OWNER
 Documents Do not scale drawings, except for estimation and approximation purposes. 	UITS
 This is an existing building; all conditions and critical dimensions must be field verified. 	
KEYNOTES A2D.01 HATCHED AREA: REMOVE BRICK FLOOR & PORTION OF SUB-GRADE AS REQUIRED FOR PLACEMENT OF 4" CRUSHED STONE & CONCRETE SLAP	130 Lincoln Street Meriden, CT 06451 203 235 7146 www.girlsincmeriden.org
A2D.03 REMOVE DIRT FLOOR AT CRAWL SPACE AS REQUIRED TO INSTALL 4" OF CRUSHED STONE, VAPOR BARRIER & 3" CONCRETE SLAB	ARCHITECT
B3D.01 REMOVE EXISTING DOOR & FRAME C1D.02 TEMPORARILY SUPPORT WOOD STAIR TO FACILITATE DEMOLITION OF BRICK FLOOR & PLACEMENT OF CONCRETE FLOOR SLAB D2D.01 DISCONNECT & REMOVE HOT WATER HEATER.	
EXISTING HW & CW PIPING TO REMAIN. SEE PLUMBING DWGS D3D.02 REMOVE BOILER. SEE MECHANICAL DEMOLITION DRAWINGS	CHRISTOPHER WILLIAMS ARCHITECTS, LLC
D3D.03 REMOVE SUSPENDED AIR HANDLING UNIT, & PORTIONS EXISTING DUCTWORK. SEE MECHANICAL DEMOLITION DRAWINGS D3D.05 DISCONNECT & REMOVE OIL TANK. SEE	203 776 0184 www.cwarchitectsllc.com
MECHANICAL DEMO DWGS D5D.01 REMOVE WALL MOUNTED ELECTRICAL PANELS. SEE ELECTRICAL DEMOLITION DRAWINGS	
	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370 iesllc.biz
	ENVIRONMENTAL CONSULTANT
	EnvíroMed
	DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	MARCH 12, 2024
	MARK DATE DESCRIPTION
	PROJECT NO: CWA PROJECT NO.: 2313
	DRAWN BY: CHK'D BY: COPYRIGHT
	SHEET TITLE
	BASEMENT DEMOLITION PLAN
1	AD100

3/12/2024 4:29:45 PM



0' 4' 8' 16' 24'

2

2

GENÉRAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work	
by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum.	
Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors or in	
establishing the extent of Work to be performed by any trade.	
intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of	OWNER
Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that	
Work can be logically interred by other parts of the Contract Documents	
3. Do not scale drawings, except for estimation and approximation purposes.	
 I his is an existing building; all conditions and critical dimensions must be field verified. 	
C1D.01 REMOVE CONCRETE STEP	130 Lincoln Street Meriden, CT 06451
C3D.01 REMOVE WOOD PORCH, RELATED FOUNDATIONS & PORTIONS OF WOOD COLUMNS. SEE DEMOLITION ELEVATION	203 235 7146 www.girlsincmeriden.org
D3D.03 REMOVE SUSPENDED AIR HANDLING UNIT, & PORTIONS EXISTING DUCTWORK. SEE MECHANICAL	ARCHITECT
D3D.04 REMOVE WALL MOUNTED FAN COIL UNIT & ALL RELATED COMPONENTS. SEE MECHANICAL	
DEMOLITION DRAWINGS D3D.07 REMOVE BASEBOARD RADIATION D3D.00 DEMOVE EXISTING ELOOD MOUNTED AID TERMINAL	
SEE MECHANICAL DEMOLITION DWGS	
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	INNOVATIVE ENGINEERING SERVICES, LLC
	AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370
	ICSIIC.DIZ
	ENVIRONMENTAL CONSULTANT
	Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING MARCH 12, 2024
	MARK DATE DESCRIPTION
	PROJECT NO: CWA PROJECT NO · 2313
	DRAWN BY:
	COPYRIGHT
	SHEET TITLE FIRST FLOOR DEMOLITION PLAN
	۵ח101
1	
Autodesk Docs://2313 Girls Inc/2313 G	irls Inc - CWA - R24.rvt 3/12/2024 2:37:32 PM





0' 4' 8' 16'

2

2

GENÉRAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work	
by the Contractor. The Contract Documents are complementary, and what is required by one shall be as	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the	
Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the	
Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade.	
2. REFERENCE KEY NOTES and SHEET KEY NOTES are intended to indicate and clarify the extent and requirements of	
the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute	OWNER
that Work as part of the Contract when the requirement for that Work can be logically inferred by other parts of the Contract	
3. Do not scale drawings, except for estimation and	
approximation purposes.	
4. This is an existing building; all conditions and critical dimensions must be field verified.	
KEYNOTES	
D3D.04 REMOVE WALL MOUNTED FAN COIL UNIT & ALL RELATED COMPONENTS. SEE MECHANICAL DEMOLITION DRAWINGS	203 235 7146 Meriden, CT 06451 www.girlsincmeriden.org
	ARCHITECT
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
	203 / /6 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370
	ENVIRONMENTAL CONSULTANT
	_
	EnviroMed
	Cleaner environment. Safer workplaces.
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	1550ED FOR BIDDING
	MARCH 12, 2024
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY:
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT
	MARK DATE DESCRIPTION MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: CHK'D BY: COPYRIGHT SHEET TITLE
	MARK DATE DESCRIPTION MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE SECOND FLOOR
	MARK DATE DESCRIPTION MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE SECOND FLOOR DEMOLITION PLAN
	MARK DATE DESCRIPTION MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: CHK'D BY: COPYRIGHT SHEET TITLE SECOND FLOOR DEMOLITION PLAN
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE SECOND FLOOR DEMOLITION PLAN
	MARK DATE DESCRIPTION MARK DATE MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO: SHEET TITLE SHEET TITLE SECOND FLOOR DEMOLITION PLAN

\square	\square
\square	\Box



NOTE: THIS DRAWING IS FOR INFORMATION ONLY. SEE THIRD FLOOR REFLECTED CEILING PLAN, MECHANICAL, & ELECTRICAL PLANS FOR EXHAUST SYSTEM & RELATED CEILING REPLACMENT

2

0' 4' 8' 16' 24'

2

	PROFESSIONAL SEAL
GENERAL NOTES	
necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are	
complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the Contract Documents exists, the Contractor shall provide the	
better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum.	
articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in	
establishing the extent of Work to be performed by any trade.	
intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of	OWNER
required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that	
Work can be logically inferred by other parts of the Contract Documents	
 Do not scale drawings, except for estimation and approximation purposes. 	
 This is an existing building; all conditions and critical dimensions must be field verified. 	
KEYNOTES	
	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
	ARCHITECT
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
	203 776 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	<u> </u>
	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370 jesllc.biz
	ENVIRONMENTAL CONSULTANT
	EnviroMod
	DESIGNED BY: JOHN LUBY
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	MARCH 12, 2024
	PROJECT NO:
	CWA PROJECT NO.: 2313
	DRAWN BY: CHK'D BY:
	COPYRIGHT
	SHEET TITLE
	AD103
1	

Autodesk Docs://2313 Girls Inc/2313 Girls Inc - CWA - R24.rvt

3/12/2024 2:37:33 PM





0' 4' 8' 16' 2

2

2

GENERAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work	
by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum.	
Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in	
establishing the extent of Work to be performed by any trade. 2. REFERENCE KEY NOTES and SHEET KEY NOTES are	
intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of	OWNER
Work as part of the Contract when the requirement for that Work as part of the Contract when the requirement for that	
Documents	
approximation purposes.	
dimensions must be field verified.	
KEYNOTES A2.01 CONCRETE SLAB ON GRADE @ BASEMENT: 4" THICK	130 Lincoln Street Meriden, CT 06451
CONCRETE W/ 6 X 6 2.0 - 2.0 W.W.F. ON CONTINUOUS 10 MIL POLY VAPOR BARRIER ON 4" OF CRUSHED STONE. TURN VAPOR RETARDER	203 235 / 146 www.girlsincmeriden.org
VERTICALLY (FULL THICKNESS OF SLAB) UP FACE OF ALL WALLS, COLUMNS & FLOOR PENETRATIONS. PROVIDE CONTINUOUS 1/2" PREFORMED FILLER &	
SEALANT AT INTERSECTION OF SLAB WITH WALLS AND COLUMNS. PROVIDE CONTROL JOINTS PER CAST IN PLACE CONCRETE SPECIFICATIONS OR AS	
A2.02 INCREASE SIZE OF EXISTING RAISED CONC PAD AS BEOURED TO ACCOMODATE NEW BOILERS	
COORDINATE PAD SIZE WITH ACTUAL BOILER DIMENSIONS. SEE MECH DWGS	
A3.01 SHOKE OF WOOD TIMBER BEAM & REMOVE LOOSE & DAMAGED BRICK AT TOP OF PIER DOWN TO SOUND MATERIAL. RECONSTRUCT PIER TO	85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
DINDERSIDE OF BEAM. REPLACE DAMAGED OR DETRIORATED BRICK IN KIND. ASSUMED TO BE MOLDED COMMON BRICK (FIELD VERIFY)	
C2.01 BROKEN LINE: REMOVE LOOSE PAINT & MORTAR, PREP & APPLY CRYSTALLINE WATERPROOFING TO INTERIOR FACE OF EXISTING EXTERIOR MASONRY	MEP ENGINEER
FOUNDATION WALLS - FULL HEIGHT (TYPICAL) D2.01 HOT WATER HEATER. SEE PLUMBING DWGS D3.02 PAD MOUNTED AIR HANDLING UNIT OFF	INNOVATIVE ENGINEERING SERVICES, LLC
D3.10 BOILERS. SEE MECHANICAL DWGS	AN INTEGRATED ENGINEERING + DESIGN FIRM
D5.01 ELECTRICAL PANELS ON WALL MOUNTED PAINTED, FIRE RETARDANT TREATED PLYWOOD PANEL(S). SEE ELECTRICAL DRAWINGS	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370 iesllc.biz
	ENVIRONMENTAL CONSULTANT
	EnvíroMed
	Cleaner environment. Safer workplaces.
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	MARCH 12, 2024
	PROJECT NO:
	CWA PROJECT NO.: 2313 DRAWN BY:
	CHK'D BY:
	COPYRIGHT
	PLAN
	A 4 0 0
1	A100

3/12/2024 3:09:20 PM


0'____4'___8'____16'____

2

GI	ENÉRAL NOTES	PROFESSIONAL SEAL
1. The i neces	ntent of the Contract Documents is to include all items ssary for the proper execution and completion of all Work	
by the comp	e Contractor. The Contract Documents are lementary, and what is required by one shall be as	
bindir Contr	ng as if required by all. Where a conflict within the ract Documents exists, the Contractor shall provide the	
Archit	tect's resolution without any increase in the Contract sum. nization of the Specifications into divisions, sections and	
article	es, and arrangement of sheets shall not control the actor in dividing the work among subcontractors, or in	
estab	Ishing the extent of Work to be performed by any trade.	
intend the W	ded to indicate and clarify the extent and requirements of /ork; they do not indicate every location or occurence of	OWNER
requir Work that V	red Work. The lack of a key note to an individual item of will not relieve the Contractor of responsibility to execute Work as part of the Contract when the requirement for that	
Work	can be logically inferred by other parts of the Contract ments	
3. Do no	ot scale drawings, except for estimation and	
4. This i	is an existing building; all conditions and critical	
dimer	nsions must be field verified.	
C2.02	EYNOTES PATCH EXISTING PLASTER OR GYPSUM BD/WD STUE	130 Lincoln Street Meriden, CT 06451
01.01	WALL AFFECTED BY REMOVAL OF EXISTING & INSTALLATION OF REPLACEMENT FAN COIL UNIT.	203 235 7146 www.girlsincmeriden.org
	SEAL ALL PENETRATIONS. PAINT ENTIRE WALL - FULL LENGTH & HEIGHT (FIELD VERIFY DIMENSIONS)	ARCHITECT
C2.04	PATCH EXISTING CMU WALL AFFECTED BY REMOVAL OF EXISTING & INSTALLATION OF	
	REPLACEMENT RADIATION. PAINT ENTIRE WALL - FULL LENGTH & HEIGHT (FIELD VERIFY DIMENSIONS	
C2.08	PROVIDE VCT FLOORING & RESILIENT BASE. ENTIRE ROOM	
C2.09	COLOR TO MATCH EXIST'G REPLACE EXISTING LOCKSET WITH LEVER HANDLE	
55.01	LOCKSET. SEE HARDWARE SPEC. MODIFY WOOD DOOR & FRAME TO RECEIVE NEW LOCKSET. PAINT	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
D0.02	ENTIRE DOOR & FRAME AFTYER INSTALLATION OF REPLACEMENT HARDWARE	2037700164 www.cwarchitectsllc.com
ט3.08 <u>D3 11</u>	SEE DWG C501 FOR CONC PAD DETAIL	MEP ENGINEER
D3.12	MECHANICAL DWG BASEBOARD RADIATION. SEE MECHANICAL DWGS	
D3.13	FAN COIL UNIT MOUNTED HIGH ON WALL. SEE MECHANICAL DWGS	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIDM
D3.14	CEILING MOUNTED FAN COIL UNIT. SEE MECHANICAL DWGS. REMOVE, MODIFY & REINSTALL PORTIONS OF EXISTING 2x2 SUSPENDED ACT	
	CEILING AS REQUIRED FOR INSTALLATION	33 North Plains Industrial Rd Wallingford, CT 06492
		203 467 4370
		iesllc.biz
		ENVIRONMENTAL CONSULTANT
		EnvíroMed
		Cleaner environment. Safer workplaces.
		CT ASBESTOS PROJECT DESIGNER LICENSE #19
		CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
		ISSUED FOR BIDDING
		MARCH 12 2024
		MARK DATE DESCRIPTION
		PROJECT NO:
		CWA PROJECT NO.: 2313
		DRAWN BY: CHK'D BY:
		COPYRIGHT
		SHEET TITLE
		FIRST FLOOR PLAN
1		A101
	1	

Autodesk Docs://2313 Girls Inc/2313 Girls Inc - CWA - R24.rvt

3/12/2024 2:37:37 PM



GENÉRAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work	
by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum	
Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the	
establishing the extent of Work to be performed by any trade.	
2. REFERENCE KEY NOTES and SHEET KEY NOTES are intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of	OWNER
required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that	
Work can be logically inferred by other parts of the Contract Documents	
Do not scale drawings, except for estimation and approximation purposes.	
 This is an existing building; all conditions and critical dimensions must be field verified. 	
KEYNOTES	130 Lincoln Street Meriden, CT 06451
WALL AFFECTED BY REMOVAL OF EXISTING & INSTALLATION OF REPLACEMENT FAN COIL UNIT. SEAL ALL PENETRATIONS. PAINT ENTIRE WALL - FULL LENGTH & HEIGHT (FIELD VERIFY	203 235 7146 www.girlsincmeriden.org ARCHITECT
DIMENSIONS) C2.03 PATCH EXISTING PLASTER OR GYPSUM BD/WD STUD	
UNIT. SEAL ALL PENETRATIONS. PAINT ENTIRE WALL - FULL LENGTH & HEIGHT (FIELD VERIFY	
C2.05 PATCH DUCT PENETRATION THRU PLASTER OR GYPSUM BD/WD STUD PARTITON. PAINT ENTIRE	
WALL (BOTH SIDES - FULL HT & LENGTH) C2.06 PATCH DUCT PENETRATION THRU PLASTER OR GYPSUM BD/WD STUD WL. PAINT ENTIRE WALL -	CHRISTOPHER WILLIAMS ARCHITECTS, LLC
FULL HT & LENGTH D3.13 FAN COIL UNIT MOUNTED HIGH ON WALL. SEE MECHANICAL DIAGS	85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
D3.15 CEILING EXHAUST FAN W/ DUCT THROUGH WALL & UP TO CAP ON ROOF. SEE MECHANICAL DWGS.	MEP ENGINEER
D3.16 CEILING EXHAUST FAN W/ DUCT THROUGH WALL EXTERIOR WALL. SEE MECHANICAL DWGS. EXACT	
LOCATION & ROUTING TO BE DETERMINED IN FIELD	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd
	Wallingford, CT 06492 203 467 4370
	iesllc.biz
	ENVIRONMENTAL CONSULTANT
	DESIGNED BY: JOHN LUBY
	CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	MARCH 12 2024
	MARK DATE DESCRIPTION
	PROJECT NO: CWA PROJECT NO.: 2313
	DRAWN BY: CHK'D BY:
	COPYRIGHT
	SHEET TITLE
	SECOND FLOOR PLAN
	A 4 0 0
1	AIUZ
Autodesk Docs://2313 Cirls Inc/2313 Ci	rls Inc - CWA - R24 rvt $3/12/2024 2.37.37 PM$



NOTE: THIS DRAWING IS FOR INFORMATION ONLY. SEE THIRD FLOOR REFLECTED CEILING PLAN, MECHANICAL, & ELECTRICAL PLANS FOR EXHAUST SYSTEM & RELATED CEILING REPLACMENT

C5 A201

0' 4' 8' 16'

PROFESSIONAL SEAL
OWNER
B B
130 Lincoln Street Meriden, CT 06451 203 235 7146 www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC
85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
MEP ENGINEER
SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370
163116.DIZ
ENVIRONMENTAL CONSULTANT
EnvíroMed
Cleaner environment. Safer workplaces.
DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
 MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024
MARCH 12, 2024





2

	GENERAL NOTES	PROFESSIONAL SEAL
	1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and articles, and arrangement of spects shall not control the	
	Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade.	
	intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of	OWNER
	that Work as part of the Contractor of responsibility to execute Work can be logically inferred by other parts of the Contract Documents	
	Do not scale drawings, except for estimation and approximation purposes.	
	 This is an existing building; all conditions and critical dimensions must be field verified. 	R R
	KEYNOTES	130 Lincoln Street Maridan CT 06451
	C5.01 1/2" GYPSUM BOARD CEILING - ENTIRE THIRD FLOOR C5.01 1/2" GYPSUM BOARD ON EXISTING 2X WD SUPPORT FRAMING. FIELD VERIFY SUPPORT FRAMING SIZE & CONFIGURATION. PROVIDE SUPPLEMENTAL WOOD FRAMING AS REQUIRED TO FASTEN GYPSUM BOARD @ 24" O.C. MAX	ARCHITECT
	C5.02 1/2" GYSPUM BOARD ON EXISTING SLOPED WOOD RAFTERS. FIELD VERFIY RAFTER SIZE & SOACING. PROVIDE SUPPLEMENTAL WOOD FRAMING AS REQUIRED TO FASTEN GYPSUM BOARD @ 24" O.C. MAX	CWJ
	C5D.02 HATCHED AREA. REMOVE SLOPED BATTEN SEAM MASONITE CEILING FROM UNDESIDE OF EXISTING WOOD RAFTERS	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
	C6.02 24 x 36 CEILING ACCESSS DOOR. SUPPLEMENTAL 2x FRAMING AS REQUIRED FOR INSTALLATION C6.03 24 x24 CEILING ACCESSS DOOR. MODIFY EXISTING/PROVIDE SUPPLEMENTAL 2x FRAMING AS REQUIRED FOR INSTALLATION. COORDINATE/VERIFY EXACT SIZE & LOCATION W/	
	INLINE EXHAUST FAN ABOVE CEILING. SEE MECH DWGS D3.18 CEILING MOUNTED EXHAUST AIR TERMINAL. SEE MECHANICAL DRAWINGS D5.02 REINSTALL OR REPLACE EXISTING CEILING MOUNTED HEAT DETECTOR. THE INTO FIRE ALARM	33 North Plains Industrial Rd
	SYSTEM. SEE ELECTRICAL DRAWINGS D5.03 SURFACE MOUNTED LIGHTING FIXTURE. SEE ELECTRICAL DWGS	Wallingford, CT 06492 203 467 4370 iesllc.biz
	ELECTRICAL DEMOLITION DRAWINGS	ENVIRONMENTAL CONSULTANT
		Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #1986 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
		MARCH 12, 2024
		MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT
\bigvee		SHEET TITLE THIRD FLOOR REFLECTED CEILING PLANS - DEMOLITION & NEW
	1	A113

3/12/2024 3:18:35 PM



0' 4' 8' 16'

GENÉRAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work	
by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the	
Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and articles, and arrangement of spects shall not control the	
Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade.	
2. REFERENCE KEY NOTES and SHEET KEY NOTES are intended to indicate and clarify the extent and requirements of	
the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of Work will not reliave the Contractor of recomposibility to every	OWNER
that Work can be logically inferred by other parts of the Contract	
Documents 3. Do not scale drawings, except for estimation and	UITS
approximation purposes.4. This is an existing building: all conditions and critical	
dimensions must be field verified.	
B3.03 CURB MOUNTED ROOFTOP EXHAUST FAN HOOD.	130 Lincoln Street Meriden, CT 06451
SEE MECHANICAL DRAWINGS. COORDINATE LOCATION W/ ROOF FRAMING, CUT OPENING IN WOOD ROOF DECK & FLASH INTO EXISTING	
ASPHALT SHINGLE ROOFING. PROVIDE SUPPLEMENTAL 2x WD SUPPORT FRAMING BELOW BOOF DECK & BLOCKING AS REQUIRED TO	ARCHITECT
SUPPORT & SECURELY ANCHOR HOOD IN PLACED3.04CURB MOUNTED ROOFTOP EXHAUST FAN HOOD.	
SEE MECHANICAL DRAWINGS. COORDINATE LOCATION W/ ROOF FRAMING, CUT OPENING IN WOOD ROOF DECK & FLASH HOOD INTO EXISTING	
SBS (MODIFIED BITUMEN) ROOFING PER ROOFING SYSTEM MFR RECOMMENDATIONS. PROVIDE	
ROOF DECK & BLOCKING AS REQUIRED TO SUPPORT & SECURELY ANCHOR HOOD IN PLACE	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
D3.15 CEILING EXHAUST FAN W/ DUCT THROUGH WALL & UP TO CAP ON ROOF. SEE MECHANICAL DWGS. EXACT ROUTING TO BE DETERMINED IN FIELD	203 776 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370 iesllc.biz
	ENVIRONMENTAL CONSULTANT
	EnvíroMed
	Cleaner environment. Safer workplaces.
	DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19
	OT LEAD FLANNER FROJEOT DESIGNER LICENSE #586
	MARCH 12, 2024
_	
	PROJECT NO:
	CWA PROJECT NO.: 2313
	DRAWN BY: CHK'D BY:
	COPYRIGHT
	SHEET TITLE
	ROOF PLAN
	A120
1 Autodesk Docs://2313 Girls Inc/2313 G	irls Inc - CWA - R24.rvt 3/12/2024 2:37:39 PM



3/12/2024 3:24:27 PM



				100 - 2 3/10	
JN					
	0'	4'	8'	16'	24'

	GENÉRAL NOTES	PROFESSIONAL SEAL
	 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade. REFERENCE KEY NOTES and SHEET KEY NOTES are 	
	 intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that Work can be logically inferred by other parts of the Contract Documents 3. Do not scale drawings, except for estimation and approximation purposes. 	OWNER
<u>THIRD FLOOR</u> 197' - 8 7/16"	 4. This is an existing building; all conditions and critical dimensions must be field verified. KEYNOTES D3.01 PAD MOUNTED CONDENSING UNIT. SEE MECHANICAL DWGS D3.05 CONDENSING UNIT LINESETS, WIRING & ENCLOSURE. SEE MECH DWGS. REUSE EXISTING 	130 Lincoln Street 203 235 7146 Meriden, CT 06451 www.girlsincmeriden.org
S <u>ECOND FLOOR</u> 186' - 9 13/16"	WALL PENETRATIONS TO INTERIOR FCU D3.06 CONDENSING UNIT LINESETS, WIRING & ENCLOSURE. SEE MECH DWGS D3.07 OUTLINE OF EXTERIOR PAD MOUNTED AHU IN FOREGROUND. SEE MECH DWGS D3.09 DUCT PENETRATIONS THRU EXISTING EXTERIOR WOOD STUD WALL SEE DETAIL	CWN
<u>FIRST_FLOOR</u> 174' - 0"		CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com MEP ENGINEER
<u>BASEMENT</u> 166' - 2 3/16"		AN INTEGRATED ENGINEERING + DESIGN FIRM 33 North Plains Industrial Rd Wallingford, CT 06492
8' 16' 24'		203 467 4370 iesllc.biz ENVIRONMENTAL CONSULTANT
		Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
		ISSUED FOR BIDDING MARCH 12, 2024
		MARK DATE DESCRIPTION
		CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT
		SOUTH & WEST BUILDING ELEVATIONS
	1	A202
	Autodesk Docs://2313 Cirls Inc/2313 C	THE IDO LINUA DUA NOT 3/10/000/ 0.37.43 DM









GENERAL NOTES	PROFESSIONAL SEAL
 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work 	
by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the	
Contract Documents exists, the Contractor shall provide the better quality or greater quantity of work in accordance with the	
Architect's resolution without any increase in the Contract sum. Organization of the Specifications into divisions, sections and	
articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in	
establishing the extent of Work to be performed by any trade.	
intended to indicate and clarify the extent and requirements of the Work: they do not indicate every location or occurence of	
required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute	OWNER
that Work as part of the Contract when the requirement for that Work can be logically inferred by other parts of the Contract	
3. Do not scale drawings, except for estimation and	
approximation purposes.	J
This is an existing building; all conditions and critical dimensions must be field verified.	
KEYNOTES	
H2F.01 FLOOR OF BASEMENT BEING DEMOLISHED. NO LEAD-BASED PAINT OR ACM IDENTIFIED ON	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
FLOOR. H2D.03 DIRT FLOOR BEING EXCAVATED IN CRAWLSPACE.	ARCHITECT
NO ACM IDENTIFIED IN CRAWLSPACE. H3D.01 REMOVE EXISTING DOOR & FRAME	
PAINTED WITH LEAD-BASED PAINT AND DISPOSE AS HAZARDOUS LEAD WASTE.	
H1D.02 WOOD STAIR IS NOT PAINTED WITH LEAD-BASED PAINT.	
H2D.01 HOT WATER HEATER BEING REMOVED. NO ACM ASSOCIATED WITH HOT WATER HEATER	
H3D.02 BOILER BEING REMOVED. NO ACM ASSOCIATED WITH BOILER. WHITE SECTIONAL ROPE	
GASKETING AND RED CEMENT IS NON-ACM. H3D.03 SUSPENDED AIR HANDLING UNIT & ASSOCIATED	85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarebitectelle.com
DUCTWORK BEING REMOVED. NO ACM ASSOCIATED WITH UNIT OR DUCTWORK.	
HSD.05 UIL TANK BEING REMOVED. NO ACM OR LEAD-BASED PAINT ASSOCIATED WITH TANK.	MEP ENGINEER
REMOVED. NO ACM ASSOCIATED WITH THE PANELS.	· · · · · · · · · · · · · · · · · · ·
PIPE ASBESTOS-CONTAINING PIPE INSULATION IS PRESENT IN CRAWLSPACE AND REMAINS IN-PLACE.	INNOVATIVE ENGINEERING SERVICES, LLC
CONDUIT LARGE WHITE METAL ELECTRICAL CONDUIT IS	
PAINTED WITH LEAD-BASED PAINT.	33 North Plains Industrial Rd
WALL.NO WALLS IN ROOM ARE NOT PAINTED WITH LEAD-BASED PAINT.	203 467 4370
WHITE METAL CEILING IS PAINTED WITH	iesllc.biz
LEAD-BASED PAINT.	
	ENVIRONMENTAL CONSULTANT
	EnviroMod
	Environed
	Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	MARCH 12, 2024
	,,
	MARK DATE DESCRIPTION
	DRAWN BY:
	CHK'D BY:
	COPYRIGHT
	SHEET TITLE
	BASEMENT
	BASEMENT ABATEMENT PLAN
	BASEMENT ABATEMENT PLAN
	BASEMENT ABATEMENT PLAN
	BASEMENT ABATEMENT PLAN H100



GE	ENERAL NOTES	PROFESSIONAL SEAL
1. The ir neces	ntent of the Contract Documents is to include all items sary for the proper execution and completion of all Work	
by the compl bindin	Contractor. The Contract Documents are ementary, and what is required by one shall be as g as if required by all. Where a conflict within the	
Contra better	act Documents exists, the Contractor shall provide the quality or greater quantity of work in accordance with the	
Archit Organ article	s, and arrangement of sheets shall not control the	
Contra establ	actor in dividing the work among subcontractors, or in is ishing the extent of Work to be performed by any trade.	
2. REFE intend the W	RENCE KEY NOTES and SHEET KEY NOTES are ed to indicate and clarify the extent and requirements of ork: they do not indicate every location or occurence of	
require Work that M	ed Work. The lack of a key note to an individual item of will not relieve the Contractor of responsibility to execute (ork as part of the Contract when the requirement for that	
Work Docur	can be logically inferred by other parts of the Contract nents	
3. Do no appro:	ot scale drawings, except for estimation and ximation purposes.	
4. This is dimen	s an existing building; all conditions and critical sions must be field verified.	
KE	EYNOTES	
H2.02	PLASTER OR WALLBOARD WALL FINISH IS PAINTED WITH LEAD-BASED PAINT. EXISTING RADIANT HEATING UNIT BEING REMOVED, NEW UNIT BEING INSTALLED, AND ENTIRE WALL IS BEING PAINTED. CONDUCT EXISTING HEATING UNIT DEMO, NEW HEATING UNIT INSTALL, AND WALL PAINT PREP IN COMPLIANCE WITH SECTION	130 Lincoln Street Meriden, CT 06451 203 235 7146 www.girlsincmeriden.org ARCHITECT
H2.04	028233. CMU WALL FINISH IS PAINTED WITH LEAD-BASED PAINT. EXISTING RADIANT HEATING UNIT BEING REMOVED, NEW UNIT BEING INSTALLED, AND ENTIRE WALL IS BEING PAINTED. CONDUCT EXISTING HEATING UNIT DEMO, NEW HEATING	CWJ
	UNIT INSTALL, AND WALL PAINT PREP IN COMPLIANCE WITH SECTION 028233.	CHRISTOPHER WILLIAMS ARCHITECTS. LLC
H3.08	AT THIS LOCATION. CUTTING THROUGH EXTERIOR WALL, PAINTED WITH LEAD-BASED PAINT, IS REQUIRED TO INSTALL NEW DUCTWORK. CONDUCT CUTTING IN COMPLIANCE WITH SECTION 028233.	85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
H3.11	NEW FLOOR-MOUNTED AIR TERMINAL BEING INSTALLED IN EXISTING FLOOR OPENING AT THIS LOCATION. NO ASBESTOS OR LEAD-BASED PAINT IMPACT ANTICIPATED.	INNOVATIVE ENGINEERING SERVICES, LLC
H3.12	BASEBOARD RADIATOR REMAINS IN-PLACE. NO ASBESTOS OR LEAD-BASED PAINT IMPACT ANTICIPATED.	AN INTEGRATED ENGINEERING + DESIGN FIRM
H3.13	FAN COIL UNIT REMAINS IN-PLACE. NO ASBESTOS OR LEAD-BASED PAINT IMPACT ANTICIPATED.	33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370
H3.14	NEW CEILING-MOUNTED FAN COIL UNIT BEING INSTALLED AT THIS LOCATION IMPACTING CEILING PAINTED WITH LEAD-BASED PAINT, CONDUCT DRILLING IN COMPLIANCE WITH SECTION 028233.	
VAT.GO	VINYL ASBESTOS FLOOR TILE & MASTIC PRESENT IN ROOM. REMOVE AND DISPOSE OF VAT & MASTIC USING NEGATIVE PRESSURE CONTAINMENT. REMOVE MASTIC USING WET MECHANICAL METHODS.	
VAT.NO	VINYL ASBESTOS FLOOR TILE & MASTIC PRESENT IN ROOM UNDER CARPET. LEAVE VAT IN-PLACE.	EnviroMed
MPF.GO	ASBESTOS-CONTAINING MUDDED PIPE FITTING INSULATION (1 FITTING) PRESENT ON DOMESTIC WATER LINE. REMOVE PIPE FITTING INSULATION AND 6" OF FIBERGLASS PIPE INSULATION TO EITHER SIDE USING THE GLOVEBAG REMOVAL TECHNIQUE.	DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
FLEX.GO	ASBESTOS-CONTAINING FLEXIBLE DUCT CONNECTORS (2) PRESENT ON HVAC UNIT. REMOVE AND DISPOSE OF THE FLEXIBLE DUCT CONNECTORS USING A NEGATIVE PRESSURE CONTAINMENT.	
SINK.NO	SINK HAS ASBESTOS-CONTAINING UNDERCOATING. LEAVE SINK IN-PLACE.	ISSUED FOR BIDDING
LBP	CEILING, WALLS, AND WOODWORK IN ROOM ARE PAINTED WITH LEAD-BASED PAINT (EXCEPT WHERE SPECIFICALLY NOTED). CONDUCT CEILING, WALL, AND WOODWORK DRILLING AND CUTTING REQUIRED IN SUPPORT OF ELECTRICAL & MECHANICAL DRAWINGS IN COMPLIANCE WITH SECTION 028233.	MARCH 12, 2024
		MARK DATE DESCRIPTION
		PROJECT NO: CWA PROJECT NO.: 2313
		DRAWN BY: CHK'D BY:
		COPYRIGHT
		FIRST FLOOR ABATEMENT PLAN
		H101



-24'

GE	ENERAL NOTES	PROFESSIONAL SEAL
1. The i neces by the comp bindir Contr better Archit Organ	ntent of the Contract Documents is to include all items sary for the proper execution and completion of all Work e Contractor. The Contract Documents are elementary, and what is required by one shall be as ng as if required by all. Where a conflict within the fact Documents exists, the Contractor shall provide the r quality or greater quantity of work in accordance with the tect's resolution without any increase in the Contract sum. hization of the Specifications into divisions, sections and	
article Contr estab 2. REFE	es, and arrangement of sheets shall not control the ractor in dividing the work among subcontractors, or in lishing the extent of Work to be performed by any trade. ERENCE KEY NOTES and SHEET KEY NOTES are ded to indicate and clarify the extent and requirements of	
the W requir Work that V Work	Vork; they do not indicate every location or occurence of red Work. The lack of a key note to an individual item of will not relieve the Contractor of responsibility to execute Vork as part of the Contract when the requirement for that can be logically inferred by other parts of the Contract ments	OWNER
3. Do no appro	ot scale drawings, except for estimation and eximation purposes.	JII
4. This dimer	is an existing building; all conditions and critical nsions must be field verified.	
	EYNOLES DEASTED OF WALL FOR DO WALL FINISH IS	130 Lincoln Street Meriden CT 06451
112.02	PAINTED WITH LEAD-BASED PAINT. EXISTING RADIANT HEATING UNIT BEING REMOVED, NEW UNIT BEING INSTALLED, AND ENTIRE WALL IS BEING PAINTED. CONDUCT EXISTING HEATING UNIT DEMO, NEW HEATING UNIT INSTALL, AND WALL PAINT PREP IN COMPLIANCE WITH SECTION 028233.	203 235 7146 www.girlsincmeriden.org
H2.03	PLASTER OR WALLBOARD WALL FINISH IS PAINTED WITH LEAD-BASED PAINT. NEW FAN CO IL UNIT BEING INSTALLED, AND ENTIRE WALL IS BEING PAINTED. CONDUCT NEW FAN COIL UNIT INSTALL, AND WALL PAINT PREP IN COMPLIANCE WITH SECTION 028233.	
H2.05	PATCHING AND PAINTING OF PLASTER OR WALLBOARD WALL AT DUCT PENETRATION BEING CONDUCTED. WALL IS PAINTED WITH LEAD-BASED PAINT. CONDUCT PATCHING AND PAINT PREP OF 2 SIDES OF WALL IN COMPLIANCE WITH SECTION 028233.	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
H2.06	PATCHING AND PAINTING OF PLASTER OR WALLBOARD WALL AT DUCT PENETRATION BEING CONDUCTED. WALL IS PAINTED WITH LEAD-BASED PAINT. CONDUCT PATCHING AND PAINT PREP OF 1 SIDE OF WALL IN COMPLIANCE WITH SECTION 028233.	MEP ENGINEER
H3.13	NEW FAN COIL UNIT BEING MOUNTED HIGH ON WALL. WALLS ARE PAINTED WITH LEAD-BASED PAINT. CONDUCT FAN COIL INSTALLATION IN COMPLIANCE WITH SECTION 028233.	33 North Plains Industrial Rd Wallingford, CT 06492
H3.15	NEW CEILING-MOUNTED EXHAUST FAN BEING INSTALLED AT THIS LOCATION IMPACTING WALL & CEILING PAINTED WITH LEAD-BASED PAINT, CONDUCT CUTTING & DRILLING IN COMPLIANCE WITH SECTION 028233.	203 467 4370 iesllc.biz
H3.16	NEW CEILING-MOUNTED EXHAUST FAN BEING INSTALLED AT THIS LOCATION IMPACTING WALL & CEILING PAINTED WITH LEAD-BASED PAINT, CONDUCT CUTTING & DRILLING IN COMPLIANCE WITH SECTION 028233.	ENVIRONMENTAL CONSULTANT
LBP	CEILING, WALLS, AND WOODWORK IN ROOM ARE PAINTED WITH LEAD-BASED PAINT (EXCEPT WHERE SPECIFICALLY NOTED). CONDUCT CEILING, WALL, AND WOODWORK DRILLING AND CUTTING REQUIRED IN SUPPORT OF ELECTRICAL & MECHANICAL DRAWINGS IN COMPLIANCE WITH SECTION 028233.	Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
		ISSUED FOR BIDDING MARCH 12, 2024
		MARK DATE DESCRIPTION
		PROJECT NO: CWA PROJECT NO.: DRAWN BY: CHK'D BY: COPYRIGHT
		SHEET TITLE SECOND FLOOR ABATEMENT PLAN
	1	H102

GE	ENERAL NOTES	PROFESSIONAL SEAL
1. The in neces by the compl bindin Contra better	ntent of the Contract Documents is to include all items sary for the proper execution and completion of all Work contractor. The Contract Documents are lementary, and what is required by one shall be as g as if required by all. Where a conflict within the act Documents exists, the Contractor shall provide the quality or greater quantity of work in accordance with the act action without any increase in the O	
Archit Organ article Contra establ	ect's resolution without any increase in the Contract sum. ization of the Specifications into divisions. sections and is, and arrangement of sheets shall not control the actor in dividing the work among subcontractors, or in lishing the extent of Work to be performed by any trade.	
2. REFE intend the W require Work that W Work	RENCE KEY NOTES and SHEET KEY NOTES are led to indicate and clarify the extent and requirements of 'ork; they do not indicate every location or occurence of ed Work. The lack of a key note to an individual item of will not relieve the Contractor of responsibility to execute Vork as part of the Contract when the requirement for that can be logically inferred by other parts of the Contract	OWNER
3. Do no appro	nents ot scale drawings, except for estimation and ximation purposes.	GIIS
dimen		
LBP	CEILING, WALLS, AND WOODWORK IN ROOM ARE PAINTED WITH LEAD-BASED PAINT (EXCEPT WHERE SPECIFICALLY NOTED). CONDUCT CEILING, WALL, AND WOODWORK DRILLING AND CUTTING REQUIRED IN SUPPORT OF ELECTRICAL	130 Lincoln Street 203 235 7146Meriden, CT 06451 www.girlsincmeriden.orgARCHITECT
	A MECHANICAL DRAWINGS IN COMPLIANCE WITH SECTION 028233. CONDUCT DRILLING REQUIRED FOR NEW CEILING SYSTEM GRID INSTALL SHOWN ON DRAWING A113 IN COMPLIANCE WITH SECTION 028233.	
CEILING NO	LEAD-BASED PAINT.	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
		203 776 0184 www.cwarchitectsllc.com MEP ENGINEER
		INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
		33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370 iesllc.biz
		ENVIRONMENTAL CONSULTANT
		EnvironMed Cleaner environment. Safer workplaces.
		DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
		ISSUED FOR BIDDING MARCH 12, 2024
		MARK DATE DESCRIPTION PROJECT NO: 2313
		DRAWN BY: CHK'D BY: COPYRIGHT
		SHEET TITLE THIRD FLOOR ABATEMENT PLAN
	_	H103

 $(\mathbf{+})$

GENERAL NOTES	PROFESSIONAL SEAL
1. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of all Work by the Contractor. The Contract Documents are	
complementary, and what is required by one shall be as binding as if required by all. Where a conflict within the Contract Documents exists, the Contractor shall provide the	
better quality or greater quantity of work in accordance with the Architect's resolution without any increase in the Contract sum.	
articles, and arrangement of sheets shall not control the Contractor in dividing the work among subcontractors, or in establishing the extent of Work to be performed by any trade.	
 REFERENCE KEY NOTES and SHEET KEY NOTES are intended to indicate and clarify the extent and requirements of the Work; they do not indicate every location or occurence of 	
required Work. The lack of a key note to an individual item of Work will not relieve the Contractor of responsibility to execute that Work as part of the Contract when the requirement for that	OWNER
Work can be logically inferred by other parts of the Contract Documents	
Do not scale drawings, except for estimation and approximation purposes.	
 This is an existing building; all conditions and critical dimensions must be field verified. 	
	130 Lincoln Street Meriden CT 06451
BEING INSTALLED. NO ACM OR LEAD-BASED PAINT IMPACT ANTICIPATED AT ROOF.	203 235 7146 www.girlsincmeriden.org
H3.04 CURB MOUNTED ROOFTOP EXHAUST FAN HOOD BEING INSTALLED. NO ACM OR LEAD-BASED PAINT IMPACT ANTICIPATED AT ROOF	ARCHITECT
H3.15 DUCT FROM CEILING EXHAUST FAN AT ROOF. NO ACM OR LEAD-PAINT IMPACT ANTICIPATED AT	
ROOF.	
	85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd
	203 467 4370
	leslic.blz
	ENVIRONMENTAL CONSULTANT
	EnvíroMed
	Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	MARCH 12, 2024
	-
	PROJECT NO:
	CWA PROJECT NO.: 2313 DRAWN BY:
	CHK'D BY:
	ROOF
	ABATEMENT
	PLAN
	H120
1	

3	

3		2
	·	

5

3

2

BASEMENT PLUMBING DEMOLITION NOTES

- (1) CONTRACTOR SHALL DISCONNECT, CAP HW, AND CW PIPING, AND REMOVE EXISTING ELECTRIC WATER HEATER.
- CONTRACTOR SHALL DISCONNECT, AND CAP EXISTING DOMESTIC MAKE-UP CW PIPING TO BOILER.

PROFESSIONAL SEAL
OWNER
130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC
85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
MEP ENGINEER
INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370 iesllc.biz
EnviroMed
Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY
CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO:
CWA PROJECT NO.: 2313 DRAWN BY:
CHK'D BY: COPYRIGHT
BASEMENT PLUMBING
DEMOLITION PLAN
100 חח

3/13/2024 9:12:18 AM

5

3

2

BASEMENT PLUMBING DRAWING NOTES

1

- 1 PROVIDE NEW DOMESTIC GAS HOT WATER HEATER. REFER TO SCHEDULES, AND DETAILS FOR ADDITIONAL INFORMATION.
- 2 CONNECT NEW 3/4" CW PIPING FROM WATER HEATER TO EXISTING CW PIPING. REFER TO HOT WATER HEATER DETAIL FOR ADDITIONAL INFORMATION.
- 3 CONNECT NEW 3/4" HW PIPING FROM WATER HEATER TO EXISTING HW PIPING. REFER TO HOT WATER HEATER DETAIL FOR ADDITIONAL INFORMATION.
- 4 EXISTING 3/4" HW, AND 3/4" CW PIPING. CONTRACTOR SHALL FIELD VERIFY ALL EXISTING DOMESTIC WATER PIPING IN MECHANICAL ROOM, AND REPLACE EXISTING INSULATION WITH NEW INSULATION.
- 5 PROVIDE NEW 1-1/2" NATURAL GAS PIPING FROM METER ABOVE ON EXTERIOR WALL OF THE BUILDING.
- 6 PROVIDE 3" INLET/EXHAUST PIPING FROM GAS WATER HEATER, AND EXTEND UP ABOVE GRADE PER WATER HEATER'S MANUFACTURER'S REQUIREMENTS.
- (7) CONNECT NEW GAS PIPING TO NEW GAS BOILER. COORDINATE WITH HVAC CONTRACTOR FOR ALL REQUIREMENTS.

PROFESSIONAL SEAL
OWNER
girls
130 Lincoln Street Meriden, CT 06451
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
MEP ENGINEER
INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370 iesllc.biz
ENVIRONMENTAL CONSULTANT
Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT
SHEET TITLE
BASEMENT PLUMBING PLAN
P.100
inner_6GB99RY4KMYC.rvt 3/13/2024 9:12:18 AM

5

FIRS	57
1	E\ Bl
2	PL CC
3	PF

2

3

2

T FLOOR PLUMBING DRAWING NOTES

1

EVERSOURCE GAS UTILITY SHALL PROVIDE NEW GAS SERVICE TO THE BUILDING FROM LINCOLN STREET. PLUMBING CONTRACTOR SHALL COORDINATE WITH EVERSOURCE GAS CO. FOR INSTALLATION OF NEW GAS METER BY EVERSOURCE.

PROVIDE 1-1/2" LOW PRESSURE GAS PIPING INTO BASEMENT BELOW.

PROFESSIONAL SEAL
OWNER
130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
203 776 0184 www.cwarchitectsllc.com
MEP ENGINEER
INNOVATIVE ENGINEERING SERVICES, LLC
AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370 iesllc.biz
ENVIRONMENTAL CONSULTANT
EnviroMed
Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO: CWA PROJECT NO.: 2313
DRAWN BY: CHK'D BY:
COPYRIGHT
PLAN
D 101
nner 6GB99RY4KMYC.rvt 3/13/2024 9:12:19 AM

		3				2				1		PROFESSIONAL SEAL
									DI LIMRIN			
										WATER		
										WATER		
-VACUUM BR	EAKER											
(TYPICAL) CHECK VALVE			CONNECT TO 3/4" CW PIPI	D EXISTING ING	NC 2/4"				WAST	E, SANITARY OR SOIL		
(TYPICAL) TCW	<u> </u>		∠ℓ ⟨𝔄_⊂ℓ	CW	NG 3/4		 \$		CD-CD-CONE	DENSATE WASTE		
		1-1,	/4" CW——						IW INDIF	RECT WASTE		OWNER
		3/4"		EXISTING 3/4"					— G — GAS	(NATURAL OR LPG)		
<u></u> -cw	H MH		<u>ż</u> ı	120° HW [<u></u>				VENT		
- 3/4"	-140°	_ @	Ī						EXIS	TING PIPING TO BE REMOVED		
3/4"	HW			3/4" HW PIPING					C PIPE	DOWN		
	UNION (TYPICAL)								PIPE	DROP		
			GAUGE						O PIPE	RISE		130 Lincoln Street Meriden, CT 06451
		ш 	5/4 CW							ANCHOR		203 235 7146 www.girlsincmeriden.org
4	4" TO 6"HIGH CONCRETE									GUIDE or SLEEVE		ARCHITECT
	CHAMFERS FOR WATER HI	ALL BRONZE T	HERMOSTATIC WITH INTEGRAL		FROM GAS			f		EXPANSION FITTING (AT BLDG. EXP.)(DINTS)	
	HEATER MANUFACTURER'S REQUIREMENTS.	CHECKS, SET N REFER TO SCH	/ALVE AT 120°F. EDULE FOR		SUPPLY					ATION ISOLATION FITTING		
F	IN. FLOOR	ADDITIONAL II	NFORMATION	GAS COCK								
	ANSION TANK (ET): < SHALL BE RATED FOR U	SF FOR										
POT	ABLE WATER APPLICATION	IS. REFER TO INFORMATION.	TC		╤ ╻ ┍	FI			ABI	BREVIATIONS		CHRISTOPHER WILLIAMS ARCHITECTS, LLC
EXIS	STING FLOOR DRAIN OR S	UMP KIT	EC				OFFLI					85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
			GROUND JOINT	/		/ 6" TRAP (DRIP LEG)			FP BACKFLOW PI	HED FLOOR REVENTER DEVICE		
			MANUAL SHUT-OFF	F VALVE					CUBIC FEET			MEP ENGINEER
WATER	HEATER PIP	ING DETAIL	WITH AN 1/8" NPT TAPPING ACCESSIE	PLUGGED BLE FOR				C E	TE CONNECT TO XP EXPANSION T	EXISTING		
				STREAM OF				G	V GAS VENT GAS			INNOVATIVE ENGINEERING SERVICES, LLC
			TO THE EQUIPMEN	T.				G' H	WH GAS WATER H HW HOT WATER	IEATER		AN INTEGRATED ENGINEERING + DESIGN FIRM
			ТҮ	PICAL GAS CONNE	CTION TO	EOUIPME	NT DETAIL	HI Hi	B HOSE BIBB C HANDICAP AC	CCESSIBLE		33 North Plains Industrial Rd
			N.T.	S.				IV N	W INDIRECT WA	STE RACT		Wallingford, CT 06492
									&P TEMPERATUR	SQUARE INCH E & PRESSURE RELIEF VALVE		203 467 4370 iesllc.biz
									YP TYPICAL			
								W	/HA WATER HAMM	IER ARRESTOR		ENVIRONMENTAL CONSULTANT
		PLUMBING	EQUIPMENT SCHEI	DULE				PLUMBING W	ATER SPECIALT	IES SCHEDULE		EnvíroMed
								MANUFACTURER/		ACCESSORIES		Cleaner environment. Safer workplaces.
<u>SYMBOL</u>	EQUIPMENT TYPE	MODEL NUMBER	<u>DESCRIPTIO</u>	<u>DN</u>	NOTES	SYMBOL	. FIXTURE TYPE	MODEL NUMBER	DESCRIPTION	AND TRIM	REMARKS	DESIGNED BY: JOHN LUBY
<u>GWH</u>	NATURAL GAS	A.O. SMITH BTX-80 PROV	IDE THE CYCLONE HE LIGHT DUTY, PC	OWER VENT, FULLY CONDENSING 50	123	ET1	EXPANSION	BELL + GOSSETT CO.	EXPANSION TANK:	-	#3	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	WATER HEATER	GALLO RECO	DN GAS WATER HEATER WITH 76,000 VERY @ 90°F RISE OF 95 GALLONS PE	BTU INPUT PER HOUR, AND ER HOUR. ELECTRICAL	1,2,5		TAINK	MODEL #PTA-20V	DIAPHRAGM, STEEL			
		REQU NEUTI	IREMENTS SHALL BE 120 VOLT, 60 HZ RALIZER KIT, MODEL #100112380.	Z, 5 AMPS. PROVIDE CONDENSATE					PRECHARGED 55 PSIG, X			
NOTES:						RPRP1		WATTS MODEL #		SHUT-OFF VALVES ATR GAP	#1245	
1. REFE 2. COO	R TO PLUMBING PLANS, A RDINATE ALL ELECTRICAL	ND DETAILS FOR EXACT LOCA REQUIREMENTS WITH ELECT	TION, AND PIPING SIZE REQUIREME RICAL CONTRACTOR.	NTS.			PRESSURE	UP TO 1" LF909	ASSE 1013 LEAD FREE	FITTING, STRAINER, TEST COCKS	#1,2,4,5	
3. PRO\	/IDE AUTHORIZED EQUIP	MENT REP. FOR STARTUP, AND	OWNER TRAINING OF EQUIPMENT.				PREVENTER	2 1/2" AND OVER LF909	P REDUCED PRESSURE ZONE ASSEMBLY,			ISSUED FOR BIDDING
									PRESSURE DIFFERENTIAL RELIEF VALVE BETWEEN			
									TWO POSITIVE SEATING CHECK VALVES.			
	FIFING					TMV	THERMOPLASTIC	MORRIS GROUP	TEMP CONTROL	SHUT-OFF VALVES	#2	
SIZE	PIPING MATERIALS	JOINTS	FITTINGS	NOTES			MIXING VALVE	INTERNATIONAL CONTROLS MV17-2	THERMOPLASTIC MIXING VALVE, LEAD FREE,	TEMP GAUGE		
	HARD COPPER TUBE:	SOLDER FILLER METALS	WROUGHT-COPPER SOLDER-	PROVIDE 1" INSULATION ON					INTEGRAL STOPS, 5 PSI PRESSURE DROP AT 15 GPM			
1/2" - 2"	ASTM B88, TYPE L WATER TUBE	ASTM B 32, LEAD-FREE ALLOYS	JOINT FITTINGS: ASME B16.22, WROUGHT-COPPER	CW, HW & HWR PIPING IN ALL AREAS		WHA	WATER HAMMER	РРР	WATER HAMMER	-	#4	
			PRESSURE FITTINGS		CW & HW		ARRESTOR	MODEL #SC-500A	ARRESTOR: CORROSION RESISTANT			
1/2" - 2"	ASTM B88, TYPE K, WATER TUBE	GROUND IF POSSIBLE IF REQUIRED FLARED	CAST IRON FLARED FITTINGS	PIPING IN ALL AREAS. PROVIDE PV FOR ALL PIPING UNDER SLAB BE	C SLEEVES ETWEEN		_		BRASS			
	DRAWN TEMPER	JOINTS ARE ACCEPTABLE		WALL AND DENTAL EQUIPM	ENT	REMARK	S:					
ALL	ASTM B 88,	SOLDER JOINTS: ASTM B 813, WATER -	ASME B16.22, WROUGHT- DWV COPPER	_				-UK SIZES.				
	DRAINAGE TOBE, DRAWN TEMPER	END OF TUBE	PRESSURE FITTINGS			2. PROV		VE ON WATER SUPPLY.				
1 / 4 !! !!	SCHEDULE 40 BLACK STEEL		BLACK MALLEABLE-IRON THREADED FITTINGS	ALL PIPING MATERIALS TO B	BE IN	2. PROV	/IDE AIR GAP FITTIN	G AND PIPE DRAIN TO NEAR	REST FLOOR DRAIN			
1/4 - 2	PIPE, ASTM A53	INKEADED	ASME B16.3 150# CLASS STANDARD PATTERN	ACCORDANCE WITH N.F.P.A	. 54	5. MOU	NT MAX OF 5'-0" ABC	OVE FINISHED FLOOR.				MARK DATE DESCRIPTION
2 1 /2" 0	SCHEDULE 40	BUTT WELDED	WROUGHT BLACK	ALL GAS PIPING MATERIALS 2-1/2"	' & LARGER							PROJECT NO:
LARGER	DLACK STEEL PIPE, ASTM A52	OR SOCKET WELDED	STEEL FITTINGS, ASTM A234	SHALL BE WELDED IN ACCORE WITH N.F.P.A. 54, A.G.A & N.I	DANCE B.F.U.							DRAWN BY:
	CCA INI CA			<u> </u>								CHK'D BY:
												COPYRIGHT
												SHEET TITLE
												PLUMBING DETAILS.
												LEGENDS, NOTES, &
												SCHEDULES
												P 200

3/13/2024 9:12:20 AM

PLUMBING SPECIFICATIONS

TECHNICAL REQUIREMENTS

4

THESE SPECIFICATIONS CALL OUT CERTAIN DU NOT INTENDED AS SUBCONTRACT DOCUMENTS BY THE CONTRACT.

PROVIDE ALL ITEMS AND WORK CALLED FOR I CONTRACT DOCUMENTS. THIS INCLUDES ALL SCAFFOLDING, SUPPORTS, TOOLS, SUPERVISI TO PROVIDE COMPLETE SYSTEMS. PERFORM S OPERABLE SYSTEMS.

INTENT OF DRAWINGS

DO NOT SCALE DRAWINGS. CHECK EXISTING CODES AND STANDARDS

INTERNATIONAL BUILDING CODE IBC, 2021 ED AMENDMENTS.

INTERNATIONAL PLUMBING CODE, 2021 EDITIO AMENDMENTS.

NFPA 101, LIFE SAFETY CODE, 2021 EDITION, NFPA 54: NATIONAL FUEL GAS CODE, 2021 EDI AMENDMENTS.

NFPA 70: NATIONAL ELECTRICAL CODE, 2020 E AMENDMENTS.

CONNECTICUT DEPARTMENT OF PUBLIC HEALTH

OCCUPATIONAL SAFETY AND HEALTH STANDAR DEPARTMENT OF ENVIRONMENTAL PROTECTIO

INTERNATIONAL ENERGY CONSERVATION COD AMENDMENTS.

STATE DEMOLITION CODE.

LOCAL BUILDING CODE.

ICC/ANSI A117.1, 2017, ACCESSIBLE AND USAE CONNECTICUT 2022 AMENDMENTS.

COMPLY WITH ALL APPLICABLE GOVERNMENTA INSURANCE UNDERWRITERS AND OTHER APPL THESE SPECIFICATIONS, CODES AND ORDINAM THE SPECIFICATIONS NOR INSTALL ANY WORK THE CONFLICT IS RESOLVED AND THE SOLUTION

<u>SUBMITTALS</u>

PRODUCT DATA: SUBMIT MANUFACTURER'S TEC SELECTED MODEL CLEARLY INDICATED, FURNIS INSTRUCTIONS.

SHOP DRAWINGS: SUBMIT MANUFACTURER'S ROUGHING-IN REQUIREMENTS, REQUIRED CLEA ANCHORAGES.

MAINTENANCE DATA: SUBMIT MAINTENANCE D ACCESSORY; INCLUDING "TROUBLE SHOOTING SHOP DRAWINGS IN MAINTENANCE MANUAL.

PROVIDE SHOP DRAWING SUBMITTALS OF ALL

CERTAIN TERMS SUCH AS "SHALL, PROVIDE, I THESE SPECIFICATIONS. THIS DOES NOT INDI SYSTEMS SHALL BE LESS THAN COMPLETE.

PERMITS AND FEES SECURE AND PAY COSTS OF PERMITS, CERTIFI

ADJUSTMENTS

UPON COMPLETION OF WORK, PERFORM THE FOR ADJUST SYSTEMS COMPONENTS FOR PROPER I OPEN AND CLOSE VALVES, SET IN PROPER OPE

ACCESSIBILITY

PLACE VALVES, UNIONS, DRAINS, AND ITEMS LOCATIONS. COORDINATE ACCESS PANELS WI

REFERENCE PUBLICATIONS AMERICAN SOCIETY FOR TESTING AND MATERI

(ANSI) PUBLICATIONS ARE REFERRED TO HERE DATES FOLLOWING PUBLICATION NUMBERS H/

COORDINATION OF WORK TRANSMIT TO OTHER TRADES ALL INFORMATIC SECTIONS IN AMPLE TIME FOR INSTALLATION.

WHEREVER WORK INTERCONNECTS WITH WORK THAT ALL TRADES HAVE THE INFORMATION NE CONNECTIONS AND EQUIPMENT. IDENTIFY ALL

ORDER THAT THE CEILING SUBCONTRACTOR V CONSULT WITH OTHER TRADES REGARDING E THE SAME MANUFACTURER.

FURNISH AND SET ALL SLEEVES FOR PASSAGE CONCRETE WALL AND FLOORS, AND ELSEWHE PASSING THROUGH BUILDING SURFACES.

PROVIDE REQUIRED SUPPORTS AND HANGERS EXCEED ALLOWABLE LOADINGS OF STRUCTUR

CONFORM THE PLUMBING WORK TO THE REQU ACCESSORIES WHICH MAY BE REQUIRED. INVI WORK, AND ARRANGE THE WORK ACCORDINGL MAY BE REQUIRED TO MEET SUCH CONDITION

ELECTRICAL CONNECTIONS

4

POWER SUPPLY WILL BE PROVIDED UNDER DIV

TO FACILITATE ELECTRICAL CONNECTIONS EQU KNOCKOUTS, CONNECTORS, TERMINAL BLOCKS AND/OR CONTACTS.

	PIPING IDENTIFICATION	PLUMBING MATER
JTIES OF THE CONTRACTOR AND HIS SUBCONTRACTOR. THEY ARE S, NOR ARE THEY INTENDED AS A MATERIAL LIST OF ITEMS REOUIRED	CONSPICUOUSLY IDENTIFY NEW PIPING WITH SELF-ADHERING VINYL PLASTIC COLOR BANDS AND PIPE MARKERS IMPRINTED WITH LEGEND, BASED ON ANSO A13.1 "SCHEME FOR THE IDENTIFICATION OF PIPING SYSTEMS".	DOMESTIC HOT AN
N THIS DIVISION OF THE SPECIFICATIONS IN ACCORDANCE WITH THE INCIDENTALS, EQUIPMENT, APPLIANCES, SERVICES, HOISTING,	APPLY LEGENDS TO FEED AND CROSS MAIN PIPING ADJACENT TO CHANGES IN DIRECTION WHERE PIPES PASS THROUGH WALLS OR FLOORS, AT INTERVALS NOT EXCEEDING 40 FEET IN STRAIGHT PIPING RUNS, AND ADJACENT TO CROSS MAIN CONNECTIONS WITH FEED MAIN.	DOMESTIC WATER FITTINGS SOLDER AND PR FITTINGS
ON LABOR, CONSUMABLE ITEMS, FEES, LICENSES, ETC., NECESSARY TART UP AND CHECK OUT EACH ITEM AND SYSTEM TO PROVIDE FULLY	MINIMUM LETTER SIZE:	NATURAL GAS PIP MALLEABLE - IRON
	1/2" FOR PIPING 3/4" TO 1-1/4" OD 3/4" FOR PIPING 1-1/2" TO 2" OD 1-1/4" FOR PIPING 2-1/2" TO 6" OD	NATURAL GAS PIP SOCKET WELDED 2-1/2" & LARGER S
SPACE CONDITIONS AT THE JOB SITE.	MINIMUM COLOR BAND WIDTH:	APPLIANCE FLEXIE
DITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022	8" FOR PIPING 3/4" TO 2" OD 12" FOR PIPING 2-1/2" TO 6" OD	OUTDOOR, APPLIA
ON, AS AMENDED BY THE STATE OF CONNECTICUT 2022	OPERATING INSTRUCTIONS	<u>VALVES</u> CHECK VALVE: ALI
	INSTRUCT OWNER'S OPERATING PERSONNEL ON PROPER CARE, MAINTENANCE AND OPERATING PROCEDURES.	PSI NON-SHOCK C
ITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.	MAINTENANCE MANUAL INCLUDE FOLLOWING IN MANUALS:	400 PSI WOG, NIB
EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022	MANUFACTURER'S DESCRIPTIVE DATA OPERATION AND MAINTENANCE INSTRUCTIONS	GAS VALVE - 2" AN TRIM OR 1-PIECE I
	REPLACEMENT PART LISTS WIRING DIAGRAMS MANUFACTURER'S WARRANTY & SERVICE CERTIFICATES	GAS VALVE - 2-1/2 HOMESTEAD, McD
RDS.	INSTRUCTIONS FOR PERIODIC CLEANING AND MAINTENANCE PROCEDURES FOR SYSTEMS START-UP AND SHUT-DOWN	<u>PIPE HANGER MAT</u>
Ν.	VALVE LOCATION AND TAG NUMBER CHARTS	BOLTING - ASTM A CAST IRON - ASTM
E, 2021, AS AMENDED BY THE STATE OF CONNECTICUT 2022	<u>CLEANING</u>	FORGOINGS - AST MALLEABLE IRON -
	UPON COMPLETION OF WORK, PERFORM THE FOLLOWING CLEANING PROCEDURES:	STEEL PIPE - ASTM STEEL PIPE - ASTM STEEL ROD - ASTM
	REMOVE PROTECTIVE COVERS AFTER PAINTING	PIPE HANGER COM
ABLE BUILDINGS AND FACILITIES, AS AMENDED BY THE STATE OF	CLEAN PIPING AND EQUIPMENT REMOVE SURPLUS MATERIALS AND RUBBISH	COMPONENTS: DE
	RESTORE DAMAGED SURFACE FINISHES	CORROSION RESIS
ICABLE CODES AND ORDINANCES. IF ANY CONFLICT ARISES BETWEEN	ADJOSTMENTS	PIPE ATTACHMENT
WHICH MAY BE IN CONFLICT WITH CODES AND ORDINANCES UNTIL ON APPROVED BY THE ENGINEER.	ADJUST SYSTEMS COMPONENTS FOR PROPER PERFORMANCE OPEN AND CLOSE VALVES, SET IN PROPER OPERATING POSITION SEAL CONTROL VALVES OPEN	SURFACE FINISH - SURFACE FINISH -
CHNICAL PRODUCT DATA, INCLUDING RATED CAPACITIES OF	GUARANTEE	SURFACE FINISH -
ASSEMBLY TYPE SHOP DRAWINGS INDICATING DIMENSIONS	SUPPLY TWO COPIES OF A WARRANTY COUNTERSIGNED AND GUARANTEED BY CONTRACTOR, STATING THAT IMPERFECT SYSTEM OPERATION AND ALL DEFECTS IN LABOR AND MATERIALS WILL BE REPAIRED WITHOUT COST TO OWNER FOR A PERIOD OF ONE YEAR FROM DATE OF SUBSTANTIAL COMPLETION, AND STATING THAT ALL PLUMBING	ESCUTCHEONS
ARANCES, AND METHODS OF ASSEMBLY OF COMPONENTS AND	EQUIPMENT HAS BEEN FULLY SERVICED AND LEFT IN PROPER OPERATING CONDITION.	ACOUSTICAL SURF
DATA AND PARTS LISTS FOR EACH TYPE OF PLUMBING FIXTURE AND G" MAINTENANCE GUIDE. INCLUDE THIS DATA, PRODUCT DATA AND	PIPE SLEEVE INSTALLATION	PREFINISHED SUR
. PLUMBING EQUIPMENT INCLUDING:	PROVIDE FOR PIPING PASSING THROUGH WALLS, PARTITIONS AND SLAB, SLEEVES SIZED AT LEAST 1 INCH LARGER THAN OD OF PIPE.	PIPE SLEEVES, SE
	SLEEVES ARE REQUIRED FOR PIPING PASSING THROUGH FIRE-RATED WALLS CONSTRUCTED OF METAL STUDS AND GYPSUM WALLBOARD.	PIPE INSULATION
NSTALL, COMPLETE, START-UP" ARE NOT USED IN SOME PARTS OF CATE ITEMS SHALL BE LESS THAN COMPLETELY INSTALLED OR THAT	TERMINATE SLEEVES THROUGH WALLS, PARTITIONS AND CEILINGS FLUSH WITH FINISHED SURFACES: THROUGH SLABS 1/2" ABOVE FLOOR FINISH IN HABITABLE SPACES AND 2" ABOVE ROUGH FINISH IN PIPE SPACES AND OTHER UNFINISHED AREAS.	PIPE: SECTIONAL MINIMUM DENSITY MAXIMUM K-FACTO
	SET SLEEVES IN PLACE BEFORE PLACING CONCRETE, OR SECURELY FASTEN AND GROUT IN PLACE WITH CONCRETE. EXERCISE CARE IN LOCATING AND SETTING OF SLEEVES TO ASSURE ACCURATE ALIGNMENT. IN ABSENCE OF SLEEVES, USE CORE DRILLED HOLES AND PROVIDE CURBS TO PREVENT PASSAGE OF WATER.	VALVES AND FITTI INSULATION, FINI BANDS.
ICATES, LICENSES, INSPECTIONS AND APPROVALS.	FILL VOID SPACES BETWEEN PIPING AND PIPE SLEEVES WITH PENETRATION SEAL, OR APPROVED ELASTROMERIC CAULKING MATERIALS.	SURFACE FINISH - SURFACE FINISH -
OLLOWING ADJUSTMENT PROCEDURES:	ESCUTCHEON INSTALLATION	WATER PIPING SPI
PERFORMANCE. ERATING POSITION.	PROVIDE ESCUTCHEONS ON PIPE PROTRUSIONS AT WALLS, PARTITIONS, CEILING AND FLOORS. ESCUTCHEONS SHALL FIT SNUGLY AROUND PIPING AND COVER SURFACE OPENING.	SHOCK ABSORBER INLET; 150 PSI WV
	FIRE STOPPING	
TH ARCHITECT.	WITH FIBROUS GLASS SEALED WITH FIRE TESTED AND APPROVED ELASTOMERIC CAULKING MATERIALS.	
	INSTALLATION - GENERAL	
IALS (ASTM) AND AMERICAN NATIONAL STANDARDS INSTITUTE EIN, BECAUSE THESE PUBLICATIONS ARE REVISED FREQUENTLY, AVE BEEN OMITTED. REFER TO LATEST EDITION.	PREPARATION: CUT PIPE AND TUBING ENDS SQUARE, REMOVE BURRS AND REAM TO ORIGINAL BORE. CLEAN JOINT SURFACES PRIOR TO ASSEMBLY. WIPE OFF EXCESS JOINING COMPOUNDS AND FLUX RESIDUE.	
ON REQUIRED FOR WORK TO BE PROVIDED UNDER THEIR RESPECTIVE	SCREWED: USE AMERICAN STANDARD TAPER PIPE THREADS CUT SHARP AND TRUE AND SUITABLE FOR NORMAL ENGAGEMENT. SCREW THREADED ITEMS UP CLOSE TO SHOULDERS WITH NOT MORE THAN THREE INCOMPLETE THREADS EXPOSED. DO NOT USE LAMP WICK, CORD, WOOL OR OTHER "WICKING" MATERIALS. REPAIR LEAKS WITH NEW MATERIALS, DO NOT PEEN OR CAULK. "TEFLON" PIPE JOINT TAPE OR JOINT COMPOUNDS COMPOSED OF RED LEAD AND GRAPHITE GROUND IN LINSEED OIL WILL BE PERMITTED. APPLIED TO MALE THREADS ONLY.	
RK OF OTHER TRADES, COORDINATE WITH OTHER TRADES TO INSURE	SOLDER: MAKE UP JOINTS WITH 95-5 TIN-ANTIMONY WIRE SOLDER AND NON-CORROSIVE FLUX. DO NOT USE 50-50	
ECESSARY SO THEY MAY PROPERLY INSTALL ALL THE NECESSARY WORK ITEMS (VALVES, DRAINS, ETC.) IN AN APPROVED MANNER IN VILL KNOW WHERE TO INSTALL ACCESS DOORS AND PANELS.	OR OTHER TIN-LEAD SOLDERS. MECHANICAL COUPLINGS: USE MANUFACTURER'S MATERIALS AND METHODS.	
QUIPMENT SO, WHEREVER POSSIBLE, MOTORS AND CONTROL ARE OF	PIPE HANGER AND SUPPORT INSTALLATION	
OF PIPES AND CONDUITS THROUGH STRUCTURAL MASONRY AND	REFER TO MSS-SP-58; STANDARD FOR PIPE HANGERS AND SUPPORTS.	
EOR PIPING FIXTURES AND FOUR MENT SOL CADING WILL NOT	SUPPORT, ANCHOR AND GUIDE PIPING SYSTEMS TO WITHSTAND STATIC AND DYNAMIC LOAD CONDITIONS, TO ALLOW FOR EXPANSION AND CONTRACTION; TO PREVENT VIBRATION AND SWAYING; TO MAINTAIN ALIGNMENT AND MINIMIZE VERTICAL DEFLECTION.	
IREMENTS HERFIN. PROVIDE OFESETS FITTINGS DRAINS AND	DO NOT SUPPORT PIPING FROM OTHER PIPING OR DUCTWORK. DO NOT USE WIRE, TAPE, METAL BAND, OR OTHER MAKE-SHIFT DEVICES AS MEANS OF SUPPORT OR ATTACHMENT.	
ESTIGATE THE STRUCTURAL AND FINISH CONDITIONS AFFECTING THE LY. PROVIDE SUCH PIPING, FITTINGS, VALVES AND ACCESSORIES AS S.	TESTING GENERAL: TEST PLUMBING SYSTEMS TO SATISFACTION OF BUILDING OFFICIAL. DO NOT CLOSE IN, CONCEAL, OR COVER UP ANY WORK UNTIL IT HAS BEEN TESTED, INSPECTED, AND APPROVED BY ENGINEER AND LOCAL	
/ISION 16, AND CONNECTIONS MADE TO ANY NEW ELECTRICAL ITEMS.	FLUSH PIPING, PRIOR TO TESTING, TO REMOVE FOREIGN MATERIAL WHICH MAY HAVE ENTERED DURING COURSE OF	
UIP ELECTRICAL ITEMS WITH NEMA ENCLOSURES HAVING ADEQUATE	INSTALLATION. ULLAN FILTERS AND STRAINERS AFTER FLUSHING.	

2

<u>TERIALS</u>

FAND COLD WATER: PIPING - HARD DRAWN COPPER TUBING TYPE "L" WITH SOLDERED JOINTS.

ATER PIPING UNDERGROUND - SOFT COPPER TUBE, ASTM B 88 TYPE K OR L AND WROUGHT - COPPER DER-JOINT, ASME B16.18 OR BRONZE FLANGE TYPE FITTINGS ASME B16.22. PE PIPING ASTM D 2239, NGS MOLDED ASTM D 3350.

PIPING 2" & UNDER SHALL BE SCHEDULE 40 BLACK PIPE, ASTM A53 WITH THREADED BLACK IRON FITTINGS ASME B16.3 150# CLASS STANDARD PATTERN.

PIPING 2-1/2" & LARGER SHALL BE SCHEDULE 40 BLACK PIPE, ASTM A53 WITH BUTT WELDED OR ED WROUGHT BLACK STEEL FITTINGS ASTM A234 150# CLASS STANDARD PATTERN. ALL GAS PIPING GER SHALL BE WELDED IN ACCORDANCE WITH N.F.P.A. 54, A.G.A. & N.B.F.U.

EXIBLE CONNECTION - INDOOR, FIXED - APPLIANCE FLEX CONNECTORS SHALL COMPLY WITH ANSI 10VABLE - APPLIANCE FLEX CONNECTORS SHALL COMPLY WITH ANSI Z21.69. PLIANCE FLEX CONNECTORS SHALL COMPLY WITH ASNI Z21.75.

ALL BRONZE, HORIZONTAL SWING, REGRINDING TYPE, Y PATTERN, RENEWABLE SEAT AND DISC, 200 OCK COLD WATER BY NIBCO OR EQUAL.

2" AND SMALLER: LEAD FREE BRONZE BODY AND BALL, TEFLON SEATS AND SEALS, LEVER HANDLE, NIBCO, APOLLO OR EQUAL.

" AND SMALLER: BRONZE PLUG VALVE, TWO-PIECE FULL PORT BRONZE BALL VALVE WITH BRONZE ECE FULL PORT BRONZE BALL VALVE WITH BRONZE TRIM. APOLLO, MILWAUKEE, OR EQUAL.

2-1/2" AND LARGER: FLANGED BRONZE PLUG VALVE OR CAST-IRON, NON-LUBRICATED PLUG VALVE. McDONALD, OR EQUAL.

MATERIAL STANDARDS

TM A307, GRADE A ASTM A48, GRADE 30 AST, A521, GRADE CA RON - ASTM A47, ASTM A197 A36, ASTM A569, ASTM A570 ASTM A53, ASTM A120 ASTM A36, ASTM A575

<u>COMPONENTS</u>

DESIGNED WITH MINIMUM SAFETY FACTOR OF 5; OF ALL METAL CONSTRUCTION; ASSEMBLED WITH RESISTANT SQUARE HEAD MACHINE BOLTS AND SQUARE OR HEX HEAD NUTS, STEEL WASHERS; IN IER AS TO PREVENT SELF-DISENGAGEMENT.

IENTS: CAPABLE OF VERTICAL ADJUSTMENT UNDER LOAD, SHAPED TO OD IF PIPING, SIZED ALLOW INSULATION.

SH - GENERALLY: CORROSION RESISTANT PAINT COATING.

SH - UNINSULATED COPPER AND BRASS PIPING; COPPER-PLATED OR PLASTIC COATED. SH - CHROME PLATED PIPING: CHROME PLATED.

FACES: PRIME COATED SHEET METAL

SURFACES: FACTORY PAINTED TO MATCH SURFACE SHEET STEEL

SURFACES: CHROME-PLATED CAST BRASS

SEALS AND ADAPTERS

TITIONS: #20 GAUGE GALVANIZED STEEL, LOCK SEAM JOINT

NAL PREMOLDED FIBROUS GLASS, FACTORY APPLIED FLAME RETARDANT VAPOR BARRIER JACKET.

ISITY: 3.5 LBS PCF FACTOR: 0.25 AT 75 DEGREES FAHRENHEIT, MEAN.

ITTINGS: PREMOLED, FIBROUS GLASS FITTING COVERS, OR FABRICATED MITERED SEGMENTS OF PIPE FINISHED WITH 20-20 COUNT GLASS CLOT EMBEDDED IN SURFACE COATING. NO STAPLES, TACKS OR

SH - EXPOSED: UNIFORM COAT OF SURFACE COATING SH - CONCEALED: NONE REQUIRED.

SPECIALTIES

BER: SA: ALL COPPER, MECHANICAL-PNEUMATIC TYPE, HERMETICALLY SEALED BELLOWS, THREADED I WWP. SIZE AND PLACEMENT DETERMINATION: PDI-WH 201 AND A.S.S.E. STANDARD 1010.

PROFESSIONAL SEAL
OWNER
130 Lincoln Street Meriden, CT 06451
203 235 7146 www.girlsincmeriden.org
ARCHITECT
85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370 iesllc.biz
ENVIRONMENTAL CONSULTANT
EnviroMod
Cleaner environment. Safer workplaces.
DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
· ·
MARK DATE DESCRIPTION
PROJECT NO: CWA PROJECT NO.: 2313
DRAWN BY:
COPYRIGHT
SHEET TITLE
PLUMBING
P.201

3/13/2024 9:12:20 AM

5

3

2

MECHANICAL DEMOLITION NOTES

1 DISCONNECT, REMOVE, AND PROPERLY DISPOSE OF EXISTING OIL FIRED, CAST IRON, HOT WATER BOILER INCLUDING BUT NOT LIMITED TO ASSOCIATED CONTROLS, BURNER, OIL PIPING, FLUE PIPING, SECTION OF HOT WATER HEATING SUPPLY & RETURN PIPING BACK TO EXISTING ISOLATION VALVES, PRIMARY IN-LINE PUMP, WIRING, SPECIALTIES, OIL FILTER,

2 DISCONNECT, REMOVE, AND PROPERLY DISCARD EXISTING FUEL OIL STORAGE TANKS INCLUDING BUT NOT LIMITED TO ASSOCIATED FUEL & VENT PIPING, MOUNTING LEGS, GAUGES, VALVES, FILTERS, OIL SUPPLY PIPING, ETC. CONTRACTOR SHALL RECLAIM & RECYCLE EXISTING FUEL OIL IN THE TANKS PRIOR TO REMOVAL.

DISCONNECT, REMOVE, AND DISCARD EXISTING VERTICAL AIR HANDLING UNIT INCLUDING BUT NOT LIMITED TO ASSOCIATED DX COOLING COIL, HOT WATER HEATING COIL, REFRIGERATION PIPING, HOT WATER HEATING SUPPLY & RETURN PIPING BACK TO EXISTING ISOLATION VALVES, HYDRONIC SPECIALTIES, CONTROL VALVE, MATCHING REMOTE CONDENSING UNIT, CONDENSATE PIPING, POWER WIRING, FILTERS, CONTROLS, ETC. EXISTING SUPPLY & RETURN DUCTWORK SHALL REMAIN. CONTRACTOR SHALL RECLAIM AND RECYCLE EXISTING REFRIGERANT PRIOR TO REMOVAL.

A REMOVE EXISTING AND INSTALL NEW DUCT INSULATION ON EXISTING DUCTWORK.

 1
PROFESSIONAL SEAL
OWNER
UITS
R R
130 Lincoln Street Meriden CT 06451
203 235 7146 www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
203 / /6 0184 www.cwarchitectsllc.com
MEP ENGINEER
INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370
lesiic.diz
ENVIRONMENTAL CONSULTANT
Enviromed
Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY
CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO:
DRAWN BY:
CHK'D BY:
MECHANICAI
DEMOLITION PLAN
DM.100

1

3/13/2024 9:12:21 AM

5

3

- TO REMOVAL.

- BRACKETS & HARDWARE.

2

MECHANICAL DEMOLITION NOTES

1 DISCONNECT, REMOVE, AND DISCARD EXISTING WALL MOUNTED DUCTLESS FAN COIL UNIT INCLUDING BUT NOT LIMITED TO ASSOCIATED REFRIGERANT PIPING, CONDENSATE PIPING, CONTROLS, POWER WIRING, ETC.

2 DISCONNECT, REMOVE, AND PROPERLY DISCARD EXISTING OUTDOOR CONDENSING UNIT INCLUDING BUT NOT LIMITED TO ASSOCIATED REFRIGERATION PIPING, VALVES, POWER WIRING, MOUNTING PAD, ETC. CONTRACTOR SHALL RECLAIM AND RECYCLE REFRIGERANT IN ACCORDANCE WITH FEDERAL, STATE, AND LOCAL ENVIRONMENTAL REGULATIONS PRIOR

(3) DISCONNECT, REMOVE, AND DISCARD EXISTING TOILET EXHAUST FAN. EXISTING EXHAUST DUCTWORK AND ELECTRICAL POWER SHALL REMAIN FOR RECONNECTION TO NEW FAN.

4 DISCONNECT, REMOVE, AND DISCARD EXISTING SUSPENDED AIR HANDLING UNIT INCLUDING BUT NOT LIMITED TO ASSOCIATED HOT WATER HEATING SUPPLY & RETURN PIPING BACK TO EXISTING ISOLATION VALVES, HYDRONIC SPECIALTIES, CONTROL VALVE, SECTION OF EXISTING SUPPLY & RETURN DUCTWORK, ALL HANGERS & SUPPORTS, ETC.

5 DISCONNECT, REMOVE, AND DISCARD EXISTING FLOOR SUPPLY GRILLE.

(6) DISCONNECT, REMOVE, AND DISCARD EXISTING FLOOR RETURN GRILLE.

DISCONNECT, REMOVE, AND DISCARD EXISTING BASEBOARD RADIATION ENCLOSURE INCLUDING BUT NOT LIMITED TO FITTINGS, END PIECES, 90° ELBOWS, MOUNTING

8 DISCONNECT, REMOVE, AND DISCARD EXISTING RETURN DUCTWORK INCLUDING BUT NOT LIMITED TO ASSOCIATED FITTINGS, ACCESSORIES, HANGERS, SUPPORTS, ETC.

1	PROFESSIONAL SEAL
	OWNER
	girls
	IIIC.®
	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
	ARCHITECT
	CWA
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd
	203 467 4370
	iesllc.biz
	ENVIRONMENTAL CONSULTANT
	EnviroMed
	DESIGNED BY: JOHN LUBY
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
NOTES	
D DUCTLESS FAN COIL UNIT	
PING, CONDENSATE PIPING,	ISSUED FOR BIDDING
DOOR CONDENSING UNIT PIPING, VALVES, POWER ID RECYCLE REFRIGERANT ENTAL REGULATIONS PRIOR	MARCH 12, 2024
ST FAN. EXISTING EXHAUST NECTION TO NEW FAN.	
R HANDLING UNIT	
IING SUPPLY & RETURN IALTIES, CONTROL VALVE, IFRS & SUPPORTS, FTC.	
GRILLE.	
GRILLE.	
ADIATION ENCLOSURE LBOWS, MOUNTING	
UPPORTS, ETC.	MARK DATE DESCRIPTION
	PROJECT NO:
	CWA PROJECT NO.: 2313
	CHK'D BY:
	COPYRIGHT
	SHEET TITLE
	FIRST FLOOR
	MECHANICAL
1	DM.101
C:\Users\cskinner\Documents\RVT2024\Girls Inc. MEP R24_cskinner	

5

3

2

MECHANICAL DEMOLITION NOTES

(1) DISCONNECT, REMOVE, AND PROPERLY DISCARD EXISTING WALL MOUNTED DUCTLESS FAN COIL UNIT INCLUDING BUT NOT LIMITED TO ASSOCIATED REMOTE OUTDOOR CONDENSING UNIT, REFRIGERATION PIPING, VALVES, CONDENSATE PIPING, POWER WIRING, CONTROLS, ETC. CONTRACTOR SHALL RECLAIM & RECYCLE EXISTING REFRIGERANT IN ACCORDANCE WITH FEDERAL, STATE & LOCAL ENVIRONMENTAL REGULATIONS PRIOR TO REMOVAL.

PROFESSIONAL SEAL				
OWNER				
130 Lincoln Street Meriden, CT 06451				
203 235 7146 www.girlsincmeriden.org				
ARCHITECT				
CHRISTOPHER WILLIAMS ARCHITECTS, LLC				
85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com				
MEP ENGINEER				
INNOVATIVE ENGINEERING SERVICES, LLC				
33 North Plains Industrial Rd				
Wallingford, CT 06492				
203 467 4370 iesllc.biz				
ENVIRONMENTAL CONSULTANT	_			
_				
EnviroMed				
Cleaner environment. Safer workplaces				
DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19				
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986				
MARCH 12, 2024				
CWA PROJECT NO.: 2313				
DRAWN BY:				
CHK'D BY:				
SHEET TITLE				
SECOND FLOOR				
DM.102				
<image/>				

4

5

2

MECHANICAL GENERAL NOTES

REFRIGERANT PIPING SHALL BE SIZED BY INSTALLER IN STRICT ACCORDANCE WITH EQUIPMENT MANUFACTURER'S INSTALLATION INSTRUCTIONS. ALL EXPOSED INTERIOR AND EXTERIOR REFRIGERANT PIPING SHALL BE ENCLOSED WITHIN PVC LINE SET SERVICE JACKET COVER KIT REUSE EXISTING REFRIGERANT PIPING WALL PENETRATIONS (WHERE OCCURRING) PROVIDE NEW REFRIGERANT PIPING WALL PENETRATIONS AT NEW FCU LOCATIONS. 4. MINIMIZE DISTURBANCE TO EXISTING ALUMINUM SIDING & WALLS. 5. SEAL ALL REFRIGERANT PIPING PENETRATIONS WEATHERTIGHT.

PR	OFESSIONAL SEAL
OV	VNER
	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
AR	CHITECT
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven. CT 06511
	203 776 0184 www.cwarchitectsllc.com
ME	EP ENGINEER
	INNOVATIVE ENGINEERING
-	AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd
	Wallingford, CT 06492 203 467 4370
i	iesllc.biz
EN	IVIRONMENTAL CONSULTANT
	EnvíroMed
C	Cleaner environment. Safer workplaces.
DCC	ESIGNED BY: JOHN LUBY T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	MARCH 12, 2024
MAI	RK DATE DESCRIPTION
PR CV	OJECT NO: VA PROJECT NO.: 2313
DR CH	AWN BY: IK'D BY:
CC	PYRIGHT
SH	
B M	ASEMENT IECHANICAL PLAN
- •	
	M.100

5

3

2

MECHANICAL GENERAL NOTES

REFRIGERANT PIPING SHALL BE SIZED BY INSTALLER IN STRICT ACCORDANCE WITH EQUIPMENT MANUFACTURER'S INSTALLATION INSTRUCTIONS.

ALL EXPOSED INTERIOR AND EXTERIOR REFRIGERANT PIPING SHALL BE ENCLOSED

3. REUSE EXISTING REFRIGERANT PIPING WALL PENETRATIONS (WHERE OCCURRING)

PROVIDE NEW REFRIGERANT PIPING WALL PENETRATIONS AT NEW FCU LOCATIONS.

5. SEAL ALL REFRIGERANT PIPING PENETRATIONS WEATHERTIGHT.

MECHANICAL DRAWING NOTES

1 PROVIDE NEW BASEBOARD RADIATION ENCLOSURE INCLUDING BUT NOT LIMITED TO FITTINGS, END PIECES, 90° ELBOWS, MOUNTING BRACKETS & HARDWARE. THOROUGHLY CLEAN EXISTING HEATING ELEMENT.

2 PROVIDE NEW FLOOR MOUNTED SUPPLY AIR GRILLE. CONNECT TO EXISTING SUPPLY

3 PROVIDE NEW FLOOR MOUNTED RETURN AIR GRILLE. CONNECT TO EXISTING RETURN AIR DUCTWORK BELOW FLOOR.

	OWNER
	girls ®
	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
	ARCHITECT
	85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
	MEP ENGINEER
	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd
	Wallingford, CT 06492 203 467 4370
	ENVIRONMENTAL CONSULTANT
	Cleaner environment. Safer workplaces. DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ORDANCE	
ULOCATIONS.	MARCH 12, 2024
LIMITED TO E.	
NG SUPPLY	
ING RETURN	MARK DATE DESCRIPTION
	PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY:
	COPYRIGHT
	SHEET TITLE FIRST FLOOR MECHANICAL PLAN

	MECH
	REFRIGERANT PIPING S WITH EQUIPMENT MAN
	ALL EXPOSED INTERIO WITHIN PVC LINE SET
	REUSE EXISTING REFR
	PROVIDE NEW REFRIGE MINIMIZE DISTURBANC
•	SEAL ALL REFRIGERAN

HANICAL GENERAL NOTES

SHALL BE SIZED BY INSTALLER IN STRICT ACCORDANCE NUFACTURER'S INSTALLATION INSTRUCTIONS.

OR AND EXTERIOR REFRIGERANT PIPING SHALL BE ENCLOSED T SERVICE JACKET COVER KIT

FRIGERANT PIPING WALL PENETRATIONS (WHERE OCCURRING) GERANT PIPING WALL PENETRATIONS AT NEW FCU LOCATIONS. NCE TO EXISTING ALUMINUM SIDING & WALLS.

ANT PIPING PENETRATIONS WEATHERTIGHT.

PROFESSIONAL SEAL
OWNER
203 235 7146 www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
203 776 0184 www.cwarchitectsllc.com
MEP ENGINEER
SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
22 North Plaina Industrial Pd
Wallingford, CT 06492
iesllc.biz
ENVIRONMENTAL CONSULTANT
Enviromed
DESIGNED BY: JOHN LUBY
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO: CWA PROJECT NO.: 2313
DRAWN BY:
COPYRIGHT
SHEET TITLE

3/13/2024 9:12:25 AM

	PROFESSIONAL SEAL
	OWNER
	130 Lincoln Street Meriden, CT 06451
	203 235 7 146 www.girlsincmeriden.org
	ARCHITECT
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
	203 776 0184 www.cwarchitectsllc.com
	INNOVATIVE ENGINEERING SERVICES, LLC
	AN INTEGRATED ENGINEERING + DESIGN FIRM
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370 iesllc.biz
	ENVIRONMENTAL CONSULTANT
	EnvíroMed
	Cleaner environment. Safer workplaces.
	DESIGNED BY: JOHN LUBY
	CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	MARCH 12, 2024
	MARK DATE DESCRIPTION
	PROJECT NO:
	CWA PROJECT NO.: 2313
	DRAWN BY: CHK'D BY:
	COPYRIGHT
	SHEET TITLE
	THIRD FLOOR
	MECHANICAL PLAN
	M.103
unner	одвууктикмтс.rvt 3/13/2024 9:12:26 AM

HORIZONTAL AIR HANDLING UNIT DETAIL SCALE: N.T.S.

4

2

	1	PROFESSIONAL SEAL
Μ	IECHANICAL LEGEND	
\sum	DIFFUSER/GRILLE - SUPPLY	
	GRILLE/REGISTER - RETURN	
	GRILLE/REGISTER - EXHAUST	
	SIDEWALL GRILLE/REGISTER - SUPPLY/RETURN/EXHAUST	OWNER
>	AIR FLOW DIRECTION INDICATOR - SUPPLY	
\/►	AIR FLOW DIRECTION INDICATOR - RETURN	UITS
	DUCT RISER - SUPPLY/COMBUSTION AIR	
	DUCT DROP - SUPPLY/COMBUSTION AIR	
	DUCT RISER - RETURN/EXHAUST	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
	DUCT DROP - RETURN/EXHAUST	ARCHITECT
	EXHAUST FAN	
С П	THERMOSTAT	
		85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
	REDUCER	MEP ENGINEER
——нтs	HOT WATER SUPPLY - HEATING	
— — — HTR — — — —	HOT WATER RETURN - HEATING	SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
RI I /RSI	REFRIGERANT PIPING - LIQUID LINE & SUCTION LINE	33 North Plains Industrial Rd
CD	CONDENSATE WASTE	Wallingford, CT 06492 203 467 4370 ieslle biz
	DIRECTION OF FLOW	
c	PIPE DOWN	ENVIRONMENTAL CONSULTANT
	PIPE DROP	
0	PIPE RISE	EnvíroMed
- 	PLUGGED OR CAPPED PIPE	Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY
	EXISTING TO BE REMOVED	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
•••••	CONNECT TO EXISTING	
		ISSUED FOR BIDDING
		MARCH 12, 2024
		MARK DATE DESCRIPTION PROJECT NO:
		CWA PROJECT NO.: 2313 DRAWN BY:
		CHK'D BY: COPYRIGHT
		SHEET TITLE
		MECHANICAL NOTES,
		LEGEND, & DETAILS

3/13/2024 9:12:26 AM

3/13/2024 9:12:27 AM

								AIR	HAND	LING	G UNIT	SCH	EDULE		
GENERAL DATA										COOLI	NG DATA				HEA
SYMBOL	MANUFACTURER / MODEL NUMBER	AREA SERVE D	LOCATION	COOLING (CFM)	HEATING (CFM)	ESP (IN) SI	PEED COC (M	TAL SENS LING COO 3H) (MI	IBLE LING W 3H) (D	EAT /B/DB EG F°)	LAT WB/DB (DEG F°)	AIR P (IN. WG)	D EVAPORAT SUCTIO TEMP (°	FOR HEATI N CAPAC F) (MBł	NG ITY I)
AHU-1	DAIKIN FTQ18TAVJUD	FIRST FLOO OFFICES (SEE PLAN	DR BASEMENT (SEE PLAN)	600	600	0.5" H	IGH 1	8.0 12	.7 77.	2/65.4	55.0/54.0	0.41	45.0	20.(7
PROVIDE R410A RE PROVIDE REFRIGER PROVIDE	WITH DISCONNECT SWIT FRIGERANT EXTRA SET OF MERV-13 ATION PIPING SHALL BE SERVICE ISOLATION VAL	CH FILTER REPLACE SIZED BY INST/ VES.	MENTS. ALLER IN ACCO	RDANCE W	ITH EQUIPME	NT MANUFA	TURER'S IN	TALLATION I	NSTRUCTIC	INS.					
PROVIDE R410A RE PROVIDE REFRIGER PROVIDE	WITH DISCONNECT SWIT FRIGERANT EXTRA SET OF MERV-13 ATION PIPING SHALL BE SERVICE ISOLATION VAL	CH FILTER REPLACE SIZED BY INST VES.	MENTS. ALLER IN ACCO	RDANCE W	ITH EQUIPME	nt Manufa	TURER'S IN	PACK	NSTRUCTIC	AIR	HAND	DLING	UNIT S	CHEDI	JLE
PROVIDE RA10A RE PROVIDE REFRIGER PROVIDE	WITH DISCONNECT SWIT FRIGERANT EXTRA SET OF MERV-13 ATION PIPING SHALL BE SERVICE ISOLATION VAL	CH FILTER REPLACE SIZED BY INST VES.	MENTS. ALLER IN ACCO	RDANCE W	ITH EQUIPME		TURER'S IN	PACK	NSTRUCTIC	AIR	HANE	DLING	UNIT S		JLE
PROVIDE R410A RE PROVIDE REFRIGER PROVIDE	WITH DISCONNECT SWIT FRIGERANT EXTRA SET OF MERV-13 ATION PIPING SHALL BE SERVICE ISOLATION VAL MANUFACTURER MODEL NUMBER	CH FILTER REPLACE SIZED BY INST VES. GI SERVING	MENTS. ALLER IN ACCO ENERAL DATA LOCATION	RDANCE W	ITH EQUIPME	ENT MANUFAG	NOMINA COOLING (TONS)	TALLATION I PACK	NSTRUCTIC AGED SENSIB COOLIN (MBH)	AIR	DX COIL E DX COIL E EAT /B/DB DEG F)	DLING DATA LAT WB/DB (DEG F)	UNIT S SUCTION TEMP (DEG F°)	FACE VELOCITY (FPM)	

NOTES: 1. PROVIDE NEMA 3R DISCONNECT SWITCH

2. PROVIDE DUCT SMOKE DETECTORS IN THE SUPPLY & RETURN AIR STREAMS 3. PROVIDE FACTORY CONTROLS, PROGRAMMABLE THERMOSTAT, LOW VOLTAGE WIRING, DEVICES, ETC. FOR FULLY OPERATIONAL SYSTEM

4. PROVIDE WITH SIDE DUCT DISCHARGE. 5. PROVIDE WITH ECONOMIZER DAMPERS, WEATHER HOODS, BAROMETRIC RELIEF

6. PROVIDE WITH EXTRA SET OF MERV-8 REPLACEMENT FILTERS.

	BOILER SCHEDULE															
			I-B-R		MAX. PSI	BURNER DATA							COMPLICATION	THEDMAL	WEIGHT	
SYMBOL	MODEL NUMBER	LOCATION	NET OUTPUT MBH	BOILER HP	WORKING (PSIG)	GAS INPUT (MBH)	GAS OUTPUT (MBH)	VOLT	PHASE	AMPS	EWT/LWT	DIAMETER	EFFICIENCY	EFFICIENCY	(LBS)	NOTES
B-1	THERMAL SOLUTIONS AMP-800	MECHANICAL 003	776.0	23.2	160	800	776	120	1	8	160/180	6"	97%	97%	560	1,2,3,4,5,6,7
B-2	THERMAL SOLUTIONS AMP-800	MECHANICAL 003	776.0	23.2	160	800	776	120	1	8	160/180	6"	97%	97%	560	1,2,3,4,5,6,7

4

4

NOTES: 1. PROVIDE DISCONNECT SWITCH.

2. PROVIDE ALL SAFETY CONTROLS (T&P RELIEF VALVE, LWCO, HIGH LIMIT, ETC.).

3. PROVIDE WITH CONDENSATE NEUTRALIZATION KIT.

4. PROVIDE WITH FACTORY CONTROLLER, SENSORS, LOW VOLTAGE WIRING FOR FULLY OPERATIONAL SYSTEM. 5. PROVIDE COMBUSTION AIR INTAKE AND VENT PIPING IN ACCORDANCE WITH MANUFACTURER'S INSTALLATION INSTRUCTIONS.

6. PROVIDE WITH FACTORY OUTDOOR RESET CONTROL WITH DOMESTIC HOT WATER PRIORITY.

7. PROVIDE GAS REGULATOR TO REDUCE GAS PRESSURE TO 7" W.C. AT BOILERS.	
--	--

						FAN	SCHED	ULE						
SYMBOL	MANUFACTURER MODEL NUMBER	LOCATION	SERVING	TYPE	AIR FLOW CFM	ESP IN. WS.	FAN RPM	DRIVE	CONTROL	BHP (W)	HP (W)	ELECTRICAL V-PH-RPM	SONES (LWA)	NOTES
EF-1	LOREN COOK GC-420	TOILETS/ SHOWERS 114	TOILETS/ SHOWERS 114	BATHROOM CEILING	200	0.4	1145	DIRECT	WALL SWITCH	0.08	1/10 HP	120-1-1725	2.9	2,3,4
EF-2	LOREN COOK ACED90C17DEC	ROOF	TOIL 119/ WC 119A	ROOF DOWNBLAST	200	0.375	1322	DIRECT	WALL SWITCH	0.23	1/4 HP	120-1-1725	5.0	1,5,6
EF-3	LOREN COOK GC-142	TOILET 204	TOILET 204	BATHROOM CEILING	100	0.25	1000	DIRECT	WALL SWITCH	0.08	1/10 HP	120-1-1725	1.9	2,3,4
EF-4	LOREN COOK GC-142	TOILET 204	TOILET 204	BATHROOM CEILING	100	0.25	1000	DIRECT	WALL SWITCH	0.08	1/10 HP	120-1-1725	1.9	2,3,4
EF-5	LOREN COOK GC-142	TOILET 209	TOILET 209	BATHROOM CEILING	100	0.25	1000	DIRECT	WALL SWITCH	0.08	1/10 HP	120-1-1725	1.9	2,3,4
EF-6	LOREN COOK GN-420	STORAGE 304	THIRD FLOOR	INLINE	200	0.25	1360	DIRECT	RUNS CONTINOUSLY	77	1/10 HP	120-1-1725	2.0	1,2,3,4
NOTES														

NOTES: 1. PROVIDE DISCONNECT SWITCH.

2. PROVIDE FACTORY WALL CAP (LOREN COOK WCR6 OR WCG-6).

PROVIDE FACTORY HANGING ISOLATOR KIT.
 PROVIDE FLEX CONNECTION AT INLET & OUTLET.

5. PROVIDE FACTORY ROOF CURB WITH SELF-ACTING BACKDRAFT DAMPER.

6. PROVIDE ECM MOTOR.

PUMP SCHEDULE													
SYMBOL	MANUFACTURER MODEL NUMBER	TYPE	LOCATION	SYSTEM SERVED	WATER/ GLYCOL	CAPCITY GPM	CAPACITY FT OF HEAD	WATER TEMP (F°)	FLANGE S/D	ELECTRICAL DATA	POWER HP/BHP	MOTOR SPEED	NOTES
P-1	ARMSTRONG 4380-0205-001.5	INLINE	MECHANICAL 003	BOILER B-1	WATER	80.0	35 FT	180° F	2"	208V-3PH	1.5	2,993	1,2,3
P-2	ARMSTRONG 4380-0205-001.5	INLINE	MECHANICAL 003	BOILER B-2	WATER	80.0	35 FT	180° F	2"	208V-3PH	1.5	2,993	1,2,3
NOTES:													

1. PROVIDE SENSORLESS CONTROL WITH INTEGRAL VFD. 2. INTERLOCK WITH BOILER CONTROLLER TO ENABLE AND CONTROL PUMP SPEED.

3. PROVIDE DIGITAL DISPLAY WITH FLOW RATE READOUT.

Δ

				LOUVE	R SCH	EDULE				
SYMBOL	MANUFACTURER MODEL NUMBER	SYSTEM SERVED	AIRFLOW (CFM)	MAXIMUM VELOCITY (FPM)	AIR P.D. (IN. WG)	NOMINAL WIDTH (IN.)	NOMINAL HEIGHT (IN.)	MINIMUM FREE AREA (SQ. FT.)	MAXIMUM WATER PENETRATION (OZ./SQ. FT.)	NOTES
L-1	RUSKIN ELF375DX	AHU-1	150	750	0.05"	12"	12"	3.21	0.01 @ 873 FPM	1,2,3
NOTES: 1. ALUMINUM CONTRUCTION. 2. COLOR/FINISH SHALL BE SELECTED BY OWNER. 3. WEATHER-RESISTANT, DRAINABLE LOUVER.										

	F	
_	-	

ULL			
	F	EATING DAT	Ą
EVAPORATOR SUCTION TEMP (°F)	HEATING CAPACITY (MBH)	EAT/LAT (DEG F°)	E

HEATING CAPACITY (MBH)	EAT/LAT (DEG F°)	EFFICIENCY	CONTROL
20.0	70.0/90.0	2" THICK MERV-13	24 VAC

FILTER DATA

2" PLEATED MERV-8 18"x24"x2"

EFFICIENCY SIZE

FILTER

TYPE

BHP

2.76

ELECTRICAL DATA	١
VOLT-PH-MCA	МОСР
208-1-4.9	15 A

ELECTRICAL DATA

PHASE

VOLT

208

NOTES 1,2,3,4

MCA/MOCP

67.0/80A

NOTES

1,2,3,4,5,6

SYMBOL	MANUFACTURER MODEL NUMBER	SYSTEM SERVED	TYPE	AIR FLOW (CFM)	TOTAL CAPACITY (MBH)	EAT DEG F° DB	LAT DEG F° DB	WATER FLOW GPM	WATER P.D. FT. H ₂ O	EWT/LWT DEG F°	AIR P.D. IN. H ₂ O	NOTES
HWC-1	CARRIER SFI COILS HWC-600	AHU-1	HW	600	28.7	50.0	94.2	2.5	0.5	180/160	0.22	1,2,3
HWC-2	CARRIER SFI COILS HWC-6000	AHU-2	HW	6,000	329.8	45.0	95.8	32.5	4.4	180/160	0.22	1,2,3

HW = HOT WATERNOTES:

3. -

1. PROVIDE WITH 5/8" COPPER TUBES AND BRANDED ALUMINUM FINS.

3. PIPE IN ACCORDANCE WITH DETAIL ON M-2.01

FIN TUBE RADIATION SCHEDULE

SYMBOL	MANUFACTURER MODEL NUMBER	HOT WATER CAPACITY BTU/HR/LF	EWT (DEG F)	CONNECTION SIZE	ROWS OF ELEMENT	ELEMENT LENGTHS	ENCLOSURE HT	NOTES
RAD-A	RITTLING MODEL ETO 3/4"C, 4-1/4 x 4-1/4 - 48	1,030	180	3/4"	1	SEE PLANS	10-5/8"	1,2,3
NOTES: 1 2								

AIR COOLED CONDENSING UNIT SCHEDULE

SYMBOL	MANUFACTURER MODEL NUMBER	SYSTEM SERVED	LOCATION	COOLING CAPACITY TOTAL MBH	SUCTION TEMP (DEG F°)	AMBIENT AIR TEMP DB/WB (DEG F°)	ELECTRICAL MCA-VOLT-PH	ELECTRICAL MOCP	NOTES
CU-1	DAIKIN RZQ18TBVJUA	AHU-1	OUTDOORS	18.0	45° F	91.0/77.0	16.5-208-1	20A	1
NOTES:									

1. PROVIDE DISCONNECT SWITCH 2. PROVIDE 4" THICK REINFORCED CONCRETE PAD.

3. COOLING AND HEATING RATINGS AT 95°F (SUMMER) AND -4°F (WINTER). 4. R410A REFRIGERANT

SYMBOL	MANUFACTURER	עדווח	TVDE	BORDER	С	NOTES		
STMBUL	MODEL NUMBER	DOTT	IIFL	TYPE	OBD	FRAME	BLADES	NOTES
S1	TITUS S300RL	SUPPLY	D.D.	DUCT MOUNTED	STEEL	STEEL	STEEL	2,3,4,7
S2	TITUS CT-541 SERIES	SUPPLY	L.F.	FLOOR	STEEL	STEEL	STEEL	2,4,5,7
R1	TITUS CT-540 SERIES	RETURN	L.F.	FLOOR	STEEL	STEEL	STEEL	2,4,5,7
E1	KREUGER S80	EXHAUST	L.F.	CEILING	-	ALUMINUM	ALUMINUM	4,5,7

TYPES:

NOTES:

1. SQUARE TO ROUND TRANSITION. 2. OPPOSED BLADE DAMPER.

3. DOUBLE DEFLECTION. 4. FACE VELOCITY 600 FPM MAX.

5. 45° DEFLECTION (FIXED). 6. 45° DEFLECTION (FULLY ADJUSTABLE).

7. NC LEVEL NOT TO EXCEED 25.

8. INSULATED PLENUM BOOT. 9. HINGED CORE WITH 1" DISPOSABLE FILTER.

HOT WATER COIL SCHEDULE

2. 2-WAY MODULATING CONTROL VALVE SHALL BE SUPPLIED BY OWNER'S ATC VENDOR, INSTALLED BY M.C.

L.F. - LOUVERED FACE D.D. - DIRECTIONAL DIFFUSER

 $\mathbf{\mathbf{X}}$ TYPICAL RDG

3/13/2024 9:12:28 AM

4

ULE				
G DATA	(NOT IN	ELECTRICAL CLUDING ELECTRI	IC HEAT)	
SENS CLG FULL LOAD MBH	GENERAL V/HZ/PH	MIN CIRC AMPS	MAX FUSE SIZE	NOTES
6.75	208/60/1	0.4 A	15 A	1,2,3
7.95	208/60/1	0.4 A	15 A	1,2,3
6.75	208/60/1	0.4 A	15 A	1,2,3
7.95	208/60/1	0.4 A	15 A	1,2,3
7.95	208/60/1	0.6 A	15 A	1,2,3

1

FCU-A - 0.75 TON OFFICE 212

CU-5 VRF PIPING DIAGRAM

SCALE: N.T.S.

PROFESSIONAL SEAL
OWNER
UITS
R R
130 Lincoln Street Meriden CT 06451
203 235 7146 www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS
85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
MEP ENGINEER
INNOVATIVE ENGINEERING SERVICES, LLC
AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd
Wallingford, CT 06492
iesllc.biz
EnviroMed
Cleaner environment. Safer workplaces.
CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12 2024
MARK DATE DESCRIPTION
PROJECT NO:
CWA PROJECT NO.: 2313
CHK'D BY:
COPYRIGHT
SHEET TITLE
DIAGRAMS
M.203

3/13/2024 9:12:29 AM

	5	4 3	2	
	MECHANICAL (HVAC) SPECIFICATIONS			
	GENERAL			
	<u>SCOPE</u> THE GENERAL SCOPE OF THE HVAC WORK IS TO REMOVE EXISTING SYSTEMS, MODIFY THE EXISTING SYSTEMS, AND PROVIDE NEW SYSTEMS AS INDICATED ON	ALL ELECTRICAL WORK CUTTING AND PATCHING	SIZES AND APPROXIMATE LOCATION OF ALL DUCTS ARE SHOWN ON THE DRAWINGS. CHECK CAREFULLY WITH THE ARCHITECTURAL DRAWINGS, DRAWINGS SHOWING WORK OF OTHER TRADES. AND EXISTING FIELD CONDITIONS TO MAKE SURE THAT THERE WILL BE NO CONFLICT BETWEEN THESE TRADES AND THE	SEAL ALL DUCTWORK IN
	THESE DOCUMENTS.	LINTELS AND STRUCTURAL FRAMING ALL CONCRETE WORK ALL PAINTING	DUCTS. DUCTS SHALL BE OFFSET AS REQUIRED TO CLEAR STRUCTURAL MEMBERS AND EXISTING FIELD CONDITIONS; IF NECESSARY, THE DIMENSIONS OF THE DUCT MAY BE ALTERED PROVIDED THE CROSS-SECTIONAL AREA IS IN NO CASE REDUCED.	COMMERCIAL DUCTS, SI RESIDENTIAL DUCTS, VI
	ALL MATERIALS NOT SPECIFICALLY MENTIONED BUT NECESSARY OR REQUIRED TO PROVIDE COMPLETE AND FULLY OPERATIONAL HVAC SYSTEMS. IT IS THE INTENTION OF THE SPECIFICATIONS AND DRAWINGS TO CALL FOR FINISHED WORK, TESTED AND READY FOR OPERATION.	THIS CONTRACTOR SHALL FURNISH THE GENERAL CONTRACTOR WITH THE SIZES AND LOCATIONS OF CHASES AND OPENINGS WHICH OCCUR IN WALLS, PARTITIONS, FLOORS, ROOFS, ETC., REQUIRED FOR THE INSTALLATION OF THE WORK CALLED FOR UNDER THIS CONTRACT. THIS WORK WILL BE DONE BY THE	FIELD QUALITY CONTROL PERFORM THE FOLLOWING FIELD TESTS AND INSPECTIONS ACCORDING TO SMACNA'S "HVAC AIR DUCT LEAKAGE TEST MANUAL" AND PREPARE TEST REPORTS:	AIR HANDLER AND ALL I
D	IT IS THE INTENT THAT ALL MECHANICAL WORK AND MATERIALS NECESSARY TO COMPLETE THE ENTIRE PROJECT IN ACCORDANCE WITH THE CONTRACT PLANS AND SPECIFICATIONS, WHETHER SPECIFICALLY MENTIONED HERE OR NOT, SHALL BE FURNISHED. ALL WORK AND MATERIALS NECESSARY TO FULFILL THIS INTENT SHALL BE SUPPLIED UNDER THE MECHANICAL SPECIFICATIONS WITHOUT ADDITIONAL COST TO THE OWNER.	GENERAL CONTRACTOR, EXCEPT CUTTING REQUIRED FOR THE INSTALLATION OF HANGERS.	DISASSEMBLE, REASSEMBLE AND SEAL SEGMENTS OF SYSTEMS TO ACCOMMODATE LEAKAGE TESTING AND FOR COMPLIANCE WITH TEST REQUIREMENTS.	EXTERIOR DUCTWORK S MEMBRANE APPLIED AN
	DEFINITIONS	PRIOR TO DELIVERY TO THE JOB SITE, BUT SUFFICIENTLY IN ADVANCE OF REQUIREMENTS NECESSARY TO ALLOW ENGINEER AMPLE TIME FOR REVIEW, CONTRACTOR SHALL SUBMIT FOR APPROVAL, FIVE (5) COPIES OF EACH SHOP DRAWING.	CONDUCT TESTS AT STATIC PRESSURES EQUAL TO MAXIMUM DESIGN PRESSURE OF SYSTEM OR SECTION BEING TESTED. IF PRESSURE CLASSES ARE NOT INDICATED, TEST ENTIRE SYSTEM AT MAXIMUM SYSTEM DESIGN PRESSURE. DO NOT PRESSURIZE SYSTEMS ABOVE MAXIMUM DESIGN OPERATING PRESSURE. GIVE SEVEN DAYS ADVANCE NOTICE FOR TESTING.	DUCT CLEANING - EXIS
	' <u>FURNISH' OR 'PROVIDE'</u> - TO FURNISH, ERECT, INSTALL AND CONNECT UP COMPLETE AND READY FOR OPERATION PARTICULAR WORK REFERRED TO, UNLESS SPECIFICALLY INDICATED OR SPECIFIED OTHERWISE.	INDICATE ON EACH SUBMISSION:	MAXIMUM ALLOWABLE LEAKAGE: COMPLY WITH REQUIREMENTS FOR LEAKAGE CLASS 3 FOR ROUND AND FLAT-OVAL DUCTS, LEAKAGE CLASS 12 FOR RECTANGULAR DUCTS IN PRESSURE CLASSES LOWER THAN AND EQUAL TO 2-INCH WG (500 PA) (BOTH POSITIVE AND NEGATIVE PRESSURES), AND LEAKAGE	COMPLY WITH NADCA A ACCORDING TO OWNER SYSTEM. COMPLETE CLE
	WORK' - LABOR, MATERIALS, EQUIPMENT, APPARATUS, CONTROLS, ACCESSORIES AND ALL OTHER ITEMS CUSTOMARILY FURNISHED AND/OR REQUIRED FOR PROPER AND COMPLETE INSTALLATION OF WORK.	PROJECT NAME AND LOCATION ARCHITECT AND ENGINEER ITEM IDENTIFICATION	CLASS 6 FOR PRESSURE CLASSES FROM 2- TO 10- WG (500 TO 2500 PA). REMAKE LEAKING JOINTS AND RETEST UNTIL LEAKAGE IS EQUAL TO OR LESS THAN MAXIMUM ALLOWABLE.	ALL AIR-MOVING AND D STATE, AND LOCAL REQ
	<u>'EXPOSED'</u> - NOT INSTALLED UNDERGROUND OR 'CONCEALED' AS DEFINED ABOVE. <u>'INDICATE' OR 'SHOWN'</u> - AS INDICATED OR SHOWN ON DRAWINGS OR SPECIFIED WITH SPECIFICATIONS.	 APPROVAL STAMP OF PRIME CONTRACTOR ALL DUCTWORK SHOP DRAWINGS AND COORDINATION DRAWINGS SHALL BE SUBMITTED ON 3/8 IN SCALE DRAWINGS AND SHALL INCLUDE LOCATIONS AND SIZES OF EXISTING FOURMENT ALONG WITH NEW WORK, DRAWINGS SHALL INDICATE LOCATIONS OF HANGERS, SUBDORES, SU	MATERIALS	VOLUME DAMPERS
	' <u>PIPING'</u> - PIPE, FITTINGS, FLANGES, VALVES, CONTROLS, HANGERS, TRAPS, DRAINS, INSULATION AND ITEMS CUSTOMARILY OR REQUIRED IN CONNECTION WITH OR RELATING TO SUCH PIPING.	SIZES OF EXISTING EQUIPMENT ALONG WITH NEW WORK. DRAWINGS SHALL INDICATE LOCATIONS OF HANGERS, SUPPORTS, EXPANSION JOINTS, GUIDES, ANCHORS AND ANCHOR LOADS.	DISSIMILAR METALS WHENEVER DISSIMILAR PIPING MATERIALS ARE CONNECTED THE TWO SHALL BE SEPARATED WITH AN 'INSULATION' CONNECTION (DIELECTRIC) FITTING.	SINGLE BLADE OR OPPO OTHER OPERATOR FOR E
	<u>'SUPPLY'</u> - TO PURCHASE, PRODUCE, ACQUIRE AND DELIVER COMPLETE WITH ALL RELATED ITEMS.	NOTED.	PIPING STEAM & CONDENSATE RETURN PIPING	SMOKE AND/OR FIRE DA
	'NOTED' - AS INDICATED ON DRAWINGS AND/OR SPECIFIED.	DUCTWORK LAYOUT, SHEET METAL DETAILS/STANDARDS COORDINATION DRAWINGS	STEAP & CONDENSATE RETORNET IT ING STANDARD WEIGHT, SCHEDULE 40, OPEN HEARTH STEEL, NATIONAL OR EQUAL. FITTINGS FOR STEEL PIPE SHALL BE AS FOLLOWS: GENERALLY, BUTT WELDING FITTINGS OVER TWO INCHES SHALL BE USED AND FITHER SOCKET-WELD OR SCREWED FOR TWO INCHES AND UNDER. WELDING FITTINGS SHALL BE	PROVIDE SMOKE AND/O PLANS FOR DESIGNATIO
	CODES, RULES, PERMITS AND FEES THIS CONTRACTOR SHALL GIVE ALL NECESSARY NOTICES, OBTAIN ALL PERMITS AND PAY ALL STATE AND LOCAL TAXES, FEES AND OTHER COSTS IN	SUBMITTALS	STANDARD FORGED STEEL WITH CHAMFERED ENDS. ALL BRANCHES SHALL BE WELDED WITH EITHER WELDOLETE OR TEES.	INSTALL IN ALL RATED
	CONNECTION WITH HIS WORK; FILE ALL NECESSARY PLANS, PREPARE ALL DOCUMENTS AND OBTAIN ALL NECESSARY APPROVALS OF ALL STATE AND LOCAL DEPARTMENTS HAVING JURISDICTION; OBTAIN ALL REQUIRED CERTIFICATES OF INSPECTION FOR HIS WORK AND DELIVERY OF SAME TO THE OWNER BEFORE REQUEST FOR ACCEPTANCE AND FINAL PAYMENT FOR THE WORK.	PRIOR TO DELIVERY TO THE JOB SITE, BUT SUFFICIENTLY IN ADVANCE OF REQUIREMENTS NECESSARY TO ALLOW ENGINEER AMPLE TIME FOR REVIEW, CONTRACTOR SHALL SUBMIT FOR APPROVAL, FIVE (5) COPIES OF EACH SHOP DRAWING.	TYPE L COPPER TUBING WITH SWEAT FITTINGS WITH 95-5 SOLDER OR STANDARD WEIGHT, SCHEDULE 40, OPEN HEARTH STEEL, NATIONAL OR EQUAL. FITTINGS FOR STEEL PIPE SHALL BE AS FOLLOWS: GENERALLY, BUTT WELDING FITTINGS OVER TWO INCHES SHALL BE USED AND EITHER SOCKET-WELD OR	PROVIDE ACCESS DOOR EACH FIRE DAMPER OR
	THIS CONTRACTOR SHALL INCLUDE IN THE WORK, WITHOUT EXTRA COST TO THE OWNER, ANY LABOR, MATERIALS, SERVICES, APPARATUS, DRAWINGS (IN ADDITION TO CONTRACT DRAWINGS AND DOCUMENTS), IN ORDER TO COMPLY WITH ALL APPLICABLE LAWS, ORDINANCES, RULES AND REGULATIONS	INDICATE ON EACH SUBMISSION: 1. PROJECT NAME AND LOCATION	SCREWED FOR TWO INCHES AND UNDER. WELDING FITTINGS SHALL BE STANDARD FORGED STEEL WITH CHAMFERED ENDS. ALL BRANCHES SHALL BE WELDED WITH EITHER WELDOLETE OR TEES, OR MATCH EXISTING MATERIALS.	WALL, INSULATED. REF
	WHETHER OR NOT SHOWN ON THE DRAWINGS AND/OR SPECIFIED. THIS CONTRACTOR SHALL PERFORM AND FILE ALL TESTS IN ACCORDANCE WITH THE CURRENT REGULATIONS OF THE STATE AND LOCAL AUTHORITIES. HE	 ARCHITECT AND ENGINEER ITEM IDENTIFICATION APPROVAL STAMP OF PRIME CONTRACTOR 	<u>CHILLED WATER PIPING</u> TYPE L COPPER TUBING WITH SWEAT FITTINGS WITH 95-5 SOLDER OR STANDARD WEIGHT, SCHEDULE 40, OPEN HEARTH STEEL, NATIONAL OR EQUAL.	MOTORIZED DAMPERS
	SHALL FURNISH AND INSTALL SIGNS REQUIRED BY THE STATE AND LOCAL AUTHORITIES. ALL MATERIALS FURNISHED AND ALL WORK INSTALLED SHALL COMPLY WITH THE RULES AND RECOMMENDATIONS OF THE NATIONAL BOARD OF FIRE	SUBMIT SUBMITTALS ON THE FOLLOWING:	FITTINGS FOR STEEL PIPE SHALL BE AS FOLLOWS: GENERALLY, BUTT WELDING FITTINGS OVER TWO INCHES SHALL BE USED AND EITHER SOCKET-WELD OR SCREWED FOR TWO INCHES AND UNDER. WELDING FITTINGS SHALL BE STANDARD FORGED STEEL WITH CHAMFERED ENDS. ALL BRANCHES SHALL BE WELDED WITH EITHER WELDOLETE OR TEES, OR MATCH EXISTING MATERIALS.	PROVIDE MOTORIZED S
	UNDERWRITERS, WITH ALL REQUIREMENTS OF LOCAL UTILITIES COMPANIES, WITH THE RECOMMENDATIONS OF THE FIRE INSURANCE RATING ORGANIZATION HAVING JURISDICTION.	 PIPING MATERIALS PIPING SPECIALTIES PIPING INSULATIONS 	CONDENSER WATER PIPING	(THE SPEC WRITER IS T
	<u>CODES, REGULATIONS AND STANDARDS</u> ALL MECHANICAL WORK SHALL BE PERFORMED IN ACCORDANCE WITH THE FOLLOWING APPROVED CODES, REGULATIONS AND STANDARDS:	 DUCT MATERIALS DUCTWORK SPECIALTIES DUCTWORK INSULATORS 	TYPE L COPPER TUBING WITH SWEAT FITTINGS WITH 95-5 SOLDER OR PVC (EXCEPT IF LOCATED IN A SPACE USED AS AN AIR PLENUM) OR MATCH EXISTING MATERIALS.	<u>CONTROLS</u> <u>GENERAL</u>
	IBC - INTERNATIONAL BUILDING CODE, 2021 EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.	7. AIR OUTLETS (RGD) 8. HEATING EQUIPMENT 9. AIR CONDITIONING EQUIPMENT	<u>CONDENSATION DRAIN PIPING</u> TYPE DWV COPPER TUBING WITH DWV SWEAT FITTINGS OR PVC (EXCEPT IF LOCATED IN A SPACE USED AS AN AIR PLENUM)	PROVIDE ALL TEMPERAT TO ACHIEVE A FULLY OP
С	IMC - INTERNATIONAL MECHANICAL CODE, 2021 EDITION. WITH 2022 CONNECTICUT AMENDMENTS. IFC - INTERNATIONAL FIRE CODE, 2021 EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.	10. FANS 11. CONTROLS 12. HYDRONIC SYSTEMS BALANCING REPORTS 13. AUX OVERTIME DAY ANOTHER DEPORTS		PROVIDE A PROGRAMMA COOL) DEADBAND (5°F
	2022 CONNECTICUT STATE FIRE SAFETY CODE	13. AIR SYSTEMS BALANCING REPORTS EQUIPMENT DEVIATION	TYPE ACR SERVICE COPPER TUBING MEETING ASTM B280: HARD DRAWN (ANY SIZE) OR SOFT DRAWN (1-5/8" ID OR SMALLER), OR AS PER MANUFACTURER'S RECOMMENDATIONS.	TESTING AND BALANCIN
	LOCAL BUILDING CODE	THE PLANS AND/OR SPECIFICATIONS INDICATE THE NAME, MODEL NUMBER OR TYPE OF EQUIPMENT OR MATERIALS SPECIFIED TO SET THE STANDARD OF THE EQUIPMENT FOR THE PROJECT. THE ENGINEER WILL ENTERTAIN THE USE OF OTHER MANUFACTURER'S EQUIPMENT OF LIKE FUNCTIONS AND EQUAL QUALITY.	PIPE INSULATION THE FOLLOWING PIPING SYSTEMS ARE TO BE INSULATED:	GENERAL COMPLETELY TEST AND
	NFPA - NATIONAL FIRE PROTECTION CODE	AND ACCEPTANCE OF SUBSTITUTES IS AT THE ENGINEER'S DISCRETION. SHOULD THE BIDDER DESIRE TO USE EQUIPMENT OR MATERIALS OR A MANUFACTURER OTHER THAN THOSE SPECIFIED OR SHOWN, HE SHALL ATTACH A RIDER TO THE BID FORM LISTING THE DEDUCTIONS AND/OR ADDITIONS TO HIS BASE BID, TOGETHER WITH THE MANUFACTURE'S NAME AND MODEL NUMBERS OF THE EQUIPMENT OR MATERIALS HE PROPOSED TO FURNISH AS CONSTITUTES' TO NO SUBSTITUTE INFORMATION TO EXPLOSE TO THE ADDITIONS TO HE EXPRESSION TO THE ADDITIONS TO HER TO ADDITIONS TO HIS BASE BID, TOGETHER WITH THE MANUFACTURE'S NAME AND MODEL NUMBERS OF THE EQUIPMENT OR MATERIALS HE PROPOSED TO FURNISH AS	STEAM SUPPLY PIPING STEAM CONDENSATE RETURN PIPING WITHIN 10'-0" OF THE FLOOR	GENERAL PIPE TEST
	NFPA 72 - NATIONAL FIRE ALARM CODE, 2019 EDITION	FURNISHED IN FULL ACCORDANCE WITH THE PLANS AND/OR SPECIFICATIONS.	CHILLED WATER SUPPLY AND RETURN PIPING CONDENSATION DRAIN PIPING DEEDIGEPANT SUCTION I INE DIDING	UNLESS OTHERWISE NO DEFECTIVE MATERIAL SI REPRESENTATIVE OF TH
	NFPA 99 - HEALTH CARE FACILITIES CODE, 2021 EDITION	CONTRACTOR SHALL KEEP ACCURATE RECORD OF ALL DEVIATIONS IN WORK AS ACTUALLY INSTALLED FROM WORK INDICATED PAYING PARTICULAR ATTENTION TO DIMENSIONING QUISIDE UNDERGROUND UTUITY LINES. THEIR OFFSETS AND VALVES.	HOT WATER & STEAM HEATING AND CHILLED WATER PIPING INSULATION	AND PLANS. UNLESS OT OF PIPE UNDER INSULAT PRIOR TO TESTING SHA
	IECC - INTERNATIONAL ENERGY CONSERVATION CODE, 2021, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.	AT THE CLOSE-OUT OF THE PROJECT THE CONTRACTOR IS TO DELIVER TO THE OWNER TWO SETS OF "AS-BUILT" DRAWINGS COPIES OF ALL APPROVED SHOP DRAWINGS	INSULATE WITH RIGID PREFORMED FIBERGLASS WITH AP-T PLUS JACKET, SCHULLER MICRO-LOK OR EQUAL. INSULATION THICKNESS SHALL BE 1" THICK FOR BELOW 1 ½" OR SMALLER PIPING, 1-1/2" THICK FOR 2" TO 3" PIPING AND 2" THICK FOR PIPING 4" AND LARGER. PROVIDE ZESTON COVERS ON ALL FITTINGS.	TESTS ARE SATISFACTO
	ICC/ANSI A117.1, 2017, ACCESSIBLE AND USABLE BUILDINGS AND FACILITIES, AS AMENDED BY THE STATE OF CONNECTICUT 2018 AMENDMENTS.	OWNER'S INSTRUCTIONS AND SYSTEM OPERATION	CONDENSATE DRAIN PIPING INSULATION INSULATE WITH 1/2" THICK HEAVY DENSITY FIBERGLASS 25 ASJ WITH VAPOR BARRIER AND LAP ADHESIVE JACKET. INSULATION ON FITTINGS SHALL BE	AUTHORITY. WICKING, OUNTIL SYSTEMS ARE PRO
	ASTM - AMERICAN SOCIETY FOR TESTING AND MATERIALS	THE CONTRACTOR IS TO INSTRUCT THE OWNER, OR HIS REPRESENTATIVE, ON THE OPERATION AND MAINTENANCE PROCEDURES FOR ALL OF THE INSTALLED SYSTEMS AND EQUIPMENT. IN ADDITION TO THE VERBAL INSTRUCTIONS, THESE INSTRUCTIONS SHALL BE WRITTEN IN LAYMAN'S LANGUAGE AND SHALL BE INSERTED IN VINYL-COVERED THREE-RING LOOSE LEAF BINDER. THIS INFORMATION IN BINDER SHALL BE FIRST SENT TO AND APPROVED BY THE	FIBERGLASS WITH PRE-MOLDED JACKET.	HOT OR CHILLED WATER
	OSHA - OCCUPATIONAL SAFETY AND HEALTH ADMINISTRATION U.L UNDERWRITERS LABORATORIES	ARCHITECT/ENGINEER BEFORE TURNING OVER TO OWNER. INSTALLATIONS	INSULATE SUCTION LINE AND LIQUID WITH ARMAFLEX INSULATION. MINIMUM INSULATION THICKNESS SHALL BE 1".	TEST WATER PIPING BY
	EPA - ENVIRONMENTAL PROTECTION AGENCY	SLEEVES	VALVES AND SPECIALTIES BALANCING FITTINGS	FOLLOWING PRECAUTIO
	COMPLY WITH REQUIREMENTS OF LOCAL AUTHORITY HAVING JURISDICTION.	PROVIDE NO. 22 GA. GALVANIZED IRON SLEEVES EXTENDED THROUGH CONSTRUCTION AT ALL PENETRATIONS THROUGH CEILINGS, WALLS AND PARTITIONS. FOR INSULATED PIPING THE SLEEVE IS TO BE SIZED TO ALLOW INSULATION TO PASS THROUGH SLEEVE, PROVIDE 1/2 INCH SPACE BETWEEN PIPE AND/OR	PROVIDE "B & G" CIRCUIT SETTER BALANCING FITTINGS ON ALL WATER SYSTEMS WHENEVER REQUIRED FOR BALANCING OF SYSTEMS.	 SYSTEM PRESSURE WATER CONTROL V
	THE MECHANICAL CONTRACTOR SHALL FURNISH STATUTORY COMPENSATION INSURANCE CERTIFICATES FOR PERSONAL AND PROPERTY DAMAGE DISABILITY/LIABILITY AS REQUIRED BY THE OWNER AND/OR AS HEREINBEFORE DESCRIBED.	INSULATION AND SLEEVE. FIRE SEAL ALL SLEEVES IN ACCORDANCE WITH BUILDING CODE AND APPLICABLE SECTIONS OF THE NFPA.	HOT AND CHILLED WATER VALVES BALL TYPE VALVES TO BE JAMESBURY, CLINCHER, OR APOLLO GATE TYPE VALVES TO BE MILWAUKEE #F-2885M (FLANGED) OS&Y TYPE VALVES TO BE IRON	REFRIGERANT PIPE TEST TEST FOR LEAKS BY FIL
	GUARANTEE AND SERVICE	EXPANSION ANCHORS	BODY, BRONZE MOUNTED OR (SCREWED), BRONZE, RISING STEM. CHECK VALVES TO BE CRANE/JENKINS VALVES.	PLUGGED. USE SEPARAT MAXIMUM TEST PRESSU
	THE CONTRACTOR SHALL GUARANTEE ALL MATERIALS AND WORKMANSHIP FOR A PERIOD OF ONE YEAR FROM THE DATE OF ACCEPTANCE OF THE INSTALLATION. IN ADDITION, THE CONTRACTOR SHALL PROVIDE, FREE OF CHARGE, ONE YEAR'S MAINTENANCE GUARANTEE ON MAINTAINED SERVICE AND ADJUSTMENT OF ALL EQUIPMENT IN THIS CONTRACT.	SUSPEND HANGERS FROM EXPANSION ANCHORS IN SOLID CONCRETE SLABS SIMILAR TO HILTI HDI. PROVIDE HANGER IN PLACE WITH DOUBLE NUTS. PROVIDE PROTECTION SHIELDS IN INSULATED PIPING. INSTALL HANGERS OVER INSULATION AND SHIELDS.	SHALL BE TRERICE UNIVERSAL ANGLE TYPE #L80732, SOLID LIQUID FILLED, 4 ½" DIAL SIZE. FURNISH WITH SEPARABLE SOCKET WITH 2" EXTENSION NECK.	A. REFRIGERANT HIG B. REFRIGERANT LOW
	ALL COMPRESSORS TO HAVE (5) FIVE YEAR EXTENDED WARRANTEES.	WHERE OVERHEAD CONSTRUCTION DOES NOT PERMIT FASTENING HANGER RODS IN REQUIRED LOCATIONS, PROVIDE ADDITIONAL STEEL FRAMING AS REQUIRED AND REVIEWED.	STEAM SPECIALTIES ALL STEAM TRAPS AND SPECIALTIES TO BE B & G OR SARCO.	APPLY SOAPY WATER MI EQUIPMENT FOR LEAKAG
В	DRAWINGS AND INTENT	HANGERS AND SUPPORTING		AIR SYSTEMS BALANCIN
	ACCESSIONES, SECTALITES, DUCINGER, AND FIFTING STSTEMS ARE DIAGRAMMATIC UNLESS SECTIONALLY DIMENSIONED, AND DO NOT NECESSIONED, AND DO NOT NECESSIONED	SIZE OF PIPE AND PROPER STRENGTH AND QUALITY AT PROPER INTERVALS SO THAT THE PIPING CANNOT BE MOVED ACCIDENTALLY FROM THE INSTALLED POSITION AS FOLLOWS:	ALL DUCTWORK SHALL BE CONSTRUCTED OF #1 QUALITY SHEETS OF GALVANIZED STEEL FREE OF CRACKS OR BLEMISHES. WHEN PITTSBURGING OR SNAP	PROCURE THE SERVICES
	FIELD, ALL EXISTING CONDITIONS.	PROVIDE CLEVIS HANGERS AT CENTER OF CENTER SPACING (UNLESS OTHERWISE NOTED)	CONSTRUCTION, HANGING METHOD, AND INSTALLED IN STRICT ACCORDANCE WITH THE CURRENT EDITION OF THE SMACNA STANDARDS, INCLUDING DUCT LEAKAGE REQUIREMENTS.	COMPONENTS. BALANCE REPORT INDICATING TH
	MISUNDERSTANDING OF MATERIAL TO BE FURNISHED, OR WORK TO BE DONE; IT BEING THAT TENDER OF PROPOSAL INDICATED WITH ITS AGREEMENT TO ITEMS AND CONDITIONS REFERRED TO HEREIN OR INDICATED ON AFOREMENTIONED DRAWINGS.	1/2 INCH PIPE OR TUBING 6 FEET 3/4 INCH OR 1 INCH PIPE OR TUBING 8 FEET 1-1/4 INCH OR LARGER (HORIZONTAL) 10 FEET	FLEXIBLE DUCTWORK FLEXIBLE DUCTS TO BE INSULATED TYPE: UL 181, CLASS 1, 2-PLY VINYL FILM SUPPORTED BY HELICALLY WOUND, SPRING-STEEL WIRE WITH FIBROUS-GLASS	WATER SYSTEMS BALAN
	MEASUREMENTS ALL MEASUREMENTS TAKEN AT THE BUILDING SHALL TAKE PRECEDENCE OVER SCALE DIMENSIONS. EVERY PART OF THE PLANS SHALL BE FITTED TO THE	1-1/4 INCH OR LARGER (VERTICAL) EVERY FLOOR LEVEL DUCT HANGING AND SUPPORTING - DUCTWORK SHALL NOT BE SUPPORTED BY OTHER DUCTWORK OR PIPING, BUT SHALL BE SUPPORTED WITH HANGERS OF	INSULATION AND POLYETHYLENE VAPOR BARRIER FILM. THE LENGTH OF FLEXIBLE DUCT IS NOT TO EXCEED 12'-0". FLEXIBLE DUCT MAY ONLY BE USED ON THE SUPPLY AIR SIDE OF LOW PRESSURE DUCT SYSTEMS.	COMPLETELY TEST AND EOUIPMENT (PUMP, COI
	ACTUAL CONDITIONS AT THE BUILDING. IF IN CONFLICT WITH SCALE DIMENSIONS, CONTACT ARCHITECT FOR CLARIFICATION.	TYPE AND AT SPACING AS PER SMACNA STANDARDS. VIBRATION AND SEISMIC CONTROL	MOISTURE LADEN AIR DUCTWORK DUCTWORK WHICH CONVEYS AIR WITH A HIGH MOISTURE CONTENT, SUCH AS SHOWER EXHAUST AIR, SHALL BE CONSTRUCTED OF ALUMINUM SHEET	REPORT. SUBMIT THE R
	THE HVAC CONTRACTOR IS TO COORDINATE WITH THE GENERAL CONTRACTOR, PRIOR TO PERFORMING WORK REQUIRING INTERRUPTION OF EXISTING SERVICES, THE CONTRACTOR SHALL SECURE FROM THE OWNER, APPROVAL OF THE PROPOSED OPERATION.	<u>QUIET OPERATION</u> - ALL WORK SHALL OPERATE UNDER ALL CONDITIONS OF LOAD WITHOUT ANY SOUND OR VIBRATION WHICH IS OBJECTIONABLE IN THE OPINION OF THE ENGINEER. IN CASE OF MOVING MACHINERY, SOUND OR VIBRATION NOTICEABLE OUTSIDE OF ROOM IN WHICH IT IS INSTALLED, OR	MATERIAL MEETING SMACNA STANDARDS.	
	WORK SHALL BE ARRANGED FOR CONTINUOUS PERFORMANCE WHENEVER POSSIBLE. THE MECHANICAL CONTRACTOR SHALL PROVIDE TEMPORARY SERVICES AND/OR CONNECTIONS WHERE REQUIRED AND/OR SCHEDULE AND PERFORM OVERTIME WORK FOR ANY OPERATION WHICH REQUIRED SHUTDOWN OF THE	ANNOYING INSIDE ITS OWN ROOM, WILL BE CONSIDERED OBJECTIONABLE BY THE ENGINEER AND SHALL BE REMEDIED IN APPROVED MANNER BY THE CONTRACTOR AT HIS EXPENSE.	SOUND (ACOUSTIC) INSULATION	
	FACILITIES AT NO ADDITIONAL COST TO THE OWNER. THE AREA OF CONSTRUCTION AND/OR ADJACENT SPACES MAY BE OCCUPIED DURING THE CONSTRUCTION PERIOD. THE CONTRACTOR IS TO TAKE ALL	PROVIDE FLEXIBLE PIPE CONNECTIONS AT ALL PIPING CONNECTED TO MOVING EQUIPMENT. PROVIDE FLEXIBLE DUCT CONNECTIONS AT ALL DUCTWORK CONNECTED TO MOVING EQUIPMENT. FLEXIBLE CONNECTIONS SHALL BE 29 OZ. NEOPRENE	PROVIDE INTERNAL SOUND INSULATION IN ALL DUCTS WITHIN 10'-0" OF THE DISCHARGE OF AN AIR HANDLING UNIT AND WHERE INDICATED ON THE DRAWINGS. THE DUCTWORK SHALL BE LINED WITH JOHNS MANVILLE PERMACOTE LINACOUSTIC. THICKNESS, UNLESS SPECIFIED OTHERWISE, SHALL BE 1". LINER SHALL BE APPLIED TO DUCT IN STRICT ACCORDANCE WITH MANUFACTURER'S INSTRUCTIONS AND SMACNA GUIDELINES, LATEST EDITION WHERE	
	NECESSARY MEASURES AND PROVIDE ALL MATERIALS TO ENSURE A SAFE ENVIRONMENT FOR THE FACILITY'S OCCUPANTS.	COATED FIBERGLASS, 6" WIDE. BURNING PROPERTIES SHALL CONFORM TO NFPA 90A. FASTEN TO DUCTWORK PER MANUFACTURER'S RECOMMENDATIONS. FABRIC SHALL NOT BE STRESSED OTHER THAN BY AIR PRESSURE. ALLOW AT LEAST ONE INCH SLACK TO INSURE THAT NO VIBRATION IS TRANSMITTED.	SOUND INSULATION IS INDICATED, DUCTWORK SIZES DENOTED ARE THE INSIDE DIMENSIONS AFTER THE INSULATION HAS BEEN INSTALLED.	
	WHEREVER AN EXISTING SYSTEM IS REMOVED, PARTIALLY REMOVED, OR MODIFIED THE REMAINING SYSTEM IS TO FUNCTION FULLY AS BEFORE.	PROVIDE VIBRATION ISOLATION SPRINGS OR PADS AT MOUNTING AND SUPPORTS FOR ALL EQUIPMENT CAPABLE OF TRANSMITTING VIBRATIONS. <u>SEISMIC RESTRAINTS</u>	COVER ALL CONCEALED UNLINED SUPPLY AIR AND OUTSIDE AIR DUCTWORK WITH FIBERGLASS DUCT WRAP HAVING A MIN. R-6, EQUAL TO JOHNS MANVILLE R- SERIES MICROLITE WITH F.R.G. VAPOR BARRIER. ALL SUPPLY DUCTS, LOCATED IN ATTIC SHALL BE INSULATED TO MINIMUM R-8. COVER ALL EXPOSED UNLINED	
	MAINTAIN CONTINUITY OF THE EXISTING AIR SYSTEMS, HYDRONIC SYSTEMS, AND CONTROL SYSTEMS TO THE AREAS NOT AFFECTED BY THIS ALTERATION. SCAFFOLDING, RIGGING AND HOISTING	SEISMIC RESTRAINTS DESIGNED AND CONSTRUCTED FOR LATERAL FORCES IN ANY DIRECTION SHALL BE PROVIDED FOR ALL MECHANICAL EQUIPMENT IN ACCORDANCE WITH THE STATE BUILDING CODE.	SUPPLY AIR AND OUTSIDE AIR DUCTWORK WITH RIGID FIBERGLASS BOARD INSULATION HAVING MIN. R-6. PROVIDE ALL TAPE, MASTICS, SEALANTS, MOUNTING PINS, AND ETC. TO INSTALL INSULATION AS RECOMMENDED BY THE MANUFACTURER.	
	UNLESS OTHERWISE SPECIFIED, CONTRACTOR SHALL FURNISH ALL SCAFFOLDING, RIGGING, HOISTING AND SERVICES NECESSARY FOR ERECTION AND DELIVERY INTO THE PREMISES OF ANY EQUIPMENT AND APPARATUS FURNISHED.	SEISMIC RESTRAINTS SHALL NOT BE REQUIRED FOR THE FOLLOWING:	THERMAL INSULATION SCHEDULE INSULATE DUCTS IN COMPLIANCE WITH THE INTERNATIONAL ENERGY CONSERVATION CODE. COMMERCIAL DUCTWORK SHALL BE INSULATED TO R-6 WHEN IN	
	THE CONTRACTOR SHALL REMOVE SAME FROM PREMISES WHEN NO LONGER REQUIRED.	 PIPING IN BOILER AND MECHANICAL ROOMS LESS THAN 1-1/4 INCH INSIDE DIAMETER. ALL OTHER PIPING LESS THAN 2-1/2 INCH INSIDE DIAMETER. RECTANGULAR AIR-HANDLING DUCTS LESS THAN 6 SQUARE FEET IN CROSS-SECTIONAL AREA. 	UNCONDITIONED SPACES AND R-8 WHEN LOCATED OUTSIDE THE BUILDING. COMMERCIAL DUCTWORK IN CONDITIONED SPACES DOES NOT REQUIRE INSULATION. RESIDENTIAL DUCTS OUTSIDE THE BUILDING ENVELOPE SHALL BE INSULATED TO A MINIMUM OF R-8. RESIDENTIAL DUCTWORK INSIDE THE BUILDINGS THERMAL ENVELOPE DOES NOT REQUIRE INSULATION. ALL EXTERIOR DUCTS TO BE INSULATED TO A MINIMUM OR R-8.	
	HOUSEKEEPING THIS CONTRACTOR SHALL BE RESPONSIBLE FOR KEEPING STOCK OF MATERIALS AND EQUIPMENT STORED ON PREMISES, AT LOCATIONS DESIGNATED FOR	 ROUND AIR-HANDLING DUCTS LESS THAN 28 INCHES IN DIAMETER. PIPING SUSPENDED BY INDIVIDUAL HANGERS 12 INCHES OR LESS IN LENGTH FROM THE TOP OF THE PIPE TO THE BOTTOM OF THE SUPPORT FOR THE HANGER. PURTE CURPENDED BY HANGERS 12 INCHES OR LESS IN LENGTH FROM THE TOP OF THE PIPE TO THE BOTTOM OF THE SUPPORT FOR THE HANGER. 		
	THIS CONTRACTOR SHALL AT ALL TIMES, KEEP THE PREMISES FREE FROM ACCUMULATION OF WASTE MATERIAL OR RUBBISH CAUSED BY HIS EMPLOYEES AT	SEISMIC RESTRAINT FOR DUCTWORK; PROVIDE REQUIRED BRACING MATERIAL. DUCTWORK SHALL BE SUPPORTED AND BRACED TO RESIST ALL DIRECTIONAL		
A	HIS WORK IN A CLEAN AND ORDERLY CONDITION.	IDENTIFICATION		
	PROTECTION OF MATERIALS AND EQUIPMENTS	ALL IDENTIFICATION LABELING IS TO COMPLY WITH ASME A13.1		
	CLOSE PIPE OPENINGS WITH CAPS OR PLUGS DURING INSTALLATION.	ALL DUCTWORK IS TO BE LABELED WITH INDICATIONS OF SERVICE, DIRECTION OF FLOW.		
	TIGHTLY COVER AND PROTECT FIXTURES AND EQUIPMENT AGAINST DIRT, WATER AND CHEMICAL OR MECHANICAL INJURY.	ALL EQUIPMENT IS TO HAVE PERMANENT LABELS INDICATING EQUIPMENT DESIGNATION.		
	AT COMPLETION OF ALL WORK, FIXTURES, EXPOSED MATERIALS AND EQUIPMENT SHALL BE THOROUGHLY CLEANED.	SIZES AND APPROXIMATE LOCATION OF PIPING SYSTEMS ARE SHOWN ON THE DRAWINGS. CHECK CAREFULLY WITHTHE ARCHITECTURAL DRAWINGS, DRAWINGS SHOWING WORK OF OTHER TRADES. AND EXISTING FIFI D CONDITIONS TO MAKE SUBE THAT THERE WILL BE NO CONFLICT BETWEEN THESE		
		TRADES AND THE PIPING SYSTEMS. PIPES SHALL BE OFFSET AS REQUIRED TO CLEAR STRUCTURAL MEMBERS AND EXISTING FIELD CONDITIONS.		
		PIPING SYSTEM. INSTALL PIPING TO ALLOW FOR PIPE EXPANSION.		

5

3

2

	PROFESSIONAL SEAL
<u>ICT SEALING</u> AL ALL DUCTWORK IN COMPLIANCE WITH THE INTERNATIONAL ENERGY CONSERVATION CODE.	
MMERCIAL DUCTS, SEAL ALL LONGITUDINAL AND TRANSVERSE JOINTS, SEAMS AND CONNECTIONS.	
ICT INSULATION - EXTERIOR DUCTWORK	
TERIOR DUCTWORK SHOULD FIRST BE WRAPPED WITH 1-1/2" THICK RIGID FIBERGLASS INSULATION, COVERED WITH #907 MASTIC, WEATHERPROOF MBRANE APPLIED AND WRAPPED FOR THE ENTIRE LENGTH, THEN RE-APPLY MASTIC AS FINAL SEALING AGENT.	OWNER
JCT CLEANING - EXISTING DUCTWORK	
CCORDING TO OWNER'S STANDARDS OR AUTHORITIES HAVING JURISDICTION. REMOVE NON-ADHERED SUBSTANCES AND DEPOSITS FROM WITHIN THE HVAC STEM. COMPLETE CLEANING IN ACCORDANCE WITH OWNER-CONTRACTOR AGREED-UPON SCOPE OF WORK. SYSTEMS AND COMPONENTS TO BE CLEANED: L AIR-MOVING AND DISTRIBUTION EQUIPMENT. DEBRIS REMOVED FROM THE HVAC SYSTEM SHALL BE DISPOSED OF ACCORDING TO APPLICABLE FEDERAL,	
ATE, AND LOCAL REQUIREMENTS.	
DUME DAMPERS	
HER OPERATOR FOR EXTERNALLY INSULATED DUCT SHALL HAVE STAND-OFF MOUNT SO OPERATION IS CLEAR OF THE INSULATION. PROVIDE VOLUME MPER IN DUCTWORK AT ALL RUN-OUT DUCT TO EACH CEILING DIFFUSER, AT ALL BRANCH DUCTS AND WHERE INDICATED.	
AOKE AND/OR FIRE DAMPERS ROVIDE SMOKE AND/OR FIRE DAMPERS AS REQUIRED, WHETHER INDICATED OR NOT, AT ALL FIRE AND SMOKE RATED PARTITIONS. REVIEW ARCHITECTURAL	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
ANS FOR DESIGNATIONS. FIRE DAMPERS SHALL BE RUSKIN IBD 2, VERTICAL OR HORIZONTAL, STYLE B OR STYLE C FOR ROUND DUCTS, OR EQUAL. EACH FALL BE FURNISHED AND INSTALLED IN ACCORDANCE WITH NFPA 90A LATEST EDITION AND BEAR U.L. LABEL AND SHALL CONFORM TO BULLETIN #UL-555. ISTALL IN ALL RATED WALLS AND CEILINGS AS REQUIRED AND/OR INDICATED ON DRAWINGS.	ARCHITECT
J <u>CT ACCESS DOORS</u> OVIDE ACCESS DOORS, SIZED AND LOCATED FOR MAINTENANCE WORK, UPSTREAM WHERE POSSIBLE, FOR EACH DUCT MOUNTED SMOKE DETECTOR AND	
CH FIRE DAMPER OR DEVICE WITHIN THE DUCT THAT REQUIRES SERVICE OR INSPECTION. ACCESS SECTIONS IN INSULATED DUCTS SHALL BE DOUBLE- ALL, INSULATED. REFER TO SMACNA STANDARDS. PROVIDE LOCK TYPE 2 (DOOR LATCH, NOT SASH LOCK).	
TORIZED DAMPERS	
DVIDE MOTORIZED SHUTOFF DAMPERS ON EXHAUST AND OUTDOOR AIR SUPPLY OPENINGS IN COMPLIANCE WITH THE IECC.	
IE SPEC WRITER IS TO EDIT THIS SECTIONS TO INCLUDE THE EQUIPMENT PERTAINING TO THE PROJECT)	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
NERAL	203 776 0184 www.cwarchitectsllc.com
ROVIDE ALL TEMPERATURE, OPERATION AND SAFETY CONTROLS, LOW VOLTAGE CONTROL WIRING, HARDWARE, SOFTWARE, AND ACCESSORIES NECESSARY D ACHIEVE A FULLY OPERATIONAL HVAC CONTROL SYSTEM. POWER WIRING SHALL BE PROVIDED BY THE ELECTRICAL CONTRACTOR.	MEP ENGINEER
OVIDE A PROGRAMMABLE THERMOSTAT(S) IN COMPLIANCE WITH THE INTERNATIONAL ENERGY CONSERVATION CODE INCLUDING SETBACK (55°F HEAT, 85° DOL) DEADBAND (5°F COMMERCIAL) AND TIMECLOCK (7 DAY COMMERCIAL)	·
INERAL	INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
MPLETELY TEST AND BALANCE HOT AND CHILLED WATER SYSTEMS AND ALL SUPPLY, RETURN AND EXHAUST AIR SYSTEMS AND PROVE THE CAPACITIES OF E SYSTEM AND THE SYSTEM COMPONENTS. SUBMIT RESULTS TO ENGINEER FOR APPROVAL	
NERAL PIPE TEST LESS OTHERWISE NOTED, TEST ALL PIPING HYDROSTATICALLY AT NOT LESS THAN 200 PSIG (# PER SQUARE INCH PRESSURE) FOR TWO HOURS AND ALL	33 North Plains Industrial Rd Wallingford, CT 06492
FECTIVE MATERIAL SHALL BE REPLACED. BEFORE MAKING FINAL APPROVAL, THE SUBCONTRACTOR SHOULD PRODUCE A WRITTEN STATEMENT, SIGNED BY A PRESENTATIVE OF THE OWNER'S UNDERWRITER, THAT THE WORK HAS BEEN COMPLETED AND TESTED IN ACCORDANCE WITH APPROVED SPECIFICATIONS D PLANS. UNLESS OTHERWISE NOTED, PERFORM PRESSURE TESTS AND OBTAIN APPROVAL OF TEST RESULTS BEFORE STARTING CLEANING OR CONCEALING PIDE LINDER INSULATION OR OTHER FINISH. INSULATION REMOVAL AND REINSTALLATION WHICH IS REQUIRED BEFORE STARTING CLEANING INSULATION WAS INSTALLED.	203 467 4370
IOR TO TESTING SHALL BE DONE BE THE CONTRACTOR AT NO EXTRA COST. STS ARE SATISFACTORY ONLY WHEN JOISTS SHOW NO VISIBLE LEAKS AND TEST PRESSURE REMAINS CONSTANT AFTER CONTINUOUS TEST PERIOD. REPAIR	IESIIC.DIZ
AKS, AND REMOVE AND REPLACE DEFECTIVE PIPE, FITTINGS AND JOISTS WITH NEW MATERIAL, UNTIL ACCEPTED BY ARCHITECT AND INSPECTING ITHORITY. WICKING, CAULKING, COMPOUNDING, PEENING, OR OTHER MAKESHIFT TYPE OF REPAIRS ARE NOT PERMITTED. REPEAT TESTS AFTER REPAIRS ITIL SYSTEMS ARE PROVEN TIGHT.	ENVIRONMENTAL CONSULTANT
OT OR CHILLED WATER PIPE TEST FSTS SHALL BE MAINTAINED AS LONG AS NECESSARY TO COMPLETELY INSPECT PIPING (MINIMUM 4 HOURS).	
EST WATER PIPING BY APPLYING HYDROSTATIC PRESSURE USING PUMP; ENSURE THAT LINES ARE VENTED OF ALL AIR.	
DLLOWING PRECAUTIONS SHALL BE TAKEN DURING PRESSURE TESTS: HOT WATER SYSTEM RELIEF VALVE SHALL BE REMOVED.	EnviroMed
SYSTEM PRESSURE GAUGES WITH SCALE RANGES LOWER THAN TEST PRESSURE SHALL BE REMOVED OR ISOLATED. WATER CONTROL VALVES SHALL BE REMOVED.	Cleaner environment. Safer workplaces.
ET FOR LEAKS BY FILLING SYSTEM WITH DRY NITROGEN. COMPRESSOR SUCTION AND DISCHARGE VALVES SHALL BE CLOSED. EXPANSION VALVES SHALL BE LUGGED. USE SEPARATE PUMP OR AVAILABLE GAS CYLINDER PRESSURE TO PRESSURIZE SYSTEM.	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
AXIMUM TEST PRESSURE SHALL NOT EXCEED 120% OF FOLLOWING MINIMUM TEST PRESSURE:	
REFRIGERANT HIGH SIDE PIPING: 250 PSIG REFRIGERANT LOW SIDE PIPING PPIY SOAPY WATER MIXTURE AND VISUALLY INSPECT FACH PIPE JOINT, VALVE PACKING AND BONNET, FLANGE, SIGHT GLASS, EITTING AND PIECE OF	
QUIPMENT FOR LEAKAGE, LIQUID LEAK DETECTOR SHALL NOT BE USED EXCEPT WITH PRIOR WRITTEN APPROVAL FROM ARCHITECT. AFTER PRESSURE TESTS AVE BEEN COMPLETED AND ACCEPTED, EVACUATE SYSTEMS TO ATMOSPHERE AND PROCEED WITH EVACUATION TESTS.	
<u>R SYSTEMS BALANCING</u> COCURE THE SERVICES OF A CERTIFIED BALANCING CO. TO PERFORM THE TESTING AND BALANCING OF THE AIR SYSTEMS.	
OMPLETELY TEST AND BALANCE ALL SUPPLY, RETURN AND EXHAUST AIR SYSTEMS AND PROVE THE CAPACITIES OF THE SYSTEM AND THE SYSTEM OMPONENTS. BALANCE THE GRILLES, REGISTERS, DIFFUSERS AND EQUIPMENT TO OBTAIN THE RESULTS INDICATED ON THE DWGS. SUBMIT A BALANCING PORT INDICATING THE RESULTS TO ENGINEER FOR APPROVAL	ISSUED FOR BIDDING
ATER SYSTEMS BALANCING	MARCH 12, 2024
OCURE THE SERVICES OF A CERTIFIED BALANCING CO. TO PERFORM THE TESTING AND BALANCING OF THE WATER SYSTEMS. MPLETELY TEST AND BALANCE ALL SUPPLY AND RETURN PIPING SYSTEMS. BALANCE FLOWS TO DESIGN/SCHEDULED LISTING FOR EACH PIECE OF	
UIPMENT (PUMP, CUIL, TERMINAL UNIT, ETC.). INCLUDE SIZE, CV VALUE OF EACH CONTROL VALVE, AND EQUIPMENT SERVED IN THE FINAL BALANCING PORT. SUBMIT THE REPORT TO THE ENGINEER FOR APPROVAL.	
	MARK DATE DESCRIPTION
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY:
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE MECHANICAL
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE MECHANICAL SPECIFICATIONS
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE MECHANICAL SPECIFICATIONS
	MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE MECHANICAL SPECIFICATIONS

_

5

3

2

DEMOLITON GENERAL NOTES

- 1. ALL EXISTING ELECTRICL EQUIPMENT AND/OR DEVICE SHOWN WITHOUT "ER" SHALL BE DISCONNECTED AND REMOVED, REMOVE ALL ASSOCIATED BACK BOX, CONDUIT AND WIRING BACK TO SOURCE OR LAST DEVICE.
- 2. "ER" INDICATED EXISTING ELECTRIOCAL EQUIPMENT AND/OR DEVICE TO REMAIN.
- 3. NO EQUIPMENT OR DEVICE THAT HAVE BEEN DISCONNECTED AND OR ABANDONED SHALL REMAIN.
- 4. THE ELECTRICAL CONTACTOR SHALL VISIT THE SITE AND BECOME FAMILIAR WITH THE EXISTING SYSTEMS AND CONIDTIONS IN AREAS OF RENOVATION.
- 5. ANY SYSTEMS OR EQUIPMENT TO REMAIN ACTIVE DURING THE RENOVATION SHALL BE KEPT IN OPERATION BY PROVIDING TEMPORARY CONNECTIONS AS REQUIRED UNTIL NEW SYSTEMS ARE INSTALLED AND OPERATIONAL.
- 6. THE ELECTRICAL CONTRACTOR SHALL COORDINATE WITH THE OWNER, CM, AND OR GENERAL CONTACTOR ANY AND ALL PHASING OF THE MECHANICAL DEMOLITION WORK IN ORDER TO SATISFY THE CONSTRUCTION SCHEDULE AND OWNERS OCCUPANCY REQUIREMENTS.
- 7. ANY ELECTRICAL EQUIPMENT TO BE REMOVED AND REUSED OR TURNED OVER TO THE OWNER, AT OWNERS REQUEST, OR AS INDICATED ON THE DRAWINGS SHALL BE CAREFULLY REMOVEED AND STORED TO PREVENT DAMAGE.
- 8. THE ELECTRICAL CONTACTOR SHALL ALSO REVIEW THE ARCHITECTURAL DEMOLITION DRAWINGS AS PART OF THIS CONTACT FOR ADDITIONAL INFORMATION AND REQUIREMENTS.
- 9. ALL SERVICE INTERRUPTIONS SHALL BE COORDINATED AND APPROVED WITH THE OWNER IN ADVANCE PRIOR TO COMMENCEMENT OF ANY WORK.
- 10. THE ELECTRICAL CONTACTOR SHALL COORDINATE HIS DEMOLITION WORK WITH THAT OF OTHER TRADES IN ORDER TO AVIOD CONFLICTS.





DEMOLITON GENERAL NOTES

 ALL EXISTING ELECTRICL EQUIPMENT AND/OR DEVICE SHOWN WITHOUT "ER" SHALL BE DISCONNECTED AND REMOVED, REMOVE ALL ASSOCIATED BACK BOX, CONDUIT AND WIRING BACK TO SOURCE OR LAST DEVICE.

 "ER" - INDICATED EXISTING ELECTRIOCAL EQUIPMENT AND/OR DEVICE TO REMAIN.

3. NO EQUIPMENT OR DEVICE THAT HAVE BEEN DISCONNECTED AND OR ABANDONED SHALL REMAIN.

 THE ELECTRICAL CONTACTOR SHALL VISIT THE SITE AND BECOME FAMILIAR WITH THE EXISTING SYSTEMS AND CONIDTIONS IN AREAS OF RENOVATION.

 ANY SYSTEMS OR EQUIPMENT TO REMAIN ACTIVE DURING THE RENOVATION SHALL BE KEPT IN OPERATION BY PROVIDING TEMPORARY CONNECTIONS AS REQUIRED UNTIL NEW SYSTEMS ARE INSTALLED AND OPERATIONAL.

 THE ELECTRICAL CONTRACTOR SHALL COORDINATE WITH THE OWNER, CM, AND OR GENERAL CONTACTOR ANY AND ALL PHASING OF THE MECHANICAL DEMOLITION WORK IN ORDER TO SATISFY THE CONSTRUCTION SCHEDULE AND OWNERS OCCUPANCY REQUIREMENTS.

 ANY ELECTRICAL EQUIPMENT TO BE REMOVED AND REUSED OR TURNED OVER TO THE OWNER, AT OWNERS REQUEST, OR AS INDICATED ON THE DRAWINGS SHALL BE CAREFULLY REMOVEED AND STORED TO PREVENT DAMAGE.

8. THE ELECTRICAL CONTACTOR SHALL ALSO REVIEW THE ARCHITECTURAL DEMOLITION DRAWINGS AS PART OF THIS CONTACT FOR ADDITIONAL INFORMATION AND REQUIREMENTS.

 ALL SERVICE INTERRUPTIONS SHALL BE COORDINATED AND APPROVED WITH THE OWNER IN ADVANCE PRIOR TO COMMENCEMENT OF ANY WORK.

10. THE ELECTRICAL CONTACTOR SHALL COORDINATE HIS DEMOLITION WORK WITH THAT OF OTHER TRADES IN ORDER TO AVIOD CONFLICTS.



3/13/2024 9:12:30 AM



5

DEMOLITON GENERAL NOTES

- SOURCE OR LAST DEVICE.
- DEVICE TO REMAIN.
- OR ABANDONED SHALL REMAIN.
- AREAS OF RENOVATION.
- ARE INSTALLED AND OPERATIONAL.
- OWNER, CM, AND OR GENERAL CONTACTOR ANY AND ALL SATISFY THE CONSTRUCTION SCHEDULE AND OWNERS OCCUPANCY REQUIREMENTS.
- AND STORED TO PREVENT DAMAGE.
- CONTACT FOR ADDITIONAL INFORMATION AND REQUIREMENTS.
- APPROVED WITH THE OWNER IN ADVANCE PRIOR TO COMMENCEMENT OF ANY WORK.
- AVIOD CONFLICTS.

2

3

1. ALL EXISTING ELECTRICL EQUIPMENT AND/OR DEVICE SHOWN WITHOUT "ER" SHALL BE DISCONNECTED AND REMOVED, REMOVE ALL ASSOCIATED BACK BOX, CONDUIT AND WIRING BACK TO

2. "ER" - INDICATED EXISTING ELECTRIOCAL EQUIPMENT AND/OR

3. NO EQUIPMENT OR DEVICE THAT HAVE BEEN DISCONNECTED AND

4. THE ELECTRICAL CONTACTOR SHALL VISIT THE SITE AND BECOME FAMILIAR WITH THE EXISTING SYSTEMS AND CONIDTIONS IN

5. ANY SYSTEMS OR EQUIPMENT TO REMAIN ACTIVE DURING THE RENOVATION SHALL BE KEPT IN OPERATION BY PROVIDING TEMPORARY CONNECTIONS AS REQUIRED UNTIL NEW SYSTEMS

6. THE ELECTRICAL CONTRACTOR SHALL COORDINATE WITH THE PHASING OF THE MECHANICAL DEMOLITION WORK IN ORDER TO

7. ANY ELECTRICAL EQUIPMENT TO BE REMOVED AND REUSED OR TURNED OVER TO THE OWNER, AT OWNERS REQUEST, OR AS INDICATED ON THE DRAWINGS SHALL BE CAREFULLY REMOVEED

8. THE ELECTRICAL CONTACTOR SHALL ALSO REVIEW THE ARCHITECTURAL DEMOLITION DRAWINGS AS PART OF THIS

9. ALL SERVICE INTERRUPTIONS SHALL BE COORDINATED AND

10. THE ELECTRICAL CONTACTOR SHALL COORDINATE HIS DEMOLITION WORK WITH THAT OF OTHER TRADES IN ORDER TO

OWNER 130 Lincoln Street Meriden, CT 06451 203 235 7146 www.girlsincmeriden.org ARCHITECT

PROFESSIONAL SEAL

CHRISTOPHER WILLIAMS ARCHITECTS, LLC

85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com

MEP ENGINEER

INNOVATIVE ENGINEERING SERVICES, LLC AN INTEGRATED ENGINEERING + E AN INTEGRATED ENGINEERING + DESIGN FIRM

33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370

iesllc.biz

ENVIRONMENTAL CONSULTANT



DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986

ISSUED FOR BIDDING

MARCH 12, 2024

MARK DATE

PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT

SHEET TITLE

SECOND FLOOR ELECTRICAL DEMOLITION PLAN

DE.102

C:\Users\cskinner\Documents\RVT2024\Girls Inc. MEP R24_cskinner_6GB99RY4KMYC.rvt

3/13/2024 9:12:31 AM

DESCRIPTION



PROFESSIONAL SEAL
OWNER
130 Lincoln Street Meriden, CT 06451
203 235 7146 www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS. LLC
85 Willow Street New Haven, CT 06511 203 776 0184 www.cwarchitectsllc.com
MEP ENGINEER
INNOVATIVE ENGINEERING SERVICES, LLC
AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd
Wallingford, CT 06492
203 467 4370 iesllc.biz
ENVIRONMENTAL CONSULTANT
_
EnviroMed
Cleaner environment. Safer workplaces.
DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO:
DRAWN BY:
CHK'D BY:
COPYRIGHT
SHEET TITLE
THIRD FLOOR
THIRD FLOOR ELECTRICAL
THIRD FLOOR ELECTRICAL DEMOLITION PLAN
THIRD FLOOR ELECTRICAL DEMOLITION PLAN
THIRD FLOOR ELECTRICAL DEMOLITION PLAN DE.103

3/13/2024 9:12:31 AM



PROFESSIONAL SEAL
OWNER
130 Lincoln Street Meriden, CT 06451
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
203 776 0184 www.cwarchitectsllc.com
MEP ENGINEER
SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370 iesllc.biz
ENVIRONMENTAL CONSULTANT
EnvíroMed
Cleaner environment. Safer workplaces.
DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
PROJECT NO:
CWA PROJECT NO.: 2313
DRAWN BY: CHK'D BY:
COPYRIGHT
BASEMENT ELECTRICAL
E.100
nner 6GB99RY4KMYC.rvt 3/13/2024 9:12:31 AM



5

3

2

DRAWING NOTES

1 PROVIDE PVC BACKBOARD FOR METER

2 LOCATE LTG FIXTURE PER DWG C501 (TYPICAL OF 4)

	ROFESSIONAL SEAL
0\	
	130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
AF	RCHITECT
	CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511
	203 776 0184 www.cwarchitectsllc.com
	INNOVATIVE ENGINEERING
-	AN INTEGRATED ENGINEERING + DESIGN FIRM
	·
	33 North Plains Industrial Rd Wallingford, CT 06492
	203 467 4370
	Iesilc.biz
	VVIRONMENTAL CONSULTANT
_	
	EnviroMod
	Enviromed
	Cleaner environment. Safer workplaces. Designed by: IOHN LUBY
0	T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
c	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
c	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	ISSUED FOR BIDDING
	T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024
	T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024
	T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024
	ISSUED FOR BIDDING MARCH 12, 2024
	ISSUED FOR BIDDING MARCH 12, 2024
	ISSUED FOR BIDDING MARCH 12, 2024
	ISSUED FOR BIDDING MARCH 12, 2024
	T ASBESTOS PROJECT DESIGNER LICENSE #19 T LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024
	TASBESTOS PROJECT DESIGNER LICENSE #19 TIEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024
	ISSUED FOR BIDDING MARCH 12, 2024
	TASBESTOS PROJECT DESIGNER LICENSE #19 TLEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 RK DATE DESCRIPTION
	TASBESTOS PROJECT DESIGNER LICENSE #19 TLEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 RK DATE DESCRIPTION ROJECT NO:
	TASBESTOS PROJECT DESIGNER LICENSE #19 TASBESTOS PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 RK DATE DESCRIPTION ROJECT NO: VA PROJECT NO: 2313
	TASBESTOS PROJECT DESIGNER LICENSE #19 TASBESTOS PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 RK DATE DESCRIPTION ROJECT NO: VA PROJECT NO.: 2313 RAWN BY:
	TASBESTOS PROJECT DESIGNER LICENSE #19 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 RK DATE DESCRIPTION ROJECT NO: VA PROJECT NO.: 2313 RAWN BY: IK'D BY:
	TASBESTOS PROJECT DESIGNER LICENSE #19 ISSUED FOR BIDDING MARCH 12, 2024 RK DATE DESCRIPTION RK DATE DESCRIPTION ROJECT NO: VA PROJECT NO:: 2313 RAWN BY: IK'D BY:
Image: Control of the second	TASBESTOS PROJECT DESIGNER LICENSE #19 TI LEAD PLANNER/PROJECT DESIGNER LICENSE #986
	Trasbestos project designer license #19 Triead planner/project designer license #986 ISSUED FOR BIDDING MARCH 12, 2024
MA PF CV DF CV F	TASBESTOS PROJECT DESIGNER LICENSE #19 TI LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024
MA PF CV DF CC SF	TASBESTOS PROJECT DESIGNER LICENSE #19 ISSUED FOR BIDDING MARCH 12, 2024 RK DATE DESCRIPTION ROJECT NO: VA PROJECT NO: 2313 RWN BY: IK'D BY: DPYRIGHT IEET TITLE IRST FLOOR LECTRICAL PLAN
	TASBESTOS PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024
MA PF CV DF CV F	TASBESTOS PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 RK DATE DESCRIPTION ROJECT NO: 2313 RAWN BY: IK'D BY: DPYRIGHT IEET TITLE IRST FLOOR LECTRICAL PLAN E.101



PROFESSIONAL SEAL
OWNER
130 Lincoln StreetMeriden, CT 06451203 235 7146www.girlsincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC
85 Willow StreetNew Haven, CT 06511203 776 0184www.cwarchitectsllc.com
MEP ENGINEER
SERVICES, LLC AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370
ENVIRONMENTAL CONSULTANT
EnvíroMed
Cleaner environment. Safer workplaces.
DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19
CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
MARK DATE DESCRIPTION
PROJECT NO: CWA PROJECT NO.: 2313
DRAWN BY:
CHK'D BY: COPYRIGHT
SECOND FLOOR
□ 102
E.102



THIRD FLOOR ELECTRICAL PLAN Scale: 1/8" = 1'-0"

3



3

1 INSTALL LIGHTING FIXTURE ON FLAT PORTIONS OF GYP BD CEILING. SEE ARCH DWGS

2

2

DRAWING NOTES

1

PROFESSIONAL SEAL	
OWNER	
203 235 7146 www.girlsincmeriden.org	
ARCHITECT	
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven, CT 06511	
203 776 0184 www.cwarchitectsllc.com	
MEP ENGINEER	
INNOVATIVE ENGINEERING SERVICES, LLC	4
AN INTEGRATED ENGINEERING + DESIGN FIRM	1
33 North Plains Industrial Rd	
203 467 4370	
iesllc.biz	
—	
EnvíroMed	
EnviroMed	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986	5.
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986	12
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986	
Cleaner environment. Safer workplaced DESIGNED BY OF DESIGNER LICENSE #18 CT ASBESTOS PROJECT DESIGNER LICENSE #986	
Cleaner environment. Safer workplaced DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #1980 ISSUED FOR BIDDING MARCH 12, 2024	
Cleaner environment. Safer workplaces Designed BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #196 TIEAD PLANNER/PROJECT DESIGNER LICENSE #986 USSUED FOR BIDDING MARCH 12, 2024	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CTASBESTOS PROJECT DESIGNER LICENSE #186 OSCILLAND PLANNER/PROJECT DESIGNER LICENSE #866 USSUED FOR BIDDING MARCH 12, 2024	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024	
Cleaner environment. Safer workplaces Designed BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #080 DISSUED FOR BIDDING MARCH 12, 2024	
Cleaner environment. Safer workplaces Designed BY: John LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #980 USSUED FOR BIDDING MARCH 12, 2024	
Cleaner environment. Safer workplaces Designed by: John Luby Crasbestros project designer License #19 Criead planner/project designer License #966 ARACH 12, 2024	
Cleaner environment. Safer workplaces Designed by: John LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #18 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #980 USSUED FOR BIDDING MARCH 12, 2024	
Enclose Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOG PROJECT DESIGNER LICENSE #18 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #200 DISSUED FOR BIDDING MARCH 12, 2024	
Encorrection Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 MARK DATE MARK DATE MARK DATE MARCH 10:	
ECCEPTION Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #18 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #988 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO: 2313 DRAWN RY-	
EncircoMedic Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASBESTOS FROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 MARK DATE MARK DATE MARK DATE PROJECT NO: 2313 DRAWN BY: CHK'D BY:	
Encircle environment. Safer workplaces DESIGNED BY: JOHN LUBY CTASBESTOS PROJECT DESIGNER LICENSE #18 CTIEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDDING MARCH 12, 2024 MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO: Z313 DRAWN BY: CHK'D BY: COPYRIGHT	
Encorrectionment: Safer workplaces DESIGNED BY: JOHN LUBY! CIT ASBESTTOS PROJECT DESIGNER LICENSE #13 CIT LEAD PLANNER/PROJECT DESIGNER LICENSE #386 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO: 2313 DRAWN BY: CHK'D BY: COPYRIGHT	
EFFECTION Cleaner environment. Safer workplaces DESIGNED BY JOHN LUBY CTASBESTOS PROJECT DESIGNER LICENSE #38 CTIEAD PLANNER/PROJECT DESIGNER LICENSE #38 ISSUED FOR BIDDING MARCH 12, 2024 MARK DATE MARK DESCRIPTION PROJECT NO: 2313 DRAWN BY: COPYRIGHT SHEET TITLE THIRD FLOORR	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO: 2313 DRAWN BY: CHK/D BY: COPYRIGHT SHEET TITLE THIRD FLOOR ELECTRICAL PLAN	
Cleaner environment. Safer workplaces DESIGNED BY: JOHN LUBY CT ASSESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #980 ISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 MA	
Cleaner environment. Safer workplaced DESIGNED BY: JOHN LUTEY CT ABBESTOS PROJECT DESIGNER LICENSE #188 CISSUED FOR BIDDING MARCH 12, 2024 MARCH 12, 2024 MARCH 12, 2024 MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE THIRD FLOOR ELECTRICAL PLAN	





	UILJ
(11)	1"C, #1/0 GROUNDING ELECTRODE CONDUCTOR (GEC). CONNECT TO DOMESTIC WATER SOURCE AND BOND TO GAS METER WITH APPROVED GROUNDING CLAMP, PROVIDE JUMPER ACROSS METERS AND CONTINUOUS GROUND PATH ON BOTH SIDES OF METERS.
(12)	PANELBOARD P1, SEE PANEL P1 SCHEDULE AND ELECTRICAL SPECIFICATIONS.
(13)	EXISTING LOAD CENTER TO REMAIN. REFEED FROM PANELBOARD P1 AS INDICATED.
(14)	EXTEND EXISTING FEEDER TO PANELBOARD P1. MATCH EXISTING WIRING AND CIRCUIT BREAKERS.
(15)	2 SETS 3 1/2" C, 2#350KCMIL, #4G.
(16)	100 AMP, 2 POLE, HEAVY DUTY SAFETY SWITCH WITH TWO (2) 70 AMP CLASS-RK1 FUSES, NEMA TYPE 1 EMCLOSURE.
(17)	30 KVA BUCK BOOST TRANSFORMER INPUT VOLTAGE 240V (1) ONE PHASE, OUTPUT VOLTAGE 208V (1) ONE PHASE.
(18)	1 1/4" C, 2#3, #8G.
(19)	PHASE PERFECT DIGITAL PHASE CONVERTER MODEL # PT-030R-BHS INPUT 240 (1) ONE PHASE, OUTPUT 208 (3) 3 PHASE.
20	PHASE CONVERTER WIRED WITH 1 1/4" C, 3#4, #8G TO AHU-2.



4 I C

2

1

RA MA MA INT	TING: 2 IN BUS: IN DEVIO ERRUPT	40/120V, ⊠ CU CE: MLO ING RATII	1 PHASE, 3 WIRE J 🔲 AL 600A MAIN BUS WITH FEED THRU LUGS NG: 42KAIC			PANE SECTI	L P1 ON 1			LOCATION: BASEMENT ELECTRICA MOUNTING: FLUSH SU TRIM: SINGLE DOOR	AL ROOM RFACE	R-IN-DOC	IR
CIVT	CKT. BKR.										CKT.	BKR.	
NO.	TRIP AMPS	POLES	DESCRIPTION	WIRE & CONDUIT	KVA	A	В	KVA	WIRE & CONDUIT	DESCRIPTION	POLES	TRIP AMPS	NO.
1	20	1	BOILER B-1	3/4"C, 2#12, #12G	0.9	1.8		0.9	3/4"C, 2#12, #12G	BOILER B-2	1	20	2
3	20	2	AHU-1	3/4"C, 2#12, #12G	1.0		1 .4	0.4	-	CONTROL PANEL	1	20	4
5	20	1	SPARE	-	0.0			1.7	3/4"C, 2#10, #10G	CU-4	2	20	6
7	50	2	AHU-2	1 1/4"C, 2#3, #8G	7.0	8.7		1.7	-	-	-	-	8
9	-	-	-	-	7.0		8 .7	1.7	3/4"C, 2#10, #10G	CU-2	2	20	10
11	20	2	CU-1	3/4"C, 2#10, #10G	1.7			1.7	-	-	-	-	12
13	-	-	-	-	1.7	3 .4		1.7	3/4"C, 2#10, #10G	CU-5	2	20	14
15	20	2	CU-3	3/4"C, 2#10, #10G	1.7		3.4	1.7	-	-	-	-	16
17	-	-	-	-	1.7			1.7	3/4"C, 2#10, #10G	CU-6	2	20	18
19	20	2	CU-7	3/4"C, 2#10, #10G	1.7	3 .4		1.7	-	-	-	-	20
21	-	-	-	-	1.7		3.4	1.7	3/4"C, 2#10, #10G	CU-8	2	20	22
23	20	1	EXH. FAN EF-1	3/4"C, 2#12, #12G	0.2			1.7	-	-	-	-	24
25	20	1	EXH. FAN EF-2	3/4"C, 2#12, #12G	0.2	0.3		0.1	3/4"C, 2#12, #12G	RECEPTACLE - BASEMENT	1	20	26
27	20	1	EXH. FAN EF-3	3/4"C, 2#12, #12G	0.1		0.4	0.3	3/4"C, 2#12, #12G	RECEPTACLE - BASEMENT	1	20	28
29	20	1	EXH. FAN EF-4	3/4"C, 2#12, #12G	0.1			0.1	3/4"C, 2#12, #12G	RECEPTACLE - EXTERIOR	1	20	30
31	20	1	EXH. FAN EF-5	3/4"C, 2#12, #12G	0.1	0.2		6.0	SEE POWER RISER	PHASE CONVERTER	2	70	32
33	20	1	EXH. FAN EF-6	3/4"C, 2#12, #12G	0.1		0.1	6.0	-	-	-	-	34
35	100	2	EXISTING PANEL	MATCH EXISTING FEEDER	12.0			1.0	3/4"C, 3#6, #10G	EXISTING RANGE	2	50	36
37	-	-	-	-	12.0	13.0		1.0	-	-	-	-	38
39	100	2	EXISTING PANEL	MATCH EXISTING FEEDER	9.6		15.4	5.8	MATCH EXISTING FEEDER	EXISTING SUBPANEL	2	60	40
41	-	-	-	-	9.6			5.8	-	-	-	-	42
	ł			TOTALS PER PHASE		30.8	32.8				I	1	
				TOTALS SECTION 1	33.6								
				TOTALS SECTION 2	7.3	1							
				GRAND TOTAL (2-SECT.)	40.9								
				L	1	1							

RA ⁻ MA MA INT	TING: 24 IN BUS: IN DEVIC ERRUPTI	40/120V, ⊠ CU CE: MLO ING RATII	1 PHASE, 3 WIRE I 🗌 AL 600A MAIN BUS NG: 42KAIC			PA SEC	NEL CTIO	P1 N 2			LOCATION: BASEMENT ELECTRIC MOUNTING: FLUSH SI TRIM: SINGLE DOOR	AL ROOM JRFACE	L-IN-DOC	DR
	СКТ	. BKR.										СКТ.	BKR.	
CKT. NO.	TRIP AMPS	POLES	DESCRIPTION	WIRE & CONDUIT	KVA	A		В	KVA	WIRE & CONDUIT	DESCRIPTION	POLES	TRIP AMPS	CKT.
43	20	2	FCU'S - 1ST FLOOR	3/4"C, 3#12, #12G	0.1	•	0.3		0.2	3/4"C, 3#12, #12G	FCU'S - SECOND FLOOR	2	20	44
45	-	-	-	-	0.1			0.3	0.2	-	-	-	-	46
47	20	2	FCU-E 1ST FLOOR	3/4"C, 3#12, #12G	0.1				0.1	3/4"C, 2#12, #12G	RECEPTACLE - RECIRC PUMP	1	20	48
49	-	-	-	-	0.1	•	0.2		0.1	3/4"C, 2#12, #12G	RECEPTACLE - RECIRC PUMP	1	20	50
51	70	2	PHASE CONVERTER	SEE RISER DIAGRAM	6.0			6 .1	0.1	3/4"C, 2#12, #12G	RECEPTACLE - EXTERIOR	1	20	52
53	-	-	-	-	6.0				0.1	3/4"C, 2#12, #12G	RECEPTACLE - EXTERIOR	1	20	54
55	20	1	RECEPTACLE - EXTERIOR	3/4"C, 2#12, #12G	0.1	•	0.2		0.1	3/4"C, 2#12, #12G	RECEPTACLE - EXTERIOR	1	20	56
57	20	1	LIGHTING - EXTERIOR RAMP	3/4"C, 2#12, #12G	0.2			• 0.4	0.2	3/4"C, 2#12, #12G	LIGHTING - THIRD FLOOR	1	20	58
59	20	1	SPARE	-	0.0				0.0	-	SPARE	1	20	60
61	20	1	SPARE	-	0.0	e	0.0		0.0	-	SPARE	1	20	62
63	20	1	SPARE	-	0.0			0.0	0.0	-	SPARE	1	20	64
65	20	1	SPARE	-	0.0				0.0	-	SPARE	1	20	66
67	20	1	SPARE	-	0.0	e	0.0		0.0	-	SPARE	1	20	68
69	20	1	SPARE	-	0.0			0.0	0.0	-	SPARE	1	20	70
71	20	1	SPARE	-	0.0				0.0	-	SPARE	1	20	72
73	20	1	SPARE	-	0.0	•	0.0		0.0	-	SPARE	1	20	74
75	20	1	SPARE	-	0.0			0.0	0.0	-	SPARE	1	20	76
77	20	1	SPARE	-	0.0				0.0	-	SPARE	1	20	78
79	20	1	SPARE	-	0.0	•	0.0		0.0	-	SPARE	1	20	80
81	20	1	SPARE	-	0.0			0.0	0.0	-	SPARE	1	20	82
83	20	1	SPARE	-	0.0				0.0	-	SPARE	1	20	84
		1	r	TOTALS PER PHASE		0.7	7 0	6.6		,				
				GRAND TOTAL	7.3									
				•		-								

PROFESSIONAL SEAL
OWNER
130 Lincoln Street Meriden, CT 06451
203 235 7 146 www.girisincmeriden.org
ARCHITECT
CHRISTOPHER WILLIAMS ARCHITECTS, LLC 85 Willow Street New Haven. CT 06511
203 776 0184 www.cwarchitectsllc.com
AN INTEGRATED ENGINEERING + DESIGN FIRM
33 North Plains Industrial Rd Wallingford, CT 06492
203 467 4370
leslic.blz
EnviroMod
DESIGNED BY: JOHN LUBY
CT ASBESTOS PROJECT DESIGNER LICENSE #19 CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986
ISSUED FOR BIDDING
MARCH 12, 2024
·
MARK DATE DESCRIPTION
MARK DATE DESCRIPTION
MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY:
MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY:
MARK DATE DESCRIPTION PROJECT NO: 2313 DRAWN BY: CHK'D BY: COPYRIGHT
MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: CHK'D BY: COPYRIGHT SHEET TITLE
MARK DATE DESCRIPTION MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: CHK'D BY: COPYRIGHT SHEET TITLE ELECTRICAI
MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE ELECTRICAL SCHEDULES
MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE ELECTRICAL SCHEDULES
MARK DATE DESCRIPTION MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE ELECTRICAL SCHEDULES
MARK DATE DESCRIPTION PROJECT NO: CWA PROJECT NO.: 2313 DRAWN BY: CHK'D BY: COPYRIGHT SHEET TITLE ELECTRICAL SCHEDULES ELECTRICAL SCHEDULES

3/13/2024 9:12:35 AM

	5		4	ABBREVIATIONS
D			A ADA AFF AWG C C/B CLG CKT DW DWG E/EM ER FL G G FJB NIC NTS P PH RR UON V W WP	AMPERE AMERICANS WITH DISABILITIES A ABOVE FINISHED FLOOR AMERICAN WIRE GAUGE CONDUIT CIRCUIT BREAKER MOUNTED IN CEILING CIRCUIT DISHWASHER DRAWING EMERGENCY POWER CIRCUIT EXISTING TO REMAIN FLOOR GROUND GROUNDING FAULT CIRCUIT INTE JUNCTION BOX NOT IN CONTRACT NOT TO SCALE POLE PHASE TO BE REMOVEWD AND RELOCATE UNLESS OTHERWISE NOTED VOLT WATT WEATHERPROOF
C				
В				
A				

		3
		ELECTRICAL LEGEND
	SYMBOL	DESCRIPTION
ACT	S _{TT}	MANUAL MOTOR STARTER WITH THERMAL OVERLOAD PROTECTION
	S _E	EMERGENCY POWER OFF SWITCH : SINGLE POLE TOGGLE SWITCH WITH RED "EMERGENCY OFF" WALL PLATE
	S _{TTE}	MANUAL MOTOR STARTER WEITH THERMAL OVERLOAD PROTECTION, 'E' WHERE SPECIFIED INDICATES EMERGENCY POWER OFF WITH RED "EMERGENCY OFF" WALL PLATE.
	₽ ^{GF}	GFCI DUPLEX RECEPTACLE
RRUPTER	₽ ^{GF/WP}	GFCI DUPLEX RECEPTACLE WITH WEATHERPROOF 'WHILE-IN-USE' COVER.
	₽ ^{USB}	DUPLEX RECEPTACLE WITH (2) USB HOSPITAL GRADE: HUBBELL USB8300_ OR APPROVED EQUAL TAMPER PROOF: HUBBELL USB20X2_ OR APPROVED EQUAL
-D	Ŷ	OUTLET TO SUIT EQUIPMENT (CONTRACTOR TO VERIFY)
		EXISTING PANELBOARD/LOAD CENTER
		120/240V PANEL
		REVENU METER
		CONDUIT AND WIRE
		HOMERUN TO PANELBOARD, NUMBER/LETTERS INDICATE CIRCUIT AND PANELBOARD TERMINATION UNLESS OTHERWISE NOTED
	Ū	JUCTION BOX
	Ø	MOTOR
	<u></u>	SAFETY DISCONNECT SWITCH
	Ľ	FUSIBLE SAFETY DISCONNECT SWITCH
	•	THERMAL LINK FOR EMERGENCY BOILER SHUT DOWN
		SWITCHED WIRE

LIGHT FIXTURE SCHEDULE

LIGHTING FIXTURE TYPE	\bigcirc ^A
	SIGMA
CATALOG NO	120 DT22 DAL 000E T
FIXTURE DESCRIPTION	
VOLTAGE	120
LAMP & DESIGNATION	EXTERIOR RAMP
DRIVER	IFD
MOUNTING	
HOUSING	ALUMINUM
LENS/LOUVER	ARCRLIC
REMÁRKS	XXXXX
	<u>B</u>
LIGHTING FIXTURE TYPE	
MANUFACTURER	LITHONIA LIGHTING
CATALOG NO.	LBL4-4800LM-80CRI-35K-MIN1-ZT-MVOLT-LSXRHL
FIXTURE DESCRIPTION	WRAPAROUND
VOLTAGE	
LAMP & DESIGNATION	THIRD FLOOR
DRIVER	HIGH EFFICIENCY
MOUNTING	SURFACE
HOUSING	STEEL
LENS/LOUVER	CURVED PRISMATIC DIFFUSER
REMARKS	WITH BUILT IN OCCUPANCY SENSOR

GENERAL NOTES	
 THE ELECTRICAL CONTRACTOR SHALL BE RESPONSIBLE FOR ALL WORK REQUIRED FOR A COMPLETE, FULLY OPERABLE INSTALLATION. ALL WORK DONE IN ACCORDANCE WITH THE LATEST APPROVED ISSUE OF THE NFPA 70, NATIONAL ELECTRIC CODE (NEC) AND APPLICABLE LOCAL CODES. 	(TO
2. PRIOR TO SUBMISSION OF NIDS GIVE WRITTEN NOTICE KTO ARCHITECT AND ENGINEER OF ANY MATERIAL OR APPRATUS THAT IS INADEQUATE, UNSUITABLE FOR THE USE, IN VOLATION OF LAWS, ORDINANCES, RULES, CODE OR ANY REGULATIONS OF AUTHORITIES HAVING JURISDICTION ANY NECESSARY ITEMS OF WORK THAT HAS BEEN OMITTED. ONTATOE AFFIRMS THAT ABSENT SUCH NOTICE, ALL SYSTEMS WILL FUNCTION SATIFACTORILY WITHOUT ADDITIONAL EXTRA CCOMPENSION.	OR
3. THE ELECTRICAL CONTACTOR SHALL VISIT THE SITE PRIOR TO SUBMITTING A BID AND ADHERE TO CONTENTS OF THE BID DOCUMENTS. ANY DEVIATIONS FROM THE INFORMASTION PROVIDED IN THE DOCUMENTS MUST BE LISTED IN WRITING. INNOVATIVE ENGINEERING SERVICES, LLC THE RIGHT TO BE COMPENSATED FOR REVIEW OF VALUE ENGINEERING OR SUBSTITUTED MATERIALS AND EQUIPMENT.	C HAS
4. ELECTRICAL CONTRACTOR SHALL RESTORE ALL AREAS DISTURBED TO THEIR ORIGINAL CONDITION. ELECTRICAL CONTRACTOR IS RESPONSIBLE ALL CUTTING, PATCHING, PAINTING, CLEAN-UP, ELECTRICAL DEBRIS REMOVAL AND GENERAL COORDINATION OF THE WORK EFFORT AS REQUIN THE INSTALLATION OF THE ELECTRICAL ITEMS OF WORK.	EFOR
5. THE DRAWINGS SHOW THE GENERAL LAYOUT AND SOME OF THE DETAIL, BUT THEY DO NOT SHOW EVERY FITTING, BEND, ETC. ELECTRICAL CONTRACTOR SHALL PROVIDE ALL SUCH MATERIALS TO MAKE A COMPLETE INSTALLATION.	
6. ALL PART NUMBERS ARE PROVIDED FOR THE CONVENIENCE OF THE CONTRACTOR. THEY ARE NOT TO BE CONSIDERED THE COMPLETE SPECIFICATION OF THE PRODUCT. THE PART NUMBER AND DESCRIPTION WILL BE THE COMPLETE SPECIFICATION. IN THE EVENT OF A DISCREPANCY BETWEEN T TWO, THE MORE STRINGENT, MORE COSTLY FEATURE/PERFORMANCE WILL BE REQUIRED.	ATION THE
7. ELECTRICAL CONTRACTOR SHALL BE RESPONSIBLE FOR TESTING OF ALL PHASES OF THE WORK AND TO DEMONSTRATE TO OWNER THAT THE EQUIPMENT IS IN FULL OPERATING ORDER.	
8. DO NOT SCALE DRAWINGS; ACTUAL FIELD MEASUREMENTS AND DIMENSIONS TAKE PRECEDENCE IN ALL CASES.	
9. ALL WORK SHALL BE DONE IN ACCORDANCE WITH THE STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT, AIA DOCUMENT A LATEST EDITION.	201
10. ELECTRICAL CONTRACTOR SHALL INSTALL ALL EQUIPMENT IN ACCORDANCE WITH MANUFACTURERS INSTRUCTIONS AND OR REQUIREMENTS FO PROPER OPERATION AND MAINTENANCE.	R
11. IN ADDITION TO THE REQUIREMENTS SHOWN FOR LOW VOLTAGE EQUIPMENT AND RACEWAYS, THE ELECTRICIAN SHALL CARRY AN ALLOWANCE FINAL COORDINATION OF ALL RACEWAYS AND SLEEVES REQUIRED TO FACILITATE THE LOW VOLTAGE CONSULTANT WORK.	FOR
12. ELECTRICAL CONTRACTOR SHALL WARRANT AND GUARANTEE ALL MATERIALS AND WORKMANSHIP FOR A PERIOD OF ONE YEAR FROM THE DATE FINAL ACCEPTANCE BY THE OWNER.	OF
13. ELECTRICAL CONTRACTOR SHALL GIVE OWNER 10 DAYS ADVANCE NOTICE OF SHUTDOWNS, OR AS DIRECTED BY THE OWNER. SHUTDOWNS TO KEPT TO A MINIMUM. AT NO TIME SHALL THE BUILDING/SPACE BE LEFT WITHOUT COMMERCIAL POWER IN FULL OPERATING ORDER.	BE
14. PROVIDE INDEPENDENT SEISMIC SUPPORT OF ALL ELECTRICAL EQUIPMENT PER THE LATEST ADOPTED VERSION OF THE INTERNATIONAL BUILDI CODE.	NG
15. ALL ELECTRICAL PENETRATIONS TO BE FIREPROOFED TO MAINTAIN INTEGRITY OF FIRE WALLS/FLOORS/CEILINGS.	

16. ALL THE WIRE SIZES ARE BASED ON COPPER, ALUMINUM IS NOT TO BE USED UNLESS NOTED OTHERWISE.

THE FIRST OUTLET; OVER 60 FEET, #10; OVER 105 FEET, #8; INCREASE CONDUIT SIZE TO SUIT. 18. CIRCUIT NUMBERS ARE INDICATED FOR INTENT ONLY. THE ELECTRICAL CONTRACTOR SHALL ADJUST ACCORDINGLY IN THE FIELD, TO BALANCE CIRCUITS EVENLY ON ALL PHASES.

19. ALL WIRING METHODS ARE TO BE IN ACCORDANCE WITH THE CURRENT ISSUE OF THE NATIONAL ELECTRICAL CODE, AND APPLICABLE LOCAL CODES. ALL WIRING IS TO BE IN CONDUIT, UNLESS SPECIFICALLY NOTED OTHERWISE. ALL WIRING IS TO BE CONCEALED.

20. ELECTRICAL CONTACTOR MUSR COORDINATE LED DRIVER WITH DIMMER SWITCH FOR COMPATIBILITY. 21. SHARED NEUTRALS ARE NOT TO BE USED. PROVIDE SEPARATE NEUTRALS FOR ALL CIRCUITS INCLUDING SWITCHES PER THE LATEST EDITION OF THE NEC.

22. FOR ALL ROOFTOP OR GRADE LEVEL HVAC EQUIPMENT, THE ELECTRICAL CONTRACTOR SHALL SUPPLY A GFCI WP, 20A RECEPTACLE FOR EQUIPMENT SERVICING. ALL DISCONNECT SWITCHES ARE TO BE HEAVY DUTY, FUSED, WEATHER PROOF (WP) DEVICES.

23. FOR ALL FLOOR BOXES WITH DATA COMMUNICATIONS SERVICES PROVIDE (2) 3/4" EMPTY CONDUITS TO HUNG CEILING OR OTHER ACCESSIBLE SPACE. INSTALL A DRAG WIRE.

24. FOR ALL WALL/CEILING BOXES FOR DATA COMMUNICATIONS PROVIDE 3/4"C EMPTY CONDUITS TO HUNG CEILING OR OTHER ACCESSIBLE SPACE. INSTALL A DRAG WIRE.

25. RISER DIAGRAMS ARE PROVIDED TO SHOW DIAGRAMMATIC GENERAL WIRING REQUIREMENTS. WIRING IS TO BE PROVIDED FOR THE PARTICULAR VENDOR/SYSTEM APPROVED FOR THE PROJECT. ALL WIRING IS TO BE CONCEALED.

26. ALL WIRING IN AIR PLENUM CEILINGS SHALL BE TEFLON COATED AND RATED FOR USE WITHIN THE PLENUM. 27. NO LOW VOLTAGE WIRING SHALL BE PERMITTED IN THE SAME RACEWAY AS POWER WIRING. 28. EMERGENCY SERVICES SHALL BE RUN IN SEPARATE RACEWAYS FROM ALL OTHER SYSTEMS. 29. PROVIDE DRAG LINES IN ALL EMPTY RACEWAYS.

30. ELECTRICAL CONTRACTOR SHALL PROVIDE SHOP DRAWINGS FOR APPROVAL FOR ALL LIGHT FIXTURES, PANELS, SWITCHES, RECEPTACLES, POKE THROUGH'S, WIREMOLD, TRANSFORMERS, GENERATOR, ATS, UPS AND SPECIALTY ELECTRICAL SYSTEMS, ETC.

31. ELECTRICAL CONTRACTOR TO VERIFY LIGHTING FIXTURE MOUNTING HEIGHT REQUIREMENTS FOR VARIOUS CEILING TYPES AND ORDER APPROPRIATE HARDWARE.

32. COORDINATE EXACT PLACEMENT OF EQUIPMENT WITH ARCHITECTURAL AND MECHANICAL PLANS, MAKE FIELD ADJUSTMENTS AS REQUIRED TO AVOID CONFLICTS, VERIFY WITH OWNER.

33. ELECTRICAL CONTRACTOR TO COORDINATE WITH ARCHITECTURAL AND MECHANICAL CONTRACTOR FOR ITEMS SUPPLIED BY THE MECHANICAL/OTHER DIVISIONS BUT INSTALLED BY THE ELECTRICAL CONTRACTOR. ELECTRICAL CONTRACTOR TO REVIEW ALL THE PLANS FOR THE PROJECT AND ASSIST IN PROVIDING COORDINATION DRAWINGS WITH OTHER TRADES.

34. ELECTRICAL CONTRACTOR TO VERIFY LOADS, SETTINGS, OVERCURRENT PROTECTION ... ETC TO INSURE COMPATIBILITY OF EQUIPMENT. 35. PROVIDE LAMICOID NAMEPLATES AND TYPEWRITTEN PANEL SCHEDULES FOR ALL PANEL BOARDS, DISTRIBUTION AND DISCONNECT EQUIPMENT, MODIFIED OR PROVIDED AS NEW AS PART OF THE SCOPE OF WORK. ALTERNATE LABELING PROVISIONS MUST BE APPROVED BY THE ENGINEER.

36. REFER TO ARCHITECTURAL DRAWINGS FOR THE EXACT LOCATION AND MOUNTING HEIGHTS OF ALL DEVICES AND OUTLETS.

37. CONTRACTOR TO PROVIDE ALL COORDINATION FOR UTILITY SERVICES INCLUDING TEMPORARY SERVICE FOR THE PROJECT. THIS INCLUDES POWER UTILITY, TELEPHONE COMPANY AND CABLE TV OPERATOR.

38. DISCONNECT SWITCHES AND CIRCUIT BREAKER USED AS SWITCHES SHALL BE INSTALLED IN ACCORDANCE WITH ALL LOCAL CODES AND THE LATEST VERSION OF THE NATIONAL ELECTRICAL CODE "NEC" SECTION 110.26 TABLE 110.26(A)(1) AND SECTION 404.8. ALL DISCONNECT SWITCHES AND CIRCUIT BREAKERS SHALL BE LOCATED SO THAT THEY MAY BE OPERATED FROM A READILY ACCESSIBLE PLACE. THEY SHALL BE INSTALLED SUCH THAT THE CENTER OF THE GRIP OF THE OPERATING HANDLE OF THE SWITCH OR CIRCUIT BREAKER, WHEN IN ITS HIGHEST POSITION, IS NOT MORE THAN 6'-7" ABOVE THE FLOOR OR WORKING PLATFORM WITH 36" CLEAR IN FRONT.

39. ALL SURFACE RACWAY LOCATED IN FINSH AREAS SHALL BE INSTALLED IN WIREMOLD SURFACE METAL RACEWAY OR APPROVED EQUAL.

AL NOTES

17. MINIMUM CONDUCTOR SIZE FOR A FULLY LOADED 20A CIRCUIT, UNLESS OTHERWISE NOTED, SHALL BE #12 FOR ALL BRANCH CIRCUIT RUNS UP TO



ELECTRICAL SPECIFICATIONS

PART 1 - GENERAL PROVISIONS FOR ELECTRICAL WORK

<u>REFERENCES</u>

THIS SECTION COVERS THE GENERAL REQUIREMENTS FOR ELECTRICAL WORK; EXAMINE ALL CONTRACT DRAWINGS AND ALL OTHER SECTIONS OF THE SPECIFICATIONS FOR ADDITIONAL WORK RELATED TO THE WORK OF THIS DIVISION.

DEFINITIONS

'PROVIDE' - TO FURNISH, INSTALL AND CONNECT UP COMPLETE AND READY FOR SAFE AND REGULAR OPERATION OF PARTICULAR WORK REFERRED TO UNLESS, SPECIFICALLY OTHERWISE NOTED.

'INSTALL' - TO ERECT, MOUNT AND CONNECT COMPLETE WITH RELATED ACCESSORIES.

'WORK' - LABOR, MATERIALS, EQUIPMENT, APPARATUS, CONTROLS, ACCESSORIES AND OTHER ITEMS REQUIRED FOR PROPER AND COMPLETE INSTALLATION.

'WIRING' - RACEWAY, FITTINGS, WIRE, BOXES, MOUNTING HARDWARE AND RELATED ITEMS.

'CONCEALED' - EMBEDDED IN MASONRY OR OTHER CONSTRUCTION CAVITY, INSTALLED IN FURRED SPACES, WITHIN DOUBLE PARTITIONS OR HUNG CEILINGS.

'SIMILAR' OR 'EQUAL' - EQUAL MATERIALS, WEIGHT, SIZE, DESIGN AND EFFICIENCY OF SPECIFIED PRODUCT.

'CONTRACTOR' - THE ELECTRICAL CONTRACTOR.

'NOTED' - AS INDICATED ON THE DRAWINGS AND/OR SPECIFICATIONS.

<u>SCOPE</u>

THIS WORK SHALL CONSIST OF THE FURNISHINGS OF ALL LABOR, MATERIALS AND SERVICES REQUIRED COMPLETE, READY FOR CORRECT OPERATION FOR ALL ELECTRICAL WORK CALL FOR BY THE ACCOMPANYING DRAWINGS AND SPECIFICATIONS. ALL ELECTRICAL WORK SHALL BE PERFORMED IN ACCORDANCE WITH THE NATIONAL ELECTRICAL CODE, STATE AND LOCAL CODES.

THE DATA INDICATED IN THESE DRAWINGS AND SPECIFICATIONS ARE AS EXACT AS COULD BE SECURED. BUT THEIR ABSOLUTE ACCURACY IS NOT GUARANTEED. DO NOT SCALE DRAWINGS. EXACT LOCATIONS, DISTANCES, LEVELS AND OTHER CONDITIONS WILL BE GOVERNED BY THE BUILDING. USE THE DRAWINGS AND SPECIFICATIONS FOR GUIDANCE AND SECURE THE ENGINEER'S APPROVAL OF CHANGES IN LOCATIONS. CIRCUITS, WHERE SHOWN ON AN ELECTRICAL DRAWINGS, ARE SO INDICATED PRIMARILY FOR THE PURPOSE OF INDICATING THE GENERAL CIRCUIT PLAN AND DO NOT NECESSARILY INDICATE THE EXACT LOCATION OF ROUTING OF THE RACEWAYS UNLESS SPECIFICALLY INDICATED. CIRCUITS SHALL BE RUN IN SUIT CONDITIONS CONSIDERING STRUCTURAL FEATURES, OTHER TRADES, CONSTRUCTION METHODS AND GOOD INSTALLATION PRACTICE.

BEFORE SUBMITTING A BID, THE CONTRACTOR SHALL VISIT THE SITE AND BECOME THOROUGHLY FAMILIAR WITH ALL EXISTING CONDITIONS UNDER WHICH THE WORK AND WORK OF OTHER TRADES WILL BE INSTALLED. THIS CONTRACT INCLUDES ALL NECESSARY OFFSETS, TRANSITIONS, MODIFICATIONS AND RELOCATION REQUIRED TO INSTALL ALL NEW EQUIPMENT IN NEW OR EXISTING SPACES. CONTRACTOR SHALL INCLUDE ANY MODIFICATIONS REQUIRED IN EXISTING ELECTRICAL EQUIPMENT FOR INSTALLATION OF NEW ELECTRICAL EQUIPMENT AND NEW EQUIPMENT OF OTHER TRADES. (LIGHTING FIXTURES, DEVICES, CONDUIT WIRING, ETC.) ALL NEW AND EXISTING EQUIPMENT AND SYSTEMS SHALL BE FULLY OPERATIONAL UNDER THIS CONTRACT BEFORE THE PROJECT IS CONSIDERED COMPLETE. THE CONTRACTOR SHALL BE HELD RESPONSIBLE FOR ANY ASSUMPTIONS THAT ARE MADE, ANY OMISSIONS OR ERRORS MADE AS A RESULT OF FAILURE TO VISIT THE SITE AND BECOME THOROUGHLY FAMILIAR WITH THE EXISTING CONDITIONS AND THE CONTRACT DOCUMENTS OF ALL TRADES.

CODES, REGULATIONS AND STANDARDS

ALL ELECTRICAL WORK SHALL BE PERFORMED IN ACCORDANCE WITH THE FOLLOWING APPROVED

CODES, REGULATIONS AND STANDARDS:

IBC - INTERNATIONAL BUILDING CODE, 2021 EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.

IFC - INTERNATIONAL FIRE CODE, 2021 EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.

2022 CONNECTICUT STATE FIRE SAFETY CODE

STATE DEMOLITION CODE

LOCAL BUILDING CODE

NFPA - NATIONAL FIRE PROTECTION CODE

NFPA 70 - NATIONAL ELECTRICAL CODE, 2020 EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.

NFPA 72 - NATIONAL FIRE ALARM CODE, 2019 EDITION

NFPA 101 - LIFE SAFETY CODE, 2021 EDITION, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.

IECC - INTERNATIONAL ENERGY CONSERVATION CODE, 2021, AS AMENDED BY THE STATE OF CONNECTICUT 2022 AMENDMENTS.

ICC/ANSI A117.1, 2017, ACCESSIBLE AND USABLE BUILDINGS AND FACILITIES, AS AMENDED BY THE STATE OF CONNECTICUT 2018 AMENDMENTS.

ANSI - AMERICAN NATIONAL STANDARDS INSTITUTE

ASTM - AMERICAN SOCIETY FOR TESTING AND MATERIALS

OSHA - OCCUPATIONAL SAFETY AND HEALTH ADMINISTRATION

U.L. - UNDERWRITERS LABORATORIES

EPA - ENVIRONMENTAL PROTECTION AGENCY

IEEE - INSTITUTE OF ELECTRICAL AND ELECTRONIC ENGINEERS

NEMA - NATIONAL ELECTRICAL MANUFACTURERS ASSOCIATION

PERMITS, FEES AND INSPECTIONS

THE CONTRACTOR SHALL GIVE ALL NECESSARY NOTICES, OBTAIN ALL PERMITS, PAY FOR ALL GOVERNMENT, STATE SALES TAXES AND APPLICABLE FEES. THE CONTRACTOR SHALL FILE ALL DRAWINGS, COMPLETE ALL DOCUMENTS AND OBTAIN ALL NECESSARY APPROVALS FROM THE PROPER AUTHORITY OR AGENCY HAVING JURISDICTION. OBTAIN ALL REQUIRED CERTIFICATES OF INSPECTION COVERING WORK. THE CONTRACTOR SHALL SEE THAT ALL REQUIRED INSPECTIONS AND TESTS ARE MADE AND SHALL COOPERATE TO MAKE THESE TESTS AS THOROUGH AND AS READILY MADE AS POSSIBLE.

MATERIALS AND WORKMANSHIP

ALL MATERIALS AND APPARATUS REOUIRED FOR THE WORK, EXCEPT AS OTHERWISE SPECIFIED, SHALL BE NEW AND OF FIRST-CLASS QUALITY. IT SHALL BE FURNISHED, DELIVERED, ERECTED, CONNECTED, FINISHED IN EVERY DETAIL AND SO SELECTED AND ARRANGED AS TO FIT PROPERLY INTO THE BUILDING SPACES. WHERE NO SPECIFIC KIND OR QUALITY MATERIAL IS GIVEN, A FIRST-CLASS STANDARD ARTICLE AS ACCEPTED BY THE ENGINEER SHALL BE FURNISHED.

ALL EQUIPMENT AND MATERIALS SHALL BE SPECIFICATION GRADE AND BEAR THE UNDERWRITER'S LABEL. NO SUBSTITUTE OR ALTERNATE EQUIPMENT, MATERIAL, ETC. WILL BE CONSIDERED FOR THIS PROJECT.

ALL WORK SHALL BE OF A QUALITY CONSISTENT WITH GOOD TRADE PRACTICE AND SHALL BE INSTALLED IN A NEAT, WORKMANLIKE MANNER. THE ENGINEER/OWNER RESERVES THE RIGHT TO REJECT ANY WORK WHICH, IN HIS OPINION, HAS BEEN INSTALLED IN A SUBSTANDARD, DANGEROUS OR IN A UNSERVICEABLE MANNER. THE CONTRACTOR SHALL REPLACE REJECTED WORK IN A SATISFACTORY MANNER AT NO EXTRA COST TO THE OWNER.

GUARANTEES

ALL WORKMANSHIP AND MATERIALS SHALL BE FULLY GUARANTEED FOR A PERIOD OF ONE YEAR AFTER ACCEPTANCE OF THE ENTIRE INSTALLATION COVERED BY THIS CONTRACT. SHOULD ANY DEFECTS OCCUR DURING THE GUARANTEED PERIOD, THE CONTRACTOR SHALL REPAIR AND/OR REPLACE ALL DEFECTIVE EQUIPMENT, MATERIAL AND/OR WORK AT NO EXTRA CHARGE TO THE OWNER.

RECORD DRAWINGS

MAINTAIN, AT THE JOB SITE, A SET OF ELECTRICAL DRAWINGS INDICATING ALL CHANGES IN LOCATION AND CIRCUITING OF THE EQUIPMENT, PANELS, DEVICES, ETC. FROM THE ORIGINAL LAYOUT. CLEARLY MARK IN RED ALL CHANGES ON THE DRAWINGS. AT THE COMPLETION OF THE PROJECT THE CONTRACTOR SHALL TURN OVER THE RECORD DRAWINGS TO THE ENGINEER/OWNER.

<u>COORDINATION</u>

ALL WORK SHALL BE COORDINATED AND CARRIED OUT IN CONJUNCTION WITH ALL TRADES AND FULL COORDINATION DRAWINGS SHALL BE CREATED IN ORDER THAT ALL WORK MAY PROCEED WITH A MINIMUM OF DELAY AND INTERFERENCE.

SHOP DRAWINGS

SUBMIT ELECTRONIC PDF FORMAT OR EIGHT (8) COPIES FOR REVIEW, DETAILED SHOP DRAWINGS OF ALL EQUIPMENT AND MATERIAL SPECIFIED. THE CONTRACTOR SHALL REVIEW ALL SHOP DRAWINGS PRIOR TO INSTALLED UNTIL CONTRACTOR HAS IN THEIR POSSESSION, APPROVED SHOP DRAWINGS FOR THE PARTICULAR IDENTIFIED.

THE FOLLOWING IS A LIST OF ELECTRICAL ITEMS THAT MUST BE SUBMITTED FOR REVIEW:

a. PANELBOARDS

- b. LOAD CENTERS c. SAFETY/DISCONNECT SWITCHES
- d. CIRCUIT BREAKERS e. LIGHTING
- f. FUSES
- q. CONDUIT, WIRE AND CABLE h. FIRE ALARM EQUIPMENT

i. DEVICES (RECEPTACLES, TOGGLE SWITCHES, ETC.)

OPERATING INSTRUCTIONS

AND MAINTENANCE INSTRUCTIONS.

THE CONTRACTOR, IN THE ABOVE-MENTIONED INSTRUCTIONS, SHALL INCLUDE THE MAINTENANCE SCHEDULE FOR THE PRINCIPAL ITEMS OF EQUIPMENT FURNISHED UNDER THIS DIVISION.

AN AUTHORIZED MANUFACTURER'S REPRESENTATIVE SHALL ATTEST IN WRITING THAT HIS EQUIPMENT HAS BEEN PROPERLY INSTALLED PRIOR TO STARTUP. THESE LETTERS WILL BE BOUND INTO OPERATING AND MAINTENANCE BOOKS.

EQUIPMENT PROTECTION

PROPERLY AND COMPLETELY PROTECT AGAINST ALL DAMAGE, ALL APPARATUS, EQUIPMENT, ETC., INCLUDED IN THIS CONTRACT. THE CONTRACTOR WILL BE HELD RESPONSIBLE FOR ANY DAMAGE TO FURNISHED APPARATUS, EQUIPMENT, ETC., UNTIL FINAL ACCEPTANCE.

PROPERTY PROTECTION

THE CONTRACTOR SHALL TAKE WHATEVER MEANS NECESSARY AND/OR REQUIRED TO PROTECT OWNER'S PROPERTY WITHIN THE WORKING AREAS FROM DUST, DEBRIS AND OTHER MATTER GENERATED BY THE WORK. NO WORK SHALL COMMENCE IN AREAS WHERE PROTECTION IS REQUIRED UNTIL APPROVAL HAS BEEN GIVEN TO THE CONTRACTOR BY THE OWNER.

MANUFACTURER'S INSTRUCTION

INSTALL ALL EQUIPMENT IN ACCORDANCE WITH MANUFACTURER'S INSTRUCTIONS OR REQUIREMENTS FOR PROPER OPERATION AND MAINTENANCE.

EQUIPMENT PAINTING AND CLEANING

THOROUGHLY CLEAN ALL ELECTRICAL EQUIPMENT DEVICES AND ENCLOSURES UPON COMPLETION OF ALL WORK. REPAINT ANY EQUIPMENT WHOSE FINISH IS DAMAGED OR RUSTED. MATCH MANUFACTURER'S ORIGINAL FINISH.

DEMOLITION/REMOVAL AND RECONNECTION

BEFORE SUBMITTING A BID, THE CONTRACTOR SHALL VISIT THE SITE AND BECOME THOROUGHLY FAMILIAR WITH ALL EXISTING CONDITIONS UNDER WHICH HIS WORK WILL BE INSTALLED.

ALL EXISTING ACTIVE CIRCUITS WHICH FEED EQUIPMENT OR DEVICES THAT ARE TO REMAIN, SHALL BE MAINTAINED IN SERVICE AND SHALL BE PERMANENTLY REFED.

ALL ITEMS BEING REMOVED SHALL REMAIN THE PROPERTY OF THE CONTRACTOR AND REMOVED FROM THE SITE THE PROPERTY OF THIS CONTRACTOR AND SHALL BE REMOVED FROM THE SITE.

ALL MATERIAL CHOSEN TO BE RETAINED BY THE OWNER SHALL BE DELIVERED BY THE CONTRACTOR TO SUCH POINT AS DESIGNATED BY THE OWNER.

MATERIAL TO BE REUSED SHALL BE CAREFULLY REMOVED AND STORED AND SHALL BE REINSTALLED IN AS-FOUND CONDITION EXCEPT AS OTHERWISE INDICATED ON THE PLANS. DAMAGE OR LOSS OF MATERIAL TO BE REUSED SHALL BE THE CONTRACTOR'S RESPONSIBILITY AND SHALL BE REPAIRED OR REPLACED WITH THE EQUIVALENT MATERIAL ACCEPTABLE BY THE OWNER.

DISCONNECT AND REMOVE ALL EXISTING ELECTRICAL WORK IN CONTRACT AREA AS INDICATED ON DRAWINGS.

ALL WIRING AND CABLING SHALL BE REMOVED BACK TO ORIGINATION PANEL, UNLESS OTHERWISE INDICTED. CONTRACTOR SHALL KEEP PREMISES FREE FROM ACCUMULATION OF WASTE MATERIAL AND RUBBISH, AND AT COMPLETION OF WORK DAY, SHALL REMOVE ALL RUBBISH AND IMPLEMENTS TO A DESIGNATED LOCATION, IF AVAILABLE, LEAVING WORK AREAS BROOM CLEAN. UNUSED OUTLET BOXES AND PLASTER RINGS SHALL BE PROVIDED WITH BLANK COVER PLATES AND MATCH DEVICE PLATES WITHIN THE ROOM.

PENETRATION SEALANT

ALL PENETRATIONS SHALL BE SEALED WITH 3M INTUMESCENT FIRE BARRIER PENETRATION SEALANT, APPLIED PER MANUFACTURER'S AND U.L. GUIDELINES.

CUTTING, PATCHING, REPAIRING AND PAINTING

THE GENERAL CONTRACTOR SHALL PERFORM ALL CUTTING, PATCHING, REPAIRING AND PAINTING FOR ALL ELECTRICAL ITEMS AND EQUIPMENT CALLED FOR UNDER THIS CONTRACT.

FIRE STOPS AND SEALS

PENETRATIONS THROUGH FIRE-RATED WALLS, CEILING OR FLOORS IN WHICH CABLES OR CONDUITS PASS SHALL BE FILLED SOLIDLY BY U.L. APPROVED FIRE-STOP MATERIALS, CLASSIFIED FOR AN HOUR RATING EQUAL TO THE FIRE RATING OF THE WALL, CEILING OR FLOOR. PROVIDE TO 3M BRAND FIRE BARRIER CP25WB CAULK OR APPROVED EQUIVALENT.

WATER, A DAMP OR CORROSIVE ATMOSPHERE, DRAFT OR DUST.

ACCESS PANELS

THE CONTRACTOR SHALL FURNISH AND INSTALL ACCESS PANELS AND DOORS AS REQUIRED FOR ACCESS TO INACCESSIBLE PULLBOXES, JUNCTION BOXES AND OTHER SPECIALTIES.

THE CONTRACTOR SHALL COORDINATE THE LOCATIONS OF ACCESS PANELS AND DOORS WITH THE GENERAL CONTRACTOR AND OTHER TRADES. FINAL LOCATIONS SHALL BE SUBJECT TO THE APPROVAL OF THE ARCHITECT.

SUBMISSION TO THE ENGINEER FOR REVIEW. NO MATERIAL OR EQUIPMENT MAY BE DELIVERED TO THE JOB SITE OR MATERIAL OR EQUIPMENT. SHOP DRAWINGS SHALL BE SPECIFIC WITH ITEMS SUBMITTED FOR APPROVAL CLEARLY

THE CONTRACTOR SHALL FURNISH TO THE ENGINEER, FOUR (4) COMPLETE BOUND SETS OF TYPEWRITTEN OR BLUEPRINTED INSTRUCTIONS FOR OPERATING AND MAINTAINING ALL SYSTEMS AND EQUIPMENT INCLUDED IN THIS DIVISION. MANUFACTURER'S ADVERTISING LITERATURE OR CATALOGS WILL NOT BE ACCEPTABLE FOR OPERATING

UNLESS OTHERWISE INDICATED. EQUIPMENT AND DEVICES THE OWNER DOES NOT WISH TO RETAIN SHALL BECOME

SEALING BUSHINGS SHALL BE USED ON CONDUIT AND CABLE ENDS TO EFFECTIVELY PREVENT THE INTRUSION OF

INTERRUPTION OF EXISTING SERVICES

PRIOR TO PERFORMING WORK REQUIRING INTERRUPTION OF EXISTING SERVICES, THE CONTRACTOR SHALL SECURE FROM THE OWNER, APPROVAL OF THE PROPOSED OPERATION. (48 HOURS IN ADVANCE) WORK SHALL BE ARRANGED FOR CONTINUOUS PERFORMANCE. THE CONTRACTOR SHALL PROVIDE ALL NECESSARY LABOR, INCLUDING OVERTIME, OF REQUIRED, TO ASSURE THAT THE EXISTING OPERATING SERVICES WILL BE SHUT DOWN ONLY DURING THE TIME ACTUALLY REQUIRED TO MAKE NECESSARY CONNECTIONS.

TEMPORARY LIGHT AND POWER

FURNISH AND INSTALL TEMPORARY ELECTRICAL SERVICE OF SUFFICIENT SIZE FOR POWER AND LIGHTING USE BY ALL TRADE CONTRACTORS DURING THE COURSE OF CONSTRUCTION. ALL TEMPORARY WORK SHALL BE DONE IN COMPLIANCE WITH ALL APPLICABLE ARTICLES IN THE NATIONAL ELECTRICAL CODE, O.S.H.A. AND WITH ALL REQUIREMENTS OF ANY AUTHORITY HAVING JURISDICTION OVER THE WORK. PROVIDE TEMPORARY POWER CONNECTION TO A CONSTRUCTION TRAILER. THE LOCATION OF THE TRAILER IS TO BE DETERMINED ON SITE PROVIDE FEEDER TO TRAILER OF SUFFICIENT SIZE TO POWER HEATING, AIR CONDITIONING, GENERAL POWER AND LIGHTING.

<u>PART 2 - PRODUCTS</u>

<u>DESCRIPTION</u>

ALL MATERIALS AND EQUIPMENT PROVIDED UNDER THIS SECTION SHALL BE NEW, FIRST GRADE, BEST OF THEIR RESPECTIVE KINDS AND IN NO WAY SHALL THEY BE LESS THAN THE QUALITY AND INTENT SET FOURTH UNDER THIS SECTION. THEY SHALL MEET THE REQUIREMENTS OF ALL STANDARDS SET UP TO GOVERN THE MANUFACTURER OF ELECTRICAL MATERIALS AND COMPLY WITH ALL APPLICABLE CODES AND STANDARDS.

<u>WIRE</u>

CONDUCTORS SHALL BE U.L. LISTED, 600 VOLTS, 90 DEG. C., SINGLE CONDUCTOR TYPE THWN/THHN. 98% CONDUCTIVITY, ANNEALED UNCOATED COPPER WITH PVC INSULATION COVERED WITH NYLON SHEATH JACKET. TESTED IN ACCORDANCE WITH THE REQUIREMENTS OF UNDERWRITERS LABORATORIES STANDARD 83. WIRE SHALL BE IDENTIFIED BY SURFACE MARKING INDICATING MANUFACTURER'S IDENTIFICATION CONDUCTOR SIZE AND METAL, VOLTAGE RATING, U.L. SYMBOL AND TYPE DESIGNATION. CONDUCTORS SHALL BE STRANDED. MINIMUM SIZE SHALL BE #12 AWG UNLESS OTHERWISE INDICATED. MANUFACTURED BY ROME CABLE, TRIANGLE WIRE & CABLE, GENERAL CABLE OR ESSEX WIRE & CABLE.

RIGID GALVANIZED STEEL CONDUIT (RGS)

RIGID STEEL CONDUIT SHALL BE FULL WEIGHT, HEAVY WALL STEEL PIPE WITH GALVANIZED PROTECTIVE COATING. MANUFACTURED BY TRIANGLE WIRE AND CABLE, ALLIED TUBE AND CONDUIT, REPUBLIC OR STEELDUCT. CONDUIT FITTINGS SHALL BE MALLEABLE IRON, CADMIUM PLATED WITH FULL THREADED HUBS.

RIGID POLYVINYL CHLORIDE CONDUIT (PVC)

RIGID POLYVINYL CHLORIDE CONDUIT SHALL BE TYPE DB, SCHEDULE 40, SUNLIGHT RESISTANT, RATED OR USE WITH 90 DEGREES C. CONDUCTORS, U.L. RATED. ALL PVC CONDUIT AND FITTINGS SHALL BE SOLVENT WELDED. MANUFACTURED BY CARLON, ELECTRI-FLEX OR PLASTILINE.

ELECTRIC METALLIC TUBING (EMT)

ELECTRICAL METALLIC TUBING SHALL BE GALVANIZED THIN WALL STEEL CONDUIT. MANUFACTURED BY TRIANGLE WIRE AND CABLE, ALLIED TUBE AND CONDUIT, REPUBLIC OR STEELDUCT. THE CONNECTORS AND COUPLINGS SHALL BE HEAVY DUTY, STEEL-ZINC PLATED, SET SCREW TYPE.

FLEXIBLE METALLIC CONDUIT (FMC)

FLEXIBLE METALLIC CONDUIT SHALL BE OF HEAVY GALVANIZED SHEET METAL STRIP IN INTERLOCKED CONSTRUCTION. MANUFACTURED BY TRIANGLE WIRE AND CABLE, AMERICAN FLEXIBLE CONDUIT OR ELECTRIC-FLEX. THE CONNECTORS SHALL BE SQUEEZE TYPE MALLEABLE IRON, CADMIUM PLATED.

LIQUID-TIGHT FLEXIBLE METAL CONDUIT (LFMC)

LIQUID-TIGHT FLEXIBLE CONDUIT SHALL BE CONSTRUCTED OF HEAVY GALVANIZED SHEET METAL STRIP, SPIRALLY-WOUND INTERLOCK CONSTRUCTION WITH AN EXTRUDED POLYVINYL GRAY JACKET. CONDUIT SHALL BE U.L. LABELED AND CONFORMED TO THE APPLICATION AND ENVIRONMENT IN WHICH IT WILL BE USED. ALL CONNECTIONS, COUPLINGS AND FITTINGS SHALL BE OF HIGH QUALITY STEEL-ZINC RATED TYPE SPECIFICALLY DESIGNED FOR THIS PURPOSE. MANUFACTURED BY O/Z GEDNEY OR ELECTRI-FLEX.

METAL CLAD CABLE (MC)

METAL CLAD CABLE SHALL BE INTERLOCKING GALVANIZED STEEL ARMOR CONSTRUCTION. COLOR CODED THERMOPLASTIC/NYLON INSULATION THHN, 90 DEGREE C., 600 VOLTS, COPPER CONDUCTORS AND INTERNAL INSULATED EQUIPMENT COPPER GROUND CONDUCTOR. MARKER TAPE AND CABLE TAPE OVER MINIMUM SIZE #12 AWG UNLESS OTHERWISE INDICATED. MANUFACTURED BY AMERICAN FLEXIBLE CONDUIT, TRIANGLE WIRE AND CABLE, GENERAL CABLE OR STANDARD CABLE.

<u>FITTINGS</u>

CONDUIT BODIES FOR RIGID GALVANIZED STEEL CONDUIT (RGS) SHALL BE MALLEABLE IRON-ZINC PLATED WITH TAPERED HUBS AND GASKETED ALUMINUM COVER.

CONDUIT BODIES FOR ELECTRICAL METALLIC TUBING (EMT) SHALL BE CAST ALUMINUM-ALUMINUM ENAMEL FINISH WITH SET SCREW HUBS AND ALUMINUM COVER.

INSULATION BUSHINGS SHALL BE HIGH IMPACT THERMOPLASTIC PHENOLIC WITH 150 DEG. C. UL TEMPERATURE RATING.

INSULATED GROUNDING BUSHINGS SHALL BE MALLEABLE IRON ZINC PLATED WITH MOLDED ON PHENOLIC INSULATION AND LAY-IN GROUNDING LUG.

CONDUIT LOCKNUTS SHALL BE HEAVY NUT STOCK STEEL-ZINC PLATED.

OFFSET NIPPLES SHALL BE MALLEABLE IRON ZINC PLATED WITH RIGID CONDUIT THREADING AND 3/4" OFFSET. CONNECTORS AND COUPLINGS FOR ELECTRICAL METALLIC TUBING (EMT) SHALL BE HEAVY STEEL-ZINC PLATED

CONDUIT STRAPS SHALL BE SNAP-TYPE, DOUBLE RIBBED STEEL-ZINC PLATED.

METAL CLAD CABLE AND FLEXIBLE METALLIC CONDUIT CONNECTORS SHALL BE MALLEABLE IRON-ZINC PLATED, MALE HUB THREADS WITH LOCKNUT.

CONDUIT FITTINGS SHALL BE MANUFACTURED BY O/Z GEDNEY, CROUSE-HINDS OR APPLETON.

SUPPORT FITTINGS

WITH PRE-SET/PRE-SHAKED SET SCREWS.

SUPPORT CHANNEL SHALL BE ROLL-FORMED #12 GAUGE STEEL, SOLID BASE OR BOLT HOLE BASE - HOT DIP GALVANIZED FINISH. COMPLETE WITH ANGLE FITTINGS, SPRING NUTS, CONDUIT SUPPORTS, 3/8" OR 1/2" THREADED RODS (SIZE REQUIRED FOR LOAD), ETC.

CABLE TIES

CABLE TIES SHALL BE FABRICATED OF ONE-PIECE HALLAR WITH NO METAL PARTS. MANUFACTURED BY BURNDY, T&B, PANDUIT OR BLACKBURN.

U.L. LISTED DUAL ELEMENT TIME DELAY TYPE.

AN INTERRUPTING RATING OF 200,000 A.I.C. U.L. LISTED CLASS 'RK1'.

INTERRUPTING RATING OF 200,000 A.I.C. U.L. LISTED CLASS 'RK1'.

OF 200,000 A.I.C. U.L. LISTED CLASS 'L' KLP-C.

WITH AN INTERRUPTING RATING OF 200,000 A.I.C. U.L. LISTED CLASS 'RK1'.

FUSES SHALL HAVE VOLTAGE RATING BASED ON DISTRIBUTION REQUIREMENT SYSTEMS.

BELOW

(SIZE 30"H x 24"W x 12"D).

<u>OUTLET BOXES</u>

INSULATION. MANUFACTURED BY STEEL CITY OR RACO.

ALIGN THEM EXACTLY ON CENTER LINES HORIZONTALLY OR VERTICALLY.

CONTROLLED.

SURFACE RACEWAY

SEE DRAWINGS FOR TYPE AND SIZE. MANUFACTURED BY WIREMOLD.

MAIN SERVICE ENTRANCE CABINET REFER TO ONE-LINE POWER RISER AND NOTES FOR REQUIREMENTS.

<u>CT CABINET</u>

REQUIREMENTS.

PANELBOARDS

VISIBLE WITH THE DOOR OPEN. REFER TO SCHEDULES FOR OTHER REOUIREMENTS.

CIRCUIT BREAKERS

HAMMER. REFER TO SCHEDULES FOR AIC RATING.

FOR CIRCUIT BREAKER SIZES 100 AMPS AND LARGER PROVIDE THE FOLLOWING: REPLICABLE ELECTRONIC TRIP; AND THE FOLLOWING FIELD-ADJUSTABLE SETTINGS:

1. INSTANTANEOUS TRIP 2. LONG AND SHORT TIME PICKUP LEVELS 3. LONG AND SHORT TIME ADJUSTMENTS

ALL MULTI-POLE BREAKERS SHALL BE EQUIPPED WITH HANDLE TIES FOR MULTI-POLE USE.

PHASE SEQUENCE AND BALANCING

THE ENTIRE SYSTEM. BALANCING ALL FEEDERS AND CIRCUITS TO WITHIN 10 PERCENT.

SAFETY/DISCONNECT SWITCHES

SIEMENS, SQUARE 'D', OR CUTLER HAMMER.

COMBINATION MOTOR STARTERS

MANUFACTURED BY ABB, SIEMENS, SQUARE D OR ALLEN BRADLEY.

PROFESSIONAL SEAL FUSES SHALL NOT BE INSTALLED UNTIL EQUIPMENT IS READY TO BE ENERGIZED. THIS MEASURE PREVENTS FUSE DAMAGE DURING SHIPMENT OF THE EQUIPMENT FROM THE MANUFACTURER TO THE JOB SITE. ALL FUSES SHALL BE FURNISHED AND INSTALLED BY THE CONTRACTOR. FUSES SHALL BE OF THE SAME MANUFACTURER, COPPER BUSSMAN, FERRAZ SHAWMUT OR LITTLEFUSE. FUSE TYPES DESCRIBE BELOW SHALL BE CIRCUIT 0 TO 600 AMPERE SHALL BE PROTECTED BY DUAL ELEMENT, TIME DELAY, CURRENT LIMITING FUSES WITH OWNER MOTOR CIRCUITS SHALL BE PROTECTED BY DUAL ELEMENT, TIME DELAY, CURRENT LIMITING FUSES WITH AN LARGE MOTORS SHALL BE PROTECTED BY TIME DELAY, CURRENT LIMITING FUSES WITH AN INTERRUPTING RATING CIRCUIT BREAKER PANELBOARDS SHALL BE PROTECTED BY DUAL ELEMENT, TIME DELAY, CURRENT LIMITING FUSES UPON COMPLETION OF THE WORK, THE CONTRACTOR SHALL PROVIDE THE OWNER WITH THE SPARE FUSES LISTED Meriden, CT 06451 203 235 7146 www.girlsincmeriden.org 10% (MINIMUM OF THREE) OF EACH TYPE AND RATING INSTALLED, 0 TO 600 AMPERES. SPARE FUSE CABINET, CATALOG NUMBER 'LSFCL, SHALL BE PROVIDED TO STORE THE ABOVE SPARES ARCHITECT OUTLET BOXES SHALL BE GALVANIZED STEEL, FLUSH OR SURFACE MOUNTED AND OF PROPER TYPE AND SIZE AS REQUIRED FOR THE PARTICULAR APPLICATION. SIZE AND TYPE DICTATED BY THE NUMBER OF DEVICES, NUMBER OF CONDUCTORS AND WIRING METHOD UTILIZED. BOXES SHALL BE ADEQUATE SIZE FOR THE INSTALLATION OF CONDUCTORS WITHOUT EXCESSIVE BENDING OR CRIMPING OF THE CONDUCTORS AND DAMAGING OF CONDUCTOR OUTLET BOXES SHALL BE SECURED FIRMLY IN PLACE TO THE BUILDING STRUCTURE AND SET TRUE AND SOUARE. CHRISTOPHER WILLIAMS ARCHITECTS, LLC PROVIDE SUITABLE MEANS TO SUPPORT OUTLET BOX TO TAKE THE WEIGHT OF THE LIGHTING FIXTURE OR DEVICE. 85 Willow Street New Haven, CT 06511 OUTLET BOXED OR BOX EXTENSION RINGS SHALL BE SET FLUSH TO THE FINISHED WALL OR CEILING. BOXES MUST 203 776 0184 www.cwarchitectsllc.com BE ATTACHED THAT THEY WILL NOT 'ROCK', 'SHIFT' OR 'MOVE IN AND OUT' WHEN DEVICES ARE USED. IN NO CASE SHALL BOXES BE INSTALLED BACK-TO-BACK IN A COMMON WALL DIVIDING TWO SPACES. WHERE MORE THAN ONE OUTLET IS SHOWN OR SPECIFIED TO BE THE SAME ELEVATION OR ONE ABOVE THE OTHER, MEP ENGINEER MULTIPLE SWITCHES SHOWN AT ONE LOCATION SHALL BE INSTALLED GANGED TOGETHER UNDER ONE WALL PLATE. SWITCHES SHALL BE ARRANGED IN AN ORDER APPROPRIATE TO THE LOCATIONS OF LIGHTING FIXTURE BEING INNOVATIVE ENGINEERING SERVICES, LLC -IES-AN INTEGRATED ENGINEERING + DESIGN FIRM 33 North Plains Industrial Rd Wallingford, CT 06492 203 467 4370 iesllc.biz THE SERVICE ENTRANCE SHALL HAVE A CONTINUOUS CURRENT RATING OF 600 AMPERES AT (240Y/120V), SINGLE (1) PHASE, THREE (3) WIRE, 60 CYCLE WITH A SHORT CIRCUIT RATING OF 100,000 AMPERES RMS SYMMETRICAL OR AS INDICATED ON SCHEDULES. U.L. LISTED AND LABELED FOR SERVICE ENTRANCE. (WALL) (FLOOR) MOUNTED ENVIRONMENTAL CONSULTANT TAMPER-PROOF ENCLOSURE WITH ANSI-61 LIGHT GRAY PAINT FINISH. FRONT ACCESSIBLE CABLE TERMINATORS. ALL BUS CONDUCTORS SHALL BE TIN PLATED ALUMINUM. 600A (FUSIBLE SWITCH) MAIN DEVICE. UTILITY, CURRENT TRANSFORMER COMPARTMENT INCLUDES FRONT HINGED COVER WITH PROVISION FOR LOCK WIRE SEALING AND BAR TYPE CURRENT TRANSFORMER MOUNTING. PROVIDE PULL SECTION WIRE COMPARTMENT. MANUFACTURED BY ABB, SIEMENS, SQUARE 'D', CUTLER HAMMER. REFER TO PLANS FOR ADDITIONAL DESIGNED BY: IOHN LUBY CT ASBESTOS PROJECT DESIGNER LICENSE #19 PANELBOARDS SHALL BE THE COMBINATION THERMAL/MAGNETIC CIRCUIT BREAKER TYPE, 3 PHASE, 4 WIRE WITH CT LEAD PLANNER/PROJECT DESIGNER LICENSE #986 THE NUMBER OF BRANCH CIRCUITS AS INDICATED ON THE SCHEDULES. PROVIDE WITH FULLY RATED HARD-DRAWN COPPER, 98 PERCENT CONDUCTIVE PHASE AND GROUND BUS. LUGS SIZED TO ACCOMMODATE WIRE QUALITY AND SIZES. PANELS SHALL BE U.L. LISTED. DOOR-IN-DOOR DESIGN. BOXES SHALL BE CORROSION RESISTANT, ZINC FINISH GALVANIZED. FRONTS SHALL BE REINFORCED STEEL POWDER FINISH PAINTED LIGHT GRAY (ANSI-61) AND SHALL BE EQUIPPED WITH CONCEALED HINGES AND CONCEALED TRIM ADJUSTING SCREWS. DIRECTORY CARD HOLDERS SHALL BE CORROSION-PROOF VALOX WITH RETRACTABLE LATCH, KEYED ALIKE. PHASE BUS SHALL BE SEQUENCED AND FULLY INSULATED RATINGS SHALL BE DISPLAYED ON THE DEAD FRONT SHIELD AND TOTALLY **ISSUED FOR BIDDING** BRANCH CIRCUIT BREAKERS SHALL BE QUICK-MAKE, QUICK-BREAK, (PLUG-IN) (BOLT-IN) THERMAL MAGNETIC TYPE WITH VISIBLE CURRENT RATING AND TRIP POSITION. MANUFACTURED BY ABB, SIEMENS, SQUARE 'D' OR CUTLER MARCH 12, 2024 ELECTRONIC TRIP CIRCUIT BREAKERS WITH RMS SENSING; FIELD-REPLACEABLE RATING PLUG OR FIELD-MAINTAIN CORRECT PHASE SEQUENCE OF ALL FEEDERS AND CIRCUITS WITH PHASE IDENTIFICATION THROUGHOUT DISCONNECT/SAFETY SWITCHES SHALL BE MOTOR RATED, METAL ENCLOSED, INTERLOCKING, FUSIBLE OR NONFUSED AS INDICATED. HEAVY DUTY TYPE, WITH APPROPRIATE VOLTAGE RATINGS, QUICK-MAKE, QUICK-BREAK MECHANISMS, SOLID NEUTRAL AND U.L. LISTED. SWITCHES SHALL HAVE PROPER TYPE METAL ENCLOSURES; STANDARD, WEATHERPROOF, DUSTPROOF, ETC., TO SUIT THEIR SPECIFIC LOCATIONS. MANUFACTURED BY ABB, MARK DATE DESCRIPTION PROJECT NO: FURNISH AND INSTALL COMBINATION MOTOR STARTER, NEMA SIZE X, NEMA TYPE X, RATED FOR (208, 480 VOLT), 3 CWA PROJECT NO.: 2313 PHASE, (NON-FUSED, FUSED, THERMAL MAGNETIC CIRCUIT BREAKER, MOTOR CIRCUIT PROTECTOR), FULL VOLTAGE, DRAWN BY: NON-REVERSING, MELTING ALLOY OVERLOADS, HAND-OFF-AUTO SWITCH AND AUXILIARY CONTACTS. CHK'D BY: COPYRIGHT SHEET TITLE ELECTRICAL SPECIFICATIONS E.204

C:\Users\cskinner\Documents\RVT2024\Girls Inc. MEP R24 cskinner 6GB99RY4KMYC.rvt

3/13/2024 9:12:36 AM

ELECTRICAL SPECIFICATIONS CONTINUED

MANUAL MOTORS

FURNISH AND INSTALL FRACTIONAL HORSEPOWER MANUAL MOTOR STARTERS WITH ON-OFF CONTROL. THERMAL OVERLOAD RELAY AND PILOT LIGHTS. MANUFACTURED BY ABB, SIEMENS, SQUARE 'D', OR ALLEN BRADLEY.

	ABB	SQUARE 'D'	ALLEN BRADLEY	<u>SIEMENS</u>
POLE, SURFACE	CR101Y11	2510FG1P	600-TAX4	SMFFG1P
POLE, SURFACE	CR101H11	2510FG2P	600-TAX5	SMFFG2P

JUNCTION BOXES, PULLBOXES AND WIREWAYS

JUNCTION BOXES, PULLBOXES AND WIREWAYS SHALL BE OF PROPER TYPE AND SIZES AS REQUIRED. CODE GAUGE, GALVANIZED STEEL WITH KNOCKOUTS AND FLANGES TO RECEIVE THE COVERS. COVERS SHALL BE FLAT, OF THE SAME MATERIAL AS THE BOX AND FASTENED TO THE BOX WITH MACHINE SCREWS. MANUFACTURED BY HOFFMAN, SQUARE 'D', OR LEE PRODUCTS.

BACKBOARDS

BACKBOARDS SHALL BE FIRE RETARDENT, HICKSON CO. (DRI-CON) 3/4 INCH TYPE AS PLYWOOD OF SUFFICIENT SIZE FOR MOUNTING OF SPECIFIED EQUIPMENT. PAINT ALL SIDES WITH TWO (2) COATS OF FIRE-RETARDENT GRAY ENAMEL PAINT.

GROUND RODS

GROUND RODS SHALL BE HIGH STRENGTH STEEL CORE WITH ELECTROLYTIC ALLY BONDED COPPER JACKET. GROUND RODS SHALL CONFORM TO THE REQUIREMENTS OF THE U.I. SPEC. NO. 467 (ANSI C-33.8-1972). MINIMUM SIZE SHALL BE 5/8 INCH DIAMETER BY EIGHT (8') FT. UNLESS OTHERWISE INDICATED. LENGTH MANUFACTURED BY ERICO, BLACKBURN PRODUCTS, ANIXTER, OR GAVIN.

WIRING DEVICES

ALL DEVICES SHALL BE COMMERCIAL SPECIFICATION GRADE, U.L. LISTED, SELF-GROUNDING, GROUND LUG, SIDE/BACK WIRED. COLOR SHALL BE SELECTED BY ARCHITECT OR OWNER UNLESS OTHERWISE INDICATED. MANUFACTURED BY HUBBELL, LEVITON, OR PASS & SEYMOUR.

ALL DEVICES SHALL BE HOSPITAL GRADE, U.L. LISTED, SELF-GROUNDING, GROUND LUG, BACK AND SIDE WIRED. COLOR SHALL BE SELECTED BY ARCHITECT OR OWNER UNLESS OTHERWISE INDICATED. MANUFACTURED BY HUBBELL, LEVITON, OR PASS & SEYMOUR.

DEVICES COLOR SHALL BE SELECTED BY ARCHITECT OR OWNER UNLESS OTHERWISE INDICATED FOR NORMAL POWER CIRCUITS AND RED FOR EMERGENCY POWER CIRCUITS.

RECEPTACLES THAT HAVE A POWER FEED THRU (FEED IN - FEED OUT) ARRANGEMENT SHALL BE PIGTAILED. FEED THRU FEATURE ON DUPLEX RECEPTACLES USE IS NOT ACCEPTABLE.

RECEPTACLES: 20A 125V	<u>HUBBELL</u>	<u>LEVITON</u>	PASS & SEYMOUR
DUPLEX RECEPTACLE	BR20	5362	CRB5362
DUPLEX RECEPTACLE - GFCI	GF20 AL	7899	2095

ALL INSTALLED RECEPTACLES UNDER THIS PROJECT SHALL BE TESTED AND CERTIFIED AS PER JOINT COMMISION CERTIFICATION REQUIREMENTS. THE CONTRACTOR SHALL UTILIZE HOSPITAL FURNISHED EQUIPMENT TO PERFORM REQUIRED TESTING. CONSULT WITH FACILITIES ENGINEER FOR REQUIREMENTS.

WALL PLATES FOR SWITCHES AND RECEPTACLES SHALL BE STAMPED STEEL FOR FURNISHED AREAS. MANUFACTURED BY HUBBELL.

RECEPTACLES LOCATED IN WET LOCATIONS SHALL BE INSTALLED WITH AN OUTLET ENCLOSURE CLEARLY MARKED 'SUITABLE FOR WET LOCATIONS WHILE IN USE'. THERE MUST BE A GASKET BETWEEN THE COVER AND THE BASE TO ASSURE A PROPER SEAL. THE ENCLOSURE MUST EMPLOY STAINLESS STEEL MOUNTING HARDWARE AND BE CONSTRUCTED OF IMPACT RESISTANT POLYCARBONATE. THE OUTLET ENCLOSURE SHALL BE U.L. LISTED. MANUFACTURED BY HUBBELL, OR APPROVED EQUAL.

EMERGENCY LIGHTING AND EXIT LIGHTING SYSTEM

FURNISH AND INSTALL WHERE SHOWN ON PLANS, A SYSTEM OF UNITS SO ARRANGED THAT IN CASE OF POWER FAILURE, THESE LIGHTS WILL AUTOMATICALLY ILLUMINATE THE EGREES AND EXIT AREAS. UNIT SHALL BE U.L. LISTED AND SHALL MEET THE REQUIREMENTS OF THE N.E.C., N.F.P.A. 101 LIFE SAFETY CODE. CONNECT EMERGENCY LIGHTING AND EXIT SIGNAGE TO LOCAL LIGHTING BRANCH CIRCUIT AHEAD OF SWITCHING DEVICE (CONSTANT POWER).

POWER AND CONTROL WIRING

FURNISH AND INSTALL ALL POWER WIRING, CONTROL WIRING (120VAC), CONDUIT AND FITTINGS FOR ALL PLUMBING, HEATING AND VENTILATING AND AIR CONDITIONING EQUIPMENT AND FINAL CONNECTIONS. MANUAL MOTOR STARTERS SHALL BE FURNISHED, INSTALLED AND WIRED BY THE ELECTRICAL CONTRACTOR, EVERY MOTOR SHALL BE PROVIDED WITH RUNNING OVERLOAD PROTECTION. UPON COMPLETION OF WORK, CHECK OUT EACH ITEM. ITEMS TO BE CHECKED ARE VOLTAGE, ROTATION AND OVERLOAD PROTECTION.

PART 3 - EXECUTION

INSTALLATION

ALL WORK, MATERIALS AND MANNER OF INSTALLING SAME SHALL BE IN STRICT ACCORDANCE WITH THE LATEST REQUIREMENTS OF THE NATIONAL ELECTRIC CODE.

ALL CONDUIT AND WIRING SHALL BE INSTALLED CONCEALED UNLESS OTHERWISE NOTED.

WIRING IN UNFINISHED AREAS SHALL BE INSTALLED EXPOSED USING EMT OR RGS CONDUIT.

WIRING IN FINISHED AREAS SHALL BE INSTALLED IN WIREMOLD RACEWAY

RACEWAYS

RACEWAYS, ENCLOSURES AND BOXES SHALL BE MECHANICALLY JOINED TO FORM A CONTINUOUS ELECTRICAL PATH.

THE CONTRACTOR SHALL PROVIDE APPROVED TYPE PULL BOXES AS REQUIRED.

MINIMUM SIZE CONDUIT SHALL BE 3/4" UNLESS OTHERWISE NOTED.

FURNISH NYLON PULL STRINGS IN ALL EMPTY CONDUIT RUNS.

FURNISH LOCKNUTS AND BUSHINGS FOR ALL CONDUIT TERMINATIONS IN ALL OUTLET BOXES, PANELS, PULL BOXES, CONDUIT STUBS, ETC.

RIGID GALVANIZED STEEL CONDUIT (RGS) SHALL BE USED FOR WIRING IN THE FOLLOWING LOCATIONS:

1. SERVICE POLE RISERS 2. BURIED UNDER GRADE FOR SERVICE ENTRANCE CONDUCTORS

- 3. WITHIN CONCRETE SLABS
- 4. EXPOSED TO MOISTURE AND MECHANICAL DAMAGE 5. EXTERIOR INSTALLATIONS

AS FOLLOWS:

1. INTERIOR PANEL FEEDERS 2. EXPOSED TO MOISTURE WITH WATER PROOF COMPRESSION FITTINGS 3. INTERIOR LIGHTING, RECEPTACLE AND POWER BRANCH CIRCUIT WIRING

RIGID POLYVINYL CHLORIDE (PVC) SHALL BE USED FOR WIRING IN THE FOLLOWING LOCATIONS: 1. BURIED UNDER GRADE FOR SERVICE ENTRANCE PRIMARY AND SECONDARY CONDUCTORS. 2. LIGHTING AND POWER BRANCH CIRCUIT WIRING BURIED UNDER GRADE 3. BELOW CONCRETE SLABS

WIRING

ETC.

MINIMUM SIZE WIRE SHALL BE #12 UNLESS OTHERWISE INDICATED. ALL WIRING SHALL BE COLOR CODED.

EXERCISE CAUTION IN PULLING CONDUCTORS INTO RACEWAYS SO AS NOT TO DAMAGE THE INSULATION. CABLE PULLING LUBRICANT SHALL BE USED TO ASSIST IN PULLING.

CIRCUITS SHALL BE SO CONNECTED TO THE PANELBOARDS THAT THE TOTAL LOAD IS DISTRIBUTED AS NEATLY AS POSSIBLE, EQUALLY BETWEEN EACH LINE AND NEUTRAL. 10% WILL BE CONSIDERED A REASONABLE AND ALLOWABLE UNBALANCE.

BRANCH CIRCUIT WIRING FOR HVAC EQUIPMENMT AND RECEPETACLES IN DRYWALL CONSTRUCTION AND ACCESSIBLE HUNG CEILING SPACE, HOME-RUN CIRCUIT SHALL BE INSTALLED WITHIN EMT RACEWAY, BRANCH WIRING WITHIN PARTITIONS SHALL BE IN METAL SHEATHED 'MC' TYPE CABLE. CABLE SHALL BE SUPPORTED FROM STRUCTURE 4" O.C. WITH APPROVED CABLE SUPPORTS. PROVIDE APPROPRIATE GROMMETS FOR HORIZONTAL RUNS IN METAL STUD PARTITIONS. CABLE SHALL NOT LAY ON CEILING STRUCTURE OR TILES. PROVIDE ANTI-SHORT BUSHINGS (RED HEAD) UNDER ARMOR JACKET AT TERMINATIONS.

COMMON NEUTRAL FOR MULTIPLE BRANCH CIRCUITS IS NOT ACCEPTABLE. PROVIDE SEPARATE NEUTRAL FOR EACH BRANCH CIRCUIT.

WIRING IN OUTLET BOXES, JUNCTION BOXES, CABINET PANELBOARDS OR EQUIPMENT SHALL HAVE A MINIMUM OF EIGHT (8") INCHES LENGTH LEADS FOR CONNECTING WIRING DEVICES TO MAKE UP CIRCUIT SPLICES.

PROVIDE FLEXIBLE METAL CONDUIT FOR DRY TYPE TRANSFORMER CONNECTIONS. LENGTH OF FLEXIBLE METAL CONDUIT DO NOT EXCEED THREE FEET (3').

<u>SPLICING</u>

SPLICING SHALL BE DONE WITH INSULATED OR NON-INSULATED CONNECTORS OF APPROPRIATE TYPES AND CURRENT-CARRYING CAPACITY. NON-INSUALTED CONNECTORS SHALL BE WRAPPED WITH INSULATING TAPE TO THE THICKNESS OF THE INSULATION OF THE CONDUCTORS BEING SPLICED. ELECTRICAL TAPE SHALL BE 3M OR SUPER 88 SCOTCH VINYL FLAME-RETARDANT, COLD AND WEATHER RESISTANT.

SPLICES, TAPS AND TERMINALS FOR CONDUCTORS #8 AWG OR LARGER SHALL BE MADE WITH U.L. LISTED BOLTED PRESSURE CONNECTORS OF BRONZE OR COPPER CONSTRUCTION, OF APPROPRIATE CURRENT CARRYING CAPACITY. EOUAL TO O/Z GEDENY, BURNDY OR BLACKBURN.

CONDUCTORS #8 AWG AND SMALLER SHALL HAVE A COLOR-CODED INSULATION.

CONDUCTORS #6 AWG AND LARGER SHALL BE IDENTIFIED WITH TAPES APPLIED NEAR THE ENDS OF THE CONDUCTORS.

FEEDERS AND BRANCH CIRCUIT CONDUCTORS SHALL BE IDENTIFIED FOR PHASE ROTATION.

PHASE A PHASE B PHASE C NEUTRAL GROUND

ALL FEEDERS, MAINS AND BRANCH CIRCUIT CONDUCTORS SHALL BE TAGGED AT BOTH ENDS WITH WIRE MARKERS IN ALL PANELS, MOTOR CONTROLS, JUNCTION BOXES, OUTLET BOXES AND DEVICE BOXES.

FURNISH AND INSTALL NAMEPLATES FOR ALL ELECTRICAL EQUIPMENT, IDENTIFYING ITEMS BY NAME, FUNCTION AND/OR CONTROL.

IDENTIFYING NAMEPLATES SHALL BE LAMINATED, PLASTIC TYPE, CONSISTING OF TWO BLACK PLASTIC SHEETS WITH ONE WHITE PLASTIC SHEET BONDED TO AND BETWEEN THE TWO OUTER BLACK SHEETS AND HAVING THE LETTERS ENGRAVED IN ONE BLACK TO THE DEPTH OF THE WHITE PLASTIC. FASTEN NAMEPLATES TO EQUIPMENT WITH SUITABLE ADHESIVES OR STAINLESS STEEL SCREWS.

ALL PANELS SHALL HAVE TYPEWRITTEN CIRCUIT DIRECTORIES IDENTIFYING ALL BRANCH CIRCUITS. PROVIDE ADDITIONAL COPY OF COMPLETE UPDATED PANEL DIRECTORY TO FACILITY ENGINEERING.

USE PLASTIC-COATED WIRE MARKERS OF THE SELF-ADHESIVE. WRAPAROUND TYPE WITH PERMANENT FACTORY-PRINTED NUMBER, LETTERS AND SYMBOLS.

WIRE MARKERS SHALL BE SECURELY ATTACHED AT BOTH ENDS, IDENTIFYING PANEL AND CIRCUIT BREAKER NUMBERS.

ALL CONDUCTORS SHALL BE PERMANENTLY TAGGED AT TIME OF INSTALLATION. LABELS SHALL BE EQUAL TO T&B, PANDUIT OR IDEAL.

ELECTRICAL METALLIC TUBING (EMT) SHALL BE USED FOR CONCEALED AND EXPOSED WIRING IN DRY LOCATIONS

4. EXPOSURE TO MOISTURE EXTERIOR FEEDERS SHALL BE INSTALLED IN RGS CONDUIT.

ALL CONDUIT SHALL BE INSTALLED IN PARALLEL AND PERPENDICULAR TO THE BUILDING LINES.

ALL CONDUIT SHALL BE SUPPORTED USING CADMIUM PLATED CONDUIT STRAPS AND HANGERS.

SEPARATE CONDUIT SYSTEMS SHALL BE INSTALLED FOR NORMAL AND EMERGENCY POWER.

PROVIDE WIRING TO ALL OUTLETS, EQUIPMENT, APPARATUS AND OTHER SPECIALTIES UNDER THIS DIVISION THAT WHICH FURNISHED OR PROVIDED UNDER OTHER DIVISIONS OR BY THE OWNER.

THE TERM 'WIRING' SHALL BE CONSIDERED TO BE COMPRISED OF THE CONDUIT, CONDUCTORS, CONNECTIONS,

ALL WIRING ON DRAWINGS IS SIZED FOR TYPE THWN/THHN COPPER CONDUCTORS.

CONDUCTOR WITHIN PANELBOARDS, JUNCTION BOXES, TROUGHS AND OTHER EQUIPMENT WHERE CONCENTRATIONS OF CONDUCTORS ARE ENCLOSED, SHALL BE NEATLY ARRANGED AND TIED WITH CABLE TIES.

INSTALL COPPER GREEN INSULATED GROUNDING CONDUCTOR IN ALL CONDUITS AND RACEWAYS.

SPLICES FOR CONDUCTORS, SIZES #10 AWG OR SMALLER SHALL BE MADE WITH U.L. LISTED SPRING-TYPE CONNECTORS OR APPROPRIATE CURRENT CARRYING CAPACITY.

CONDUCTOR IDENTIFICATION

	120/240V/1PH
	BLACK RED
	-
_	WHITE
1	GREEN

IDENTIFICATION

<u>GROUNDING</u>

ALL ELECTRICAL WORK SHALL BE GROUNDED AND BONDED IN FULL CONFORMANCE WITH THE LATEST APPROVED EDITION OF THE NATIONAL ELECTRICAL CODE AND LOCAL REQUIREMENTS.

ALL ELECTRICAL EQUIPMENT, PANELBOARD ENCLOSURES, MOTOR FRAMES, SAFETY SWITCHES, METAL ENCLOSURES, ELECTRICAL DEVICE CLOSURES AND ALL OTHER EQUIPMENT SHALL BE MADE TO FORM A CONTINUOUS CONDUCTING, GROUND PATH OF LOW IMPEDANCE FOR GROUND FAULT CIRCUITS AND OPERATION OF THE CIRCUIT PROTECTIVE DEVICES WITHIN EACH CIRCUIT.

PROVIDE GROUNDING CONDUCTOR IN ALL RACEWAYS.

GROUND CONNECTIONS WITH THE GROUNDING CONDUCTORS SHALL BE MADE AT EACH OUTLET BOX, MOTOR AND OTHER EQUIPMENT COMPONENTS BY MEANS OF A POSITIVELY SECURED GROUNDING CLAMP, SCREW OR CLIP. CONNECTIONS TO GROUNDING RODS, OTHER GROUNDING ELECTRODE CONDUCTORS SHALL BE MADE WITH CADWELL TYPE, EXOTHEMIC WELD PROCESS UNLESS OTHERWISE NOTED. CONNECTIONS TO PIPES SHALL BE MADE WITH APPROVED BRONZE OR BRASS CLAMPS.

BONDING SHALL BE PROVIDED TO ASSURE ELECTRICAL CONTINUITY AND THE CAPACITY TO SAFELY CONDUCT ANY FAULT CURRENT LIKELY TO BE IMPOSED.

ALL DEVICES (DISCONNECTS, RECEPTACLES, ETC.), SHALL BE GROUNDED TO CONDUIT SYSTEM WITH SIX (6") INCH SOLID COPPER #12 AWG INSULATED WIRE (GREEN) CONNECTED TO GROUND SCREW IN DEVICE AND FASTENED TO BACKBOX WITH 10-32x3/8" SLOTTED HEXAGON HEAD WASHER FACE GROUND WITH GREEN DYE FINISH.

SEISMIC RESTRAINT

SEISMIC LATERAL RESTRAINTS DESIGNED TO RESIST HORIZONTAL MOVEMENT IN ANY DIRECTIONS SHALL BE INSTALLED IN ALL SUSPENDED CONDUITS 2-1/2 INCHES IN DIAMETER OR GREATER. QUANTITY AND LOCATION OF THE LATERAL RESTRAINTS SHALL BE BASED ON THE CONDUIT SYSTEM LAYOUT AND IN GENERAL, SHALL BE INSTALLED AT CONDUIT BENDS, JUNCTION BOXES AND APPROXIMATELY EVERY 20 FEET ALONG CONDUIT RUNS. SEISMIC LATERAL RESTRAINTS ARE NOT REQUIRED FOR ANY PIPING SUSPENDED BY INDIVIDUAL HANGERS 12 INCHES OR LESS IN LENGTH FROM TOP OF PIPING TO THE BOTTOM OF THE SUPPORT FOR THE HANGER.

END OF ELECTRICAL SPECIFICATIONS

PROFESSIONAL SEAL	-
OWNER	
	rs
	C.
130 Lincoln Street 203 235 7146	Meriden, CT 06451 www.girlsincmeriden.org
ARCHITECT	
CHRISTOPHER WIL	LLIAMS ARCHITECTS, LLC
85 Willow Street 203 776 0184	New Haven, CT 06511 www.cwarchitectsllc.com
MEP ENGINEER	
	TIVE ENGINEERING ES, LLC
	SRATED ENGINEERING + DESIGN FIRM
33 North Plains Indus Wallingford, CT 0649	strial Rd 02
203 467 4370 iesllc.biz	
ENVIRONMENTAL CO	NSULTANT
_	
Enví	roMed
Cleaner environm Designed by: John	nent. Safer workplaces. LUBY
CT ASBESTOS PROJECT I CT LEAD PLANNER/PROJ	DESIGNER LICENSE #19 ECT DESIGNER LICENSE #986
ISSUED F	OR BIDDING
MARC	H 12, 2024
MARK DATE	DESCRIPTION
PROJECT NO: CWA PROJECT NO.: 2	2313
CHK'D BY:	
ELECTRICA	
	D D

E.205